# FINAL ENVIRONMENTAL IMPACT REPORT NO. 573 FOR THE CIVILIAN REUSE OF MCAS EL TORO AND THE AIRPORT SYSTEM MASTER PLAN FOR JOHN WAYNE AIRPORT AND PROPOSED ORANGE COUNTY INTERNATIONAL AIRPORT

SCH NO. 98101053

#### **EIR TEXT VOLUME 2B**

August 2001

Prepared for:

COUNTY OF ORANGE
MCAS El Toro Local Redevelopment Authority
10 Civic Center Plaza
Santa Ana, California 92701
(714) 834-3000

Contact: Bryan Speegle

Prepared by:

LSA ASSOCIATES P&D CONSULTANTS

#### **TABLE OF CONTENTS**

## FINAL ENVIRONMENTAL IMPACT REPORT NO. 573 FOR THE CIVILIAN REUSE OF MCAS EL TORO AND THE AIRPORT SYSTEM MASTER PLAN FOR THE JOHN WAYNE AIRPORT AND PROPOSED ORANGE COUNTY INTERNATIONAL AIRPORT

#### **VOLUME 1**

LIST	OF AC	CRONYN	AS	LOA-1
1.0	SUN	MMARY	AND INTRODUCTION	1-1
	1.1	Genera	l Introduction	1-1
		1.1.1	Proposed Project	1-1
	1.2	History	and Background	
		1.2.1	MCAS El Toro	
		1.2.2	John Wayne Airport	
	1.3	Relation	nship to EIR No. 563	
	1.4		of Controversy	
	1.5		o be Resolved	
	1.6		nmental Setting ("Existing Conditions")	
		1.6.1	No Project Alternative	
	1.7	Organiz	zation of the EIR	
	1.8	Referen	nced Documents and Availability of Studies and Reports	1-8
	1.9	Summa	rry of Significant Effects of the Proposed Project and Mitigation	
		Measur	res and Alternatives that will Reduce or Avoid that Effect	1-10
		1.9.1	Land Use	1-10
		1.9.2	General Plan Consistency	
		1.9.3	Transportation and Circulation	1-12
		1.9.4	Noise	1-13
		1.9.5	Air Quality	1-15
		1.9.6	Soils, Geology and Seismicity	
		1.9.7	Hydrology and Water Quality	1-18
		1.9.8	Public Services and Utilities	1-18
		1.9.9	Natural Resources and Energy	1-19
		1.9.10	Recreation	1-20
		1.9.11	Hazardous Wastes and Hazardous Materials Use	1-21
		1.9.12	Socioeconomics	
		1.9.13	Risk of Upset	
		1.9.14	Summary of Cumulative Effects of The Proposed Project And	
			Mitigation Measures That Would Lessen or Avoid That Effect.	1-24
		1.9.15	Level of Significance After Mitigation	

2.0	PROJECT HISTORY AND SUMMARY OF THE PROPOSED PROJECT 2-							
	2.1	Project	t Background	2-1				
		2.1.1	History of the El Toro Military Base Closure Process	2-1				
		2.1.2	MCAS EL Toro Community Reuse Plan (CRP) And Final EIR 5	63 . 2-2				
		2.1.3	Litigation Regarding Final EIR 563 and the Courts' Judgments					
		2.1.4	Transition From CRP to ASMP and Open Space Plan					
	2.2	Project	t Purpose and Need					
		2.2.1	Aviation Demand	2-7				
		2.2.2	Other Purpose and Need Factors	2-9				
	2.3	Project	t Objectives	2-11				
		2.3.1	General Project Objectives	2-11				
		2.3.2	Aviation Related Planning Objectives	2-12				
		2.3.3	Ability of The Proposed Project And Alternatives to Satisfy					
			The Project Objectives	2-13				
	2.4	Public	Coordination and Participation	2-14				
		2.4.1	Notice of Preparation					
		2.4.2	Public Scoping Meeting	2-15				
		2.4.3	Distribution of the Draft EIR	2-15				
		2.4.4	Distribution of Draft Supplemental Analysis	2-15				
	2.5	Relatio	onship to Federal Environmental Processes					
3.0	DES	DESCRIPTION OF THE PROPOSED PROJECT 3-						
210	3.1		action and Project Summary					
	***	3.1.1	Airport System Master Plan					
		3.1.2	Base Transition Plan					
	3.2		Location					
	3.3		Overview					
		3.3.1	OCX Proposed Aviation Facility Improvements					
		3.3.2	Nonaviation Revenue Support Uses					
		3.3.3	Base Transition Plan					
		3.3.4	Interim Aviation Uses					
		3.3.5	JWA Facility Improvements for the Proposed Project					
		3.3.6	Proposed Aviation Activity for the Two-Airport System					
	3.4		ary of CRP Project Addressed in EIR No. 563					
	J. 1	3.4.1	Comparison of Proposed Project with the Community Reuse Plan					
	3.5	Phasing						
	J.J	3.5.1	Phase 1 (2000-2005)					
		3.5.2	Phase 2 (2006-2010)					
		3.5.3	Phase 3 (2011-2015)					
		3.5.4	Phase 4 (2016-2020)					
	3.6		ry of Proposed Actions					
	5.0	3.6.1	County Implementing Actions and Discretionary Approvals					
	3.7		d Uses of the EIR					
	3.1	3.7.1	Approvals For Which The EIR Will Be Used					
		J.7.1	White and in the the the carrier committee	・・・シーロブ				

4.0		ENVIRONMENTAL SETTING, IMPACTS AND MITIGATION						
		MEASURES4-1						
		ntroduction/Organization and Content						
	4.1		Jse					
		4.1.1	Summary of Conclusions in Final EIR No. 563					
		4.1.2	Final EIR No. 563 Supplemental Analysis					
		4.1.3	Environmental Setting/Existing Conditions					
		4.1.4	Methodology					
		4.1.5	Thresholds of Significance					
		4.1.6	Project Impacts					
		4.1.7	Mitigation Measures					
		4.1.8	Level of Significance After Mitigation	4.1-30				
		4.1.9	No Project/No Activity Comparison	4.1-31				
	4.2	Genera	al Plan Consistency					
		4.2.1	Summary of Conclusions in EIR No. 563	4.2-1				
		4.2,2	Final EIR No. 563 Supplemental Analysis					
		4.2.3	Environmental Setting/Existing Conditions	4.2-2				
		4.2.4	Methodology					
		4.2.5	Thresholds of Significance					
		4.2.6	Project Impacts					
		4.2.7	Mitigation Measures					
		4.2.8	Level of Significance After Mitigation					
		4.2.9	No Project/No Activity Comparison					
	4.3	Transp	ortation And Circulation	4.3-1				
		4.3.1	Summary of Conclusions in Final EIR No. 563	4.3-1				
		4.3.2	Final EIR No. 563 Supplemental Analysis					
		4.3.3	Environmental Setting/Existing Conditions					
		4.3.4	Methodology	4.3-9				
		4.3.5	Thresholds of Significance					
		4.3.6	Project Impacts					
		4.3.7	Mitigation Measures					
		4.3.8	Level of Significance After Mitigation					
		4.3.9	No Project/No Activity Comparison					
	4.4	Noise		4.4-1				
		4.4.1	Summary of Conclusions in Final EIR No. 563	4.4-1				
		4.4.2	Final EIR No. 563 Supplemental Analysis					
		4.4.3	Environmental Setting/Existing Conditions					
		4.4.4	Methodology					
		4.4.5	Regulatory Setting					
		4.4.6	Project Impacts					
		4.4.7	Potential Mitigation Approaches					
		4.4.8	Mitigation Measures	4.4-174				

	4.4.9	Level of Significance After Mitigation	4.4-179
	4.4.10	No Project/No Activity Comparison	4.4-180
4.5	A i= 000	ality	151
4.3	4.5.1	Summary of Conclusions in EIR No. 563	
	4.5.2	Summary of Conclusions In Final EIR No. 563 Supplemental	
	4.3.4	Analysis	
	4.5.3	Summary of Conclusions In Draft Supplemental Analysis to	4.,2**4
	4.3.3	Draft EIR No. 573	151
	4.5.4	General Approach and Methodology	
	4.5.5	Existing Conditions	
	4.5.6	Existing and Future Background Concentrations	
	4.5.7	Methodology for Emissions Inventories	
	4.5.8	Methodology for Air Dispersion Modeling	
	4.5.9		
		Federal and State Regulatory Framework	
	4.5.10	Methods of Determining Significance	
	4.5.11	Project Impacts	
	4.5.12	Mitigation Measures	
	4.5.13	Environmental Impacts of Air Quality Mitigation Measures	
	4.5.14	Level of Significance After Mitigation	
	4.5.15	Consistency with AQMP	
	4.5.16	Health Risk Assessment of Toxic Air Contaminants	4.5-150
VOLUME 2	***************************************		
4.6		m And Topography	
	4.6.1	Summary of Conclusions in Final EIR No. 563	
	4.6.2	Final EIR No. 563 Supplemental Analysis	
	4.6.3	Environmental Setting/Existing Conditions	4.6-2
	4.6.4	Methodology	4.6-3
	4.6.5	Thresholds of Significance	4.6-3
	4.6.6	Project Impacts	4.6-4
	4.6.7	Mitigation Measures	4.6-8
	4.6.8	Level of Significance After Mitigation	4.6-8
	4.6.9	No Project/No Activity Comparison	4.6-9
4.7	Soils, G	eology And Seismicity	4.7-1
	4.7.1	Summary of Conclusions in Final EIR No. 563	
	4.7.2	Final EIR No. 563 Supplemental Analysis	
	4.7.3	Environmental Setting/Existing Conditions	
	4.7.4	Methodology	
	4.7.5	Thresholds of Significance	
	4.7.6	Project Impacts	
	4.7.7	Mitigation Measures	
Table of Contents		County of Orange Final EI	K NO. 373

	4.7.8	Level of Significance After Mitigation	4.7-18
	4.7.9	No Project/No Activity Comparison	
4.8	Hydrolo	ogy and Water Quality	4.8-
	4.8.1	Summary of Conclusions in Final EIR No. 563	
	4.8.2	Final EIR No. 563 Supplemental Analysis	
	4.8.3	Environmental Setting/Existing Conditions	
	4.8.4	Methodology Related to Hydrology and Water Quality	
	4.8.5	Thresholds of Significance	
	4.8.6	Project Impacts	
	4.8.7	Mitigation Measures	
	4.8.8	Level of Significance After Mitigation	
	4.8.9	No Project/No Activity Comparison	
4.9	Biologic	cal Resources	4.9-1
	4.9.1	Summary of Conclusions in EIR No. 563	4.9-1
	4.9.2	Final EIR No. 563 Supplemental Analysis	4.9-2
	4.9.3	Environmental Setting/Existing Conditions	4.9-2
	4.9.4	Methodology	4.9-23
	4.9.5	Thresholds of Significance	4.9-24
	4.9.6	Project Impacts	4.9-27
	4.9.7	Mitigation Measures	
	4.9.8	Level of Significance After Mitigation	
	4.9.9	No Project/No Activity Comparison	4.9-57
4.10	Public S	ervices And Utilities	
	4.10.1	Summary of Conclusions of Final EIR No. 563	4.10-1
	4.10.2	Final EIR No. 563 Supplemental Analysis	
	4.10.3	Environmental Setting/Existing Conditions	
	4.10.4	Methodology	
	4.10.5	Thresholds of Significance	4.10-18
	4.10.6	Proposed Transit Serving Improvements – MCAS El Toro	
	4.10.7	Project Impacts	4.10-20
	4.10.8	Phasing and Base Transition Plan	
	4.10.9	Mitigation Measures	
	4.10.10	Level of Significance After Mitigation	
	4.10.11	No Project/No Activity Comparison	4.10-54
4.11		Resources and Energy	
	4.11.1	Summary of Conclusions in EIR No. 563	4.11-1
	4.11.2	Effects Determined to be Mitigated to Below a Level of	
		Significance	
	4.11.3	Final EIR No. 563 Supplemental Analysis	
	4.11.4	Environmental Setting/Existing Conditions	4.11-3
	4.11.5	Methodology	4 11-9

	4.11.6	Thresholds of Significance	4.11-9
	4.11.7	Proposed Project Natural Resources and Energy	4.11-11
	4.11.8	Project Impacts	
	4.11.9	Mitigation Measures	
	4.11.10	Level of Significance After Mitigation	4.11-30
	4.11.11	No Project/No Activity Comparison	
4.12	Aestheti	cs, Light and Glare	4.12-1
	4.12.1	Summary of Conclusions in FEIR No. 563	4.12-1
	4.12.2	Final EIR No. 563 Supplemental Analysis	4.12-1
	4.12.3	Environmental Setting/Existing Conditions	4.12-2
	4.12.4	Methodology	
	4.12.5	Thresholds of Significance	
	4.12.6	Project Impacts	4.12-7
	4.12.7	Precautionary Mitigation Measures	
	4.12.8	Level of Significance After Mitigation	
	4.12.9	No Project/No Activity Comparison	
4.13	Cultural	Resources	4.13-1
	4.13.1	Summary of Conclusions in Final EIR No. 563	4.13-1
	4.13.2	Final EIR No. 563 Supplemental Analysis	
	4.13.3	Environmental Setting/Existing Conditions	
	4.13.4	Methodology	
	4.13.5	Thresholds of Significance	4.13-4
	4.13.6	Project Impacts	
	4.13.7	Mitigation Measures	
	4.13.8	Level of Significance After Mitigation	
	4.13.9	No Project/No Activity Comparison	
4.14	Recreation	on	4.14-1
	4.14.1	Summary of Conclusions In Final EIR No. 563	4.14-1
	4.14.2	Final EIR No. 563 Supplemental Analysis	
	4.14.3	Environmental Setting/Existing Conditions	
	4.14.4	Methodology	
	4.14.5	Thresholds of Significance	
	4.14.6	Proposed Recreational Amenities – MCAS El Toro	
	4.14.7	Project Impacts	
	4.14.8	Mitigation Measures	
	4.14.9	Level of Significance After Mitigation	
	4.14.10	No Project/No Activity Comparison	
		, , , , , , , , , , , , , , , , , , ,	

4.15	Public S	Safety	
	4.15.1	Summary of Conclusions in Final EIR No. 563	4.15-1
	4.15.2	Final EIR No. 563 Supplemental Analysis	4.15-2
	4.15.3	Environmental Setting/Existing Conditions	4.15-2
	4.15.4	Methodology	4.15-11
	4.15.5	Thresholds of Significance	4.15-12
	4.15.6	Proposed Project Features Relevant to Aviation Safety	Issues 4.15-15
	4.15.7	Project Impacts - Air Safety	4.15-16
	4.15.8	Mitigation Measures	4.15-29
	4.15.9	Level of Significance After Mitigation	4.15-31
	4.15.10	No Project/No Activity Comparison	4.15-31
4.16	Hazardo	ous Wastes and Hazardous Materials Use	
	4.16.1	Summary of Conclusions in Final EIR No. 563	4.16-1
	4.16.2	Final EIR No. 563 Supplemental Analysis	4.16-3
	4.16.3	Environmental Setting/Existing Conditions	4.16-3
	4.16.4	Methodology	4.16-22
	4.16.5	Threshold of Significance	4.16-23
	4.16.6	Project Impacts	4.16-24
	4.16.7	Mitigation Measures	4.16-52
	4.16.8	Level of Significance After Mitigation	4.16-57
	4.16.9	No Project/No Activity Comparison	4.16-57
4.17	Socioec	onomics	
	4.17.1	Summary of Conclusions In Final EIR No. 563	
	4.17.2	Final EIR No. 563 Supplemental Analysis	
	4.17.3	Environmental Setting/Existing Conditions	
	4.17.4	Methodology	
	4.17.5	Thresholds of Significance	
	4.17.6	Project Impacts	
	4.17.7	Mitigation Measures	
	4.17.8	Level of Significance After Mitigation	
	4.17.9	No Project/No Activity Comparison	4.17-22
4.18		Upset	
	4.18.1	Summary of Conclusions in Final EIR No. 563	
	4.18.2	Final EIR No. 563 Supplemental Analysis	
	4.18.3	Environmental Setting/Existing Conditions	
	4.18.4	Methodology	
	4.18.5	Thresholds of Significance	
	4.18.6	Proposed Project Land Uses	
	4.18.7	Project Impacts	4.18-14
	4.18.8	Mitigation Measures	
	4.18.9	Level of Significance After Mitigation	
	4.18.10	No Project/No Activity Comparison	4.18-34

#### **VOLUME 2B**

5.0	CUI	MULAT	IVE IMPACTS	5-1
	5.1	CEQA	Definition Of Cumulative Impacts	5-1
		5.1.1	Methodology	5-1
		5.1.2	Geographic Area	5-5
		5.1.3	Criteria for Selection	5-6
	5.2	Summa	ary Of Conclusions In Final EIR No. 563	5-6
		5.2.1	Land Use	5-7
		5.2.2	Transportation and Circulation	
		5.2.3	Noise	5-7
		5.2.4	Hydrology and Water Quality	5-8
		5.2.5	Public Service and Utilities	5-8
		5.2.6	Natural Resources and Energy	5-9
		5.2.7	Cultural Resources	5-9
		5.2.8	Socioeconomics	5-9
	5.3	Reason	ably Foreseeable Probable Future Projects	5-10
		5.3.1	County of Orange Jurisdiction Land Use Projects	5-13
		5.3.2	City Jurisdictions Land Use Projects	
		5.3.3	Transportation Land Use Projects	5-42
		5.3.4	Other Land Use Projects	5-45
		5.3.5	Non-Planned Land Use Projects	5-48
	5.4	Cumula	ative Impacts By Resource Area	5-49
		5.4.1	Potential Cumulative Impacts Related to Land Use	5-49
		5.4.2	Potential Cumulative Impacts Related to General Plan	
			Consistency	5-54
		5.4.3	Transportation and Circulation	5-56
		5.4.4	Potential Cumulative Impacts Related to Noise	5-62
		5.4.5	Potential Cumulative Impacts	5-64
		5.4.6	Topography	5-66
		5.4.7	Soils, Geology, and Seismicity	5-67
		5.4.8	Potential Cumulative Impacts Related to Hydrology and	
			Water Quality	5-67
		5.4.9	Potential Cumulative Impact Related to Biological Resources	5-68
		5.4.10	Potential Cumulative Impacts Related to Public Services and	
			Utilities	
		5.4.11	Natural Resources and Energy	5-75
		5.4.12	Potential Cumulative Impacts Related to Aesthetics, Light and	
			Glare	5-78
		5.4.13	Potential Cumulative Impacts Related to Cultural Resources	5-79
		5.4.14	Potential Cumulative Impacts Related to Recreation	5-80
		5.4.15	Potential Cumulative Impacts Related to Public Health and Safety	.5-81
		5.4.16	Potential Cumulative Impacts Related to Hazardous Materials and	İ
			Hazardous Waste	5-82

		5.4.17	Potential Cumulative impacts Related to Socioeconomics	5-83				
		5.4.18	Potential Cumulative Impacts Related to The Risk of Upset					
6.0	LO	NG-TERI	M IMPLICATIONS OF THE PROPOSED PROJECT	<b> 6-</b> ]				
	6.1							
		Propose	ed Project Should it be Implemented	<b>6-</b> 1				
	6.2	Growth	Inducing Impacts of the Proposed Project	6-2				
	6.3		Definition of Growth Inducing Impacts					
	6.4	Potentia	d Growth Inducing Impacts	6-3				
		6.4.1	Potential Growth Inducing Impacts of The Proposed Project	6-3				
7.0	SIG	NIFICAL	NT UNAVOIDABLE ADVERSE IMPACTS	7-1				
	7.1	Overvie	w	7-1				
	7.2	Signific	ant Unavoidable Adverse Impacts Under The Proposed Project	7-1				
	7.3	Signific	ant Unavoidable Adverse Impacts If Mitigation By Other Entities is	ļ				
		-	plemented					
8.0	<b>AL</b> T	[ERNAT]	IVES	8-1				
	8.1	Introduc	etion	8-1				
		8.1.1	Final EIR No. 563 Alternatives	8-1				
		8.1.2	ASMP Alternatives	8-2				
		8.1.3	Introduction to EIR Alternatives	8-3				
	8.2	No Proje	ect/No Activity Alternative (Alternative E): JWA – Status Quo					
		Aviation	n Roles; No Aviation Reuse at Former MCAS El Toro	<b>8-</b> 9				
		8.2.1	Aviation Uses					
		8.2.2	Nonaviation Revenue Support Uses	8-9				
		8.2.3	Attainment of Project Objectives	8-9				
		8.2.4	Environmental Impacts of the No Project/No Activity					
			Alternative	8-10				
		8.2.5	Conclusions	8-28				
	8.3	ETRPA	Nonaviation Plan Alternative	8-30				
		8.3.1	Aviation Uses	8-30				
		8.3.2	Nonaviation Revenue Support Uses	<b>8-</b> 30				
		8.3.3	Phasing: Build Out Over 20 Years	<b>8-3</b> 1				
		8.3.4	Attainment of Project Objectives	8-32				
		8.3.5	Environmental Impacts of the ETRPA Nonaviation Plan					
			Alternative	8-32				
		8.3.6	Feasibility	<b>8-8</b> 5				
		8.3.7	Conclusions	8-87				
	8.4	Alternati	ve A: JWA - Status Quo Aviation Roles; OCX - Full Domestic	8-88				
		8.4.1	Aviation Uses	8-88				
		8.4.2	Nonaviation Revenue Support Uses	8-88				
		8.4.3	Attainment of Project Objectives					
		811	Environmental Impacts of Alternative A	8-80				

	8.4.5	Feasibility	8-110
	8.4.6	Conclusions	8-110
8.5	Alterna	tive C: JWA – Short-Haul; OCX – Medium-Haul to Full	
	Interna	tional (Linked) Air Service	8-111
	8.5.1	Aviation Uses	8-111
	8.5.2	Nonaviation Revenue Support Uses	8-111
	8.5.3	Attainment of Project Objectives	8-112
	8.5.4	Environmental Impacts of Alternative C	8-112
	8.5.5	Feasibility	
	8.5.6	Conclusions	8-133
8.6	Alterna	tive F: JWA - Short- to Limited Long-Haul With Limited General	l
	Aviatio	on; No Aviation Reuse at Former MCAS El Toro	8-134
	8.6.1	Aviation Uses	
	8.6.2	Nonaviation Revenue Support Uses	8-135
	8.6.3	Attainment of Project Objectives	8-135
	8.6.4	Environmental Impacts of Alternative F	8-135
	8.6.5	Feasibility	
	8.6.6	Conclusions	8-152
8.7	Alterna	tive G: JWA - Limited International; No Aviation Reuse at	
	Former	MCAS El Toro	8-153
	8.7.1	Aviation Uses	8-153
	8.7.2	Nonaviation Revenue Support Uses	8-154
	8.7.3	Attainment of Project Objectives	8-154
	8.7.4	Environmental Impacts of Alternative G	8-154
	8.7.5	Feasibility	8-173
	8.7.6	Conclusions	8-174
8.8	Alternat	tive J: JWA - Status Quo Aviation Roles; OCX Full International	
	With W	idely-Spaced Runways	
	8.8.1	Aviation Uses	8-175
	8.8.2	Nonaviation Revenue Support Uses	
	8.8.3	Attainment of Project Objectives	
	8.8.4	Environmental Impacts of Alternative J	8-176
	8.8.5	Feasibility	8-178
	8.8.6	Conclusions	
8.9	Alternat	ive OCX Airport Runway Layout (Wildlands Ranch Alternative).	8-179
	8.9.1	Aviation Uses	8-179
	8.9.2	OCX Facility Improvements for the Wildlands Ranch	
		Alternative (October 5, 1999, Submittal)	
	8.9.3	Nonaviation Revenue Support Uses	
	8.9.4	Environmental Impacts of Wildlands Ranch Alternative	
	8.9.5	Conclusions	
8.10	Land Us	se Alternatives at Former MCAS El Toro	
	8.10.1	Nonaviation Revenue Support Area Alternative 1	8-199
	8.10.2	Environmental Impacts of Nonaviation Revenue Support Area	
		Alternative 1	8-201

		8.10.3	Nonaviation Revenue Support Area Alternative 2	8-205
		8.10.4	Environmental Impacts of Nonaviation Revenue Support Area	
			Alternative 2	8-207
	8.11	Alterna	tive K: JWA - Status Quo Aviation Roles; Alternative Airport Site-	•
		Full D	omestic to Full International; No Aviation Reuse at MCAS El Toro.	8-212
		8.11.1	CEQA Requirements for Alternative Sites	8-212
		8.11.2	Previous Studies of Alternative Airport Sites	8-212
		8.11.3	Alternative Sites Evaluated for EIR No. 573	8-214
	8.12	Enviro	nmentally Superior Alternative	8-215
		8.12.1	Introduction	8-215
		8.12.2	No Project/No Activity Impacts Summary	8-215
		8.12.3	Environmentally Superior Alternative	
	8.13	Compa	rison of the Environmental Impacts of the Project Alternatives	8-217
			tives Considered but Rejected	
		8.14.1	One-Airport Scenarios Not Carried Forward	<b>8-2</b> 19
		8.14.2	Unlinked Two-Airport Scenarios Not Carried Forward	<b>8-</b> 221
		8.14.3	Linked Two-Airport Scenarios Not Carried Forward	8-225
9.0	INV	ENTOR	Y OF MITIGATION MEASURES	9-1
	9.1		ction	
		9.1.1	Standard Conditions of Approval	
	9.2	Mitigat	ion Related to Land Use	
		9.2.1	Final EIR No. 563 Mitigation Measures	
		9.2.2	Additional Precautionary Mitigation Measures	
	9.3	Mitigat	ion Related to General Plan Consistency	
		9.3.1	Final EIR No. 563 Mitigation Measures	9-3
		9.3.2	Additional Mitigation Measures	
	9.4	Mitigati	ion Related To Transportation and Circulation	
		9.4.1	Final EIR No. 563 and EIR No. 563 Supplemental Analysis	
			Mitigation Measures	9-4
		9.4.2	DEIR No. 573 Additional Mitigation Measures	
	9.5	Mitigati	ion Related To Noise	9-11
		9.5.1	Final EIR No. 563 Mitigation Measures	9-11
		9.5.2	Proposed Mitigation Measures	
	9.6	Mitigati	on Related to Air Quality	
		9.6.1	Final EIR No. 563 Mitigation Measures	9-13
		9.6.2	Mitigation Measures Updating Final EIR No. 563 Mitigation	9-15
		9.6.3	Additional Mitigation Measures	
	9.7	Mitigati	on Related To Soils, Geology And Seismicity	9-26
		9.7.1	Final EIR No. 563 Mitigation Measures	
	9.8	Mitigati	on Related To Hydrology And Water Quality	
		9.8.1	Mitigation Measures Identified in Final EIR No. 563	
		9.8.2	Mitigation Measures Identified in Final EIR No. 563, Final	-
			Sunnlemental Analysis	0_28

	9.9	Mitigat	ion Related To Biological Resources	9-3
		9.9.1	Final EIR No. 563 Mitigation Measures	9-31
		9.9.2	Additional Mitigation Measures	9-31
	9.10	Mitigat	ion Related To Public Services And Utilities	9-35
		9.10.1	Final EIR No. 563 Mitigation Measures	9-35
		9.10.2	Additional Mitigation Measures	9-36
	9.11	Mitigat	ion Related To Natural Resources And Energy	9-37
		9.11.1	Final EIR No. 563 Mitigation Measures	9-37
		9.11.2	Final EIR No. 563 Supplemental Analysis Mitigation Measures.	9-37
	9.12	Mitigati	ion Related To Aesthetics, Light And Glare	9-38
		9.12.1	Final EIR No. 563 Mitigation Measures	9-38
	9.13	Mitigati	ion Related To Cultural Resources	9-39
		9.13.1	Final EIR No. 563 Mitigation Measures	9-39
	9.14	Mitigati	ion Related To Recreation	
		9.14.1	Final EIR No. 563 Mitigation Measures	9 <b>-4</b> 0
		9.14.2	Additional Mitigation Measures	9-40
	9.15	•	ion Related To Public Safety	
	9.16	Mitigati	ion Related To Hazardous Wastes and Hazardous Materials Use	9-41
		9.16.1	Final EIR No. 563 Mitigation Measures	9-41
		9.16.2	Additional Mitigation Measures	
	9.17	Mitigati	on Related To Socioeconomics	
		9.17.1	Final EIR No. 563 Mitigation Measures	
	9.18		on Related To Risk Of Upset	
		9.18.1	Additional Mitigation Measures	
	9.19	_	on Related To Cumulative Impacts	
	9.20	Standard	d Conditions	9-46
10.0	ENV	TRONM	ENTAL EFFECTS FOUND NOT TO BE	
	SIG	NIFICAN	T	10-1
11.0	SUP	PLEME	NTAL PHASING ANALYSIS	11-1
	11.1	Introduc	rtion	11-1
	11.2	Descript	tion of The Project at Phase 2	11-2
		11.2.1	Proposed Project at Phase 2 - OCX Aviation Facility Improvement	nts11-5
		11.2.2	Nonaviation Revenue Support Uses at Phase 2	11-9
		11.2.3	Base Transition Plan at Phase 2	. 11-10
		11.2.4	Interim Aviation Uses at Phase 2	. 11-10
		11.2.5	JWA Facility Improvements for the Proposed Project at Phase 2	. 11-10
	11.3	Impacts	Analysis	. 11-29
		11.3.1	Land Use	11-29
		11.3.2	General Plan Consistency	11-30
		11.3.3	Transportation and Circulation	11-30
		11.3.4	Noise	11-34
		11.3.5	Air Quality	11-46
		11.3.6	Landform and Topography	11-55

	11.3.7	Soils, Geology and Seismicity	11-56
	11.3.8	Hydrology and Water Quality	11-57
	11.3.9	Biological Resources	11-57
	11.3.10	Public Services and Utilities	11-58
	11.3.11	Natural Resources and Energy	11-60
	11.3.12	Aesthetics, Light and Glare	11-63
	11.3.13		
	11.3.14	Recreation	11-64
	11.3.15	Public Safety	11-66
	11.3.16	Hazardous Wastes and Hazardous Materials Use	11-68
	11.3.17	Socioeconomics	11-69
	11.3.18	Level of Significance After Mitigation	
	11.3.19	Risk Of Upset	11-72
12.0	PERSONS A	ND ORGANIZATIONS CONSULTED	12-1
13.0	LIST OF PRI	EPARERS AND CONTRIBUTORS	13-1
	13.1 County of	of Orange	13-1
	13.2 Consulta	ents	13-2
14.0	REFERENCE	ES	14-1

#### **VOLUME 3: GRAPHICS**

#### **APPENDICES**

- A Notice of Preparation and Initial Study
- B Comments on the Notice of Preparation
- C Technical Reports: Land Use Compatibility and General Plan Consistency
- D Technical Report: Traffic Analysis
- E Technical Report: Noise Analysis
- F Technical Report: Air Quality Analysis
- G Technical Report: Hydrology
- H Technical Report: Biological Resources
- I Technical Report: Hazardous Materials
- J Alternative Project Sites
- K Consistency with SCAG Policies
- L County of Orange Standard Conditions of Approval (1998)
- M Technical Report: Public Safety
- N Air Quality Technical Analysis
- O Supplemental Traffic and Circulation Technical Analysis

#### LIST OF TABLES

Table 1-1 Project Related Documents	1-9
Table 3-1 Proposed Project – Uses/Acreages By Planning Area	3-4
Table 3-2 Facilities at MCAS El Toro Proposed for Interim Nonaviation Reuse	3-28
Table 3-3 Major Buildings at MCAS El Toro Considered for Potential Aviation Reuse	3-30
Table 3-4 Summary of Facility Requirements for Proposed Project at JWA, 2020	
Table 3-5 Requirements for General Aviation at JWA, 2005 to 2020 [a]	3-33
Table 3-6 Summary Of Aviation Activity Forecasts For The Proposed Project, 2005	3-35
To 2020	
Table 3-7 Comparison of the Community Reuse Plan and the Airport System Master Plan.	
Table 3-8 Comparison of Proposed Project and the 1996 Community Reuse Plan	
Table 3-9 Summary List of Improvements at OCX and JWA Recommended in the Airport System Master Plan to 2020 [a]	
Table 3-10 Demolition/Construction Schedule at OCX	
Table 3-11 Proposed Nonaviation Revenue Support Uses and Locations	
Table 3-11 Proposed Nonaviation Revenue Support Uses and Locations         Table 3-12 Proposed Agricultural Use Parcels at OCX, Phase 1	
Table 4.1-1 Proposed Agricultural Use	
	4.1-17
Fable 4.2-1 Summary: Orange County General Plan Consistency Topics Identified in         EIR No. 563	4.2-3
Table 4.3-1 Comparison of CRP and Proposed Project Trip Generation	4.3-1
Гable 4.3-2 Long-Range (Year 2020) Impacted Arterial Miles (EIR No. 563)	4.3-1
Table 4.3-3 Existing (1995) Impacted Arterial Miles (EIR No. 563 FSA)	4.3-2
Table 4.3-4 Existing Trip Generation Summary	. 4.3-5
Table 4.3-5 Deficient Locations Based on Existing Levels of Service	. 4.3-7
Table 4.3-6 Analyzed Scenarios	. 4.3-9
Table 4.3-7 Roadway Level of Service (LOS) Descriptions	4.3-12
Fable 4.3-8 Trip Generation Summary - Proposed Project at Build Out	4.3-19
Table 4.3-9 Proposed Project Access Improvements	4.3-20
Table 4.3-10 Existing Plus Proposed Project Impact Summary	4.3-26
Table 4.3-11 Build Out (2020) Proposed Project Impact Summary	4.3-28
Table 4.3-12 Trip Generation Summary - Base Transition Plan Pre-Phase 1 Land Uses	
Table 4.3-13 Trip Generation Summary - Proposed Project (Phase 1)	4.3-31
Table 4.3-14 Phase 1 (2005) Proposed Project Impact Summary	
Table 4.3-15 Trip Generation Summary - Proposed Project (Phase 2)	
Table 4.3-16 Phase 2 (2010) Proposed Project Impact Summary	
Table 4.3-17 Trip Generation Summary - Proposed Project (Phase 3)	
Table 4.3-18 Phase 3 (2015) Proposed Project Impact Summary	
Table 4.3-19 Proposed Project Access Improvements	
Table 4.4-1 Factors That Affect Individual Annovance to Noise	4 4-5

Table 4.4-2 Land Uses Within the 1981 AICUZ 65 dB CNEL Contour	4.4-15
Table 4.4-3 1998 MCAS El Toro Aircraft Operations	4.4-16
Table 4.4-4 Average Daily MCAS El Toro Aircraft Operations/Percent by Time of Day for Calendar Year 1998	4 4-16
Table 4.4-5 MCAS El Toro Aircraft Types by Percent of Daily Operations	*** 4.4-10
for Calendar Year 1998	4.4-17
Table 4.4-6 Average Daily Jet Operations by Flight Corridor at the MCAS El Toro in	1110
Calendar Year 1998	
Table 4.4-7 1998 MCAS El Toro Average Daily Runway Usage	
Table 4.4-8 Existing 1998 MCAS El Toro CNEL Area In Acres	4.4-21
Table 4.4-9 Land Use Summary for the 1998 Existing Condition and 1981 (AICUZ) 65 dB CNEL Contours	4.4-21
Table 4.4-10 Specific Receptor Locations and Existing Land Uses	
Table 4.4-11 CNEL at Sensitive Receptor Locations for Calendar Year 1998	
MCAS El Toro Military Operations	4.4-23
Table 4.4-12 SENEL for Military F/A-18 Operations	4.4-24
Table 4.4-13 SENEL for Various Military Aircraft Operations	
Table 4.4-14 1998 JWA Air Carrier Operations by Percent of Aircraft Type	
Table 4.4-15 JWA 1998 CNEL at Noise Monitoring Sites	
Table 4.4-16 Sample Energy Average SENEL by Noise Monitoring Sites for JWA for the Fourth Quarter of 1998	
Table 4.4-17 Time Above Values for Existing 1998 JWA Aircraft Operations in Average	
Minutes Per Day	
Table 4.4-18 Arterial Road Vehicle Mix Data (Traffic Distribution By Time of	
Day Shown as a Percent of Average Daily Traffic)	4.4-32
Table 4.4-19 Contour Distances for Existing Conditions on Orange County Roads	4.4-33
Table 4.4-20 County of Orange Land Use Compatibility Criteria	4.4-50
Table 4.4-21 County of Orange Land Use Compatibility Criteria Explanations and	
Definitions	4.4-51
Table 4.4-22 FHWA Noise Abatement Criteria for Highway Construction Projects	4.4-52
Table 4.4-23 Annual and Daily Year 2020 Operations by Time of Day at OCX	4.4-55
Table 4.4-24 Year 2020 Daily Operations by Runway at OCX	4.4-55
Table 4.4-25 Interim Year Operations At OCX Under the Proposed Project	4.4-55
Table 4.4-26 Summary of OCX Year 2020 Operations by Aircraft Type	4.4-57
Table 4.4-27 Interim Year Fleet Mix at OCX	4.4-58
Table 4.4-28 Year 2020 Daily Departures at OCX by Stage Length for the	
Proposed Project	4.4-62
Table 4.4-29 Land Use Comparison: 1998 Military and 2020 Proposed Project	4.4-66
Table 4.4-29a Land Use Comparison Table, Military, CRP and Alternative B Year 2005, 2010, 2015 and 2020	. 4.4-67

Table 4.4-30 Comparison of 1998 And 2020 CNEL At Sensitive Receptor Locations	
For The Proposed Project	4.4-69
Table 4.4-30a Comparison of CNEL, Alternative B Years 2005, 2010, 2015, and 2020	4.4-70
Table 4.4-31 Corridor Comparison of 1998 Military Operations and Noise and the	
	4.4-73
Table 4.4-32 Aggregation of Aircraft Operations into Aircraft	4.4-74
Table 4.4-33 Comparison of Existing Military Jet Average Daily Operations with	
Civilian Aviation Jet Operations by Flight Corridor	4.4-75
Table 4.4-34 Time Above Data at Sensitive Receptors for 1998 Military and 2020	
1 3	4.4-77
Table 4.4-34a Time Above Data for Year 2020 Alternative B at Sensitive Receptor	4 4 770
Locations Compared with Years 2005, 2010 and 2015, 24 Hour Exposure	4.4-/9
Table 4.4-34b Time Above Data for Year 2020 Alternative B at Sensitive Receptor	
Locations Compared With Years 2005, 2010 and 2015, Night Hours Only (10 p.m. – 7 a.m.)	4.4-80
Table 4.4-35 Time Above 65 dBA for Existing Military and the Proposed Project	7.7-00
Operations	4.4-81
Table 4.4-36 2020 Annual and Daily Operations by Time of Day at JWA	
Table 4.4-37 Summary of JWA Operations by Aircraft Type for the	
	4.4-83
Table 4.4-38 2020 Daily Operations by Runway for JWA	4.4-84
Table 4.4-39 2020 Aircraft Stage Lengths for JWA for the Proposed Project	
(Average Daily Departures)	4.4-84
Table 4.4-40 Aircraft Takeoff Profiles for JWA	4.4-85
Table 4.4-41 Land Use Comparison for 1998 and 2020 for JWA (Area in Square Miles)	4.4-86
Table 4.4-42 Comparison of 1998 and 2020 CNEL at JWA Sensitive Receptor Locations	
	4.4-87
Table 4.4-43 Time Above Data for JWA 1998 and 2020 Under the Proposed Project at	
Sensitive Receptor Locations	
Table 4.4-44 Road Noise Level Changes for Future Conditions	
Table 4.4-45 Road Noise Level Increases Greater than 1.5	
Table 4.4-46 Contour Distances for Future Conditions Under the Proposed Project 4.	
Table 4.4-47 Utilization of RW 34 Right Turn by Aircraft Type and Stage Length4.	4-150
Table 4.4-48 Comparison of Years 1998 and 2020 CNEL at Sensitive Receptor	
Locations for Mitigation Alternative 1, Night Preferential Runway System 4.	4-153
Table 4.4-49 Time Above Data for Year 2020 Alternative B at Sensitive Receptor	
Locations Compared With Night Preferential Runway Alternative 1,	A 15A
Night Hours Only (10 pm – 7 am)	7-1J4
Receptor Locations Compared With Night Preferential Runway Alternative	
	4-155

Table 4.4-51 Comparison of Years 1998 and 2020 CNEL at Sensitive Receptor
Locations for Mitigation Alternative 2, Night Curfew
Table 4.4-52 Time Above Data for Year 2020 Alternative B at Sensitive Receptor
Locations Compared With Night Curfew Alternative 2, Night Exposure
(10  pm - 7  am)
Table 4.4-53 Time Above Data for Year 2020 Alternative B at Sensitive Receptor
Locations Compared With Night Curfew Alternative 2, 24 Hour Exposure4.4-159
Table 4.4-54 Comparison of Years 1998 and 2020 CNEL at Sensitive Receptor
Locations for Mitigation Alternative 3, Night SENEL 86 Limit
Table 4.4-55 Time Above Data for Year 2020 Alternative B at Sensitive Receptor
Locations Compared With 86 SENEL Night Limit Alternative 2, Night Exposure
(10 pm – 7 am)
Table 4.4-56 Time Above Data for Year 2020 Alternative B at Sensitive Receptor
Locations Compared With 86 SENEL Night Limit Alternative 2,
24 Hour Exposure 4.4-163
Table 4.4-57 Comparison of Years 1998 and 2020 CNEL at Sensitive Receptor
Locations for Mitigation Alternative 4, Right Turn Runway 34 Departures 4.4-164
Table 4.4-58 Time Above Data for Year 2020 Alternative B at Sensitive Receptor
Locations Compared With Runway 34 North Right Turn Alternative 4,  Night Exposure (10 pm – 7 am)
Table 4.4-59 Time Above Data for Year 2020 Alternative B at Sensitive Receptor
Locations Compared With Runway 34 North Right Turn Alternative 4,
24 Hour Exposure
Table 4.4-60 Summary of Land Use Impacts, Alternative B Year 2020
Table 4.4-61 Proposed Project Hour by Operations Per Year 2020
Table 4.4-62 Impact of Curfew on Annual Commercial Operations at OCX in 2020 4.4-171
Table 4.5-1A Regionwide Emissions Inventory Existing Conditions Plus Project
(Pounds/Day Unless Noted) (Phase 2)
Table 4.5-1B Regionwide Emissions Inventory Existing Conditions Plus Project
(Pounds/Day Unless Noted) (Phase 4)
Table 4.5-1C Regionwide Emissions Inventory Phase 2 Proposed Project/No Project
(Pounds/Day Unless Noted)
Table 4.5-1D Regionwide Emissions Inventory Phase 4 Proposed Project/No Project
(Pounds/Day Unless Noted)
Table 4.5-2 Ambient Air Quality at North Orange County Air Monitoring Station
Table 4.5-3 Ambient Air Quality at Central Orange County Air Monitoring Station 4.5-15
Table 4.5-4 Ambient Air Quality at North Coastal Orange County Air Monitoring Station . 4.5-16
Table 4.5-5 Ambient Air Quality Saddleback Valley Air Monitoring Station
Table 4.5-6 1998 Aircraft Operations Emissions (Pounds/Day)
Table 4.5-7 1998 Ground Support Equipment Emissions (Pounds/Day)
Table 4.5-8 1998 Energy Consumption Emissions MCAS El Toro Site (Pounds/Day) 4.5-20
Table 4.5-9 1998 Energy Consumption Emissions John Wayne Airport (Pounds/Day) 4.5-20
14010 (15 > 1770 Entered Communication Dimension Point in alternation (1 communication 110-20

Table 4.5-10 Existing Airport Vehicular Trips Emissions (Pounds/Day)	. 4.5-22
Table 4.5-11 Summarized Air Pollutant Emissions for Existing Conditions (1998)	
(Pounds/Day Unless Noted)	. 4.5-22
Table 4.5-12 Existing and Future Background Concentrations	. 4.5-24
Table 4.5-13 National and California Ambient Air Quality Standards	. 4.5-34
Table 4.5-14 NAAQS and CAAQS Attainment Status - South Coast Air Basin	. 4.5-35
Table 4.5-15 Thresholds of Significance for Air Pollutants in the South Coast Air Basin	. 4.5-38
Table 4.5-16 Unmitigated Construction Traffic and Equipment Exhaust Emissions,	
Proposed Project	
Table 4.5-17 Unmitigated Fugitive Dust Emissions, Proposed Project	. 4.5-41
Table 4.5-18 Unmitigated PM10 Construction Concentrations in Peak Year	
(Phase 2, Year 3)	. 4.5-43
Table 4.5-19 Unmitigated NO2 Construction Concentrations in Peak Year (Phase 2, Year 2)	. 4.5-44
Table 4.5-20 Unmitigated CO Construction Concentrations in Peak Year	
(Phase 2, Year 2)	. 4.5-45
Table 4.5-21 Unmitigated Operational Emissions - Aircraft Operations (Pounds/Day)	. 4.5-47
Table 4.5-22 Unmitigated Operational Emissions - Ground Support Equipment (Pounds/Day)	. 4.5-48
Table 4.5-23 Unmitigated Operational Emissions - Energy Consumption (Pounds/Day)	-
Table 4.5-24 Unmitigated Operational Emissions - Fuel Storage and Dispensing	
(Pounds/Day)	. 4.5-51
Table 4.5-25 Unmitigated Operational Emissions - Airport Roadways (Pounds/Day)	. 4.5-53
Table 4.5-26 Unmitigated Operational Emissions - Airport Parking Lots (Pounds/Day)	. 4.5-54
Table 4.5-27 Unmitigated Operational Emissions – Surface Traffic (Pounds/Day)	4 5 55
* 1	. 4.5-55
Table 4.5-28 Unmitigated Operational Emissions - Surface Traffic (Pounds/Day)	1 5 56
Orange County International Emissions Surface Traffic (Bounds/Day)	4.3-30
Table 4.5-29 Unmitigated Operational Emissions - Surface Traffic (Pounds/Day)  Both Airports	4 5-56
Table 4.5-30 Total Unmitigated Operational Emission Inventory	7.5-50
(Pounds Per Day Unless Noted)	4.5-57
Table 4.5-31 Existing Conditions (1998) - Pollutant Concentrations - JWA (Worst Case	
Operations and Meteorology)	4.5-60
Table 4.5-32 Existing Conditions (1998) - Pollutant Concentrations - MCAS El Toro	
(Worst Case Operations and Meteorology)	4.5-61
Table 4.5-33 Phase 2 Proposed Project (Alternative B) - Pollutant Concentrations -	
JWA (Worst Case Operations and Meteorology)	4.5-62
Table 4.5-34 Phase 2 Proposed Project (Alternative B) - Pollutant Concentrations -	
OCX (Worst Case Operations and Meteorology)	4.5-63

Table 4.5-35 Phase 4 Proposed Project (Alternative B) - Pollutant Concentrations -	
JWA (Worst Case Operations and Meteorology)4.5-6	54
Table 4.5-36 Phase 4 Proposed Project (Alternative B) - Pollutant Concentrations -	
OCX (Worst Case Operations and Meteorology)4.5-6	55
Table 4.5-37 Phase 4 No Project (Alternative E) - Pollutant Concentrations -	
JWA (Worst Case Operations and Meteorology)4.5-6	56
Table 4.5-38 Existing Conditions (1998) Plus Proposed Project (Phase 2) - Pollutant	
Concentrations - JWA (Worst Case Operations and Meteorology)	58
Table 4.5-39 Existing Conditions (1998) Plus Proposed Project (Phase 2) - Pollutant	
Concentrations - OCX (Worst Case Operations and Meteorology)	59
Table 4.5-40 Existing Conditions (1998) Plus Proposed Project (Phase 4) - Pollutant	
Concentrations - JWA (Worst Case Operations and Meteorology)	70
Table 4.5-41 Existing Conditions (1998) Plus Proposed Project (Phase 4) - Pollutant	
Concentrations - OCX (Worst Case Operations and Meteorology)	1
Table 4.5-42 Existing One Hour CO Concentrations - Year 1998 No Project -	
Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections	
with the Highest Volume and Worst Level of Service (LOS)	4
Table 4.5-43 Existing Eight Hour CO Concentrations - Year 1998 No Project - Predicted Eight	
Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest	
Volume and Worst Level of Service (LOS)4.5-7	5
Table 4.5-44 Phase 1 No Project – Predicted One Hour Ambient Carbon Monoxide	
Concentration for Intersections with the Highest Volume and Worst	
Level of Service (LOS)4.5-7	6
Table 4.5-45 Phase 1 No Project – Predicted Eight Hour Ambient Carbon Monoxide	
Concentration for Intersections with the Highest Volume and Worst Level of	. ,
Service (LOS)	7
Table 4.5-46 Phase 1 Proposed Project – Predicted One Hour Ambient Carbon Monoxide	
Concentration for Intersections with the Highest Volume and Worst Level of	0
Service (LOS)	0
Table 4.5-47 Phase 1 Proposed Project- Predicted Eight Hour Ambient Carbon Monoxide	
Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)	a
Table 4.5-48 Phase 2 No Project – Predicted One Hour Ambient Carbon Monoxide	"
Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	n
Table 4.5-49 Phase 2 No Project – Predicted Eight Hour Ambient Carbon Monoxide	•
Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	1
Table 4.5-50 Phase 2 Proposed Project – Predicted One Hour Ambient Carbon Monoxide	-
Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	2

Table 4.5-51 Phase 2 Proposed Project – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-52 Phase 3 No Project – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-53 Phase 3 No Project – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-54 Phase 3 Proposed Project – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-55 Phase 3 Proposed Project – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-56 Phase 4 No Project – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-57 Phase 4 No Project – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-58 Phase 4 Proposed Project - Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-59 Phase 4 Proposed Project - Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-60 Year 1998 Proposed Project - Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)
Table 4.5-61 Year 1998 Proposed Project - Predicted Eight Hour Ambient Carbon  Monoxide Concentration for Intersections with the Highest Volume and  Worst Level of Service (LOS)
Table 4.5-61A Maximum CO Concentrations in PPM Due to JWA, OCX, and Off-Airport         Traffic
Table 4.5-61B MCAS El Toro ASMP Regional Vehicle/Transit Miles for No Project and Proposed Project Alternatives
Table 4.5-61C MCAS El Toro ASMP Regional Vehicle/Transit Miles in Phase 4
Table 4.5-62 Regional Vehicular Traffic Emissions (Tons/Day)
Table 4.5-63 Regional Vehicular Traffic Emissions (Pounds/Day)    4.5-99
Table 4.5-64A Regionwide Emissions Inventory Phase 2 Proposed Project/No Project  (Pounds/Day Unless Noted)  4 5-102

Table 4.5-64B Regionwide Emissions Inventory Phase 4 Proposed Project/No Project	
(Pounds/Day Unless Noted)	
Table 4.5-65 Significance of Unmitigated Air Quality Impacts in Phase 1	
Table 4.5-66 Significance of Unmitigated Air Quality Impacts in Phase 2	
Table 4.5-67 Significance of Unmitigated Air Quality Impacts in Phase 3	4.5-105
Table 4.5-68 Significance of Unmitigated Air Quality Impacts in Phase 4	4.5-106
Table 4.5-69 Potential Construction Mitigation Percentage Reductions	4.5-122
Table 4.5-70 Operation Mitigation Percentage Reductions	4.5-123
Table 4.5-71 Potential GSE/APU Emissions Reductions (lbs/day)	4.5-123
Table 4.5-72 Potential Emissions Reductions for Energy Consumption (lbs/day)	4.5-123
Table 4.5-73 Total Potential Project Emissions Reductions (Lbs/Day)	4.5-124
Table 4.5-74 Estimated Emission Reductions for Proposed Mitigation Measures –	
Phase 1	4.5-125
Table 4.5-75 Estimated Emission Reductions for Proposed Mitigation Measures –	
Phase 2	4.5-126
Table 4.5-76 Estimated Emission Reductions for Proposed Mitigation Measures –	
Phase 3	4.5-127
Table 4.5-77 Estimated Emission Reductions for Proposed Mitigation Measures –	
Phase 4	4.5-128
Table 4.5-78 Mitigated Construction Traffic and Equipment Exhaust Emissions,	
Proposed Project	
Table 4.5-79 Mitigated Fugitive Dust Emissions, Proposed Project	
Table 4.5-80 Mitigated Airport Operational Emissions Inventories (tons per year)	4.5-131
Table 4.5-81 Mitigated Operational Emissions Inventories for Traffic Emissions	
(tons per year)	4.5-132
Table 4.5-82 Total Mitigated Operational Emission Inventory (Pounds Per Day	4 5 100
Unless Noted)	4.5-133
Table 4.5-83 Mitigated Phase 2 Proposed Project (Alternative B) -	4 5 104
Pollutant Concentrations - JWA (Worst Case Operations and Meteorology)	. 4.5-134
Table 4.5-84 Mitigated Phase 2 Proposed Project - Pollutant Concentrations -	4 5 125
OCX (Worst Case Operations and Meteorology)	. 4.5-133
Table 4.5-85 Mitigated Phase 4 Proposed Project (Alternative B) - Pollutant  Concentrations - JWA (Worst Case Operations and Meteorology)	15 126
Table 4.5-86 Mitigated Phase 4 Proposed Project - Pollutant Concentrations -	. 4.3-130
OCX (Worst Case Operations and Meteorology)	A 5_137
Table 4.5-87 Existing Conditions (1998) Plus Mitigated Proposed Project (Phase 2)	.4.5-137
Pollutant Concentrations - JWA (Worst Case Operations and Meteorology)	4 5_138
Table 4.5-88 Existing Conditions (1998) Plus Mitigated Proposed Project (Phase 2) -	. 4.5-156
Pollutant Concentrations - OCX (Worst Case Operations and Meteorology)	4 5_130
Table 4.5-89 Existing Conditions (1998) Plus Mitigated Proposed Project (Phase 4) -	. T.J" X J J
Pollutant Concentrations - OCX (Worst Case Operations and Meteorology)	4 5-140

Table 4.5-90 Existing Conditions (1998) Plus Mitigated Proposed Project (Phase 4)	
Pollutant Concentrations - JWA (Worst Case Operations and Meteorology)	4.5-141
Table 4.5-91 Significance of Air Quality Impacts After Mitigation - Phase 1	4.5-143
Table 4.5-92 Significance of Air Quality Impacts After Mitigation - Phase 2	4.5-144
Table 4.5-93 Significance of Air Quality Impacts After Mitigation - Phase 3	4.5-145
Table 4.5-94 Significance of Air Quality Impacts After Mitigation - Phase 4	4.5-146
Table 4.5-95 Emissions Estimates for Existing Conditions (1998) at MCAS El Toro	
and JWA	4.5-154
Table 4.5-96 List of Toxic Air Contaminant Emission Locations at OCX in Phase 4	4.5-155
Table 4.5-97 TAC Emissions Under Project Conditions (Phase 2 and Phase 4) at OCX 4.5-156	
Table 4.5-98 List of Toxic Air Contaminant Emission Locations at JWA in 1998 and	
Phase 4	. 4.5-157
Table 4.5-99 TAC Emissions Under Project Conditions (Phase 2 and Phase 4) at JWA	. 4.5-158
Table 4.5-100 Worst-Plausible Excess Lifetime Cancer Risk for the MEIs	
(Risk per million) for El Toro and OCX	. 4.5-163
Table 4.5-101 Worst-Plausible Chronic Noncancer Hazard Indices for El Toro and	
OCX	. 4.5-164
Table 4.5-102 Worst-Plausible Acute Noncancer Hazard Indices for El Toro and	
OCX	. 4.5-164
Table 4.5-103 Worst-Plausible Excess Lifetime Cancer Risk for the MEIs	4 1
(Risk per million) for JWA	
Table 4.5-104 Worst-Plausible Chronic Noncancer Hazard Indices for JWA	
Table 4.5-105 Worst-Plausible Acute Noncancer Hazard Indices for JWA	. 4.5-166
Table 4.5-106 Central Tendency Excess Lifetime Cancer Risk for the MEIs	45 167
(Risk per million)1	. 4.3-107
(Risk per million) for Proposed Project	4 5_160
Table 4.5-108 Worst-Plausible Chronic Noncancer Hazard Indices for Proposed Project.	
Table 4.5-109 Worst-Plausible Acute Noncancer Hazard Indices for Proposed Project	
Table 4.5-110 Summary of Criteria Pollutant Impacts After Mitigation	
· · · · · · · · · · · · · · · · · · ·	. 4.3-1/4
Table 4.5-111 Analysis of Mitigation Measures for OCX: Worst-Plausible Excess  Lifetime Cancer Risk for the MEIs (Risk per million)	. 4.5-177
Table 4.5-112 Analysis of Mitigation Measures for OCX: Worst-Plausible Chronic	
Noncancer Hazard Indices for the MEIs	. 4.5-177
Table 4.5-113 Analysis of Mitigation Measures for OCX: Worst-Plausible Acute	
Noncancer Hazard Indices for the MEIs	. 4.5-177
Table 4.5-114 Analysis of Mitigation Measures for JWA: Worst-Plausible Excess	
Lifetime Cancer Risk for the MEIs (Risk per million)	. 4.5-178
Table 4.5-115 Analysis of Mitigation Measures for JWA: Worst-Plausible Chronic	
Noncancer Hazard Indices for the MEIs	4.5-178

Table 4.5-116 Analysis of Mitigation Measures for: Worst-Plausible Acute	
Noncancer Hazard Indices for the MEIs	4.5-179
Table 4.7-1 Soil Associations on the MCAS El Toro Site	4.7-4
Table 4.7-2 Faults and Seismicity Characteristics in the MCAS El Toro Site Area	4.7-7
Table 4.7-3 Soil Associations on the JWA Site	4.7-9
Table 4.7-4 Faults and Seismicity Characteristics in the JWA Area	4.7-11
Table 4.8-1 Summary of 100 Year Flows at MCAS El Toro Under Existing Conditions.	4.8-4
Table 4.8-2 Summary of 100 Year Flows at JWA Under Existing Conditions	4.8-13
Table 4.8-3 Summary of 100 Year Flows and Related Capacity Under the Proposed	
Project With Existing Drainage Facilities	4.8-18
Table 4.8-4 Summary of 100 Year Flows and Related Capacity Under the Proposed	
Project With Project Drainage Facilities	
Table 4.8-5 Proposed Project Flow Capacity Compared to Existing Flow Capacity	
Table 4.8-6 NPDES Permit Runoff Concentrations for MCAS El Toro Site	4.8-22
Table 4.8-7 Percent Change from Existing Conditions of Outflow and Impervious  Drainage Area For Proposed Project	4.8-23
Table 4.8-8 Bee Canyon/Agua Chinon Estimated Outfall Point Concentrations for	
Proposed Project	4.8-24
Table 4.9-1 Plant Communities in the Habitat Reserve	4.9-9
Table 4.9-2 Special Interest Plant Species on the MCAS El Toro Site	4.9-14
Table 4.9-3 Special Interest Wildlife Species on the MCAS El Toro Site, Habitat Reserve	e,
JWA, and Upper Newport Bay Ecological Reserve	4.9-15
Table 4.9-4 Wildlife Species Known, Expected or With Limited Potential to Occur in the	
Habitat Reserve	
Table 4.9-5 Noise Exposure for Biological Resource Receptor Locations	
Table 4.10-1 OCTA Bus Transit Boardings, 1994-1998	
Table 4.10-2 Existing Daily Bus Transit Trips	4.10-8
Table 4.10-3 Existing Metrolink Boardings/Alightings at Irvine Transportation Center, 1998	4.10-8
Table 4.10-4 Fire Stations and Their Facilities at and in the Vicinity of JWA	
Table 4.10-5 Average Daily Transit Trips at JWA and the El Toro Site Under Project	
Build Out (2020)	. 4.10-23
Table 4.10-6 Average Daily Transit Trips at JWA and El Toro Site Under Project	4 10 24
Phase 1 (2005)	. 4.10-24
Phase 2 (2010)	. 4.10-24
Table 4.10-8 Average Daily Transit Trips at JWA and the El Toro Site Under Project Phase 3 (2015)	4 10-25
Table 4.10-9 Proposed Project Solid Waste Generation	
Table 4.10-10 Estimated Daily Demand for Electricity for the Nonaviation Revenue	V-20
Support Uses	. 4.10-33

Table 4.10-11 Estimated Daily Demand for Natural Gas for Nonaviation Revenue	
Support Uses	4.10-38
Table 4.10-12 Estimated Daily Demand for Domestic Water for the Nonaviation	
Revenue Support Uses	4.10-41
Table 4.10-13 Estimated Daily Demand for Recycled Water for the Nonaviation	
Revenue Support Uses	4.10-43
Table 4.10-14 Estimated Daily Waste Water Generated by the Nonaviation Revenue	
Support Uses	
Table 4.11-1 Estimated Daily Jet Fuel Use by Commercial Aircraft	
Table 4.11-2 Summary of Agricultural Land Impacts	
Table 4.11-3 Annual Electricity Demand (GWh)	4.11-14
Table 4.11-4 Annual Natural Gas Demand (Therms)	4.11-17
Table 4.11-5 Annual Automotive Gasoline Demand (Gallons)	4.11-18
Table 4.11-6 Annual Diesel Fuel Demand (Gallons)	4.11-19
Table 4.11-7 Construction-Related Energy Consumption – MCAS El Toro	4.11-20
Table 4.11-8 Estimated Jet Fuel Consumption for JWA and OCX Under the	
Proposed Project	4.11-21
Table 4.11-9 Projection of Jet Fuel Loaded onto Commercial Aircraft in the	
Southern California Air Service Area (1998 And 2020)	4.11-23
Table 4.11-10 Forecast of Passengers and Cargo in the Air Service Area	4.11-23
Table 4.15-1 Nationwide Existing Airport-Related Military Aircraft Accident Rates	
(1973-1993)	4.15-4
Table 4.15-2 NTSB Aircraft Accident Rates for the United States (1982-1998)	
(On and Off Airport Combined)	4.15-7
Table 4.15-3 Accident Summary for JWA (1964 – 1998)	
(On and Off Airport Combined)	
Table 4.15-4 JWA Summary of Airport Related Accidents by Phase 1980-1998	4.15-8
Table 4.15-5 Summary Comparison of Aircraft Accident Potential For MCAS El Toro	4 1 5 1 7
and JWA, Proposed Project, 2020 [a]	
Table 4.15-6 Proposed Project Forecast Mix of Large and Small Aircraft	
Table 4.16-1 Installation Restoration Program Sites at MCAS El Toro	4.16-8
Table 4.17-1 Population, Housing, and Employment Trends and Forecasts for	4 1 5 4
1990 to 2020	4.17-4
Γable 4.17-2 Demographic and Socioeconomic Characteristics in 1990 Project Site,	4 17 0
Surrounding Area and Orange County	4.1 /-8
Table 4.17-3 Comparison of Socioeconomic Characteristics of Final EIR No. 563,	4 17 11
CRP to the Proposed Project	
Table 4.17-4 Projected On-Site Employment Proposed Project	4.1 /-13
Table 4.17-5 Existing and Planned Residential Development in Planned Communities	11716
in the El Toro Study Area	

Table 4.18-2 Accident Severity Categories	. 4.18-11
Table 4.18-3 Proposed Project Estimated Daily Jet Fuel Tank Truck Supply Operations	
by Year	. 4.18-13
Table 4.18-4 Jet-A Physical and Chemical Properties	. 4.18-15
Table 4.18-5 Jet-A Bulk Fuel Storage Tank Release Scenarios – MCAS El Toro(a)	. 4.18-17
Table 4.18-6 Jet-A Tank Truck Highway Accident Release Scenarios – Single Incident(a)	. 4.18-18
Table 4.18-7 Proposed Project Jet Fuel Tank Truck Daily Accident Probabilities MCAS El Toro Site	. 4.18-20
Table 4.18-8 Proposed Project Bulk Jet Fuel Storage Tank Daily Accident Probabilities – MCAS El Toro Site	
Table 4.18-9 Estimated Daily Aircraft Induced Bulk Fuel Storage Facility Accident Probabilities – MCAS El Toro Site - Runway 7 Arrivals	
Table 4.18-10 Proposed Project Jet Fuel Tank Truck Daily Accident Probabilities –	
JWA Site	4.18-28
Table 4.18-11 Summary of Weighted JWA On-Site Fatal Accident Potential Under Proposed Project (2020)	
Table 4.18-12 Bulk Jet Fuel Storage Tank Daily Accident Probabilities – OCX with	
Pipeline Fuel Supply Option	4.18-33
Table 5.1-1 Regional Growth Projections 2000 to 2020	5-3
Table 5.1-2 Comparison of Year 2020 OCP-96M and Post-2020 Cumulative  Development Forecasts	5-4
Table 5.3-1 Reasonable Foreseeable Probable Future Projects County of Orange  Jurisdiction Land Use Projects	
Table 5.3-2 Reasonably Foreseeable Probable Future Projects (July 1999) City Jurisdictions Land Use Projects	
Table 5.3-3 Unspecified Projects in the City of Irvine	
Table 5.3-4 Reasonably Foreseeable Probable Future Projects Related Transportation  Land Use Projects	5-43
Table 5.3-5 Reasonable Foreseeable Probable Future Projects Other Related Land Use	····· J <del>-1</del> J
Project	5-46
Table 5.4-1 Cumulative Impacts With Cumulative and Proposed Projects	
Table 5.4-2 Post-2020 Non-Committed, Planned (MPAH) Improvements	
Table 5.4-3 Post-2020 Unplanned Mitigation Improvements	
Table 5.4-4 Summary of Off-Site Cumulative Coastal Sage Scrub Impacts (Acres by Phas	
Table 5.4-5 Summary of Off-Site Agricultural Soils Impacts (Acres by Phase)	-
Table 8.1-1 Summary Comparison of Principal Aviation Characteristics and	
Major Impacts for the CRP, Existing Conditions and Alternatives	8-6
Table 8.1-2 Summary of Aviation Activity at Orange County Airports Under Alternative Airport System Development Scenarios	8-7
Table 8.2-1 Residential and School Land Uses Within 65 CNEL	

Table 8.2-2 Trip Generation Summary - No Project/No Activity Alternative	. 8-12
Table 8.2-3A Regionwide Emissions Inventory Phase 2 Proposed Project/No Project (Pounds/Day Unless Noted)	. 8-15
Table 8.2-3B Regionwide Emissions Inventory Phase 4 Proposed Project/No Project (Pounds/Day Unless Noted)	. <b>8-</b> 16
Table 8.2-4 Phase 2 No Project/No Activity Pollutant Concentrations – JWA (Worst Case Operations and Meteorology)	. 8-17
Table 8.2-5 Phase 4 No Project/No Activity Pollutant Concentrations – JWA (Worst Case Operations and Meteorology)	. 8-18
Table 8.2-6 Phase 2 No Project – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)	. 8-19
Table 8.2-7 Phase 2 No Project – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of	. 8-20
Table 8.3-1 Trip Generation Summary - ETRPA Nonaviation Plan Alternative	
Table 8.3-2 Existing Plus ETRPA Nonaviation Plan Alternative Impact Summary	
Table 8.3-3 ETRPA Nonaviation Plan Alternative Mitigation Improvements	
Table 8.3-4 Traffic Noise Model Results Existing Conditions	
Table 8.3-5 Traffic Noise Model Results - Existing Plus ETRPA Nonaviation Plan - Existing Segments With 1.5 dB or Higher Traffic Noise Increase	8-52
Table 8.3-6 Traffic Noise Model Results - Existing Plus ETRPA Nonaviation Plan - New Segments	8-53
Table 8.3-7 Year 2020 No Project vs. Existing No Project Conditions - Existing Segments With 1.5 dB or Higher Traffic Noise Increase	8-54
Table 8.3-8 Traffic Noise Model Results - Year 2020 No Project Condition –  New Segments	8-58
Table 8.3-9 – Traffic Noise Model Results - Year 2020 With ETRPA Nonaviation Plan vs.	<b>8-</b> 59
Table 8.3-10 Traffic Noise Model Results Comparison - Number of Segments With Traffic Noise Increase Over Existing No Project Condition	8-64
Table 8.3-11 Traffic Noise Model Results Comparison – Number of Segments With Traffic Noise Increase – Year 2020 ETRPA Nonaviation Plan vs. Year 2020 No Project	8-65
Table 8.3-12 Phase 4 ETRPA Nonaviation Plan Alternative - Project Direct Air Pollutant	8-69
Table 8.3-13 Year 1998 Existing Conditions Plus ETRPA Nonaviation Plan - Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)	8-71
Fable 8.3-14 Year 1998 Existing Conditions Plus ETRPA Nonaviation Plan - Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)	<b>8-7</b> 2

Table 8.3-15 Phase 4 ETRPA Non-Aviation Plan – Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and	
Worst Level of Service (LOS)	
Table 8.3-16 Phase 4 ETRPA Non-Aviation Plan – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and	
Worst Level of Service (LOS)	
Table 8.4-1 Trip Generation Summary - Alternative A	8-91
Table 8.4-2 Summary Comparison of Traffic Impacts for Alternative A to  Existing Conditions and Existing Conditions Plus Project	8_Q7
Table 8.4-3 Land Use Comparison with Noise Contours for 1998 Military and Year 2020 Alternatives for El Toro	
Table 8.4-4 Land Use Comparison with Noise Contours for 1998 and Year 2020	
Alternatives for John Wayne Airport	8-96
Table 8.4-5 Project Direct Air Pollutant Emissions (Pounds/Day) - Phase 4 Alternative A.	8-98
Table 8.4-6 Regionwide Emissions Inventory Alternative A Phase 4	
(Pounds/Day Unless Noted)	8-99
Table 8.4-7 Phase 4 Alternative A – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of	
	. 8-101
Table 8.4-8 Phase 4 Alternative A - Predicted Eight Hour Ambient Carbon Monoxide	
Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)	<b>₽</b> _102
Table 8.5-1 Trip Generation Summary – Alternative C	
Table 8.5-2 Summary Comparison of Traffic Impacts for Alternative C to Existing	, 0-117
Conditions and Existing Conditions Plus Project	8-115
Table 8.5-3 Alternative C Impact Summary	
Table 8.5-4 Phase 4 Alternative C – Project Direct Air Pollutant Emissions (Pounds/Day)	
Table 8.5-5 Regionwide Emissions Inventory Alternative C Phase 4	
Table 8.5-6 Phase 4 Alternative C – Predicted One Hour Ambient Carbon Monoxide	
Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	8-123
Table 8.5-7 Phase 4 Alternative C – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	
Table 8.6-1 Trip Generation Summary - Alternative F	8-136
Table 8.6-2 Summary Comparison of Traffic Impacts for Alternative F to Existing	
Conditions and Existing Conditions Plus Project	
Table 8.6-3 Phase 4 Alternative F - Project Direct Air Pollutant Emissions (Pounds/Day)	8-141
Table 8.6-4 Regionwide Emissions Inventory Alternative F Phase 4	
(Pounds/Day Unless Noted)	8-142

Table 8.6-5 Phase 4 Alternative F – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of  Service (LOS)	8-144
Table 8.6-6 Phase 4 Alternative F – Predicted Eight Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	
Table 8.7-1 Trip Generation Summary - Alternative G	8-157
Table 8.7-2 Summary Comparison of Traffic Impacts for Alternative G to Existing  Conditions and Existing Conditions Plus Project	8-158
Table 8.7-3 2020 Alternative G - Project Direct Air Pollutant Emissions (pounds/day)	8-163
Table 8.7-4 Regionwide Emissions Inventory Alternative G Phase 4	
(Pounds/Day Unless Noted)	8-164
Table 8.7-5 Phase 4 Alternative G – Predicted One Hour Ambient Carbon Monoxide  Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	8-166
Table 8.7-6 Phase 4 Alternative G – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of	
Service (LOS)	8-167
Table 8.9-1 Summary of Arrivals by Type of Aircraft Impacted by Short "V"  Alternative Airport Runway Layout at OCX in 2020	8-190
Table 8.9-2 Summary of Impact Short "V" Alternative Airport Runway Layout on Projected Aircraft Arrivals at OCX in 2020	
Table 8.10-1 Nonaviation Revenue Support Area Alternative 1	
Table 8.10-2 Nonaviation Revenue Support Area Alternative 2	
Table 8.13-1 Comparison of Key Impacts of Alternatives Relative to the Proposed Project	8-218

#### LIST OF FIGURES

### All figures are bound separately in the Graphics Volume in the order listed below; therefore, no page numbers are included in the listing below.

- Figure 1-1 Planning Areas at MCAS El Toro
- Figure 2-1 1996 Community Reuse Plan for MCAS El Toro
- Figure 2-2 Air Service Area and Airports
- Figure 2-3 Distance of Domestic Markets from Orange County
- Figure 2-4 Air Passenger Demand Forecasts in ASA, 1970-2020
- Figure 2-5 Air Cargo Demand Forecasts in ASA, 1975-2020
- Figure 3-1 The Proposed Project at OCX Year 2020
- Figure 3-2 Proposed Land Uses at OCX Year 2020
- Figure 3-3 Proposed Land Uses at JWA
- Figure 3-4 Proposed Development Phases for OCX
- Figure 3-5 Project Location Map
- Figure 3-6 Regional and Local Access
- Figure 3-7 Proposed Terminal Area Development at OCX
- Figure 3-8 Concept of Proposed Terminal Plan
- Figure 3-9 Conceptual Cross-Section of Proposed Terminal and Roadway
- Figure 3-10 Proposed Terminal Roadways and Phasing at OCX
- Figure 3-11 Trabuco/ETC Interchange Option
- Figure 3-12 Composite Open Space Plan
- Figure 3-13 Proposed On-Site Regional Park-MCAS El Toro Site
- Figure 3-14 Proposed General Aviation Development on East Side of JWA
- Figure 3-15 Proposed General Aviation Development on West Side of JWA
- Figure 4.1-1 Proposed Project Surrounding Jurisdictions MCAS El Toro
- Figure 4.1-2 Proposed JWA Project Surrounding Jurisdictions
- Figure 4.1-3 Existing Land Uses in the Vicinity of the Proposed El Toro Project
- Figure 4.1-4 General Plan Land Uses in the Vicinity of the Proposed El Toro Project
- Figure 4.1-5 Existing Land Uses in the Vicinity of the Proposed JWA Project
- Figure 4.1-6 General Plan Land Uses in the Vicinity of the Proposed JWA Project
- Figure 4.2-1 County of Orange General Plan Designations MCAS El Toro
- Figure 4.2-2 City of Irvine General Plan Designations Planning Area 30
- Figure 4.2-3 City of Irvine Planning Areas Within Project Vicinity MCAS El Toro
- Figure 4.3-1 Traffic Analysis Study Area
- Figure 4.3-2 El Toro Circulation Plan
- Figure 4.4-1 Examples of Various Sound Levels
- Figure 4.4-2 Effect of Atmospheric Absorption on Sound Levels
- Figure 4.4-3 Single and Cumulative Noise Metric Definitions

```
Figure 4.4-4 Examples of Typical Outdoor CNEL Levels
```

Figure 4.4-5 Speech Interference and Noise Levels

Figure 4.4-6A Causes and Prevalence of All Awakenings

Figure 4.4-6B Sleep Interference and Noise Levels

Figure 4.4-7 Comparison of Logistic Fits to Original 161 Data Points of Schultz (1978) and USAF Analysis with 400 Points

Figure 4.4-8 1981 AICUZ Military Noise CNEL Contours

Figure 4.4-9 Historical Jet Operations at MCAS El Toro

Figure 4.4-10 Runway and Corridor Names

Figure 4.4-11 Flight Tracks for Military Operations for Calendar Year 1998

Figure 4.4-12 Calendar Year 1998 Military Noise Contours, CNEL 60, 65 and 70 dB

Figure 4.4-13 Specific Point Receptor Locations Used for the Detailed Analysis

Figure 4.4-14 SENEL 85 dB Contours for F/A 18 Operations

Figure 4.4-15 JWA Existing Flight Tracks

Figure 4.4-16 JWA 1998 Contours CNEL 60, 65, 70 and 75 dB

Figure 4.4-17 JWA Noise Monitoring Locations

Figure 4.4-18 SENEL 85 dB Contours for Various Aircraft Departures at JWA

Figure 4.4-19 SENEL 85 dB Contours for Arrivals at JWA

Figure 4.4-20 JWA 1985 Master Plan CNEL 60, 65, 70 and 75 dB

Figure 4.4-21 Federal Aviation Regulation Part 150 Land Use Guidelines

Figure 4.4-22 Proposed Project 2020 Operations by Time of Day

Figure 4.4-23 Proposed Project 2020 Operations by Runway

Figure 4.4-24 Proposed Project 2020 Operations by Aircraft Type

Figure 4.4-25 Typical Aircraft Departure Noise Levels

Figure 4.4-26 Typical Aircraft Arrival Noise Levels

Figure 4.4-27 El Toro Generalized Civilian Flight Tracks and Existing Buffer Zone

Figure 4.4-28 Typical JWA Track Dispersion Overlaid on OCX

Figure 4.4-29 Flight Tracks at OCX for the Proposed Project

Figure 4.4-30 Aircraft Altitude Profiles, Runways 34 Departures

Figure 4.4-31 Aircraft Altitude Profiles, Runways 07 Departures

Figure 4.4-32 Aircraft Altitude Profiles, Runways 34 Arrivals

Figure 4.4-33 El Toro Average Wind Speed and Direction by Quadrant

Figure 4.4-34 Proposed Project 2020 CNEL 60, 65 and 70 dB Contours

Figure 4.4-35 Proposed Project 2005 CNEL 60, 65 and 70 dB Contours

Figure 4.4-36 Proposed Project 2010 CNEL 60, 65 and 70 dB Contours

Figure 4.4-37 Proposed Project 2015 CNEL 60, 65 and 70 dB Contours

Figure 4.4-38 SENEL 86 dB Contours for Various Aircraft for Runway 34 Departures

Figure 4.4-39 SENEL 86 dB Contours for Various Aircraft for Runway 07 Departures

Figure 4.4-40 SENEL 86 dB Contours for Various Aircraft for Runway 34 Arrivals

- Figure 4.4-41 SENEL 86 dB Contours for Various Aircraft for Arrivals on Runway 16 and Departures on Runway 16
- Figure 4.4-42 Single Event Histograms and Bar Charts for 2020 for the Proposed Project
- Figure 4.4-43 JWA 2020 CNEL Contours 60, 65 and 70 dB Under the Proposed Project
- Figure 4.4-44 2020 No Project Alternative CNEL 60, 65 and 70 dB at JWA
- Figure 4.4-45 Proposed Project 2020 Combined Aircraft and Road Noise CNEL Contours at OCX
- Figure 4.4-46 Proposed Project 2020 Combined Aircraft and Road Noise CNEL Contours at JWA
- Figure 4.4-47 Noise Abatement Takeoff Procedures Example
- Figure 4.4-48 Mitigation Alternative 1: Night Preferential Runway System
- Figure 4.4-49 Mitigation Alternative 2: Full Curfew CNEL Contours
- Figure 4.4-50 Mitigation Alternative 3: Night 86 SENEL Limit
- Figure 4.4-51 Mitigation Alternative 4: Runway 34 Right Turn CNEL Contours and Flight Tracks
- Figure 4.4-52 Conceptual Noise Monitoring Locations at OCX
- Figure 4.5-1 Receptor Locations MCAS El Toro
- Figure 4.5-2 Receptor Locations OCX
- Figure 4.5-3 Receptor Locations JWA
- Figure 4.5-4 Unmitigated 24-Hour PM10 Dispersion in Peak Year
- Figure 4.5-5 Unmitigated Annual PM10 Dispersion in Peak Year
- Figure 4.5-6 Unmitigated 1-Hour NOX Dispersion in Peak Year
- Figure 4.5-7 Unmitigated Annual Hour NOX Dispersion in Peak Year
- Figure 4.5-8 Unmitigated 1-Hour CO Dispersion in Peak Year
- Figure 4.5-9 Unmitigated 8-Hour CO Dispersion in Peak Year
- Figure 4.5-10 EDMS Dispersion Receptor Locations MCAS El Toro
- Figure 4.5-11 EDMS Dispersion Receptor Locations OCX
- Figure 4.5-12 EDMS Dispersion Receptor Locations JWA
- Figure 4.5-13 CO Hot Spot Modeling Locations Existing Conditions
- Figure 4.5-14 CO Hot Spot Modeling Locations Existing Plus Proposed Project Conditions
- Figure 4.5-15 CO Hot Spot Modeling Locations Phase 4 No Project
- Figure 4.5-16 CO Hot Spot Modeling Locations Phase 4 Proposed Project
- Figure 4.5-17 Mitigated 24-Hour PM10 Dispersion in Peak Year
- Figure 4.5-18 Mitigated Annual PM10 Dispersion in Peak Year
- Figure 4.5-19 1-Hour NOX Dispersion in Peak Year
- Figure 4.5-20 Mitigated Annul NOX Dispersion in Peak Year
- Figure 4.5-21 Mitigated 1-Hour CO Dispersion in Peak Year
- Figure 4.5-22 Mitigated 8-Hour CO Dispersion in Peak Year
- Figure 4.5-23 MEI Receptors Existing Conditions

- Figure 4.5-24 MEI Receptors at Build Out
- Figure 4.5-25 Estimated Lifetime Excess Cancer Risk Within 3 Miles of Regional Airports Relative to No Project
- Figure 4.6-1 Topographical Analysis of MCAS El Toro
- Figure 4.6-2 Topographical Analysis of John Wayne Airport
- Figure 4.6-3 Rough Grading Plan
- Figure 4.7-1 Geologic Map of MCAS El Toro
- Figure 4.7-2 Expansivity Index of Soils at MCAS El Toro
- Figure 4.7-3 Generalized Depth to Shallow Groundwater Aquifer at MCAS El Toro Irvine Sub-Basin
- Figure 4.7-4 Earthquake Epicenter and Fault Map of Southern California
- Figure 4.7-5 Expansivity Index of Soils at JWA
- Figure 4.7-6 Generalized Depth to Shallow Groundwater at JWA
- Figure 4.8-1 Project Area Drainage Network
- Figure 4.8-2 Flood Plain Map of Existing Conditions at MCAS El Toro
- Figure 4.8-3 Existing Hydrology MCAS El Toro
- Figure 4.8-4 Water Quality Outfalls Under Existing Conditions at MCAS El Toro
- Figure 4.8-5 Project Area Groundwater Elevations
- Figure 4.8-6 Existing Hydrologic Conditions at JWA
- Figure 4.8-7 Flood Plain Map Existing Conditions John Wayne Airport
- Figure 4.8-8 Flood Plain Map Under the Proposed Project at MCAS El Toro
- Figure 4.8-9 Backbone Storm Drain System Under the Proposed Project at MCAS El Toro
- Figure 4.9-1 Existing Uses in the Federal Habitat Reserve
- Figure 4.9-2 Central and Coastal Subregion of the NCCP
- Figure 4.9-3 NCCP Vegetation Communities on the MCAS El Toro Site
- Figure 4.9-4 NCCP Vegetation Communities in the Federal Habitat Reserve Including Gnatcatcher and Cactus Wren Locations
- Figure 4.9-5 NCCP Vegetation Communities and Special Interest Plant Species at JWA and Vicinity
- Figure 4.9-6 Special Interest Wildlife Species at JWA and Vicinity
- Figure 4.9-7 Noise Receptor Locations for Biological Resources
- Figure 4.9-8 1981 AICUZ, Proposed Project Noise Exposure with Biological Resources/1998 Burn Area at MCAS El Toro
- Figure 4.9-9 1998 65 CNEL Noise Contour with Biological Resources at JWA
- Figure 4.9-10 Proposed Project 65 CNEL Noise Exposure with Biological Resources at JWA
- Figure 4.9-11 Drainage Channels and Jurisdictional Wetlands
- Figure 4.10-1 Existing Setting at MCAS El Toro Public Services
- Figure 4.10-2 Irvine and Saddleback Valley Unified School Districts
- Figure 4.10-3 Existing Transit Routes Serving Project Zones
- Figure 4.10-4 Transit Routes and Facilities

- Figure 4.10-5 Proposed Backbone Cable Television and Fiber Optic Facilities
- Figure 4.10-6 Proposed Backbone Communications Distribution Facilities
- Figure 4.10-7 Proposed Backbone Electrical Distribution Facilities
- Figure 4.10-8 Proposed Optional Pipeline Fuel Supply System For OCX
- Figure 4.10-9 Potential Bulk Fuel Storage Facility With Fuel Supplied by Truck
- Figure 4.10-10 Proposed Backbone Natural Gas Facilities
- Figure 4.10-11 Proposed Backbone Domestic Water Facilities
- Figure 4.10-12 Proposed Backbone Recycled Water Facilities
- Figure 4.10-13 Proposed Backbone Sanitary Sewers
- Figure 4.12-1 Visual Character Areas
- Figure 4.12-2 Viewshed Vantage Points
- Figure 4.12-3 Vantage Point 8 Visual Simulation MCAS El Toro Site
- Figure 4.12-4 Vantage Point 9 Visual Simulation MCAS El Toro Site
- Figure 4.12-5 Vantage Point 10 Visual Simulation MCAS El Toro Site
- Figure 4.12-6 Vantage Point 11 Visual Simulation MCAS El Toro Site
- Figure 4.14-1 Regional Recreational Facilities
- Figure 4.14-2 County Bikeways and Regional Riding and Hiking Trails
- Figure 4.14-3 City of Irvine Conservation and Open Space Map
- Figure 4.14-4 City of Irvine Trails Map
- Figure 4.14-5 City of Lake Forest Planned Bikeways Map
- Figure 4.14-6 City of Lake Forest Open Space Map
- Figure 4.14-7 Proposed On-Site Regional Park MCAS El Toro Site
- Figure 4.14-8 Proposed Golf Courses MCAS El Toro Site
- Figure 4.14-9 Project Aviation Noise Impacts on Recreational Facilities
- Figure 4.14-10 Project Aviation Noise Impacts on County Bikeways and Regional Riding and Hiking Trails
- Figure 4.15-1 Off-Site Accident Locations at MCAS El Toro (1965 to 1998)
- Figure 4.15-2 Aircraft Accident Rates at MCAS El Toro from 1966 to 1998
- Figure 4.15-3 Existing MCAS El Toro Accident Potential Zones
- Figure 4.15-4 Commercial Aircraft Accident Rates for 1967 to 1998
- Figure 4.15-5 General Aviation Accident Rates for 1977 to 1998
- Figure 4.15-6 Proposed OCX Arrival and Departure Routes
- Figure 4.15-7 Plot of Commercial Aircraft Landing Accidents/Incidents Relative to Runway
- Figure 4.15-8 Percent of Landing Accidents/Incidents by Distance From Runway Approach End
- Figure 4.15-9 Plot of Commercial Aircraft Departure Accidents/Incidents Relative to Runway
- Figure 4.15-10 Percent of Departure Accidents/Incidents by Distance From Runway Departure End
- Figure 4.15-11 Air Safety Zones Proposed for OCX
- Figure 4.16-1 Installation Restoration Program Sites at MCAS El Toro
- Figure 4.16-2 Proposed Project With Existing Hazardous Waste Sites MCAS El Toro

- Figure 4.17-1 Project Sites, Surrounding Area and Orange County
- Figure 4.18-1 Accident Frequency/Severity Screening Matrix Figure 8-1 Alternative A
- Figure 8-1 Nonaviation Alternative
- Figure 8-2 Alternative A
- Figure 8-3 Alternative A Year 2020 CNEL 60, 65, and 70 dB
- Figure 8-4 Alternative C
- Figure 8-5 Alternative C Year 2020 CNEL 60, 65, and 70 dB
- Figure 8-6 Alternative F
- Figure 8-7 JWA Year 2020 Alternative F CNEL Contours 60, 65, and 70 dB
- Figure 8-8 Alternative G
- Figure 8-9 JWA Year 2020 Alternative G CNEL Contours 60, 65, and 70 dB
- Figure 8-10 Alternative J
- Figure 8-11 Alternative J Year 2020 CNEL 60, 65, and 70 dB
- Figure 8-12A Wildlands Ranch Plan Alternative 1
- Figure 8-12B Wildlands Ranch Plan Alternative 2
- Figure 8-13A Nonaviation Revenue Support Area Alternative 1
- Figure 8-13B Nonaviation Revenue Support Area Alternative 2

## 5.0 CUMULATIVE IMPACTS

This section addresses potential cumulative impacts to the environment that could be associated with implementation of the Proposed Project in concert with one or more other past, present, and reasonably foreseeable probable future projects.

This section also includes a list of approved and planned projects, their environmental consequences, a brief discussion of their cumulative impacts on the environment and a description of the combined cumulative impact of the projects and the Proposed Project.

#### 5.1 CEQA DEFINITION OF CUMULATIVE IMPACTS

The CEQA Guidelines (§15130) require that a project's cumulative impacts be discussed when "...the incremental effect is cumulatively considerable..." According to CEQA Guideline §15065(c), the term cumulatively considerable means "...that the incremental effects of an individual project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects..." Specifically, CEQA Guideline §15355 defines cumulative impacts as:

"...two or more individual effects which, when considered together, are considerable or which compound or increase other environmental impacts.

- a) The individual effects may be changes resulting from a single project or a number of separate projects.
- b) The cumulative impact from several projects is the change in the environment which results from the incremental impact of the project when added to other closely related past, present, and reasonably foreseeable probable future projects. Cumulative impacts can result from individually minor but collectively significant projects taking place over a period of time."

## 5.1.1 Methodology

When addressing cumulative impacts, Section 15130(b) of the CEQA Guidelines notes that the elements necessary to provide an adequate discussion of significant cumulative impacts encompass either:

- "a) A list of past, present, and probable future projects producing related or cumulative impacts, including, if necessary, those projects outside the control of the agency, or
- b) A summary of projections contained in an adopted general plan or related planning document, or in a prior environmental document which has been adopted or certified, which described or evaluated regional or area-wide conditions

contributing to the cumulative impact. Any such planning document shall be referenced and made available to the public at a location specified by the lead agency;"

## Existing Setting/Existing Conditions

The Environmental Setting/Existing Conditions for this section is the Orange County Preferred–1996 Modified (OCP-96M) growth forecasts. As discussed in Section 4.17, these forecasts include the CRP and growth in nearby unplanned/unentitled areas analyzed in EIR 563.

## Related Projects Included in OCP-96M

Section 5.3 identifies individual projects identified by cities/agencies within the vicinity of the Proposed Project. The County reviewed these projects and determined that all of the projects were consistent with the year 2020 projections contained OCP-96M adopted by Orange County Council of Governments and the County except the City of Irvine General Plan post-2020 unspecified projects.

#### Availability of OCP-96M

The OCP-96M forecasts are contained in and form the basis of the adopted County Land Use Element (pg. II-5 through II-12), which is available at the County Planning and Development Services Department, 300 North Flower St., Santa Ana, CA. In addition to the County Land Use Element, OCP-96M was approved by the Orange County Council of Governments in 1997 and transmitted to the Southern California Association of Governments (SCAG) as the County input to the regional planning program. SCAG considers the forecasts in regional plans, including the Regional Transportation Plan. The Board of Supervisors adopted OCP-96M on July 22, 1997. Table 5.1-1 summarizes the OCP-96M forecasts for Orange County and the SCAG Baseline Projections for the year 2000 and 2020.

## Post-2020 Traffic Analysis

In response to NOP comments (e.g., page 805, No. 6), Section 4.5.3 includes an analysis of development intensification in the traffic study area and build out of the MPAH. This scenario is titled "post-2020" to distinguish it from the year 2020 OCP-96M forecasts. Table 5.1-2 compares the year 2020 OCP-96M forecasts and the post-2020 forecasts analyzed herein for the seven Community Analysis Statistical Areas (Figure 4.17-1) where post-2020 impacts are anticipated by commentors to occur.

Table 5.1-1
Regional Growth Projections 2000 to 2020

		2000			2020	
Geographic Area	Population	Housing Units	Employment	Population	Housing Units	Employment
Orange County <sup>1</sup>	2,865,830	990,311	1,381,695	3,244,600	1,154,528	2,116,600
Los Angeles County	9,818,235	3,131,606	4,557,889	12,249,088	3,984,119	5,817,641
Riverside County	1,687,764	547,361	515,712	2,815,987	917,960	960,777
San Bernardino County	1,772,539	564,962	617,055	2,830,050	904,942	1,103,362
SCAG Region <sup>2</sup>	16,999,453	5,434,377	7,441,154	22,352,394	7,320,286	10,573,759

SCAG region includes Orange, Los Angeles, San Bernardino, Riverside, Ventura, and Imperial counties. SCAG projections for Orange County and OCP-96M differ slightly so totals for region do not add.

NOTES: [1] Orange County Preferred-1996 Modified. Prepared by California State University at Fullerton. July 1997.

[2] SCAG Projections. Adopted April, 1998.

Table 5.1-2 Comparison of Year 2020 OCP-96M and Post-2020 Cumulative Development Forecasts

			Population	1		Housing	1	E	mploymen	ŧ
Location	CAA	2020	P-2020	Diff.	2020	P-2020	Diff.	2020	P-2020	Diff.
North Newport Beach (JWA Area)	47	45,356	46,870	+1,514	19,762	20,398	+636	22,040	22,175	+135
John Wayne Airport	48	5,671	6,996	+1,325	2,540	4,578	+2,038	59,983	70,904	+10,921
Irvine Industrial Complex	49	11,625	11,650	+25	4,172	4,247	+75	100,566	118,144	+17,578
South Irvine	50	36,760	47,627	+10,867	15,109	19,571	+4,462	19,279	36,701	+17,422
Central Irvine	51	83,487	93,026	+9,539	28,625	34,048	+5,423	38,387	63,841	+25,454
North Irvine	52	94,983	108,821	+13,838	35,377	41,320	+5,943	70,069	80,388	+10,319
East Irvine Industrial	53	0	0	0	0	0	0	40,526	68,353	+27,827
Total		277,882	314,990	+37,108	105,585	124,162	+18,577	350,850	460,506	109,656

Notes: CAA: Community Analysis Area

CAA boundaries are shown on Figure 4.17-1. Year 2020 forecasts are based on OCP-96M.

## Cumulative Analyses in Section 4.0

Certain impact analysis are best understood and/or are required by policy/regulation (e.g., Measure M/CMP) to include a cumulative impact component. These analyses include issues such as traffic and traffic related noise, regional air quality and other impact categories. Wherever this is the case, Section 4.0 includes a cumulative analysis (e.g., Transportation) and the discussion for these impacts will refer the reader to the appropriate subsection in Section 4.0.

## 5.1.2 Geographic Area

The geographic area addressed in this cumulative impact analysis can vary according to the nature and characteristics of each environmental resource. For example, certain project-related impacts tend to be site-specific with limited potential for cumulative impacts beyond the general vicinity of the project site. In contrast, other project-related impacts may exhibit a greater potential for affecting a larger geographic area, perhaps on a regional scale. Consequently two geographic areas have been defined for the purposes of this cumulative impact analysis.

The first area is the general vicinity of the Proposed Project, encompassing the communities of Costa Mesa, Irvine, Lake Forest, Laguna Woods, Mission Viejo, Newport Beach, Santa Ana, and Tustin. This geographic area generally coincides with that addressed in the impact evaluation section for Transportation and Circulation (Section 4.3). Environmental resources evaluated for cumulative impact potential within this area include:

- i) Land Use
- ii) General Plan Consistency
- iii) Transportation and Circulation
- iv) Noise
- v) Local Air Quality
- vi) Topography
- vii) Soils, Geology, and Seismicity
- viii) Hydrology and Water Quality
- ix) Biological Resources
- x) Public Services and Utilities
- xi) Aesthetics, Light and Glare
- xii) Cultural Resources
- xiii) Recreation
- xiv) Public Health and Safety
- xv) Hazardous Materials and Hazardous Waste
- xvi) Area Socioeconomics

The second geographic area encompasses the County and a substantial portion of the Southern California region, generally the territory within the jurisdiction of the South Coast Air Quality Management District (SCAQMD), comprising the Counties of Los Angeles, Orange, San Bernardino (non-desert portion), and Riverside. Environmental resources evaluated for cumulative impact potential within this area include:

- i) Regional Air Quality
- ii) Natural Resources and Energy
- iii) County-wide Socioeconomics

For these categories, the cumulative impact analysis focuses primarily on the potential impacts of reasonably anticipated future projects. Impacts of past and present projects are reflected in the baseline environmental conditions in the Proposed Project area, and are discussed, as appropriate, in Section 4.0.

#### 5.1.3 Criteria for Selection

The following criteria were considered in identifying those past, present, and reasonably anticipated projects that could potentially result in cumulative impacts:

- i) Projects that have an application for construction and/or operation pending before an agency with permit/approval authority, and/or;
- ii) Proposed projects that have the potential to generate environmental impacts that, when addressed collectively with the Proposed Project, could result in cumulative impacts to the environment, and/or;
- iii) Proposed projects that are of a similar character, could affect similar environmental resources, or are located in geographic proximity to the Proposed Project.

# 5.2 SUMMARY OF CONCLUSIONS IN FINAL EIR NO. 563

The following paragraphs summarize the conclusions derived from the cumulative impact analyses of the MCAS El Toro Community Reuse Plan, (CRP) Final Environmental Impact Report No. 563 CRP (Final EIR No. 563), as well as the Final Supplemental Analysis to EIR No. 563. Also summarized are CEQA findings, facts in support of findings and overriding considerations, relative to cumulative impact issues (Final EIR No. 563 (Vol. 6A), Attachment A). In addition, specific mitigation measures pertaining to cumulative impacts from the Mitigation Monitoring Plan contained within Final EIR No. 563 as supplemented by the FSA are noted, where appropriate and applicable.

CEQA Guidelines Sections 15130(d) and (e) and 15183 permit reliance on a prior EIR to streamline cumulative impacts analysis. The cumulative impacts analysis in EIR No. 563

was not challenged in court. To the extent permissible under CEQA and except as otherwise stated herein, this EIR incorporates by reference and relies upon the EIR No. 563 Cumulative Impacts discussions.

#### 5.2.1 Land Use

Final EIR No. 563 found that the CRP 65 dBA CNEL noise contour encompassed a smaller area than the existing MCAS El Toro Policy Implementation Line (PIL) due to a projected quieter commercial aircraft fleet mix. Therefore, restricted land uses in this area could be removed, modified or otherwise converted to different, less restrictive uses at subsequent levels of planning. According to Final EIR No. 563, these potential changes could increase the overall land use intensity surrounding the site. It was further noted that implementation of the CRP would add to the overall urbanization of Orange County and the area surrounding MCAS El Toro.

Final EIR No. 563 judged these potential changes/effects to constitute a significant cumulative impact related to land use, which could not feasibly be mitigated to below a level of significance. Remaining unavoidable effects were judged to be acceptable when balanced against specific overriding economic, legal, social, technological, and other considerations.

## 5.2.2 Transportation and Circulation

In Final EIR No. 563, traffic impacts associated with the CRP were found to be significant. It was anticipated that the CRP would result in significant cumulative adverse impacts related to transportation and circulation when combined with other planned or approved projects. Although changes and alterations were incorporated into the design of the CRP to substantially avoid or mitigate significant environmental effects, the identified cumulative effects remained significant and unmitigable. No feasible measures were identified to mitigate this impact to below a level of significance. However, the County of Orange Board of Supervisors determined that this significant effect was acceptable because of specific overriding economic, legal, social, technological, and other considerations described in the Statement of Overriding Considerations.

#### 5.2.3 Noise

Final EIR No. 563 concluded that the CRP could potentially contribute to background traffic noise exposure along the roadway segments of Trabuco Road. This was considered to be a significant adverse cumulative impact, which feasibly could not be mitigated to below a level of significance. The Board determined that the remaining unavoidable effects were acceptable when balanced against the specific overriding economic, legal social technological and other considerations described in the Statement of Overriding Considerations.

## 5.2.4 Hydrology and Water Quality

Final EIR No. 563 found that some cumulative water quality impacts would occur due to the intensification of urban uses on the MCAS El Toro site, including roads, parking areas and structures, which could result in increased urban pollutants in rain and irrigation runoff from the site. When coupled with other projects in the San Diego Creek watershed, it was determined that the proposed CRP would contribute to a cumulatively significant adverse impact related to water quality in the area.

Although changes and alterations were incorporated into the design of the CRP to substantially avoid or mitigate significant environmental effects, the identified cumulative effects remained significant and unmitigable. No feasible measures were identified to mitigate this impact to below a level of significance. However, the Board determined that this significant effect was acceptable because of specific overriding economic, legal, social, technological, and other considerations described in the Statement of Overriding Considerations.

#### 5.2.5 Public Service and Utilities

Final EIR No. 563 found that the CRP was not anticipated to result in significant adverse impacts on public services and utilities after mitigation. However, it also noted that, in those areas which are no longer subject to aviation-related noise due to potential future changes in the PIL, more intensive development might be permitted. Final EIR No. 563 indicated that such development could create a higher demand on public services and utilities than previous uses. In addition, Final EIR No. 563 noted several related projects in the vicinity of MCAS El Toro that would require public services and utilities. In conjunction with these projects, the CRP was found to result in significant cumulative adverse impacts related to public services and utilities described in the Statement of Overriding Considerations.

Although changes and alterations were incorporated into the design of the CRP to substantially avoid or mitigate significant environmental effects, the identified cumulative effects remained significant and unmitigable. No feasible measures were identified to mitigate this impact to below a level of significance. However, the Board determined that this significant effect was acceptable because of specific overriding economic, legal, social, technological, and other considerations.

## 5.2.6 Natural Resources and Energy

Final EIR No. 563 found that the loss of agricultural land on MCAS El Toro was a significant unavoidable adverse impact, and that the loss of agricultural land is also a significant adverse impact of many related projects. Although changes and alterations were incorporated into the design of the CRP to substantially avoid or mitigate significant environmental effects, the identified cumulative effects remained significant and unmitigable. No feasible measures were identified to mitigate this impact to below a level of significance. However, the Board determined that this significant effect was acceptable because of specific overriding economic, legal, social, technological, and other considerations described in the Statement of Overriding Considerations.

## 5.2.7 Cultural Resources

Final EIR No. 563 found that implementation of the CRP had the potential to significantly and adversely affect certain Cold War era historic resources on the MCAS El Toro site. Although changes and alterations were incorporated into the design of the CRP to substantially avoid or mitigate significant environmental effects, the identified cumulative effects remained significant and unmitigable. No feasible measures were identified to mitigate this impact to below a level of significance. However, the Board determined that this significant effect was acceptable because of specific overriding economic, legal, social, technological, and other considerations.

#### 5.2.8 Socioeconomics

Final EIR No. 563 found that implementation of the CRP was expected to result in several socioeconomic impacts of potential significance when considered cumulatively with related projects. These impacts would be related to jobs/housing balance and low/moderate income housing needs. Although changes and alterations were incorporated into the design of the CRP to substantially avoid or mitigate significant environmental effects, the identified cumulative effects remained significant and unmitigable. No feasible measures were identified to mitigate this impact to below a level of significance. However, the Board determined that this significant effect was acceptable because of specific overriding economic, legal, social, technological, and other considerations described in the Statement of Overriding Considerations.

A Supplemental Analysis (SA) for Final EIR No. 563 was prepared in response to a January 6, 1998 ruling and writ of mandate, issued by the San Diego Superior Court, which found that certain portions of EIR No. 563 were inadequate or incomplete under CEQA. The SA was not required to address CRP cumulative impacts.

# 5.3 REASONABLY FORESEEABLE PROBABLE FUTURE PROJECTS

This section addresses those projects that met the criteria outlined under Subsection 5.1.1, Methodology. Varying levels and completeness of information, depending on the current development status of the particular project, is provided for each of the projects addressed below. This information may range from a simple project description to a comprehensive environmental review. Some of the cumulative projects were previously included in Final EIR No. 563, as provided below. In such cases, the local jurisdictions responsible for these projects were contacted, and the information on these projects was updated.

## County Projects

- Foothill Ranch Planned Community
- Portola Planned Community
- Saddleback Meadows

#### City Projects

- Sakioka Property
- Lower Peters Canyon Specific Plan
- Westpark Project
- Moulton Parkway Commercial Development
- Foothill-Aliso Commercial Center
- Saddleback Valley Church
- First American Title
- MacArthur Place
- Nexus Twin Towers
- One Hutton Center

#### **Transportation Related Projects**

- Alton Parkway Extension
- Foothill Transportation Corridor

#### Other Related Land Use Projects

- Natural Community Conservation Plan
- MCAS Tustin Reuse Plan
- James A. Musick Facility

Other projects were identified by surveying other land use authorities within the planning area of the Proposed Project. These surveys consisted of formal and informal inquiries designed to acquire existing available environmental documentation and project descriptions. These projects included the following:

#### City Jurisdiction Land Use Project

- Segerstrom Homes Ranch
- General Plan Amendment/Zone Changes for 440 acres south of MCAS El Toro
- Oak Creek Project
- Westpark Project
- Revised City of Irvine Planning Areas 51, 35, and 30, Annexation, General Plan Amendment, Pre-Zoning, and Zone Change Project
- GPA/ZC to allow housing in the Spectrum area of Irvine
- North Irvine Middle School
- Northwood High School
- Alicia Skate Park
- Allen Oldsmobile/Cadillac
- Enterprise Car Sale
- La Paz Park
- Mission Imports Remote Parking Lot
- Moulton Elementary Field Improvements
- Shapell/Aliso Creek
- Shea Business Properties Ocean Ranch II
- St. Anne's School
- Tutor Time/Tentative Parcel Map No. 98-171
- Town Centre
- Home Depot Center
- Baker Ranch Planned Community
- A.J. West Ranch
- Los Angeles Athletic Company Project
- Mammoth Equities Project
- Painted Trails
- Stone Ridge
- County Suites Hotel
- Shea Properties Apartments
- Acero Street Office Project
- Heritage Villas Senior Housing
- Camino Capistrano Mini-Storage
- Legacy Partners Office Building
- Mission Viejo Mall Expansion

- Lexus Auto Dealership
- Rockwell Site Expansion
- Koll Center Expansion
- Hertz Extended Stay Hotel and Office Project
- Amendment 891 in Newport Beach
- Conerant Expansion Project (GPA 96-39(f))
- OCTA CenterLine Project

#### Transportation Land Use Projects

- I-405/SR-55 HOV Access Improvement Project
- LAX Expansion Project
- Proposed High Speed Rail Project

The following subsections provide a discussion of probable future projects that, when considered in combination with the Proposed Project have the potential to result in related or cumulative impacts. These projects are presented as follows:

- i) County of Orange Jurisdiction Land Use Projects
- ii) City Jurisdictions Land Use Projects
- iii) Transportation Land Use Projects
- iv) Other Land Use Projects
- v) Non-Planned Areas Land Use Projects

The jurisdiction of the County of Orange encompasses the unincorporated territory of Orange County. The city jurisdictions include the Cities of Costa Mesa, Irvine, Laguna Woods, Mission Viejo, Lake Forest, Newport Beach, and Santa Ana. Laguna Niguel is located outside, but adjacent to, the area of cumulative project effects, so future projects in Laguna Niguel are not discussed in detail in this section of the EIR. The potential for cumulative impacts associated with other projects in the vicinity of JWA and MCAS El Toro also is discussed.

One important consideration in the present cumulative impact discussion focuses on the removal or reduction of the existing PIL, which restricts the development of noise sensitive land uses in areas included within the PIL. Removal of this noise constraint in the remaining undeveloped areas could encourage or might allow different land use mixes than previously indicated on local jurisdiction's General Plans in the PIL area, since certain noise sensitive uses could be deemed compatible with reduction or removal of the PIL. The principal area where residential or noise sensitive uses could be expanded is the undeveloped or underdeveloped areas of east Irvine, which are now subject to the military 65 dB CNEL line and the military accident potential zones.

## 5.3.1 County of Orange Jurisdiction Land Use Projects

There are no proposed land use projects under County jurisdiction within the vicinity of JWA. JWA is surrounded by the incorporated cities of Costa Mesa, Irvine and Newport Beach.

The County of Orange has identified a number of approved and planned projects in the vicinity of MCAS El Toro that potentially contribute to cumulatively significant adverse impacts. A summary of the projects that fall under the County of Orange jurisdiction is provided in Table 5.3-1. As noted in Subsection 5.1.1, the County determined that these projects are consistent and included in OCP-96M. Each is discussed separately, below.

For the present cumulative impact analysis, the OCP-96M was used to assess population, housing and employment projections for the MCAS El Toro and JWA site vicinities, as well as surrounding areas of the County. This information was then used to evaluate the potential for cumulative impacts to occur as a result of implementing the Proposed Project. OCP-96M Projections were approved by the Orange County Council of Governments in 1997 and were transmitted to the Southern California Association of Governments (SCAG) as the County input to the regional planning program. SCAG incorporated OCP-96M into its program, including the Regional Transportation Plan.

## 5.3.1.1 Foothill Ranch Planned Community

The Foothill Ranch Planned Community comprises four primary land uses: residential, commercial, industrial, and open space. In general, residential developments would be limited to 3,900 dwelling units to be located between Whiting Ranch Wilderness Park and an Urban Activity Center between Portola Parkway and the Foothill Transportation Corridor (FTC). Commercial development within this Planned Community would be limited to 750,000 sq. ft., including a 500,000 sq. ft. regional commercial center. Industrial uses include a business park along the FTC in the Urban Activity Center. In addition, office, research and development uses would be developed. In total, these uses would not exceed 6,925,000 sq. ft. on approximately 415 acres. At present, approximately 95 percent of Foothill Ranch is built-out (Orange County, 1999e).

## 5.3.1.2 Portola Hills Planned Community

The Portola Hills Planned Community covers approximately 1,006 acres in foothills of the Santa Ana Mountains. The site lies west of El Toro Road and south of Santiago Canyon Road in unincorporated areas of Orange County. This project area is adjacent to the Foothill Ranch Planned Community to the west and south, and partially lies within the existing MCAS El Toro PIL.

Table 5.3-1 Reasonable Foreseeable Probable Future Projects County of Orange Jurisdiction Land Use Projects

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Foothill Ranch Planned Community	County of Orange	x	X	x	X			Development of approximately 6,925,000 sq.ft. located between Whiting Ranch Wilderness Park and an Urban Activity Center between Portola Parkway and the Foothill Transportation Corridor (FTC) Residential: 3,900 dwelling units Commercial: 750,000 sq.ft., including 500,000 sq.ft regional commercial center. Industrial: 6,925,000 sq. ft. business park on 415 acres				X		95% built out
Portola Hills Planned Community	County of Orange	X	X	X				Development of approximately 1,006 acres located west of El Toro Road and South of Santiago Canyon Road Residential: approximately 2,200 residential units on 493 acres Commercial: approximately 200 acres developed into business  park and commercial areas  Open Space: approximately 400 acres of dedicated open space and open space easements				X		95% built out
Saddleback Meadows	County of Orange	x	x					Development of a 229.2 acre site located approximately 0.25 mile south of the intersection of El Toro and Live Oak Canyon Roads Residential: 235 single family detached residential units, associated infrastructure, a wildlife movement corridor, hiking/riding trail, and biological resource enhancement on a 229.2-acre site.		x				In litigation, no development onsite

The Portola Hills Planned Community comprises approximately 2,200 residential units on 493 acres, a business park and commercial area totaling approximately 200 acres, and dedicated open space and open space easements totaling approximately 400 acres. As of July 1999, Portola Hills is 95 percent built out (Orange County, 1999e).

#### 5.3.1.3 Saddleback Meadows

The Saddleback Meadows project proposes development of 235 single-family detached residential units, associated infrastructure, a wildlife movement corridor, hiking/riding trail, and biological resource enhancement on a 229.2-acre site. The site is approximately 0.25 mile south of the intersection of El Toro and Live Oak Canyon Roads, immediately north of the Hidden Ridge community and east of El Toro Road in the Foothill Trabuco Specific Plan area of unincorporated Orange County. As of November 1999, the Saddleback Meadows project was being litigated, and no development had occurred on the project site (Orange County, 1999c).

## 5.3.2 City Jurisdictions Land Use Projects

The regional city jurisdictions include the Cities of Costa Mesa, Irvine, Laguna Niguel, Laguna Woods, Mission Viejo, Lake Forest, Newport Beach, and Santa Ana. The related projects that fall under city jurisdictions are outlined in Table 5.3-2.

## 5.3.2.1 City of Costa Mesa Jurisdiction Land Use Projects

## Sakioka Property

This project consists of the planned expansion of existing structures on a 47-acre site located at 555 and 575 Anton Boulevard. The project includes the construction of four office buildings, two restaurants, a retail outlet, a 200-room hotel, one clubhouse and three parking structures. The project would add approximately 546,600 sq. ft. of commercial use. The City of Costa Mesa City Council approved the project in April 1999. The first phase of the project, is underway. The entire project should be built out between the year 2000 to 2005. The City is currently processing a General Plan Amendment to allow an additional 100 rooms to be constructed in the project Hotel. The City Planning Commission may consider this amendment in November 1999 (letter from City dated Sept. 30, 1999)

Table 5.3-2
Reasonably Foreseeable Probable Future Projects
City Jurisdictions Land Use Projects

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	tion	Π	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Sakioka Property	City of Costa Mesa			X				47 acres located at 555 and 575 Anton Boulevard Commercial: 4 office buildings, 2 restaurants, Retail outlet, hotel, parking structures		X		)		This project was approved in April 1999 by the City Council. The City is processing a General Plan Amendment (GPA) to allow the development of up to 300 rooms in the Project Hotel. The City Planning Commission may consider the EPA in Nov. 1999. The project would be completed within the year 2000-2005.
Segerstrom Home Ranch	City of Costa Mesa			X	***************************************			This project is located on 93 acres between Harbor Boulevard and Fairview Road. The project includes:  300,000 square feet IKEA Home Furnishings Store  1,994,17 square feet of office spaces	X		***************************************			The City is currently processing a General Plan Amendment to allow for the development of this project. A Draft EIR is also being prepared on this project. If approved, the IKEA Store would be developed within the year 2000 to 2005. The office development would be constructed in equal increments through the year 2020.
General Plan Amendment/Zone Change for 440 Acres South of MCAS El Toro (Alternative to the Proposed Project)	City of Irvine			x		х	x	342 acres located within the MCAS El Toro boundary Development of multi-use entertainment center, including:  Sports stadium, arena Hotel/convention center Cultural uses Multi-modal transit corridor, including motor vehicles, shuttles, trans, and/or light rail, and pedestrian/bicycle trail system	x					Final EIR pending and completion expected in early 2000. Project is currently in litigation with County of Orange. May be superseded by Annexation GPA

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Lower Peters Canyon Specific Plan	City of Irvine	x	x	x			x	Approximately 1,400 acres proposed for up to 10,568 residential dwelling units.  Open Space: a special use park, a community park, and six neighborhood parks  Residential: 10,568 residential dwelling units  Commercial: retail center, libraries and schools		x		X		Construction complete in some areas.
Oak Creek Project	City of Irvine	X	x	x	X			1,250 acres in Planning Area 12 Residential: 4,050 units Commercial: 1,105,000 sq. ft commercial; 470,000 sq.ft non-residential uses, 535,400 sq.ft new institutional uses. Industrial: 2,871,080 sq.ft. Open Space: 339 acres		X		x		Approximately 70% complete. Construction of residential units continuing.
West Park Project	City of Irvine		X					350 acres in Planning Area 38 to provide 3,850 units.		X		x	<b>†</b>	Approximately 95% complete.
Revised City of Irvine Planning Areas #51, #35, and #30, Annexation, General Plan Amendment, Pre-Zoning, and Zone Change Project (An alternative to the Proposed Project)	City of Irvine	X	X	X			X	Annexation of approximately 4,298 acres within the MCAS El Toro site, plus 113 acres in adjacent unincorporated area of Orange County to include:  James A. Musick Jail Irvine Ranch Water District Parcels  Additional aspects of project include:  Amendment to Master Plan of Arterial Highways  General Plan Amendments to City Elements	X		WWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWW		***************************************	NOP issued August 1999; Draft EIR issued 11-24-99; considered by the City to be Phase II of Millennium Plan (ETRPA Nonaviation Plan).
GPA/ZC to allow housing in the in the Spectrum area.	City of Irvine	X				A PROPERTY AND A PROP	X	The City of Irvine has initiated a General Plan Amendment, zone change and possibly an amendment to the Irvine Center Development Agreement to allow housing in the spectrum area of the City.	X		**************************************			An EIR will be prepared on the proposed GPA/ZC, with the EIR tentatively to be distributed for public review in February 2000. Consideration of this project has also been tentatively scheduled for consideration by the City's PC in April 2000 and by the City Council in May 2000.
North Irvine Middle School	Irvine Unified School District						х	20 acre site located south of Hicks Canyon Wash and east of Hicks Canyon Road. This project will provide facilities to accommodate 750 students.		Х	x			Neg. Dec. February 1992 Completion for 2003.

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Northwood High School	Irvine Unified School District						x	43 acre site located at future intersection of Portola Parkway and Yale Avenue. This project will provide facilities to accommodate 2,400 students.		х			х	Approved 1995. Construction completed in Fall 1999; portion of school now open
Alicia Skate Park	City of Laguna Niguel						X	4 acre public skate board park	x					Preliminary Planning Stages.
Allen Oldsmobile/Cadillac Parcel Map 97-126	City of Laguna Niguel			X				Expansion of existing automobile dealership located at Camino Capistrano and Paseo de Colinas.		x		X		Under Construction.
Enterprise Car Sales	City of Laguna Niguel			X				2 acres located at 28112 Camino Capistrano to provide construction for a used car sales and rental facility		X		X		Approved May 1999.
La Paz Park	City of Laguna Niguel	x						6.3 acre public park with lighted active sports fields	x					DEIR released for Public Review in July 1999.
Mission Imports Remote Parking Lot	City of Laguna Niguel			X				2.4 acre site located at 1 Star Drive consists of surplus parking area with night lighting		X		x		Approved December 1998.
Moulton Elementary Field Improvements	City of Laguna Niguel						x	4 acre site located at 29851 Highlands Avenue consists of expansion to active recreational facilities and parking areas at the Moulton Elementary School.		х				Approved by City.
Shapell/Aliso Creek	City of Laguna Niguel		x					9 acre site in Tentative Tract 15670 Residential: 46 detached single-family dwelling units		X		X		Under Construction.
Shea Business Properties- Ocean Ranch II	City of Laguna Niguel			x				Site located at 32451 Golden Lantern to provide 45,000-sq.ft. office building and health club on a vacant pad adjacent to an existing commercial center.		x				Approved July 1999.
St. Anne's School	City of Laguna Niguel						x	Site of existing private schools located at 32451 Bear Brand Road. This project will provide:  29,235 sq.ft. classroom building  33,065 sq.ft. classroom/administration building		x			x	Construction completed July 1999.
Tutor Time/Tentative Parcel Map No. 98-171	City of Laguna Niguel			X				Located at Aliso Creek Road, east of La Paz Road. This project will provide a new 12,000-sq.ft. day care facility.	x			x		Approved January 1999.
Moulton Parkway Commercial Development	City of Laguna Woods			X				30,000-sq. ft. office project. This project includes 26,178 square feet of space for the Los Angeles Times.				x		L.A. Times sorting/ distribution building.
Town Centre	City of Laguna Woods			x	***************************************			This project includes 7 building which include 253,270 square feet of space in the Town Centre commercial project which is located on 14 acres on El Toro Road. The project includes: 140 room (80,000 square feet) Hotel; 40,000 square feet Medical Offices; 7,000 square feet Restaurant; 45,570 square feet garden office; 68,000 square feet of offices; 5,900 square feet of services retail; and 6,800 square feet of retail use.		x	x			In Building Plan Check Process. Project to begin construction in early 2000.
Home Depot Center	City of Laguna Woods			x				This 215,773 square foot commercial center consists of a Home Depot, Office Max, Rite Aide, Stater Brothers and in-line retail stores. This project also includes a bank and pad for a future restaurant.					x	This project has already been constructed.

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Foothill-Aliso Commercial Center	City of Lake Forest Influence County of Orange	888		x				Approximately 16-acre site, previously zoned open space, located on Portola Parkway. This property consists of several elements, including:  10,000 sq. ft. on approximately 8-acre parcel  40,000 sq. ft. office building  7,000 sq. ft. office and food facility on approximately 3-acre parcel  3-acre right-of-way for Portola Parkway	x	***************************************		x		To date, 40,000 square feet of office space has been developed. The City of Lake Forest Planning Commission will consider the development of 110,000 square feet of commercial development on this project site in November 1999. It is anticipated that this project will be completed between the year 2000 and 2005.
Baker Ranch Planned Community	City of Lake Forest			x	X			4.3 million square feet of business park development on 370 gross acres located south of the Foothill toll road with the City of Lake Forest City boundary to the west and Bake Parkway to the east.	X					The property owner is currently processing a new tentative tract map for this project.
A.J. West Ranch	City of Lake Forest			X	Х			933,000 square feet of business park development on 77 acres of land located south of the Foothill toll road and east of Baker Parkway.		X		х		This project will be developed between the year 2000 – 2005.
Saddleback Valley Church	City of Lake Forest			X		***************************************	x	This project is located on the northwest comer of El Toro Road and Portola Parkway and includes 800,000 square feet of Church related development. Some commercial development may also be included in this project. The County of Orange approved 310,000 square feet of development as part of a master plan for the Church. Some Church buildings have been constructed to date. Precise development plans for the remaining 489,000 square feet have not been submitted to the City of Lake Forest	x	×		x		It is anticipated that this project will be built out by 2005.
Los Angeles Athletic Company (LAACO) Project	City of Mission Viejo			x				3 acre site located north of El Toro Road and east of Marguerite Parkway to provide 77,095 sq. ft. self-storage facility.		X		x		Approved June 1999. Construction to begin on or before September 1999
Mammoth Equities Project	City of Mission Viejo	x		x				3 lot (~16.5 acre) site located at the southeast corner of El Toro Road and Marguerite Parkway. Commercial: 49,000 sq.ft. office building on ~2.5 acres Open Space: remainder of property		X				Approved June 1999. Construction to begin Fall 1999.
Painted Trails	City of Mission Viejo	x	x					Site located north of the Foothill toll road, east of El Toro Road, and west of Upper Oso Reservoir. The project will accommodate 578 single-family dwelling units and several private parks.		X		x		Approved 1990.

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	ric-acaga	In Progress	Complete	Comments
Stone Ridge	City of Mission Viejo	x	x					340 acre site located east of Olympiad Road between Alicia Parkway and Fieldcrest Street. The project will accommodate:  468 single-family dwelling units open space and local parks		x	***************************************	<b>x</b>		Approved 1991.
Country Suites Hotel	City of Mission Viejo						X	East of Marguerite Parkway and west of Foothill toll road ramps at Los Alisos; 28941 Los Alisos Boulevard 118 room hotel		X	1	ĸ		Approved December 1998; under construction; January 2000 occupancy.
Shea Properties Apartments	City of Mission Viejo		x					East of Marguerite Parkway and west of the Country Suites Hotel; 28401 Los Alisos Boulevard 230 apartment units		x	;	ĸ		Approved August 1998; Under construction; May 2000 occupancy.
Acero Street Office Project	City of Mission Viejo			x				225,000 square feet office buildings located on Acero Street	x					Proposed development; preliminary discussions, projected January 2001 occupancy.
Heritage Villas Sr. Housing	City of Mission Viejo		x					Southeast corner of Oso parkway and Country Club Drive; 26836 Oso Parkway; 143 senior housing units		x		x		Approved October 1998; under construction; April 2000 occupancy.
Camino Capistrano Mini- Storage	City of Mission Viejo						x	Camino Capistrano, northerly of Crown Valley Parkway; 107108 square feet of mini-storage, with a caretaker residence on 4.79 acres	x					Development application scheduled for City action in October 1999; projected February 2001 completion.
Legacy Partners Office Bldg.	City of Mission Viejo			x				27101 Puerta Real 27201 Puerta Real 232,000 square feet of office in two buildings		x				Approved June 1999; Preliminary site work begun; October 2000 completion.
Mission Viejo Mall Expansion (Shops at Mission Viejo)	City of Mission Viejo			x				Shops at Mission Viejo expansion; 500,000 square feet of commercial		x				Approved May 1997; buildout by June 2000, with 250,000 sq. ft. open in 1999 (Nordstrom/Saks).
Lexus Auto Dealership	City of Mission Viejo						x	38,614 square feet of auto dealership	***************************************	x				Approved June 1998; September 2000 completion.
Rockwell Site Expansion	City of Newport Beach			x				Expansion of 442,000 sq.ft. existing office structure General Plan and Planned Community Amendments to allow additional 566,000sq ft office building expansion.	x					Under environmental review.

Project	Jurisdiction	Open Space	Residential	Commertial	Industrial	Transportation	Offier	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Koll Center Expansion	City of Newport Beach			x				250,000 sq.ft., 10-story existing office structure located at intersection of MacArthur Boulevard and Jamboree Road.  Commercial: Demolition and reconstruction of parking structure office building expansion.  Also includes a GPA, amendment of Planned Community Development Plan and Traffic Study pursuant to the Newport Beach TPO and an EIR	X					Under environmental review. It's anticipated that a Draft EIR would be distributed for review in January 2000. Project to be considered by the City's Planning Commission in March or April 2000. The City Council tentatively scheduled to consider the project in April 2000.
Holtze Extended Stay Hotel and Office Project	City of Newport Beach			х				Former automobile dealership located at 1300 Quail Drive and Dove Avenue.  100,000 sq.ft. office building, 350 bed hotel located at 1300 Quail Drive.	***************************************	,	ĸ			Mitigated Negative Declaration approved by City Council. Plan check stage — unknown date to commence construction.
Amendment 891	City of Newport Beach	x	X	*			X	<ul> <li>10-acre elementary school site</li> <li>216 acres for parks, open space and habitat restoration</li> <li>The project includes GPA pre-zoning, pre-annexation zoning, Local Coastal Permit and a development agreement.</li> </ul>	X					The County of Orange is acting as the lead agency on this EIR currently being prepared on this project.
Conerant Expansion Project GPA 96-3(f)	City of Newport Beach			X				This commercial project is located at 4311 Jamboree Road. The project includes a GPA and PC amendments to allow for the development of an additional 566,000 sq. ft. of space for a total of 1,008,775 sq. ft. of allowable building area on the project site.	X					A Draft EIR has been prepared on this project and is out for public review. Review of the project by the City's Planning Commission and City Council has been tentatively scheduled for February 2000.

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory		Approved	Pre-design	In Progress	Complete	Comments
First American Title	City of Santa Ana			x			W	211,769 sq.ft. development located near the intersection of MacArthur Boulevard and Main Street and includes construction of 3 additional office buildings at 1,2 and 3 First American Way, as follows:  87,500 sq.ft., 3-story 64,059 sq.ft, 2-story 60,210 sq.ft., 2-story		×		X		Project is part of the MacArthur Place project.
MacArthur Place (Brea Properties)	City of Santa Ana			х				Mixed-use project located at the intersection of MacArthur Boulevard and Main Street. Residential: 29,419 sq. ft complex to residential housing, including 280 apartments and condominiums Commercial: 8 building complex to provide office and retail space		X		-		Environmental review is complete.
Nexus Twin Towers (Nexus)	City of Santa Ana	and the standard designation of the standard designation o		x				Development of 456,414 sq.ft. for an office building atop an existing parking garage, located at 1 East Majestic Drive Commercial:. office building will be 456,414 sq.ft., parking garage will be 220,366 sq.ft.		,		x		One twin tower is in plan check phase. Construction of other twin tower is complete.
One Hutton Center	City of Santa Ana			x				13 story office building project located near the northeast corner of MacArthur Boulevard and Main Street to provide:  240 room hotel  5,000 sq.ft. restaurant  4,740 sq.ft. conference hall  7 story parking structure (875 parking spaces)		,	<b>S</b>			Approved July 1992
OCTA Centerline Project	OCTA					X			x					Draft EIR released September 1999for public review.

## Segerstrom Home Ranch

The project is located on a 93 acre site between Harbor Boulevard and Fairview Road. The project includes a 300,000 square feet for an IKEA Home Furnishing Store and 1,994,177 square feet of office space. The City is processing a General Plan Amendment to allow for this use. A Draft EIR is currently being prepared on this project. It is anticipated that this project would be considered by City decision makers sometimes in the year 2000. It is also anticipated that the IKEA Store will be developed within the year 2000 to 2005. The office development would be constructed in equal increments through the year 2020 (letter from City dated Sept. 30, 1999).

## 5.3.2.2 City of Irvine Jurisdiction Land Use Projects

## General Plan Amendment/Zone Change for 400 Acres South of MCAS El Toro

Approximately 340 acres of the MCAS El Toro site are within the City of Irvine corporate boundary. The City has proposed GPA 21633-GA and a Zone Change, which would allow the 340 acres to be developed as a multi-use entertainment center. The center would provide areas for a sports stadium, an arena, a hotel/convention center, and other cultural uses. A multi-modal transit corridor has been proposed to traverse the site. This corridor is intended to accommodate a number of transit options, including motor vehicles, shuttles, trams, or light rail. It also features a pedestrian/bicycle trail system. An EIR was approved for this project in 1996. It is the subject of litigation by the County. That EIR may be superseded by the City's 1999 EIRs for the Millennium Plan or Millennium Plan II. The Final EIR for the Millennium Plan was not approved in June 1999, and a revised Draft EIR was issued in November 1999. The Final EIR is expected to be reviewed again in early 2000 (City of Irvine, 1999c).

Although this portion of the MCAS El Toro site lies within the City of Irvine corporate boundary, the federal, state or County government likely will be the ultimate property owner, in which case the property would not be subject to City of Irvine planning and zoning laws and regulations. Because this is a project alternative, it would not cause cumulative impacts in conjunction with the Proposed Project. For an analysis of nonaviation land uses at former MCAS El Toro, refer to Chapter 8.0, Alternatives.

## Lower Peters Canyon Specific Plan

The Lower Peters Canyon Specific Plan encompasses approximately 1,400 acres on a site bounded by the west leg of the Eastern toll road (ETR) and the planned alignment of Portola Parkway to the north, Culver Drive to the east, I-5 to the south, and Jamboree Road to the west. The Lower Peters Canyon Specific Plan contains land use regulations and development standards for all uses proposed for this site, including an allowance for up to

10,568 residential dwelling units. Currently, development is underway for up to 8,000 residential dwelling units, retail commercial land uses, a special use park, a community park, six neighborhood parks, a library and schools on 1,400 acres. The site recently was incorporated within the City of Irvine, between the East Tustin Planned Community in the City of Tustin and the Northwood community in the City of Irvine. Elements of the Lower Peters Canyon Specific Plan currently are being implemented in phases. Mixed uses have been constructed, as of July 1999 (City of Irvine, 1999b).

## Oak Creek Project

The Oak Creek project lies within City of Irvine Planning Area 12 on a 1,250-acre site, located north of I-5, east of Sand Canyon Avenue, south of I-405, and west of Jeffrey Road. This mixed use project includes the planned development of 4,050 new residential units, 1,105,000 sq. ft. of new commercial uses, 2,871,080 sq. ft. of new industrial uses, 470,000 sq. ft. of new non-residential uses, 535,400 sq. ft. of new institutional uses, and 339 acres of conservation/open space use. The residential component of the Oak Creek project, entailing development of single-family dwelling units on a 350-acre site, will be constructed in various phases. This project was approved by the City of Irvine, and construction began on a portion of the project in July 1999 (City of Irvine, 1999b).

## West Park Project

This project is located in City of Irvine Planning Area 38 on a 350-acre site. It is bounded on the north by Irvine Center Drive, on the south by Barranca Parkway, on the east by Culver Drive and on the west by Harvard Drive. Ninety-five percent of the proposed 3,850 units had been constructed, as of July 1999. Development of the remainder of the West Park Project was still under City of Irvine review, as of July 1999 (City of Irvine, 1999b).

Revisions to City of Irvine Planning Areas 51, 35, 30, annexation of certain areas into the City, General Plan Amendment, Pre-Zoning, and zone change.

The City of Irvine has proposed to amend Planning Areas 51, 35, and 30, and to adopt the necessary pre-zoning and a zone change for these areas to allow for land uses outlined in the ETRPA Nonaviation Plan referred to by the City as Phase II of the Millennium Plan. (See Chapter 8.0, Alternatives, for a discussion of the Millennium Plan.)

This project would also provide for the annexation of approximately 4,298 acres of the MCAS El Toro site into the City of Irvine. Also to be annexed into the City would be the James A. Musick Jail and Irvine Ranch Water District parcels in the vicinity of the MCAS El Toro, with these land uses comprising an additional 113 acres.

This project also calls for an amendment to the Master Plan of Arterial Highways for on-site roadway classified as Major, Primary and Secondary Arterial in the above areas.

This project will also include, at a minimum, amendments to the following City of Irvine General Plan Elements: Land Use; Circulation, Parks and Recreation, Open Space and Conservation, Noise and Safety.

In its NOP dated August 13, 1999, the City of Irvine stated that a Program Final EIR will be prepared for the project that includes a detailed project description and an analysis of the potential environmental effects of the project, identification of mitigation measures necessary to reduce significant environmental impacts and alternatives that could avoid or substantially reduce significant effects of the project. The Draft EIR was released for public review the day before Thanksgiving, November 24, 1999.

This project would not co-exist with the Proposed Project and thus does not contribute to cumulative impacts in conjunction with the Proposed Project.

## Unspecified Future Projects

The City of Irvine has indicated that it anticipates that a number of new residential, commercial, office, warehouse and institutional type projects will be developed within the various planning areas in the City in the years to come. No specific information was received on the exact location of these projects. These projects are based on projections being made by the City on development that could occur throughout the City. Proposed development includes the land uses proposed by the revised Millennium Plan which is an alternative project to the proposed project. These projects are summarized in Table 5.3-3.

## General Plan Amendment and Zone Change to Allow Housing in the Spectrum Area

The City of Irvine has initiated a General Plan Amendment and Zone Change to allow housing in the Spectrum area of the City. An amendment to the Irvine Center Development Agreement may also be required for this to occur. The City is in the process of retaining a consultant to prepare an EIR on this project. The EIR is tentatively scheduled to be distributed for public review in February 2000. Consideration of this project has also been tentatively scheduled for review by the City's Planning Commission in April 2000 and by the City Council in May 2000.

## 5.3.2.3 Irvine Unified School District (IUSD)

There are two school projects currently under consideration by the Irvine Unified School District (IUSD), Northwood High School (which has been partially built) and North Irvine Middle School. According to the IUSD, the City of Irvine Planning Area 27 also proposes a possible elementary school development during the year 2003 as part of the City's Final EIR (IUSD, 1999b). However, according to the IUSD no planning, other than general discussion, has been initiated for this potential school.

**Table 5.3-3 Unspecified Projects in the City of Irvine** 

Land Use		Pla	anning Are	a l			Pl	noning Are	a 2			Pla	inning Are	s 4			Plans	ing Area	6	
	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+
Estate [1]			<b>†</b>		222					25									<b></b>	263
Single Family Detached [1]					4,310					2,496	629	307	1,664	100						
Condominium [1]					<b>T</b>		1		1			1,276	537	365				ļ	<u> </u>	
Apartment [1]												139	2,913	70						
Commercial [2]			1		23,769		<b>1</b>		<b> </b>	406,300	406,300	339,299	329,401							
Office [2]					***					300,226	300,226	150,000	899,774							
Warehouse [2]							T													
R&D [2]						<b></b>					•••••									
Institutional [2]							1	1	l		•••••	Ī			<b></b>				<b>.</b>	

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

**Table 5.3-3 Unspecified Projects in the City of Irvine** 

Land Use		Pla	inning Area 1	2		Planning Area 13								
	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+				
Estate [1]														
Single Family Detached [1]	573	259	400	175		_			1					
Condominium [1]	552	525	712	568										
Apartment [1]	1,039		359											
Commercial [2]			150,000						ļ					
Hotel [2]						61,512								
Vehicle Related Comm. [2]	51,684													
Office [2]					1,197,980	1,487,652	181,687		1,630					
Warehouse [2]	347,962	250,505			77,550	497,638								
R&D [2]	73,580	250,000	1,452,358	120,881	485,531	208,164	61,512		2,161,622	1,073,476				
Institutional [2]	95,360				195,854									

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

**Table 5.3-3 Unspecified Projects in the City of Irvine** 

Land Use			Planning Area	17		Planning Area 18							
	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Yer 2005	Year 2015	Year 2020+			
Estate [1]													
Single Family Detached [1]					1,680					200			
Condomínium [1]					350					550			
Apartment [1]	1					***************************************							
Commercial [2]					300,000								
Hotel [2]													
Vehicle Related Comm. [2]													
Office [2]		1							<u> </u>				
Warehouse [2]													
R&D [2]			200,000		710,000								
Institutional [2]													

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

**Table 5.3-3** Unspecified Projects in the City of Irvine

Land Use		Pl	anning Area	22			1	Planning Area	27		Planning Area 30					
	Existing Development as of 6/30/99	Year 2006	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	
Estate [1]		100	100		200			500		475			<u> </u>		1	
Single Family Detached [1]			***************************************					600		580				T		
Condominium [1]		***************************************	***************************************													
Apartment [1]			***************************************													
Commercial [2]		***************************************	***************************************												2,906,013	
Commercial/Office [2]										<b>†</b>					261,360	
Commercial Stadium (seats) [3]			***************************************									<u> </u>			(85,000)	
Commercial/Recreation [2]			***************************************										<b>1</b>	1	328,006	
Hotel [2]							<b></b>			<u> </u>		İ				
Vehicle Related Comm. [2]										<b>-</b>		<b></b>			78,400	
Office [2]										<u> </u>			1			
Warehouse [2]					<u> </u>					1					-	
R&D		•••••													1,190,059	
Institutional										283,293						

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

**Table 5.3-3 Unspecified Projects in the City of Irvine** 

Land Use			Planning Area	31		Planning Area 32						
	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+		
Estate [1]								1		1		
Single Family Detached [1]						***************************************				***************************************		
Condominium [1]						***************************************						
Apartment [1]												
Commercial [2]	47,710				35,908	292,421				939,844		
Commercial/Office [2]												
Commercial Stadium (seats) [3]												
Commercial/Recreation [2]									***************************************			
Hotel [2]	76,359											
Vehicle Related Comm. [2]												
Office [2]	240,309	223,520				2,596,912						
Warehouse [2]	311,186					1,610,405						
R&D	446,000	626,380	1,906,156		870,090	314,492				1		
Institutional									1			

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

Table 5.3-3 **Unspecified Projects in the City of Irvine** 

Land Use			Planning Area 3	)			Plat	ning Area 34				Plan	ning Area	35	
	Existing Development as of 6/34/39	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/36/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Ver 2006	Year 2005	Year 2015	Year 2020+
Estate [1]								***************************************							
Single Family Detached [1]															
Condominium [1]															
Apartment [1]				······································				***************************************							
Military Housing []									··········						
Commercial [2]	499,547	157,673	643,195		92,560	149,220	213,767	400,000		579,443	100,595				
Auto Dealer [2]						84,882					172,033				489,691
Auto Repair [2]						35,643					79,973		<b></b>		
Commercial/Office [2]															
Commercial Stadium (seats)				****************				······································							
Commercial/Recreation [2]								······································					<u> </u>		
Health Club						41,067									
Hotel [2]	142,357	407,335									67,672				
Vehicle Related Comm. [2]								······································							
Office	2,323,185	646,365	1,000,000		3,997,703	213,275		······································			3,353,209	150,000			242,058
Warehouse	_					143,407					7,195,422	<b>}</b>		***************************************	
R&D							252,799	1,000,000		4,972,227	823,371	100,000		·····	
Institutional													<b>†</b>		

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

**Table 5.3-3 Unspecified Projects in the City of Irvine** 

Land Use			nning Area 38		•	.ty 01 11 ·		lanning Area 39	)	
	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+
Estate [1]										
Single Family Detached [1]	871				25					
Condominium [1]	608				396					
Apartment [1]	1,264				686					
Military Housing [ ]	552				701					
Commercial [2]										
Auto Dealer [2]										
Auto Repair [2]										
Commercial/Office [2]										!
Commercial Stadium (seats)										
Commercial/Recreation [2]										58,450
Health Club										
Hotel [2]										
Vehicle Related Comm. [2]										
Office										
Warehouse										
R&D								ŧ		
Institutional										

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

**Table 5.3-3 Unspecified Projects in the City of Irvine** 

Land Use			Planning Area	51		Planning Area 30 [4]							
	Existing Development as of 6/30/99	Year 2000	Year 2005	Year 2015	Year 2020+	Existing Development	Year 2000	Year 2005	Year 2015	Year 2020+			
Estate [1]													
Single Family Detached [1]					772								
Condominium [1]					176								
Apartment [1]			113		2,200								
Military Housing [ ]													
Commercial [2]													
Auto Dealer [2]							151,000			50,000			
Commercial/Office [2]			506,000		4,134,000								
Commercial Stadium (seats) [3]										85,000 [3]			
Commercial/Recreation [2]				-	519,000		41,000						
Health Club [2]													
Hotel [2]													
Vehicle Related Comm. [2]							_			ŀ			
Office [2]													
Warehouse [2]													
R&D [2]			1,576,000		6,603,000								
Institutional [2]		1			513,000		1,003,000			673,000			

- Number in dwelling units
   Number in square feet
   85,000 seats for the stadium shown, however, no square footage is included in Planning Area total.
   Portion of PA 30 within the area covered by the Millennium Plan.

#### IUSD - North Irvine Middle School

The proposed middle school is located south of Hicks Canyon Wash and east of Hicks Canyon Road in the City of Irvine. The school will be located on 20 acres and will accommodate 750 students. The school is in its pre-planning and pre-design phase. It is anticipated to open in Fall 2003 (IUSD, 1999).

## IUSD - Northwood High School

This high school is located at the northwest corner of the future intersection of Portola Parkway and Yale Avenue in the City of Irvine. The project is located on approximately 43 acres and ultimately will accommodate 2,400 students (The Planning Center, 1995). The Northwood High School project was approved in 1995 and was opened in September 1999 (IUSD, 1999).

## 5.3.2.4 City of Laguna Niguel Jurisdiction Land Use Projects

The City of Laguna Niguel requested that several projects within its jurisdiction be considered within the cumulative impact analysis of the Proposed Project (City of Laguna Niguel, 1999). These projects are shown in Table 5.3-2 and include:

- i) Alicia Skate Park, which is a new proposed park for skate boarding activities
- ii) Allen Oldsmobile/Cadillac Tentative Parcel Map 97-126, which is the expansion of an existing automobile dealership located at Camino Capistrano and Paseo De Colinas.
- iii) Enterprise Car Sales, which is a new used car sales lot and rental facility on 2 acres located at 28112 Camino Capistrano in Laguna Niguel.
- iv) La Paz Park, which is a new 6.3 acre public park with lighted sports fields.
- v) Mission Imports Remote Parking Lot, which is a new 2.4 acre parking lot located at 1 Star Drive which would provide a surplus parking area with night lighting.
- vi) Moulton Elementary Field Improvements, which would provide for the expansion of active recreational facilities and parking area at this elementary school.
- vii) St. Anne's School, which would provide for the expansion of an existing private school with the project to include 29,235 square feet of classroom building and 33,065 square feet of classroom/administration building at 32451 Bear Brand Road.
- viii) Shapell/Aliso Creek/Tentative Tract No. 15670, which is a 9 acre tract of single-family dwelling units. Forty-six homes will be built in this new tract of homes.
- ix) Shea Business Properties-Ocean Ranch II, which will provide 45,000 square feet of office dwelling and health club, located at 32451 Golden Lantern Street in the city.
- x) Tutor Time/Tentative Parcel Map No. 98-171, which is located at Aliso Creek Road, east of La Paz Road and will provide 12,000 square feet day-care facility.

However, these projects fall outside of the planning area associated with the Proposed Project. As a result, the potential for these projects to contribute to significant cumulative impacts in combination with the Proposed Project is considered to be minimal and unlikely. Consequently, they have not been included within this cumulative impact analysis.

# 5.3.2.5 City of Laguna Woods (formerly Leisure World) Jurisdiction Land Use Projects

The unincorporated community of Leisure World incorporated to form the City of Laguna Woods in March 1999. The City is currently considering the following projects.

## Moulton Parkway Commercial Development

This 20,178 square foot commercial project will provide sorting and distribution space for the Los Angeles Times. The project presently is slated to commence construction in Fall 1999 (letter from City of Laguna Woods dated 9/29/99).

#### Town Centre

This project includes 7 buildings that contain 253,270 square feet of space in the Town Centre commercial project. This project is located on 14 acres on El Toro Road. The project includes:

- a) 140 room (80,000 square feet) hotel
- b) 40,000 square feet of medical offices
- c) 7,000 square feet restaurant
- d) 45,570 square feet of garden office
- e) 68,000 square feet of offices
- f) 5,900 square feet of service retail
- g) 6,800 square feet of retail uses

The project is in building plan check, with construction anticipated to begin in early 2000.

# Home Depot Center

This project contains 215,773 square feet of commercial use. This project includes the following stores:

- a) Home Depot
- b) Office Max
- c) Rite Aide
- d) Stater Brothers
- e) In-line retail

- f) Bank
- g) Pad for future restaurant use

Most of the structures in this project have already been constructed and are in use.

## 5.3.2.6 City of Lake Forest

## Commercial Use on Portola Parkway

A conceptual site plan has been proposed for commercial uses of up to 110,000 sq. ft. on an approximately 9-acre parcel. The site is located on Portola Parkway, between El Toro Road and Saddleback Parkway, in unincorporated Orange County within the sphere of influence of the City of Lake Forest. The City of Lake Forest is processing this project which would be located on approximately 6 acres of land. The project includes a Site Plan/Use Permit (PA 95-0097) for the construction of a 40,000 sq. ft. office building and 7,000 sq. ft. office/food facility. The remaining 3 acres provide right-of-way for Portola Parkway. The project consists of several elements, including:

- i) General Plan Land Use Amendment on a 16-acre site of which 9 acres will be used for the project;
- ii) Zone Change from Planned Community Open Space to Community Commercial Sign Restriction; and
- iii) Community Profile Amendment from Open Space/Conservation to Community Commercial.

As of July 1999, the city was still reviewing a planning application for the site, but no development has been approved. (City of Lake Forest, 1999)

# Baker Ranch Planned Community Project

The County of Orange originally approved a Planned Community and Development Agreement for this commercial and industrial project in 1988. The project is located on 370 gross acres of land and is located south of the Foothill transportation corridor and west of the City of Lake Forest city boundary and east of Baker Parkway. The 1988 tentative tract map has expired and the property owner is currently processing a new tentative tract map and revised area plan for 4.3 million square feet of business park development on the project site with the City of Lake Forest. It is estimated that the project will be completed between 2000 - 2010. (Letter from City dated 9-27-99)

#### A.J. West Ranch

This project is located south of the Foothill transportation corridor and east of Baker Parkway. The project includes the development of 933,000 square feet of business park

development. This project would be developed between the year 2000 and 2005. (Letter from City dated 9-27-99)

## Saddleback Valley Church

This project is located on the northwest corner of El Toro Road and Portola Parkway and includes the development of 800,000 square feet of church related development. Some commercial development may also be included in this project. The County of Orange originally approved 310,000 square feet of development as part of a master plan for this Church. Some Church buildings have already been constructed on the project site. Precise development plans for the remaining 489,000 square feet have not been submitted to the City of Lake Forest. It is anticipated that this project will be built out by the year 2005. (Letter from City dated 9-29-99)

## 5.3.2.7 City of Newport Beach Jurisdiction Land Use Projects

## Holtze Extended Stay Hotel and Office Project

The Holtze Hotel and Office Project, located at 1300 Quail Drive and Dove Avenue in Newport Beach, is the location of a former automobile dealership. The Newport Beach City Council approved a Mitigated Negative Declaration (MND) for the 100,000 sq. ft. office building and the 350-bed extended stay hotel. The project was in the plan check stage, as of July 1999. The planned construction date is unknown. (City of Newport Beach, 1999)

# Koll Center Expansion

The Koll Center Expansion project would entail expansion of parking space and facilities associated with an existing 250,000 sq. ft., 10-story office tower in Koll Center Newport, located at the intersection of MacArthur Boulevard and Jamboree Road. The project includes demolition and reconstruction of an existing parking structure, as well as the addition of a new parking structure. The project involves a General Plan Amendment, an amendment to the Planned Community Development Plan (Zoning Document), and a Traffic Study pursuant to the Newport Beach Traffic Phasing Ordinance. The project is undergoing environmental review by the City, with an EIR to be distributed in January 2000. The project is tentatively scheduled to be reviewed by the City's Planning Commission in March 2000. The City Council is tentatively scheduled to consider the project in April 2000. (Letter from City of Newport Beach, Dated October 27, 1999)

## Rockwell Site Expansion

The project involves a General Plan Amendment and a Planned Community Amendment to allow for the development of an additional 566,000 sq. ft. (for a total of 1,008,000 sq. ft.) of allowable building area on the project site. The project was undergoing environmental review by the City, as of July 1999. (City of Newport Beach, 1999)

#### Amendment 891

This project is located on 412 acres of land at 5400 West Pacific Coast Highway. The proposed planned community would include the following uses:

- 1,750 dwelling units
- 75 hotel/motel units
- 70,000 sq. ft. of commercial uses
- 10-acre elementary school site
- 216 acres for parks, open space and habitat restoration

The project includes general plan amendment pre-zoning, pre-annexation zoning, a Local Coastal Permit, and a development agreement. The County of Orange is acting as the Lead Agency on the EIR currently being prepared on this project.

# Conerant Expansion Project, GPA 96-3(f)

This commercial project is located at 4311 Jamboree Road. The project includes an amendment to the City's General Plan and an amendment to the planned community covering the project site to allow for the development of an additional 566,000 sq. ft. of commercial space for a total of 1,008,775 sq. ft. of allowable building area on the project site.

A Draft EIR has been prepared on the project and is out for public review. Review of the project by the City's Planning Commission and City council has been tentatively scheduled for February 2000.

# 5.3.2.8 City of Santa Ana Jurisdiction Land Use Projects

#### First American Title

This project is part of the MacArthur Place project in the City of Santa Ana, and involves the construction of the new headquarters for First American Title. The project has recently been completed and included the construction of three separate office buildings at 1, 2 and 3 First American Way, near the intersection of MacArthur Boulevard and Main Street in the City of Santa Ana. The floor areas of the three buildings are 87,500 sq. ft. (3-story), 64,059 sq. ft.

(2-story) and 60,210 sq. ft. (2-story), respectively. The environmental review process for this project has been completed (City of Santa Ana, 1999b).

## MacArthur Place (Brea Properties)

MacArthur Place is a mixed-use development project that will involve the construction of retail space, condominiums, apartments and other uses. Located at the intersection of MacArthur Boulevard and Main Street in the City of Santa Ana, a portion of the 8-building development is in the plan-check stage of the development process. This portion includes 280 apartments, which would provide 29,419 sq. ft. of residential space. The environmental review process has been completed (City of Santa Ana, 1999a).

### Nexus Twin Towers (Nexus)

This project is also part of the MacArthur Place project. The Nexus Twin Towers project involves the development of an office building atop an existing parking garage, located at 1 East Majestic Drive near the intersection of MacArthur Boulevard and Main Street in the City of Santa Ana. The area of the office building, now in the plan check stage of the planning process, will be 456,414 sq. ft. The office and parking garage combined will be 676,780 sq. ft. The environmental review process is complete (City of Santa Ana, 1999b).

#### One Hutton Center

This project involves the construction of a proposed office building, consisting of 12 stories and a 13<sup>th</sup> level penthouse for mechanical equipment, a 240-room hotel, a 5,000 sq. ft. restaurant, a 4,740 sq. ft. conference hall, and a 7-story parking structure to accommodate 875 parking spaces. The project is located near the northeast corner of MacArthur Boulevard and Main Street. The environmental review was completed in July 1992. No construction has been scheduled, as of July 1999, and the Conditional Use Permit has been extended (City of Santa Ana, 1999b).

# 5.3.2.9 City of Mission Viejo Land Use Projects

# Los Angeles Athletic Company (LAACO) Project

The LAACO project is located north of El Toro Road and east of Marguerite Parkway. This project includes the development of 77,095 sq. ft. self-storage facility on a 3-acre site. The project was approved in June 1999. It is in the early stage of development. Construction is expected to commence on or before September 1999. (LAACO, 1999)

# Mammoth Equities Project

The Mammoth Equities project is located at the southeast corner of El Toro Road and Marguerite Parkway. The project includes the development of 49,000 sq. ft. of commercial

office space on an approximately acre site. However, only 2.5 acres is to be developed due to zoning restrictions. The project was approved in June 1999. As of July 1999, the project was in the plan check stage. (Mammoth Equities, 1999)

#### Painted Trails

This project is located north of the Foothill Transportation Corridor, east of El Toro Road and west of Upper Oso Reservoir. The project will accommodate 578 single-family dwelling units and several private parks. The project is covered by Final Program EIR for the Mission Viejo General Plan, which was certified by the Mission Viejo City Council in 1990. Portions of the project have been constructed, as of July 1999. (City of Mission Viejo, 1999)

## Stone Ridge

This project is located east of Olympiad Road between Alicia Parkway and Fieldcrest Street. The project will accommodate 468 single-family dwelling units, open space, and local parks on approximately 340 acres. The project is covered by previously certified Final Program Final EIR MV89-2 and the City of Mission Viejo General Plan Final Program EIR and Initial Study/Addendum IS GP91-1, which serve as the Program EIRs for the Stone Ridge project. Portions of the project have been constructed, as of July 1999. (City of Mission Viejo, 1999)

## Country Suites Hotel

This project is located east of Marguerite Parkway and west of the Foothill Transportation Corridor ramp at Los Alisos Boulevard. The project consists of a 118 room hotel. The project was approved by the City in December 1998 and is currently under construction. Occupancy of the hotel is expected in January 2000. (Letter from City of Mission Viejo dated 10-12-99)

# Shea Properties

This proposed project is located east of Marguerite Parkway and west of the Country Suites Hotel at 28401 Los Aliso Boulevard. The project includes 230 apartment units. The project was approved by the City in August 1998 and is currently under construction. The project should be ready for occupancy by May 2000. (Letter from City dated 10-12-99)

## Acero Street Office Project

This project is located on Acero Street in Mission Viejo. The project includes 225,000 square feet of office space. The project developer is currently discussing this project with the City. Should the project be approved by the City, project occupancy is anticipated to occur in January 2001. (Letter from City dated 10-12-99)

## Heritage Villas Senior Housing

This project is located on the southeast corner of Oso Parkway and Country Club Drive at 26836 Oso Parkway. The project includes 143 senior housing units. The project was approved by the City in October 1998 and is under construction. The project should be ready for occupancy in April 2000. (Letter from City dated 10-12-99)

## Camino Capistrano Mini-Storage

This project is located on 4.79 acres of land on Camino Capistrano, northerly of Crown Valley Parkway. The project includes 107,108 square feet of mini-storage space with a caretaker residence. The project is currently being considered by the City, with City action scheduled in October 1999. The project is scheduled for completion in February 2001. (Letter from City dated 10-12-99)

## Legacy Partners Office Building

This project is located at 27101 and 27201 Puerta Real Street in Mission Viejo. The project consists of 232,000 square feet of office space in two separate buildings. The project was approved by the City in June 1999. Preliminary site work has begun on this project. The project is scheduled for completion in October 2000. (Letter from City dated 10-12-99)

# Mission Viejo Mall Expansion

The Mission Viejo Mall has been expanded to include an additional 500,000 square feet of commercial space. The project was approved by the City in 1997, with build out expected in June 2000. The Mall has already been expanded by 250,000 square feet, with new Nordstrom and Saks 5<sup>th</sup> Avenue stores opened in Summer 1999. (Letter from City dated 10-12-99)

# Lexus Auto Dealership

This project is located north of Avery Parkway and east of Marguerite Parkway at 28242 Marguerite Parkway. The project consists of a 38,164 square foot auto dealership. The project was approved in June 1998, with the project scheduled for completion in September 2000. (Letter from City dated 10-12-99)

# 5.3.3 Transportation Land Use Projects

The projects related to transportation land use includes infrastructure improvement, airport expansion, a proposed high speed rail system, and a regional transit system being proposed by the Orange County Transit Authority. An overview of these projects, including identification of jurisdiction, are included in Table 5.3-4.

## 5.3.3.1 I-405/SR-55 HOV Access Improvements Project

Significant traffic congestion occurs along portions of Interstate 405 (I-405), State Route 55 (SR-55), and Bristol Street in the vicinity of the City of Costa Mesa, which impacts both regional mobility and local access. To reduce congestion and encourage the use of buses and high occupancy vehicles (HOV), the Orange County Transportation Authority (OCTA) has developed the I-405/SR-55 Transitway Project. Elements of the Transitway Project include:

- i) Direct HOV connector ramps between I-405 and SR-55
- ii) Access improvements along I-405
- iii) HOV drop-ramps at Bear Street
- iv) HOV drop-ramps at Von Karman Avenue

The focus of the project is construction of access improvements along I-405 in the City of Costa Mesa between Bristol Street and SR-55. A project to construct direct HOV connector ramps between I-405 and SR-55 was initiated in 1993 by OCTA under the "Measure M" freeway program, following approval of a Project Report (OCTA, 1993) and supported by the California Department of Transportation. The project is scheduled to begin construction in November 1999. (Caltrans, 1999)

# 5.3.3.2 Alton Parkway Extension

Alton Parkway is proposed as a major arterial highway with a 120-foot wide cross section, providing 6 travel lanes and a 14-foot median. The proposed 2.17-mile extension of Alton Parkway, from Irvine Boulevard to the Foothill toll road (FTR), is within the North El Toro Regional Circulation Area (County of Orange) and is part of the Foothill Circulation Phasing Program (FCPP). A portion of this proposed extension passes along the eastern boundary of the MCAS El Toro site. Although construction of the extension is not part of the Proposed Project, conveyance of right-of-way for this future FCPP improvement is planned as part of the Proposed Project. The proposed extension of Alton Parkway, along the eastern boundary of the MCAS El Toro site and through a portion of Lake Forest are the only unfinished segments of this master planned arterial. These proposed extensions are an important component of the FCPP. The FCPP is designed to phase and coordinate road development with building permits in southeast Orange County.

Table 5.3-4
Reasonably Foreseeable Probable Future Projects Related Transportation Land Use Projects

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
I-405/SR-55 HOV Access Improvements Project	Caltrans					X		Construction of access improvements within the City of Costa Mesa to include:  • Direct HOV connector ramps between I-405 and SR-55  • Access improvements along I-405  • HOV drop-ramps at Bear Street HOV drop-ramps at Von Karman Avenue		x	x	x		Construction to begin November 1999.
Alton Parkway Extension	County of Orange					X		Modification to Alton Parkway to include 120-foot wide cross section to provide 6 travel lanes and 14-foot median		X		x		Status Pending
Foothill Transportation Corridor (FTC) North and South	County of Orange					X		Approximately 30 mile connection between ETR Orange County inland foothills to I-5 near San Diego County.  North Segment: in the vicinity of MCAS El Toro (~7.5 miles)  South Segment: connects Oso Parkway to San Diego County line.		x		x		North segment completed April 1995.
LAX Expansion Project	City of Los Angeles					X		Expansion project for the LAX to provide medium- and long-range air transportation services through the following:  new terminal  4 new concourses  automated people mover system  new cargo facilities  redevelopment of the Century Cargo Complex.  Alternative 1:  new additional north runway  relocation of 3 existing runways  reconfiguration of north central terminal area  Alternative 2:  2 new runways  a new concourse	x					No environmental documentation prepared to date.
Proposed High Speed Rail Project	Southern California Association of Governments (SCAG)					Х		<ul> <li>\$6.2 billion project funded through Project California's private consortium (no public operating subsidy) to improve linkages to and between airports.</li> <li>Core System to connect Riverside, Orange, and Los Angeles counties along 1-5 from MCAS El Toro site to Anaheim, Union Station, and Palmdale through ETC to SR-91 and Riverside.</li> <li>Statewide System to connect in Palmdale and at Union Station in Los Angeles</li> </ul>	x					

## 5.3.3.3 Foothill Transportation Corridor (FTC) North and South

The FTC connects with the eastern leg of the ETC in Orange County's inland foothills and is ultimately planned to connect with I-5 near the San Diego County border. The total length of the FTC is approximately 30 miles, divided into the North and South Segments. The North Segment is in the vicinity of the MCAS El Toro site. Construction began on the North Segment in November 1990. The first section from Portola Parkway North to Portola Parkway South opened on October 16, 1993. The second section, which extended the FTC south to Antonio Parkway, opened on April 7, 1995. The completed section of the FTC is approximately 10.2 miles long. Construction of the section north of Portola Parkway North began in 1995 and is complete. The South Segment of the FTC from Oso Parkway to the San Diego County line is undergoing environmental review (TCA, 1999).

## 5.3.3.4 Los Angeles International Airport Expansion Project

The City of Los Angeles proposes to expand the Los Angeles International Airport (LAX). New features would include a new terminal and four new concourses, an automated people mover system, new cargo facilities, and redevelopment of the Century Cargo Complex. Two alternatives are also being proposed. Alternative 1 would include a new runaway to the north, relocation of three existing runways, and reconfiguration of the north central terminal area. The second alternative would include the construction of two new runways and a new concourse. Environmental review is pending.

# 5.3.3.5 Proposed High Speed Rail Project

A high-speed rail system is proposed by the Southern California Association of Governments (SCAG) on several freeway rights-of-way with limited stations to improve linkages to and between the region's airports. The core system would connect Riverside, Orange, and Los Angeles counties along I-5 from the MCAS El Toro site to Anaheim, Union Station, and Palmdale, as well as from the MCAS El Toro site through the Eastern Transportation Corridor to SR 91 and Riverside. This rail system also would connect with the statewide high-speed rail system in Palmdale and at Union Station in downtown Los Angeles. It would be constructed and operated through Project California, a private consortium, at a cost of \$6.2 billion, on public right-of-way, with no public operating subsidy.

# 5.3.3.6 Orange County Transit Authority Transit Project

A transit project between the Fullerton Transportation Center and the Irvine Transportation Center is currently being studied by the OCTA. The project could use a combination of at grade as well as elevated transit ways. An EIR was released in September, 1999, for public review and comments. The OCTA has decided to pursue further environmental analysis on the 28 mile Fullerton to Irvine Centerline, and to complete a supplemental environmental

analysis on the full 28 miles. In addition, environmental analysis will be prepared for the first phase minimum operating segment from Irvine to Costa Mesa. The OCTA Board will address the issue of preparing preliminary engineering and final environmental analysis in Summer of 2000.

# 5.3.4 Other Land Use Projects

Several projects fall outside the categories referred to previously, but are significant for evaluation under cumulative impact analysis. These projects are outlined in Table 5.3-5.

## 5.3.4.1 Natural Community Conservation Plan

The Natural Community Conservation Plan (NCCP) provides the planning framework to preserve natural biotic communities in Orange County. Since the Orange County Board of Supervisors has approved the reserve design, it helps establish a geographic context of preserved open space, which assists in determining the cumulative impacts of individual project actions.

In early 1991, the California State Resources Agency developed a concept for a multiple species habitat conservation plan. This pilot plan, which focuses on coastal sage scrub habitat, was developed in collaboration with a number of landowners. A primary element of the plan was establishment of a Scientific Review Panel (SRP), composed of five conservation biologists, which was to develop and oversee data collection procedures and to formulate regional and subregional goals and criteria for future reserves. Local jurisdictions are responsible for developing local area NCCPs to be approved by the United States Department of the Interior, Fish and Wildlife Service (USFWS). A number of Orange County landowners and the County of Orange assembled data collected in 1992 on coastal sage scrub species, according to guidelines released by the SRP in December 1991, and developed the Orange County NCCP. A total of approximately 78,600 acres are enrolled countywide by public and private landowners, compared to a total area of 393,655 acres in southern California south of metropolitan Los Angeles.

The Reserve Design, which is the essential premise of the NCCP and identifies lands to be set aside, was prepared on April 22, 1994 which included special provisions for the Central Coastal Subarea in Orange County. The NCCP was adopted by the County of Orange in April 1996 upon certification of its EIR for the project, which was not challenged in court. The federal and state governments, various municipalities and landowners entered into an implementation agreement for the NCCP shortly thereafter. In general, the Central subarea Reserve Design incorporated committed open space along with areas contemplated as open space in conjunction with the approval of certain development projects in other areas. This

Table 5.3-5
Reasonable Foreseeable Probable Future Projects Other Related Land Use Project

Project	Jurisdiction	Open Space	Residential	Commercial	Industrial	Transportation	Other	Brief Project Description And Building Inventory	Proposed	Approved	Pre-design	In Progress	Complete	Comments
Central/Coastal Natural Community Conservation Plan	County of Orange	x					x	Approximately 78,600 acres enrolled by public and private landowners of the 393,655 total acreage located south of metropolitan Los Angeles.				x		Approved in 1991 On-going
MCAS Tustin Reuse Plan	City of Tustin		x	x	X	:		Conceptual land use plan containing a variety of uses including residential, commercial, office and industrial uses				x		EIR/EIS to be considered for approval in early 2000
James A. Musick Facility	County of Orange						X	Expansion of existing County Jail facility located on Irvine Boulevard to provide:  additional 20,000-sq. ft. sheriff's station  relocation of interim care facility  increase inmate capacity from 1,256 to 7,584 inmates  extension of Alton parkway to site entrance.		x				Approved by the County in October 1998 – currently under litigation

open space system has been augmented by adding reserve areas known to contain substantial populations of the California gnatcatcher and cactus wren, and to provide linkages of natural habitat. The Reserve Design incorporates over 21,000 acres of coastal sage scrub and its matrix of other associated habitats, including lands necessary for connectivity (i.e. contiguous open space to promote genetic exchange). Existing, planned and/or proposed regional open space lands in the Central Subarea include a total of 8,379 acres of coastal sage scrub in Weir Wilderness Park, Santiago Oaks Regional Park, Irvine Regional Park, Open Space Area 31 in Gypsum Canyon, Peters Canyon Regional Park, the Loma Ridge open Space system, miscellaneous open space associated with the East Orange Planned Community and Limestone-Whiting Wilderness Park. These areas are generally northwest, north and northeast of the MCAS El Toro site. A 998-acre habitat reserve has been identified on the MCAS El Toro site in the Reserve Design. This 998-acre habitat reserve would partially implement the adopted NCCP in Orange County as it is a portion of the total reserve design for the subarea. Also, see the discussion in Section 4.9 (Biological Resources) for more information on the NCCP.

The County is currently in the process of preparing a Southern Region NCCP which will cover additional lands in South County. The area covered by this plan will be added to the previously approved Central Coastal HCCP and provide additional land for the preservation of natural biotic communities in Orange County.

#### 5.3.4.2 MCAS Tustin Reuse Plan

The City of Tustin has conducted comprehensive planning for future reuse of the MCAS Tustin facility, which has been identified for closure by the Federal government. A Reuse Study has been completed which identifies a long-range land use and transportation plan for redevelopment of the Base. A task force of elected officials and representatives from various area jurisdictions assisted in the planning process and adopted a conceptual land use plan. This plan shows a variety of uses including residential, commercial, office and industrial uses. The City identified a Preferred Alternative, the Local Reuse Alternative (LRA) for the reuse of the Base that addresses issues, opportunities and constraints associated with the site and that attempts to be practical in economic terms and to respond to future opportunities (City of Tustin, 1999a).

A joint Environmental Impact Report/Environmental Impact Statement (Final EIR/EIS) for the reuse of MCAS Tustin is scheduled to be completed in February 2000. The federal government is currently scheduled to consider formal approval of the MCAS Tustin Reuse Plan in early 2000. The City Council of the City of Tustin will consider approval of a specific Plan for the site at that time (City of Tustin, 1999b).

## 5.3.4.3 James A. Musick Facility

The James A. Musick Facility is a County-operated detention facility located southeast of the future extension of Alton Parkway and northwest of existing Bake Parkway, in unincorporated Orange County. The facility is in the sphere of influence of the City of Irvine, and is immediately adjacent to the City of Lake Forest. The municipal boundary of the City of Irvine borders the property on the south/southwest.

The James A. Musick facility is being considered for expansion to 7,584 inmates, with a full range of classifications, from minimum to maximum-security. Inmate bookings and releases are proposed to be handled at the facility for a full classification of inmates to assist in handling a shortfall in capacity for incarceration facilities in Orange County. This project is proposed to include a 20,000 sq. ft. Sheriff's Station on the site, and could include an Interim Care Facility (ICF) in an area adjacent to the substation. The Sheriff's Station would provide improved law enforcement services to the area. The ICF is a residence facility operated by the County Mental Health Board for young people unable to function in a foster or group home or Juvenile Hall due to emotional or psychiatric instability. The youths are confined in the home and are not free to come and go.

Alton Parkway will be extended northeast from Irvine Boulevard/Trabuco Road to the site entrance. Although the Musick Drive entrance may still be used for inmate buses and deliveries, the primary traffic impact of the facility, staff and visitor access will be absorbed on the Alton Parkway extension. Bake Parkway will be used to access the Sheriff's Station. The County originally approved the project in November 1996 and re-approved in October 1998. However, the project is in litigation at this time (Orange County, 1999a).

# 5.3.5 Non-Planned Land Use Projects

A general assessment was made as part of FEIR No. 563 of areas currently designated for agricultural and open space uses, which are adjacent to the MCAS El Toro site and within the PIL. These areas are referred to as non-planned areas. It is conceivable that these areas could convert to other uses regardless of the disposition of MCAS El Toro. Land uses anticipated to be developed between the year 2005 and year 2020 were included as future prospects in the OCP-96M. Land uses anticipated to be developed beyond the year 2020 were previously estimated by the Irvine Company. According to the Irvine Company, potential land uses for these areas may include residential, industrial and institutional uses. However, at the time specific project proposals are submitted for consideration, environmental documentation must be prepared for these projects.

**Cumulative Impacts** 

### 5.4 CUMULATIVE IMPACTS BY RESOURCE AREA

This section analyzes potential cumulative impacts to the environment that could be associated with implementation of the Proposed Project in concert with cumulative development and growth, including the above-listed past, present, and reasonably foreseeable future projects. For the cumulative impact analysis, OCP-96M was used to assess population, housing and employment projections for the MCAS El Toro and JWA site vicinities, as well as surrounding areas of the County. The County reviewed and determined that the cumulative projects were included in OCP-96M. These data then were used to evaluate the potential for cumulative impacts as a result of implementing the Proposed Project. See Table 5.4-1 for a summary of cumulative impacts by resource areas.

The threshold of significance used to determine whether the cumulative projects considered would create a significant impact on the environment was taken from the CEQA Environmental Checklist, and is as follows:

(i) Does the project have impacts that are individually limited but cumulatively considerable? Cumulatively considerable means that the incremental effects of an applicable project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects

The threshold of significance for each impact category is provided separately below.

# 5.4.1 Potential Cumulative Impacts Related to Land Use

The threshold of significance used to determine cumulative land use impacts is as follows:

(i) Does the project have land use impacts that are individually limited, but cumulatively considerable?

## 5.4.1.1 Cumulative Impacts

FEIR No. 563 found that the CRP project would have significant unavoidable adverse cumulative land use impacts because it would add to the overall urbanization of Orange County and the area around MCAS El Toro. However, under the Proposed Project, the non-aviation land uses have been reduced substantially in intensity, as compared to the CRP, consisting now primarily of parks, open space, and golf courses, and approximately 87 acres of business park development (Table 3-7). Under existing conditions, approximately 2,130 acres of land are devoted to open space land uses, including agriculture, golf course, recreation, restricted open space, and the open space surrounding the ordnance storage and disposal site. Under the Proposed Project condition, approximately 2,056 acres of land

Table 5.4-1
Cumulative Impacts With Cumulative and Proposed Projects

<del> </del>		T		······						,	····	7******				
Project	Land Use	Transportation & Circulation	Noise	Air Quality	Topography	Geology and Soils	Water Quality	Biological	Public Service/Utilities	Natural Resources/Energy	Aesthetics	Cultural Resources	Recreation	Public Health & Safety	Hazardous Materials	Socioeconomic
405/SR-55 HOV Access																
Improvements																
Amendment 891	~~~~~~~~~	U	U	U			U	***************************************		U	***************************************		***************************************	U		
A.J. West Ranch	······································	1		YES					***************************************		······································	<b>1</b>	***************************************			
Alton Parkway Extension				YES					······································		***************************************		***************************************			
Baker Ranch Park				YES							***************************************					YES
Camino Capistrano Mini- Storage				YES												
Country Suites Hotel	······································	YES	***************************************	YES							***************************************					
Conerant Expansion Project		YES	***************************************	YES			YES			1						
Foothill Aliso Commercial Center	<u></u>			YES									-			YES
Foothill Ranch Planned Community				YES												
Foothill Transportation Corridor				YES												
GPA/Zone Change for 440 Acres South of MCAS El Toro		YES1		YES			YES¹			YES1					_	YES
Heritage Villas Sr. Housing				YES		}										
High Speed Rail (SCAG)										YES						
Holtze Extended Stay Hotel and Office Project				YES												
Home Depot Center		YES		YES					<u> </u>	***************************************						YES
James A. Musick Facility		YES		YES			<b>T</b>			YES						
Koll Center Expansion		U	U	U			U			U		1		U		U
LAACO Project				YES						1			[		<b> </b>	1
LAX Expansion Project		U	U	U			U			Ū				U		YES
Legacy Partners Office Bldg.				YES					1							YES
Triangue de la companie de la compan		<del></del>						***************************************								

These projects are not compatible with and would not be constructed with the Proposed Project, and thus the Proposed Project causes no cumulative impacts with these competing alternative projects.

#### Legend

- No significant impact after mitigation

YES - Unavoidable potentially significant adverse impact

U - Unknown

Project	Land Use	Transportation & Circulation	Noise	Air Quality	Topography	Geology and Soils	Water Quality	Biological	Public Service/Utilities	Natural Resources/Energy	Aesthetics	Cultural Resources	Recreation	Public Heatth & Safety	Hazardous Materials	Socioeconomic
Lower Peters Canyon Specific Plan				YES			YES			YES						
MacArthur Place	ļ <u>.</u>			YES									ļ			
Mammoth Equities Project		-		YES						<del>                                     </del>			ļ		-	
MCAS Tustin Reuse Plan	<del>                                     </del>	1-		YES			YES							<u> </u>		<u> </u>
Mission Viejo Mall Exp.				YES			IES					-	ļ			YES
Moulton Parkway Commercial	<del>                                     </del>	<del>                                      </del>			_					<del>                                     </del>			<u> </u>			<del> </del>
Development		YES		YES	_			:								YES
Natural Community				_												
Conservation Plan																
Nexus Twin Towers				YES	_											YES
North Irvine Middle School				YES						YES			Ì			
Northwood High School				YES						YES						
Oak Creek Project				YES			YES			YES						
OCTA Centerline				YES												
One Hutton Center				YES												YES
Painted Trails				YES												
Portola Hills Planned Community				YES			YES									
Revised PAS 51, 35 & 30, Annexation, GPA, Pre-Zone, Zone Change		YES¹	YES¹	YES¹	_		YES¹	YES <sup>1</sup>		YES						
Rockwell Site Expansion				YES												YES
Saddleback Meadows				YES												
Saddleback Valley Church			YES	YES												
Sakioka Property				YES						YES						YES
Segerstrom Home Ranch		1		YES			<del> </del>			YES		1				YES
Spectrum Housing GPA/2C	<del>                                     </del>	U	U	U		<del>                                     </del>	U	U		U	<del>                                     </del>	<del>                                     </del>	<del>                                     </del>	U		U
<del></del>					<u></u>			_								

These projects are not compatible with and would not be constructed with the Proposed Project, and thus the Proposed Project causes no cumulative impacts with these competing alternative projects.

#### Legend

- No significant impact after mitigation
- Unavoidable potentially significant adverse impact YES

Ū - Unknown

Project	Land Use	Transportation & Circulation	Noise	Air Quality	Topography	Geology and Soils	Water Quality	Biological	Public Service/Utilities	Natural Resources/Energy	Aesthetics	Cultural Resources	Recreation	Public Health & Safety	Hazardous Materials	Socioeconomic
Stone Ridge		YES							<u> </u>	***************************************						
Shea Properties Apts.		YES		YES							······································					]
Town Centre	***************************************	YES	U	U			U		·	U	******			U		U
West Park Project	***************************************	YES							<u> </u>	YES	······					
Unspecified Projects in Irvine <sup>2</sup>	***************************************	YES <sup>2</sup>	YES <sup>2</sup>	YES <sup>2</sup>						YES <sup>2</sup>						YES <sup>2</sup>

To the extent these impacts are attributable to alternatives to the Proposed Project, they do not constitute cumulative impacts with the Proposed Project.

Legend

- No significant impact after mitigation

YES

- Unavoidable potentially significant adverse impact

U - Unknown

would be allocated to open space uses including agriculture, golf course, recreation, restricted open space, and habitat areas, which represent a one percent reduction in open space. However, the project would have a significantly beneficial cumulative impact because: (a) recreational uses would be increased and open to the public (i.e., golf courses, regional park); and (b) open space areas would be developed and maintained for habitat values, not for military uses. Under existing conditions, recreational uses were generally limited to military personnel with the exception of limited public access to the stables and golf course. Wildlife habitat areas were not managed for wildlife purposes and were not enrolled in programs such as the NCCP. In summary, the project would not add significantly to the urbanization, and the project recreation and open space components would mitigate, to a degree, the urbanization of cumulative development. Therefore, the Proposed Project would not have a significant adverse cumulative impact related to increased urbanization.

Similar to the FEIR No. 563 CRP project, under the Proposed Project, the land uses restricted by the military 65 CNEL dB PIL are anticipated to be converted to different, less restrictive uses for those portions where the PIL and the project 65 dB CNEL contour no longer overlap. Since the announced closure of MCAS El Toro, two proposals for conversion of land classified business park/town center commercial to residential uses have been considered by the Board of Supervisors. The Board denied a proposal for residential uses in the Portola Hills Planned Community, and approved a proposal to reallocate a maximum of 1,800 dwelling units to Planning Areas 40, 52, and 53 in the Aliso Viejo Planned Community. Note, however, that this was a reallocation of dwelling units from other planning areas, so there was no net increase in dwelling units. In addition, the City of Mission Viejo approved a reallocation of a maximum of 825 dwelling units to replace planned light industrial uses in Planning Area 11 in the Mission Viejo Planned Community. No other conversions of nonresidential uses to noise sensitive uses within the MCAS El Toro 65 dB CNEL contour have been approved.

There appear to be no additional sites where conversion of existing planned uses are contemplated south or east of the El Toro site. However, proposals for residential development are anticipated to the west and southwest of the El Toro site in the City of Irvine's Planning Area 33 (approximately 1,200 dwelling units), Planning Area 6 (approximately 5,780 dwelling units), and Planning Area 9 (approximately 8,900 dwelling units) as a result of the conversion of MCAS El Toro to civilian uses and the proposed changes in the 65 dB CNEL contour PIL. These conversions are also a result of the proposed reduction in the land affected by the military runway protection zones (i.e., APZ I and APZ II) compared to the proposed project safety zones. The reduction in the runway protection zone area permits residential uses where the land is also outside the OCX 65 dB CNEL contour, or an intensification of commercial and industrial occupancies (e.g., Baker Ranch Planned Community) where the land would be within the OCX 65 dB CNEL contour. Intensification is anticipated in Planning Areas 13, 32 to 35, and 39 in Irvine and Baker Ranch in the City of Lake Forest. However, no change in the land use is anticipated. The net increase in residential development in Planning Areas 6, 9, and 33 is, however, unknown at this time because of the preliminary information available. That is, it is unknown if these

dwelling units represent increases in the City of Irvine's General Plan, the County's General Plan, or the OCP-96M residential allocations, or if these dwelling units are reallocations of dwelling units from other planning areas or community analysis areas. According to preliminary information, the residential units would be (a) reallocated within the OCP-96M forecasts and (b) located outside the OCX 65 dB CNEL contour and runway protection zones. The residential units would require General Plan and zoning amendments and related CEQA documents. The County and City General Plans include Growth Management Elements, public services and facilities phasing policies, ordinances, and land use standards which, when applied to development proposals, are anticipated to reduce land use impacts to a level of insignificance.

The Irvine General Plan includes post-2020 development (Table 5.3-3) not included in the year 2020 OCP-96M. However, this development is not allocated to specified projects; therefore, the land use impacts, if any, are speculative for purposes of this section.

#### 5.4.1.2 Conclusions

Portions of the existing agricultural uses will be retained on the MCAS El Toro site as part of the Proposed Project. Agricultural uses are compatible with the proposed aviation and non-aviation uses. Therefore, there will be no significant cumulative impacts of the Proposed Project from agricultural operations. Section 5.4.11 addresses potential cumulative impacts related to loss of farmlands.

In the JWA area, the Proposed Project would have no land use compatibility impacts since the service level at the airport would be reduced from the current level, and the surrounding area is almost fully developed. No reduction in the PIL is proposed. Therefore, the project would not result in cumulative impacts on land use.

In conclusion, (a) the project would result in fewer cumulative land use impacts than existing forecasts which are based on the CRP; (b) changes in the project since EIR No. 563 have reduced the cumulative impacts of the project on increased urbanization to a level of insignificance; and (c) the project would reduce off-site aircraft restrictions that would permit up to 15,880 dwelling units to be reallocated to Irvine Planning Areas 6, 9, and 33. These impacts are anticipated to extend adjacent land uses and, therefore, would not divide existing communities, result in land use incompatibilities, or impair emergency response plans.

# 5.4.2 Potential Cumulative Impacts Related to General Plan Consistency

The threshold of significance used to determine cumulative General Plan consistency impacts is as follows:

(i) Does the project have General Plan consistency impacts that are individually limited, but cumulatively considerable?

## 5.4.2.1 Cumulative Impacts

Any effects of General Plan inconsistencies are either 1) direct effects, limited to the affected jurisdiction requiring a General Plan Amendment; or 2) indirect effects, which are generally secondary land use effects addressed in Section 4.1, Land Use, of this EIR. Implementation of the Proposed Project will require an amendment to several Orange County General Plan Elements at the time of project approval. These amendments will occur prior to project implementation or prior to development applications for off-site lands; therefore, there are no project related effects relating to General Plan consistency.

As analyzed in Section 5.4.1, the project would reduce the land area within the aircraft 65 dB CNEL contour and the runway protection zones (e.g., the Military Accident Potential Zones I and II) compared to the existing military AICUZ Study policies. This reduction would permit noise sensitive land uses to be proposed inside the military 65 dB CNEL contour PIL (but outside the OCX 65 dB CNEL contour) and would permit higher intensity occupancies (e.g., employees) at nonresidential uses within the military APZ I and APZ II zones (but outside the OCX safety zones). As noted in the previous section, the City of Mission Viejo and the County have approved residential reallocations in (respectively) Mission Viejo Planned Community Planning Area 11 and Aliso Viejo Planned Community Planning Areas 40, 52, and 53 within the military 65 dB CNEL contour PIL, but outside the OCX 65 dB CNEL contour PIL.

These approvals (and the denial of the Portola Hills Planned Community proposal) appear to represent the extent of conversions of land uses to the south and east of the El Toro site. However, the City of Irvine has initiated a proposal to permit approximately 1,200 dwelling units in Planning Area 33, and applications are anticipated for up to 5,780 dwelling units in Planning Area 6 and approximately 8,900 dwelling units in Planning Area 9 of the Irvine General Plan, which are located within the military 65 dB CNEL contour PIL. In addition, development applications are anticipated in Irvine Planning Areas 13, 32 to 35, and 39, and for Baker Ranch in Lake Forest to intensify occupancies within the former military APZ I and APZ II. These proposals would require amendment of the AELUP and the City and/or County General Plans to replace the military PIL and APZ I and APZ II policies with the OCX 65 dB CNEL contour and OCX runway protection zones.

#### 5.4.2.2 Conclusions

The City and County General Plans include Growth Management Elements, policies, and development standards that will ensure that development proposals are compatible with the respective General Plans. In addition, each project will be required to comply with CEQA. Based on these facts, no significant cumulative impact is anticipated regarding General Plan consistency.

# 5.4.3 Transportation and Circulation

The threshold of significance used to determine cumulative transportation and circulation impacts is as follows:

i) Does the project have transportation and circulation impacts that are individually limited, but cumulatively considerable?

## 5.4.3.1 Cumulative Analysis Methodology and Results

Section 4.3 includes a cumulative analysis of the traffic impacts of the project and OCP-96M development forecasts for the County. Refer to Section 4.3 for the year 2020 cumulative effects.

Comments on the NOP (e.g., pg. 805, No. 6) request that this EIR analyze the traffic impacts of development which exceeds OCP-96M. In response, the following analysis was completed of an intensified land use scenario for the vicinity of the El Toro site and JWA. To avoid conflict with the adopted OCP-96M forecasts, the scenario is designated "post-2020." The post-2020 cumulative setting also analyzes full development of the Proposed Project and its associated site access plan, as well as implementation of the circulation system improvements which are necessary to mitigate the year 2020 traffic impacts. The traffic conditions are analyzed herein based on the Table 5.1-2 in the traffic analysis study area (Figure 4.3-1).

The analysis involved preparing vehicle traffic forecasts and levels of service which were utilized to identify circulation deficiencies. A traffic shares analysis was then performed for each location where non-committed improvements are required (for example, according to the MPAH) or where deficiencies are forecast in order to identify those locations that would carry a significant amount of project traffic under cumulative post-2020 conditions (refer to Section 15.0 in the 1999 Traffic Analysis Technical Report for detailed summaries of the traffic volumes, levels of service and traffic shares for intersections and arterial roadways within the traffic analysis study area and Section 15.0 in the 2001 Traffic Analysis Technical Report Addendum for detailed summaries of the traffic volumes, levels of service and traffic shares for freeway/transportation corridor mainline segments and ramps within the traffic analysis study area).

Table 5.4-2 summarizes the non-committed but planned Master Plan of Arterial Highways improvements for the locations that are forecast to carry a significant amount of project traffic. Table 5.4-3 summarizes unplanned improvements which, if implemented, would improve the levels of service at the locations which are forecast to operate deficiently and to carry a significant amount of project traffic.

Mitigation measures are presented below identifying the project's responsibility to participate in the long-range implementation of these non-committed circulation improvements which are either currently planned or which are needed to address deficiencies at locations where the project is forecast to contribute significant levels of traffic.

In conclusion, all the post-2020 impacts can be accommodated by the existing Master Plan of Arterial Highways except 1) Moulton Parkway from Ridge Route to Glenwood and 2) Laguna Canyon Road from Bake Parkway to Santa Maria. The project impact on these highway segments is 3% and 4%, respectively, and the improvements would be required without the project.

With the exceptions of Moulton Parkway (Ridge Route to Glenwood) and Laguna Canyon Road (Bake Parkway to Santa Maria), the planned and unplanned improvements shown in Tables 5.4-2 and 5.4-3 will be implemented as development projects are entitled by the County and Cities per their General Plans. The County and City Growth Management Plans, and Measure M/CMP require that conditions be applied to projects, fee programs be established, and/or public works programs be amended as growth and development proceed to meet Measure M/CMP and General Plan level of service policies. Therefore, no new or additional ordinances, regulations or conditions of approval would be required to implement the improvements with the two exceptions noted. Mitigation measures are proposed to ensure that the post-2020 impacts are addressed.

# 5.4.3.2 Mitigation Measures For Post 2020 Impacts

T-12 Prior to issuance of the first building permits for each phase of the Airport System Master Plan development, the County will determine if the phase's traffic plus cumulative development traffic requires any improvement listed in Tables 5.4-2 and 5.4-3. If any improvement is required, the County will enter into a cooperative agreement with the lead local jurisdictions responsible for the cumulative development impacts to participate on a fair share basis in the implementation of the planned and unplanned long-range circulation improvements listed in Table 5.4-2 and 5.4-3. The agreements will commit the County to participate in the implementation process (i.e., the establishment of funding mechanisms, the preparation of design plans, the performance of feasibility assessments, etc.) based on the project fair share percentages identified in Tables 5.4-2 and 5.4-3. Where the County is the lead, the County will prepare cooperative agreements and use its best efforts to obtain fair share participation by non-lead local jurisdictions.

Table 5.4-2
Post-2020 Non-Committed, Planned (MPAH) Improvements

Location	Jurisdiction	Improvement	Project Share
FREEWAY/TRANSPORTATION CO.	RRIDOR IMP		
Eastern Toll Road (ETR) East Leg (SR-	1	Convert to a toll-free facility and improve to	6%
91 to ETR West Leg)	Caltrans and	provide two high occupancy vehicle (HOV)	
	TCA, Planned	lanes	
	(MPAH)		
ETR East Leg (ETR West Leg to 1-5)		Convert to a toll-free facility and improve to	13%
	Caltrans and	provide two HOV lanes and interchanges at	
	TCA, Planned	Culver Dr. and Jeffrey Rd.	
1-405 at Von Karman	(MPAH)	Company HOV many	5%
1-405 at von Karman	Caltrans	Construct HOV ramps	376
ARTERIAL IMPROVEMENTS	Canrais		
	C	Communication 1	2%
Aliso Creek (Laguna Canyon to El Toro)	County	Construct as four-lane secondary arterial	
Culver (ETR to Santiago Canyon) Irvine (PA2 East Access Rd to Alton)	County County	Construct as six-lane major arterial Improve to six-lanes	2% 32%
Jeffrey (ETR to Santiago Canyon)		1 1	32%
	County	Construct as four-lane primary arterial	5% 6%
Jeffrey (Irvine to Portola)	County	Improve to six-lanes	3%
Jeffrey (Portola to ETR)	County	Construct as four-lane primary arterial	3%
Portola (FTR to Alton)	County	Construct as four-lane primary arterial  Construct as four-lane secondary arterial	
Santa Maria (west of Moulton to Laguna Canyon)	County/ Laguna Hills	Construct as four-lane secondary arterial	2%
Trabuco (Jeffrey to Sand Canyon)	County	Improve to four-lanes	10%
Irvine (Yale to Jeffrey)	County/Irvine	Improve to six-lanes	7%
Portola (Culver to Jeffrey)	County/Irvine	Improve to six-lanes	7%
Portola (ETR to Culver)	County/Irvine	Improve to six-lanes	4%
Sand Canyon (Irvine to 1-5)	County/Irvine	Improve to six-lanes	14%
El Toro (Aliso Creek to SJHTC)	County/	Improve to six-lanes	2%
	Caltrans		
	Laguna		
	Beach		
Portola (FTR to El Toro)	County/ Lake Forest	Improve to eight-lanes	2%
Chapman (Jamboree to Santiago	County/	Improve to six-lanes	3%
Canyon)	Orange		
Bake (Lake Forest to Laguna Canyon)	Irvine	Construct as six-lane major arterial	3%
(Old) Laguna Canyon (Alton to Lake	Irvine	Improve to four-lanes and realign to Lake	1%
Forest)		Forest Dr	
Laguna Canyon (I-405 to Lake Forest)	Irvine	Improve to six-lanes	4%
Lake Forest (Bake to Laguna Canyon)	Irvine	Construct as four-lane primary arterial	2%
Michelson (Sand Canyon to Laguna	Irvine	Construct as four-lane secondary arterial	1%
Canyon)			
Sand Canyon (I-5 to Oak Canyon)	Irvine	Improve to six-lanes	15%
Sand Canyon (I-405 to Michelson)	Irvine	Construct as six-lane major arterial	1%

Location	Jurisdiction	Improvement	Project Share
ARTERIAL IMPROVEMENTS (cont)			
University (Michelson to I-405)	Irvine	Improve to six-lanes	5%
Yale (I-405 overcrossing)	Irvine	Construct two-lane overcrossing	5%
Ridge Route (west of Moulton to Bake)	Irvine/	Construct as four-lane secondary arterial	4%
	Laguna Hills		
Ridge Route (I-5 overcrossing)	Laguna Hills/	Construct four-lane overcrossing	6%
	Laguna Woods/		
	Lake Forest		
Ridge Route (Moulton to Avd. Carlota)	Laguna Hills/	Improve to four-lanes	1%
-	Laguna Woods		
Los Alisos (Muirlands to Avd. Carlota)	Laguna Hills/	Improve to six-lanes	4%
	Lake Forest/		
	Mission Viejo		
El Toro (north of Trabuco)	Lake Forest	Improve to six-lanes	4%
El Toro (Trabuco to I-5)	Lake Forest	Improve to eight-lanes	5%
Chapman (west of Newport)	Orange	Improve to six-lanes	1%
Jamboree (Tustin Ranch to Chapman)	Orange/	Improve to six-lanes	2%
	Tustin		
Irvine (Prospect to Newport)	Tustin	Improve to six-lanes	2%
Red Hill (north of Bryan to El Camino	Tustin	Improve to six-lanes	2%
Real)			
INTERSECTION IMPROVEMENTS		<del></del>	
Culver & Michelson	Irvine	Add second NB left-turn lane	1%
Jamboree & Michelson	Irvine	Add third WB and EB through lanes and free	3%
		NB right-turn lane	
Bake & Irvine/Trabuco	Irvine/	Add third NB through lane	17%
	Lake Forest		
Prospect & Irvine	Tustin	Add second SB left-turn lane	1%
Abbreviations: NB - northbound	EB – ea	stbound	***************************************
SB – southbound	WB - w	restbound	

Table 5.4-3
Post-2020 Unplanned Mitigation Improvements

Location	Jurisdiction	Improvement	Project Share
DEFICIENT INTERSECTIONS	<b>'</b>	<u>-</u>	
Aliso Creek & Laguna Hills	County	Convert WB shared second left-turn/third through lane to second left-turn lane, add third WB through lane and eliminate east-west split phasing	2%
Laguna Canyon & Santa Maria	Caltrans/ Laguna Woods	Add third NB through lane	4%
Moulton & Glenwood	County	Add fourth NB through lane	3%
Moulton & Laguna Hills	County	Add fourth NB through lane	3%
Culver & Michelson	Irvine	Add second EB through lane	1%
Irvine Center & Bake	Irvine	Add fourth NB through lane	4%
Jeffrey & Alton	Irvine	Add third EB and WB through lanes	4%
Jeffrey & Irvine Center	Irvine	Add free WB right-turn lane	4%
Jeffrey & Walnut/I-5 SB	Irvine	Convert WB through lane to shared through/second right-turn lane	3%
Laguna Canyon & Bake	Caltrans/Irvine	Add third SB through lane	3%
Sand Canyon & I-5 NB Ramps	Irvine	Convert EB left-turn lane to shared second left-turn/second right-turn lane [1]	
Sand Canyon & Irvine Center	Irvine	Add fourth WB through lane	11%
Sand Canyon & Oak Canyon	Irvine	Reconstruct WB approach to provide a left- turn lane, a right-turn lane and a shared through/second right-turn lane	14%
Technology & Barranca	Irvine	Add SB right-turn lane and convert second SB through lane to shared second through/ second right-turn lane	2%
Von Karman & Michelson	Irvine	Add third SB through lane and second SB left- turn lane	3%
Yale & Irvine	Irvine	Add second SB and EB left-turn lanes	4%
Yale & Walnut	Irvine	Add second EB left-turn lane	3%
Bake & Jeronimo	Irvine/ Lake Forest	Add second NB left-turn lane	9%
Irvine Center & Lake Forest	Irvine/ Laguna Hills	Add fourth NB through lane	3%
Lake Forest & Avd. Carlota	Irvine/ Laguna Hills	Convert second WB right-turn lane to shared second left-turn/second right-turn lane and add second EB right-turn lane and NB right-turn lane	3%
El Toro & Avd. Carlota	Laguna Hills	Convert NB right-turn lane to shared fourth through/right-turn lane and convert WB through lane to shared through/second right-turn lane	4%
Moulton & Alicia	Laguna Hills	Add fourth NB through lane	3%

Location	Jurisdiction	Improvement	Project Share
DEFICIENT INTERSECTIONS (cont)		***************************************	.1
Laguna Hills & Paseo Valencia	Laguna Hills/	Add EB right-turn lane and convert third EB	4%
	Laguna Woods	through lane to shared third through/second right-turn lane	
Moulton & Ridge Route	Laguna Hills/ Laguna Woods	Add fourth NB and SB through lanes	3%
Moulton & Santa Maria	Laguna Hills/ Laguna Woods	Add fourth NB and SB through lanes	3%
El Toro & Jeronimo	Lake Forest	Add EB right-turn lane	6%
Los Alisos & Muirlands	Lake Forest/ Mission Viejo	Add second EB left-turn lane	4%
Alicia & Muirlands	Mission Viejo	Add second SB left-turn lane and SB, EB and WB right-turn lanes	4%
La Paz & Muirlands/I-5 NB	Mission Viejo	Add second NB left-turn lane	4%
Jamboree & Chapman	Orange	Add fourth NB through lane	2%
DEFICIENT ARTERIAL ROADWAYS	<u> </u>		L
Laguna Canyon (Bake to Santa Maria)	Caltrans/ County/Irvine	Improve to six-lanes	4%
Moulton (El Toro to Glenwood)	Laguna Woods	Improve to eight-lanes	3%
Moulton (Ridge Route to Santa Maria)	Laguna Hills/ Laguna Woods	Improve to eight-lanes	3%
Moulton (Santa Maria to El Toro)	Laguna Woods	Improve to eight-lanes	3%
<b>DEFICIENT FREEWAY SEGMENTS</b>			······································
I-5 (Lake Forest to south of La Paz)	Caltrans	Implementation of Caltrans Traffic Operations Strategies (TOPS)	4%-5%
I-405 (Jamboree to Jeffrey)	Caltrans	Implementation of Caltrans TOPS	11%
I-405 (Sand Canyon to SR-133)	Caltrans	Implementation of Caltrans TOPS	10%
SR-55 (north of Irvine/Fourth)	Caltrans	Implementation of Caltrans TOPS	4%
DEFICIENT FREEWAY/TRANSPORT	TATION CORRID	OR RAMPS	·
I-5 at Alton (NB Direct On-Ramp)	Caltrans/Irvine	Add a third mixed-flow lane at the ramp meter	13%
I-5 at Sand Canyon (NB On-Ramp)	Caltrans/Irvine	Convert HOV preferential lane to a Add second metered mixed-flow lane	26%
I-5 at Sand Canyon (SB Off-Ramp)	Caltrans/Irvine	Add second drop lane from freeway mainline to off-ramp	16%
I-405 at Sand Canyon (NB Direct On- Ramp)	Caltrans/Irvine	Improve to a two-lane on-ramp with two mixed-flow lanes at the ramp meter	4%
I-405 at Sand Canyon (SB Off-Ramp)	Caltrans/Irvine	Add second drop lane from freeway mainline to off-ramp	2%
I-5 at El Toro (NB Loop On-Ramp)	Caltrans/ Lake Forest	Convert HOV preferential lane to a Add second metered mixed-flow lane	4%
I-5 at Lake Forest (NB Off-Ramp)	Caltrans/ Lake Forest	Add second drop lane from freeway mainline to off-ramp	2%
ETR East Leg (SR-241) at Santiago Canyon (SB On-Ramp)	Caltrans/TCA/ County	Provide three mixed-flow lanes at the on-ramp meter	2%
ETR East Leg (SR-241) at Santiago Canyon (NB Off-Ramp)	Caltrans/TCA/ County	Add second drop lane from ETR mainline to off-ramp	7%
Abbreviations: NB – northbound SB – southbound	EB – easth WB – wes		

<sup>[1]</sup> Only required with the Trabuco Road/ETR interchange option.

T-13 Prior to issuance of the first building permits for the Airport System Master Plan Development, the County will use its best efforts to obtain a cooperative agreement with the applicable lead jurisdictions to process amendments to the Orange County Master Plan of Arterial Highways (MPAH) as appropriate through the Orange County Transportation Authority (OCTA) for the unplanned long-range circulation improvements listed in Table 5.4-3 [i.e., Moulton Parkway and Laguna Canyon Road] with the County's participation in the MPAH process being based on the project fair share percentages identified in Table 5.4-3.

<u>Implementation of the proposed</u> These mitigation measures are, in whole or in part, within the responsibility and jurisdiction of another public agency or entity. Should these other entities not implement the mitigation measures, the relevant project impacts may remain significant after mitigation.

Note: Pursuant to CEQA Guidelines Section 15091(a)(2), portions of these measures are within the responsibility and jurisdiction of another public agency, and not the County of Orange.

#### 5.4.3.3 Conclusions

The potential cumulative traffic impacts of the Airport System Master Plan Proposed Project would be reduced to below a level of significance based on the implementation of mitigation measures included in Section 4.3. For the post 2020 conditions, the impacts of cumulative development would be reduced to a level of insignificance with Mitigation Measures T-12 and T-13. In the event that Traffic Operations Strategies (TOPS) improvements are not implemented, or if such improvements fail to mitigate the identified impacts to below a level of significance, the freeway/tollway mainline cumulative impacts will remain significant and unavoidable. Regarding off-site highway improvements that are not under the sole control of the County and are subject to the implementation procedures described in Mitigation Measure T-12, in the event such off-site improvements are not implemented by the jurisdiction in which the improvements are located, the cumulative impacts at those unimproved locations will remain significant and unavoidable.

# 5.4.4 Potential Cumulative Impacts Related to Noise

# 5.4.4.1 Aviation Noise Impacts

The project noise analysis (Section 4.4) indicated that the Proposed Project would not result in significant impacts on existing and cumulative OCP-96M noise sensitive land uses by reference to traditional and generally accepted 65 dB CNEL contour significance criterion. However, the development of a civilian airport at MCAS El Toro would result in a significantly greater number of airplanes approaching/departing OCX and operations during nighttime hours compared to when the MCAS El Toro was used as a military airport. Noise

from nighttime use of the Proposed Project will cause some sleep disturbance even after mitigation.

Section 4.4 also analyzes the aviation noise impacts of JWA on existing and cumulative OCP-96M noise sensitive land uses.

Within Orange County, JWA and OCX would be the sources of commercial aircraft noise impacts, although Fullerton Airport would generate cumulative GA aircraft noise impacts and Los Alamitos AFRC would generate cumulative military aircraft noise impacts in Orange County. In addition to airports in the County, GA and military aircraft overfly the County on approaches and departures to regional airports. While individually these flights generate less than significant noise impacts (see Section 4.4 for significance discussion) in the vicinity of JWA and the El Toro site, they have generated noise complaints in Southern Orange County since military aircraft ceased regular operations at MCAS El Toro. Since Section 4.4 concludes that OCX would have significant aircraft noise impacts after mitigation, the project is anticipated to result, together with other aircraft noise, in a significant aviation noise impact.

Currently, regional commercial airports including LAX, Long Beach, Ontario, Burbank, San Diego, and March AFB include existing and/or planned noise sensitive land uses within the aircraft 65 dB CNEL contour. The cumulative impact of the project would be to reduce or avoid increases in aviation operations and, therefore, reduce cumulative regional aircraft noise impacts on noise sensitive uses, including sleep disturbance. Because regional airports already include large numbers of noise sensitive uses (Table 8.2-1) within the 65 dB CNEL contour, unlike OCX and JWA, the net cumulative impact of the project is anticipated to be beneficial. That is, the project impacts in the vicinity of OCX and JWA after mitigation are anticipated to be less than No Project Alternative effects at regional airports.

#### 5.4.4.2 Conclusions

In summary, the project is anticipated to have a locally significantly adverse aviation noise impact, but a regionally beneficial cumulative aviation noise impact. As discussed in Section 4.4, mitigation measures would reduce local aviation noise impacts, but local impacts would remain significant after mitigation.

# 5.4.4.3 Highway Noise Impacts

Section 4.4 includes a cumulative analysis of the project highway noise impacts together with traffic generated by OCP-96M growth forecasts. Please refer to Section 4.4 for impacts and conclusions.

Section 5.4.3.1 includes an analysis of post-2020 cumulative traffic impacts including highway improvements that would reduce post-2020 impacts to a level of insignificance. All but two improvements are included in the MPAH, which is the basis of the County and

City Noise Elements for highway noise policy and regulation of development impacts. The two highway improvements not included on the MPAH are Laguna Canyon Road between Bake Parkway and Santa Maria Avenue, to which the project would contribute four percent of the cumulative growth, and Moulton Parkway between Ridge Route and Glenwood, to which the project would contribute three percent of the cumulative growth in traffic and traffic noise.

The County and City General Plans include Growth Management Elements/Plans and Noise Element policies, programs, and standards, and development regulations that ensure that highway noise impacts would be analyzed and mitigated during the review of the General Plan and zoning amendments required for the currently unplanned post-2020 cumulative development. In addition, these amendments would be subject to CEQA.

#### 5.4.4.4 Conclusions

Based on County and City General Plan Growth Management Elements/Plans and Regulations, post-2020 highway noise impacts are anticipated to be reduced to a level of insignificance.

However, as noted above, local aviation noise impacts would remain significant after mitigation. Therefore, the combination of highway and aviation noise in the vicinity of the El Toro site is anticipated to be cumulatively significant after mitigation.

# 5.4.5 Potential Cumulative Impacts

#### 5.4.5.1 CEQA Threshold

The CEQA threshold of significance used to determine cumulative air quality impacts is as follows: Does the project have air quality impacts that are individually limited, but cumulatively considerable?

## 5.4.5.2 Construction Impacts

The Proposed Project is expected to be constructed in four phases, lasting approximately 20 years. Ozone precursors ROC and NO<sub>X</sub> from construction equipment exhaust and fugitive dust from soil disturbance generated during construction of each of the related projects identified in Section 5.0 of Draft EIR No. 573, as well as from the Proposed Project, will cumulatively affect the region's air quality.

The South Coast Air Quality Management District (SCAQMD) has included the construction emissions within the South Coast Air Basin (Basin) in its preparation of the emissions inventory in the regional Air Quality Management Plan (AQMP) and, therefore, in its projection of the attainment of these air pollutants. Each related project, similar to the Proposed Project, would be required to implement all feasible mitigation measures to reduce

the air pollutant emissions. Because the South Coast Air Basin is nonattainment in ozone and PM<sub>10</sub>, cumulative construction emissions are anticipated to remain significant and unavoidable for cumulative effects under OCP-96M forecasts.

## 5.4.5.3 Operational Air Quality Impacts

Although the Basin is in nonattainment for federal CO standards, concentrations of carbon monoxide (CO) monitored in Orange County stations have been below the State and federal standards for the past five years. Cumulative traffic volumes based on the project for the JWA and El Toro sites and on OCP-96M forecasts for offsite development were used to assess the CO concentrations at most affected intersections in the project vicinity in Section 4.5 of Draft EIR No. 573. No exceedance of the State or federal CO standards was identified in-for any future years including Phase 1 (2005), Phase 2 (2010), Phase 3 (2015), and Phase 4 (2020). No exceedance of the State or federal CO standards was found related to aircraft emissions from the Proposed Project in all future phasing years. Therefore, cumulative local CO hot spot impacts would be less than significant in all future phasing years. For the post-Phase 4 traffic conditions, all intersection impacts would be mitigated to acceptable levels of service with the recommended mitigation measures. Based on this, no CO hot spots are anticipated.

Due to the low background concentrations for  $SO_{2x}$  in Orange County, no exceedance of the State and federal  $SO_{2x}$  standards was found related to the aircraft exhaust in all future phasing years. The Basin is in attainment with State and federal  $SO_{2x}$  standards. Cumulative emissions of  $SO_{2x}$  are not anticipated to result in local concentrations of  $SO_{2x}$  in Orange County to exceed the State or federal standards.

A possible exceedance of the State's 1-hour NO<sub>2</sub> ambient air quality standard (0.25 ppm) was identified for OCX at Alton Parkway Business Park (0.373 ppm in Phase 2 and 0.493 ppm in Phase 4), the Irvine Transportation Center (0.580 ppm in Phase 2 and 0.670 ppm in Phase 4), and Old Towne Irvine (0.487 ppm in Phase 2 and 0.557 ppm in Phase 4). Based on the levels projected in Phase 2 and Phase 4, it is anticipated that the State's 1-hour NO<sub>2</sub> standard would also be exceeded at these three locations around OCX. Therefore, it is anticipated that the projected NO<sub>2</sub> concentrations would exceed the State's 1-hour standard at three sites in the vicinity of OCX for all Tour phases.

Similarly, a possible exceedance of the State's 1-hour NO<sub>2</sub> standard was identified for JWA at Executive Park (0.277 ppm) in Phase 4. Based on a linear interpolation calculation between the Phase 2 NO<sub>2</sub> level (0.219 ppm) and Phase 4 (0.277 ppm) at Executive Park, it is anticipated that the projected NO<sub>2</sub> level (0.248 ppm) at Executive Park in Phase 3 would be below the State's 0.25 ppm standard. Therefore, NO<sub>2</sub> concentration is projected to exceed the State's 1-hour NO<sub>2</sub> standard at one site in the vicinity of JWA after Phase 3.

There are no projected future NO<sub>2</sub> concentrations that would exceed the federal annual arithmetic mean (AAM) standard (0.0534 ppm) at both OCX and JWA in all future phasing years. There are no exceedances of the NO<sub>2</sub> standard modeled in any of the community areas surrounding either OCX or JWA. There would be no exceedance of the State's 1-hour NO<sub>2</sub> standard at all receptor locations around both airports. However, there would be one exceedance of the

federal annual arithmetic mean (AAM) standard (0.0534 ppm) for NO<sub>2</sub> concentration at Irvine Transportation Center around OCX for the Proposed Project in Phases 2, 3, and 4. Cumulative development identified in Section 5.0 of Draft EIR No. 573 would result in the emission of NO<sub>2</sub>. It is not anticipated that NO<sub>2</sub> emissions from the related projects and the OCP-96M/post-2020 forecasts would increase the NO<sub>2</sub> concentrations around the project sites by a significant amount. Nevertheless, the NO<sub>2</sub> exceedances are anticipated to be significant after mitigation.

The projected PM<sub>10</sub> concentrations would exceed the State's 24-hour standard (50 ug/m³) at all receptor locations around JWA and OCX in all future phasing years, but would be below the federal 24-hour standard (150 ug/m³). The are no projected future PM<sub>10</sub> concentrations that would exceed the federal annual arithmetic mean (AAM) standard (50 ug/m³) at both JWA and OCX at all receptors in all phasing years.

The SCAQMD has included the operational emissions from existing and future development within the South Coast Air Basin in its preparation of the emissions inventory in the regional Air Quality Management Plan (AQMP) and, therefore, in its projection of the attainment of these air pollutants. Future offsite development is based on SCAG and OCP-96M forecasts. Each cumulative or related project, similar to the Proposed Project, would be required to implement all feasible mitigation measures to reduce the air pollutant emissions.

#### 5.4.5.4 Conclusions

The Proposed Project would result in additional air pollutant emissions in the project vicinity. Therefore, because the South Coast Air Basin is nonattainment in State and federal standards for ozone, and PM<sub>10</sub>, and nonattainment in State standards for CO, cumulative construction and operational emissions would remain significant and unavoidable in all four project phases (Phase 1, Phase 2, Phase 3, and Phase 4) under both the OCP-96M and the post-Phase 4 cumulative development scenarios.

# 5.4.6 Topography

The threshold of significance used to determine cumulative topography impacts is as follows:

(i) Does the project have topography impacts that are individually limited, but cumulatively considerable?

# 5.4.6.1 Cumulative Impacts

Existing topography varies substantially among the various related project development areas. Locally, these topographic features may be altered, depending on construction/development specifics of the individual projects. However, the County and City General

Plans, hillside grading policies, and grading ordinances are anticipated to mitigate topographic impacts in the project area.

#### 5.4.6.2 Conclusions

The project will not have a significant adverse impact on topography, and is not anticipated to contribute to a significant cumulative impact.

# 5.4.7 Soils, Geology, and Seismicity

The threshold of significance used to determine cumulative soils, geology, and seismicity impacts is as follows:

(i) Does the project have soils, geology, and seismicity impacts that are individually limited, but cumulatively considerable?

## 5.4.7.1 Cumulative Impacts

The geologic impacts vary substantially among projects within the cumulative study area, including the Lower Peters Canyon Specific Plan, Foothill Ranch, Portola Hills, Tustin MCAS Reuse Plan and the Eastern Transportation Corridor areas. Cumulative development covers a wide geographic area and includes numerous geologic features.

However, none of the projects in the cumulative study area are anticipated to result in significant adverse impacts related to geology, seismicity and soils after mitigation. Engineering and design features required by County and City General Plans and development regulations are anticipated to adequately address any potentially significant adverse impacts related to soils, geophysical, and reasonably foreseeable seismic impacts.

#### 5.4.7.2 Conclusions

The project is not anticipated to result in significant adverse impacts due to soils, geology, or seismicity. Therefore, cumulatively significant adverse impacts related to soils, geology and seismicity in the area are not anticipated.

# 5.4.8 Potential Cumulative Impacts Related to Hydrology and Water Quality

The threshold of significance used to determine cumulative hydrology and water quality impacts is as follows:

(i) Does the project have hydrology and water quality impacts that are individually limited, but cumulatively considerable?

## 5.4.8.1 Cumulative Impacts

The potential impacts of cumulative development will be mitigated to acceptable levels for hydrology as a result of flood control improvements required by County and City General Plans and development regulations and Master Plans of Drainage for project approval. These flood control features, such as construction of retarding basins, the lining of flood control channels, and ultimate implementation of area flood control master plans, will contribute to reducing flood control hazards associated with these projects. Regional hydrology will be improved as a result of these projects and their associated flood control requirements.

For water quality, the increase in urban runoff from cumulative development will continue to provide a source of heavy metals and fertilizers to the downstream watersheds, particularly San Diego Creek, Upper Newport Bay, and ultimately the Pacific Ocean.

Pursuant to the State and federal Clean Water Act, applicable regulatory agencies established total maximum density load (TMDL) pollutants inland to improve the water quality of receiving waters. Any cumulative impacts from surface water runoff quantities from cumulative development will be mitigated to below a level of significance by mitigation measures included in these projects in conformance with federal, State, County, and local regulations and policies. The implementation of Best Management Practices included in required National Pollution Prevention Plan Permits and Storm Water Pollution Prevention Plans is anticipated to reduce pollutants in runoff water to acceptable levels.

#### 5.4.8.2 Conclusions

Based on the foregoing, cumulative impacts are anticipated to be reduced to acceptable levels after mitigation.

# 5.4.9 Potential Cumulative Impact Related to Biological Resources

The threshold of significance used to determine cumulative biological resource impacts is as follows:

(i) Does the project have biological resource impacts that are individually limited, but cumulatively considerable?

## 5.4.9.1 General Impacts

The regional context for the consideration of potential cumulative adverse impacts on biological resources is the Central and Coastal NCCP Subregions. The NCCP Central

subregion area includes most of the frontal slopes of the foothills west of the Santa Ana Mountains from State Route 91 to the natural areas near Aliso Creek. The NCCP Coastal Subregion of the NCCP focuses on the San Joaquin Hills south of the MCAS El Toro site. However, because the biological resources on the MCAS El Toro site and on the sites of most of the related projects are generally in the Central Subarea, this discussion focuses primarily on potentially significant cumulative adverse impacts on biological resources in that area.

Although the project is not expected to have significant impacts, cumulative development in this area could result in significant adverse impacts on biological resources as described in the environmental documents for each project and/or could contribute to cumulatively significant adverse impacts on biological resources in the Central Subarea. These projects include the Lower Peters Canyon Specific Plan, the Foothill Ranch Planned Community, the Portola Hills Planned Community, Saddleback Meadows, Northwood High School in Irvine, the Alton Parkway Extension, the improvements to Rockfield Road and the north segment of the Foothill Transportation Corridor.

The Proposed Project would also contribute to the need for off-site highway improvements which will impact biological resources. Specifically, the project would contribute 3% of the cumulative total traffic growth at off-site highway improvements on Laguna Canyon Road, south of El Toro Road that will have impacts on off-site coastal sage scrub habitat, as shown in Table 5.4-4. These highway improvements, and the off-site coastal sage scrub habitat impacts, would result with or without the project.

Table 5.4-4
Summary of Off-Site Cumulative Coastal Sage Scrub Impacts (Acres by Phase)

Impact Type	2005	2010	2015	2020	Total
Direct	0	0	0	0	0
Indirect	2.62	0	0	0	2.62
Total	2.62	0	0	0	2.62

The primary potential significant cumulative adverse impact on biological resources is the continued fragmentation of ecosystems resulting from the incremental loss of native habitat and the separation of areas of native habitat from other areas of native habitat by intervening urban and suburban development. Habitat fragmentation has been linked to the reduction of sustainable population levels for many species, particularly free ranging mammals such as the mountain lion, bobcat and mule deer, that require large expanses of land. However, a coordinated open space plan that enhances and preserves remaining areas supporting native habitats would reduce such impacts on biological resources. The central/coastal NCCP, other open space preservation activities, and the Habitat Reserve and Wildlife Habitat Area included in the Proposed Project will provide connectivity among a number of large, protected open spaces in this part of the County. Therefore, the Proposed Project will not

contribute to a cumulatively significant adverse impact on biological resources, but will contribute beneficially to long-term preservation and protection of biological resources in the NCCP Central Subregion.

#### 5.4.9.2 Conclusions

The project is not anticipated to result in significant adverse impacts, but would have beneficial impacts by preservation of habitat areas and construction of new habitat areas. Therefore, the project would not add to cumulative adverse impacts of related projects or OCP-96M.

#### 5.4.9.3 Alton Parkway Impacts

The extension of Alton Parkway was committed and funded to serve off-site development prior to the Department of Defense decision to close MCAS El Toro. The Proposed Project, consistent with the CRP and EIR No. 563, contemplates requesting conveyance of the right-of-way for the extension to the County. Construction of Alton Parkway is and has been since the late 1980s, an independent project. The Alton Parkway extension is required without the Proposed Project, and was funded separately from the planning of the reuse project. Therefore, any impacts of the construction and operation of Alton Parkway are analyzed as potential cumulative impacts.

As currently aligned, construction of the extension would impact biological resources. The biological resources potentially impacted include coastal sage scrub, southern willow scrub, coast live oak woodland, ephemeral drainages and washes, and primarily annual, non-native grassland. Site preparation activities are estimated to impact an estimated five acres of coastal sage scrub. In addition, based upon the 1992 NCCP baseline target species data, one pair of California gnatcatchers is present within the impact area of the roadway. Although specific delineations have not been completed, there appears to be less than one acre of Waters of the U.S. (Army Corps of Engineers jurisdiction) and up to four acres of riparian scrub, which would be impacted if the Alton Parkway is constructed as presently proposed. The four acres of riparian scrub would include one acre of Waters of the U.S. These acreages occur within the impact area of the roadway. Riparian scrub would include willow and mulefat communities along Borrego Wash and a tributary. A specific jurisdictional wetland delineation has not been conducted for Borrego Wash, but it is likely to be a subset of the acreage defined for Waters of the U.S.

#### 5.4.9.4 Conclusions

These impacts prior to mitigation (see Section 5.4.9.7) would be considered a significant cumulative impact for biological resources.

#### 5.4.9.5 Marshburn Channel Impacts

Marshburn Channel improvements are required as a part of regional off-site development and are required even without the Proposed Project. Marshburn Channel, between Irvine Boulevard and Trabuco Road, is a gunite-lined open trapezoid channel that is planned to be improved to a concrete-lined open channel. In addition, and along Irvine Boulevard, the existing open channel is planned to be improved to a reinforced concrete box. The Marshburn Channel does convey Waters of the U.S. An estimated 1.9 acres of Waters of the U.S. would be impacted as part of this regional project. The 1.9 acre impact is considered a cumulative impact. There are no substantive biological resources contained within the cross section of this drainage.

#### 5.4.9.6 Conclusions

No cumulative significant adverse biological resource impacts are anticipated to occur as a result of the Marshburn Channel improvements.

#### 5.4.9.7 Mitigation Measures

Section 4.9 includes mitigation measures for construction activities that would reduce the impacts of the Alton Parkway extension on biological resources. In addition to these impacts, the following mitigation measure is incorporated:

B-17 The County is the lead agency for a redesign of Alton Parkway. It will redesign the extension to minimize biological impacts to one acre or less of Waters of the U.S. and jurisdictional wetlands.

#### 5.4.9.8 Level of Significance After Mitigation

The cumulative impacts associated with the Alton Parkway extension on biological resources have been determined to be significant prior to mitigation. While the application of project mitigation measures (Section 4.9) to the Alton Parkway extension would reduce non-wetlands impacts, these impacts would remain significant after mitigation. With the incorporation of Mitigation Measure B-17, the total Waters of the U.S. impact for the Alton Parkway extension can be reduced to approximately 0.7 acre, which includes 0.45 acre along Borrego Wash and 0.25 acre along the local tributary. In addition, the total linear distance of the crossing of these drainages will also be reduced, as well as the amount of acreage of riparian scrub. Therefore, due to the small streambed acreage directly impacted and Mitigation Measure B-17, the cumulative riparian resource impacts are less than significant after mitigation.

### 5.4.10 Potential Cumulative Impacts Related to Public Services and Utilities

The threshold of significance to determine cumulative public services impacts is as follows:

(i) Does the project have public service impacts that are individually limited, but cumulatively considerable?

#### 5.4.10.1 Potential Impacts Related to Public Services

#### Cumulative Impacts

Cumulative offsite development is anticipated to increase the demand for public services, including police, fire and emergency medical, schools, libraries, transit, and solid waste. In combination with cumulative development projects, the Proposed Project at the MCAS El Toro site will also contribute to the demand for public services. Although the Proposed Projects' incremental contribution is not anticipated to be significant based on implementation of mitigation measures and the standard conditions of approval listed in Section 4.10.

Cumulative developments will be required under County and City General Plan Growth Management Elements, public service and facilities policies, and development regulations to provide for mitigation that will reduce significant public services impacts. Cumulative impacts on the provision of schools, police, and fire and emergency services will be analyzed in the CEQA and fiscal impact documents required by County and City General Plans for all future cumulative projects, and these impacts are anticipated to be reduced to below a level of significance with the addition of staff, facilities, or the payment of applicable fees. The OCTA will be able to determine the need for transit service, and plan accordingly, as reasonably foreseeable future projects develop. With adherence to the IWMD recommendations, and State, local, and federal regulations that address the disposal of waste, the cumulative projects are not expected to have a significant cumulative impact on solid waste services.

#### 5.4.10.2 Conclusions

In conclusion, no significant cumulative effects are anticipated.

#### 5.4.10.3 Potential Impacts Related to Utilities

The threshold of significance to determine cumulative utility impacts is as follows:

(i) Does the project have utility impacts that are individually limited, but cumulatively considerable?

Cumulative development is anticipated to increase the demand for the following utilities.

#### Cable Television Services

As stated in Section 4.10, the project with Mitigation Measures U-1, U-2, U-3 will not result in significant short or long-term impacts related to cable television.

Cumulative development will result in an increase in the demand for cable television services. The cable television providers in the region have or would provide cable television, telecommunications, data systems and Internet services to these projects. Short-term construction related impacts to cable television facilities will be mitigated to below a level of significance by mitigation measures included in these projects as required by County and City development policies. Therefore, the Proposed Project, when considered with other projects in the region, will not contribute to a cumulatively significant adverse impact related to cable television services.

#### Communication Facilities and Services

As stated in Section 4.10.7, the project with Mitigation Measure U-3 would result in no significant adverse impacts on Communication Facilities and Services.

Cumulative development will result in increases in the demand for communication facilities and services. It is anticipated that the communication providers in the region have or would also provide telecommunications, data systems and Internet services to these projects. Short-term construction related impacts on communication facilities will be mitigated to below a level of significance by mitigation measures included in these projects as required by County and City development policies. Cable television providers master plan their facilities, infrastructure, services and programming based on planned and anticipated land uses and market based demand in their service areas. Communications providers master plan their facilities, infrastructure and services based on planned and anticipated land uses and market based demand in their service areas. Therefore, the Proposed Project, when considered with other projects in the region, will not contribute to a cumulatively significant adverse impact related to communication facilities and services.

#### Electrical Facilities and Services

As stated in Section 4.10.7, the project with Mitigation Measures U-1, U-2, and U-3 would be reduced to a level of insignificance.

Cumulative development will result in an increase in the demand for electric facilities and services. County and City development policies require will-serve commitments from Southern California Edison (SCE) or San Diego Gas and Electric ensuring that infrastructure and supplies are adequate to serve these projects. Short-term construction related impacts to

electrical facilities will be mitigated to below a level of significance by mitigation measures required by County and City development policies. Therefore, the Proposed Project, when considered with other projects in the region, is not anticipated to contribute to a cumulatively significant adverse impact related to electric facilities and services.

#### Fuel Facilities

As stated in Section 4.10.7, the fuel facility demand at JWA would be reduced in the project case, and the impacts of the project at the El Toro site would not be significant.

Cumulative development will result in an increase in the demand for fuel. County and City development policies require will-serve commitments from service providers prior to development. Therefore, the Proposed Project, when considered with other aviation related projects in the region, is not expected to contribute a cumulatively significant adverse impact related to fuel and fuel facilities.

#### Natural Gas

As stated in Section 4.10.7, the project impacts with Mitigation measures U-1, U-2, and U-3 would be reduced to a level of insignificance for natural gas.

Cumulative development will result in an increase in the demand for natural gas facilities and services. County and City development policies require will-serve commitments prior to project development from the Southern California Gas Company (SCGC). It is anticipated that short-term construction related impacts created by the construction of new natural gas facilities will be mitigated to below a level of significance by County and City development policies. Therefore, the Proposed Project, when considered with other projects in the region, is not anticipated to contribute to a cumulatively significant adverse impact related to natural gas services and facilities.

#### Domestic Water

As stated in Section 4.10.7, the project with Mitigation Measures U-1, U-2, and U-4 would not result in significant adverse short or long-term domestic water impacts.

Cumulative development will result in an increase in the demand for domestic water. County and City development policies require will-serve commitments prior to project development from water districts. It is anticipated that short-term construction related impacts created by the construction of new domestic water facilities will be mitigated to below a level of significance by County and City development policies. Other projects in the region are also anticipated to increase the demand for domestic water in the region. The water districts master plan their facilities, infrastructure, and services based on planned and anticipated land uses, adopted forecasts including OCP-96M, and market based demand in their service area. Each project is reviewed independently by the applicable water district to

ensure that sufficient facilities are in place and available domestic water supplies are adequate to properly service each project. Therefore, the Proposed Project, when considered with other projects in the region, is not anticipated to contribute to a cumulatively significant adverse impact related to domestic water resources.

#### Recycled Water

As stated in Section 4.10.7, the project with Mitigation Measures U-1 and U-2 would not result in a significant adverse impact on recycled water facilities.

Cumulative development will result in an increase in the demand for recycled water. Short-term construction impacts will be created by the development of new recycled water facilities, including disruption of recycled water service or accidental damage to existing recycled water facilities. Other projects in the region are also anticipated to increase the demand for recycled water. The water districts master plan their facilities, infrastructure and services, including recycled water, based on planned and anticipated land uses, adopted growth forecasts including OCP-96M, and market based demand in its service area. Each project is also reviewed independently by the water district to ensure that sufficient facilities are in place and available recycled water supplies are adequate to properly service each project. Therefore, the Proposed Project, when considered with other projects in the region, is not anticipated to contribute to a cumulatively significant adverse impact related to recycled water.

#### Sanitary Sewers

As stated in Section 4.10.7, the project with Mitigation measures U-1 and U-2 would not result in a significant adverse impact on sanitary sewers.

Cumulative development will increase the generation of wastewater. County and City development policies require will-serve commitments prior to project development from the applicable water district. Other projects in the region would also increase the generation of wastewater in the region. As noted earlier, the water districts master plan their facilities, infrastructure and services, including sanitary sewers, based on planned and anticipated land uses and market based demand in its service area. Each project is also reviewed independently by the water district to ensure that sufficient facilities are in place to properly serve the wastewater collection and treatment needs of each project. Therefore, the Proposed Project, when considered with other projects in the region, is not anticipated to contribute to a cumulatively significant adverse impact related to wastewater.

#### 5.4.11 Natural Resources and Energy

The threshold of significance to determine cumulative natural resource and energy is as follows:

(i) Does the project have natural resources and energy impacts that are individually limited but cumulatively considerable?

#### 5.4.11.1 Mineral Resources

As stated in Section 4.11.8, there are no mineral resources on the JWA site and any resources on the El Toro site will not be used for commercial purposes. Mineral resources available in the region are anticipated to be adequate for the project. In conclusion, the Proposed Project's effects on mineral resources would not be significant.

Cumulative development will require the use of building materials including such mineral resources as sand, gravel and other similar construction materials. County and City Resource Element policies and the Surface Mining Recovery Act will ensure that these materials are available in sufficient supply to provide for the development of these future projects since these materials are readily available in the County and region. Also, the affected area does not include active mineral mining sites and the cumulative projects will not affect extraction operations or its resources. Finally, sand, gravel, and construction materials are proposed to be recycled on the project site, reducing project demand.

#### 5.4.11.2 Conclusions

Therefore, the Proposed Project, when considered with other projects in the region, is not anticipated to contribute to a cumulatively significant impact related to mineral resources.

#### 5.4.11.3 Agricultural Resources

As stated in Section 4.11.8, there are no agricultural resources on the JWA site and the project would result in the loss of 902 acres (87%) of the agricultural resources at the El Toro site. This impact is significant, and would remain significant after mitigation. This conclusion is the same as EIR 563.

The depletion of agricultural and farmland throughout the region is a significant adverse impact that can, in most cases, not be mitigated. Although the California State Legislature provides incentives to retain farmlands for agricultural purposes under the Williamson Act, the policy does not guarantee the long-term preservation of agriculture lands. The Proposed Project will result in the loss of substantial agricultural farmland on the project site due principally to be construction of nonaviation uses including golf courses, a business park, and public facilities (see Section 4.11.7.1). In addition, the project would result in off-site direct (due to project improvements) and indirect (due to cumulative traffic improvement requirements) impacts, as shown in Table 5.4-5.

Table 5.4-5
Summary of Off-Site Agricultural Soils Impacts (Acres by Phase)

Impact Type	2005	2010	2015	2020	Total
Direct	9.76	0.66	0	3.43	13.85
Indirect	0	34.00*	0	0	34.00
Total	9.76	34.66	0	3.43	47.87

<sup>\*</sup> Trabuco Road/ETC Interchange

#### 5.4.11.4 Conclusions

Some of the other related projects will also result in the incremental loss of agricultural and farmland in the Orange County area. Therefore, these projects will contribute to the cumulatively significant loss of agriculture resources (see Section 4.11). This was the same conclusion made in Section 7.3.10.1 (Potential Cumulative Impacts Related to Natural Resources and Energy Under the CRP) of Final EIR No. 563.

#### 5.4.11.5 Water

As stated in Section 4.11.8, there are no surface water resources at the El Toro or JWA sites, and no groundwater modification is planned. Therefore, the Proposed Project would result in no significant adverse impacts to water resources.

Cumulative development will also result in the use of additional amounts of water which will result in a significant adverse cumulative impact on water. County and City development policies require will-serve commitments prior to development from water purveyors. The regional water districts master plan their capacity based on planned development and adopted growth forecasts such as OCP-96M. Water reclamation and recycling, groundwater recharge and other supply strategies are required by County and City policies, as well as water district policies and programs, in order to meet the demand of continued growth and development. County and City Growth Management Elements and public services and facilities policies ensure development is phased in coordination with water capacity.

Cumulative projects are also required to comply with CEQA, including water impact mitigation programs. However, since water is a finite resource, cumulative development and the project will contribute individually small but cumulatively significant demands for a limited resource.

#### 5.4.11.6 Conclusions

Cumulative development impacts on water demand are significant adverse impacts.

#### 5.4.11.7 Energy

As stated in Section 4.11.8.2, electricity demand at JWA would decrease with the project, and increase at the El Toro site. The increase at the El Toro site would contribute substantially less than 1% of the peak SCE service in year 2020. Regarding natural gas, the demand at JWA would also decline under the project, but increase at the El Toro site. Again, the increase at El Toro would be substantially less than 1% of statewide and ASA growth. Section 4.11.8.2 analyzes the project impacts on highway, construction and aviation fuels for JWA and the El Toro site. For each of these impacts, Section 4.11.8.2 determines that the project would result in substantially less than 1% increase in demand, but the project's encouragement of the use of jet fuel is an impact that cannot be mitigated to a level below the level of significance.

Cumulative development projects vary in contribution to cumulative adverse impacts on energy resources. Some of the proposed projects and projects under construction will create increased need for energy as well as services and facilities required for the supply of these resources. On the other hand, projects described under Section 5.3.3 (Transportation Projects) of the Project Draft EIR may contribute to beneficial impacts on such energy resources as fossil fuels by reducing trip length and vehicular fuel consumption in the region. However, considered cumulatively, cumulative development will have an adverse impact on energy resources in the region because nonrenewable energy resources are finite.

#### 5.4.11.8 Conclusions

Therefore, the Proposed Project, considered with other projects in the region, will contribute to a cumulatively significant adverse impact related to energy.

## 5.4.12 Potential Cumulative Impacts Related to Aesthetics, Light and Glare

The threshold of significance to determine cumulative aesthetic, light and glare impacts is as follows:

(i) Does the project have aesthetic, light and glare impacts that are individually limited but cumulatively considerable?

#### 5.4.12.1 Cumulative Impacts

The Proposed Project would change the views of the site from the existing closed military base to an airfield with new airport terminal, aviation support use buildings and nonaviation revenue support uses, which is considered a beneficial impact. Cumulative projects have or will result in a change in the aesthetics of the area in which they are constructed. However, each project has or will be required to comply with CEQA and County or City development

policies or regulations, which are anticipated to mitigate their aesthetic impacts to an acceptable level. Therefore, the Proposed Project, when considered with other related projects in the region, will not contribute to a cumulatively significant adverse impact related to aesthetics.

#### 5.4.12.2 Conclusions

The Proposed Project would not create light and glare substantially greater than that generated by 1998 military base operations because of the similar type of use proposed. Cumulative development will incrementally add sources of light and glare to the area where they are constructed However, each project has or will be required by County or City development policies to mitigate light and glare impacts to below a level of significance. Therefore, the Proposed Project, considered with other projects in the region, will not contribute to a cumulatively significant adverse impact related to light and glare.

### 5.4.13 Potential Cumulative Impacts Related to Cultural Resources

The threshold of significance to determine cumulative cultural resource impacts is as follows:

(i) Does the project have cultural resource impacts that are individually limited but cumulatively considerable?

#### 5.4.13.1 Cumulative Impacts

The Proposed Project, considered by itself, will not generate any significant adverse impacts to cultural resources since the studies required by the mitigation measures in FEIR No. 563 concluded that 1) the post-war era buildings at MCAS El Toro were not eligible for listing on the National Register of Historic Places, and 2) the archaeological site CA-ORA-1462 is not considered a unique resource or eligible for the National Register. Significant paleontological sites are not known to be present; however, the on-site soil formations have the potential for these resources. Standard construction monitoring procedures will be required for the project to ensure that no significant impacts occur if such paleontological resources are discovered during grading activity. Application of those procedures will reduce any potentially significant impact to paleontological resources to less than significant.

The Proposed Project, considered in conjunction with cumulative development projects in the study area, would not contribute to cumulative impacts on cultural resources, since all development projects are subject to County and City development policies or requirements. These mitigation requirements will ensure that adequate procedures, such as research, site testing, construction monitoring, and data recovery, take place as necessary. Therefore, impacts caused by other cumulative projects approved in the study area have been or will have been mitigated to appropriate levels.

At JWA, there are no cultural resources on the site that would be disturbed by the Proposed Project improvements.

#### 5.4.13.2 Conclusions

The Proposed Project, when considered with other projects in the region, will not contribute to a significant cumulative impact to cultural resources.

### 5.4.14 Potential Cumulative Impacts Related to Recreation

The threshold of significance to determine cumulative recreation impacts is as follows:

(i) Does the project have recreational impacts that are individually limited but cumulatively considerable?

#### 5.4.14.1 Cumulative Impacts

The Proposed Project considered by itself will not create any significant impacts to recreational facilities that cannot be mitigated to a level below significance. One potentially significant impact would occur by the project due to inconsistency with the Open Space Conservation Map in the County General Plan Resources Element. This condition would be corrected upon implementation of Mitigation Measure LU-1, which would amend the County General Plan elements to be consistent with the Proposed Project. In the long term, the Proposed Project does not physically disrupt any existing parks, off-road trails, or other recreational facilities. The project provides for open space and recreational uses that would be compatible with the proposed commercial airport and aviation related uses. In that the project provides recreational uses that would be available for public use, it adds recreational components to the MCAS El Toro area that were previously not available for public use.

Considered in combination with the reasonably foreseeable future projects in the cumulative study area, the Proposed Project does not cause increased use of existing recreational facilities, nor does it cause a need for more recreational facilities to be built.

#### 5.4.14.2 Conclusions

In conclusion, the recreational effects of the Proposed Project would not be cumulatively significant.

# 5.4.15 Potential Cumulative Impacts Related to Public Health and Safety

The threshold of significance to determine cumulative public health and safety impacts is as follows:

(i) Does the project have public health and safety impacts that are individually limited but cumulatively considerable?

#### 5.4.15.1 Cumulative Impacts

Cumulative development will create incremental public health and safety impacts to the degree that projects require the use of hazardous materials for construction and operation. However, transportation, storage use, and disposal of these materials are strictly regulated by existing federal, State, County, and in some areas, city regulations. In addition, each cumulative project must comply with CEQA. Therefore, these projects are not anticipated to create significant public health and safety impacts.

Cumulative aircraft safety impacts related to the Proposed Project would result from increases in aviation activity at other airports in the Air Service Area. Within the Air Service Area, the number of passengers is projected to increase from 91.8 MAP in 1996 to 178.6 in 2020, a 95 percent increase. Air cargo handled in the region is projected to increase from 2.5 million tons in 1996 to 8.9 million tons in 2020, a 256 percent increase. While these increases are not expected to affect air safety in the immediate vicinity of JWA and OCX beyond that estimated for the Proposed Project, they will tend to increase the risk of mid-air collisions in the Air Service Area.

Because general aviation aircraft were involved in the two significant commercial aircraft mid-air collisions in the Air Service Area (Cerritos and San Diego), general aviation flying is seen as a significant factor in the risk of mid-air collisions. The number of general aviation aircraft based in the region (and presumably the number of general aviation operations) is projected to remain essentially the same in 2020 as in 1997. The stabilizing of general aviation activity in the region is expected to moderate potential increases in mid-air collision risk.

Growth in urbanization around JWA and OCX would tend to increase the potential for personal injury to people on the ground or property damage in the event that an off-site accident occurred. However, property around JWA is essentially fully built out, and the proposed safety zones at OCX will mitigate this potential impact to a level of insignificance (Section 4.15).

Although cumulative growth has the potential to increase the risks associated with aircraft accidents, the cumulative impacts are not estimated to be significant relative to Existing Conditions or the Proposed Project due to the extremely small likelihood of occurrences.

#### 5.4.15.2 Conclusions

The Proposed Project, considered with other projects in the region, will not contribute to a cumulative significant adverse aviation impact related to public health and safety.

### 5.4.16 Potential Cumulative Impacts Related to Hazardous Materials and Hazardous Waste

#### 5.4.16.1 Asbestos and Lead-Based Paint

The threshold of significance to determine cumulative hazardous materials and hazardous waste impacts is as follows:

(i) Does the project have hazardous materials and hazardous waste impacts that are individually limited but cumulatively considerable?

As discussed in 5.4.15 above, cumulative development has the potential to use hazardous materials for construction and operation. However, the transport, storage, use, and disposal of these materials are strictly regulated by existing federal, State, County, and, in some areas, city regulations. In addition, each cumulative project must comply with CEQA. Therefore, these projects are not anticipated to create significant hazardous conditions.

Under the Proposed Project, a number of buildings that contain asbestos and lead-based paint will remain on the site at the time of Marine Corps conveyance of the site to the County. However, according to DoD policy, the Marine Corps is required to maintain all asbestos-containing materials (ACM) and lead-based paint (LBP) in a manner protective of human health and the environment and in compliance with all applicable federal, state and local laws pertaining to these materials.

#### 5.4.16.2 Conclusions

Therefore, it is anticipated that all structures containing ACM and LBP will be received by the County in a condition that will not contribute to a cumulatively significant impact related to ACM and LBP.

### 5.4.16.3 Hazardous Materials Use and Hazardous Waste Generation

Land uses under the Proposed Project and potentially foreseeable future projects potentially include hazardous materials and generate hazardous wastes such as petroleum wastes, spent solvents, used oils and oily wastes, and other miscellaneous hazardous wastes. These substances will be managed and controlled based on existing federal, state and local

regulations governing the storage, handling, transportation, and disposal of hazardous materials and wastes. Therefore, since these materials will be managed and controlled, they are not anticipated to contribute to a cumulative significant impact related to the use and disposal of hazardous materials.

As analyzed in Section 4.16, the right-of-way for northeasterly extension of Alton parkway is located in the immediate vicinity of landfill Site 2 (within 1000 feet) and therefore may be adversely affected by the imposition of institutional controls for Landfill Site 2. DON recognizes the future construction of the Alton Parkway extension and states in the Final ROD for Site 2 that "... In developing the proposed remedy for Site 2, DON intends that all relevant parties (including the DON, FFA signatories, and the County of Orange) will cooperate with one another to ensure that all proposed projects (the remedy for Site 2, the construction of Alton Parkway, and improvements to Borrego Canyon Wash) are designed, constructed and maintained in a prompt and reasonable manner." Therefore, implementation of DON's selected remedy is anticipated to reduce impacts related to the presence of Site 2 on Alton Parkway to below a level of insignificance.

#### 5.4.16.4 Conclusions

The Proposed Project, considered with other projects in the region, will not contribute to a cumulative significant adverse impact related to hazardous materials and hazardous waste.

### 5.4.17 Potential Cumulative impacts Related to Socioeconomics

The threshold of significance to determine cumulative socioeconomic impacts is as follows:

(i) Does the project have socioeconomic impacts that are individually limited but cumulatively considerable?

#### 5.4.17.1 Cumulative Impacts

When considered together with related projects, the Proposed Project is anticipated to result in significant adverse cumulative socioeconomic impacts. Section 4.17 includes a cumulative analysis of the project compared to OCP-96M (refer to Section 4.17 for the analysis, mitigation measures, and conclusions.) The Proposed Project would redesignate substantial portions of the MCAS El Toro site for aviation, recreational, institutional, and commercial/industrial uses. As discussed earlier in Section 4.17 (Socioeconomics), redevelopment of the site with these uses could result in significant impacts relative to: (a) inducing substantial growth and concentration of non-resident employee population; (b) increasing demand for all types and prices of housing, while not directly supplying any new housing to meet this demand; and (c) reducing the supply of available housing in the County.

The project is not, however, anticipated to result in cumulative demand off site for additional support development. The project includes sufficient land and development (Section 3.0) to accommodate all ancillary uses such as car rentals, hotel rooms, flight catering, etc. In addition, the project includes 99 acres of aviation industrial uses and 87 acres of business park uses, which will meet the support needs of aviation and nonaviation uses proposed. Finally, approximately 200 acres of land adjacent to Planning Area 1 is classified for employment development in the County Land Use Element. According to the County General Plan, this area would accommodate 2.5 to 5.0 million square feet of light and service industries and professional-administrative office uses, and up to 26,000 employees. This development is included in OCP-96M and the traffic and other impact categories related to OCP-96M in Section 4.3. Based on these directly provided support uses and already planned adjacent uses, no further support uses would be required for the project.

Several of the related projects could produce similar impacts, cumulatively increasing the magnitude of these impacts. Specifically:

- The owners of unplanned and undeveloped areas surrounding the MCAS El Toro site could propose land uses not currently accounted for in the adopted regional growth forecasts. These proposals will be subject to review including CEQA analysis by the affected cities. The County and City of Irvine General Plans include Growth Management Elements which include policies and standards for these types of proposals which could mitigate impacts to a level of insignificance. However, potential land uses developed at these sites could impact the distribution of regional growth and thus cumulatively increase the magnitude of inconsistency with adopted regional growth forecasts if the affected cities chose to override the growth forecasts.
- ii) The related projects listed above could result in substantial growth and concentration of both resident and non-resident population, cumulatively increasing the impact of the Proposed Project.
- iii) The commercial and industrial projects that do not include a residential component (i.e., Sakioka Property, Koll Center Expansion, Rockwell Site Expansion, Hotel and Office Development on Former Car Dealership, Nexus Twin Towers, First American Title Project, Foothill Aliso Commercial Center, Shea Business Properties, Enterprise Car Sales, commercial development along El Toro Road, office development along Moulton Parkway, LAX Master Plan and other such cumulative projects) could cumulatively induce substantial non-residential development without providing any housing to accommodate potential new employees at these sites. These projects, in concert with the Proposed Project, could cumulatively increase demand for all types and prices of housing, while not directly supplying any new housing to meet this demand.

Each related project would be required to comply with CEQA and mitigation measures will be or have been considered together with findings. Mitigation Measure SE-1 in Section 4.17 would reduce the cumulative impacts noted above in relation to adopted forecasts.

However, cumulative socioeconomic impacts are anticipated to be significant after mitigation as analyzed in Section 4.17.

#### 5.4.17.2 Conclusions

The project would incrementally add to the cumulative adverse impacts noted in Section 4.17 and impacts will remain significant after mitigation.

# 5.4.18 Potential Cumulative Impacts Related To The Risk of Upset

The threshold of significance to determine risk of upset impacts is as follows:

(i) Does the project have risk of upset impacts that are individually limited but cumulatively considerable?

#### 5.4.18.1 Cumulative Impacts

Cumulative impacts pertaining to risk of upset refers to the likelihood or potential of the related projects and the Proposed Project to contribute to community hazards associated with explosions, fires, or release of hazardous substances in combination with other sources of such hazards. The risk of upset conditions associated with the other related projects include the spill of fuel and other hazardous materials during construction of these projects and the loading of fuel from aboveground or belowground storage tanks. The primary risk of upset associated with the Proposed Project relates to the potential exposure of nearby residents and/or persons using nearby roadways resulting from the planned on-site storage and use of aviation fuel, as well as its highway transport to the MCAS El Toro and JWA sites. Section 4.18 includes a mitigation measure that would reduce highway transport impacts due to OCX to a level of insignificance. No new or additional impacts are proposed at JWA.

#### 5.4.18.2 Conclusions

Fires or explosions involving the bulk storage of flammable or combustible liquids, while potentially serious or catastrophic, are isolated local events, typically restricted to industrial or similar areas. Because no facilities storing similar large volumes of such materials are known to exist or to be included in the related impacts near the related project areas or near the bulk fuel storage installations at OCX or JWA, the potential for these projects to contribute to significant cumulative community risk of upset is remote.

Therefore, the Proposed Project, considered with the other projects in the region, will not contribute to a cumulative significant adverse impact related to risk of upset conditions.

# 6.0 LONG-TERM IMPLICATIONS OF THE PROPOSED PROJECT

# 6.1 SIGNIFICANT IRREVERSIBLE ENVIRONMENTAL CHANGES THAT WOULD BE CAUSED BY THE PROPOSED PROJECT SHOULD IT BE IMPLEMENTED

CEQA Guidelines Section 15126.2 describes the issues for this section as follows:

"Uses of nonrenewable resources during the initial and continued phases of the project may be irreversible since a large commitment of such resources makes removal or nonuse thereafter unlikely. Primary impacts and, particularly, secondary impacts (such as a highway improvement which provides access to a previously inaccessible area) generally commit future generations to similar uses. Also irreversible damage can result from environmental accidents associated with the project. Irretrievable commitments of resources should be evaluated to assure that such current consumption is justified."

The construction of the land uses under the Proposed Project would require the commitment of a substantial amount of building materials such as concrete, asphalt, aggregate, steel, and other materials used in the construction of buildings, parking areas, streets, and infrastructure. However, the commitment of building materials under the Proposed Project would not be a significant adverse impact because the demand for these materials would be phased consistent with the phasing of the construction of the land uses, and these resources are generally considered to be readily available and in sufficient quantity in this region.

The Proposed Project would alter part of the site from the existing agricultural uses to urban uses. This would cause the irreplaceable loss of use of the agricultural soils on the Marine Corps Air Station (MCAS) El Toro site. This impact is significant and cannot be mitigated to a level below the level of significance (see Section 4.11, Natural Resources and Energy).

The implementation and operation of the land uses under the Proposed Project would require a long-term, irretrievable commitment of natural resources, including water, jet fuel, gasoline, diesel fuel, natural gas, and electricity. As described earlier in Section 4.10 (Public Services and Utilities) and Section 4.11 (Natural Resources and Energy), it is expected that the demand for these types of resources is within the capacity and capability of the applicable service providers to satisfy. Therefore, the commitment of these types of resources to the long-term operation of the land uses under the Proposed Project would not result in a significant adverse impact, except for consumption of jet fuel as compared to the existing condition. However, to the extent these resources are non-renewable and have finite limits, the project will have an impact on the long-term supply in generations to come absent technology changes to decrease or eliminate reliance on them.

The long-term operation of the airport and public streets under the Proposed Project would require a commitment on the part of the applicable parties to maintain and operate these

facilities. This is not expected to be a significant adverse impact of the Proposed Project, although it will require a commitment of financial, personnel, and facility resources by the applicable operators and agencies.

The protection of the wildlife habitat area and open space on the MCAS El Toro site is a beneficial effect of the Proposed Project related to natural resources.

In summary, the Proposed Project will not result in a significant adverse impact related to the commitment of resources for construction and operation of the proposed land uses on the MCAS El Toro site with the exception of encouragement of large amounts of jet fuel consumption when compared to the existing condition, as acknowledged in Section 4.11 and Chapter 5.0. However, the commitment of agricultural land for urban uses is a significant adverse impact of the Proposed Project related to natural resources that cannot be mitigated to below a level of significance. This impact was acknowledged in EIR No. 563 and the Supplemental Analysis and does not represent a new impact. The Proposed Project preserves additional acres of agriculture that were not preserved in the CRP project.

### 6.2 GROWTH INDUCING IMPACTS OF THE PROPOSED PROJECT

The text that follows is taken from EIR No. 563 because the growth inducing impacts of the Proposed Project have not changed substantially since EIR No. 563. The analysis provided in EIR No. 563 was not challenged in court. The text and analysis have been updated to reflect both changed conditions and the current Proposed Project.

No additional text has been inserted from the <u>EIR No. 563</u> Supplemental Analysis because this topic was not required to be addressed in the Supplemental Analysis.

Text that is no longer applicable is shown in strikeout. New text is shown in underline.

Note: The double underline and *italicized strikeout* in Chapter 6 reflects changes made to EIR No. 563 text for inclusion in EIR No. 573. Underline text indicates additions and strikeout indicate deleted text to reflect changes made to the Final EIR.

### 6.3 CEQA DEFINITION OF GROWTH INDUCING IMPACTS

Sections 15126(g), 15126(d) and 15126.2(d) of the CEQA Guidelines requires that an environmental impact report discuss the ways in which a proposed project could foster economic or population growth, or the construction of additional housing, whether directly or indirectly.

Direct growth inducing impacts are generally associated with the provision of urban services and the extension of infrastructure to an undeveloped area. The provision of services and facilities to a site can reduce development constraints for other nearby areas and can serve to induce further development in the vicinity of a project. Indirect, or secondary, growth inducing impacts consist of growth induced in the region by the additional demand for housing, employment, and goods and services associated with population increases caused by, or attracted to, new development.

Section 7.0 of this EIR 5.0 in this EIR No. 573 (Cumulative Impacts) discusses the potential cumulative impacts of the reuse alternatives and other existing, approved and planned projects in the vicinity of the Marine Corps Air Station (MCAS) El Toro site and JWA. To avoid extensive and unnecessary repetition, the information in Section 7.0 5.0 is not repeated in this section, although some of the identified cumulative impacts may be considered as potentially growth inducing as well as cumulative impacts. Section 4.17 (Socioeconomics) describes the source for the employment projections assumed under the reuse alternatives. It is important to remember that, in effect, a number of the defined objectives of the project (see Section 1.0 2.0 in this EIR No. 573) seek to foster economic growth and job creation.

### 6.4 POTENTIAL GROWTH INDUCING IMPACTS BY ALTERNATIVE

## 6.4.1 Potential Growth Inducing Impacts Of The Proposed Project (Reuse Alternative A)

The Proposed Project would result in a net increase over existing conditions of 23,500 jobs at MCAS El Toro (an increase of 400 percent) and 1,258 jobs at JWA (a 60 percent increase) in 2020, based on the total number of new jobs minus the existing military jobs at MCAS El Toro and existing jobs at JWA.

With regard to direct growth inducing impacts, the influence of the proposed project which would generate approximately 35,800-26,200 new jobs on the MCAS El Toro site and 3,300 jobs at JWA for a total of 29,500 jobs. The growth inducing impact of this increase in jobs would be greatest on nearby areas designated for urban development, or which are likely to become urban development property in the foreseeable future, but which are currently undeveloped. These include parcels of vacant land north, south and west of the MCAS El Toro site, some of which are currently in agricultural production. In the immediate vicinity of the MCAS El Toro site, in the area south of Loma Ridge, west of Limestone-Whiting Wilderness Park, north of the proposed alignment of Portola Parkway and west of the east leg of the Eastern Transportation Corridor (ETC), there are three distinct areas shown on the County of Orange Community Profile Map designated for low (1.2) dwelling units per acre (du/ac)) and medium (6.5 1.4-du/ac) density residential uses. These combined areas cover well over 1,000 acres. There are no specific development plans for these areas at the present time. The buildout of the land uses under the proposed project (Reuse Alternative A) and the associated infrastructure improvements on the MCAS El Toro site would make it easier to develop these areas since utilities and services would be increased and improved in the area near brought closer to them.

It is important to recognize that most some of the land use designations in these areas, particularly the low density residential designations, are subject to change as the Natural Community Conservation Planning (NCCP) process.—avolves. The Reserve Design under the NCCP includes large parts of these areas as natural reserves, particularly Upper Rattlesnake Canyon and Hicks Canyon. Although the majority of the NCCP Reserve is committed to public ownership, the Reserve will set aside an additional 3,000 acres of land previously planned for residential use antitled land along the frontal slopes of Loma Ridge through separate dedication agreements so that it will not now be developed. However, development of new residential uses north, northwest, west and south of the MCAS El Toro site is more likely to occur in response to local and regional economic conditions and forces than in response to the development of other projects in the vicinity of the MCAS El Toro site. The Central/Coastal Orange County Subregion NCCP has now been adopted. This does not change the conclusions of this section.

With regard to indirect growth impacts, the estimated 24,758 29,400 new jobs which could locate on the MCAS El Toro and JWA site under the proposed project would generate increased demand for goods and services. This increased demand for goods and services would create additional jobs in the surrounding region. In addition, it is expected that a portion of the visitors to Orange County that would use El Toro in the year 2020 would spend less time and money in Orange County not visit the region if commercial air service were not available at El Toro and they had to enter the region through another airport outside of Orange County (Technical Report No. 16, Economic Benefits Study, November 5, 1999). This is due to expected constraints in the regional aviation system as existing and planned airports reach capacity. Based on a study of the employment and economic impacts of the reuse alternatives at MCAS El Toro (EPS, July 1996), this additional employment could amount to an estimated 112,800 jobs in Orange County, bringing the total estimated number of direct and indirect jobs generated under this reuse alternative to 148,600 compared to 11,500 jobs under continued military use.

Multiplier impacts will include increases in employment created by successive rounds of spending and re-spending of income generated by direct economic activities associated with development on the project site under the Proposed Project. Indirect and multiplier effects of the direct on-site jobs will create another 16,200 indirect and multiplier effect jobs by 2020, for a total of 45,700 jobs in Orange County being supported by economic activities occurring on El Toro and JWA. The average personal income of these direct, indirect and multiplier jobs is projected to be \$37,400, higher than the average personal income per job countywide (Technical Report No. 16, Economic Benefits Study, November 5, 1999).

In addition, visitors that use OCX and JWA will spend a portion of their trip duration in Orange County. During their stay, these visitors spend money on lodging, food and beverage, retail goods and services, recreation, and ground transportation. These expenditures support additional job growth in the vital tourism and business travel services sector of the Orange County economy. Airport activity will also generate additional expenditures by persons booking airline tickets using Orange County travel agents and by airline crews on layover in the County. By 2020, these expenditures in Orange County are expected to support 76,900 jobs (Technical Report No. 16,

Economic Benefits Study, November 5, 1999). Multiplier effects of these visitor expenditures will support another 23,500 jobs, for a total of 100,400 jobs supported by visitors using OCX and JWA. As is typical with the visitor service sector employment in Orange County and elsewhere, the average personal income due to these jobs is lower than the Countywide average, at \$22,600 per job (Technical Report No. 16, Economic Benefits Study, November 5, 1999).

In total, economic activity to be generated by aviation and nonaviation activities at OCX and JWA, as well as by tourism-related jobs generated by visitors to Orange County using OCX and JWA, are anticipated to support a total of 146,100 jobs County-wide by 2020 (Technical Report No. 16, Economic Benefits Study, November 5, 1999).

While the direct employment generated by the Proposed Project activity will be onsite at the OCX and JWA sites, the multiplier effect employment, as well as the visitor-related employment, will largely occur off the project site. Some of this indirect and multiplier effect employment will be located in areas adjacent to OCX and JWA, as well as at locations throughout the County. As noted in Section 4.17, the impacts of the Proposed Project related to inducing substantial growth or concentration of non-resident employee population cannot be mitigated to below a level of significance.

Some of this indirect employment would locate in areas adjacent to, or in the vicinity of the MCAS El Toro site. It is anticipated that increased demand will be placed on the recreational, restaurant and other business and personal services available at the Irvine Spectrum area south and west of MCAS El Toro site and other areas to be developed in the future surrounding the MCAS El Toro site. This is a significant growth inducing impact of the proposed project. (Reuse Alternative A).

Employees of the aviation, commercial, industrial, institutional and recreational uses on the MCAS El Toro site under the proposed project (Reuse Alternative A) are expected to be drawn from the existing and projected labor pool in the southern California region. Some employees may choose to relocate to housing closer to the MCAS El Toro site. This would occur over time, as the land uses under the proposed project (Reuse Alternative A) are developed. The resulting increased demand for housing over time would be satisfied by existing housing resources and new residential units that would be developed in the MCAS El Toro area. As noted in Section 4.17, impacts of the Proposed Project related to increased resident population growth and the related demand for all types and prices of housing in the surrounding area cannot be mitigated to below a level of significance. Based on the amount of planned and undeveloped residential land in the surrounding area, a substantial portion of this induced housing growth would be absorbed in residential projects currently under development. According to Economic and Planning Systems, Inc. (EPS, July 1996), There are approximately 51,000 55,000 additional housing units available or yet to be built in residential planned communities. Impacts of the Proposed Project related to reducing the supply of available housing in the County, however, cannot be mitigated to below a level of significance. Regardless of the housing in the planned communities, the proposed project (Reuse Alternative A) will generate a need to develop, and would result in growth inducing impacts related to, housing.

The direct impacts of the Proposed Project will slightly reduce the forecasted jobs/housing ratio in a jobs-rich area relative to the adopted region growth forecasts; the Proposed Project will not result in significant adverse impacts related to the projected jobs/housing ratio.

The conclusion is different than the conclusion reached in Final EIR No. 563, which found that the CRP would have a significant adverse impact related to projected jobs/housing ratio. However, that finding was based on a comparison with the jobs/housing ratio expected under the previously adopted Orange County Projections 1992 Modified regional growth forecasts. Those forecasts have been superseded by Orange County Projections 1996 Modified (OCP-96 Modified), which was used in this analysis, and result in a revised finding of no significant adverse impact.

Another growth inducing effect associated with the proposed project (Reuse Alternative A) is the conversion of agricultural uses on the north and south side of the MCAS El Toro site to urban uses. Under the proposed project (Reuse Alternative A), a regional park and airport parking and the Marshburn Retarding Basin—would replace the existing agricultural uses on the north part of the MCAS El Toro site. Business park uses would replace agricultural uses at the southern end of the site adjacent to the I-5 Freeway. Research and development, light industrial and institutional land uses would replace the existing agricultural uses at the end of Runways 34L and 34R on the southern part of the base. There will still be a loss of agricultural land, but 138.93 acres will be preserved for agriculture compared to no acres of agriculture in the CRP project. Conversion of these areas would reduce the total amount of agricultural land in unincorporated Orange County and could also negatively impact continued farming in the remaining agricultural areas in the vicinity of the MCAS El Toro site. Therefore, the conversion of this agricultural land will result in significant growth inducing impacts adversely affecting agricultural resources in this part of Orange County.

Under the proposed project (Reuse Alternative 4), the total area within the 65 dB CNEL noise contour would probably be reduced compared to the Policy Implementation Line (PIL) identified for MCAS El Toro in the current Noise Element of the Orange County General Plan. See the discussion regarding the 65 dB CNEL contour PIL and related issues in Section 4.4 (Noise). The project includes a If, ultimately, the Noise Element amendment is amended to define a reduced 65 dB CNEL PIL; therefore, the it is possible that land uses outside the redefined PIL could be converted to different uses or could be developed essentially as they are designated in the existing General Plans. The areas most likely to be converted include Planning Areas 6, 9, and 33 of the City of Irvine's General Plan located northwest, west, and southwest of the MCAS El Toro site. These areas are currently restricted by the Airport Environs Land Use Plan due to military aircraft noise and accident potential zones, which will be reduced or eliminated under the Proposed Project.

For analysis purposes, Planning Area 9 is estimated to accommodate about 8,900 dwelling units (which would generate approximately 83,700 daily trips), Planning Area 6 is estimated to include 5,800 dwelling units (which would generate approximately 54,300 daily trips), and Planning Area 33 would accommodate about 1,200 dwelling units (which would generate approximately 8,600 daily trips). The total projected increase would be about 15,900 dwelling units and approximately 146,600 daily trips. This scenario represents one potential conversion approach of this area to

provide evidence of probable impacts. However, development plans have not been proposed at this time and, therefore, this scenario represents a preliminary estimate. Note that OCP-96 Modified, which was adopted by the Orange County Council of Governments and the Board of Supervisors in 1997, includes employment in Planning Areas 6 and 9 that was forecast to result from development of the CRP. Approximately 650 acres of nonresidential land use areas could potentially be developed instead as residential or other noise sensitive uses, although this number could be larger or smaller depending upon the criteria used to select a redefined PIL. This potential land use conversion could also be reflected in the future average daily traffic (ADT) generated in these areas due to a different land use mix. With the approximate acreage identified above, the ADT in the area may decrease by 84,265 from the ADTs which would result from existing General Plan permitted land uses, based on typical trip generation factors assigned to those land uses. This trip generation difference, when applied to the average trip length, in turn, has implications for vehicle miles travelled (VMT) and regional air pollutant emissions. These potential impacts are analyzed and discussed in Section 2.0 5.0 (Cumulative Impacts).

An analysis of the post-2020 traffic impacts A discussion of non-planned areas adjacent to the MCAS El Toro site, but within the existing PIL is has been included in Section 2.0 5.0 (Cumulative Impacts). These non-planned areas are currently designated for agricultural and open space uses, but may be proposed for other uses in the future. Based upon the speculative nature of such changes, no significant adverse growth inducing impacts can be identified with certainty at this time.

# 7.0 SIGNIFICANT UNAVOIDABLE ADVERSE IMPACTS

#### 7.1 OVERVIEW

This section of this Environmental Impact Report (EIR) summarizes the potential significant unavoidable adverse impacts that could result from the implementation of the Proposed Project. These impacts are adverse impacts that cannot be avoided and that cannot be mitigated to below a level of significance based on implementation of the mitigation measures provided earlier in Chapter 4.0 (Environmental Setting, Impacts and Mitigation Measures) of this EIR. The summary of significant unavoidable adverse impacts provided in this section is based on the detailed environmental analysis provided earlier in Chapter 4.0.

The impacts of the Proposed Project found not to be significant or that can be mitigated to below a level of significance based on implementation of the mitigation measures described earlier in Chapter 4.0 of this EIR are discussed in the detailed environmental analysis provided earlier in Chapter 4.0.

### 7.2 SIGNIFICANT UNAVOIDABLE ADVERSE IMPACTS UNDER THE PROPOSED PROJECT

As described in detail earlier in Chapter 4.0, the Proposed Project is anticipated to result in the following significant adverse impacts that cannot be mitigated to below a level of significance after implementation of relevant existing regulations and the identified mitigation measures:

- Noise impacts due to increased aircraft operations and nighttime aircraft operations in the vicinity of the El Toro site.
- Noise impacts on the use of 1) the proposed on-site recreational facilities; 2) existing local parks and open space areas in the northern part of the City of Lake Forest; 3) future off-site trails, including portions of the future Borrego Canyon Bikeway, the future Jeffrey Road Bikeway, and the future Hicks Canyon Trail; and 4) portions of Class II on-road bikeways on Alton Parkway, Portola Parkway, Bake Parkway, and Lake Forest Drive.
- Loss of agricultural soils.
- Encouragement of use of large amounts of jet fuel when compared to the existing conditions.
- Short-term air quality impacts during construction.
- Public acute health impacts due to potential toxic air contaminant emissions during operation.
- Demand for all types and prices of housing, including low and moderate income housing, and impacts of inducing substantial growth or concentration of non-resident employee population, and reducing the supply of available housing in the County.

- Air Quality: short-term impacts during construction and long-term regional air quality impacts and local air quality impacts due to aircraft and associated operations at JWA and OCX.
- Public cancer risk and noncancer acute health hazard impacts due to potential toxic air contaminant (TAC) emissions during operation.
- Public health and safety risk of upset conditions if the County, despite every reasonable effort
  to lease or otherwise obtain the use of the Norwalk Pipeline and the Santa Fe Pipeline for the
  purpose of conveying jet fuel to the MCAS El Toro site, is unable to do so to the extent
  necessary to reduce the potential risk of upset impacts associated with highway truck
  transport of jet fuel to a level below significant.

The Proposed Project will also contribute to potentially significant cumulative adverse impacts related to: noise; air quality; toxic air contaminants; water quality; water; fossil fuels; and socioeconomic impacts related to low and moderate income housing; and impacts of inducing substantial growth or concentration of non-resident employee population, and reducing the supply of available housing in the County to low and moderate income housing. The Proposed Project's contribution to cumulative impacts in the areas of traffic, biological resources, and hydrology/water quality are reduced to below a level of significance with the proposed mitigation measures.

# 7.3 SIGNIFICANT UNAVOIDABLE ADVERSE IMPACTS IF MITIGATION BY OTHER ENTITIES IS NOT IMPLEMENTED

The following environmental impacts would remain significant after mitigation in the event that other public agencies and/or entities partially or wholly responsible for the mitigation measures provided in Section 6.4 do not implement those measures:

- Project related land use incompatibility with General Plans of adjacent jurisdictions and/or planned land uses of adjacent landowners (ref. Mitigation Measure LU-2).
- Land use impacts of the Proposed Project to surrounding cities (ref. Mitigation Measure LU-3).
- Undesirable land uses or developments (ref. Mitigation Measure LU-6).
- Inconsistency with SCAG Core RTP Objective Meet Air Plan Emissions Budgets (ref. Mitigation Measure GPC-4).
- Lack of Airport Land Use Commission approval of the ASMP (ref. Mitigation Measure GPC-1).
- Potential zoning inconsistencies with future development proposals within the Proposed Project's 65 CNEL contour (ref. Mitigation Measure GPC-2).
- Inconsistency of the Proposed Project with the City of Irvine's General Plan designations for the portion of the former MCAS El Toro site within the City's jurisdiction (ref. Mitigation Measure GCP-3).

- Intersection and arterial circulation impacts in non-County jurisdictions for which the County would be responsible for partial contribution to the improvements (ref. Mitigation Measure T-4).
- Circulation impacts to highways not under the sole control of the County (Mitigation Measure T-9).
- Circulation impacts related to project facilities that connect to right-of-way or property under the jurisdiction of non-County entities (ref. Mitigation Measure T-10).
- Noise impacts upon existing sensitive land uses (Mitigation Measure N-1).
- Traffic noise impacts at portions of Trabuco Road between Yale and Jeffrey Road and along Jeffrey Road south of Irvine Boulevard (ref. Mitigation Measure N-6).
- Public cancer risk and noncancer acute health hazard impacts due to potential toxic air contaminant (TAC) emissions during operation.
- Noise impacts from project construction in incorporated off-site areas (ref. Mitigation Measure N-7).
- Impacts to existing wells and pipelines owned by The Irvine Company (ref. Mitigation Measure U-4).
- Impacts to transit services (connection to the Irvine Transportation Center) (ref. Mitigation Measure PS-4).
- Loss of prime agricultural land that would otherwise be preserved as agricultural by County Sheriff-Coroner's Department (ref. Mitigation Measure NRE AG-1).
- Potential unavailability of agricultural lands for lease (ref. Mitigation Measure NRE AG-2).
- Inconsistency with the Southern California Association of Governments Regional Plan Population, Housing and Employment Growth Forecasts (ref. Mitigation Measure SE-1).
- Potential risk of upset due to highway truck transport of jet fuel (Mitigation Measure RU-1).
- Project related cumulative traffic and circulation impacts (ref. Mitigation Measure T-12).
- Inconsistency with the County's Master Plan of Arterial Highways relative to cumulative long-range circulation improvements (ref. Mitigation Measure T-13).

#### 8.0 ALTERNATIVES

#### 8.1 INTRODUCTION

#### 8.1.1 Final EIR No. 563 Alternatives

Final Environmental Impact Report (EIR) No. 563 for the Community Reuse Plan (CRP) for the Marine Corps Air Station (MCAS) El Toro considered a number of possible reuse alternatives including:

- i) Reuse Alternative A, which proposed a commercial passenger and cargo airport at El Toro, surrounded by nonaviation uses including a Habitat Reserve, educational and institutional uses, residential uses, recreation and open space uses, research and development/light industrial uses, a meeting center, mixed retail/office/commercial uses, office and conference center uses, and multimodal surface transportation center. Reuse Alternative A assumed commercial operations at JWA would cease. The County of Orange Local Redevelopment Authority (LRA) adopted Reuse Alternative A as the CRP.
- ii) Reuse Alternative B, which proposed a commercial airport limited to cargo and general aviation operations at the MCAS El Toro site, surrounded by nonaviation uses including a Habitat Reserve, educational and institutional uses, residential uses, recreation and open space uses, research and development/light industrial uses, office and conference center uses, and a multimodal surface transportation center. Reuse Alternative B assumed all commercial passenger operations would be provided by JWA.
- iii) Reuse Alternative C, which proposed a wide range of nonaviation uses at El Toro, including a Habitat Reserve, visitor oriented attractions, residential uses, recreation and open space uses, research and development/light industrial uses, educational and institutional uses, mixed retail/office/commercial uses and a multimodal surface transportation center. Reuse Alternative C assumed all commercial, cargo and general aviation passenger operations would be provided by JWA.
- iv) No Project Alternative D, which assumed the military would retain ownership and operation of the MCAS El Toro site and that operations would continue at 1994 levels.
- v) No Development Alternative E, which assumed the military would leave the site and the site would be vacant and unplanned.

County of Orange Final EIR No. 573

The 970 acre Habitat Reserve in Planning Area 6 is subject to a federal agency to federal agency transfer, and is not part of the Proposed Project.

Final EIR No. 563 also considered secondary alternatives to Alternatives A and B, which essentially considered different airport configurations or operating conditions compared to Alternatives A and B. Final EIR No. 563 also considered alternative sites for the proposed airport use, as described in detail later in Section 8.12.5.1 (Alternative Sites Evaluated in Final EIR No. 563). Section 15126.6(f)(2)(c) of the CEQA Guidelines permits reliance in this document on the analysis provided in EIR No. 563:

"Limited new analysis required. Where a previous document has sufficiently analyzed a range of reasonable alternative locations and environmental impacts for projects with the same basic purpose, the lead agency should review the previous document. The EIR may rely on the previous document to help it assess the feasibility of potential project alternatives to the extent the circumstances remain substantially the same as they relate to the alternative."

#### 8.1.2 ASMP Alternatives

With the starting point of the CRP and the Board's direction to develop a two airport system, the Airport System Master Plan (ASMP) analyzed a broad range of airport development options for MCAS El Toro and JWA. Preliminary screening of these options described "families" of potential airport system solutions, and consisted of two components for each airport:

- i) an airport role (type of service provided) and
- ii) airport facility improvements.

Technical Report 6, Alternatives Definition Report, developed in April, 1998, was prepared to analyze preliminary screening scenarios that cover a broad range of possible airport system options for Orange County. Other documents used to conduct the alternatives analysis include: Working Paper 2, List of Preliminary Project Planning Issues; Technical Report 1, Airport System Feasibility; Technical Report 2, Planning and Performance Parameters; Technical Report 3, Existing Facilities; and Technical Report 4, Aviation Demand Forecasts. The selection of alternatives analyzed by the ASMP (and also the present EIR) focused on alternatives to the Proposed Project which meet the planning goals and criteria established by the following:

- Orange County Board of Supervisors December 11, 1996, Resolution No. LRA R96-02, which adopted the Community Reuse Plan (CRP) for MCAS El Toro and initiated the ASMP.
- ii) Policies established in the Orange County General Plan by Measure A, approved in 1994.
- iii) The need, as part of the Master Development Program (MDP) planning process, to address issues of unique importance to the planning of an airport system in Orange County.

- iv) The need to address issues of special importance to the public and the Board of Supervisors.
- v) California Environmental Quality Act (CEQA) criteria for the definition of alternatives.

Please see the ASMP (Technical Report 17) for a complete description of the ASMP alternatives evaluation.

#### 8.1.3 Introduction to EIR Alternatives

Section 15126.6(a) of the CEQA Guidelines indicates the scope of alternatives to a Proposed Project that must be evaluated:

"An EIR shall describe a range of reasonable alternatives to the project, or to the location of the project, which would feasibly attain most of the basic objectives of the project but would avoid or substantially lessen any of the significant effects of the project, and evaluate the comparative merits of the alternatives. An EIR need not consider every conceivable alternative to a project. Rather it must consider a reasonable range of potentially feasible alternatives that will foster informed decisionmaking and public participation. An EIR is not required to consider alternatives which are infeasible. The lead agency is responsible for selecting a range of project alternatives for examination and must publicly disclose its reasoning for selecting those alternatives. There is no ironclad rule governing the nature or scope of the alternatives to be discussed other than the rule of reason."

As described in detail earlier in Chapter 4.0, the Proposed Project is anticipated to result in significant adverse impacts that cannot be mitigated to below a level of significance after implementation of relevant standard conditions of approval, regulations, and mitigation measures. In summary, these unavoidable impacts are as follows:

- Significant adverse noise impacts due to increased aircraft operations and nighttime aircraft operations.
- Significant unavoidable adverse noise impacts to the use of 1) the proposed on-site
  recreational facilities; 2) existing local parks and open space areas in the northern part of
  the City of Lake Forest; 3) future off-site trails, including portions of the future Borrego
  Canyon Bikeway, the future Jeffrey Road Bikeway, and the future Hicks Canyon Trail;
  and 4) portions of Class II on-road bikeways on Alton Parkway, Portola Parkway, Bake
  Parkway, and Lake Forest Drive.
- Significant loss of agricultural resources.
- Short-term air quality impacts during construction.

- Significant adverse impact to public health due to potential toxic air contaminant emissions during operation.
- Significant adverse impacts related to the demand for all types and prices of housing, including low and moderate income housing, and impacts of inducing substantial growth or concentration of non-resident employee population, and reducing the supply of available housing in the County.

The Proposed Project will contribute to potentially significant cumulative adverse impacts related to: land use related to the change in the area covered by the 65 dBA CNEL contour; noise; air quality; water quality; energy resources; water; and socioeconomic impacts related to low and moderate income housing; and impacts of inducing substantial growth or concentration of non-resident employee population, and reducing the supply of available housing in the County to low and moderate income housing.

In this light, this chapter presents a reasonable range of alternatives to the Proposed Project. These alternatives include the following:

- i) No Project/No Activity
- ii) ETRPA Nonaviation Plan Alternative
- iii) Alternative A: JWA Status Quo Aviation Roles (Reduction to 6 Million Annual Passengers or MAP); OCX Full Domestic Service Airport (19 MAP)
- iv) Alternative C: JWA Short-Haul Domestic (10.1 MAP); OCX Full International Service Airport, excluding Short-Haul Domestic (23.4 MAP)
- v) Alternative F: JWA Short, Medium and Limited Long Haul Passenger and Cargo Service with No Operational Limitations and No General Aviation Use (14 MAP); OCX No Aviation Reuse
- vi) Alternative G: JWA General Aviation and Cargo/Passenger Service from Short Haul to Limited International (25 MAP); OCX No Aviation Reuse
- vii) Alternative H: JWA Status Quo (10.8 MAP); OCX Limited Use (10 MAP)
  Domestic Service Airport
- viii) Alternative I: JWA Status Quo Aviation Roles (7 MAP); OCX Limited Use (15 MAP) Domestic Service Airport
- ix) Alternative J: JWA Reduced Service (5.4 MAP); OCX Full International Service Airport at OCX at (28.8 MAP) with Widely Separated North/South (N/S) Runways
- x) Alternative OCX Airport Runway Layout (Wildlands Ranch Plan Alternative)
- xi) Land Use Alternatives at OCX Nonaviation Land Use Component
- xii) Alternative K: Off-Site Alternatives (JWA 8.4 MAP)
- xiii) Alternatives Considered. But Rejected
- xiv) For comparison purposes, the data regarding the CRP, as adopted in December, 1996, are carried forward.

As indicated above, this section also presents a number of alternatives to the Proposed Project not carried forward for further analysis and the rationale for their exclusion. Table 8.1-1 provides a summary comparison of the aviation characteristics, trip generation, vehicle miles traveled, aircraft noise impacts, and air quality emissions for existing conditions and each alternative analyzed herein. In addition, the last part of this section presents a matrix comparison of the environmental impacts of the alternatives considered in detail.

A summary of the aviation activity under the Proposed Project and the aviation alternatives to be carried forward is provided in Table 8.1-2.

Table \$.1-1
Saumary Comparison of Principal Aviation Churacteristics and Major Impacts for the CRP, Existing Conditions and Alternatives

SUBMARY OF AVIATION CHARACTERISTICS  Million Annual Emberages Little OCK 37MA Total 3.  Adams Africa Decresses OCK 447 PMA 377 Total 417  MILLION AND AND HAPPACTS Transpartension 18 Total Size ADT Generated 19 ADT Generated Repeated Vehicles Miles Trevelled Responds Totals Miles Trevelled Seige (No. of Residences Within Each CRUEL)	13	Existing Conditions	No Project	Proposed Project	ETRPA								Alternative OCX	Lond	
SHIRE ANNUAL PROCESSOR AND A STATE OF THE AND A STA	0	6				A	C	F	G	N		1	Reservey Design	Une	. K
OCX PWA Treal 3 Anamel Aircraft Operations CCX OCX PWA 177 Treal 177 Treal 178 Treal 1	0						·						,	·	
TOTAL TOTAL	0				l		1		Į				1	( 1	ĺ
Treal  3 Annual Aircraft Chokrasises  CCC  441  WA 377  Tonel  812  BURNALARY OF MAJOR IMPACTS  Transpacturing  E1 Two Size ADT Generated  WA ADT Generated  Regulated Vehicles Miles Trecelide  Regulated Vehicles Miles Trecelide  Notes (No. 48 Renderans Within Kach CNEL)			a	26.8		19	23.4	. 6	6	10	15	24.1	Ser	28.8	
Anamol Aircraft Operations OCX 198A 37: Total SUMMARY OF MAJOR IMPACTS Transmertanion ET Two Size ADT Generated St Two Size ADT Generated Request Vehicles Miles Trevelled Request Vehicles Miles Trevelled String (No. of Reprinted No. of Reputer Vehicles Miles Trevelled String (No. of Reputers No. of Reputer No. of Reputer No. of Reputer No. of Reputer No. of Reputer No. of Reputers No. of Reputer No. of No. of Reputer No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of No. of		7.5	8.4	54	8.4		10.1	14	25	10.8	7	5.4	Analysis	3,4	8.4
OCX PWA 377 PWA 377 Total BUMMARY OF MAJOR IMPACTS Transmitted El Tron Size ADT Generated 505. PWA ADT Generated Regional Verbids Miles Trevelled Regional Temb Miles Trevelled String (No. of Replication William Temple)	ы )	7.5	9.4	34.2	8.4	23	33.5	14	25	20.3	72	342	, ,	34.2	<b>k</b> 4
TWA STOOM STANDER IMPACTS  Internated to the Control of Standard S	-	)			,										ĺ
Took  ### ### ### ### ### ### ### ### ### #	.000	30,217	0	300,608	σ	252,400	1911,3800	0	0	314,200	202,900	300,600	See	360,660	
BEURRARHY OF MAJOR IMPACTS Transpacturing ET Two Size ADT Generated PMA ADT Generated Regaust Vehicles Miles Trevelle Regaust Treash Miles Trevelle Miles Trevelle Miles Trevelle Miles Trevelle Miles Trevelle Miles Trevelle	,09G	417,725	462,690	426,700	462,000	432,300	464,296	172,000	347,600	319,800	444,196	426,790	Asaiysh	126,700	462,00
Transportation  El Toro Sile ADT Generated  10 Toro Sile ADT Generated  100.	,000	447,942	462,900	727,300	462,000	684,700	656.00Q	172,100	347,600	634,800	6531,900	727,300		727,300	462,00
El Toro Sie ADT Generated 305.  WA ADT Generated Regular Vehicle Miles Trevelled N. Regional Vehicle Miles Trevelled N.  Voice (No. of Residences Within Kach CNEA)	- {	1												j 1	
El Toro Sie ADT Generated 305.  WA ADT Generated Regular Vehicle Miles Trevelled N. Regional Vehicle Miles Trevelled N.  Voice (No. of Residences Within Kach CNEA)	- {	ì							i				į l	1 /	ŧ
PWA ADT Generated Regional Vehicles Miles Travelled Regional Transh Miles Travelled Regional Transh Miles Travelled Noise Clic. of Residences Within Each CNEA	240(1)	25,400		176.123	391,322	138,157	152,273	39 (,3229)	391,322	109,900	142,400	176,123	See	150,22214	
Regional Transk Miles Travelled	-	47,450	51.706	32,690	51,706	36,274	48.376	#3,943	116,424	48.390	44,000	32,690	Assivais	32,690	51.70
Saint Clin. of Residences Within Each CNEL		321,922,797	447.831.642	442.069.732	447.787.665	442.115.204	See	442,679,260		442.289.168	462 158 695	See	, .—,.— 1	See	See N
	A	464,798	494,777	502.388	506,371	501.478	Project	494,777	498,585	501,478	201,478	Project		Project	Ртојев
OCX ENEL 70												1	ļ		ĺ
	• !	0	D	9		ę	a			0	9	8	Sec		i e
	12	9	6	0		0	ŧ	0	•	9		525	Amelytis	0	
OCX CNEL 60 2.	134	0	0	1,837	} e	1,312	717	ا ہ	Đ	394	787	3,451		1,#37	
IWA CNEL 70	a Ì	ا د		0		0	ě	[ ه ]	446	e		· a !	NA Í	0 1	(
IWA CHEL 65	)	134	236	79	236	79	236	184	1,463	236	236	70	NA	79	236
JWA CNEL 60	,	682	991	\$77	997	557	1,623	997	6,954	1,023	991	577	NA.	577	997
Total CNES, 30	ا د	a		a		a		1	44ó		a		أسعا		ء ا
Total CNEL 65		134	236	29	NA.	79	236	184	1.863	236	236	464	Assilvata	76	134
Total CREL 60 2:		6472	997	2,414	NA.	1,369	1,110	997	6,954	1,417	1,784	3,914	,	2,414	597
his Onabley (Data are the change from No Project)*  Air Quality Regional Pollutant Emissions ((builty))													, ,		
CO As							!					_	!		١.
		1,521,279	2,971,017	34,789	-62,954 <sup>19</sup>	31,064	-35,724	8,522	-20,958	No New	No New	Ser	Sec	See	Ser
NOs -d.		483,451	597,999	-3.316	-7,931	4,585	-5,698	-2.459	-5319	Lines	Des	Project	Audysia	Project	No Fro
ROC 3.		109,255	\$6,629	1,698	9,440	-1,269	-1,546	-1,007	-2,530	Avadable	Ayeliable	Date		Deta	Dana
50x -53	499	44,967 \$.863	57,062 11,428	49 -225	-52,483 1,836	-139 -246	#5 -320	-#9 -10	-279 20			i			í

<sup>\*</sup> This inflammance has been applied from the Draft EIR or reflect the analyses in the USA.

 CMP angusts on for the D Toro in only. TNA I now methods.
 CMP angusts on for the D Toro in only. TNA I now methods.
 Administrar in med is would repet to a sequenciar pain as ET year and to the ETEPA Alternative SI ADT for LLU J.111 ADT or executed in 13,144
 Administrar in the THEPA Alternative in the CMP and I are the CMP and I

County of Orange Florid ESR No. 373

Table 8.1-2
Summary of Aviation Activity at Orange County Airports Under Alternative Airport System Development Scenarios

	r Prop	osed Project		e siete Ferenda	ternative A		1	Iternative (			Alternative 2020	E
Type of Operation	OCX	2020 JWA	Total		JWA	Total	oex .	2020 JWA	Total	oex.	THE PROPERTY AND ADDRESS OF THE PARTY OF THE	Total
Air Passengers (millions)												
Domestic	20.6	5.4	26.0	18.7	6.0	24.7	14.0	10.1	24.1	**	8.4	8.4
International	8.2	**	8.2	0.3		0.3	9.4	***	9.4	**	-	
Total	28.8	5.4	34.2	19.0	6.0	25.0	23.4	10.1	33.5	**	8.4	8.4
Air Cargo (millions U.S. tons)												
Domestic	1.19	0.02	1.21	1.21	0.02	1.23	1.18	0.05	1.23		0.05	0.03
International	0.82	***	0.82	0.04	***	0.04	0.84		0.84			
Total	2.01	0.02	2.03	1.25	0.02	1.28	2.02	0.05	2.07	***	0.05	0.0
Based Aircraft	14	570	584	20	567	587	9	503	512	**	582	582
Aircraft Operations												
Passenger	251,100	67,500	318,600	196,000	75,100	271,100	150,200	147,000	297,200	**	95,100	95,10
All-Cargo	26,600		26,600	22,600		22,600	26,000		26,000	**		
General Aviation	22,000	359,000	381,000	33,000	357,000	390,000	15,000	317,000	332,000		366,700	366,700
Military	900	200	1,100	800	200	1,000	600	200	800	**	200	20
Total	300,600	426,700	727,300	252,400	432,300	684,700	191,800	464,200	656,000		462,000	462,000
							<u></u>					

<sup>&</sup>lt;sup>1</sup> Alternative J has the same activity levels as the Proposed Project.

Table 8.1-2
Summary of Aviation Activity at Orange County Airports Under Alternative Airport System Development Scenarios

A.	Alternative F 2020				Alternativ 2020	G	er gegen Harer ta	Alternatis 2020	eН	Alternative I 2020			
Type of Operation	OCX	JWA	Total	ocx	JWA	Total	ocx .	JWA	Total	ocx	JWA	Total	
Air Passengers (millions)													
Domestic	***	14.0	14.0		24.7	24.7	9.9	10.8	20.7	14.8	7.0	21.9	
International	**	**		**	0.3	0.3	0.1	**	0.1	0.2	•	0.2	
Total	404 304	14.0	14.0		25.0	25.0	10.0	10.8	20.8	15.0	7.0	22.0	
Air Cargo (millions U.S. tons)													
Domestic	***	0.18	0.18	***	1.23	1.23	1.06	0.05	1.11	1.15	0.03	1.18	
International			****	***	0.04	0.04	0.03		0.03	0.04	**	0.04	
Total	*****	0.18	0.18	**	1.28	1.28	1.09	0.05	1.14	1.19	0.03	1,22	
Based Aircraft	***		We Aft		20	20	294	293	587	12	567	579	
Aircraft Operations													
Passenger	***	161,700	161,700	***	273,900	273,900	106,300	135,100	241,400	159,40 0	90,700	250,100	
All-Cargo	••	3,700	3,700		28,300	28,300	22,600	***	22,600	22,600	***	22,600	
General Aviation		6,600	6,600		45,300	45,300	185,200	184,600	369,800	27,300	357,200	384,500	
Military	~=	100_	100	••	100	100	100	100	200	600	200	800	
Total	**	172,100	172,100	**	347,600	347,600	314,200	319,800	634,000	209,90	448,100	658,000	
										0			

# 8.2 NO PROJECT/NO ACTIVITY ALTERNATIVE (ALTERNATIVE E): JWA – STATUS QUO AVIATION ROLES; NO AVIATION REUSE AT FORMER MCAS EL TORO

This section presents the potential impacts of the No Project/No Activity Alternative as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

#### 8.2.1 Aviation Uses

Under the No Project/No Activity Alternative, JWA would continue to operate as it does presently, providing general aviation service, short- and medium-haul domestic passenger service (with limited long-haul service), and very limited all-cargo service. JWA would be constrained to 8.4 MAP in the year 2020 under this alternative. There would be no aviation reuse of MCAS El Toro, and the site would remain vacant and undeveloped. Therefore, aviation demand projected to use Orange County airports under the Proposed Project would need to use other airports in the region. This is discussed further in Section 8.2.4.1 below.

# 8.2.2 Nonaviation Revenue Support Uses

Under the No Project/No Activity Alternative, the MCAS El Toro would remain vacant and undeveloped, with no nonaviation uses.

# 8.2.3 Attainment of Project Objectives

This alternative would not meet any of the general project objectives identified in Chapter 2, Section 2.3 3.0 regarding base redevelopment. This alternative would not meet the aviation objectives relating to passenger and cargo demand, service opportunities, industry competition, economic growth, business activities, existing land use restrictions, or General Plan implementation. The No Project/No Activity Alternative would meet or partially meet the aviation objective relating to general aviation by maintaining GA uses at JWA.

# 8.2.4 Environmental Impacts of the No Project/No Activity Alternative

#### 8.2.4.1 Land Use

A vacant and undeveloped site at MCAS El Toro would be incompatible with adjacent or nearby land uses. No activity at the El Toro site would be inconsistent with the Base Realignment and Closure (BRAC) objectives to transfer closed bases and mitigate economic loses in the community. No activity would eliminate revenue to the United States Department of Defense (DOD) to offset maintenance. No activity would lead to decay and vandalism. This alternative is comparable to the Proposed Project at JWA in that no significant adverse impacts to land use would occur. This alternative would not avoid or lessen impacts compared to the project.

As discussed at the outset of this section, under the No Project/No Activity alternative, aviation demand projected to use Orange County airports under the Proposed Project would need to use other airports in the region. To evaluate the ability of the regional aviation system to accommodate this additional demand, the allocation model used to project commercial aviation demand at each airport in the region for the Proposed Project in 2020 was used to forecast and analyze the redistribution of regional demand under the No Project/No Activity Alternative. The model and assumptions used to project demand under the Proposed Project is documented in Appendix B, Technical Report 6, Alternatives Definition Report, April 17, 1998, revised October 15, 1999.

Additional research was undertaken in late 1998 to identify potential constraints at commercial service airports in the region that might affect their ability to accommodate future commercial aviation demand. This research found that capacity at three other airports in the region could be limited due to existing airfield or other constraints, as follows:

- (i) LAX: limited to 96 MAP based on alternatives under consideration in the LAX Master Plan.
- (ii) Ontario International Airport: limited to 20 MAP based on potential existing airfield capacity.
- (iii) Burbank Airport: limited to 15 MAP based on potential runway capacity and other information provided by airport staff.

No other airports in the region were determined to be capacity limited when compared to potential levels of demand. These assumptions were incorporated into the No Project forecast and the model was rerun. The No Project forecast shows that demand at other airports would increase to absorb demand not accommodated at OCX.

The increase in passenger demand at other airports in the region over the level anticipated under the Proposed Project will increase the number of commercial aircraft operations at

these airports, causing associated increases in the noise from these aircraft operations (see Table 8.2-1 below). The increased travel distances and times required for some passengers to reach these alternative airports would also affect regional transportation and circulation, as well as air quality. These issues are discussed below.

Table 8.2-1
Residential and School Land Uses Within 65 CNEL

· Airport	Residential . Units	Number of People	Schools
San Diego (1998)	10,300	25,400	8
Burbank	1,400 (Dec. 1998)	4,140 (Dec. 1998)	4 (mid-1989 forecast for 2000)
Oxnard (forecast for 2003)	56	193	0
John Wayne Airport	134	300	0
LAX	31,335	84,054	36 <sup>I</sup>
OCX (2020)	0	0	0

Sources: JWA: Noise Abatement Quarterly Report, June 30, 1998.

LAX: Quarterly Report, Second Quarter 1998

# 8.2.4.2 General Plan Consistency

This alternative would be incompatible with Policies 13.1 through 13.7 of the County Land Use Element and Policy 5 of the County Public Services and Facilities Element regarding MCAS El Toro as regulated by Measure A. Amendments to the AELUP, Noise Element, Safety Element, and possibly the Land Use Element would be necessary to reflect that the aviation noise contours and associated land use restrictions would no longer be applicable around the El Toro site. Therefore, this alternative would have greater adverse impacts related to General Plan consistency than the Proposed Project. This alternative would not avoid or lessen impacts compared to the project.

# 8.2.4.3 Transportation and Circulation

The transportation and circulation impacts for the No Project/No Activity Alternative were analyzed based on existing roadway conditions plus committed improvements and OCP-96 development growth for 2020. The AM and PM peak hour and ADT traffic generated by JWA and the former MCAS El Toro site under 2020 No Project/No Activity conditions is summarized in Table 8.2-2. Refer to Section 14.0 in the Traffic Analysis Technical Report (Appendix D) for detailed information on the methodology applied to produce trip generation estimates for the No Project/No Activity Alternative, and for detailed summaries of the No Project/No Activity Alternative traffic volumes and associated LOS for the circulation system in the traffic analysis study area.

<sup>&</sup>lt;sup>1</sup> Total schools estimated from land use map, 1 not insulated, 35 are sound insulated.

Table 8.2-2
Trip Generation Summary - No Project/No Activity Alternative

	. AN	1 Peak F	lour	PN.	1 Peak E	lour -		Existing
Project Component	In	Out	Total	In	Out	Total	ADT	ADT
El Toro Site	0	0	0	0	0	0	0	25,400
JWA	1,552	1,037	2,589	2,047	2,050	4,097	51,706	47,450
Total	1,552	1,037	2,589	2,047	2,050	4,097	51,706	72,850

In conclusion, the No Project/No Activity Alternative would result in no new or additional local impacts related to transportation and circulation. In comparison, as discussed in detail in Section 4.3.6.6 of this Draft EIR No. 573, as supplemented, under the Proposed Project phasing years, four intersection locations, two arterial roadway segments, one continuous freeway mainline segment, and one freeway ramp would be significantly impacted under Phase 1 conditions (2005); five intersection locations, two arterial roadway segments, one continuous freeway mainline segment, and one freeway ramp would be significantly impacted under Phase 2 conditions (2010); and nine intersection locations, two arterial roadway segments, one continuous freeway mainline segment, and two freeway ramps would be significantly impacted under Phase 3 conditions (2015). At Phase 4 build out, the Proposed Project would result in significant impacts not previously identified at four freeway/tollway mainline segments and four freeway tollway ramps. See Supplemental Analysis, Section 4.3.6.5. In each case, however, the identified impacts will be mitigated to a level below significant during the applicable phasing year (see Section 4.3.7.2, Table 4.3-20).

This alternative would avoid the transportation and circulation impacts of the Proposed Project at the El Toro and JWA sites. However, regional vehicle miles traveled would be greater than the Proposed Project under this alternative. Since the impacts of the Proposed Project would be mitigated to a level of insignificance, this alternative would not avoid a significant impact; however, this alternative would reduce less than significant highway impacts near the El Toro site. In conclusion, the increase in VMT due to this alternative would result in a worse impact than the Proposed Project, and this impact would not be mitigatable except through expansion of airport facilities in the County.

#### 8.2.4.4 Noise

#### Aircraft Noise

This alternative will not result in an increase in aircraft generated CNEL or SENEL contours at the MCAS El Toro site since no airport would be developed on the MCAS El Toro site. No airport expansion would occur at JWA, and the CNEL contours would be expected to increase over 1998 conditions in proportion to the anticipated growth in activity to JWA's currently authorized service level. This alternative would, however, lead to an increase in the 65 dB CNEL contour at regional airports (see Section 8.2.4.1), which would increase the existing adverse impacts of these airports on noise sensitive land uses (Table 8.2-1).

Therefore, compared to the Proposed Project, this alternative would avoid aviation noise at the El Toro site, but would increase aviation noise impacts on a regional basis.

# **Ground Transportation**

The Proposed Project Noise Study analyzed the potential increase in noise on the road network surrounding the El Toro site and JWA for this alternative. The Federal Highway Administration standard (an increase of 1.5 dB) for a significant noise increase was used in this study. This alternative would not increase noise levels for any roadway link. In comparison, under the Proposed Project, while roadway noise impacts at two roadway links will be significant, these impacts will be reduced to a level below significance with project mitigation.

#### 8.2.4.5 Air Quality

The air quality impacts of the No Project/No Activity Alternatives were identified by analyzing the short-term impacts (construction), regional air quality impacts (total air pollutants emissions), local air quality impacts due to traffic carbon monoxide (CO), and local impacts due to aircraft and associated operations under each development scenario (i.e., Phase 1, Phase 2, Phase 3, and Phase 4).

As summarized below, the No Project/No Activity Alternative would result in greater regional air quality impacts caused by motor vehicle, aircraft, and aviation related activity emissions when compared to the Proposed Project. These impacts would be greater in all phasing years than under the Proposed Project's development scenarios. The reason for these increased air quality impacts results primarily because aviation related activity would reach or exceed the operating capability of many regional airports producing significant delay. Under the Proposed Project condition, air traffic would be efficiently accommodated at JWA and OCX. The No Project/No Activity Alternative, however, would avoid the significant and unavoidable construction impacts of the Proposed Project, the significant and unavoidable local air quality impacts due to aircraft operations at OCX and JWA, and the significant and unavoidable toxic air contaminant impacts of the Proposed Project.

# Short-Term Air Quality Impacts (Construction)

Under this alternative, no runway improvements at JWA would be necessary, and there would be no aviation reuse of MCAS El Toro. Additionally, no nonaviation land uses are planned for the El Toro and JWA sites under this alternative. Therefore, short-term construction emissions under this alternative would be less than those of the Proposed Project during any development phase and would not be significant. Similarly, peak daily local emissions, including both equipment exhaust and fugitive dust, would be less than those of the Proposed Project under all development phases and would not be significant. Therefore, compared to all phases of the Proposed Project, this alternative would avoid significant and unavoidable local construction emission impacts of the Proposed Project.

#### Operational Air Quality Impacts

#### **Emissions Inventories**

Emissions projected to occur under the No Project/No Activity Alternative in Phase 2 and Phase 4 of project development in comparison to the Proposed Project are shown in Tables 8.2-3A and 8.2-3B, respectively. As can be seen from the tables provided, the No Project/No Activity Alternative would result in operational emissions impacts that exceed the Proposed Project. Although there is sufficient existing capacity at airports in the region to absorb the projected unconstrained demand without expansion of runway capacity in Orange County, the failure to provide sufficient airport capacity in Orange County to meet the locally generated demand will result in greater average highway trip lengths and, therefore, increased vehicle miles traveled (VMT) by air passengers and shippers. addition, accommodating future demand without the project at other Basin airports would increase average delay time at those airports. This would result in increased aircraft emissions due to longer taxi times and LTO cycle times. Longer aircraft taxi times generate major increases in the amount of aircraft emissions. Therefore, for the No Project/No Activity Alternative, emissions at other regional airports would be higher per operation than at OCX. All of these factors would result in significant regional air quality emissions for the No Project/No Activity Alternative that exceed the Proposed Project in all phasing years.

#### **Dispersion Analysis**

An airport emissions dispersion analysis was conducted for JWA for the No Project/No Activity Alternative. Tables 8.2-4 and 8.2-5 show that no local criteria pollutant hot spots from airport operations were found under this project alternative in Phase 2. In Phase 4, however, there would be one exceedance of the State 1-hour NO<sub>2</sub> standard of Executive Park. Therefore, the No Project/No Activity Alternative would result in a significant local air quality impacts in Phase 4. In comparison, under the Proposed Project, there will several exceedances of the 1-hour State standard for NO<sub>2</sub> projected at OCX and JWA and continued exceedances of the State 24-hour standard for PM<sub>10</sub> projected at OCX and JWA. Therefore, the No Project/No Activity Alternative would avoid a number of significant and unavoidable local air quality impacts due to aircraft operations at OCX and JWA.

For the No Project/No Activity Alternative, at intersections in the vicinity of JWA, the CAL3OHC model was used to assess the CO concentration.

Tables 8.2-6 and 8.2-7 show that the 1-hour and 8-hour CO concentrations would be below the State and federal CO standards. Therefore, no CO hot spots would occur from vehicular traffic trips under this alternative. Similarly, under the Proposed Project, no CO hot spots would occur from vehicular traffic trips.

Table 8.2-3A
Regionwide Emissions Inventory Phase 2 Proposed Project/No Project (Pounds/Day Unless Noted)

		No Proj	ect (Phase 2)			7		Proposed Project (Phase 2)					
		CO	NOx	ROC	SO <sub>x</sub>	PM <sub>10</sub>		CO	NO <sub>x</sub>	ROC	\$O <sub>x</sub>	PM <sub>10</sub>	
Aircraft	El Toro				**1	; ,			7,877.30	790.56	548.57	96.08	
	JWA	7,237.35	3,117.14	415.07	246.83				1,267.48	279.81	107.33	22.53	
	Other Airports	64,338.22	70,647.13	9,401.21	5,385.93				62,802.79	8,350.06	4,785.16	683.42	
<u> </u>	Total Regional	71,575.57	73,764.27	9,816.28	5,632.76		1		71,947.57	9,420.43	5,439.67	802.03	
GSE	El Toro		1		1	§	GSE El Toro		1,115.99	366.61	55.33	44.04	
	JWA	5,914.70	634.43	181.56	16.21				422.61	104.15	11.35	18.62	
	Other Airports	90,189.58	9,056.56	2,668.50	586.59		•		8,059.30	2,374.68	522.02	295.52	
	Total Regional	96,104.28	9,690.99	2,850.06	602.80				9,597.90	2,845.44	588.70	358.18	
Energy	El Toro		***		]		Energy El Toro		407.70	3.80	41.80	14.00	
Į.	JWA	20.30	117.10	1.10	12.00				84.60	0.80	8.70	2.90	
ı	Others	492.00	2,832.00	26.00	290.00	: .	II.	1	2,522.00	24.00	257.10	86.00	
l	Total Regional	512.30	2,949.10	27.10	302.00	1			3,014.30	28.60	307.60	102.90	
Fuel	El Toro	**	**			'	Fuel El Toro			48.94		-	
I	JWA	***	*	10.23		1 -	JWA	- 1	-	4.76		[	
	Other Airports		**	472.61	1	l'	Other Airports			420.57	· ·		
	Total Regional		AL SE	482.84	<u> </u>		Total Regional		***	474.27	**		
Airport Roadways	El Toro		**				Airport Roadways El Toro		87.12	29.98	4.16	4.87	
	JWA	147.64	18.07	8.41	0.55				8.96	4.04	0.30	0.56	
ĺ	Other Airports	3,864.54	803.26	271.90	37.98			· · · · · · · · · · · · · · · · · · ·	871.30	311.48	38.49	47.16	
<u> </u>	Total Regional	4,012.18	821.33	280.31	38.53	52.75			967.38	345.50	42.95	52.59	
Airport Parking	El Toro		**				Airport Parking El Toro	1 1	30.36	9.89	2.98	2.77	
	JWA	120.73	9.92	16.28	3.01			1 1	4.66	7.63	1.41	0.13	
	Other Airports	2,492.05	226.97	43.40					201.98	38.63	45.95	18.14	
<u> </u>	Total Regional	2,612.78	236.89	59.68	54.65	20.66			237.00	56.15	50.34	21.04	
Roads	El Toro		1		i!	-	Roads El Toro		5,280.00	1,548.00	305.00	2,233.00	
ĺ	[				, '	1		15,350.00	4,602.00	1,421.00	296.00	1,030,00	
Ĺ	JWA	6,937.00	2,238.00	600.00	112.00	1		1 1	1,047.00	280.00	53.00	445.00	
ĺ	Other Airports <sup>2</sup>	2,965,980.00	559,703.00	111,572.00					554,910.00	110,200.00	45,328.00	6,487.00	
1		2,960,002.00	558,499.00	110.864.00		1 .		2,928,553.00	551,497.00	107,936.00	45,683.00	6,536.00	
ĺ	Total Regional <sup>2</sup>	2,972,917.00	561,941.00	112,172.00				2,967,854.00	561,237.00	112,028.00	45,682.00	9,165.00	
		2,966,939.00	560,737.00	111,464.00	45,867.00			2,947,147.00	557,146.00	109,637.00	46,032.00	8,911.00	
TOTAL (pounds/	day)	3,147,734.11	649,403.58	125,688.27	52,385.74			3,140,398.56	647,001.15	125,198.39	52,111.26	10,501.74	
		3,141,756.11	648,199.58	124,980,27	52,497.74	10,543.65		3,110,691.56	642,910.15	122,807,39	52,461.26	10,247.74	
1 Revised cale	culation of averag	e trip length. T	his revision de	es not impact a	iny of the sign	ificance	Change from No Project		(2,402.43)	(489.88)		(26.91)	
	ons made in conne						(pounds/day	() 22,064.55	(5,289,43)	(2,172.88)	(36.48)	(295,91)	
	cal correction.	-					Change from No Project (tons/year	r) (1,338.74)	(434.44)	(89.40)	(50.09)	(4.91)	
= Typograpino	cal correction.						onang.	(4,026.78)	(965.32)	(396.55)		(\$4.00)	
~ ~ ~									V		<u> </u>		
Source: CH2M HIL	L, P&D Consultan	its, and LSA Asse	ociates, Inc. 200	<b>)</b>			SCAQMD Threshold for		55	55	150	150	
							Operation (pounds/day	)					

**Table 8.2-3B** Regionwide Emissions Inventory Phase 4 Proposed Project/No Project (Pounds/Day Unless Noted)

		No Proj	ect (Phase 4)					Proposed P	roject (Phase 4			
		CO	NO <sub>x</sub>	ROC	SO <sub>X</sub>	PM <sub>10</sub>		CO	NO <sub>X</sub>	ROC	SO <sub>x</sub>	PM <sub>10</sub>
Aircraft	El Toro		]	**			Aircraft El Toro	7,358.95	13,629.82	1,029.16	859.23	130.05
	JWA	7,061.00	3,025.85	402.78	239.64	44.48	JWA	6,014.95	1,800.92	302.24	146.18	29.64
	Other Airports	76,353.47	83,463.81	11,136.94	6,362.99	908.97	Other Airports	64,573.57	70,877.49	9,423.64	5,400.40	771.29
	Total Regional	83,414.47	86,489.66	11,539.72	6,602.63	953.45	Total Regional	77,947.47	86,308.23	10,755.04	6,405.81	930.98
GSE	El Toro			prop.	4K M*	==	GSE El Toro	17,053.53	1,573.31	506.85	75.93	63.69
	JWA	5,610.84	597.89	171.83	14.93	26.54	JWA]	4,001.17	481.47	128.31	12.64	21.23
	Other Airports	106,529.38	10,697.31	3,151.94	692.84	391.71	Other Airports	90,572.75	9,095.58	2,679.98	589.11	333.50
	Total Regional	112,140.22	11,295.20	3,323.77	707.77	418.28		111,627.45	11,150.36	3,315.14	677.68	418,42
Energy	El Toro				*~		Energy El Toro	108.60	624.60	5.80	64.10	21.40
ļ	JWA	31.60	182.20	1.70	18.70	6.20		20.30	117.10	1.10	12.00	4.00
	Others	641.00	3,691.00	34.00	376.70	126.00		544.00	3,132.00	29.00	319.90	107.00
	Total Regional	672.60	3,873.20	35.70	395.40	132.20	L	672.90	3,873.70	35.90	396.00	132.40
Fuel	El Toro				**		Fuel El Toro	414		89.31		1
	JWA			9.14	***		JWA	de de		5.87		
	Other Airports		1	558.24			Other Airports			474.65		
	Total Regional	**		567.38	**		Total Regional	**		569.83		
Airport Roadways	El Toro				**		Airport Roadways El Toro	587.85	119.27	27.04	7.16	9.76
	JWA	117.92	13.70	3.99	0.56	1.17	JWA	75.34	9.02	2.59	0.38	0.76
	Other Airports	3,673.64	745.48	169.04	44.86	60.95		4,370.07	772.62	185.01	43.44	53.23 63.75
Line Ballia	Total Regional	3,791.56	759.18	173.03	45,42	62.12		5,033.26	900.91	214.64	50.98	3.85
Airport Parking	El Toro JWA	96.38	7.40	0.00	3.05	0.28	Airport Parking El Toro	367.45 61.04	31.64 4.69	5.07	9.76	
	Other Airports	2,296.37	197,72	9.98 31.70	60.99	24.07	1	1,952.50	168.12	6.32 26.95	1.93	0.18 20.47
	Total Regional	2,392.75	205.12	41.68	64.04	24.07		2,380.99	204.45	38.34	51.86 63.55	24.50
Roads	El Toro	2,372.73	203.12	41,00	04,04	24.33	Total Regional     Roads   El Torol		4.4.4	1,193.00	396.00	2,947.00
Roads	EI 10IU	**			**	**	Roads El loros	14,631.00 13,166.00	5,781.00 5,116.00	1,193.00 1,091.00	390.00	2,590,00
	JWA	4,569.00	1,848.00	359.00	112.00	946.00	JWA	2,889.00	1,168.00	227.00	71.00	598.00
	Other Airports <sup>2</sup>	2,772,382.00	495,123.00	71,538.00	48,963.00	8,883.00		2,754,719.00	489,484.00	70,413.00	48,535.00	6,269.00
	Outer reaports	2,764,036.00	493,529.00	70,589.00	49,075.00	8,892.00	Other Auports	2,722,511.00	483,968.00	66,692.00	48,996.00	6,445.00
	Total Regional <sup>2</sup>	2,776,951.00	496,971.00	71,897.00	49,075.00	9,829.00	Total Regional <sup>2</sup>	2,772,239.00	496,433.00	71,833.00	48,996.00	9,814.00
	TOTAL ITO BIONAL	2,768,605.00	495,377.00	70,948.00	49,187.00	0,838.00	1 out regionar	2,738,566.00	490,252.00	68,010.00	49,457,00	9,633.00
TOTAL (pounds	/day)	2,979,362.60	599,593.36	87,578.28	56,890.26	11,419.40		2,969,901.07	598,870.65	86,761.89	56,590,02	11,384.05
(F 3 m m m		2,971,016.60	597,999.36	86,629.28	57,002.26	11,428.40	(poursonal)	2,936,228.07	592,689.65	82,938.89	57,051.02	11,203.05
I Revised cal	culation of averag	ze trip length T	his revision de	es not impact a	ny of the sign	ificance	Change from No Project	(9,461.53)	(722.71)	(816.39)	(300.24)	(35.35)
	ons made in conn			and mapped M			(pounds/day)	(34,788.53)	(5,309,71)	(3,690.39)		(225,25)
		······································					Change from No Project (tons/year)	(1,726.53)	(131.89)	(148.99)	(54.79)	(6.45)
= <u>rypograpn</u>	ical correction.						minima man i salaas familiami	(6,348.91)	(969.02)	(673.50)	8.00	(41.13)
Course CHOLETTI	11 D&D Consider	nte and ICA to	naiotas Ina 30	Λ1			SCAOMD Thurst III S		`'	` ` ` ` ` `	ļ	
Source: Ch2M HI	LL, P&D Consultar	nts, and LSA ASS	ociates, inc. 20	VI			SCAQMD Threshold for	550	55	55	150	150

Operation (pounds/day)

Table 8.2-4
Phase 2 No Project/No Activity Pollutant Concentrations – JWA (Worst Case Operations and Meteorology)

A CONTRACT OF THE PARTY OF THE	Pollutants Concentrations (ppm)											
Receptors	COG	ppm)	NO.	(ppm)		SO, (ppm)	700	PM <sub>is</sub> (µ	g/m³)			
and the second second	1-Hour!	8-Hour	·1-Hour	AAM' -	l-Hour*	24 Hour	AAM?	24 Hour	AAM"			
110 - Monte Vista High School	5.0	3.0	0.128	0.01505	0.022	0.007	0.002	84.511	33.8			
2 - Newport Beach Golf Course	5.4	3.0	0.148	0.01511	0.023	0.007	0.002	84.6	33.8			
3 - Santa Ana Country Club	5.0	3.1	0.128	0.01515	0.023	0.007	0.002	84.7	33.8			
4 - Residential Area East of Campus Drive	5.5	3.0	0.190	0.01510	0.024	0.007	0.002	84.7	33.8			
5 - Sheraton Newport Beach	5.7	3.1	0.143	0.01537	0.023	0.007	0.002	84.9	33.8			
6 - County Superintendent of Schools	5.2	3.1	0.166	0.01554	0.023	0.007	0.002	85.1	33.9			
7 - Fire Station	6.3	3.4	0.201	0.01748	0.027	0.008	0.002	85.6	34.1			
8 - Executive Park	6.5	3.5	0.229	0.01614	0.024	0.007	0.002	85.1	33.9			
9 - Sky Park	5.5	3.1	0.160	0.01530	0.022	0.007	0.002	84.7	33.8			
Federal Standard	35 ppm	9.0 pm	N/A	0.0534 ppm	N/A	0.14 ppm	0.030 ppm	150 μg/m³	50 μg/m <sup>3</sup>			
State Standard	20 ppm	9.0 ppm	0.25 ppm	N/A	0.25 ppm	0.04 ppm	N/A	50 μg/m <sup>3</sup>	N/A			

Source:

CH2M Hill and LSA Associates, Inc., 2001.

NOTE:

- [1] Includes ambient 1-hour CO concentration of 4.6 ppm and 1-hour CO concentration reported by EDMS.
- [2] Includes ambient 8-hour CO concentration of 2.9 ppm and 8-hour CO concentration reported by EDMS.
- [3] Includes ambient 1-hour NO2 concentration of 0.089 ppm and 48.9 percent of the 1-hour NOX concentration reported by EDMS.
- 4] Includes ambient AAM NO2 concentration of 0.0150 ppm and 48.9 percent of the annual NOX concentration reported by EDMS.
- [5] Includes ambient 1-hour SO2 concentration of 0.020 ppm and 1-hour SOX concentration reported by EDMS.
- [6] Includes ambient 24-hour SO2 concentration of 0.006 ppm and 24-hour SOX concentration reported by EDMS.
- [7] Includes ambient AAM SO2 concentrations of 0.002 ppm and AAM SOX concentration reported by EDMS.
- [8] Includes ambient 24-hour PM10 concentration of 84.3 µg/m3 and 24-hour PM10 concentration reported by EDMS.
- [9] Includes ambient AAM PM10 concentration of 33.8 µg/m3 and AAM PM10 concentration reported by EDMS.
- [10] Receptor number corresponds to Figure 2-12.
- [11] Numbers in bold represent concentrations that exceed federal or State standards.

Table 8.2-5
Phase 4 No Project/No Activity Pollutant Concentrations – JWA (Worst Case Operations and Meteorology)

ran kang tanggan dan dan dan dan dan dan dan dan dan d	Pollutan (Concentrations (ppm)											
Receptors	CO (p	pm)	NO <sub>2</sub> (	ppm)		SO <sub>2</sub> (ppm)		PM <sub>10</sub> (	μ <b>g/m³)</b>			
	1-Hour <sup>1</sup>	8-Hour <sup>1</sup>	- I-Hour	TO AAM	: 1-Hour <sup>s</sup>	24 Hour	AAM'	24 Hour	AAM!			
1 <sup>10</sup> - Monte Vista High School	5.2	3.0	0.176	0.01559	0.031	0.008	0.002	87.8	31.5			
2 - Newport Beach Golf Course	5.9	3.1	0.220	0.01572	0.038	0.008	0.002	88.0	31.5			
3 – Santa Ana Country Club	5.4	3.2	0.172	0.01579	0.030	0.008	0.002	88.2	31.5			
4 - Residential Area East of Campus Drive	5.9	3.1	0.32811	0.01569	0.049	0.009	0.002	88.1	31.5			
5 - Sheraton Newport Beach	6.5	3.2	0.214	0.01620	0.035	0.008	0.003	88.5	31.6			
6 - County Superintendent of Schools	5.5	3.2	0.222	0.01649	0.033	0.008	0.003	88.6	31.6			
7 - Fire Station	7.7	3.6	0.367	0.02070	0.045	0.010	0.003	89.5	31.9			
8 - Executive Park	8.1	4.0	0.410	0.01813	0.048	0.009	0.003	88.9	31.7			
9 - Sky Park	6.2	3.3	0.254	0.01615	0.037	0.008	0.003	88.2	31.6			
Federal Standard	35 ppm	9.0 pm	N/A	0.0534 ppm	N/A	0.14 ppm	0.030 ppm	150 μg/m <sup>3</sup>	50 μg/m <sup>3</sup>			
State Standard	20 ppm	9.0 ppm	0.25 ppm	N/A	0.25 ppm	0.04 ppm	N/A	50 μg/m <sup>3</sup>	N/A			

Source:

CH2M Hill and LSA Associates, Inc., 2001.

NOTE:

- Includes ambient 1-hour CO concentration of 4.6 ppm and 1-hour CO concentration reported by EDMS.
- [2] Includes ambient 8-hour CO concentration of 2.9 ppm and 8-hour CO concentration reported by EDMS.
- [3] Includes ambient 1-hour NO<sub>2</sub> concentration of 0.092 ppm and 48.9 percent of the 1-hour NO<sub>X</sub> concentration reported by EDMS.
- [4] Includes ambient AAM NO2 concentration of 0.0155 ppm and 48.9 percent of the annual NOx concentration reported by EDMS.
- [5] Includes ambient 1-hour SO<sub>2</sub> concentration of 0.023 ppm and 1-hour SO<sub>2</sub> concentration reported by EDMS.
- [6] Includes ambient 24-hour SO<sub>2</sub> concentration of 0.007 ppm and 24-hour SO<sub>X</sub> concentration reported by EDMS.
- [7] Includes ambient AAM SO<sub>2</sub> concentrations of 0.002 ppm and AAM SO<sub>2</sub> concentration reported by EDMS.
- [8] Includes ambient 24-hour PM<sub>10</sub> concentration of 87.5 µg/m<sup>3</sup> and 24-hour PM<sub>10</sub> concentration reported by EDMS.
- [9] Includes ambient AAM PM<sub>10</sub> concentration of 35.1 µg/m<sup>3</sup> and AAM PM<sub>10</sub> concentration reported by EDMS.
- [10] Receptor number corresponds to Figure 2-12.
- [11] Numbers in bold represent concentrations that exceed federal or State standards.

**Table 8.2-6** Phase 2 No Project - Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

INTY	CINTERESCOTTO STREET	RECI	SE THE	L RVC	i ika	(c) B1764	REC	1167		gre.	RECH	DIV RECL	III recidia
	CITY OF ORANGE <sup>13</sup>				50,000,000,000,000				200000000000000000000000000000000000000				
345	Jamboree & Chapman	7.0	7.0	7,1	6.9	6.9	6.7	6.9	7.0	6.6	6.7	6.7	6.8
	CITY OF SANTA ANA 13												
154	MacArthur & Main	7.2	7.2	7.2	7.3	7.1	6.9	6.9	7.2	6.9	7.2	6.8	7.0
152	Main & Sunflower	7.1	7.2	6.8	7.2	6.7	7.0	6.6	6.5	6.7	6,9	7.2	7,0
90	Grand & Edinger	7,1	7.1	7.2	7.2	6.8	7.0	6.8	7.2	6.7	7.0	6.8	7.0
114	Red Hill & Dyer/Barranca	7.0	7.1	7.0	7.2	6.7	6,8	6.6	6.8	6.5	6.9	6.7	6.9
	CITY OF TUSTIN <sup>13</sup>												
93	Newport & Edinger	7.3	7.2	6.9	7.1	6.8	7.0	6.6	6.7	6.7	6.8	6.7	7.0
54	Jamboree & El Camino Real	7.1	7.2	7.2	7.0	6.7	7.0	6.6	6.9	6.8	6.7	7.0	7.0
105	Red Hill & Warner	7.0	6.9	7.0	6.9	6.7	6.9	6.6	6.7	6.6	6.9	6.8	6.7
115	Von Karman & Barranca	6.9	6.9	7,0	6.7	6.6	6.6	6.6	6.7	6.5	6.4	6.6	6,6
94	Red Hill & Edinger	6.8	7.0	7.2	6.9	6.8	7.0	6.7	7,0	6.7	6.8	6.8	6.7
	CITY OF IRVINE 14												
116	Jamboree & Barranca	5,9	6.0	5.7	5.7	5.4	5.6	5.4	5,5	5.6	5.6	5.5	5.8
156	Jamboree & Main	5.9	5.8	5.7	5,5	5.5	5.7	5.5	5.7	5.4	5.5	5,5	5.8
175	Jamboree & Michelson	5.7	5,6	5.7	5.6	5.2	5.6	5,3	5.3	5.5	5.4	5.5	5.5
134	Jamboree & Alton	5.7	5.6	5.8	5.5	5,3	5.5	5.2	5.2	5,3	5.3	5.3	5.6
151	Red Hill & MacArthur	5,8	5.6	5,5	5.6	5.2	5.4	5.3	5,4	5.5	5.5	5,4	5.5
98	Culver & Irvine Center	5.5	5.5	5.7	5.7	5.2	5.4	5.4	5.6	5.2	5.3	5.4	5,5
155	Von Karman & Main	5.5	5.6	5.6	5.6	5.3	5.5	5,2	5.4	5.1	5,4	5.4	5.3
177	Culver & Michelson	5.4	5.5	5,3	5.2	5.0	5.3	5.1	5.2	5,3	5.2	5.4	5.4
195	MacArthur & Jamboree	5,6	5.5	5.4	5.4	5,1	5.4	5,0	5.2	5.3	5,3	5.3	<b>5.3</b>
	CITY OF LAGUNA HILLS	14											
280	El Toro & Avd. Carlota	5.4	5.3	5.4	5.4	5.1	5.2	5.0	5.2	5.2	5.2	5.1	5.3

\* - Concentrations are in parts per million (ppm); federal I hour CO standard is 35 ppm; State I hour CO standard is 20 ppm

<sup>1 -</sup> REC1 SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> RECS W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient one-hour CO concentration, 6.1 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

Table 8.2-7

Phase 2 No Project – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

	CITY OF ORANGE 13				S 25 C K 440 C K 45	11-012-110-091-0-2-11-1-1-1			C. C. S. D. S. S. S. L. S.		ALL DESCRIPTION OF THE PARTY OF	11. To 37. 10. 11. 15. 15.	
<b>1</b> 5	Jamboree & Chapman	5.2	5.2	5.3	5.2	5.2	5.0	5.2	5.2	5.0	5.0	5.0	5.1
	CITY OF SANTA ANA 13												
54	MacArthur & Main	5.4	5.4	5.4	5.4	5.3	5.2	5.2	5.4	5.2	5.4	5.1	5.2
52	Main & Sunflower	5.3	5.4	5.1	5.4	5.0	5.2	5,0	4.9	5.0	5.2	5.4	5.2
0	Grand & Edinger	5.3	5.3	5.4	5.4	5.1	5.2	5.1	5.4	5.0	5.2	5.I	5.2
14	Red Hill & Dyer/Barranca	5.2	5.3	5.2	5.4	5.0	5,1	5.0	5.1	4.9	5.2	5.0	5.2
	CITY OF TUSTIN <sup>13</sup>												
3	Newport & Edinger	5.4	5.4	5.2	5.3	5.1	5.2	5.0	5.0	5.0	5.1	5.0	5.2
ı	Jamboree & El Camino Real	5.3	5.4	5.4	5.2	5.0	5.2	5.0	5.2	5.1	5.0	5.2	5.2
15	Red Hill & Warner	5.2	5.2	5.2	5.2	5.0	5.2	5,0	5.0	5.0	5.2	5.1	5.0
5	Von Karman & Barranca	5.2	5.2	5.2	5.0	5.0	5.0	5.0	5,0	4.9	4.8	5.0	5.0
ŀ	Red Hill & Edinger	5.1	5.2	5.4	5.2	5.1	5.2	5.0	5.2	5.0	5.1	5.1	5.0
	CITY OF IRVINE 14												
16	Jamboree & Barranca	3.8	3.9	3.7	3,7	3.5	3.6	3.5	3.5	3,6	3.6	3,5	3.7
6	Jamboree & Main	3.8	3.7	3.7	3.5	3.5	3.7	3,5	3.7	3.5	3.5	3.5	3.7
75	Jamboree & Michelson	3.7	3.6	3.7	3.6	3.3	3.6	3.4	3.4	3.5	3.5	3.5	3.5
34	Jamboree & Alton	3.7	3.6	3,7	3.5	3.4	3.5	3.3	3.3	3.4	3.4	3.4	3.6
51	Red Hill & MacArthur	3.7	3.6	3.5	3.6	3.3	3.5	3.4	3.5	3.5	3.5	3.5	3.5
8	Culver & Irvine Center	3.5	3.5	3,7	3.7	3.3	3.5	3.5	3.6	3.3	3.4	3.5	3.5
55	Von Karman & Main	3.5	3.6	3.6	3.6	3.4	3.5	3.3	3.5	3.3	3.5	3.5	3.4
77	Culver & Michelson	3.5	3.5	3.4	3.3	3.2	3.4	3.3	3.3	3.4	3.3	3.5	3.5
)5	MacArthur & Jamboree	3.6	3.5	3.5	3.5	3.3	3.5	3.2	3.3	3.4	3.4	3.4	3.4
	CITY OF LAGUNA HILLS	14											
	El Toro & Avd. Carlota	3.5	3.4	3.5	3.5	3.3		3.2	3.3	3.3	3.3	3.3	

Note: \* - Concentrations are in parts per million (ppm); federal and State 8 hour CO standard is 9 ppm

<sup>1 -</sup> REC1 SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S, DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E, DEPARTURE - MID BLOCK

<sup>8 -</sup> REC8 W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient eight-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

<sup>14 -</sup> The ambient eight-hour CO concentration, 2.9 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

#### Toxic Air Contaminants

Under this alternative, no runway improvements at JWA would be necessary and there would be no aviation reuse of MCAS El Toro. In addition, no nonaviation uses are planned. Therefore, toxic air contaminant impacts would likely be less than under the Proposed Project.

#### 8.2.4.6 Topography

The No Project/No Activity Alternative would not involve construction at the MCAS El Toro site and, therefore, would not result in impacts related to topography. Therefore, this alternative would avoid topographic impacts of the Proposed Project at the El Toro site. However, since the project impacts are insignificant, no significant impacts would be avoided.

Under the No Project/No Activity Alternative, no changes would be made to existing operations at JWA. Therefore, no changes to existing topographic conditions at JWA would occur. This is also the case under the Proposed Project.

#### 8.2.4.7 Soils, Geology and Seismicity

The No Project/No Activity Alternative would not involve construction or development at MCAS El Toro, and would therefore not result in impacts related to soils or geologic features. Since MCAS El Toro would be closed and remain vacant and unoccupied under this scenario, it would not expose residents, employees or visitors to potential seismic effects.

Under the No Project/No Activity Alternative, no changes would be made to existing operations at JWA. Therefore, no changes to existing conditions regarding soils, geologic features or seismicity would occur at JWA.

This alternative would avoid impacts of the Proposed Project at the El Toro site. However, since the project impacts are insignificant, no significant impacts would be avoided.

# 8.2.4.8 Hydrology and Water Quality

Under the No Project/No Activity Alternative, existing on-base flooding would continue, and necessary improvements would not be made. In addition, improvements such as Marshburn Channel would not be made and regional flood control plans would not be implemented. In contrast, under the Proposed Project, improvements to the existing storm drain system at MCAS El Toro will be made resulting in beneficial impacts.

No groundwater will be pumped from the MCAS El Toro site under this alternative so there will be no impacts to local groundwater levels or basin storage under this alternative.

Groundwater quality impacts under this alternative will be similar to those discussed for the Proposed Project.

With respect to water quality, under the No Project/No Activity Alternative, sedimentation impacts due to erosion at the MCAS El Toro site would be significant. In comparison, under the Proposed Project, improvements to the drainage system will reduce water quality impacts to a level below significant.

Under this alternative, JWA will require no new construction. Therefore, this alternative will not result in impacts related to hydrology and water quality.

In summary, this alternative would result in worse impacts than the project, and would not avoid or lessen project impacts.

#### 8.2.4.9 Biological Resources

Under the No Project/No Activity Alternative, the MCAS El Toro site would remain vacant and undeveloped, which would not result in direct adverse impacts to biological resources. There would be no improvements to channels or streambeds, and they would be retained at the MCAS El Toro site. However, agricultural activities would cease, and foraging would not occur. Depending upon the amount of time that this alternative continued, some areas may become more naturalized, and some wildlife may increase in numbers as a result. There would be no aircraft flyovers in the federal Habitat Reserve as part of this alternative. Under the Proposed Project, however, a Wildlife Habitat Area will be created. Consequently, no beneficial impacts associated with the creation of coastal sage scrub on the eastern portions of the MCAS El Toro site will result under the No Project/No Activity Alternative. This alternative would not result in any project impacts at the El Toro site, but since the project would have no significant impacts, this alternative would not avoid any identified significant impacts. This alternative would preclude implementation of the Wildlife Habitat Area; therefore, this alternative would have a significant adverse regional wildlife impact.

Under the No Project/No Activity Alternative for JWA, there would be no adverse biological resource impacts at JWA or in Upper Newport Bay, since there is no physical improvements and no substantial change in aircraft operations. The current indirect impacts on biological resources in Upper Newport Bay result from existing commercial operations at JWA. These impacts include noise, motion, and startle effects from direct aircraft flyovers. These impacts would continue under the No Project/No Activity Alternative. This alternative would not avoid or lessen these impacts compared to the project.

#### 8.2.4.10 Public Services and Utilities

Under the No Project/No Activity Alternative, the MCAS El Toro site would remain vacant and undeveloped, which would require no utilities. However, a large-scale, vacant site with standing buildings, such as the El Toro site, would require some form of police security, and a plan to utilize nearby fire stations for fire and emergency medical services. However, the site would generate no revenues to offset costs requiring a subsidy from federal and/or local agencies. The lack of police and fire services under the No Project/No Activity Alternative results in significant adverse impacts. In addition, the proposed OCFA station on Irvine Boulevard that is part of the Proposed Project would not be developed, and OCFA would be required to obtain another site for relocation of the Spectrum/Lake Forest temporary OCFA station. In addition, the candidate OCFA station site in Planning Area 4 would not be developed, and OCFA would be required to obtain another site to serve the Irvine area west of the El Toro site. This alternative would preclude all the public facilities proposed in the ASMP, which would be a significant adverse impact to State, County, and special district operators.

JWA would remain status quo operations, and therefore, no change to the existing public service and utilities conditions would occur.

With respect to utilities, as described in Section 4.10 (Public Services and Utilities), the Proposed Project is not anticipated to result in significant adverse impacts related to utilities. Therefore, the No Project/No Activity Alternative would not avoid a significant impact. Utilities demand at JWA under the No Project/No Activity Alternative would be similar to existing demand and could be served without significant adverse impacts after mitigation, similar to the Proposed Project.

In summary, this alternative would not avoid impacts, but would lessen impacts compared to the project. However, this alternative would generate new, significant, adverse impacts by precluding all the public facilities included in the ASMP.

# 8.2.4.11 Natural Resources and Energy

As noted in Section 4.11 (Natural Resources and Energy), the Proposed Project would not result in significant adverse impacts to natural resources and energy, with the exception of impacts to consumption of jet fuel in the region (when compared to existing conditions) and to agricultural resources at MCAS El Toro, which could not be mitigated to below a level of significance. This alternative anticipates no activity at the El Toro site, so all agricultural operations would cease. However, the Prime Agricultural Soils would not be lost to development. There are no natural or agricultural resources at JWA.

Under this alternative, energy consumption associated with construction activities at El Toro would be eliminated, and this component of the alternative's energy consumption would be less than that of the Proposed Project. From a regional standpoint, however, this alternative

would not meet the forecasted increase in air service demand, a substantial portion of which would have been met by the Proposed Project. Under this circumstance, it would be necessary for the shortfall in air service demand to be met at other regional airports which, in turn, would entail energy (jet fuel) consumption on a par with that of the Proposed Project. As noted in Section 4.11, if for any reason the regional demand for air passenger and cargo service was not fully met, the Proposed Project would have a greater impact on consumption of jet fuels than the No Project/No Activity Alternative. In addition, providing air services equivalent to those of the Proposed Project at other regional airports also could increase overall highway travel-related fuel consumption, as air travelers drive to other, more distant airports within the ASA. Consequently, the long-term regional energy consumption implications of this alternative will be equivalent to, and possibly greater than, those of the Proposed Project, so long as regional air passenger and air cargo demand is met elsewhere.

In summary, this alternative would avoid the loss of Prime Agricultural Soils and lessen impacts on energy resources compared to the Proposed Project.

#### 8.2.4.12 Aesthetics, Light and Glare

The No Project/No Activity Alternative would eliminate all activities and potential revenue for maintenance activities and lead to decay and vandalism. This would result in a significant adverse impact on aesthetics. No new or additional light or glare impacts would occur at either the El Toro site or the JWA site. Although this alternative would decrease the level of light and intensity of glare at El Toro, this was not identified as a potentially significant impact under the Proposed Project.

In summary, this alternative would have significant adverse aesthetic impacts not identified under the Proposed Project.

#### 8.2.4.13 Cultural Resources

With the No Project/No Activity Alternative, no future uses would be developed on the former Marine base site. Any cultural resources on the site would not be disturbed under the No Project/No Activity Alternative. The Proposed Project would also have no significant impacts on cultural resources; therefore, this alternative would not avoid project impacts.

The No Project/No Activity Alternative anticipates status quo operations at JWA. As such, there would be no additional or new impacts on cultural resources in the JWA area. Similarly, the Proposed Project would not impact cultural resources in the JWA area.

#### 8.2.4.14 Recreation

Assuming no future development of the MCAS El Toro site under the No Project/No Activity Alternative, there would not be any physical impacts to area recreational facilities (trails and parks). However, under the No Project/No Activity Alternative, the recreational

facilities proposed as part of the project at the El Toro site would not be provided. The demand for these recreational uses in South County would be increased. This alternative would not avoid impacts, but would significantly reduce recreational facilities, which would be a significant adverse impact of this alternative.

At JWA, status quo operations would continue under this alternative, and no additional impacts or changes to existing impacts on use of recreational facilities in the area would occur.

## 8.2.4.15 Public Health and Safety

#### Aviation Safety

Under the No Project/No Activity Alternative, the potential air carrier and air cargo accident risks at JWA would increase over the Proposed Project by approximately 40.9% to reflect the number of increasing aviation activity at JWA and the potential accident risks for general aviation at JWA would slightly increase by 2.1% correspondingly. Since there would be no aviation activity at OCX, there would be no aviation risks. Compared to the Proposed Project, this alternative would avoid impacts at the El Toro site, but would increase impacts at JWA.

This alternative would avoid the health risks of aviation toxic air contaminants at the El Toro site, but increase them at JWA compared to the project.

#### 8.2.4.16 Hazardous Materials and Hazardous Wastes

Under the No Project/No Activity Alternative, no new construction would occur at the MCAS El Toro site and JWA would continue to operate at 8.4 MAP. Remedial investigations and response actions would continue at all IRP sites at El Toro, consistent with the current program requirements of industrial cleanup standards. This is also the case under the Proposed Project.

Under the No Project/No Activity Alternative, no new hazardous materials would be used or stored and no new hazardous waste would be generated from the El Toro site. Hazardous waste handling practices would remain unchanged at JWA. Likewise, there would be no impacts associated with the new use of hazardous materials or new generation of hazardous waste materials at the El Toro site under the Proposed Project.

In comparison to the Proposed Project, over the long term, existing structures with Asbestos-Containing Building Materials (ACBMs) and lead-based paint would no longer be maintained under the No Project/No Activity Alternative. Structures containing asbestos and lead paint would deteriorate over the long term, a condition which could represent a human health hazard. This would be a significant adverse impact associated with this alternative.

This alternative would avoid new hazardous materials impacts, but would result in worse asbestos and lead paint hazards compared to the project.

#### 8.2.4.17 Socioeconomics

This alternative would result in a reduction of 24,300 jobs compared to the Proposed Project. Under this alternative, an estimated 5,200 jobs would be generated at JWA, a net increase of 3,100 jobs over existing 1998 conditions at JWA. However, this would be a significant reduction from the project case.

As with the Proposed Project, economic activity at JWA under the No Project/No Activity Alternative, as well as expenditures by visitors arriving by air through JWA, would stimulate additional off-site job growth. However, the total number of on-site and off-site jobs stimulated by the airport system would be significantly lower under the No Project/No Activity Alternative than under the Proposed Project.

Given the fewer number of jobs generated under this alternative, at 5,200 jobs versus 29,500 jobs under the Proposed Project, the magnitude of impacts related to induced growth or concentration of population and employment in the area, and increasing demand for housing, including low and moderate income housing, beneficial socioeconomic impacts would be significantly lower under the No Project/No Activity Alternative than under the Proposed Project.

In summary, this alternative would not avoid or lessen adverse impacts compared to the Proposed Project. This would be true under all development scenarios.

# 8.2.4.18 Economic Implications

To provide a point of comparison regarding the potential unrealized economic benefits to Orange County associated with the No Project/No Activity Alternative, the level of economic benefits generated under this scenario was also estimated.

Without the development of commercial aviation facilities at MCAS El Toro, the Orange County air service deficiencies are projected to increase significantly by 2020, even if JWA were to expand to its maximum passenger capability. In 2020, the air passenger capacity deficiency at Orange County airports would range from 9.2 to 14.8 million origin and destination passengers (excluding connecting passengers) depending on the extent to which JWA could be expanded. The 2020 air cargo deficiency would be approximately 2.0 million tons without development of OCX.

The potential economic implications associated with the No Project/No Activity Alternative are twofold. First, passengers served in the year 2020 in Orange County would be reduced from 34.2 MAP to 8.4 MAP, leading to substantial reductions in the output, income, and

employment associated with the direct (provision of service) activity. Second, while the air passengers and cargo projected to use OCX and JWA under the Proposed Project could be accommodated at other airports in the region, there would be some reduction in the level of visitor expenditures in Orange County from these air passengers, as well as a potential loss of economic competitiveness for the County.

Direct (provision of service) benefits to Orange County's economy generated by the No Project/No Activity Alternative in 2020 are projected to amount to \$1.3 billion in output; \$496 million in personal income; and 13,600 jobs. In terms of potential unrealized direct economic benefits, in 2020 the No Project/No Activity Alternative generates \$2.9 billion less in total output, \$1.2 billion less in personal income, and 32,000 fewer jobs than the airport related direct (provision of service) benefits associated with the Proposed Project.

Indirect benefits under the No Project/No Activity Alternative would be generated by use of aviation services provided at JWA. These use of service benefits include expenditures by visitors arriving on commercial and general aviation flights at JWA, aircrew layovers from commercial flights using JWA, and revenue to local travel agencies from Orange County residents booking flights from JWA. The total economic benefits (including indirect and induced activity) generated by use of service provided at JWA in 2020 under the No Project/No Activity Alternative amounts to 34,100 jobs, \$784 million in personal income, and \$1.9 billion in output.

The total economic benefits (including indirect and induced activity) generated by both provision and use of service provided at JWA in 2020 under the No Project/No Activity Alternative amounts to 47,700 jobs, \$1.3 million in personal income, and \$3.2 billion in output. In 2020 the No Project/No Activity Alternative generates \$6.5 billion less in total output, \$2.7 billion less in personal income, and 98,000 fewer jobs than the benefits associated with the Proposed Project.

However, these differences overstate the level of potential unrealized indirect (use of service) economic benefits associated with the No Project/No Activity Alternative. It is anticipated that air passengers projected to use OCX and JWA under the Proposed Project could be accommodated at other airports in the region under the No Project/No Activity Alternative. Thus, the visitors to Orange County expected to use OCX and JWA under the Proposed Project will still spend time and money in Orange County under the No Project/No Activity Alternative.

As regional ground access travel times increase, which regional transportation planning agencies expect will occur, reaching Orange County from airports outside of the County will become less convenient and more time-consuming. Because visitors (both business and pleasure) to the region arriving by air desire convenient, fast transportation between their origin and destination, this will place leisure and business destinations in Orange County at a competitive disadvantage in the region, potentially leading to reductions in the amount of time spent (and associated expenditures) in Orange County under the No Project/No Activity Alternative.

There is no reliable method to quantify this reduction in visitor expenditures caused by less convenient access to the County by air passengers. However, the magnitude of the impact could be significant, and would result in economic benefits generated by visitors to Orange County arriving by air that are less than the level estimated under the Proposed Project.

#### 8.2.4.19 Risk of Upset

Implementation of the No Project/No Activity Alternative will not result in significant adverse impacts to public health and safety related to risk of upset conditions. This alternative would avoid impacts compared to the Proposed Project. However, since the Proposed Project impacts are insignificant after mitigation, no significant impacts would be avoided by this alternative.

#### 8.2.5 Conclusions

The No Project/No Activity Alternative would:

- (i) Not meet any of the general project objectives, and would not meet the aviation objectives relating to passenger and cargo demand, service opportunities, industry competition, economic growth, business activities, existing land use restrictions, or General Plan implementation;
- (ii) Not avoid impacts on land uses, General Plan consistency, and regional air quality emissions;
- (iii) Result in new or additional significant adverse impacts to regional VMT, regional air quality emissions, hydrology, public services, aesthetics, recreation, aviation safety at JWA, asbestos and lead paint hazards, and economics; and
- (iv) Avoid or lessen impacts on topography; soils, geology, and seismicity; aviation noise at the El Toro site, including sleep disturbances and recreation uses; construction related air quality impacts; toxic air contaminants at El Toro; local air quality impacts at OCX due to aircraft operations; utilities; Prime Agricultural Soils; energy resources; aviation safety at El Toro; new hazardous materials and wastes; and risk of upset. However, the Proposed Project would have no significant impact after mitigation in these categories except for sleep disturbance, jet fuel consumption, local air quality, construction related air quality, toxic air contaminants, and agricultural resources.

In summary, the No Project/No Activity Alternative would avoid unmitigatable project impacts on agricultural resources, local air quality impacts at OCX, toxic air contaminants near the El Toro site, and noise impacts on sleep disturbance and recreation uses. However, this alternative would increase significant aviation noise and air quality impacts at regional airports, including toxic air contaminants and sleep disturbance due to increased service

levels at other regional airports. This alternative would result in new or additional impacts in several categories, including significant increases in regional VMT and regional air quality emissions as a result of the failure to meet the locally generated demand in Orange County. Specifically, under the No Project/No Activity Alternative, there will be greater average highway trip lengths and, therefore, increased VMT by air passengers and shippers. In addition, accommodating Orange County demand at other airports in the region would increase average delay time at those airports resulting in increased aircraft and GSE emissions.

#### 8.3 ETRPA NONAVIATION PLAN ALTERNATIVE

This section presents the potential impacts of the ETRPA Nonaviation Plan Alternative as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

# 8.3.1 Aviation Uses

No aviation reuse activities are proposed for MCAS El Toro under the ETRPA Nonaviation Plan Alternative.

# 8.3.2 Nonaviation Revenue Support Uses

The El Toro Reuse Planning Authority (ETRPA), composed of the cities of Irvine, Lake Forest, Dana Point, Laguna Beach, Laguna Hills, Laguna Niguel, and Mission Viejo, prepared a nonaviation plan (Millennium Plan, April 1998) for the El Toro site. In October of 1997, the Board of Supervisors directed that, if ETRPA timely delivered to the County a nonaviation development proposal for El Toro in form and detail adequate for analysis in the Master Development Plan EIR, the proposal would be analyzed as an alternative in the EIR. The Board also directed that environmental comparison to the Proposed Project in the EIR be provided on all environmental categories where the nonaviation alternative would result in significantly different impacts than the Proposed Project. At a minimum, the alternative will be analyzed for noise, air quality, and traffic impacts.

EIR No. 563 included analysis of a nonaviation development plan for the El Toro site (Alternative C), which was analyzed at a level of detail equal to the aviation alternatives (Alternatives A and B). The EIR No. 563 nonaviation alternative included a land use mix similar in key areas to the ETRPA Nonaviation Alternative. While there are differences in the individual land uses and land use locations, the overall objective of both alternatives is similar; that is, to establish a nonaviation planned community with a mix of residential, employment, institutional, and open space/recreation uses, including a major visitor serving commercial use component. In its certification of Final EIR No. 563, the LRA determined that the nonaviation alternative would not meet the objectives of the project and would have certain impacts greater than the CRP.

Since the nonaviation alternative was rejected during certification of Final EIR No. 563, the nonaviation plan is not required to be carried forward for analysis in Draft EIR No. 573. However, the LRA directed that the nonaviation alternative will be analyzed in case an aviation plan is determined to be infeasible at a future date.

The land uses assumed on the former MCAS El Toro site under this alternative are organized around four districts, three of which would be developed with mixes of various land uses and are referred to as an Arts and Culture district, an Education, Research and Technology (ERT) district, and a Sports and Entertainment district. The fourth district is designated as an undeveloped Habitat Reserve district, which would not generate an appreciable amount of vehicle traffic. Each of the three developed districts contains a mixed-use village as its core activity center. The mixed-use villages are envisioned as intensive activity areas composed of both residential and nonresidential uses that establish the theme for each district. Figure 8-1 depicts the ETRPA Nonaviation Plan Alternative.

Under this alternative and as in the No Project/No Activity Alternative, JWA is assumed to continue providing general aviation and short and medium-haul domestic air passenger services at a service level of 8.4 MAP (an average of 23.0 thousand passengers per day), all of which are non-connecting passengers. JWA is also assumed to continue to annually handle approximately 6.4 thousand tons of domestic belly cargo and 13.6 thousand tons of air express cargo.

# 8.3.3 Phasing: Build Out Over 20 Years

Development and build out of this alternative is proposed to occur over a 20-year period, in four 5-year phases. However, the feasibility of this absorption rate for the proposed uses is questionable. Phase One development would encompass a 1,826-acre area north and south of Irvine Boulevard at the western boundary of MCAS El Toro nearest the Eastern Transportation Corridor. Uses designated for the Phase One area include industrial, high-technology, and commercial uses, as well as a small portion of the ERT Village. Other Phase One uses include a sports stadium, auto center, office and industrial uses, and a 995-acre habitat area.

Phase Two includes an Arts and Culture Village, park space (Central Park), and a resort hotel/conference center with related golf course. This phase also includes development of residential areas related to the Village and areas adjacent to the Village.

Phase Three includes ERT uses in the southwest corner of MCAS El Toro and Entertainment/Mixed-Use areas. Phase Three developments also include an Outdoor Sports Complex and single-family residential development north of Irvine Boulevard.

Phase Four development includes additional residential areas near Central Park and south of Trabuco Drive. Also included are residential areas adjacent to the existing golf course and research and development areas. Phase Four also plans for the final development of Central Park.

# 8.3.4 Attainment of Project Objectives

This alternative would meet the general project objectives of development and surrounding land use compatibility. The ETRPA Nonaviation Plan Alternative would not meet the general project objectives of economic opportunities, timely implementation, and special planning of the aviation related objectives, with the exception of preserving general aviation opportunities (but not the objective of enhancing these opportunities).

# 8.3.5 Environmental Impacts of the ETRPA Nonaviation Plan Alternative

#### 8.3.5.1 Land Use

Under the ETRPA Nonaviation Alternative, the MCAS El Toro site is proposed to be developed with a variety of nonaviation uses including parks and open space, residential areas, employment uses, and an arena/stadium. As with the Proposed Project, the proposed perimeter land uses along the northeast and southeast portions of the site are primarily open space such as golf, habitat, and park areas. These uses are similar in intensity or less intense than the existing and General Plan approved uses off-site. The northwest portion of the MCAS El Toro site under this alternative would be developed with business, technology, education research and development and village uses. Villages include a range of residential densities, retail, office, and hotel uses. These uses are consistent with the employment uses provided for in the Orange County General Plan in the adjacent areas. Nonaviation Alternative perimeter uses for the southwest portion of the site, abutting the business park uses in the City of Irvine, include business park, transportation center (adjacent to the Irvine Transportation Center), entertainment uses, and a stadium near the confluence of I-5 and I-405. These proposed uses are compatible with the existing business park/light industry in this area. There are no significant land use conflicts associated with the proposed land uses of the ETRPA Nonaviation Alternative.

This alternative does not include any agricultural uses, therefore there is no impact of agriculture on more urbanized development. The loss of agricultural acreage is addressed in Section 8.12.4.11, Natural Resources and Energy. Concerns that an airport would attract undesirable land uses such as sexually oriented businesses is not an issue since there is no airport use proposed in the ETRPA Nonaviation Alternative.

The ETRPA Nonaviation Alternative does not address any changes to JWA, therefore, the impacts are the same as the No Project/No Activity Alternative E. There are no significant land use impacts at JWA associated with the ETRPA Nonaviation Alternative.

In summary, the impacts of this alternative related to land use are generally less than or comparable to the impacts under the Proposed Project.

#### 8.3.5.2 General Plan Consistency

The ETRPA Nonaviation Alternative would require several General Plan amendments. This alternative is not consistent with the current Public Facilities and Open Space designations of the Orange County General Plan, and would require an amendment to the Land Use Element. An amendment to the Noise Element of the County General Plan and the AELUP would be needed to eliminate aviation noise contours relating to the MCAS El Toro site. The ETRPA Nonaviation Alternative is not consistent with the City of Irvine General Plan. The ETRPA Nonaviation Alternative would require amendments to the same elements as the Proposed Project with the exception of the Safety Element of the Orange County General Plan and, therefore, would result in comparable impacts to General Plan consistency as the Proposed Project.

#### 8.3.5.3 Transportation and Circulation

The transportation and circulation impacts with full build out of the ETRPA Nonaviation Plan Alternative were analyzed for two scenarios. The first scenario analyzed the effects of the alternative on the existing roadway system without any mitigating improvements and without the impacts of committed growth and development, for the purpose of determining the significance of this alternative's impacts. In order to identify specific project related roadway improvements required with full build out of the ETRPA Nonaviation Plan Alternative, the second scenario analyzed the effects of the alternative based on existing roadway conditions plus committed improvements and foreseeable development as represented by OCP-96 development growth for 2020. Traffic generation characteristics of JWA and the former MCAS El Toro site under this alternative were determined according to two components: 1) the ERTPA Nonaviation Plan Alternative land uses at the former MCAS El Toro site, and 2) aviation uses at JWA.

The AM and PM peak hour and ADT trips generated by the nonaviation land uses at the former MCAS El Toro site and by the aviation operations at JWA with build out of this alternative are summarized in Table 8.3-1. Refer to Section 13.0 in the 1999 Traffic Analysis Technical Report for detailed information on the methodology applied to produce trip generation estimates for the ETRPA Nonaviation Plan Alternative.

Table 8.3-1
Trip Generation Summary - ETRPA Nonaviation Plan Alternative

Project Component	Tin	M Peak Ont	Hour of	h 7	PM. Reak Out	Flour ( Total	Average Daily Traffic
Former MCAS El Toro Site			<u> </u>				_
ETRPA Nonaviation Plan	23,778	7,421	31,199	11,756	25,155	36,911	339,616
Alternative Land Uses							
JWA	1,552	1,037	2,589	2,047	2,050	4,097	51,706
TOTAL	25,330	8,458	33,788	13,803	27,205	41,008	391,322

The circulation plan that is proposed to provide access to the former MCAS El Toro site as well as to facilitate the project's on-site circulation needs under this alternative is described in detail in Section 13.0 of the 1999 Traffic Analysis Technical Report. The plan is comprised of a number of arterial roads proposed to be constructed both on- and off-site, and an ETC East Leg access system that provides full access at Trabuco Road and improved access at Irvine Boulevard. No changes to the connections which currently provide access between JWA and the surrounding circulation system are envisioned with development of this alternative.

## Existing Conditions Plus Alternative Build Out Impact Analysis

The impacts of the ETRPA Nonaviation Plan Alternative on existing conditions were identified by superimposing full build out of the project onto the existing circulation setting. This scenario analyzes the impacts of the project without any mitigating improvements and without the impacts of committed growth and development, and is intended to identify the uniquely applicable potential significant effects of the alternative for the purpose of determining the significance of the alternative's impacts.

The on-site and site access plans for this alternative were applied in the existing plus project analysis with the exception of the ETC access system improvements (since the ETC had not yet been constructed under 1997 conditions) and road extensions associated with the El Toro circulation plan which assume unplanned off-site alignments. The resulting existing plus project peak hour LOS were compared with corresponding results for existing conditions (refer to Section 13.0 in the 1999 Traffic Analysis Technical Report for detailed summaries of the existing plus project traffic volumes and LOS as well as comparisons between existing and existing plus project conditions for intersections and arterial roadways within the traffic analysis study area, and refer to Section 13.0 in the 2001 Traffic Analysis Technical Report Addendum for comparable information for freeway/tollway mainline segments and freeway/tollway ramps within the traffic analysis study area). Table 8.3-2 summarizes the intersection locations, arterial roads, freeway ramps, and freeway mainline segments that are significantly impacted under existing plus ETRPA Nonaviation Plan Alternative conditions.

This scenario (i.e., build out of the 20 year project without the consideration of committed improvements to the roadway network or the impacts of other growth and development) will never actually occur and is analyzed to determine the significance of this alternative's potential traffic impacts. Potential impacts identified in the existing plus ETRPA Nonaviation Plan Alternative analysis would be mitigated through the implementation by others of committed (non-project related) roadway improvements during the actual phased development of this alternative and through the implementation of specific project mitigation measures identified based on the existing plus committed impact analysis that is summarized below for this alternative.

Table 8.3-2
Existing Plus ETRPA Nonaviation Plan Alternative Impact Summary

Location:	Jurisdiction :	Location	Jurisdiction
IMPACTED INTERSECTIONS			
Bake & Portola	County	Jeffrey & I-405 SB Ramps	Irvine
E. Central Park & Irvine	County	Jeffrey & Walnut/I-5 SB	Irvine
Millennium & Barranca	County	Sand Canyon & I-5 NB Ramps	Irvine
Millennium & Central Park	County	Sand Canyon & I-5 SB Ramps	Irvine
Millennium & Irvine	County	Technology & Barranca	Irvine
Millennium & Jeronimo	County	Irvine Center & Lake Forest	Irvine/ Laguna Hills
Millennium & Marine	County	Bake & Irvine/Trabuco	Irvine/ Lake Forest
Millennium & Rockfield	County	Bake & Jeronimo	Irvine/ Lake Forest
Research & Irvine	County	Bake & Toledo	Irvine/ Lake Forest
Sand Canyon & Trabuco	County	Jamboree & Irvine	Irvine/Tustin
Trabuco & Irvine	County	La Paz & Cabot/I-5 SB	Laguna Hills
W. Central Park & Irvine	County	Bake & Commercentre	Lake Forest
W. Central Park & Portola	County	El Toro & Rockfield	Lake Forest
Alton & Irvine	County/Irvine	Lake Forest & Jeronimo	Lake Forest
Alton & Toledo	Irvine	Lake Forest & Rockfield	Lake Forest
Bake & I-5/I-405 NB Ramps	Irvine	Lake Forest & Trabuco	Lake Forest
Bake & I-5/I-405 SB Ramps	Irvine	Los Alisos & Muirlands	Lake Forest/ Mission Viejo
Bake & Rockfield	Irvine	Alicia & Jeronimo	Mission Viejo
I-5 HOV Ramps & Barranca	Irvine	Alicia & Muirlands	Mission Viejo
I-5 NB Ramps & Alton	Irvine	La Paz & Muirlands/I-5 NB	Mission Viejo
Jeffrey & Alton	Irvine	Newport & Old Irvine	Tustin
Jeffrey & I-405 NB Ramps	Irvine		- L
IMPACTED ARTERIAL ROADS			
Irvine (Jeffrey to Research)	County	Laguna Canyon (south of El Toro)	County/ Laguna Beach
Irvine (Millennium to Trabuco)	County	Irvine (Alton to Bake)	Irvine
Portola (W. Central Park to FTC)	County	Sand Canyon (Trabuco to I-5)	Irvine
Laguna Canyon (I-405 to SR-73)	County/Irvine	Bake (north of Irvine/Trabuco)	Irvine/ Lake Forest
IMPACTED FREEWAY SEGMENT	S		•
I-5 (Sand Canyon to north of I-5)	Caltrans	SR-55 (I-5 to MacArthur)	Caltrans
I-405 (MacArthur to north of SR-55)	Caltrans	SR-55 (I-405 to SR-73)	Caltrans
IMPACTED FREEWAY RAMPS	· ·	4	
I-5 at Bake (SB Loop On-Ramp)	Caltrans/Irvine	I-5 at Sand Canyon (SB Off-Ramp)	Caltrans/ Irvine
I-5 at Bake (NB Direct On-Ramp)	Caltrans/Irvine	I-405 at Sand Canyon (NB Direct On-Ramp)	Caltrans/ Irvine
I-5 at Jeffrey (SB Off-Ramp)	Caltrans/Irvine	SR-133 at Barranca (SB On-Ramp)	Caltrans/Irvine
I-5 at Sand Canyon (NB On-Ramp)	Caltrans/Irvine	I-5 at La Paz (SB Off-Ramp)	Caltrans/ Laguna Hills

#### Alternative Build Out Impact Analysis

In order to identify project impacts that require specific project related roadway improvements, traffic conditions were analyzed based on build out of the ETRPA Nonaviation Plan Alternative. The traffic forecasts were prepared based on the circulation system that is committed to be in place within the study area by 2020 and OCP-96 development growth for 2020. Peak hour levels of service with and without the ETRPA Nonaviation Plan Alternative were compared in order to identify the locations on the existing plus committed circulation system that require specific project related improvements to address the traffic impacts of the ETRPA Nonaviation Plan Alternative.

Table 8.3-3 summarizes the intersection locations, arterial road and freeway/tollway ramps which are significantly impacted by this alternative at build out (refer to Section 13.0 in the 1999 Traffic Analysis Technical Report for detailed summaries of the traffic volumes and LOS as well as comparisons between existing plus committed conditions with and without this alternative for intersections and arterial roadways within the traffic analysis study area, and refer to Section 13.0 in the 2001 Traffic Analysis Technical Report Addendum for comparable information for freeway/tollway mainline segments and freeway/tollway ramps within the traffic analysis study area). The summary table also identifies circulation improvements which serve as mitigation measures for this alternative's impacts as well as the project's obligation (full share or fair share) to implement the proposed mitigation improvements.

Implementation of the circulation improvements identified in Table 8.3-3 would effectively mitigate to a level of insignificance all of the project impacts identified with project conditions with the exception of the intersection of Bake Parkway and the I-5/I-405 northbound ramps. As noted in the summary table, no feasible improvements that would address the impacts of this alternative were able to be identified at this location.

Table 8.3-3
ETRPA Nonaviation Plan Alternative Mitigation Improvements

Location	Jurisdiction	Improvement	Fonding Obligation	Project Share
IMPACTED INTERSECTION	The second secon	- Amprovement	Confuces	Juaic
ETC East Leg NB & Irvine	County	Convert second EB free right-turn lane to third through lane and convert NB left-turn lane to shared left-turn/second right-turn lane	Fully fund	100%
ETC NB Off & Santiago Canyon	County	Convert NB shared second left-turn lane/right-turn lane to right-turn lane and add a NB shared second left-turn lane/second right-turn lane	Fully fund	100%
Lake Forest & Portola	County	Add NB right-turn lane and convert second NB through lane to shared second through/second right-turn lane	Fully fund	100%
Millennium & Alton	County	Add fourth SB through lane	Fully fund	100%
Millennium & Irvine	County	Add third EB and WB through lanes	Fully fund	100%
Moulton & Laguna Hills	County	Convert SB right-turn lane to free right-turn lane	Fully fund	100%
Research & Irvine	County	Add third and fourth WB through lanes, third EB through lane, second NB and EB left-turn lanes and dual SB right-turn lanes	Fully fund	100%
Sand Canyon & Trabuco	County	Add third NB and SB through lanes, second EB through lane and second WB left-turn lane	Fully fund	100%
Trabuco & Irvine	County	Add third EB and WB through lanes and second NB right-turn lane	Fully fund	100%
W. Central Park & Irvine	County	Add third EB and WB through lanes	Fully fund	100%
Jeffrey & Irvine	County/Irvine	Add third EB and WB through lanes	Fair share	22%
Alton & Jeronimo	Irvine	Add second EB and WB through lanes	Fully fund	100%
Bake & I-5/I-405 NB Ramps	Irvine	No feasible improvements could be identified	Not applicable	
Bake & I-5/I-405 SB Ramps	Irvine	Convert second EB right-turn lane to third left-turn lane	Fully fund	100%
Culver & Irvine	Irvine	Add fourth WB through lane	Fully fund	100%
Jeffrey & Walnut/I-5 SB	Irvine	Convert WB through lane to shared through/second right-turn lane	Fully fund	100%
Sand Canyon & I-5 NB Ramps	Irvine	Add third SB through lane, NB right- turn lane, NB shared third through lane/second right-turn lane, second EB through lane, second WB through lane and second WB left-turn lane	Fully fund	100%

Location	Jurisdiction	Improvement	Funding Obligation	Projec Share
IMPACTED INTERSECTION	The State of the Control of the Cont		- Congress	
Sand Canyon & I-5 SB	Irvine	Add third NB through lane	Fully fund	100%
Ramps	II VIIIC	And the couple the	Tuny runa	10070
Sand Canyon & Irvine Center	Irvine	Add second SB right-turn lane	Fully fund	100%
Technology & Barranca	Irvine	Add dual SB right-turn lanes	Fully fund	100%
Lake Forest & Avd. Carlota	Irvine/	Convert second WB right-turn lane to	Fair share	17%
	Laguna Hills	shared second left-turn/second right-		
		turn lane		
Bake & Jeronimo	Irvine/	Add second NB and SB left-turn lanes	Fully fund	100%
	Lake Forest			
Bake & Toledo	Irvine/	Add WB right-turn lane	Fully fund	100%
	Lake Forest			
Jamboree & Portola	Irvine/Tustin	Convert second WB through lane to	Fully fund	100%
		second right-turn lane	-	
Jamboree & Tustin Ranch	Irvine/Tustin	Add fourth NB through lane	Fair share	3%
Alicia & Paseo Valencia	Laguna Hills	Add second EB right-turn lane	Fair share	15%
El Toro & Paseo Valencia	Laguna Hills	Add EB right-turn lane	Fully fund	100%
La Paz & Cabot/I-5 SB	Laguna Hills	Add EB right-turn lane	Fair share	19%
Laguna Hills & P. Valencia	Laguna Hills/	Add EB right-turn lane and convert	Fair share	13%
	Laguna Woods	third EB through lane to shared third		
		through/second right-turn lane		
El Toro & Jeronimo	Lake Forest	Add second SB left-turn lane	Fair share	27%
El Toro & Rockfield	Lake Forest	Add fourth NB and SB through lanes	Fair share	21%
		and EB right-turn lane		
Los Alisos & Rockfield	Lake Forest	Add second NB left-turn lane	Fully fund	100%
Alicia & Jeronimo	Mission Viejo	Add second NB left-turn lane	Fair share	20%
Red Hill & I-5 NB Ramps	Tustin	Add WB shared second left-turn/	Fully fund	100%
		second right-turn lane		
Red Hill & Irvine	Tustin	Convert NB right-turn lane to shared	Fair share	8%
		second through/right-turn lane		
IMPACTED ARTERIAL RO	DADS			
Irvine (ETC East Leg to Research)	County	Improve to six lanes	Fully fund	100%
Irvine (Jeffrey to Sand	County	Improve to six lanes	Fully fund	100%
Canyon)		-	_	
Santiago Canyon (east of	County	Improve to six lanes	Fully fund	100%
ETC)		-		
Trabuco (ETC East Leg to	County	Improve to eight lanes	Fully fund	100%
Research)	-			
Trabuco (Jeffrey to Sand	County	Improve to four lanes	Fully fund	100%
Canyon)				
Laguna Canyon (south of El	Laguna	Improve to four lanes	Fair share	14%
Toro)	Beach/County			
Irvine (Yale to Jeffrey)	Irvine	Improve to six lanes	Fully fund	100%
Jamboree (north of Tustin	Tustin/Irvine	Improve to six lanes	Fair share	3%
Ranch)				

Location ."	Jurisdiction	Improvement	Funding Obligation	Project Share
IMPACTED FREEWAY/TO	LLWAY RAMP	S		
I-5 at Alton (NB Direct On-	Caltrans/	Add a third mixed-flow lane at the	Fully fund	100%
Ramp)	Irvine	ramp meter		
I-5 at Bake (SB Loop On-	Caltrans/	Convert HOV preferential lane to a	Fully fund	100%
Ramp)	Irvine	Add second metered mixed-flow lane	-	
I-5 at Jamboree (NB Off-	Caltrans/	Add second drop lane from freeway	Fair share	13%
Ramp)	Irvine	mainline to off-ramp		
I-5 at Sand Canyon (NB On-	Caltrans/	Convert HOV preferential lane to a	Fully fund	100%
Ramp)	Irvine	Add second metered mixed-flow lane	_	
I-405 at Sand Canyon (NB	Caltrans/	Convert HOV preferential lane to a	Fully fund	100%
Direct On-Ramp)	Irvine	Add second metered mixed-flow lane	_	
I-5 at La Paz (SB Off-Ramp)	Caltrans/	Add second drop lane from freeway	Fair share	22%
	Laguna Hills	mainline to off-ramp		
I-5 at El Toro (NB Loop On-	Caltrans/	Convert HOV preferential lane to a	Fair share	12%
Ramp)	Lake Forest	Add second metered mixed-flow lane		
ETC East Leg (SR-241) at	Caltrans/	Add second drop lane from tollway	Fully fund	100%
Santiago Canyon (NB Off-	TCA/	mainline to off-ramp	_	
Ramp)	County	_		
ETC East Leg (SR-133) at	Caltrans/	Provide two lanes from the SB I-5	Fully fund	100%
Trabuco (NB Off-Ramp)	TCA/	connector ramp in addition to one lane		
	County	from the NB I-5 connector ramp		
FTC (SR-241) at Portola East	Caltrans/	Add second drop lane from tollway	Fair share	17%
(NB Off-Ramp)	TCA/	mainline to off-ramp		
	Lake Forest			
	A Same		Project Sl	are of
Lecation:	Jurisdiction	Improvement :	Future T	raffic.
IMPACTED FREEWAY/TO	LLWAY SEGMI	ENTS		
ETC (north of FTC/SR-133)	Caltrans/	Implementation of Caltrans Traffic	16%	
	TCA	Operations Strategies (TOPS)		
FTC (Alton to south of	Caltrans/	Implementation of Caltrans TOPS	11%-13%	
Portola East)	TCA			
I-5 (Alton to north of SR-55)	Caltrans	Implementation of Caltrans TOPS	7%-17%	
I-5 (El Toro to La Paz)	Caltrans	Implementation of Caltrans TOPS	10%-11%	
I-405 (Jamboree to north of	Caltrans	Implementation of Caltrans TOPS	8%-12%	
SR-55)		-		
I-405 (Culver to Sand	Caltrans	Implementation of Caltrans TOPS	15%-16%	
Canyon)				
SR-55 (Edinger to north of	Caltrans	Implementation of Caltrans TOPS	4%-6%	
Irvine/Fourth)		_		
SR-55 (I-405 to SR-73)	Caltrans	Implementation of Caltrans TOPS	8%	
		- eastbound	1	
Abbreviations: NB – northbo	Juliu ED -	- eastoonia		

In comparison, as discussed in detail in Section 4.3.6.6 of this Draft EIR No. 573, as supplemented, under the Proposed Project phasing years, four intersection locations, two arterial roadway segments, one continuous freeway mainline segment, and one freeway ramp would be significantly impacted under Phase 1 conditions (2005); five intersection locations, two arterial roadway segments, one continuous freeway mainline segment, and one freeway ramp would be significantly impacted under Phase 2 conditions (2010); and nine intersection locations, two arterial roadway segments, one continuous freeway mainline segment, and two freeway ramps would be significantly impacted under Phase 3 conditions (2015). At Phase 4 build out, the Proposed Project would result in significant impacts not previously identified to four freeway/tollway mainline segments and four freeway/tollway ramps. See Supplemental Analysis, Section 4.3.6.5. In each case, however, the identified impacts will be mitigated to a level below significant during the applicable phasing year (see Section 4.3.7.2, Table 4.3-20). Please refer to the Comparison of Alternative Impacts to Proposed Project Impacts, which follows below, for a facility-by-facility comparison of the ETRPA Nonaviation Alternative and the Proposed Project at build out.

#### Comparison of Alternative Impacts to Proposed Project Impacts

This alternative generates 339,616 daily trips from the MCAS El Toro site compared to the Proposed Project's 176,123 daily trips at MCAS El Toro. This alternative would not decrease traffic at JWA, whereas the Proposed Project would decrease JWA trips by 14,760.

## **Comparison to Existing Conditions**

A comparison of the impacts of the Existing plus ETRPA Nonaviation Plan Alternative and Existing plus Proposed Project is as follows:

#### Impacted Intersections

The following intersections are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

- Bake & Portola
- Sand Canyon & Trabuco
- Bake & I-5/I-405 SB Ramps
- Bake & Rockfield
- Jeffrey & Alton
- Jeffrey & I-405 NB Ramps
- Jeffrey & I-405 SB Ramps
- Jeffrey & Walnut/I-5 SB
- Sand Canyon & I-5 NB Ramps
- Sand Canyon & I-5 SB Ramps

- Irvine Center & Lake Forest
- Bake & Irvine/Trabuco
- Bake & Toledo
- Los Alisos & Muirlands
- Alicia & Jeronimo
- Newport & Old Irvine

The following intersections are impacted by the ETRPA Nonaviation Plan Alternative only:

- E. Central Park & Irvine
- Millennium & Barranca
- Millennium & Central Park
- Millennium & Irvine
- Millennium & Jeronimo
- Millennium & Marine
- Millennium & Rockfield
- Research & Irvine
- Trabuco & Irvine
- W. Central Park & Irvine
- W. Central Park & Portola
- Alton & Irvine
- Alton & Toledo
- Bake & I-5/I-405 NB Ramps
- I-5 HOV Ramps & Barranca
- I-5 NB Ramps & Alton
- Technology & Barranca
- Bake & Jeronimo
- Jamboree & Irvine
- La Paz & Cabot/I-5 SB
- Bake & Commercentre
- El Toro & Rockfield
- Lake Forest & Jeronimo
- Lake Forest & Rockfield
- Lake Forest & Trabuco
- Alicia & Muirlands
- La Paz & Muirlands/I-5 NB

#### Impacted Arterial Roads

The following arterials roads are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

• Laguna Canyon (I-405 to SR-73)

The following arterials are impacted by the ETRPA Nonaviation Plan Alternative only:

- Irvine (Jeffrey to Research)
- Irvine (Millennium to Trabuco)
- Portola (W. Central Park to FTC)
- Laguna Canyon (south of El Toro)
- Irvine (Alton to Bake)
- Sand Canyon (Trabuco to I-5)
- Bake (north of Irvine/Trabuco)

The Proposed Project impacts the following additional arterials:

- Laguna Canyon (south of El Toro)
- Culver (Bryan to Trabuco)

#### Impacted Freeway Ramps

The following freeway ramps are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

- I-5 at Sand Canyon NB On
- I-5 at Sand Canyon SB Off
- I-405 at Sand Canyon (NB Direct On-Ramp)

The following freeway ramps are impacted by the ETRPA Nonaviation Plan Alternative only:

- I-5 at Bake SB Loop On-ramp
- I-5 at Bake NB Direct On-ramp
- I-5 at Jeffrey (SB Off-Ramp)
- I-5 at La Paz (SB Off-Ramp)
- SR-133 at Barranca (SB On-Ramp)

The following freeway ramp is impacted by the Proposed Project only:

• I-5 at Culver - SB Off

# Impacted Freeway Mainline Segments

The following freeway mainline segments are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

• I-5 (Jeffrey to north of SR-55)

The following freeway mainline segments are impacted by the ETRPA Nonaviation Plan Alternative only:

- I-5 (Jeffrey to Sand Canyon)
- I-405 (MacArthur to north of SR-55)
- SR-55 (I-5 to MacArthur)
- SR-55 (I-405 to SR-73)

# Comparison to Existing Plus Committed Conditions

#### Impacted Intersections

The following intersections are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

- ETC East Leg NB & Irvine
- Sand Canyon & Trabuco
- Jeffrey & Irvine
- Sand Canyon & I-5 NB Ramps
- Sand Canyon & I-5 SB Ramps
- Alicia & Paseo Valencia
- La Paz & Cabot/I-5 SB
- El Toro & Rockfield
- Alicia & Jeronimo
- Red Hill & I-5 NB Ramps
- Red Hill & Irvine

The following intersections are impacted by the ETRPA Nonaviation Plan Alternative only:

- ETC NB Off & Santiago Canyon
- Lake Forest & Portola
- Millennium & Alton
- Millennium & Irvine
- Moulton & Laguna Hills
- Research & Irvine

- Trabuco & Irvine
- W. Central Park & Irvine
- Alton & Jeronimo
- Bake & I-5/I-405 NB Ramps
- Bake & I-5/I-405 SB Ramps
- Culver & Irvine
- Jeffrey & Walnut/I-5 SB
- Sand Canyon & Irvine Center
- Technology & Barranca
- Lake Forest & Avd. Carlota
- Bake & Jeronimo
- Bake & Toledo
- Jamboree & Portola
- Jamboree & Tustin Ranch
- El Toro & Paseo Valencia
- Laguna Hills & P. Valencia
- El Toro & Jeronimo
- Los Alisos & Rockfield

The following intersections are impacted by the Proposed Project only:

- Sand Canyon & Irvine
- Jeffrey & Trabuco
- Tustin Ranch & Irvine

### Impacted Arterial Roads

The following arterial roads are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

- Irvine (Jeffrey to Sand Canyon)
- Laguna Canyon (south of El Toro)

The ETRPA Nonaviation Plan Alternative alone impacts the following arterials:

- Irvine (ETC East Leg to Research)
- Santiago Canyon (east of ETC)
- Trabuco (ETC East Leg to Research)
- Trabuco (Jeffrey to Sand Canyon)
- Irvine (Yale to Jeffrey)

The Proposed Project alone impacts the following arterials:

- Irvine (ETC East Leg to PA 2 East Access Road)
- Portola (ETC West Leg to Culver)

### Impacted Freeway/Tollway Ramps

The following freeway/tollway ramps are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

- I-5 at Jamboree (NB Off-Ramp)
- I-5 at La Paz (SB Off-Ramp)
- I-405 at Sand Canyon (NB Direct On-Ramp)
- FTC (SR-241) at Portola East (NB Off-Ramp)

The ETRPA Nonaviation Plan Alternative alone impacts the following freeway/tollway ramps:

- ETC East Leg (SR-241) at Santiago Canyon (NB Off-Ramp)
- ETC East Leg (SR-133) at Trabuco (NB Off-Ramp)
- I-5 at Alton (NB Direct On-Ramp)
- I-5 at Bake (SB Loop On-Ramp)
- I-5 at El Toro (NB Loop On-Ramp)
- I-5 at Sand Canyon (NB On-Ramp)

The following freeway ramp is impacted by the Proposed Project only:

• I-5 at Red Hill (SB On-Ramp)

### Impacted Freeway/Tollway Mainline Segments

The following freeway/tollway mainline segments are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative:

- FTC (Alton to south of Portola East)
- I-5 (Alton to north of SR-55)
- I-405 (Jamboree to north of SR-55)
- I-405 (Jeffrey to Sand Canyon)

The following freeway/tollway mainline segments are impacted by the ETRPA Nonaviation Plan Alternative only:

- ETC (north of FTC/SR-133)
- I-5 (El Toro to La Paz)
- I-405 (Culver to Jeffrey)
- SR-55 (Edinger to north of Irvine/Fourth)
- SR-55 (I-405 to SR-73)

The following freeway mainline segment is impacted by the Proposed Project only:

• I-5 (I-405 to Alton)

# Level of Significance After Mitigation

Under this alternative, impacts to one location cannot be mitigated to below a level of significance. With the Proposed Project, all impacts will be reduced to below a level of significance.

### 8.3.5.4 Noise

#### Aircraft Noise

The noise impacts of this alternative would be comparable to "No Project" Compared to existing conditions at JWA.—, this alternative would create significant adverse noise impacts since JWA would have to be expanded to handle additional passengers that could not be accommodated on the MCAS El Toro site. Therefore, it is anticipated that there would be a sizable increase in the 60 and 65 CNEL noise contour around JWA in comparison to both the 1998 and 1985 airport noise contours. The total number of jet carrier SENEL events would also increase. However, a possible consequence of this alternative is pressure on the County to expand JWA to respond to growth in aviation demand, and to relax existing restrictions on the use of JWA, including nighttime restrictions. Even with the mitigation measures proposed for the Proposed Project and the ETRPA Nonaviation Alternative, the aviation impacts of this alternative would not be reduced to below a level of insignificance because of the sleep disturbance that would be created by the increased number of airplanes that would use JWA and the possible increase in nighttime operations if existing restrictions were relaxed.

#### Traffic Noise

Noise associated with vehicular traffic for this project alternative was conducted using the FHWA highway noise model. The FHWA model uses traffic volumes, vehicle mix, average vehicle speeds, road geometry, and sound propagation path characteristics to predict hourly A-weighted LEQ values adjacent to a road. Vehicle mix is reported in terms of the number of automobiles, medium trucks, and heavy trucks. The truck categories are defined in the FHWA model by number of axles and weight. To compute a CNEL value for roads, the hourly data for a 24 hour period are used according to the CNEL formula. Vehicle distribution over the 24 hour day must be known, that is, the percentage of vehicles in the daytime period between 7 a.m. and 7 p.m., in the evening period between 7 p.m. and 10 p.m., and in the night period between 10 p.m. and 7 a.m. To determine the location of noise contours, noise levels are calculated at a large number of distances and the location of constant value CNEL is determined.

Table 8.3-4 shows the existing road links included in the Airport System Master Plan (ASMP) with traffic volumes provided in the traffic report by Austin-Foust Associates. Table 8.3-5 shows road links that would have a potential noise increase greater than 1.5 dB over the existing conditions for the ETRPA Nonaviation Plan Alternative (Existing Plus ETRPA Nonaviation Plan Conditions). A total of 12 road links would have a traffic noise increase between 1.5 and 3.0 dB. A total of 15 road links would have an increase of more than 3 dB over their corresponding existing conditions levels. The noise level increase along these road links, due to the implementation of the ETRPA Nonaviation Plan Alternative, would be considered significant. Except along Portola Parkway west of Jamboree Road, where the 60 dB CNEL noise contour would remain within the roadway right-of-way, existing residences along these other road links may be exposed to traffic noise exceeding 65 dB CNEL.

Table 8.3-6 shows the noise levels along the new road links that would be constructed with the implementation of the ETRPA Nonaviation Plan Alternative under the existing condition. Because no road traffic exists for comparison with these road links, their impacts are based on whether there is a potential for existing residences adjacent to these road links to experience noise level exceeding 65 dB CNEL. Except for areas along portions of East Central Park and East Culture, most of these road links would have the 65 dB CNEL noise contour extend outside the right-of-way and potentially impact residences along the road.

Table 8.3-7 shows the noise levels along existing road links that would have 1.5 dB or more noise increases in year 2020 under the no project scenario (Alternative E) over the existing conditions. There would be 58 road links that would have 1.5 to 3.0 dB increase in traffic noise over their corresponding existing level. A total of 115 road links would have 3 dB or

per S	yalem Master Plan - ASMP 1997	<b>⊢</b>			NEL RESUL			
Kiernnc	e CNELs for existing segments)				CNEL at	76 dB	65 dB	68 dE
ID#	M	1	Spd.	lo K	56R.C.R.	CNEL	CNEL	CNE
3	New Segment Numes 17th 6/0 Prospect	22100	(mpk)	Lanca	(dBA)	(FL)	<u>(Ft)</u>	(FL)
1			30	4	56	< RdHW	52	111
2	17th eAs SR-53	32600	30	6	67.7	< RdHW	67	144
4	17th e/o Yorba	27600	30	6	67	< RdHW	60	130
30	17th w/o Newport	15900	30	4	64.6	< RdHW	42	90
	Ist w/o El Camino Real	12100	30	4	65.1	< RdHW	46	98
31	Ist w/o Newport	15300	30	4	64.4	< Rahw	40	87
28	in w/o Tustin	18500	35	4	66.9	29	62	134
29	ist w/o Yorbs	17800	30	4	65.1	< Rahw	45	97
13	4th w/o SR-55	29000	40	6	70.3	50	108	232
12	4th w/o Tustin	20288	40	6	68.7	40	86	184
459	Alicia e/o Marguerite	25800	50	6	72.1	71	153	330
463	Alicis n/o I-5	60300	45	6	74.7	103	222	477
46 i	Alicia n/o Jeronimo	40500	45	6	73	78	168	362
462	Aliela n/o Muirtanda	56300	45	6	74.4	98	212	456
460	Alicia n/o Trabuca	34400	45	6	72.3	70	151	325
454	Alicia s/o I-5	49100	45	6	73.8	90	193	416
466	Alicis s/o Moulton	36600	45	6	72.5	73	158	341
465	Alicia s/o Paseo Valencia	43300	45	6	73.3	82	176	379
381	Alise Creek e/o El Toro	18400	55	ă	71.7	68	146	315
382	Alisa Creek s/o Glenwood	18300	55	į.	71.7	68	146	315
383	Alise Crk s/o Laguna Hills	26500	55	4	73.3	87	187	403
104	Alson s/o Culver	24200	45	i	70.7	56	120	258
114	Alton c/o 1-5	42300	55	ě	75.3	118	254	548
113	Alton e/a Irvine Center	24900	55	6	73	83	179	385
102	Alton s/o Jambores	18300	50	ä	70.7	57	179	
111	Alton e/o Laguna Canyon	14700	55	- 2	70.7	19	128	262
106	Alton c/o Lake	20600	45	7	70.7	59 50	108	275
100	Alton e/o Red Hill	14700	50	- 1	69.7	30 40		232
110	Alten e/o Sand Canyon	16900	55	- 1		•••	106	229
165	Alten e/o West Yalc Loop	22900	33 45	1	71.3	65	140	301
395	Alten n/o Jeronima				70.5	34	116	251
396		22700	55	6	72.6	78	168	362
	Alten n/o Muirlands	32500	55	6	74.2	100	215	463
394	Alton s/o Irvina	12100	55	6	69.9	52	111	239
397	Alten No Muirlands	28800	55	6	73.6	92	199	429
390	Alton s/o Portols	3800	55	6	64.8	< RdHW	52	111
103	Alton w/o Culver	15900	50	6	70	52	111	239
107	Alten w/o East Yale Loop	20100	50	4	71.1	60	130	279
112	Alten w/o Irvine Center	14400	55	4	70.6	58	126	271
101	Alton w/o Jamborec	16900	50	4	70.3	54	116	251
108	Alton w/o Jeffrey	25900	50	4	72.2	71	153	330
99	Alton wie Red Hill	4306	50	4	64.4	< RdHW	46	100
109	Alton w/o Sand Canyon	15000	50	á	69.8	50	184	232
363	Avd Carlota a/o El Toro	15400	40	à	67.5	33	71	153
360	Avd Carlots e/o Lake Forcst	18400	45	à	67.1	32	68	146
361	Avd Carlota e/o Ridge Route	14700	45	ì	68.6	40	86	184
362	Avd Carlota w/o El Toro	29800	45	i	71.6	64	138	297
399	Bake n/o Commercentre	30800	55	4	73.9	97	208	449
405	Bake n/o I-5	64900	58	ì	76.1	132	283	610
400	Bake n/o Irvine/Trabuco	36200	55	6	74.6	100		
402	Rake n/o leronimo	43700	50	6	74.0	101	232	500
403	Bake n/o Muirlands	58600	50 50	6			218	470
40.5 404	Bake n/o Rockfield				75.7	124	266	574
406		52000	50	6	75.2	115	247	532
	Bake s/o I-5	4500	50	6	64.6	< Rohw	48	103
401	Bake s/e [rvine/Trabuco	41200	50	6	74.2	97	206	449
398	Bake s/o Portola	30800	55	4	73.9	97	208	449
136	Baker e/o \$8-55	15100	40	4	67.4	33	70	15

	ystem Master Plan - ASMP 1997 a CNELs for exhibing segments)			٠	NEL RESULT	70 dB	65 dB	64 d1
		1	Spd.	# of	SOR.C.R.	CNEL	CNEL	60 dE
ID#	New Segment Names	ADT	(mph)	Lanes	(dBA)	(Ft.)	(Ft)	
133	Baker w/o SR-35	27100	40	4	70	48	103	(FL) 222
87	Barranca e/o Culver	25100	45	7	70.9	57	124	266
96	Burranca e/o Irvine Center	16100	55	- 1	71.1	63	136	292
85	Burranca c/o Jamborec	24700	50	6	72	69	149	
94	Barrance e/o Laguna Carryon	3200	55	å	64.1			320
89	Barrance c/o Lake	19000	30 30	- 1		< ROHW	46	98
83	Barranca c/o Red Hill	33000		-	70.8	58	126	271
93	Barrence e/o Send Canyon	4400	50	4	73.2	84	102	391
73 98			55	4	65.5	26	57	122
70 88	Barranca c/o Technology	14800	50	4	69,7	49	106	229
347	Barranca e/o West Yale Loop	22300	45	4	70.4	53	115	247
	Barranca w/o Alton	13300	50	4	69.3	46	98	212
<b>\$</b> 6	Barranca w/o Culver	22300	50	6	71.5	65	140	301
90	Burrance w/o East Yale Loop	18300	50	4	70.7	57	122	262
95	Barranca w/o Irvine Center	10900	55	4	69.4	48	104	225
84	Barranca w/o Jamboese	28900	56	5	72.6	77	166	357
9‡	Barranca w/o Jeffrey	16700	50	4	70.3	53	115	247
97	Barranca w/o Technology	18400	55	4	71.7	68	146	315
142	Birch o/o MacArthur	9500	40	4	65.4	< RdHW	52	111
199	Birch n/o North Bristol	14800	46	4	67.3	32	69	149
200	Birch s/e North Bristol	5900	40	4	4.5	< RdHW	37	840
201	Birch s/o South Bristol	5900	40	2	63.3	17	37	80
143	Direct w/o Jamborgo	7400	40	4	64.3	< RdHW	44	94
145	Bristol e/o Red Hill	30000	45	4	71.7	64	138	297
144	Bristol w/o Red Hili	25200	45	4	70.9	57	124	266
202	Browning n/o Bryan	2900	40	i	60.3	< RAHW	< RdHW	50
203	Browning n/o El Camino Real	2400	40	i	59.4	< RdHW	< RdHW	44
204	Browning n/o Walnut	2600	35	i	58.4	< RdHW		
39	Bryan e/o Culver	7800	45	4			17	36
38	Bryan e/o Curver	11400	45		65.8	26	57	122
35				4	67.5	34	72	156
	Bryan w/o Browning	12400	40	4	66.6	28	61	132
37	Bryan w/o lamboree	12600	45	4	67.9	36	78	168
40	Bryan w/o Jeffrey	5200	50	4	65.2	25	53	115
34	Bryun w/o Rad Hill	14300	40	4	67.2	32	68	146
36	Bryan w/o Tustin Ranch	12200	45	4	67.8	35	76	163
139	Campus e/o Jambores	19100	45	4	69.7	48	103	222
137	Campus e/o MacArthur	20600	40	4	68.8	40	86	184
141	Campus e/o University	20700	45	i i	70.1	50	108	232
195	Compus s/o North Bristol	31400	45	6	71.9	66	142	306
196	Campus s/o North Bristol	27500	45	4	71.3	61	132	283
138	Campus w/o Jamborce	15000	48	1	67.4	33	70	
140	Campus w/o L/niversity	13300	45					351
238	Carlana vo Michelana	2300	45 35	2	69.5	46	100	215
23 323	Commercentre w/o Bake	500		6	57.8	« RaHW	« RallW	< K41
323 255			45	4	53.9	< RdHW	« RallW	< RdH
	Culver n/o Alton	44406	55	6	75.5	124	266	374
254	Culver n/o Barrence	43100	55	6	75.4	120	258	557
249	Culver n/o firyng	16890	45	3	69.2	44	94	202
257	Culver n/o 1-405	50900	45	4	74	91	196	422
252	Culver n/o Irvine Center	43200	55	6	75.4	120	258	557
256	Culver n/o Main	45300	55	6	75.6	124	266	574
250	Culver n/o Trabace/1-5	27300	45	3	71,3	60	130	279
253	Culver n/o Warner	44290	55	6	75.5	122	262	565
258	Culver No 1-405	49500	45	ě	74	96	193	416
259	Culver s/o Michelson	35000	45	6	72.5	72	116	335
260	Culver s/o University	37196	45	6	72.6	75		
152	Del Mar e/o Newport (NB)	9600	45 30				161	346
153		,		2	62.4	14	30	64
	Del Mar w/o Irvine	6400	30	2	60.6	< RdHW	23	49
82	Dyer e/o SR-55	43300	40	6	72	66	142	386
81	Oyer w/o Hotel Terrace	32500	40	6	70.8	54	116	251
273	East Yale Loop n/o Alton	11500	45	4	67.5	34	73	158
272	East Yale Loop n/o Barranca	10100	45	4	66.9	31	67	144

County of Orange

**Table 8.3-4** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results Existing Conditions

12/17/99

	ystem Master Plan - ASMP 1997				MEL RESULT			
ferance	(CNRLs for existing segments)				CMEL at	78 dB	65 48	60 dB
			Spd.	ø of	SOR.C.R.	CNEL	CNEL	CNEL
ID#	New Segment Names	ADT	(mph)	Lasts	(68A)	(Ft.)	(FL)	(FL)
274	East Yale Loop s/o Alton	13300	45	4	68.t	37	80	173
64 62	Edinger o/o Jamboros	20600	50	6	71.2	61	132	283
91	Edinger #6 Red Hill	17900 28906	55 40	6	71.6	67	144	311
60	Edinger w/o Red Hill Edinger w/o SR-55	32600		4	70.2	50	102	232
60 44	El Camino Roal e/e Red Hill	3 26UU	45 35	4	72	68	146	315
41	El Camino Real n/o Main	6400	35 35	4	63.7 62.3	< RdHW < RdHW	38 31	82
46	El Camino Real w/o Jamboree	73500	35 35	;	65.5	< RdHW	.31 50	66 108
42	El Camino Real w/o Hewport	2900	35	•	63.7	< RdHW	38	82
43	El Camino Real w/o Red Hill	11106	35	- 1	64.7	< RdHW	44	95
45	El Camino w/o Tustin Ranch	3500	35	- 1	61.6	< RdHW	27	93 50
444	El Toro no Laguna Canvon	17100	35	2	66.3	27	59	128
431	El Toro e/o Marguerise	11700	55	ż	69.7	51	109	236
432	El Toro e/o Santa Margarita	#500	55	6	68.3	41	22	190
439	El Turo n/o Bridger/I-5	52800	35	6	71.4	SB	126	271
430	El Toro n/o Glenn Ranch	11100	55	ž	69.1	48	184	225
436	El Toro n/o Juronimo	33800	50	6	73.3	86	184	197
437	El Toro n/o Muirlands	36300	50	6	73.6	90	193	416
438	El Toro n/o Rockfleid	42200	40	6	71.9	63	140	301
434	El Toro n/o Trabuco	27500	50	5	72.4	75	161	346
443	El Toro s/o Aliso Creek	29000	45	6	71.5	63	136	292
440	El Toro s/o Avd Carlota	34200	35	6	69.5	44	94	202
435	El Toro s/o Trabuco	3150B	50	6	73	82	176	379
442	El Tore w/o Moulton	19700	50	6	71	59	128	275
441	El Toro w/o Pases Valencia	35300	40	6	71.1	57	124	266
433	El Toro w/o Sente Murgarita	19000	55	6	71.8	70	151	325
559	FTC s/e Alton	21000	65	6	74.1	103	222	477
560	FTC s/o Lake Forest	18000	65	6	73.5	92	199	429
558	FTC s/o Porsola	23000	63	6	74.5	109	236	508
361	PTC s/o Santa Margarita	20000	65	6	73.9	100	215	463
308	Glenn Ranch n/o Portols	10600	55	4	69.3	47	101	218
309	Glenn Ranch w/o El Tore	4100	55	4	65.2	25	54	116
455	Glenwood w/o Alise Creek	11900	50	4	68.4	40	<b>\$7</b>	187
454	Gleawood w/o Moulton	8800	50	4	67.5	35	75	161
171	Grand n/o Dyer	22800	40	6	69.2	43	92	199
170	Grand s/o Edinger	26900	45	ő	71.2	60	130	279
242	Flaryard n/o Alton	11300	50	4	58.6	41	88	190
241	Hervard n/o Barrance	11100	50	4	68.5	40	87	187
239	Hurvard n/o Irvine Center	9600	50	2	67.4	37	79	171
243	Harvard n/o Mala	17080	50	4	70.3	54	116	251
245	Harvard n/o University	13900	45	2	68.3	.39	<b>\$</b> 3	179
240 244	Hurvard s/o Irvine Center	10300	50	4	68.2	39	13	179
244 246	Harvard ato Main	17600	50	4	70.5	55	118	254
	Harvard sto University	12100	45	4	67.7	35	76	163
159	Holt n/a Irvine Halt s/a Irvine	6900 6000	40	4	64	< RoHW	42	90
186	1-405 n/o 1-5	149000	30 65	4	60.3 82.6	< Rahw 179	< RdHW	47
580	1-405 n/o Jamboree	251000		11			\$17	1761
378	1-405 n/o SR-55	251000	63 63	14 13	84.9 85.1	540	1163 1199	2506
378 582	1-405 s/o Cuiver	217000	65	13	85.1	557 485	1199	2584
38£ 581	1-405 s/o Jambores	237000	65	12	84.7			2251
5#3	i-405 s/o Jeffrey	211000	63 63	12	84.2	516 477	1111	2393 2216
584	1-405 s/o Sand Carryon	200000	65	10	43.9		998	
5#5	1-405 s/c SR-133	179000	65	12	23.4	463 429	924	2149 1991
579	1-405 No SR-55	265000	65	14	85.1	429 557	924 1199	1991 2584
562	1-103 80 3K-33 1-1 a/e SR-55	293000	65 65	14	\$5.1 \$5.6	357 592	1275	2364 2748
576	I-S sto Alicia	280000	65	12	85.4	392 574	1237	2548
57t	1-5 s/o Alton	288000	65	10	85.4	***	1029	
573	1-5 s/o Bake	333000	63 63	10 20	84.1 86.1	477 649	1398	2216 3013
27.3	1-5 s/a Culver	244000	65	70 12	\$4.5	524	1126	3013

	ritem Master Plan - ASMP 1997	<b></b>			NEL RESULT	78 dB	65 dB	60 d)i
HIT DEC	(CNELs for existing segments)		Spd.	# of	CNEL at	70 dH CNEL	43 dH CNEL	69 dB
D #	M' M' Mitana	ADT	(mph)		(dBA)	(FL)	(Ft)	
	New Segment Mames			Lanes				(FL)
575	I-3 s/o El Toro	299000	65	13	85.7	601	1295	2790
572	1-5 a/o 1-405	357000	65	22	\$6.4	680	1464	3155
567	I-5 s/o Jamborne	246000	65	14	14.1	532	1145	2468
<del>\$69</del>	1-5 n/o Jeffley	238000	65	12	84.7	516	1111	2393
577	1-5 No La Pax	268000	65	10	85.2	565	1218	2624
574	1-5 s/o Lake Forest	319000	65	16	86	629	1356	2922
564	I-5 s/o Newport	250000	65	13	\$4.9	540	1163	2506
565	I-5 ato Red Hill	244000	65	14	14.8	524	1128	2430
570	I-5 a/o SR-133	124000	65	10	84.4	300	1077	2321
563	1-5 a/o SR-55	273000	65	14	85.3	365	1218	2624
566	1-5 y/o Trestin Ranch	244000	65	14	84.8	324	3128	2430
66	Irvine Center o/o Culver	22000	55	6	72.5	77	166	357
69	Irvine Center e/o Sand Cyp	12800	55	4	70.1	54	116	251
71	Irvine Center n/o Altro	12700	55	6	70,1	53	115	247
72	Irving Center No Alton	25600	55	6	73.1	46	184	397
371	Irvine Center no Bake	30400	60	ă	74.8	113	243	524
370	Irvine Center No 1-403	27900	60	7	74.5	106	229	492
70		13300	55	6	70.3	55	118	254
	Irvine Center w/o Bermock	21600	55 55	6	72.4	76	163	351
65	Irvine Center w/o Culver		55			68		
67	Irvine Center w/o Jeffrey	18300		6	71.7		146	315
68	Irvine Center w/o Sand Cyn	16900	55	4	71.3	65	140	301
23	Irvine e/o Culver	23000	50	6	71.6	66	142	306
27	Irvino e/o ETC East Leg	19700	40	4	4.5	39	#3	179
21	frvine e/o Jamboree	22900	50	6	71.6	66	142	306
16	Irvine e/o Prospect	28000	40	4	70.1	49	106	229
26	Irvine e/o Sand Canyon	19700	65	4	73.9	98	212	456
14	Irvine e/o SR-55	36600	40	6	71.3	58	126	27 1
15	Irvine e/o Yorba	29300	35	5	68.9	39	#4	182
198	Irvino n/o Del Mar	25700	35	4	68.3	36	77	166
197	Irvine s/o South Bristol	31600	45	4	71.9	67	144	311
325	Irvine wio Alton	18900	63	4	73.7	95	205	442
326	Irvine w/o Bake	24200	55	6	72.9	82	176	379
18	Irvine w/o Browning	24200	45	4	70.7	56	120	258
20	Trying w/o Jambores	25500	50	6	72.1	71	153	330
24	irvine w/o Jeffley	22500	50	5	71.5	65	140	36
17	Irvine w/o Red Hill	29700	40	4	70.4	51	109	236
25	Irvine w/o Sand Convon	17400	65	7	73.3	91	196	422
19	Irvine w/o Tustin Ranch	23800	45	6	79.7	55	118	254
		30700	50	ı	72.9	80	173	374
230	Jamborse n/o Allon		50	i		86	184	•
229	Jamborce n/o Barranca	34000			73.3	90 59		39
223	Jemborec n/o Bryan	26200	45	5	71.1		128	275
227	Jamboree n/o Edinger	41400	50		74.2	98	212	456
224	Jambores n/o El Camino Roal	33400	45	8	72.1	69	149	329
232	Jamboroe n/o 1-405	60100	50	8	75.8	126	271	583
225	Jambores n/o I-5	44700	45		73.4	\$4	182	39
222	Jambone n/o Irvine	22900	45	5	70.5	54	116	25
237	Jumbores n/o MacAstitus	31200	50	6	73	80	173	374
231	Jumboree n/o Main	34800	50		73.4	87	187	403
221	Jamborce n/o Portola	22000	50	4	71.5	64	138	29
220	Jamboree n/a Tustin Ranch	19000	55	4	71.8	70	151	32:
235	Jemboree s'o Campus	34900	50	6	73.5	87	187	40:
228	Jersboree s/o Edinger	39800	50	i	74	95	203	442
233	Jaraborne s/o 1-405	67900	50		76.3	136	292	629
226	Jamborne s/o I-5	41600	45	6	73.1	80	173	37
234	Jambores s/o Michelson	45000	50	6	74.6	103	222	47
284		30900	55	3	73.9	97	208	44
283	Jaffrey n/o Alton	32300	55	3	74.1	100	215	46
483	Juffrey n/o Barranca					88		40:
279 285	Jeffrey n/o Brysn Jeffrey n/o 1-405	21200 38000	60 55	6 5	73.3	111	190 239	516

County of Orange

Table 8.3-4 (Cont.)

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results Existing Conditions

er z	COMPANIES STREET LAND				CNEL		65 d.W	EP 93	Ketteren	Contraction of the Laboratory and the manager of	
		;	Pds.	101	SORCE	SE C	CNEL	S S			
ž.	New Segment Comes	ADT	(mpk)	14000	(dBA)	(Ft.)	Œ	(Ft.)	*0	New Segrephia Name	ADT
y 20 1	ville	7800	Ş	*	72.5	2	121	368	23		1600
COM A	rabuco	24900	8	*	*	***	212	ž	2		8078
X 0.8 42	/sanut/1-5	32200	25	**	7	2	314	54	123		24000
space and	Alkla	16200	ş	•	9	=	8	2 2	-	Main main Liter & other	746.00
asimo ek	Allere	900	=	•	9	ř	2 2		: :	A first and Tolk and the second	2000
operation	El Teen	2070	; ;		<u>.</u>	ř	× !	3		TABLE WING THE PASSES	
			<b>;</b> ;	* -	₹ ?	2	3	777	-	WILL BOY DAY SHIP	
	The state of the s	200	2 :		7.00	*		£	2	PARTITION OF AUGUS	
	LOS ASSECT	8	Ç	•	P.69	\$	<b>5</b>	717	#2.F	Margacrite are Jeronimo	27900
опіто ек	Margrateite	9100	ş	*	5,99	23	G	*	64.4	Margarite n/o La Puz	12500
onimo w/	o El Toro	14200	*	-	41.4	2	2	<u> </u>	474	Macausarian ann Ofermentan	27.00
ORIGIN WA	o Lake Found	200	*	•	2 67		: :			Manual of the Control	3
marino wa	Memority	13000	:	. ,		; ;	? (	2		The second secon	
		2000	2:		<b>e</b> ;	,	2	=		SAME TO SERVICE	
1 00 m	Salt Salt Salt Salt Salt Salt Salt Salt		÷	•	69.7	+	2	**	473	Marguerite s/o El Toro	10200
FREE EVO A	40thins	13500	÷	~	£8.8	Ç	\$	6	7.00	Margarette s/o (_a Paz	3,500
Pac aro h	4wirlands/1-5	39800	÷	7	72.9	1,	168	(%)	- S	McFadden w/o Walnut	13260
PartoC	*ebot/1-5	19300	¥.	•	Ç	ç	3		22	Michelena ath Onlyer	2001
Bre on L	femilion	90.5	::	,	: 5	3 :	5 3	2		Sharpers about the state of the state of	1500
	Townson .	20701	<b>.</b>	0	à	?	Z,	3	-	MICHERON CO JANDOCTC	28700
- Pac 200	ased Valencia	7800	Ç	•	7.69	÷	S.	208	521	Michelson are MacAnthur	19500
<b>egunta Cun</b>	vom m/o Alton	200	Ş	*	58.7	< TABLE	A British	4	129	Michelson with Cultura	i exces
Soums Car	wor may fill Trams	9030	×		3			? ;		The state of the s	
	Contract of the contract of th	2000	2 ;	• •	60.4	4	s :	<u></u>	97	PARTITION WAS INCIDENT.	Ž
	YOU DO NIH IC	27300	2	m	73.4	\$	6		126	Michelaga w/o Jamborce	23.00
aguna Can	you ale Allen	906	÷	~	£.13	*	=	*	101	Michelson w/o University	4700
Spins Car	you go El Tom	TOWN.	*	*	£		: 3		160	Mandles with Alfair	2
	-4- 10E	00000	2 2			8 3	2	2		MINISTER VIEW	3
agunt Cya	FO 1-403	0000	Z,	+	68.7	<b>**</b>	2	2	2	Mouthors n/o El Toro	37600
Agrant Hills	s e/o Moullon	3000	R	•	8.69	90	80	217	378	Moulton n/o La Paz	20,00
Actor Hills	: w/o Aliso Crk	3190	\$	4	963	WHAL A	:	4	324	Mouther sin Cl Ton	2077
1			? :	۰ و		1	ñ	2	-	SACOMERON SOLD LONG	200
Marie Free	Mary Manager	200	₹:	٥	78.2	2	~	<b>34</b>	376	Mouthon sto Chemicod	26800
Mr. Forts	6-1 o/u	53200	2	*	=======================================	<b>\$</b>	126	13	370	Mouthon s/o La Par	9065
ske Forest	n/o Jeronimo	22100	Ş	*	73.4		***	. 5	123	Mushen ato 5 ate Paper	23675
the Forest	We Muirlands	27.RAM	\$	· ¥	10.	: :	3	, ;	-	Management of the Contract of	
A. E.	the County of	9	? :	٠.	2	? :	5	67	-	MODEL IN STREET	77100
Marc Publication	ING PARKERS	301	2	•	- 10	7	8	<u> </u>	354	Mustands and Amos	2024
ake Forest	No Rock field	35800	9	40	71.2	*	7.	ž	376	Mariefunds of a Alten	13,000
ake Forest	nio Trabum	26700	9	•	Ē		:	1	5	Bar Laborator atta El Truss	
A. C.	4 6			, ,	} ;	2 :	3	ξ.	-	• ,	2
TO LO LO	No Porton	3	2	-	6.7	Ç	<u>.</u>		286	Musicands e/o Lake Forest	202
Ake Forest	s/o Rametho	3500	2	•	69.3	4	8	211	353	Mujrimda e/o Los Allans	E SHOW
ake Forces	s/o Trabuco	27600	9		5	*	3		36:	Maintends and El Trans	90.00
the Court	And Cardon	7,500	2	<b>.</b>		2 ;	ē :	9	50		20%
TO LONG SE	BIDLE OAV DA	2	q	0	7	33	79	-		Minetiends w/o Lake Forest	200
uce Forest	w/o Moutton	10400	÷	9	67.1	× Ran v	39	3	<u> </u>	Newport n/o I-5	34700
Ox Alison &	to Marguerite	9969	Ş	-	3	5	3	**	**	Newmont ada Old Invited	23000
or Allene n	for terrorismo	CONTRACT.	5		:	: :	1 3			Manual of the second	
	to Desirely de		3 !	•	7	• ;	2	764		Manager of the property	201.67
1 Company 40	A MANUEL MANUEL	000	2	•	75.6	76	163	33	107	Newport sto Irvine	2808
OE A33905 D	vo Kockfield	200	2	*	72.6	£	38	357	163	Newport sto Main	29400
os Alues a	A S. Margarita	330	3	•	64.8	=	63	771	**	Memorine a/a Walnut	13100
cs A jisas n	to Trabaco	24100	5	•	* 1.	2	144		5	Month British mats Direct	CORY
ne Aliane	'o And Carbosa	11700	: =	. 4		3 \$		-	_	The same of the sa	
A line	1 Part 1		; ;	9 1		* 1	2	3		Cardinary over the control	304
	O MANAGEMENT	Aler.	2		70.8	ñ	2	797	-	NOTE DETECT AND JAMES OFFICE	200
	A D. Margarita	4000	2	*	69.3	<b></b>	03	=======================================	734	Olympiad e/o Marguerite	2
ac Arthur C	10 SR-53	38900	\$	4	73.9	3	707	¥2,	38	P. Vakeocia e/o La Paz	7000
ne Arthur e	√o [-403	46100	\$	**	71.5	35	184		75	P Valencia clo And Curions	25.55
Inc. A. reham r	do lambone	26700		4		:		·	175	The state of the s	
	7-3-4-7-		3 :			ā	è		rot.	F. VINCONCILL NO E: 1 CHO	23877
EATHER.	NO MEN	17100	2	9	ŗ	t	<b>\$</b>	<del>-</del>	*	P. Velencia s/o Los Alisos	32800
acArdus s	Vo Michelson	22.00	÷	**	74.2	z	202	438	367	P. Valenciu w/o Alicie	27600
acArthur s	% Blirch	19300	S	4	70.0	÷	7,5	Ē	3	P. Valencia with La Paz	5061
ac Ambisio a	the Comment	37,400	: 5	1 4		:	3			Mary 1 and 1	
	The state of the s	2000	2		***	2	€.	÷	3	PRUMPING ENG BAR-33	OTHE!
EARTH I	A MICHERON	1000	\$	**	ž	6	208	\$	133	Productino w/to SR-55	2883
ac Arthur .	WO SR-55	3350	\$	*	75.3	116	ž	9		Portola e/o Sand Carron	228.00
fain e/o Cu	- Fee	14900	\$	•	* 09	9	195		7.	Postola ada Sil Tour	2
in the first		out of		۲ ۶		1	3	9			20707
	220000	20	2	•	72.7	ļ.	<b>2</b>	<u>~</u>	Ĭ	Portote sto Clenn Ranch	2528
MIN CO MIN	CARRE	35200	2	-	717	*	•	•			2
				,	:	-	70.	-	215	Portous w/o Allon	

(Reference CNELs for existing negments)	(Reference CVELA for existing neaments)							
			Spet.	jo #	200	i de		Case
10	New Segment Name	ADT	(Mole)	Lanca	(dBA)	É	9	
23	Main w/o Culver	0091	2	*	139	G	8	1
77	Makn w/o El Carsido Real	808	2		62.3	1	2	Ş
2	Main w/o Jamborge	24000	9	•	:	. 3	<u> </u>	3 2
6	Main with MacArthus	26500	\$	•	2:	3 2	2 2	2 5
£	Main wio Newcoot	10700	8	٠.,		William >	2	9
=	Main wie Red Hill	24600	9		2	3	3	5 5
476	Marguarito ato Alicia	19208	\$	~	\$	#	101	2
#2. **	Margarite are Jeronimo	24900	Ş	*	70.9	57	22	262
479	Marguerise no Le Puz	32500	¥	•	72	*	146	315
5	Marguerite n/o Ofympiad	23300	₽	*	70.6	¥	116	23
č	Margardite n/o S. Margarita	9200	ş	*	5.93	\$	3	28
Ę	Marguerine n/o Traffueo	2004	÷	*	7.69	春	*	22
₽ ;	Marguerine sto El Toro	10200	9	*	67	33	<b>89</b>	971
2	Marguerite s/o (_a Paz	3,500	<b>*</b>	•	4.0	£9	₹	3
8	McFadden who Walnut	12200	35	•	- 59	< RoHW	÷	9
2	Michelson ato Culver	2000	Š	~	£9	<del>7</del>	9	20
2	Michelson e/o Jumborte	24700	÷	•	7.7.	63	35	282
2	Michelica ato MacArthur	9300	*	*	8.69	7	호	223
2	Michelson was Culver	18600	*	•	\$	÷	101	218
27	Michelson w/o Harvard	2068	÷	~	\$	÷	0	718
176	Michelaga w/o Jamborce	25100	÷	*	70.9	5	72	3
131	Michelson w/o University	4700	33	M	609	WHEN >	Z	3
7	Moulton no Alicia	24400	22	•	72.9	5	2	ž
ž	Moulton n/o El Toro	37600	2	•	72.2	×	<u> </u>	3
378	Moulton n/o La Paz	20700	÷	· <b>v</b> a	100	. 5	5 2	3
373	Moutton s/o El Toro	35600	<b>.</b>	• •	72.4	2 2	ž	13.5
376	Mouthon sto Olerwood	26800		•		: 2	3	2 2
2	Mouthon s/o 1 a Pax	1690	; ;	•	663	:	2	2 4
11	Mostless of the Forest	91.	) <b>4</b>	•	F		2 9	
373	Months at Ridge Route	1100	7	•	: =	2 5		9 ;
354	Muintands air Aliesa	17800	2 \$	. •	:	2	5	9 5
348	Murfands alo Alten	13300	Ş		9	4		
333	Musclands eto El Toro	1096			9	=	Š	: :
330	Maintands 6/o Lake Fornsi	107.0	*	٠ ٩		7 =	2	3 5
383	Muintands elo Los Allace	OUNG!	\$	•	***	2	: 3	3 5
35	Maintands w/o El Toro	19200	-	• •		3	3	2 5
Ž	Ministerds w/o Lake Forces	9099	<b>.</b>	•	3	? 3	3	1 2
3	Newport n/o [-5	34700	20	• ••	***	3	3	3 5
9	Newport n/o Old Irvine	27900	**	•	2.3	,	2	375
593	Newport 4/o 1-5/Niroon	29100	33	•	8.8	æ	I	2
162	Newport s/o Irvino	26900	33	-	68.5	£	٤	-
<b>59</b>	Newport s/o Nain	29400	33	-	0,36	2	ž	182
2	Newport s/o Walnut	12100	ñ	•	33	× MdHW	4	2
2	North British who Birch	20197	<b>\$</b>	•	71.2	28	2	23
2 :	North Bristol w/o Campus	34600	<b>\$</b>	•	5	=	23	330
¥ ;		200	÷ :	<b>-</b>	68.9	7	<del>-</del>	\$
	Chympiad e/o Margverito	2	Ž.	*	- <del>8</del>	#	<b>\$</b>	128
	r. verscal on La Faz	2	÷:	•	63.9	7,7		2
	T. VARIABLE SO AND CAPACITY	2	<b>?</b> :	•	63.9	20	2	2
3	T. VINCENTIA NO ES SONO	20477	₽:	•	69.7	<b>:</b>	5.	<u>\$</u>
3	S Colombia and Allein	2007	Ç;	۰ د	72.	3 :	9 :	~ :
3	9 Valencia tele 1 - Ban	1369	Ç	<b>.</b>	9		3:	
2	Pauladro eta Sta. Sc	2	2 \$	<b>5</b> •	97.0	× X X	ς (	2
Ξ	Physical arts 62.44	552	? \$	• •	200	ą į	2 8	2 :
} =	Pressilla adir Reard Consum	33800	7		3 ;	÷ .	3 ;	2 ;
: :	Portols ado St Tom	2020	2 2		Š t	<u>.</u>	2	
7	Portota sto Clene Ranch	2670	2 \$		3 6	7 5	2 2	553
910	Promise wife Athen		ì	•	2			•
			:	*	9	:		2

County of Orange

Traffic Noise Model Results Existing Conditions Table 8.3-4 (Cont.)

SOURCE: LSA Associates, Inc., 9/99 **EIR No. 573** 

8-50

	ntem Master Plan - ASMP 1997				MEL RESULT			
terance	: CNELs for existing segments)	1			CNEL M	74 4B	65 48	44 4K
		I	Spd.	# at	500.C.B.	CNEE	CNILL	CNEL
1D #	New Segment Names Portols wie Glenn Rauch	ADT	(mph)	Lancy	(dBA)	(Ft)	(Ft)	(Ft.)
31.3	Portols w/a Jambores	31500	55	4	74	98	212	456
312	Portola w/a Lake Forest	600	43 15	4	54.7	< rahw	< RAHW	« Rafff
18	Poetola w/o Sand Cunyon	27300 15100	33 65	4	73.4	88	190	410
157	Prospect n/o Irving	10000	63 48	•	72.7	13	179	385
158	Prospect s/o Irvine	10900	35	4	65.6	25	53	115
322	Rancho e/o Lake Forest	3300	33 50	1	64.6 63.2	< Rahw < Rahw	44	94
172	Red Hill n/o Bryan	15300	40	•	67.5		39 70	14
180	Red Hill n/o Dyer/Barranca	30700	50	6	72.9	33 80	173	151
177	Red Hill n/o Edinger	36900	40	5	71.3	50 50	128	374
173	Red Hill s/o El Camino Real	22700	35	4	67.7	33	128 70	275 (51
174	Red Hill a/o L-1	38800	33		70.1	53 47	101	218
182	Red Hill n/o MacArthur	31500	56	é	73	82	176	379
183	Red Hill n/o Main	15200	50	è	69.8	50	101	232
179	Red Hill o/o Warner	32300	43	6	72	64	146	315
186	Red Hill s's Baker	16100	50	- 4	70.1	52	113	243
121	Red Hill sto Dyer/Barrance	31600	50	i	73	<b>82</b>	176	379
178	Red Hill sto Edinger	31500	45	6	69.2	67	144	311
175	Red Hill s/s I-5	38200	35	6	70	47	101	218
184	Red Hill We Main	20500	50	6	71.1	61	132	283
185	Red Hill sto Paularino	18400	50	- i	70.7	57	122	262
176	Red Hill of Walnut	34600	40	i	71	37	122	262
427	Ridge Route e/o Moulton	7600	45	ż	65.7	26	56	120
423	Ridge Route n/o Jeronimo	9800	45	4	66.8	31	66	142
424	Ridge Route to'o Muirlands	220G	40	4	65.3	< Ranw	48	104
423	Ridge Route n/o Rockfield	\$00G	40	4	64.7	< RdHW	46	98
422	Ridge Route ato Trabuco	\$500	45	4	66.2	28	60	130
428	Ridge Route w/o Mosition	\$80G	35	4	637	< Rollw	38	\$2
292	Ridgeline ata University	14300	45	4	68.5	39	84	182
359	Rackfield o'c El Terro	16800	40	4	67.9	35	76	163
357	Rockfield #/o Lake Forest	21600	45	4	70.2	52	311	239
358	Rockfield w/o El Toro	19600	45	4	69.8	48	194	223
356	Rockfield w/o Laka Forest	12800	40	4	68.4	37	80	173
315	S. Margarita e/o Los Alisos	22600	50	6	71.6	63	140	301
319 299	S. Margarise e/o Marguerite	25200	50	5	72	70	151	325
299 300	Send Caryon s/o Alton	16600	55	6	71.2	64	136	297
293	Send Carryon n/o 1-405	18600	55	4	71.7	69	149	320
293	Sand Carryon s/o I-5	25200	60	6	74	100	215	463
294	Sand Carryon n/o Irvine Sand Carryon n/o Trabuco	12600	60 60	*	71	63	136	292
196	Sand Carryon and Francico	17600 21800	50 50	•	12.5	78	168	362
297	Sand Cv3 n/o Irvine Center	,		4	74	64	138	297
298	Sand Cyn No Irvine Center	21600 16700	30	6	71.4	63	136	292
187	Santa Ana s/o Bristol	10400	55 45	4	71.3	64	138	297
317	Sacta Margarita e/o El Toro	1	42 50	4	67.1	32	64	146
429	Santa Maria w/o Moulton	28800 9600	30 45	6	72.6	77	166	357
587	SIHTC 7/0 Lague Carryon	53600	43 65	4	66.7	30	65	140
589	S/HTC #0 Aliso Creek	47000	63	6 6	74.2	190	410	882
588	SINTC NO El Tora	47000	65	6	77.6	176	379	617
150	South Bristol wit Birch	20200	43		77.6	176	379	E17
149	South Bristol w/o Campus	28900	45	3	76.5	49	106	229
151	South Bristol w/o Jambarac	17100	43 43	3	69.2	63 44	136 95	292
602	58-133 e/o I-405	20000	65	6	73.9	100	93 215	205
601	SR-133 e/o I-5	20000	65	6	73.9	901		463
193	SR-55 Ma Dyer	240000	65	10	73.9 84.7	100 524	215	463
592	SR-35 n/o Edinger	247806	65	12	14.1	532	1128	2430
596	SR-35 No levine/Fourth	221000	65	12	M.6	532 492	1145	2468
594	SR-33 No MecArthur	231000	65	10	\$4.7	492 516	1061	2265
391	SR-33 No MucAdina	243000	65 65	10	84.6		1111	2393
396	SR-55 e/a 1-405	138000	65	*	82.3	532 362	1143 780	246 <b>8</b> 1681

	retem Master Plan - ASMP 1997 : CNELs for existing segments)	<b> </b>			NEL ARSULT	74 4B	65 dB	64 45
er ener	Covered the streeting to Smooth?	1	Spd.	# of	SORCA	CNEL	CNEL	CME
D#_	New Segment Names	ADT	(dab)	Lanes	(dBA)	(FL)	(Ft.)	(F4)
595	SR-55 No MacArthur	236000	65	12	\$4.6	516	1111	2393
597	SR-55 No SR-73	128000	65		82	341	734	1561
59E	3R-73 m/o 8R-55	\$7000	6.5	6	80.3	266	574	1237
600	SR-73 n/o Campus/Irvine	44008	55	ě	77.3	168	362	780
599	SR-73 a/o SR-55	79000	65	6	79.9	247	532	1145
59	Sycamore w/o Red Hill	7400	30	2	61.3	< RAHW	25	54
74	Technology e/o Barranca	9800	35	4	64.1	< Rativ	40	67
334	Taicda e/o Altan	7200	45	4	65.5	25	53	115
336	Toledo e/o Lake Forest	7190	43	•	63.4	25	53	115
337	Totado w/o Si Toro	5600	45	•	64.4	< RdHW	46	98
335 533	Tolodo ye/o Lake Forest Trabuco e/o Alfeia	9900	45 45	4	66.9 68.4	31 39	66 84	142 182
327	Trabuco e/o Beke	22700	55	i	77.6	78	164	362
331	Trabuco e/o El Toro	23300	50	4	71.7	67	144	311
329	Trabuco e/o Lake Forest	28600	50	5	72.6	77	166	357
332	Trabuco e/s Las Alisus	20600	43	ä	70	50	165	232
330	Trebuco w/e El Toro	26300	50	6	72.2	72	156	335
48	Trabasco w/o Jeffrey	4100	55	ě	65.2	25	54	116
328	Trabacco w/o Lake Forest	28000	50	6	72.5	76	163	351
49	Trabuce w/o Sand Carryon	3900	35	2	60.1	< RdHW	22	47
47	Trabuce w/s Yale	13200	55	6	70,3	55	112	254
207	Tustic Reach n/o Bryan	16600	45	6	69.1	44	94	202
209	Turtin Ranck n/o J-5	29600	43	6	71.6	64	138	297
206	Trustin Ranck n/o levine	15200	45	6	68.7	41	88	190
205	Trustin Rancis ain Portola	2790	50	5	62.3	< RdH₩	< rahw	73
201	Trustin Renck s/o Bryan	20300	45	6	70	49	106	229
210	Tastin Ranch s/o 1-5	18400	45	6	69.6	46	100	21:
154	Tustin wa 4th	14700	43	4	68.6	40	86	184
289	University of Culver	19600	45	4	69.8	41	104	22:
296	University of Harvard	15800	50	6	76	52	111	231
288	University a/o Yala	19300	45 50	•	69.8 70.6	48 57	103 122	222 262
291 286	University e/o Compus	35200	35	6 5	74.5	37 104	225	48
	University s/o 1-403		33 45	4	71.8	5194 65	223 140	30
287 169	University w/o Michelson Valencia w/o Red Hill	30600 3700	35	2	59.9	< RdHW	21	46
218	Von Karman n/o Birch	13500	40	4	66.9	30	65	146
217	Von Karman n/o Campus	17100	50	- 7	70.4	54	116	25
219	Von Kurmen e/o MacArthur	10900	40	Ä	66	26	57	123
215	Von Karman s/c Main	16300	50	6	70.1	52	113	24
216	Von Karman a/o Michelum	21300	50	ě	71.4	63	136	29
214	Von Karmen s/o Berranca	16400	50	6	70.2	52	113	24
57	Walnut a/o Culvor	17700	45	i	69.4	46	98	21
55	Walenst e/o Jumborpe	9800	35	i i	64.1	< RAHW	40	#7
52	Walms a/o Red Hill	14600	35	4	65.8	25	13	11
56	Walnut w/o Culver	13600	35	i	65.5	< RdHW	51	10
34	Walnut w/o Jambores	10000	40	4	63.6	25	53	11:
32	Waterat w/o Jeffrey	10400	35	i i	64.4	< Rati W	42	91
51	Walcast w/o Red Hill	16800	35		66.5	27	58	12
53	Walnut w/o Tuntin Ranch	15400	35	4	66.1	25	35	14
80	Werner e/o Culver	6900	40	4	64	< RdHW	42	94
79	Warner Wo Culver	3100	40	•	60.6	< rankw	24	52
78	Warner w/o Harvard	1400	40	4	57.1	< Party	< Rahw	31
73	Warner w/o Red Hill	14800	40	6	67.3	< rany	69	14
269	West Yele Loop a/o Alton	7200	45	4	65.5	25	53	11
268	West Yele Loop a/o Barranca	9900	45	4	66.9	31	66	14.
270	Wort Yele Loop n/o Main	9000	45	4	66.4	29	62	13
267	West Yale Loop n/u Warner	7200	45	4	65.5	23	53	11:
271	West Yele Loop so Main	15300	45	•	68.7	41	86	19
262	Yate n/o Bryan	10300	45	4	67	32	68	14
261	Yalo n/o brvine	9400	45	4	66.8	31	66	14
265	Yale n/o levine Center	14100	45	4	64.4	.19	14	16,
263	Yallo nës Trubuco	13300	45	•	68.1	37	80	17:
276	Yale n/o University	2000	35	4	57.2	< SqHW	< RdHW	30
264	Yale n/o Walnut	13800	45	4	68.3	39	<b>8</b> )	17
266 155	Yale a/o Irvine Center Yorks n/o Irvine	11400 5700	45 35	•	67.5	)4 « Rahw	72	15
	T CHITTER SLACE THUS KNAP	1 10 70 10	- 17	4	62.5	e sand	32	64

County of Orange

Table 8.3-4 (Cont.)

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results
Existing Conditions

CCTM2.8	MCAS El Toro		Airpo	rt System	Master Pian	- ASMP Exic	ting				Nonavia	ition Plan - I	ixisting			
					CNEL at	70 dB	65 dB	60 dB				CNEL at	7 <b>0 dB</b>	65 dB	60 dB	CNEL
			Spd.	# of	50ft.C.R.	CNEL	CNEL	CNEL		Spd.	# of	50ft.C.R.	CNEL	CNEL	CNEL	Increase
ID#	Existing Segment Names	ADT	(dqm)	Lanes	(dBA)	(Ft.)	(Ft.)	(Ft.)	ADT	(dqm)	Lanes	(dBA)	(Ft.)	(Ft.)	(Ft.)	dBA
394	Alton s/o Irvine	12100	55	6	69.9	49	106	229	24000	55	6	73.8	90	193	416	3.9
406	Bake s/o I-5	4500	50	6	64.6	< RdHW	47	101	9000	50	6	68.3	39	83	179	3.7
347	Barranca w/o Alton	13300	50	4	69.3	45	97	208	20000	50	4	71.8	66	142	306	2.5
97	Barranca w/o Technology	18400	55	4	71.7	65	140	301	27000	55	4	74.3	97	208	449	2.6
23	Irvine e/o Culver	23000	50	6	71.6	64	138	297	33000	50	6	74	92	199	429	2.4
26	Irvine e/o Sand Canyon	19700	65	4	70.1	51	109	236	45000	65	4	78.6	187	403	869	8.5
325	Irvine w/o Alton	18900	65	4	73.7	88	190	410	70000	65	4	80.5	251	540	1163	6.8
326	Irvine w/o Bake	24200	55	6	72,9	78	168	362	68000	55	6	78.3	179	385	830	5.4
24	levine w/o Jeffrey	22500	50	5	71.5	63	136	292	36000	50	5	74.4	98	212	456	2.9
25	Irvine w/o Sand Canyon	17400	65	4	73.9	91	196	422	38000	65	4	77.9	168	362	780	4
279	Jeffrey n/o Bryan	21200	60	6	73.3	83	179	385	30000	60	6	75.8	122	262	565	2.5
338	Jeronimo e/o Alton	8000	45	4	65.9	27	57	124	13000	45	4	68.6	40	87	187	2.7
5	Portola w/o Jamboree	600	45	4	54.7	< RdHW	< RdHW	< RdHW	1000	45	4	57.5	< RdHW	< RdHW	34	2.8
356	Rockfield w/o Lake Forest	18800	40	4	68.4	39	84	182	28000	40	4	70.5	54	116	25 i	2.1
299	Sand Canyon n/o Alton	16600	55	6	71.2	60	130	279	29000	55	6	74.6	101	218	470	3.4
300	Sand Canyon n/o I-405	18600	55	4	71.7	65	140	301	27000	55	4	74.3	97	208	449	2.6
295	Sand Canyon n/o I-5	25200	60	6	74	92	199	429	62000	60	6	79	199	429	924	5
294	Sand Canyon n/o Trabuco	17600	60	4	72.5	73	158	341	32000	60	4	76.1	128	275	592	3.6
297	Sand Cyp p/o Irvine Center	21600	50	6	71.4	62	134	288	40000	50	6	74.8	104	225	485	3.4
298	Sand Cyn s/o Irvine Center	16700	55	4	71.3	61	132	283	29000	55	4	74.6	101	218	470	3.3
74	Technology n/o Barranca	9800	35	4	64.1	< RdHW	44	94	17000	35	4	66.6	30	64	138	2.5
334	Toledo e/o Alton	7200	45	4	65.5	25	54	116	11000	45	4	67.9	36	78	168	2.4
335	Toledo w/o Lake Forest	9900	45	4	66.9	31	67	144	14000	45	4	68.9	42	91	196	2
327	Trabuco e/o Bake	22700	55	6	72.6	75	161	346	43000	35	6	76.3	132	283	610	3.7
48	Trabuco w/o leffrey	4100	55	4	65.2	< RdHW	52	111	7000	55	4	68.4	39	84	182	3.2
328	Trabuço w/o Lake Forest	28000	50	6	72,5	73	158	341	47000	50	6	75.5	116	251	540	3
49	Trabuco w/o Sand Canyon	3900	35	2	60.1	< RdHW	24	51	15000	35	2	66.1	27	59	128	6

County of Orange

**Table 8.3-5** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results
Existing plus ETRPA Nonaviation Plan Existing Segments with 1.5 dB or Higher
Traffic Noise Increase

12/17/99

CCTM2.8	EXISTING MCAS EI Toro		·····	C	NEL RESUL	TS		
ETRPA N	onaviation Plan -				CNEL at	70 dB	65 dB	60 dB
New Segm	ients		Spd.	# of	50ft.C.R.	CNEL.	CNEL	CNEL
ID#	New Segment Names	ADT	(mph)	Lanes	(dBA)	(FL)	(Ft.)	(Ft.)
500	Arts Village e/o E. Culture	11000	35	4	64.1	< RdHW	44	94
499	Arts Village e/o Millennium	13000	35	4	64.9	< RdHW	49	106
510	Astor e/o E. Central Park	5000	35	4	60.7	< RdHW	26	56
200	Birch s/o North Bristol	0	40	2	0	0	0	0
495	Bryan e/o Millennium	4000	50	4	64.2	< RdHW	44	95
494	Bryan e/o Research	4000	50	4	64.2	< RdHW	44	95
493	Bryan e/o Sand Canyon	6000	50	4	66	27	58	126
543	E. Central Park n/o Irvine	4000	35	4	59.8	< RdHW	< RdHW	48
545	E. Central Park n/o Trabuco	2000	35	4	56.7	< RdHW	< RdHW	30
547	E. Central Park s/o Astor	3000	35	4	58.5	< RdHW	< RdHW	40
544	E. Central Park s/o Irvine	11000	35	4	64.1	< RdHW	44	94
546	E. Central Park s/o Trabuco	3000	35	4	58.5	< RdHW	< RdHW	40
497	E. Culture e/o Millennium	4000	35	4	59.8	< RdHW	< RdHW	48
498	E. Culture n/o Trabuco	7000	35	4	62.2	< RdHW	33	70
491	Irvine e/o E. Central Park	42000	35	4	70	50	108	232
490	Irvine e/o Millennium	37000	35	4	69.4	46	98	212
488	Irvine e/o Research	39000	35	4	69.6	47	101	218
516	Jeronimo w/o Alton	13000	35	4	64.9	< RdHW	49	106
515	Jeronimo w/o Millennium	8000	35	4	62.8	< RdHW	36	77
513	Marine e/o Research	24000	35	4	67.5	34	73	158
512	Marine e/o Sand Canyon	26000	35	4	67.9	36	78	168
514	Marine w/o Millengium	27000	35	4	68	37	79	171
540	Millennium n/o Alton	34000	35	4	69	43	92	199
534	Millennium n/o Arts Village	7000	35	4	62.2	< RdHW	33	70
542	Millennium n/o Bake	34000	35	4	69	43	92	199
539	Millennium n/o Barranca	46000	35	4	70.4	53	115	247
532	Millennium n/o Bryan	10000	35	4	63.7	< RdHW	41	88
538	Millennium p/o Jeronimo	47000	35	4	70.5	54	116	251
537	Millennium n/o Marine	34000	35	4	69	43	92	199
541	Millennium n/o Rockfield	42000	35	4	70	50	108	232
535	Millennium n/o Trabuco	21000	35	4	67	32	68	146
533	Millennium s/o Bryan	9000	35	À	63.3	< RdHW	39	83
536	Millennium s/o Trabuco	29000	35	4	68.4	39	84	182
485	Portola e/o W. Central Park	29000	35	4	68.4	39	84	182
511	Quantum e/o Research	7000	35	à	62.2	< RdHW	33	70
520	Research n/o Bryan	10000	35	à	63.7	< RdHW	41	88
519	Research n/o Irvine	14000	35	4	65.2	< RdHW	52	111
523	Research n/o Murinc	17000	35	4	66	27	58	126
521	Research n/o Trabuco	22000	35	4	67.2	33	70	151
524	Research s/o Marine	11000	35	4	64.1	< RdHW	44	94
522	Research s/o Trabuco	11000	35	4	64.1	< RdHW	44	94
355	Rockfield w/o Bake	18000	40	4	68	37	79	171
333 518	Sand Canyon s/o Irvine	31000	60	4	75.4	115	247	532
508	Trabuco e/o E. Central Park	27000	35	4	68	37	79	171
506	Trabuco e/o Millennium	27000	35	2	68	37	79	171
503	Trabuco e/o Rescarch	33000	35 35	4	68.9	42	91	196
503 501	Trabuco e/o Sand Canyon	47000	35 35	4	70.5	54	116	251
		•	35		68.5	40		
504	Trabuco e/o W. Central Park Trabuco w/o E. Central Park	30000 29000	35 35	2 4	68.4	40 39	86 84	184 182
507		27000	35 35	2	68	39 37	84 79	171
505	Trabuco w/o Millennium	1		4	1	< RdHW	79 44	94
527	W. Central Park n/o Bryan	11000	35	_	64.1			
526	W. Central Park n/o Irvine	10000	35	4	63.7	< RdHW	41	88
528	W. Central Park n/o Trabuco	11000	35	4	64.1	< RdHW	44	94
525	W. Central Park s/o Portola	12000	35	4	64.5	< RdHW	46	100
530	W. Central Park s/o Quantum	11000	35	4	64.1	< RdHW	44	94
529	W. Central Park s/o Trabuco	8000	35	4	62.8	< RdHW	36	77
496	W. Culture n/o Trabuco	9000	35	4	63.3	< RdHW	39	83
489	Irvine e/o W. Central Park	40000	35	4	69.8	48	104	225

County of Orange

**Table 8.3-6** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results
Existing plus ETRPA Nonaviation
Plan - New Segments

C 1 M2.5	MCAS El Toro		Aiı	rport Syster	n Master Pla					No Pr	oject Altern	utive 2020 - 1				
					CNEL at	70 dB	65 dB	60 4B				CNEL at	70 dB	65 dB	60 dB	CNEL
***			Spd.	# of	50(LC.R.	CNEL	CNEL	CNEL		Spd.	# of	50n.C.R.	CNEL	CNEL	CNEL	Increase
ID#	Existing Segment Names	ADT	(mph)	Lanes	(ABb)	<u> (FL)</u>	(FL)	(FL)	ADT	(mph)	Lanes	(dBA)	(FL)	(Ft.)	<u>(Ft.)</u>	dBA
28	1st w/o Tustin	18500	35	•	66.9	31	67	144	27000	35	6	68,6	40	87	187	1.7
12	4th w/o Tustin	20200	40	6	68.7	41	88	190	29000	40	6	70.6	55	118	254	1.9
111	Alton e/o Laguna Canyon	14700	55	4	70.7	56	120	258	22000	55	*	73.4	84	182	391	2.7
100	Alton e/o Red Hill	14700	50	4	69.7	48	103	222	25000	50	6	72.8	77	166	357	3.1
110	Alton e/o Sand Canyon	16900	55	4	71.3	61	132	283	27000	55	4	74.3	97	208	449	3
395	Alton n/o Jeronimo	22700	55	6	72.6	75	161	346	35000	55	6	75.4	115	247	532	2.8
396	Alton n/o Muirlands	32500	55	6	74.2	95	205	442	52000	55	6	77.1	149	320	690	2.9
394	Alton s/o Irvine	12100	55	6	69.9	49	106	229	30000	55	6	74.8	104	225	485	4.9
390	Aiton L'o Portola	3800	55	6	64.8	< RdHW	48	104	29000	55	6	74.6	101	218	470	9.8
101	Alton w/o Jambures	16900	50	4	70.3	52	113	243	29000	50	6	73.4	84	182	391	3.1
406	Bake s/o 1-5	4500	50	6	64.6	< RdHW	47	101	18000	50	6	71.3	61	132	283	6.7
136	Baker e/o SR-55	15100	40	4	67.4	34	72	156	30000	40	4	70.8	57	122	262	3.4
96	Barranca e/o Irvine Center	16100	55	4	71.1	59	128	275	29000	55	4	74.6	101	218	470	3.5
94	Barrance e/o Laguna Canyon	3200	55	4	64.1	< RdHW	44	94	15000	55	4	71.7	65	140	301	7.6
93	Barranca e/o Sand Canyon	4400	55	4	65.5	25	54	116	16000	55	4	72	68	146	315	6,5
98	Barranca e/o Technology	14800	50	4	69.7	48	103	222	27000	50	4	73.1	80	173	374	3.4
347	Barrunca w/o Alton	13300	50	4	69.3	45	97	208	26000	50	4	72.9	78	168	362	3.6
95	Barranca w/o Irvine Center	10900	55	4	69.4	46	98	212	19000	55	4	72.8	77	166	357	3.4
84	Barranca w/o Jamboree	28900	50	6	72.6	75	161	346	49000	50	6	75.7	120	258	557	3.1
97	Barranca w/o Technology	18400	55	4	71.7	65	140	301	33000	55	4	75.2	111	239	516	3.5
201	Birch s/o South Bristol	5900	40	2	63.3	18	39	83	9000	40	2	65.5	25	54	116	2.2
39	Bryan e/o Culver	7800	45	4	65.8	26	57	122	12000	45	4	68.3	39	83	179	2.5
38	Bryan e/o Jamboree	11400	45	4	67.5	34	73	158	26000	45	4	71.6	64	138	297	4.1
40	Bryan w/o Jeffrey	5200	50	4	65.2	< RdHW	52	111	10000	50	4	68,8	42	90	193	3.6
139	Campus e/o Jamborce	19100	45	4	69.7	48	103	222	30000	45	4	72,2	70	151	325	2.5
238	Carlson s/o Michelson	2300	35	6	57.8	< RdHW	< RdHW	< RdHW	8000	35	6	63.4	< Rahw	39	84	5.6
323	Commercentre w/o Bake	500	45	4	53.9	< Rahw	< RaHW	< RdHW	3000	45	4	62.2	< RdHW	33	70	8.3
249	Culver n/o Bryan	16800	45	3	69.2	44	95	205	42000	45	6	73.7	88	190	410	4.5
250	Culver n/o Trabuco/i-5	27300	45	3	71.3	<del>6</del> 1	132	283	62000	45	6	75,4	115	247	532	4.1
152	Del Mar e/o Newport (NB)	9600	30	2	62.4	16	34	72	16000	30	2	64.4	21	46	98	2
82	Dyer e/o SR-55	43300	40	6	72	68	146	315	79000	40	6	75	108	232	500	3
81	Dyer w/o Hotel Terrace	32500	40	6	70.8	57	122	262	79000	40	6	75	108	232	500	4.2
64	Edinger e/o Jamboree	20600	50	6	71.2	60	130	279	37000	50	6	74.5	100	215	463	3.3
62	Edinger e/o Red Hill	17900	55	6	71.6	64	138	297	57000	55	6	77.5	158	341	734	5.9
61	Edinger w/o Red Hill	28900	40	4	70.2	52	111	239	62000	40	6	73.9	91	196	422	3.7
60	Edinger w/o SR-55	32600	45	4	72	68	146	315	59000	45	6	75.2	111	239	516	3.2
41	El Camino Real n/o Main	6400	35	4	62.3	< RdHW	33	71	10000	35	4	64,3	< RdHW	45	97	2
42	El Camino Real w/o Newport	8900	35	4	63.7	< RdHW	41	88	18000	35	4	66.9	31	67	144	3.2
431	El Toro e/o Marguerito	11700	55	2	69.7	48	103	222	21000	55	2	73.2	82	176	379	3.5
432	El Toro e/o Santa Margarita	8500	55	6	68.3	39	83	179	24000	55	6	73.8	90	193	416	5.5
430	El Toro n/o Glenn Ranch	11100	55	2	69.5	46	100	215	16000	55	2	72	68	146	315	2.5
436	El Toro n/o Jeronimo	33800	50	6	73.3	83	179	385	52000	50	6	76	126	271	583	2.7
437	El Toro n/o Muirlands	36300	50	6	73.6	87	187	403	58000	50	6	76.4	134	288	620	2.8
438	El Toro n/o Rockfield	42200	40	6	71.9	67	144	311	64000	40	6	74.1	94	202	435	2.2
440	El Toro s/o Avd Carlota	34200	35	6	69.5	46	100	215	50000	35	6	713	61	132	283	1.8
435	El Toro s/o Trabuco	31500	50	6	73	79	171	368	50000	50	6	75.8	122	262	565	2.8
559	FTC s/o Alton	21000	65	6	74.1	94	202	435	91000	65	6	81.7	301	649	1398	7.6
560	FTC No Lake Forest	18000	65	6	73.5	86	202 184	397	78000	65	6	81	27l	583	1256	7.5
558	FTC s/o Portola	23000	65	-	74.5	80 100	215	397 463	94000	65	6	81.8	306		1420	7.3
338 561		20000	65	6	1	91		403 422	.,		-			659		
301	FTC s/o Santa Margarita	10600	55	6 4	73.9 69.3	45	196 97	208	88000 27000	65 55	6	81.5 74.3	292 97	629 208	1356 449	7.6

County of Orange

**Table 8.3-7** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results
Year 2020 No Project vs. Existing No Project Conditions Existing Segments with 1.5 dB or Higher Traffic Noise Increase

CTM2.8	MCAS El Toro		Air	rport Systes	n Master Pla					No Pr	ofect Alten	native 2020 - 1				
		***************************************			CNEL at	70 dB	65 dB	60 AB			- 4	CNEL at	70 dB	65 dB	60 dB	CNEL
#185 Ir			Spd.	# of	SOR.C.R.	CNEL	CNEL	CNEL		Spd.	# of	50ft.C.R.	CNEL	CNEL	CNEL	Increase
ID# 309	Existing Segment Names	ADT	(mph)	Lanes	(dBA)	(FL)	<u>(FL)</u>	(FL)	ADT	(mph)	Lanes	(dBA)	(Ft.)	<u>(FL)</u>	(Ft.)	dBA
309 454	Glenn Ranch w/o El Toro	4100	55	4	65.2	< Rahw	52	111	9000	55	4	69.5	46	100	215	4.3
170	Glenwood w/o Moulton	#800 26900	50	4	67.5	34	73	158	13000	50	•	69.9	49	106	229	2.4
	Grand s/o Edinger		45	6	71.2	60	130	279	46000	45	6	74.1	94	202	435	2.9
244	Harverd s/o Main	17600	50	4	70.5	54	116	251	25900	50	4	72,8	77	166	357	2.3
246 159	Harvard s/o University Holt n/o Irvine	12100 6900	45 40	4	67,7	35	76	163	18000	45	4	70	50	108	232	2.3
160	Holt s/o fryine	6000		4	64	< RdHW	43	92	13000	40	4	67.1	32	69	149	3.1
66	Irvine Center e/o Culver	22000	30 55	•	60.3	< RdHW	24	52	9000	30	4	61.9	< RdHW	31	67	1.6
69	Irvine Center e/o Sand Cyn	12800	55	6 4	72.5	73	158	341	49000	55	6	76.9	144	311	669	4.4
71	Irvine Center e/o Sand Cyn Irvine Center e/o Alton	12800	55		70.1	51	109	236	40000	55	6	76	126	271	583	5.9
72	Irvine Center s/o Alton	25600	55	6	70.1	51	109	236	34000	55	6	75.3	113	243	524	5.2
371	Irvine Center s/o Arion Irvine Center s/o Bake	30400	55 60	6	73.1	80	173	374	47000	55	6	76.7	140	301	649	3.6
370	Irvine Center s/o I-405	27900	60	4	74.8	104	225	485	66000	60	6	79.3	208	449	967	4.5
70			55	6	74,5	100	215	463	62000	60	6	79	199	429	924	4.5
65	Irvine Center w/o Barranca	13300 21600	33 55	-	70.3	52	113	243	42000	55	6	76.2	130	279	601	5.9
67	tryine Center w/o Culver	18300		6	72.4	72	136	335	43000	55	6	76.3	132	283	610	3,9
68	Irvine Center w/o Jeffrey	16900	55	6	71.7	65	140	301	46000	55	6	76.6	138	297	639	4.9
23	Irvine Center w/o Sand Cyn Irvine c/o Culver	23000	55 50	4	71.3	61	132	283	40000	55	6	76	126	271	583	4.7
27	Irvine e/o ETC East Leg	19700	40	4	71.6 68.9	64	138	297	42000	50	6	75	108	232	500	3.4
21	Irvine e/o Jamboree	22900	50	6	71.6	42 64	91 138	196 297	41000	40 50	•	72.1	69	149	320	3.2
16	Irvine e/o Prospect	28000	40	4	70.1	51			40000		6	74.8	104	225	485	3.2
26	Irvine e/o Sand Canyon	19700	65	7	73.9	8) 31	109	236 422	41000	40	4	72.1	69	149	320	2
15	Irvine e/o Yorba	29300	35	š	68.9	42	196 91	196	48000 47000	65 35	4	78.9	196 59	422	910	5
325	Irvine w/o Alton	18900	65	4	73.7	88	190	410	,		4	71.1		128	275	2.2
18	Irvine w/o Browning	24200	45	- 1	70.7	56	120	258	32000 44000	65 45	6	77.1 73.9	149	320 1 <b>96</b>	690	3.4
20	Irvine w/o Jamboree	25500	50	6	72.1	59	149	320	44000	50	6	75.2	9k 111	239	422	3.2
24	Irvine w/o Jeffrey	22506	50	5	71.5	63	136	320 292	41000	50 50	5	74.9	106	239	516 492	3.4
17	Irvine w/o Red Hill	29700	40	á	70.4	53	115	247	47000	40	6	72.7	76	163	351	2.3
25	Irvine w/o Send Canyon	17400	65	4	73.3	83	179	385	42000	65	4	78.3	179	385	830	2.3
19	Irvine w/o Tustin Ranch	23800	45	6	70.7	36	120	258	39000	45	6	73.4	84	182	39t	2.7
230	Jamboree n/o Alton	30700	50	1	72.9	78	168	362	73000	50		77.4	156	335	723	4.5
229	Jamboree n/o Barranca	34000	50	å	73.3	83	179	385	88000	50	i	78.2	176	379	817	4.9
223	Jamboree n/o Bryan	26200	45	•	71.1	59	128	275	47000	45	6	74.2	95	205	442	3,1
227	Jambores n/o Edinger	41400	50		74.2	95	205	442	104000	50	4	79	199	429	924	4.8
224	Jamborec n/o El Camino Real	33400	45	å	72.1	69	149	320	51000	45	•	74.5	100	215	463	2.4
222	Jambores n/o frvine	22900	45	5	70.5	54	116	251	34000	45	6	72.8	77	166	357	2.3
231	Jamboree n/o Main	34800	50	Ė	73.4	ũ	182	391	69000	50		77.2	151	325	701	3.8
221	Jamboree n/o Portola	22000	50	4	71.5	63	136	292	42000	50	6	75	108	232	500	3.5
220	Jamboree n/o Tustin Ranch	19000	55	7	71.8	56	142	306	47000	55	4	76.7	140	301	649	4.9
228	Jamboree s/o Edinger	39800	50	ì	74	92	199	429	107000	50	ì	79.1	202	435	938	5.1
279	Jeffrey n/o Bryan	21200	60	6	73.3	83	179	385	34000	50 60	6	76.4	134	288	620	3.1
281	Jeffrey n/o 1-5	29806	60	6	74.7	103	222	477	59000	60	6	78.8	193	416	896	4.5
280	Jeffrey n/o Trabuco	24900	60	6	74	92	199	429	36000	60	6	76.6	138	297	639	
282	Jeffrey s/o Walnut/I-5	32200	55	5	74.1	94	202	435	48000	55	6	76.8	142	306	659	2.6
305	Laguna Canyon n/o Alton	1500	45	4	58.7	< RdHW	< RdHW	433	7000	45	4	65.9				2.7
306	Laguns Canyon s/o Alton	3100	45 45	2	58.7 61.8	< KOHW	< KOHW 31	66	12000	45	2		27	57	124	7.2
456	Laguna Canyon Do Alton  Laguna Hills e/o Moulton	15000	•5 50	4	69.8	45	31 104	90 225	23000	43 50	4	68.3 72.4	3 <del>9</del> 72	#3 156	179 335	6.5
458	Laguna Hills w/o Aliso Crk	3100	50	6	62.9	< RdHW	36	223 78	29000	50 50	•					2.6
457	Laguna Hills w/o Moulton	16500	50					239			6	73.4	84	182	391	10.5
413				6	70.2	52	111		50000	50	6	75.8	122	262	565	5.6
	Lake Forest n/o Trabuco	26700	50	4	72.3	71	153	330	44000	50	4	75.2	111	239	516	2.9
412	Lake Forest s/o Rancho	13500	50	4	69.3	45	97	208	37000	50	4	74.5	100	215	463	5.2
445	Los Alisos e/o Marguerise	6900	50	4	66.4	29	62	134	16000	50	4	70.8	57	122	262	4.4
446	Los Alisos n/o S. Margarita	7500	50	4	66.8	31	66	142	11000	50	4	69.2	44	95	205	2.4
120	Main e/o MacArthur	33200	50	6	73.2	82	176	379	51000	50	6	75.9	124	266	574	2.7
121	Main w/o Jamborec	24000	50	6	71.8	66	142	306	37000	50	6	74.5	100	215	463	2.7

County of Orange

**Table 8.3-7 (Cont.)** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results
Year 2020 No Project vs. Existing No Project Conditions Existing Segments with 1.5 dB or Higher Traffic Noise Increase

MI.	MCAS El Toro		Aiı	rport System	n Master Pla				1	No Pr	oject Alten	native 2020 - 1				
_			Spd.	# of	CNEL at 50ft.C.R.	70 dB CNEL	65 db CNEL	60 4B CNEL		Spd.	# of	CNEL at	70 dB CNEL	65 dB CNEL	60 4B CNEL	CNE4
D#	Existing Segment Names	ADT	(dem)	Lanes	(dBA)	(PL)	(FL)	(F4.)	ADT	(soph)	Lunes	(dBA)	(FL)	(Ft)	(FL)	dBA
119	Main w/o MacArthur	26500	50	4	72.3	71	153	330	46000	50	6	75.4	115	247	532	3.1
518 474	Main w/o Red Hill	24600	50	4	71.9	67	144	311	41000	50	6	74.9	106	229	492	3
173	Marguerite n/o S. Margarita	9200	45	•	66.5	29	63	136	14000	45	4	68.9	42	91	196	2.4
131	Marguerite s/o El Toro	10200	45	•	67	32	68	146	19000	45	4	70.3	52	113	243	3.3
32 77	Michelson w/o University Moulton n/o Alicia	4700 24400	35 55	2	60,9	12	27	57	9000	35	2	63.9	20	42	91	3
74	Moulton n/o El Toro	37600	22 45	6	72.9 72.7	78 76	168 163	362	46000	35	6	76.6	138	297	639	3,7
78 78	Moulton n/o Le Paz	20700	45	6	70.1	/6 51	103	351 236	69000 40000	45 45	6	75.9	124	266	574	3.2
75	Moulton s/o El Toro	35600	45	6	72.4	72	156	335	76006	43 45	6	73.5 76.3	86 132	184 283	397	3.4
76	Moulton s/o Glenwood	26800	55	6	73.3	83	179	385	63000	55	6	78	171	263 368	610 792	3.9 4.7
79	Moulton s/o La Paz	16900	45	6	69.2	44	95	205	32000	45	6	72.5	73	158	341	3.3
72	Moulton s/o Lake Forest	33900	45	6	72.2	70	151	325	61000	45	6	75.3	113	243	524	3.1
73	Moulton s/o Ridge Route	33700	45	6	72.2	70	151	325	63000	45	6	75.5	116	251	540	3.3
48	Muirianda e/o Alton	13200	50	4	69.2	44	95	205	19000	50	ä	71.6	64	138	297	2.4
66	Newport s/o Walgat	12100	35	- À	65	< Ratiw	50	108	25000	35		68.3	39	83	179	3.3
16	Portole n/o El Turo	29200	55	6	72.1	69	149	320	50000	55	6	77	146	315	680	4.9
11	Portoia w/o Bake	11300	55	6	69.6	47	101	218	20000	55	6	73	79	171	368	3.4
5	Portola w/o Jamburee	600	45	4	54.7	< RdHW	< RdHW	< RdHW	8000	45	Ä	66.5	29	63	136	11.4
22	Rancho e/o Lake Forest	3300	50	4	63,2	< Rahiw	38	82	41000	50	4	74.9	106	229	492	11.3
12	Red Hill s/o MacArthur	31500	50	6	73	79	171	368	47000	50	6	75.5	116	251	540	2.5
27	Ridge Route e/o Moulton	7600	45	2	65.7	26	56	120	11000	45	2	67.9	36	78	168	2.2
99	Sand Canyon n/o Alton	16600	55	6	71.2	60	130	279	34000	55	6	75.3	113	243	524	4.1
00	Sand Canyon n/o 1-405	18600	55	4	71.7	65	140	301	36000	55	4	75.5	116	251	540	3.8
95	Sand Canyon n/o 1-5	25200	60	6	74	92	199	429	44000	60	6	77.5	158	341	734	3.5
3	Sand Canyon n/o Irvine	12600	60	4	71	58	126	271	18000	60	4	73.6	87	187	403	2.6
4	Sand Canyon n/o Trabuco	17600	60	4	72.5	73	158	341	33000	60	4	76.3	132	283	616	3.8
7	Sand Cyn n/o Irvine Cemer	21600	50	6	71.4	62	134	288	39000	50	6	74.7	103	222	477	3.3
3	Sand Cyn e/o Irvine Center	16700	55	4	71.3	61	132	283	34000	55	6	75.3	113	243	524	4
	Santa Ana s/o British	10400	45	4	67.1	32	69	149	15000	45	4	69.2	44	95	205	2.1
17 17	Santa Margarita e/o El Toro	28800	56	6	72.6	75	161	346	50000	50	6	75.8	122	262	565	3.2
19	SJHTC n/o Lagues Curryon SJHTC s/o Aliso Creek	53000 47000	65 63	6	78.2 77.6	176	379	\$17	108000	65	6	82.4	335	723	1557	4.2
88 88	SJHTC s/a El Toro	47000	65	6 6	77.6	161 161	346 346	745 745	99000	63	6	81.6	297	639	1377	1
98 51	South Bristol w/o Jamborec	17100	45	3	69.2	44	340 95	205	26000	65	6 3	82	315	680	1464	4.4
)2	SR-133 n/o 1-405	20000	65	6	73.9	91	93 196	422		45	6	71.6	64	138	297	2.4
10	SR-133 #0 I-5	20000	65	6	73.9	91	196	422	42000 47000	65 65	6	78.3 78.8	179 193	385 416	\$30 \$96	4.4
97	SR-55 #0 SR-73	128000	65	8	62	313	680	1464	181000	65	8	84.7	477	1029	2216	4.9
91	SR-73 n/o SR-55	87000	65	6	80.3	243	524	1128	136000	65	6	83.4	391	843		2.7
36	SR-73 s/o Campus/Irvine	44000	65	ő	77.3	153	330	712	66000	65	6	80.3	243	524	1813 1128	3.1
19	SR-73 Wo SR-55	79000	65	6	79.9	229	492	1061	116000	65	6	\$2.7	351	757	1630	2.8
12	Trabuco w/o Jeffrey	4100	55	- I	65.2	< RdHW	52	111	14000	55	4	71.4	62	134	288	6.2
9	Trabuco w/o Sand Canyon	3900	35	ž	60.1	< RdHW	24	51	21000	35	2	67.6	35	75	161	7.5
37	Tustin Ranch n/o Bryan	16600	45	6	69.1	44	94	202	28000	45	6	71.9	67	144	311	2.8
<b>39</b>	Tustin Ranch n/o I-5	29600	45	6	71.6	64	138	297	43000	45	ă	73.8	90	193	416	2.2
16	Tustin Ranch n/o trvine	15200	45	6	68.7	41	88	190	34000	45	6	72.8	77	166	357	4.1
5	Tustin Ranch n/o Portole	2700	50	6	62.3	< RdHW	< Rahw	71	17000	30	ă	71.1	59	128	275	8.8
)B	Tustin Ranch s/o Bryan	20300	45	š	70	50	108	232	29000	45	6	72.1	69	149	320	2.1
0	Tustin Ranch t/o 1-5	18400	45	6	69.6	47	101	218	43000	45	6	73.8	90	193	416	4.2
59	Valencia w/o Red Hill	3700	35	2	59.9	« RdHW	23	49	15000	35	4	66.1	27	59	128	6.2
8	Von Karman n/o Birch	13500	40	ã	66.9	31	67	144	20000	40	À	69	43	92	199	2.1
17	Von Karman n/o Campus	17100	50	Ä	70.4	53	115	247	25000	50	Ä	72.8	77	166	357	2.4
5	You Karman n/o Main	16300	50	6	70.1	51	109	236	32000	50	6	73.8	90	193	416	3.7
6	Von Karman n/o Michelson	21500	50	6	71.4	62	134	288	33000	50	6	74	92	199	429	2.6
4	Von Karman s/o Barranca	16400	50	6	70.2	52	113	239	36000	50	6	74.4	98	212	456	4.2
54	Walnut w/o Jamboree	10000	40	4	65.6	25	55	118	15000	40	ă	67.8	36	77	166	2.2
19	Watter w/o Culver	3100	40	i	60.6	< RdHW	25	55	18000	40	Ä	68.6	40	87	187	8
78	Warner w/o Harvard	1400	40	4	57.1	< RAHW	< RdHW	32	16000	40	6	68	37	79	171	10.9
75	Warner w/o Red Hill	14800	40	6	67.3	< RdHW	71	153	49000	40	6	72	68	146	315	4.7
67	West Vale Loop a/o Warner	7200	45	4	65.5	25	54	116	11000	45	ä	67.9	36	78	168	2.4
55	Yorbs n/o irvine	6700	35	4	62.5	< RdHW	34	73	14000	35	7	65.8	26	57	122	3.3

County of Orange

**Table 8.3-7 (Cont.)** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results
Year 2020 No Project vs. Existing No Project Conditions Existing Segments with 1.5 dB or Higher Traffic Noise Increase

higher noise increase over their corresponding existing no project level. Although these noise increases would be considered significant, they are due to area growth and planned development in the region. Table 8.3-8 shows the noise levels along new road links that would be constructed under the year 2020 no project scenario. Many of these new road links would have the 65 dB CNEL affecting sensitive uses adjacent to the roads.

Table 8.3-9 shows the road links that would have a potential noise increase greater than 1.5 dB under the 2020 ETRPA Nonaviation Plan Alternative over the existing plus committed (2020) scenario. Although a total of 58 road links with 2020 ETRPA Nonaviation Plan Alternative would have a 1.5 to 3.0 dB in noise level increase over their corresponding existing plus committed scenarios and a total of 111 road links would have a 3 dB or more noise level increase over their corresponding existing plus committed scenario, in order to determine the impacts associated with the 2020 ETRPA Nonaviation Plan Alternative, a comparison with the year 2020 existing plus committed scenario was also conducted. A total of 60 road links under the 2020 ETRPA Nonaviation Plan Alternative would have an increase of up to 3.0 dB in noise levels over their corresponding existing plus committed (2020) scenario. Only two road links would have noise level increases of more than 3 dB. Along Commercentre Drive west of Bake Parkway, there would be a 3.7 dB increase. Along Trabuco Road east of Sand Canyon Road, there would be a 10.6 dB increase with the implementation of the 2020 ETRPA Nonaviation Plan Alternative. These noise level increases would be considered significant with the 2020 ETRPA Nonaviation Plan Alternative.

However, no existing sensitive use along these segments of the two road links would have more than 3 dB noise increases. Any future sensitive uses proposed along these two road links would require mitigation or rejection. In addition, a total of 41 road links would have lower traffic noise with ETRPA Nonaviation Plan Alternative. A total of 91 road links would have no measurable change in traffic noise in year 2020 with implementation of the ETRPA Nonaviation Plan Alternative.

Table 8.3-10 summarizes the number of road links that would have noise level increase over their corresponding existing levels for the existing plus ETRPA Nonaviation Plan Alternative, 2020 No Project, and 2020 ETRPA Nonaviation Plan Alternative. Table 8.3-11 summarizes the number of road links that would have noise level increase with 2020 ETRPA Nonaviation Plan Alternative over their corresponding 2020 No Project levels.

# Comparison of Alternative Impacts to Proposed Project Impacts

This alternative would not have any of the aviation noise impacts of the project at the MCAS El Toro site.

CCTM	12.8 2020 MCAS El Toro		***************************************	C	NEL RESUL	TS	······································	
NO PF	OJECT - New Segments				CNELat	70 dB	65 dB	60 dB
			Spd.	# of	50ft.C.R.	CNEL	CNEL	CNEL
TD#	New Segment Names	ADT	(mph)	Lanes	(dBA)	(Ft.)	(Ft.)	(Ft.)
392	Alton n/o Commercentre	35000	55	6	74.8	104	225	485
393	Alton n/o Irvine	35000	55	6	74.8	104	225	485
391	Alton n/o Rancho	28000	55	6	73.9	91	196	422
92	Barranca w/o Sand Canyon	18000	35	4	66.3	28	61	132
200	Birch s/o North Bristol	10000	40	2	65.4	25	53	115
248	Culver n/o Irvine	29000	45	6	71.5	63	136	292
511	East Access n/o Irvine	0	35	4	0	0	0	0
509	East Access s/o Irvine	0	35	4	0	0	0	0
63	Edinger w/o Jamboree	38000	55	6	75.2	111	239	516
512	ETC Connector (N & S)	0	35	4	0	0	0	0
513	ETC Connector (N)	0	35	4	0	0	0	0
552	ETC East Leg n/o Irvine	58000	65	6	79.1	202	435	938
550	ETC East Leg n/o Jeffrey	96000	65	6	81.3	283	610	1315
553	ETC East Leg s/o Irvine	48000	65	6	78.3	179	385	830
551	ETC East Leg s/o Jeffrey	96000	65	6	81.3	283	610	1315
555	ETC West Leg n/o Irvine	69000	65	6	79.9	229	492	1061
554	ETC West Leg n/o Portola	65000	65	6	79.6	218	470	1013
556	ETC West Leg s/o Irvine	60000	65	6	79.3	208	449	967
557	FTC s/o ETC East Leg	79000	65	6	80.4	247	532	1145
500	Irvine e/o Perimeter Rd	47000	35	4	70.5	54	116	251
22	Irvine w/o Culver	46000	50	6	74.8	104	225	485
236	Jamboree n/o California	38000	50	6	74	92	199	429
304	Laguna Canyon n/o Barranca	1000	45	4	56.9	< RdHW	< RdHW	31
385	Laguna Canyon s/o Bake	46000	55	4	76	126	271	583
386	Laguna Cyn n/o Aliso Creek	30000	55	4	74.2	95	205	442
384	Laguna Cyn s/o Lake Forest	46000	35	4	70.4	53	115	247
303	Laguna Cyn s/o Technology	3000	35	4	58.5	< RdHW	< RdHW	40
502	Marine Way e/o Sand Canyon	3000	35	4	58.5	< RdHW	< RdHW	40
167	Newport n/o Edinger	33000	35	6	68.9	42	91	196
168	Newport s/o Edinger	15000	35	6	65.5	< RdHW	54	116
302	Oak Canyon e/o Sand Canyon	6000	35	4	61.5	< RdHW	29	63
506	Perimeter Rd e/o Air Cargo	0	35	2	0	0	0	Ö
504	Perimeter Rd n/o Marine Way	l ŏ	35	2	Ö	Ö	ō	ō
507	Perimeter Rd s/o Astor	ا o	35	2	lo	ō	Ö	ō
503	Perimeter Rd s/o Irvine	l ö	35	2	l o	ō	ō	ŏ
505	Perimeter Rd s/o Marine Way	lõ	35	2	l ŏ	ŏ	0	ŏ
8	Portola e/o Culver	21000	50	2	71.4	62	134	288
6	Portola e/o Jamboree	25000	50	6	72.2	70	151	325
315	Portola n/o Rancho	27000	65	4	75.8	122	262	565
7	Portola w/o Culver	29000	50	2	72.8	77	166	357
9	Portola w/o Jeffrey	14000	35	2	65.2	24	52	111
320	Rancho e/o Alton	10000	55	4	69.4	46	98	212
321	Rancho e/o Bake	36000	55	4	74.9	106	229	492
355	Rockfield e/o Alton	0	40	4	0	0	0	0
301	Sand Canyon s/o I-405	3000	50	4	63	< RdHW	37	79
73	Technology e/o Oak Cyn	2000	35	4	56.7	< RdHW	< RdHW	30
501	Trabuco e/o Sand Canyon	2000	35	4	56.7	< RdHW	< RdHW	30
211	Tustin Ranch n/o Edinger	41000	45	6	73	79	171	368
	Tustin Ranch s/o Edinger	26000	45 35	6	67.9	36	78	168
212	Ţ.	42000	35 35	6	70	50	78 108	232
213	Tustin Ranch s/o Warner Warner e/o Red Hill	l .	50	6	75.4	115		
76		52000			73.4	68	247	532 315
<i>77</i> 510	Warner w/o Jamboree	24000	50	6	1		146	315
	West Access n/o Irvine	0	35	4	0	0	0	0

County of Orange

**Table 8.3-8** 

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results Year 2020 No Project Condition - New Segments

,		,										_													*****																					_	_					
		<b>M</b>	3:	<b>;</b> ;	7:	3 =	7	*	*	7	¥ ;		7 -	7	7	<b>7</b> ;	7 =	7	# :	₹ :	7 4	₹	7	7 7	7 7	7	<b>7</b>	<b>.</b>	7	7	7 -	7	¥:	<b>7 7</b>	•	• •	• •	•	•	•		• •	•	•	•		•	•		ı <b>•</b>	•	•
		ã	2:	2 5		2 2		315	3	**	E I	1 :	2	8	424	3 :	i	\$	Ī:	9:	ž	22	3	<b>8</b> 5	2 2	1533	ž:	. E	*	2:	2 5	. \$	<b>.</b>	3 5	Ξ	2 ?	E	2	5 G	Ē	<b>÷</b> :	Ė	2	3	= =	12	*	323	25	=	5	
							. 52:	•	*	<b>.</b>	₹:	2 2	* 52		2	<b>7</b> 2.3	0 52		<b>:</b>	*:		• •	52 5	R 5	2 22	~	æ:	9 9		₩:	2 3	2	<b>5</b> :	7 2	**	m =		<b>2</b> :	2 16	2	2:	~ 2		<b>.</b>	2 1	<u> </u>	<u></u>	<b>•</b> :	• •		<b>%</b>	₽.
ı	200	ı	,	_				2	7	22	22 9	2 5	2 2	2	2	= 1	2 2	\$	**	2 3	9 5	**	<b>*</b>	2 2	: =	2	£ ;	8 23	1 77	2:	2 %	2 2	\$:	13	25	¥ei∷	: 2	# ?	2 2	=												#
- VII S		Œ	2		<u> </u>	2	Ξ	#	8	æ	₹;	<b>a</b> 8	£ <u>=</u>	ğ	Z.	ğ ;	2 2	Ξ	= :	3 3	2	Ξ	3	2 2	1	200	2	2 2	ĸ	# 1	3 5	8	2	8 2	8.2	2 3	8	\$ 2	? ;	3	Ξ\$	3 3	C PARTY	=	≣ a	<u>.</u>	20	3 :	<b>3</b> 5	; <b>\$</b>	E	*
Non-Arieties Abernative 2026 - All Se	8 C W	7	£ ;	ī	,	2	ř	Ľ	74.5	ř.	1		7	F	7	2,5		ř	*		į į	2	ž				7		74.4	5		Z	7	7.7	2	7 2	1	2,	7 6	ř	2	i	3	3	7	F	74.5	t i	: 3	3	72.8	44.
N. A.	*	3	•	•				•	<b>-</b>	•	•				•				•••			•	•		•	•	•			* '	• •	. •	•			* *				-	•	• •		<b>.</b>	•		•	•	• •	-	•	•
MAN	•	1																																																		
	J.	1	<b>%</b> 2	¥ ¥	¥ \$	3	2	Я	25	3	¥ ;	2 \$	3	2	3		3 23	8	<b>R</b> :	2	2 2	2	¥ :	2 2	3	\$	\$ :	2 \$	3,	<b>\$</b> \$	2 3	\$	3;	8.3	3	÷	2.	₽:	\$ \$	\$	<b>,</b>	2 2	2	<b>2</b>	\$ \$	<b>.</b>	8	\$ \$	2 \$	\$	<b>R</b> :	\$
		VOZ	2004	2000	1000	31000	26000	21000	2000	8	3	34000	8	32000	33,000	20087	\$ 000 P	\$ 7000	41000	4	28000	43600	000		000	106000	0007	71000	34000	42000	900	63000	9	100	<b>45000</b>	29000	47000	41000	100	39009	44000	28000	1000	00	100	\$6000	17000		ieono	000	2500	<b>\$100</b> 0
	•					•••••	••••																					******	*****																					,ş,		
							٠																			#		Ē.					.91											¥ O								
,		N. N.				j	T	8	Colonia	Î	Ĵ	į	î	Alton	an) (mar	4	Alto		8 2			ì	i.	• .	e L	AL Carro		j	3	7	1	3	Ports		ed reine			2 2	1	Jan Ch	¥	ğ	3	i de la companya de l		į	×	Ŧ		b	<b>**</b> :	=
		Sec.	O Contract	o Brasion			A 100 KE	Lwb Ah	a wo Te	2 2 2	3		4	A I	S. T.	o levino	Q.	my mob 1-3	Den sio Traba		a to lead	who March	o MacAn		West Loss and Dr.	7 Se 1.	Contract	A 18.55	o Jembo	the Parent So I		nn ave Rockfe		ATC SO ALM C	2	ficia w/o Jamil . w/o Taritia	O Yearba	o Prosper	4	S Tage	10 Jan 10		mo Aced mets between	ino Read w/o N		of a final	ete Jimi		W/o Hery	Ę	T Pour	-
		Ž	Amon		, C		j	Berna			Carrie		2	frying C	A Mess of	Altere	2	À			Altre	Mais at			ETC	SHIC	1		Main W	1	100	10m	1		SE-72	1000	a lar	1		in the w		2	2			3		Variation	Name of	A SECOND	A lice of	i ser
			£ \$	2 :	: 8	2	ž	ì	E	S.	I a	2 3	2	=	2	I :	<b>:</b> =	Ħ	÷	\$ 2	2	=	2 ;	5 5	3	ž	3 1	<b>:</b>	2	2 :	3	5	3.3	£ 3	Ş.	n 2	*	<b>*</b> :	==	2	3 :	; =	<b>=</b>	<b>;</b>	3 2	3	3	r F	= =	£	2	=
7		7	Z ,		1 1	<b>1</b>	¥	2	w	ed y	W 1		4 500	<b>.</b>	*	<b>W</b> 1	•	¥	10	4 6	6 64	M	iai s	4 14	Z	<u></u>	-	4 54	14	M 1	1 14	W	Z Z	E tel	<b>M</b> (	N 8	86	W) 1	4 544	ted	<b>*</b>	3 14	<b>W</b>	<b>84</b> 6	4 2	in:	<b>94</b>	** 3	Z 94	<b>34</b>	<b>W</b> 1	×
	Š	1	¥ :	≘ \$	\$ 5	3 3	9	362	7	ž	ž ;	2 5	3	Ä	Ş	**	ŧ	8	¥ :	9 9	Ē	332	2	2	90	1557	3:	2 2	<b>46</b>	2 3	: 3	2	ē ;	117	2	2 %	£	2	3 8	Z	2:	Ź	2	Ξ:	2 7	įĕ	<b>3</b>	512	2 2	9	<u>S</u>	Ę
	i S	4	£ :	3 2	į	Ē	2	2	ž.	E :	오 :	:	8	3	Ž.	a :	) <u>e</u>	9 ;	78	÷ ₹	ï	#	4 :	9 #	4	5	<b>R</b> F	3 8	233	<u> </u>	Ē	707	2 5	3	7.	3 =	2	<u> </u>	<u> </u>	2	Š:	: <u>#</u>	÷	2 }	5 2	<u> </u>	57	¥ !	ę	2	<u>3</u>	2
	d	l					22	F	= :	2:	3 5	: 3	*	•	-	<b>3</b> 8	<b>3</b> \$	2	<b>#</b> :	<u>.</u> 5	Z	2	<b>#</b> :	. A	2	2	2 2	2 2	8	2 5	2	*	<u> </u>		99	2 2	*	\$ ×	t =	<b>3</b> :	= 3	Į,	***		==	· <u>#</u>	8	3 5	1 =	<b>.</b>	<b>2</b> 3	ŧ
	ند ع	ı					_	•-				_																										•	-				٠							•	*	-
1	10		7 ;	į×	76.3	×	3	2	2	Z :	į.	77.	R	5	2	?;	5	7.	***	2 2	<u> </u>	3		: 5	2	3	2;	2.2	74.5	2.5	2	74.8	ři	: :	3	7 2	7.7	2:	32	2	7	7	3	3	25	2.17	2	E &	. 4	\$	7	
THE RESIDENCE OF	*	FIRM	۰-	• •		. 40	•	•	₹.	•	• •	•	• •	*	•	•	•	•	<b>•</b> •			40	• •	• •	•	•	• 4	• •	•	40 4	•	•	•		•	+ 40	•	• •	•	•	** *	. 4	•	<b>.</b>		- 🛥	<b>.</b>	* 4	•	*	•	•
2	Spe	3	នន	3 8	3 33	2	2	2	× :	g :	2 2	3 2	: \$	35	<b>x</b> :	25	2	3:	<b>8</b> 7	2 2	2	2	25	2 3	59	<b>3</b>	2 2	3 3	X	<b>\$</b> \$	<b>.</b>	\$	3 :	<b>.</b>	3:	\$ \$	32	<b>\$</b> \$	2	<b>\$</b> :	3 ≥	: \$	22	<b>x</b> :	<b>;</b>	32	오 !	<b>\$</b> 5	£ \$	\$	<b>3</b> :	R
		7					2	9	<b>Q</b> (	2 4	2 4		· •	ģ	۰ و	2 9		9	2 5	9 9	9	9	2 -		9	8	2 5	ż 🕏	ģ	2 9		*	2 5		<b>9</b> -	. 9	ē	<b>2</b> =	2 0	9	2 4		9	2 4			9 1	2 9		P	<b>9</b> 4	•
l		V	9000	8	2	9	300	2	2	25	2 9	200	200	200 × 1	2	, é	8	366	9 5		8	*	200	3	\$	5	•	2	2	8 6	520	Ī	25	\$	3	8 8	200	8 5	3	286		2092	2001	8 8	200	37000	2		3	9081	2,5000	2
			72	5 #		Š	8		ŝ	₽.	Surrance arts Stand Consus		<b>"</b> ,	F		5	2 6		ţ	į,	:				*	SURTIC ario Lagrana Canyon	2 5							-#	<b>3</b>					4			j.	Ed Capacido Mante with Mark post. Editores miles Cit. 64								
	2	Existing Segment Name	Alich Mo Irvis: Birth sto Yeath Batani		20.00	5	ar of	A least	Bertmeck w/o Technology		ļ	1	101	1 A	j	Aston \$10 Evide Lake Forms on Stootho	Invine Center No Attor		g į	9	Post of	cArchur.			- P	Ĭ	:	5	No.	Toetha Ranch avo (-5 Von Knomen avb Main	omino.	El Toro néo Racifició	\$ 4 **	ő	SR-73 ato Campastiliviae		<b>.</b>	per .	Taring.	Creine with Tuestin Ranch	NO COL	ano de	El Camino Resi arte Main		7	型型	an there		-	E S	3	Ž
		2	Alcon to the second	101	j	j	The state of	Barranca w/o Alton	2			3	3	Contract	j :	Anton too Living Lake Forms do S	Centle	leffley at 1-3		1	A Non w/o Infinbono	No Me	family are Mark Arthur family man be to family	foffiny are Bryss	Yes !	3	j	Dyer c/o 58-53	Main w/o Jambonx	Tuertia Ranch avo (-) Von Kromim avo Ma	4	No.	1	2	0 1 2 1	town Tustis	10 YOU	In time 4/0 Prospect	In its w/o Browning	ach Tu	Posts was summores for each Tourish	Br. ser of a flamborer	die P.	El Carego Rest wh	Edinger min Red Hill	Edinger sto Rad Hill	Edinger of Jambere	Warmer with Auditor	Water n.fo Herverd	Warner w/o Cubres	Alten e/o Rod Hill Majo sefe Beef Hill	Š
CC 1 mare process to little		Exhai	2 4	6	Į	Strains Comber and Stand Cym	Smead C					-	-	S.	Y.	1	- L		El Toris sée Trabusco Torisma Constant cés Cultura	P. C.	A	į	Ĩ	Į.	٤	Ę.		6	N.	- N	į.	1	1	SHITC at Alian Creat		46 wh	Jerino.	1	F	Ĕ.		, in	2		1	Eding	2		*	W. R. S.	V V	MICH STO PART SESS
-		┚						٠.	N							2 **					₫	* :	<b>R</b> =					2	<b></b> ,	2 2				200																-	٠.	
	i		9		1	4	72	2		K 4		•	<b>3</b>	•	_,	ς	-	π:	-		-	-	**	1	÷.	₽.		-	<b></b>	~ ~	*		2 2		2.		<b>22</b> :	= =	=	= ;	3 2	<b>7</b>	~	# 5	3	3	3,	e c	=		<u> </u>	. :

SOURCE: LSA Associates, Inc., 9/99

County of Orange

**Table 8.3-9** 

Table 8.3-9 (Cont.)	
County of Orange	
EIR No. 573	

SOURCE: LSA Associates, Inc., 9/99

12/17/99

		OCTINES MCAS ELTIMS		78.5	Total Atlanta	Contra cutte.	I		7		CCTM24 MCA4 to Two							I	
				Sof.	*	CALL &							Sof	3	MACA		2 d 2 d	e zi	Inches to
	2	Existing Sections Propess	ADT	1	-	(PA)	ı	l	3		Existing Second Number	YOU	(Mark)	1	(WW)	Œ	ã	3	¥
	និ	Immborne ato Teatra Reach	2000	æ	*	74.7	2	ž	<u></u>	2	Members are Tunin Ranch		2	+	**	7		\$	3
	ā	Sombauer arts Posterie	9000	<b>R</b> :	•	r,	2	E	*	ä	Manipones w/o Portolia	43000	2	•	ž	5		5,	7
	2 5	PERSONAL PROPERTY SCHOOL	900	<b>:</b>	•	2	± ;	ā	2	E :	Months to Ridge Louis	9005	<b>\$</b>	•	7	#		3,	3
The control of the		Section and the section of the secti	2007	2 :	•	ê	5:		1	* :	MONTH NO. E. LONG.	200	<b>Q</b> :	•	e ;	2		ž	3
March Sales   March Sales	•	The state of the s	1777	2 :		į		2 5			The state of the s		2 :	• •	2	x		ā :	3 ;
	į	Comment Comments & State Comment		7 =		2			? ?	ì		448	2 :	• •	į ;	5 1		i i	7 :
March Color	=	Indus ato Implores	000	3	, ,	3	: ₫	i i		Ę	Parties of the land of the lan	90017	2 5		7	2 2		Ş	7 -
Compared by   Compared by	2	Invites with Cultrer	900	2		17.	3	ă		: 2	Print of Column	7,000	t 5			3			i
The control of the co	Ş	brytas Center w/o Jeffrey	44000	<b>*</b>	•	7	2	i i	8	1 \$	Porter County wite Actions	900,	1 22	•	2	3		1	3
The control of the	ន	Sambane at irrine	34000	<b>.</b>	•	1.1	E	<u>\$</u>	32	3	lembors no invise	15600	. 2	•	ř	#		Z	3
The control of the	Ë	Monthes are Alicia	00094	2	•	36.	2	ř.	3	ï.	Months are Ations	47000	\$	•	ž	3		3	9.6
The control of the	-	Portols are Culture	00010	2	_	77	2	3	Z	-	Postola eto Culver	32000	2	~	31.6	3		ž	7
The control of the	z	britis wie Jeffing	41000	Z	~	74.9	ž	£1	₩. **	2	Print wite before	(Amp)	\$	•	ř	8		<b>3</b>	3
The control of the	Z.	Envises ado Sand Carryon	41600	\$	•	76.0	¥.	ā	=	×	Bryline evo Samel Clery on	90006	3	•	ř	2		ž	2
March   Marc	î	Lagrant Hills to Moulion	23000	£	<b>~</b>	ř	t	*	22	*	Lagrana PHEs are Monthon	2000	×	•	72.6	2		34.	6.1
Marcon to big   Control	•	Prefede with Culver	28000	2	~	7	E	3	357	•	Pontale tule Culture	30000	7	~	£	*		7	0.1
Column 1   Column 2	£	Berment with Invite Conter	9001	2	~	7.1	F	3	357	Z	Seramon and Prime Conter	20000	*	. •				3	
Comparison   Com	¥	Sand Compute ado 3-5	44000	8	•	77.5	45	Ī	7	£	Seed Curve ato 1-5	900	3	•	F	=======================================		S	3
March of Prince   1980   1981   1982   198	2	Class Ranch sto Persols	27000	\$2		Z	*	Ž	. \$	2	Gen Tart of Parks	9	#	•		9		Ş	15
The first control of the control o	*	Pomotis ado El Tamo	20000	3	-	£	3			ž	President St. Time	18	\$	* *	Ē	2		į	1 6
From the control between the control of the control	33	Santa Marcarita eto El Toro	2000	9	•		2	×		;	Carto Managha da Pi Tom	ě	: 5	. •	: ×	2		ē	:
Trick when the control of the contro	=	Erolan Comercia Bates	9	3	• •		ij	11		į	The Course of Marie		2 2	F ×		3 ;		ì	7 :
Trick data bargers	£	FTC s/s Partole	900	1			Į į				STT als Browds		1 3	• •		? :		1	3 :
Trick in the fact Control of the con	35	The state of the s	# ions	3 2	• •	1	į				CTT AT THE		2 4	* 4	1			!	1:
New and with fine control of the con	*	TTC ate Sents Menualty	900	1 2			2		2	3	TTC (1 )		3			į			3 :
	•	Postole w/o ketter	14000	=	-	*	7			•	Percent with Saffani		=			ž×		<u> </u>	15
The order   1000   1	2	India who Sand Conver	42000	: 2		7			9		Project at the Sund Courses	8047	;		1	1		3	1
The control of the co	3,	Bris co Caver	12008	7	-	3	2			2	Part of Calve	1	2 4			•		:	1:
March   Marc	#	Traducts and Jeffrey	14000	**	-	71.4	3			4	Todans with Juffers	9	. 2	• •	-	2		3	12
March Late Format	343	Actibery ato Watered-5	1000			7.7	2		\$	787	Jeffer at Walnut	31000	: =		Ē	3		9	1 2
Fig. 10   Fig. 10   Fig. 11   Fig.	Ę	Movelton s'e Labs Forzat	4	. 2	•	2	ΙΞ		ă	E	Moulton sée Late Posess	9005	3	•	ž	Ξ		3	12
Comparison of Action 1970   Comparison of Action 1970   Comparison of Action of Acti	¥	FTC 4/0 Labs Forest	78000	2	•	=	12		*	3	PIC ate Lafer Novem	13000	3	* **	7	2		1353	2
March of S. Magnerian   1900   50   61   62   64   64   64   64   64   64   64	Ħ	Ranches the Lake Force	41000	2	-	74.9	201		~	=	Reacto ste Lake Yours	45000	8	-	7	Ξ		\$24	3
Extra   Color   March   Marc	\$	Los Alisce ado S. Mangarita	11686	S	-	£.5	1		ě	\$	Los Albana néo S. Marganeta	7000	2	-	8	Ç		712	2
First Light of March   Warmer   St. 1   13   13   13   15   15   15   15	홋	Sand Cueyta also 1-403	34000	22	<b>~</b>	7.7.	<b>9</b> .		3	矣	Sand Conyon a/o 3-405	34000	2	*	2.5	2		372	3
Fire backed by the control of the	8	ETC East Lag also Jefflers	94000	2	-	2	î		315	š	ETC East Log als heftery	185000	Ž.	49		ž		134	3
Section with classes   Section with classes	<b>7</b>	ETC East Leg also leftles	2009	<b>3</b>	-	3	7		213	ī	ETC Cast Lag ato Active;	10,000	3	•	#.7	Ā		K.	3
Second Contract   1900   15   17   17   17   17   18   18   18   18	<b>\$</b> {	5R-133 a/a (-5	2004	<b>3</b> :	•	7	2		ž	ğ	SR-131 av 15	3200	2	-	2	Ã		<b>£13</b>	3
Figure 10   Figure 20   Figu	3	NE-131 WO 1-403	700	28	•	2	Ž.		2	3	SR-133 a/o F-409	#600e	2	•		2		7	3
Particular   1900   15   15   15   15   15   15   15	ŧ ;	Imbusto a to Sand Carrytin	90	<b>n</b> :		***	2		=	\$	Trainings with Sand Convon	2400#	2	~	3	2		5	2
Base Chairs which with the control of the c	Ē	Tortals w/o biggs	8	<b>a</b> :	•	P.	r i		*	=	Portole w/o Bule	2002	æ:	-0	77	<b>:</b>		*	\$
Seed Chapter to be bridged.  Seed Chapter to	5 1	Karasa on mare		g :	•	7	5		Z :	7	Amothe ets Bake	*	2	•	7	<b>±</b>		3	ž
Second Control Contr	Ę	Sand Limiter to Allon		2	•	2	2:		2	ž.	Same Carrier arts Albert		<b>X</b> :	**	ř	ž		7.	3
Out Carrier of Seal Carrier         Control Carrier <t< th=""><th></th><th></th><th>2000</th><th>2 :</th><th>•</th><th></th><th>2 :</th><th></th><th></th><th>Ē,</th><th>AND CANADA STATE</th><th></th><th>¥ :</th><th></th><th>2:</th><th><b>F</b> ?</th><th></th><th>1</th><th><u>.</u></th></t<>			2000	2 :	•		2 :			Ē,	AND CANADA STATE		¥ :		2:	<b>F</b> ?		1	<u>.</u>
Allowed to be compared by the	3			2			253		•	Ę			2 :	•	e ;	2		Ř	<b>.</b>
FTC East Leg to bries	1	Alem s'o Pensis		: 5	, ,		101			į			::		1;	***		? :	2 4
FTC fine Ling who forms   48800   65   64   713   174   315   17	33	ETC East Lag ado by inc	38000	: 2			ä		7	9	ETC Car Les etc State		: 5	•	ā	2			::
Bytes Chiefer   1900   59   4   411   11   11   11   11   11   1	333	ETC Sant Last ado Sevien	41900	3	•	_	Z		2	Ē	FTC flue (As de levies	4300	1 7	•	į	į		3	-
Sear Communication of Communicatio	₹	Bry to mito Arthery	1000	\$	•		<b></b>		<u> </u>	\$	Or in wie keller	3000	8	*	3	\$		21	. :
Altern to National State	Ē:	Stand Cya not lardes Comor	34000	2	•		2		Ë	E	Sand Cya ado levies Conter	93000	2	•	ž	*		25	7
Final Conference   Final Confe	Ē:	Africa and Remarks	38400	2	۰	13.9	5		Z 2	ž	Altan née Rancho	38006	æ	•	2,00	Ξ		3.5	2
Almost de Commentante         35000         55         4         74.4         164         325         44.5         Manual de Albon         45000         45         4         74.2         46.9         4         4.4         74.5         15.5         4.4         4.4         74.5         15.0         4.4         74.7         15.9         4.4         74.7         15.9         4.4         74.7         15.9         4.4         74.7         15.9         4.4         74.7         15.9         4.4         74.7         15.9         15.0         15.0         4.4         74.7         15.9         15.0	£	Invite ado ETC East Log	<del>2</del>	\$	*	7	2		2	2	trying wto ETC East Log	57000	\$	•	77	S		ş	*
Description   1,000   55	2	ARTHUR COMMERCIALISM	3006	æ:	•	7	₫ :		Z ;	ž	Alter also Communication	52000	2	•	Ž	ž		ŝ	<u></u>
December 2015   1900   25    1900   25    1900   25    25	3 5	TANK AND VIDE	0000	s :	• •	F.	<u>\$</u> :		# :	S :	Bryling s.Yo.A.lines	900	2 :	→ '	į	<b>*</b>		9	2
Commerciative Balas (2000 4) 4 (4.2 c Aufrity 1) 200 E 131 Commerciative Balas (2000 5) 4 (4.2 c Aufrity 1) 200 E 131 Commerciative Balas (2000 5) 4 (4.2 c Aufrity 1) 200 E 131 Commerciative Balas (2000 5) 4 (4.2 c Aufrity 1) 200 E 131 Commerciative Balas (2000 5) 5 (4.1 c Aufrity 1) 200 E 131 Commerciative Balas (2000 5) 5 (4.1 c Aufrity 1) 200 E 131	1	TOTAL STATE OF THE PARTY OF THE		8.4	• •		ş ;	F :		5	EMON See Allem	1,000	a :	•	1,1	2		2	2
Tributes to Stand Carolom 250 4 671 C Staff W & Address of Stand Carolom 550 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	1 5		The state of	2 :	• •	2 :		3 ;	3 6	1	2 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1		ጵ :		17	# :		2	2.4
Walant viol Samblerer         150.00         4         67.38         34         75         16 w. of Tunding         27000         50         4         73.4         75         156         155         341           Edingut viol Samblerer         36000         55         4         75.3         111         275         516         6         55         111         275         111         275         111         275         111         275         111         2700         50         4         75.1         119         316 </th <th>3</th> <td>Taken to California</td> <td></td> <td>3 3</td> <td></td> <td></td> <td>A CHIEF CO.</td> <td>***</td> <td>2 5</td> <td>3</td> <th>The state of the s</th> <td></td> <td>0 ;</td> <td>٠,</td> <td>2</td> <td><b>≒</b> :</td> <td></td> <td>Z :</td> <td>2</td>	3	Taken to California		3 3			A CHIEF CO.	***	2 5	3	The state of the s		0 ;	٠,	2	<b>≒</b> :		Z :	2
Wildrage with Emboure         1908 40         4         673 34         77         164 Emboure         5 4         75 31         11         279 314         8         4         61         11         279 314         11         279 314         11         279 314         11         279 314         11         279 314         11         279 314         11         279 314         11         270 314         11         270 314         12         270 314         12         270 314	*		!	:	•	į		*	E 2	į	The same of the same to the same of the sa	23000	2 5	• •	1	2 #		3;	1
Editing to Lamboure 19900 55 6 75.3 111 239 516 76 63 26 75.1 109 336 6 75.1 109 33	* *	Walnut nice implement	280	9	•	82.9	*		1	1 7	THE PARTY OF THE P	3	*	•	3	2		Ę	
Tockwology, ctr Out, C, n. 2020 33 4 56,3 c. fulfilly < Rufrilly 39 N 32 Technology and Burmers and Sand Canada. 35 4 57 13 247 332 N 36 Warmer and Sand Canada. 35 4 66,3 28 61 13 N 37 Warmer and Sand Canada. 36 4 78.3 53 111 22 247 332 N 37 Warmer and Sand Canada. 36 4 78.3 53 111 22 24	2	Edicates w/o Jamborce	000#4	: 22		2	ŧ =			t G		••••							
Tochmology o'n Oth Cyn 2000 33 4 34.7 childry < RAFFW 39 N 73 Tochmology o'n Blanmar win Same (2000 50 6 755.4 115 247 532 N 74 Tochmology o'n Blanmar win Same (2000 50 6 776.3 24 64.3 24 61.3 N 75 Warmen win Manner of Blanmar win Same (2000 50 4 776.3 53 111 N 75 Warmen win Same (2000 50 4 776.3 53 111 N 75 Warmen win Same (2000 50 6 776.3 53 111 N 75 Warmen win Same (2000 50 7	3	,		:					*	*	Effect of Market	37860	\$	•	*	5	336	18 F	
Warmer die Rud Hölle 92000 50 6 755.4 115 247 572 M 74 Fechichidage als Barmers 13000 35 4 64.9 < RAFFW 69 Ramer of Run Hanner 16000 50 4 78.2 53 115 M 92 Warmer wire Manner of	₽.	Tochmology o've Out. Cyn.	2000	Ħ	•	**	· WHEN .	WHEN.	2	E	1								
Manufacture (Manufacture 18000 55 4 66.3 28 61 122 M 92	* ‡		50,5	97		:	***	,	×.	Z 1	Technology arts Barranca	3000	×	**	3	****	\$	5	
Burmins wio Sand Charlos 1800 55 4 64.3 28 61 122 N 92 minimum market and mass 1800 51 100 100 100 100 100 100 100 100 10				ę	•		=	Š		E #	W. and M. Company	4	5	•	;	\$	Ξ	21.1	
	47	Barranca w/o Sand Canvos	13000	×		<b>1</b>	*	3	. X	닿		<u> </u>	ŧ	,	<u>.</u>	1	•		

SOURCE: LSA Associates, Inc., 9/99

County of Orange

Table 8.3-9 (Cont.)

CCTM2.4	CCTM1.8 MCAS El Tees		Ž	Prepar Am	THE MARKET	All Septem		٦	ي	CCTM3.8 MCAS EL Tere		New.	Lotation An	Orstaline Jake	· All Department	Ž.	-	
			Š	ž	Spe. For CNEL A 2015 A	e d		4 E				ş	ĭ	Spd. wit subsciences. Collect of	t i	# 2 2 5	\$ 13 \$ 13 \$ 13 \$ 13 \$ 13 \$ 13 \$ 13 \$ 13	
ě	Cristins Separat News	VOV	Man	2,100	THE STATE OF	(FL)		Т		Contract August Acres	104	(Name)	Lann	(QBV)	<u>a</u> :	3	2	4BA
Ξ	Alben efe Lagana Canyon	73000	33	*	77.4	3	3	361	2 ==		*	2	•	1	ŝ	=	3	
<u> </u>	Manport and Edinger	88 	2	•	Ì	7	<b>∓</b>			Section 14 and 16 in	5	2	•	1	7	3	•	
<u>ē</u>	Birch tre South Britain	2006	\$	7	3	*	×		Ä		<u> </u>	3	•	3	2	ξ	2	
22	Tunto Reach sty Water	2000	25	• •	* ;	2 %	<u>=</u> 7	59										
ñ			;		!	:	ŝ		Ä	.i Learbonn ario Portois	41000	*	•	, ¥	×	138	\$33	
E M	Jeffrey súa Trabuca	2	3	<b>4</b> 5	7	5	*	\$		•		į		-				
: =		****						-				2 5	•	7	R:	£ :	<b>.</b>	
1									•	Managements and Martinarian	36	7, 5	• •	n s k ž	£ 2	i F	? =	
ě								. =		5	31800	2	. •	Ē	3	3 5	3 8	
6								_	•	-	000M	2		7	2	Z	ā	
											9004	₩:	*	3	2.	3	25	
1 2	Rockfeld of Alva	•	ş	•	*	•	-	-	5 5 1	F Jacobina ela Altan	2000	÷	~	į	\$	=	=	
2		•	ŧ		•	•	•			5 Book Mold with Bake	1000	\$	-	3	~	2	E	
2	Alum No Muidends	2000	*	•	Ē	\$	330	2	×					_				
2		·							÷	Control Comment and And Control	1600	<b>Z</b> 2	•	F.		<u>\$</u> :	ž:	
ž									: <del>4</del>	Catola of Milesnian	13000	: #	•					
ê								_	¥	7 Postoble arto E. Centural Park	34000	â	*			2	8	
8 5								_,,	# 1	A Trying on't Resourch	90067	2:	•			=	Ħ	
\$								- 4	# ¥	STATES AND W. CAMERA PLAN	900	= :	•			# 3	# :	
<u>=</u>								. pří	· ¥	further m/o E. Central Park	- B	3 22	• •				2 2	
≆ :									¥	2 Bryste w/o Sand Canyon	900	2	-			9	Ξ	
8									¥.	9 Boyano arto Samel Campton	12008	*	*			3	8	
2									6 A	A Digital At Military	8 5	<b>\$</b> \$	•			<b>2</b>	<b>2</b> :	
ž								. 44	· ¥	W. Culture and Tradesis	2	* 32				2 2	÷ ÷	
Ē								_	¥	7 E. Culture ato Millionnium	999	*	•			N. Partie	: 3	
<b>8</b> 8									# \$ Z >	C. Cuitages also Tradence	<u> </u>	×:	* •		· RHW	<b>#</b> \$	2.3	
Ξ	living ale Perimeter Rd	47000	×	•	¥.	x	=	<u> </u>	. ž			2	•			E	<u>.</u>	
8 9	A STATE OF THE PARTY OF THE PAR		;	•	ž	The state of the	A BLARMAN		• •	105 Arts Village of E. Cuthere	900	2	-	¥.	* BLAHFW	7	đ	
3 2	TO CAME AND THE CASE OF THE PARTY OF THE PAR	•	2	•	7	* TOTAL V	X MCM	7	2 3	2 Touchures with CVF Ears?	į	;	,	ŧ	:	:	;	
=	Perimeter Rd sto Union	9	22	~	*	a	•	•				2		1	2	ž	3	
3 2	Principle 24 April 14 April 20		:	,	•	•	•			Trabaco ele Resencia	36000	2	*	11.1	\$	9	£	
¥		,	•	•	•	•	•	. <del></del>		4 Impleco e/o W. Croost Park	4600	2	-	¥	÷	=	7.	
53	Perketoter Rd ato Markoe Way	-	2	~	-	٥	ф	=				!		!	:	•		
3	Perference Rd ero Air Como		2	~	•	G	-	- <b>-</b>		Trabuce w/e Milkowians	\$ \$	=	~	ī	\$	<b>=</b>	ñ	
5 3			; ;			. ,			\$	Trabuco sée Millemeisen	47900	ñ	~	2	×	*	ñ	
è	TURNATURE AND SOUTH	»	Q	~	•	<b>.</b>	-	-:		d Toloman C. Committee	NAME OF THE OWNER, OWNER, OWNE	=		ì	;	:	;	
5	West Actes to Irvine	•	×	7	•	o	•			•		2		į	2	<b>.</b>	Ę.	
<b>3</b> 3	East Access who Irvine	6	=	*	•	•	-			# Trabaco afo E Central Park	20006	2	-	70.7	ş	2	#.	
8			;	•		•		•		Trabucto w/o Alton	43000	2	•	ž,	~	8	ž	
2 :	West Access wie lived	•	*	*	•	•	=	*							:			
2 =	Em Arrest no Irvine	•	*	•	•	0	21			6 Apprehis Course Paris	8	2	*	*	< PLAFFIX	*	×	
= :										Counting of Research	7000	*	*	67	· RAFE	2	2	
2 2	ETC Consciou (M. M. S.)		2	•		٥	<b>9</b>	<u></u>	2 2 2 2	2 Martine after Sund Carn con	23,000	=	4	4	=	2	€	
9	ETC Connector (N)	٥	22	•	۰		•	-	-				,		;	:	:	
2 2								** **	**	Marine c/o Menumeth	21000	<b>#</b> :	••	3	<b>:</b>	£;	Ē ;	
3											a de la	2 =			2 4 1	2 :	2 4	
346										. –	8	: 2			252	2 4	? =	
2:									7	3 Same Caryon ato Pomola	200	3	-		11	1	·	
# #									æ :	A Same Caryon are irritre	23000	3	•	Z	I	303	3	
2										Charles and invok		2 2	•	3	22.0	<b>3</b> 3	= 8	
=									Ş		17000	: =	•	3	1	2	3 5	
ı																		

ınge	
for a	
ntyo	
$Co\pi$	
	0000
	2

SOURCE: LSA Associates, Inc., 9/99

**EIR No. 573** 

Traffic Noise Model Results Year 2020 with ETRPA Nonaviation Plan vs. Year 2020 No Project - All Segments

Table 8.3-9 (Cont.)

CTM2.8 MCA5 EL Tero		No	Present Alt	armenye 2010	· All Segme	cast#		1 ICC	M28 MCAS El Toro		Nex	Avistina A	terantive III	a · VII galim	ginds		I
		Sp4.	**	CNEL at	7948 CNEL	65 dB CNEL	64 48 CNEL	1   "		Ī	Sp4.	# ed	CNEL #	CNEL.	45 48 CNEL	64 43 CNEL	CMES.
ID # Existing Segment Names	AOT	(wph)	Lenes		(FL)	(ft)	(FL)	ا ا	Existing Segment Names	ADT	(mak)	Lene	(CA)	(FL)	(FL)	(Ft.)	dBA
ID # Existing Segment Names				199001	126.7				T Remarch s'o Torbuco	4000	33		1 63.3	< 1.45 W	32	111	1
523	1							N S	3 Research n/o Marine	15000	35	4	65.3	25	34	116	
524	1							N s	4 Resourch six Marine	13000	35	4	64.9	< <b>乳血</b> 砂	49	106	
525	1			1				N   3	5 W. Course Park are Portels.	10000	35	4	63.7	< RAHW	41	#5	
526	1			1				N 5	6 W. Control Park ale Invins	13000	35	4	64.3	< \$.45°W	49	106	
527	1			1				IN S	7 W. Control Park ale Bryan	9000	35	4	61.3	< 8.0HW	39	#3	ll
528	(			1				IN S	W. Cantral Park ale Trabuco	13000	35	4	64.9	< 14W	49	106	II
319	1			1				N s	9 W. Control Park ato Trabaco	10000	35	4	63.7	< 2.05W	41	34	II
530	1			1				N S	6 W. Control Park ale Quantum	11000	35	4	64.1	< RATH	64	94	II
531	1			1				N S	1 Milliongian a/o trvine	23000	55	4	67.3	33	73	153	II
532				1				N S	2 Millennium ale Brynn	20800	35	4	64.7	36	63	140	ll
533	1							N S	in felificaminus ato firyun	17000	33	4	66	27	54	126	II
534	1			1				lw s	Milmaine als Arts Village	13000	35	4	64.9	< Retw	49	184	ll .
535	1			1				N 3	5 Millenning a/o Tenhoco	24000	33	4	47.5	14	73	156	II
536								IN 5	6 Milanaiam ale Trabuco	30000	33	4	48.5	40	16	184	II
537	1							N S	7 Millennium 2/s Marine	34000	35	4	49	43	92	(99	ll .
538	1			1				N S	Additional on the Jaconstone	44000	35	4	76.3	52	111	239	ll .
539	1			1				N   3	9 Millennium ale Burenca	44008	13	4	79.2	52	114	239	Į
340	1			1				N s	6 Millendore pie Alten	33008	35	4	68.9	42	91	196	
541	l			1				N s	I Millennium nio Rockfield	44000	15	4	76.2	52	lu	239	
542	l			l				INI 3	13 Millionalism são Buke	36000	35	4	693	45	97	204	ll
545	- 1							N 3	II E. Control Park nie frying	4000	35	4	59.3	< Rativ	< RAHW	48	11
544	i							N I	14 E. Control Park nio Sevine	10000	55	4	43.7	< RAHW	41	11.8	ll .
545	l			1					Li E. Central Park não Trabuso	3000	35	4	58.5	< RAHW	< RAFW	40	II
<del>34</del>	1			1				1 1	4 E Concret Park e/o Trabuco	3000	35	4	58.5	< RAHW	< 24fW	40	II
Si	1			1					17 E. Control Park n/o Astor	3000	35	i	58.5	< RAHW	< Rabby	40	II
34	1			1					6 Altes n/o Trabuco	55000	35	i	71.1	59	128	275	II
603	ı			1					3 Lagues Cyn ale 1-405	45000	35	i	79.7	56	120	258	II

EIR No. 573 County of Orange Table 8.3-9 (Cont.)

SOURCE: LSA Associates, Inc., 9/99

Number of Existing Segments related to the Noise Level Increase

Scenarios	Existing No Project	<b>Existing NonAviation Pla</b>	2020 No Project	2020 NonAviation Plan
1.5<#<3dB	•	12	51	58
3 <#< 6dB	-	12	95	92
6 <#< 9dB	-	3	15	12
9 <#< 12dB	-	0	5	6
# > 12 dB		0	0	

EIR No. 573 County of Orange Table 8.3-10

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results Comparison Number of Segments with Traffic Noise Increase Over Existing No Project Condition Number of Segments related to the Noise Level Increase

Scenario	2020 NoPj vs 2020 NonAv
(-3)<#<0dB	41
# = 0 dB	91
0 <#< 3dB	60
3 < # < 6dB	1
6 < # < 9dB	0
9 <#< 12dB	1
# > 12 dB	0

EIR No. 573	County of Orange	<b>Table 8.3-11</b>

SOURCE: LSA Associates, Inc., 9/99

Traffic Noise Model Results Comparison Year 2020 ETRPA Nonaviation Plan vs. Year 2020 No Project -Number of Segments with Traffic Noise Increase

# **Comparison to Existing Conditions**

A comparison of the impacts of the Existing plus ETRPA Nonaviation Plan Alternative and Existing plus Proposed Project is as follows.

## Impacted Roadway Links

The following roadway links are impacted by both the Proposed Project and the ETRPA Nonaviation Plan Alternative with 1.5 dB or higher traffic level increases over the existing conditions:

- Irvine w/o Sand Canyon
- Irvine e/o Sand Canyon
- Irvine w/o Alton
- Irvine w/o Bake
- Jeffrey n/o Bryan
- Portola w/o Jamboree
- Sand Canyon n/o I-5
- Sand Canyon n/o Irvine Center
- Sand Canyon s/o Irvine Center
- Sand Canyon n/o Alton
- Sand Canyon n/o I-405
- Trabuco w/o Jeffrey
- Trabuco w/o Sand Canyon

The following roadway links are impacted by the ETRPA Nonaviation Plan Alternative only with 1.5 dB or higher traffic noise level increases over the existing conditions:

- Alton s/o Irvine
- Bake s/o I-5
- Barranca e/o Technology
- Barranca w/o Alton
- Irvine e/o Culver
- Irvine w/o Jeffrey
- Jeronimo e/o Alton
- Rockfield w/o Lake Forest
- Sand Canyon n/o Trabuco
- Technology n/o Barranca
- Toledo e/o Alton
- Toledo w/o Lake Forest

- Trabuco e/o Bake
- Trabuco w/o Lake Forest

The following roadway links are impacted by the Proposed Project only with 1.5 dB or higher traffic noise level increases over the existing conditions:

- Irvine e/o ETC East Leg
- Jeffrey n/o Trabuco
- Jeffrey n/o I-5
- Sand Canyon s/o I-5

The same four areas of residential development that may be significantly impacted by traffic noise impacts associated with the Proposed Project would be significantly impacted by the ETRPA Nonaviation Plan Alternative as well, as discussed in Section 4.4, Noise.

# **Comparison to Existing Plus Committed Conditions**

A comparison of the impacts of the year 2020 ETRPA Nonaviation Plan Alternative and year 2020 Proposed Project is as follows:

#### Impacted Roadway Links

The following roadway links are impacted by the Proposed Project and the ETRPA Nonaviation Plan Alternative with 1.5 dB or higher traffic noise level increases over the 2020 No Project conditions:

- Irvine e/o ETC East Leg
- Irvine w/o Alton

The following roadway links are impacted by the ETRPA Nonaviation Plan Alternative only with 1.5 dB or higher traffic noise level increases over the 2020 No Project conditions:

- Alton n/o Commercentre
- Bake s/o I-5
- Commercentre w/o Bake
- Rancho e/o Alton
- Trabuco w/o Sand Canyon

The following roadway links are impacted by the Proposed Project only with 1.5 dB or higher traffic noise level increases over the 2020 No Project conditions:

- ETC East Leg s/o Irvine
- SR-133 s/o I-5
- SR-133 n/o I-405
- Trabuco e/o Sand Canyon

# 8.3.5.5 Air Quality

The air quality impacts of the ETRPA Nonaviation Plan Alternative were identified by analyzing the short-term impacts (construction), regional air quality impacts (total air pollutants emissions), local air quality impacts due to traffic carbon monoxide, and local impacts due to aircraft and associated operations in comparison to the Proposed Project's impacts. As summarized below, the ETRPA Alternative would result in additional significant regional air quality impacts that would be greater than the Proposed Project under all development scenarios due to Orange County generated demand being serviced at other regional airports outside of the County similar to the No Project/No Activity Alternative (Alternative E). This alternative would also result in significant local CO hot spot air quality impacts associated with vehicle emissions not identified under the Proposed Project. In addition, it is likely that construction emissions would be significant and would be greater than under the Proposed Project due to the proposed greater intensity of use at the MCAS El Toro site. This alternative, however, would avoid the significant local air quality impacts of the Proposed Project resulting from aircraft emissions at OCX and JWA.

# Short-Term Air Quality Impacts (Construction)

Under this alternative, no significant runway improvements would be made at JWA. MCAS El Toro, however, would be developed with nonaviation uses in greater intensity and density than under the Proposed Project. Therefore, total construction emissions are anticipated to be greater than those of the Proposed Project in all phasing years due to higher density or intensity land uses being proposed at the MCAS El Toro site.

# Operational Air Quality Impacts

#### **Emissions Inventories**

Under this project alternative, JWA will serve 8.4 MAP in Phase 4. No aviation reuse at MCAS El Toro would occur as the site would be developed based on the nonaviation land uses proposed in the ETRPA Plan. Although there would be no emissions associated with aviation uses at MCAS El Toro, there would be emissions associated with energy consumption and vehicular trips by the nonaviation uses. Project build out air pollutant emissions associated with airport operations at JWA and nonaviation land uses at OCX are shown below in Table 8.3-12 for this alternative.

Table 8.3-12
Phase 4 ETRPA Nonaviation Plan Alternative –
Project Direct Air Pollutant Emissions (Pounds/Day)

William Willia	CO.	NO <sub>E</sub>	Roc1	SO <sub>X</sub>	PM10
Aircraft	7,061.00	3,025.85	402.78	239.64	44.48
MCAS El Toro	0.00	0.00	0.00	0.00	0.00
JWA	7,061.00	3,025.85	402.78	239.64	44.48
GSE/APU	5,610.84	597.89	171.83	14.93	26.54
MCAS El Toro	0.00	0.00	0.00	0.00	0.00
JWA	5,610.84	597.89	171.83	14.93	26.54
Fuel Storage/Dispensing	0.00	0.00	9.14	0.00	0.00
MCAS El Toro	0.00	0.00	0.00	0.00	0.00
JWA			9.14		
Airport Roadways	117.92	13.70	3.99	0.56	1.17
MCAS El Toro	0.00	0.00	0.00	0.00	0.00
JWA	117.92	13.70	3.99	0.56	1.17
Airport Parking	96.38	7.40	9.98	3.05	0.28
MCAS El Toro	0.00	0.00	0.00	0.00	0.00
JWA	96.38	7.40	9.98	3.05	0.28
Energy Consumption	161.20	500.70	313.70	18.70	6.80
MCAS El Toro	129.60	318.50	312.00	NA	0.60
JWA	31.60	182.20	1.70	18.70	6.20
Vehicular Traffic	20,714	7,186	1,964	605	3,511
MCAS El Toro	16,145	5,338	1,605	493	2,565
JWA	4,569	1,848	359	112	946
Total (lbs/day)	33,761	11,332	2,875	882	3,590

Source: CH2M Hill and LSA Associates, Inc., 2001

Total project direct emissions under this alternative would be significantly higher than the emissions under the Existing Conditions (1998). The increase in emissions for each criteria pollutant exceeds the applicable SCAQMD threshold. Accordingly, the ETRPA Alternative would result in significant operational air quality impacts.

Regional air pollutant emissions, including airport operations at other airports in the region and VMT required for air travel passengers to get to these airports, would be similar to those shown in Tables 8.2-3A and 8.2-3B for the No Project/No Activity Alternative. When compared to the regional air quality emissions associated with the Proposed Project, this project alternative would have higher regional CO, NO<sub>x</sub>, SO<sub>x</sub>, ROC, and PM<sub>10</sub> emissions than the Proposed Project. Because of the conclusions reached in connection with the No Project/No Activity Alternative, this would likely be true in all phasing years under the ETRPA Alternative, as well. As discussed in connection with the No Project/No Activity scenario, the failure to provide sufficient airport capacity in Orange County to meet the

ROC emissions obtained by multiplying HC emissions reported by EDMS by a factor of 1.14.

locally generated demand will result in increased VMT and increased aircraft emissions as a result of longer taxi times and LTO cycle time as average delay time at these regional airports increases.

# **Dispersion Analysis**

No airport emissions dispersion analysis was conducted for this project alternative. However, because local criteria pollutant hot spots were found under the No Project/No Activity Alternative, which has the same annual aircraft LTO operations at JWA, it is expected that criteria pollutant hot spots from aircraft exhaust emissions would also occur under this alternative. This alternative, however, would avoid the significant and unavoidable local air quality impacts at OCX and JWA due to aircraft and associated operations.

With regard to vehicle emissions at intersections in the vicinity of the project sites, the CAL3QHC model was used to assess CO concentrations. Tables 8.3-13 through 8.3-16 show the one-hour and eight-hour CO concentrations under the Existing Conditions (1998) plus ETRPA Nonaviation Plan scenario. Because of the CAL3QHC modeling selection criteria, not all intersections modeled for the Existing Conditions (1998) scenario were modeled for the Existing Conditions plus ETRPA Nonaviation Plan. However, for those intersections that were modeled under both scenarios, the increase in CO concentration would be 0.6 ppm or smaller for the one-hour CO concentrations, which is less than the 1 ppm threshold established by the SCAQMD. However, some of the increases in CO concentrations would exceed the thresholds of significant changes (0.45 ppm) for the eighthour CO concentration. Based on this analysis, the ETRPA Nonaviation Plan Alternative would result in significant local air quality impacts related to vehicle CO hot spot concentrations. In comparison to the Proposed Project, the ETRPA Alternative would result in exceedances of the eight-hour CO standard not present under the Proposed Project. Therefore, the ETRPA Alternative would result in significant local air quality impacts not identified under the Proposed Project.

#### Toxic Air Contaminants

Under this alternative, although there would be no runway improvements at JWA and there would be no aviation reuse of MCAS El Toro, MCAS El Toro would be developed with nonaviation uses that are higher in density and intensity than are proposed under the project (Alternative B). Therefore, although it is difficult to predict, it is likely that this alternative would result in toxic air contaminant impacts that would be significant.

Table 8.3-13

Year 1998 Existing Conditions Plus ETRPA Nonaviation Plan - Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

DITA	INTERSECTING STREETS:	BACI'	RECY	RECY.	, wee,	RECY.	BECS	* RECT	RECE	es RECY	RECIP	RECIT	REC12
***************************************	CITY OF TUSTIN <sup>IS</sup>												
26	Jamboree & Irvine	13.1	13.0	13.0	13.5	12.8	12.7	12.6	12.9	12.6	12.7	12.6	12.7
	CITY OF IRVINE"												
238	Bake & Irvine/Trabuco	8.6	9,0	8.4	8.4	8.0	8.6	8.0	7.9	8.1	8.2	8,2	8.6
268	Bake & Rockfield	8.2	8.2	8.4	8.3	8.0	8,1	8, 1	8.1	8.0	8.3	7,8	8,1
175	Jamboree & Michelson	8,6	8.4	8,5	8,2	7.9	8.2	7.9	8.0	8.0	8,2	8.0	8.3
156	Jamboree & Main	8.4	8.4	8.4	8.2	8.0	8,1	7.9	7.9	7.9	8, [	7.9	8.3
237	Alton & Irvine	8.3	8.7	8.2	8.1	7.9	8.1	8.1	8.3	7.7	7.7	8.0	8.2
68	Sand Canyon & Trabuco	8.3	8.0	8.6	8.5	8.1	7.9	7.8	8.3	7.7	8.1	7.7	7.8
320	Trabuco & Irvine	8.1	8.1	8.2	7.9	7.8	7.7	7.8	8.1	7.7	7.5	7.7	7.9
338	Millennium & Alton	8.3	8.3	8.2	8.3	8.0	8.1	7.9	8.2	7.9	8 2	7.9	8.0
151	Red Hill & MacArthur	8.3	8.1	8.0	8.3	7.8	7.9	7.8	8.0	7.9	7.9	7.9	8.0
319	E. Central Park & Irvine	7.8	8.3	7.9	7.8	7.6	7.9	7.7	7.7	7.4	7.3	7.6	7.7
130	Technology & Barranca	8.0	7.9	7.9	8.2	7.5	7.3	7.6	7.7	7.6	7.7	7.6	7.7
317	W. Central Park & Irvine	8.0	8.1	7.9	8.3	7.9	8.0	7.6	7.9	7.7	8.0	7.6	7.7
316	Research & Irvine	8.1	8.2	8.0	8.2	7.7	7.7	7.6	7.8	7.8	7,9	7.6	7.7
318	Millennium & Irvine	7.9	8.2	7.9	8.0	7.7	7.7	7.7	7.8	7.4	7.4	7.8	7.9
337	Millennium & Barranca	8.5	8.0	8.1	8.3	7.9	7.8	7.9	8.0	8.1	8.2	7.9	8.2
32	Sand Canyon & Irvine	8.3	8.2	8.2	<b>8</b> . 1	8.1	8.3	7.9	8.1	7.6	7.6	79	7.8
	CITY OF LAGUNA BILLS"												
280	El Toro & Avd. Carlota	8.4	7.9	8.2	8.1	7.6	7.9	7.5	7.7	7.8	7.9	7.8	7.9
	CITY OF LAKE FOREST"												
269	Lake Forest & Rockfield	8.5	8.6	8.5	8.5	8.0	8.6	79	8.0	7.9	8 2	8.0	8.5
	CITY OF MISSION VIEJOH												
265	Alicia & Muirlands	8.4	8.3	8.3	8.2	7.8	8.0	7.8	8.0	7.8	8.0	7.9	8, 1

Note: \* - Concentrations are in parts per million (ppm); federal one-hour CO standard is 35 ppm; State one-hour CO standard is 20 ppm.

<sup>1-</sup>REC1 SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> RECS S. DEPARTURE - MID BLOCK

<sup>6-</sup>REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> RECU W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient one-hour CO concentration, 12.0 ppm, the second highest one-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1993 and 1997, is added to the calculated one-hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 7.0 ppm, the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1993 and 1997, is added to the calculated one-hour levels.

Table 8.3-14 Year 1998 Existing Conditions Plus ETRPA Nonaviation Plan - Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

DYT#	INTERSECTING STREETS	REC1	REC2	BEC?	· REC#	Dr. et	RECS	RECE	RECT'	RECT	RECS	RECIO"	RECI!"	REC113 <sup>th</sup>
100000	CITY OF TUSTIN <sup>D</sup>		Sec. 10000 (111)	1 404 525 X-H-2-11	orden moderation (1)	ST900 102 1	8 74 × 5 5	2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	25 25 25 25 26 AC 1			7 16.2 EOE 31	01 01002 TWY-P 18.00	as/ar/
26	Jamboree & Irvine	8.8	8.7	8.7	9.1	14	8.6	8.5	8.4	8,6	8.4	8,5	8.4	8.5
	CTTY OF IRVINE"													
238	Bake & Irvine/Trabuco	5.2	5.5	5.1	4.8		5.2	4.8	47	4.9	4.9	4.9	4.9	5.2
268	Bake & Rockfield	5.2	5.5	5.1	4.8		5 2	4.8	4.7	4.9	4.9	4.9	4.9	4.9
175	Jamboree & Michelson	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4,9	4.9	4.9	5.0
156	Jamboree & Main	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	49	5.0
237	Alton & Irvine	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4.9
68	Sand Canyon & Trabuco	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4,7
320	Trabuco & Irvine	5.2	5.5	5.L	4.8		5.2	4.8	4.7	4.9	4,9	4.9	4.9	4.7
338	Millennium & Alton	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4,9	4.9	4.9	4.8
151	Red Hill & MacAnhur	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4.8
319	E. Central Park & Irvine	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4.6
130	Technology & Barranca	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4.6
317	W. Central Park & Irvine	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4,9	4.9	4.9	4.9	4.6
316	Research & Irvine	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4.6
318	Millennium & Irvine	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4,9	4.9	4.9	4.7
337	Millennium & Barranca	5.2	5.5	5.1	4.8		5.2	4.8	4.7	4.9	4.9	4.9	4.9	4,9
32	Sand Canyon & Irvine	5.2	5.5	5. t	4.8		5.2	4.8	4.7	4.9	4,9	4.9	4.9	4.7
	CITY OF LAGUNA HILLS													
280	El Toro & Avd. Carlota	5.2	5.5	5.1	5.1		4.8	5.2	4.8	4.7	4.9	4.9	4.9	4.7
	CITY OF LAKE FOREST"													
269	Lake Forest & Rockfield	5.2	5.5	5.1	5.1		4.8	5.2	4.8	4.7	4.9	4,9	4.9	5.2
	CITY OF MISSION VIETO"													
265	Alicia & Muirlands	5.2	5.5	<b>5</b> .1	5.1		4.8	5.2	4.8	4.7	4.9	4.9	4.9	4,9
Note.	* - Concentrations are in parts p	er million (opn	n): federal eight.	hour CO sta	ndard is 9 n	om.								

Concentrations are in parts per million (ppm); federal eight-hour CO standard is 9 ppm.

<sup>1 -</sup> RECI SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> REC8 W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK 11 - REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient eight-hour CO concentration, 8.0 ppm, the second highest eight-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1993 and 1997, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

<sup>14 -</sup> Number in bold represents exceedance of the standards.

<sup>15 -</sup> The ambient eight-hour CO concentration, 4.1 ppm, the second highest eight-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1993 and 1997, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

Table 8.3-15

Phase 4 ETRPA Non-Aviation Plan – Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

NTA	INTERSECTING STREETS	RECI	PECS.	RECO	RECE	. RECT	RECE	RECT	SACA	* BECT	RECIO	- RECU!	BEC11
	CTTY OF SANTA ANAD								274 146 33 14				
54	MacArthur & Main	7.0	7.2	7.1	7.1	6.8	6.8	6.6	7.1	6.8	7.0	6.7	6.8
52	Main & Sunflower	6.9	7.1	6.7	7.0	6.6	6.9	6.5	6.5	6.7	6.8	7.0	6.8
	CITY OF TUSTIN"												
3	Newport & Edinger	7.2	7.0	6.8	7.2	6.7	6.6	6.6	6.7	6.7	6.9	6.6	6.9
	CITY OF IRVINE"												
16	Jamboree & Barranca	5.8	5.9	5.6	5.6	5.3	5.5	5.3	5.5	5.5	5.4	5.5	5.7
56	Jamboree & Main	5.7	5.6	5.6	5.4	5.2	5.4	5.3	5.4	5.4	5.3	5.4	5.6
8	Culver & Irvine Center	5.6	5.5	5.7	\$.6	5.3	5.4	5.3	5.5	5.2	5.5	5.4	5.5
34	Jamboree & Alton	5.6	5.6	5.8	5.6	5.1	5.5	5.2	5.4	5.3	5.4	5.4	5.6
75	Jamboree & Michelson	5.8	5.4	5.4	5.5	5.2	5.3	5.3	5.3	5.4	5.4	5.5	5,4
100	Jeffrey & Irvine Center	5.5	5.6	5.6	\$.7	5.3	5.5	5.2	5.3	5.3	5.6	5.2	5.3
151	Red Hill & MacArthur	5.7	5.4	5.5	5.6	5.0	5.4	5.4	5.3	5.3	5.4	5.3	5.7
155	Von Karman & Main	5.5	5.6	5.8	5.5	5.2	5.6	5.3	5,4	5.1	5.4	5.4	5.4
16	Research & Irvine	5.2	5.2	5.2	5.3	5.0	5.0	5.0	5.1	5.0	5.3	5,0	5.1
58	Sand Canyon & Trabuco	5.2	5.3	5.3	5.2	5.1	5.0	5.1	5.3	5.0	5.1	5.1	4.9
1	Jeffrey & Irvine	5.3	5.5	5.4	5.5	5.2	5.3	5.1	5,2	4.9	5.2	5.0	5.1
53	Red Hill & Main	5.4	5.5	5.5	5,5	5.1	5,3	5.1	5.2	5.2	5.2	5.1	5.3
120	Trabuco & Irvine	5.4	5.5	5.6	5.5	5.2	5.3	5.2	5.3	5.1	5,3	5,2	5.3
	CITY OF LAGUNA BEACH"												
199	Moulton & El Toro	5.6	5.4	5.5	\$,5	5.1	5,3	5,4	5.5	5.0	5.3	5.1	5.1
	CITY OF LAGUNA HILLS"												
280	El Toro & Avd. Cariota	5.5	5,3	\$.5	5.4	5.0	5.3	5.1	5.4	5.3	5.2	5.1	5,3
	CITY OF LAKE FOREST"												
271	El Toro & Rockfield	5.5	5.4	5.6	5.5	5.2	5.3	5.3	5.5	5.2	5.4	5.0	5.3
	CITY OF SAN JUAN CAPISTRA	NO"											
287	Laguna Hills & Paseo Valencia	5.1	5.5	5.2	5.1	5.0	5.2	4.9	5.0	4.8	4.9	5.1	5,0
Note:	- Concentrations are in parts per m - REC1 SW CORNER - REC2 SE CORNER - REC3 NE CORNER - REC4 NW CORNER - REC5 S. DEPARTURE - MID I - REC6 N. APPROACH - MID B	BLOCK	)										

<sup>8 -</sup> RECS W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient one-hour CO concentration, 6.1 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Vailey Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

Table 8.3-16

Phase 4 ETRPA Non-Aviation Plan -- Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

MIN	INTERSECTING STREETS	REC!	HEC	RECA	RECT	· PLCS	HCF	J RECT!	RECE	REC.	REC10"	, KALII	RECL
***************************************	CITY OF SANTA ANA"									<u> </u>			
54	MacArthur & Main	5.2	5.4	5.3	5.3	5.1	5.1	5.0	5.3	5.i	5.2	5.0	5.1
52	Main & Sunflower	5.2	5.3	5,0	<b>S.2</b>	5.0	5.2	4.9	4.9	5.0	5.1	5.2	5.1
	CITY OF TUSTIN												
3	Newport & Edinger	5.4	5.2	5.1	5,4	5 0	5.0	5.0	5.0	5,0	5.2	5.0	5.2
	CITY OF IRVINE"												
16	Jamboree & Barranca	3.7	3.8	3.6	3.6	3.4	3.5	3.4	3.5	3.5	3.5	3.5	3.7
66	Jamboree & Main	3.7	3.6	3.6	3.5	3.3	3.5	3.4	3.5	3.5	3.4	3.5	3.6
:	Culver & Irvine Center	3.6	3.5	3.7	3.6	3.4	3,5	3,4	3.5	3.3	3.5	3.5	3.5
4	Jamboree & Alton	3.6	3,6	3.7	3.6	3.3	3.5	3.3	3.5	3.4	3.5	3,5	36
15	Jamboree & Michelson	3.7	3.5	3.5	3.5	3.3	3.4	3.4	3.4	3.5	3.5	3.5	3.5
Ю	Jeffrey & Irvine Center	3.5	3.6	3.6	3.7	3.4	3.5	3.3	3.4	3.4	36	3.3	3.4
i	Red Hill & MscArthur	3.7	3.5	3.5	3.6	3.2	3.5	3.5	3 4	3.4	3.5	3.4	3.7
5	Von Karman & Main	3.5	3,6	3.7	3.5	3.3	3.6	3.4	3.5	3.3	3.5	3.5	3.5
6	Research & Irvine	3.3	3.3	3.3	3.4	3.2	3.2	3.2	3.3	3.2	3.4	3.2	3.3
	Sand Canyon & Trabuco	3.3	3.4	3.4	3.3	3.3	3.2	3.3	3.4	3.2	3.3	3.3	3.1
	Jeffrey & Irvine	3.4	3.5	3.5	3.5	3.3	3.4	3.3	3.3	3.1	3.3	3,2	3.3
3	Red Hill & Main	3.5	3.5	3.5	3.5	3,3	3.4	3.3	3,3	3.3	3,3	3.3	3.4
0	Trabuco & Irvine	3.5	3.5	3,6	3.5	3.3	3.4	3,3	3.4	3.3	3.4	3.3	3,4
	CITY OF LAGUNA BEACH												
19	Moulton & El Toro	3,6	3.5	3.5	3.5	3.3	3.4	3.5	3.5	3.2	3.4	3.3	3.3
	CITY OF LAGUNA HILLS"												
<b>3</b> 0	El Toro & Avd. Carlota	3.5	3.4	3.5	3.5	3.2	3.4	3.3	3.3	3.4	3.3	3.3	3.4
	CITY OF LAKE FOREST"												
<b>'</b> 1	El Toro & Rockfield	3,5	3.5	3.6	3,5	3,3	3,4	3.4	3.5	3.3	3.5	3,2	3.4
	CITY OF SAN JUAN CAPISTI	RANO <sup>14</sup>											
17	Laguna Hills & Paseo Valencia	3.3	3,5	3.3	3.3	3.2	3.3	3.1	3.2	3.0	3.1	3.3	3.2
Note:	* - Concentrations are in parts per		,	<i>3.3</i>	2,3	J.#	3.3	₩. ₹	₹.#	3,6	J.1	3.3	

- I RECI SW CORNER
- 2 REC2 SE CORNER
- 3 REC3 NE CORNER
- 4 REC4 NW CORNER
- 5 REC5 S. DEPARTURE MID BLOCK
- 6 REC6 N. APPROACH MID BLOCK
- 7 REC7 E. DEPARTURE MID BLOCK
- 8 REC8 W. APPROACH MID BLOCK
- 9 REC9 N. DEPARTURE MID BLOCK
- 10 REC10 S. APPROACH MID BLOCK
- 11 REC11 W. DEPARTURE MID BLOCK
- 12 REC12 E. APPROACH MID BLOCK
- 13 The ambient eight-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.
- 14 The ambient eight-hour CO concentration, 2.9 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

# 8.3.5.6 Topography

The Nonaviation Plan Alternative was prepared by ETRPA at a General Plan level of detail, which is insufficient to determine at this time the topographic effects of this alternative.

Usage of JWA under the ETRPA Nonaviation Plan Alternative would be the same as the current usage, and therefore would not entail potential topographic impacts.

Under the Proposed Project, due to the relatively flat to gently sloping topography, both before and after grading, and the lack of any unusual or unique topographic features on the site, no significant adverse impacts to topography at MCAS El Toro are anticipated. No modifications to the existing topography at JWA are proposed.

# 8.3.5.7 Soils, Geology and Seismicity

Usage of MCAS El Toro under the ETRPA Nonaviation Plan Alternative would entail development of the site for nonaviation uses. Potential geophysical impacts of the ETRPA Nonaviation Plan Alternative project would likely be similar to those of the Proposed Project, but would differ in detail, depending upon the specific types or locations of structures or other features to be constructed.

Usage of JWA under the ETRPA Nonaviation Plan Alternative would be the same as the current usage, and therefore would not raise potential impacts related to soils, geologic features or seismicity. Likewise, under the Proposed Project no significant modifications would be made and, therefore, no impacts are anticipated.

# 8.3.5.8 Hydrology and Water Quality

Under the ETRPA Nonaviation Plan Alternative, the potential impacts at the MCAS El Toro site related to hydrology and water quality would be similar to those identified under the Proposed Project because both alternatives have approximately the same impervious acreage and open space areas. However, under the Proposed Project, design improvements are incorporated into the project that will reduce impacts to a level below significance. It is unclear whether the ETRPA Nonaviation Alternative includes similar design improvements.

Water quality constituents will be different under the ETRPA Nonaviation Plan Alternative as compared to the Proposed Project due to the differences associated with construction, operation, and runoff. Impacts to water quality resulting from construction, operations, and runoff under the Nonaviation Alternative can be mitigated using BMPs and other permit requirements.

Groundwater quality impacts under the ETRPA Nonaviation Plan Alternative would be the same as those discussed under the Proposed Project. No groundwater will be pumped from the MCAS El Toro site, so there will be no impacts to local groundwater levels or basin storage under either alternative.

Under this alternative, JWA would maintain the same volume of passenger traffic and would require no new construction. Therefore, the Nonaviation Alternative would not result in impacts related to hydrology and water quality at the JWA site.

# 8.3.5.9 Biological Resources

The direct impacts of the ETRPA Nonaviation Plan Alternative on biological resources will be limited primarily to the loss of some coastal sage scrub habitat, non-native plant communities, including agricultural habitats, non-native grasslands, disturbed/developed land, and their associated wildlife species. Approximately 758 acres of agricultural land will be impacted under the ETRPA Nonaviation Plan Alternative, compared to approximately 620 acres under the Proposed Project. There will not be any direct impacts to the federal Habitat Reserve.

There is some native Venturan-Diegan sage scrub habitat on the MCAS El Toro site outside the Habitat Reserve. This area occurs on a knoll and appears to be at least partially impacted by the Nonaviation Alternative, as indicated by the residential designation in this portion of the site. There is also a 20 acre parcel, south of Alton Parkway that does contain some coastal sage scrub. The nonaviation alternative does include a park/open space designation at this location. Depending upon the configuration of the uses, there may be some potential coastal sage scrub impacted at this location. These areas include California gnatcatcher habitat. In addition, streambed habitat is also impacted by the Nonaviation Alternative. These streambeds vary in terms of plant species they support. The streambeds include mulefat scrub, willow scrub, cattails, as well as highly disturbed and scoured sandy washes. Specific streambeds impacted include San Diego Creek, Serrano Creek, Borrego Wash, and portions of Agua Chinon Wash. It appears that Agua Chinon is retained north of Irvine Boulevard.

In contrast to the Proposed Project, there will be no potential beneficial impacts from the ETRPA Nonaviation Plan Alternative since the proposed wildlife habitat area along the eastern perimeter of the MCAS El Toro site is not included. In contrast, under the Proposed Project, the addition of native plant communities would potentially provide a wildlife habitat area between large habitat areas in the Coastal and Central Subregional NCCP/HCP Reserve System.

The ETRPA Nonaviation Plan Alternative assumes that JWA continues its existing role at an approved service level of 8.4 MAP, with no facility improvements. Therefore, this alternative would not result in any direct impacts on biological resources at JWA or the Upper Newport Bay.

Biological resources on MCAS El Toro that may experience indirect impacts as a result of the ETRPA Nonaviation Plan Alternative are primarily limited to resources in the federal Habitat Reserve. These potential impacts may occur with construction of the facilities associated with the Residential, Business/Technology, Village and Outdoor Sports districts. These indirect impacts may consist of construction dust, noise, introduction of non-native plants and animals, and increased human presence, similar to the Proposed Project. However, due to the distance of these impacts from the Habitat Reserve, potential indirect impacts on biological resources would be minimized and not considered significant.

#### 8.3.5.10 Public Services and Utilities

The City of Irvine GPA, Zone Change, and Annexation EIR (March, 1999) concluded that, compared to the existing conditions, development of the ETRPA Alternative on the MCAS El Toro site could: 1) create potential short-falls in fire protection services and facilities; 2) may exacerbate overcrowding at area schools; 3) create significant traffic noise levels such that three existing schools would be within 65 dB CNEL noise contours; 4) potentially disrupt domestic water services to adjacent areas; 5) create the need for new sources of water to serve proposed development on site; and 6) require additional sewage treatment capacity and damage existing sewer treatment facilities on site. The City's EIR concluded that, with prescribed mitigation, these impacts would be reduced to below a level of significance.

Similarly, as described in Section 4.10 (Public Services and Utilities), after mitigation, the Proposed Project is not anticipated to result in significant adverse impacts related to utilities.

# 8.3.5.11 Natural Resources and Energy

The ETRPA Alternative would increase the consumption of energy compared to the existing condition. The ultimate build out and development of the ETRPA Nonaviation Alternative will require construction with a greater level of total energy consumption over a 20 year build out period compared with the existing (1998) setting and the Proposed Project. As a mixed-use development, operational energy consumption by the ETRPA project would likely be substantially less than under the existing setting (1998) but substantially more than the Proposed Project. However, both the ETRPA Plan Alternative and the Proposed Project would not result in significant adverse impacts related to energy resources.

Implementation of the ETRPA Nonaviation Alternative would eliminate all of the existing agricultural uses on the MCAS El Toro site. Some areas could remain in agricultural production until such time as development is phased in, or indefinitely, if certain areas are not developed. In comparison, the Proposed Project plans to reserve 139 acres of existing agricultural land. The loss of agricultural land is considered significant for either the ETRPA Alternative or Proposed Project cases; however, the ETRPA Alternative would have an incrementally larger impact on agricultural resources than the Proposed Project. Both the

ETRPA Alternative and Proposed Project would have greater impacts to agricultural resources compared with the existing setting.

#### 8.3.5.12 Aesthetics, Light and Glare

At the MCAS El Toro site, the ETRPA Nonaviation Plan Alternative (ETRPA) would change the existing aesthetics of the site from a military aviation base to a mixed use urban planned community. The visual character of the site would change from an airfield with perpendicular runways and aviation support buildings, military community buildings, military housing, and recreational facilities to a more modern "village" development with business/technology, education, research, entertainment, retail, residential (low to high density), community parks and open space uses, and an outdoor sports complex. Buildings would include single to multi-story structures. The ETRPA Alternative also includes preservation of the natural habitat in the northeastern panhandle area of the site, as does the Proposed Project. A multi-modal transportation system is proposed to include bus, rail, and potential people-mover facilities.

Compared to the existing 1998 setting, development of the MCAS El Toro site with the ETRPA Alternative would visually appear to further intensify the surrounding urban setting, with office/commercial uses in the site vicinity and residential subdivisions in the surrounding hillsides. Specific potential aesthetic impacts of the ETRPA plan discussed in the City of Irvine's GPA/ZC EIR include that new buildings proposed as part of the ETRPA plan may be several stories in height, which would be visible to motorists on adjacent roadways and from residences located west and at higher elevations southeast and northeast of the MCAS El Toro site. New public roadways proposed with the ETRPA plan would provide public views of future development within the site. Potential adverse aesthetic impacts could occur if adjacent structures have highly different architectural styles, massing, or building density (i.e., if residential and industrial structures are proposed near each other).

The ETRPA Alternative would not necessarily represent a significant aesthetic impact compared with the existing setting, in that it would not adversely affect (e.g., obstruct) any scenic vistas or highways. The Proposed Project would, in contrast, maintain the overall visual character of the former military base, with modernization of airport support facilities, and provision of more open space and recreation areas than is currently provided. The ETRPA Alternative would not necessarily substantially degrade the existing visual character or quality of the site or its surroundings; however, it would intensify the urbanization of the site by removing the runways and filling that area with a combination of buildings interlaced with community open areas and landscaping.

The City of Irvine GPA/ZC EIR also determined that new development within the site may create light and glare impacts on adjacent residents. The Proposed Project will also generate light and glare, but at levels that are more similar to the existing setting than the levels of the ETRPA plan. With either the ETRPA plan or the Proposed Project, light and glare will be kept to below the level of significance with implementation of mitigation measures that

would minimize light intrusion and spillover onto adjacent properties, and that would minimize glare from buildings and light sources.

At JWA, the ETRPA Nonaviation Plan Alternative would maintain status quo operations and would not change the existing aesthetic, light, or glare conditions. Therefore, the ETRPA Alternative would have approximately the same effects as the Proposed Project.

#### 8.3.5.13 Cultural Resources

Redevelopment of the site with all nonaviation uses would have the same less than significant effects as the Proposed Project on cultural resources on the property since none of the cultural resources on the site are considered potentially significant. As with the Proposed Project, potential impacts of unknown archaeological resources during ground disturbance would be mitigated through implementation of standard construction monitoring measures.

The ETRPA Nonaviation Alternative assumes status quo operations at JWA. As such, there would be no additional or new effects on cultural resources in the JWA area, as there are no known archaeological, paleontological or historic resources on the already developed airport property.

#### 8.3.5.14 Recreation

With the ETRPA Nonaviation Plan Alternative, the specific recreational features and facilities proposed with the Proposed Project would not be provided. However, the ETRPA Nonaviation Plan Alternative would provide recreational facilities, including a 360 acre community park, community and neighborhood parks totaling 168 acres, open space linkages to surrounding off-site open space areas, walking paths, hiking trails, off-road bikeways, a sports stadium, and a hotel conference center with golf course. No significant long-term impacts of the ETRPA plan on parks and recreational facilities are anticipated (City of Irvine GPA, Zone Change and Annexation DEIR for MCAS El Toro and James A. Musick Branch Jail, March 31, 1999) as the City of Irvine would provide for the parks and recreational needs of the site under the ETRPA plan.

Physical effects on adjacent off-site recreational trails are likely to be the same level of magnitude as that of the Proposed Project, assuming that there would be some temporary disruptions to on-road bikeways for street improvements to serve the nonaviation plan improvements. The primary difference between the ETRPA Nonaviation Plan Alternative and the Proposed Project in terms of recreational impacts would be the lack of aviation related noise under the ETRPA Nonaviation Plan Alternative. Because there would be no exposure of planned bicycle trails and riding and hiking trails to aircraft noise from the proposed OCX, the noise related impacts to recreation would be less than under the Proposed Project.

The status quo operations of JWA under the ETRPA Nonaviation Plan Alternative would have no change in effect on recreational facilities in the JWA area.

#### 8.3.5.15 Public Health and Safety

#### Aviation Safety

Under the ETRPA Nonaviation Alternative, there would be no aviation activity at MCAS El Toro. Since there is no aviation activity at OCX, there would be zero aviation risks. Under the Proposed Project, there would be no significant adverse impacts related to aviation safety at the MCAS El Toro site or at JWA relative to on-airport and off-airport fatal accidents per million operations.

Compared to the existing conditions, there would be virtually no changes in the number of air carrier and air cargo operations and general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo and general aviation accident risks at JWA would remain the same as the existing conditions. There would be no significant adverse impacts related to aviation safety at JWA.

#### 8.3.5.16 Hazardous Materials and Hazardous Wastes

#### Hazardous Materials/Waste Usage

Most of the proposed uses under the ETRPA Nonaviation Plan Alternative would not yield large quantities of hazardous waste. However, hazardous waste generation could result from the proposed Orange County Transit Authority (OCTA) rail and bus maintenance facility, light industrial uses, and research and development uses. Compared to the Proposed Project, which would accommodate jet fuel storage and aircraft maintenance, the ETRPA Nonaviation Plan Alternative would involve substantially smaller quantities of hazardous material. All hazardous materials used, or generated, would be regulated by existing federal, state and local regulations. By meeting the regulatory guidelines, potential impacts associated with hazardous material use, or generation, would be maintained to below a level of significance. The potential impacts of the ETRPA Nonaviation Plan Alternative related to hazardous materials are generally described in the following sections for the different land uses proposed on the site under this alternative.

#### Habitat Reserve

The federal Habitat Reserve, outlined under the ETRPA Nonaviation Plan Alternative, would fall under the jurisdiction of a federal agency, which is the same as assumed under the Proposed Project. The ETRPA Nonaviation Plan also includes 686 acres of recreation uses in the southern portion of this area. However, the potential impacts associated with remediation activities addressed under the Installation Restoration Program (IRP) for Sites 1, 2, and 17 are the same for both the ETRPA Nonaviation Plan Alternative and the Proposed

Project. Since the areas under the ETRPA Nonaviation Plan Alternative are designated for use as a Habitat Reserve, the potential impacts associated with the presence of hazardous waste and the likelihood of future hazardous waste generated materials are anticipated to be less than significant.

#### Education, Research & Technology (ERT) District

Under the ETRPA Nonaviation Plan Alternative, the ERT District consists of an integrated, higher intensity grouping of high-density residential, retail and office uses, such as Village, Business/Technology, Education, Research & Technology Campus, Entertainment/Mixed Use, Parks/Open Space, Retail, and a Sports Complex. Development of the ERT District would encompass all or part of IRP Sites 3, 7, 11, 12, and 14.

#### **ERT Village**

A potential impact associated with the ERT Village residential land use outlined in the ETRPA Nonaviation Plan Alternative is its relation to the Base Realignment and Closure (BRAC) remediation activity at MCAS El Toro. One of the areas included within the ERT Village land use is IRP Site 3 near the eastern end of the loop formed by the proposed East Culture Road. Development of the ERT Village residential land use, which would overlie or directly abut IRP Site 3, would conflict with California Health and Safety Code (H&SC) Section 25202.5. Under this statute, a minimum buffer of 2,000 feet is required for residential development in the vicinity of a hazardous waste disposal facility. This is a significant adverse impact.

Under H&SC Section 25202.5, disposal of hazardous wastes, at a site with a buffer zone of less than 2,000 feet from residential land uses, is only allowable if it can be proven to the satisfaction of the California Department of Toxic Substance Control (DTSC) that the buffer zone is sufficient to protect present and future public health and safety. Therefore, development of residential uses overlying or within 2,000 feet of IRP Site 3 under the ETRPA Nonaviation Plan Alternative would result in significant adverse impacts related to hazardous wastes.

Another area of concern related to land use development within IRP Site 3 is that construction activities could result in greater potential impacts under the ETRPA Nonaviation Plan Alternative than those potential impacts under the Proposed Project. Construction activities in the area may require earth moving and excavation to accommodate foundations, subterranean parking, or footings for multi-story structures. Excavation in this area could result in the unearthing of hazardous wastes associated with IRP Site 3 and resultant exposures to construction workers and future residents on the site to levels that may exceed those deemed acceptable from a health protective perspective. Subsurface chemical concentrations in the soil are unknown, therefore a subsurface assessment of soil contamination would be required prior to any construction activities in the area where subsurface excavation is planned in order to more accurately characterize the risks

associated with disturbance of soils at the site. Impacts associated with construction worker exposures to contaminants likely could be mitigated to below a level of significance through implementation of personal protective equipment appropriate to the potential health threat posed by the site.

#### **Business/Technology**

The Business/Technology use areas of the ETRPA Nonaviation Plan Alternative would entail Research and Development, and Light Industrial uses. The Research and Development uses would consist of a variety of business and high technology uses, including production and service establishments, scientific laboratories, new technology training centers, professional/ administrative offices, and other supporting services. Police and fire stations are also a possibility within the Business/Technology land use designation.

The Light Industrial uses would encompass communications equipment manufacturing, electronics, pharmaceuticals, plastics, furniture and fixtures, printing and publishing, wholesaling, warehousing and distribution centers, professional/administrative offices, and other supporting uses.

Under the ETRPA Nonaviation Plan Alternative, proposed developments within the Business/Technology area encompass IRP Sites 11 and 12, and a portion of IRP Site 7. As discussed in Section 4.16 (Hazardous Wastes and Hazardous Materials), the Remedial Investigation/Feasibility Study (RI/FS) process for IRP Sites 7, 11, and 12 has not been completed; therefore, human health risk assessment data are not available for these sites and potential impacts associated with development of these sites cannot be fully evaluated. However, the type of land uses proposed for the site under the ETRPA Nonaviation Plan Alternative is not generally considered by EPA to be as sensitive as residential uses. Therefore, potential impacts associated with development under the proposed ETRPA Nonaviation Plan Alternative may not be significant.

As discussed in Section 4.16 (Hazardous Wastes and Hazardous Materials), the Department of the Navy (DON), with the approval of EPA and Cal-EPA, has been using industrial cleanup standards for IRP sites at MCAS El Toro. Based on this standard, development of the proposed uses within the Business/Technology land use areas of the ETRPA Nonaviation Plan Alternative is not anticipated to result in significant adverse impacts related to the presence of hazardous waste sites. Should conditions at Sites 7, 11, and 12 pose human health hazards which exceed acceptable levels under the industrial exposure scenario, remedial action will be prescribed by the DON, and agreed to by the U.S. EPA and Cal-EPA, which would reduce potential impacts to below a level of significance. These remedial actions could have an adverse effect on the land uses proposed.

Under the ETRPA Nonaviation Plan Alternative, IRP Site 14 (Battery Acid Disposal Area) lies within the ERT District, near the proposed intersection of Quantum Road and Research Parkway. The site is currently undergoing remedial investigation, and no human health risk assessment data are available. The ETRPA Nonaviation Plan Alternative proposes more

intense development overlying IRP Site 14, compared with that of the Proposed Project; thus, the potential for adverse impacts related to residual hazardous wastes is greater under the ETRPA Nonaviation Plan Alternative. However, under the ETRPA Nonaviation Plan Alternative, the ERT District would not likely consist of highly sensitive uses such as residential development; therefore, potential impacts associated with development of the IRP Site 14 area are anticipated to be less than significant following implementation of any prescribed remedial action.

#### Park and Open Space

Park and open space uses would comprise approximately 50 percent of the proposed ETRPA Nonaviation Plan Alternative. A network of open space corridors would interconnect the activity centers of the site, linking parks and recreational facilities to surrounding open space areas and other proposed uses. IRP Site 5 (Perimeter Road Landfill), a portion of IRP Site 7 (Drop Tank Drainage Area No. 2), and IRP Site 16 (Crash Crew Pit No. 2) are located within the area proposed for park and open space uses under the ETRPA Nonaviation Plan Alternative. Under the Proposed Project, IRP Site 5 is located in an open space area proposed for use as a Secondary Habitat Corridor. The potential impacts associated with the presence of IRP Site 5 under the ETRPA Nonaviation Plan Alternative are, therefore, similar to those of the Proposed Project, in terms of the proposed site development. Consequently, no significant adverse impacts related to the existence of IRP Site 5 are anticipated, provided that the selected presumptive remedy for the site remains intact.

Under the ETRPA Nonaviation Plan Alternative, proposed residential development also directly abuts IRP Site 5, which would be inconsistent with the California Health and Safety Code buffer zone requirements, as described regarding IRP Site 3 development.

Under the ETRPA Nonaviation Plan Alternative, a portion of IRP Site 7 and all of IRP Site 16 underlie an area proposed for park and open space land uses. Development of the proposed uses in this area will likely require some surface grading activities; however, no deep excavation is anticipated. Similar to the impacts of the Proposed Project, potential impacts would most likely be associated with exposing contaminated soils during construction. Because the ETRPA Nonaviation Plan Alternative proposes no structural development for human occupation in the Open Space area overlying IRP Sites 7 and 16, this usage will entail a relatively low level of risk to the public. Development of these sites under the Proposed Project would result in coverage by asphalt or concrete surfaces, which could aid in the reduction of potential contaminant migration. Park and Open Space uses under the ETRPA Nonaviation Plan Alternative likely would not provide this same benefit. Without more specific human health risk data for a portion of IRP Site 7 and all of IRP Site 16, the potential impacts associated with disturbance of these sites cannot be fully addressed; however, it is possible that some remedial action may be required before the sites can be developed for the proposed Millennium Park/Open Space uses.

#### **Entertainment/Mixed Use**

In response to a request by OCTA, the ETRPA Nonaviation Plan Alternative was modified to include approximately 50 acres of land for a rail and bus maintenance facility. The proposed facility would be located in an area northwest of the regional transportation center that was previously identified for entertainment/mixed use and research and development.

Site 8, the Defense Reutilization and Marketing Office (DRMO) Storage Yard, is located within the boundaries of the proposed OCTA Maintenance Facility. The RI/FS process for Site 8 has not been completed; therefore, human health risk assessment data are not available for this site and impacts associated with development of this site cannot be fully evaluated. The DON, with the approval of EPA and Cal-EPA, has been using industrial cleanup standards for IRP sites at MCAS El Toro. Based on these standards, development of the proposed uses within the OCTA Maintenance Facility area, under the ETRPA Nonaviation Plan Alternative, would not result in significant adverse impacts related to hazardous wastes. Should conditions at IRP Site 8 pose human health hazards that exceed acceptable levels under the industrial exposure scenario, remedial action will be prescribed by the DON, and agreed to by the U.S. EPA and Cal-EPA, to reduce potential impacts to below a level of significance.

#### 8.3.5.17 Socioeconomics

Under the ERTPA Nonaviation Plan Alternative, the military would leave MCAS El Toro and the site would be converted into a mixed-use urban center, emphasizing high technology industries, education and recreation. This alternative would also provide a range of housing types on the site. JWA would continue to operate at a maximum of 8.4 MAP. As under the Proposed Project, no housing units will be constructed at JWA.

Under this alternative, almost 56,000 jobs would be generated, including 50,700 jobs at El Toro and 5,200 jobs at JWA in 2020 as shown earlier in Table 8.2-3. This represents a net increase of 48,100 jobs at El Toro and 3,100 jobs at JWA, over existing 1998 conditions. There would be approximately 13,600 people residing at El Toro under this Alternative in 5,900 housing units. In total, this alternative support generated 55,900 jobs, 13,600 persons, and 5,900 housing units on the project site. This figure is significantly higher than the number of jobs, persons, and housing units expected under the Proposed Project. As with the Proposed Project, economic activity occurring at El Toro and JWA, as well as expenditures by visitors arriving by air through JWA, would stimulate additional off-site job growth. The total number of on-site and off-site jobs stimulated by the airport system would be similar to the level under the Proposed Project.

Given the level of employment and population growth generated by this alternative, this would be considered a significant adverse impact under the threshold of significance related to inducing substantial growth or concentration of population or housing.

The ratio of 9.5 jobs for each housing unit under this alternative is lower than the jobs/housing ratio anticipated on the JWA and MCAS El Toro site under the adopted regional forecasts. It is also significantly lower than the jobs/housing ratio forecast under the Proposed Project. However, since this alternative would produce a jobs/housing ratio in the surrounding area that is higher than the ratio expected under the adopted regional growth forecasts, and since these areas are considered by SCAG to be housing poor, this would be a significant adverse impact of this alternative.

The impacts of this alternative related to housing demand, including low and moderate income housing needs, would be higher than under the Proposed Project as a substantially higher number of jobs would be generated under this alternative compared to the Proposed Project. However, this alternative would also provide a range of housing types on site, partially accommodating the increased demand for all types of housing in the area. Even with this housing, the impact of this alternative would be significantly adverse.

The ERTPA Nonaviation Plan Alternative would generate almost 56,000 jobs at El Toro and JWA, which is substantially higher than what was adopted in the regional forecasts, and would also result in an increase in on-site housing units over what was adopted in the regional forecasts. Therefore, this alternative would result in significant adverse impacts related to inconsistency with adopted regional forecasts.

#### 8.3.5.18 Risk of Upset

Ultimate build out and development of the ETRPA Nonaviation Alternative may include land uses (e.g. the OCTA rail and bus maintenance facility, light industrial uses, and research/ development uses) that could result in a presently unidentified potential for risk of upset conditions. Compared with the Proposed Project, which would accommodate jet fuel transport and storage facilities and operations, the ETRPA Nonaviation Alternative likely would involve a lower potential for risk of upset conditions at the MCAS El Toro site. Risk of upset at JWA would be essentially the same as under the Proposed Project. On this basis, it would appear that implementation of the ETRPA Nonaviation Alternative would result in an unknown, but likely lower potential for adverse impacts to public health and safety than the Proposed Project.

# 8.3.6 Feasibility

Benefits generated by the proposed aviation uses on the MCAS El Toro site under the Proposed Project and benefits of the alternative nonaviation reuse plan are not mutually exclusive. Virtually all of the uses proposed under the nonaviation plan can be successfully developed at other locations in the County. In fact, the ability of many of these uses to attract tenants may be enhanced by airport system development and the improved economic competitiveness of the County resulting from Proposed Project implementation. Therefore, the potential benefits of the proposed aviation plan and the components of the nonaviation

plan if developed at other locations in the County are complementary, and in fact, taken together may be greater than the sum of the individual parts.

The converse, however, is not true. There is no feasible alternative site for an international airport in Orange County. Thus, reuse of MCAS El Toro for nonaviation purposes would preclude development of significant international and expanded domestic air service capabilities in Orange County, with the attendant loss of the potential quantifiable and non-quantifiable benefits to the County's economy (Technical Report 16, Economic Benefits Study).

In October, 1998, a consultant team of INTERRA, BBC Research & Consulting, and Urban Design Camp prepared an independent analysis of the ETRPA Nonaviation Plan for the Orange County Regional Airport Authority titled: Development Feasibility Analysis: El Toro Non-Aviation Reuse Alternative Millennium Plan, October 1998. In summary, the analysis concluded:

- (i) The office market absorption projections are reasonable.
- (ii) The current supply of industrial land in southern Orange County would meet demand for 30 years without this alternative. Therefore, the alternative would face considerable competition on cost/revenue pressure.
- (iii) The southern Orange County retail market is already well served and, therefore, the alternative's commercial center and Power Center are highly speculative.
- (iv) Actual residential competition in southern Orange County is 90 percent higher than the competition assumed in the alternative.
- (v) ETRPA underestimated the backbone public services costs for the alternative by \$38 million for water and sewer capacity, \$9 million for grading, and \$31.25 million for highway and street lighting improvements.
- (vi) ETRPA underestimated the development costs for schools, fire stations, police facilities, libraries, open space, and landscape amenities and other costs by \$226.15 million.
- (vii) ETRPA estimated that demolition costs would be about \$32.85 million, which is too low. (Note: the ASMP estimates that the ETRPA Alternative demolition costs would be \$193 million.)
- (viii) Based on these costs, the ETRPA Alternative would result in a net loss (cost versus revenue) of \$210 million.

In addition, the sports facilities included in the ETRPA Alternative would result in a net loss in cost versus revenue, requiring public subsidies according to the INTERRA analysis and

Sports, Jobs and Taxes, the Economic Impact of Sports Teams and Stadiums, Roger Noll and Andrew Zimbalist, Editors (1997).

#### 8.3.7 Conclusions

In conclusion, the ETRPA Alternative would:

- (i) Would not meet any of the general project objectives, and would not meet the aviation objectives relating to passenger and cargo demand, service opportunities, industry competition, economic growth, business activities, existing land use restrictions, or General Plan implementation;
- (ii) Would not avoid impacts on land uses, General Plan consistency, regional air quality emissions, toxic air contaminants, and construction emissions;
- (iii) Would result in new or additional significant adverse impacts on traffic, regional VMT, regional air quality emissions, local air quality impacts due to traffic CO hot spots, agricultural soils, hazardous wastes, socioeconomics, economics, and adverse effects of aviation noise on a regional basis; and
- (iv) Would avoid aviation noise at the El Toro site, including sleep disturbances and on recreation uses; toxic air contaminants at El Toro associated with airport operations; local air quality impacts at OCX due to aircraft and associated operations; aviation safety effects at El Toro; and aviation risk of upset at El Toro.

In summary, the ETRPA Nonaviation Plan Alternative would avoid unmitigatable project impacts on toxic air contaminants near the El Toro site associated with airport operations and aviation noise impacts on sleep disturbance and recreation uses. However, this alternative would result in new or additional impacts in several categories, including traffic, significant increases in regional VMT, regional air quality emissions, construction emissions, local CO hot spot air quality impacts at OCX, and further loss of agricultural soils.

# 8.4 ALTERNATIVE A: JWA – STATUS QUO AVIATION ROLES; OCX – FULL DOMESTIC

This section presents the potential impacts of Alternative A as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

This alternative was selected for analysis because it has the potential to lessen aircraft noise, traffic, and local air quality impacts of the Proposed Project while still feasibly attaining most of the objectives of the project.

#### 8.4.1 Aviation Uses

Under Alternative A, OCX provides short-, medium- and long-haul domestic and limited (Mexican and Canadian) international air passenger service for an estimated 19.0 MAP, 12 percent (2.2 MAP) of which are passengers with connecting flights. OCX is also forecast to annually handle approximately 0.04 million tons of international cargo, and 1.21 million tons of domestic cargo. This alternative includes an on-airport 500-room hotel. Fuel for aircraft operations at OCX is assumed to be delivered by trucks.

Under Alternative A, JWA would continue to serve general aviation, as well as provide primarily short- and medium-haul domestic passenger service. JWA will serve 6.0 MAP in 2020 under this alternative, which is less than the current service level of 7.5 MAP.

No major runway improvements, such as the lengthening of a runway, would be made at JWA. On the MCAS El Toro site, Alternative A would reuse existing Runways 16R/34L and 7R/25L and reconstruct Runways 16L/34R and 7L/25R, offset 800 feet and 700 feet, respectively, from their parallel twins to meet FAA runway separation requirements for operations under visual conditions. Figure 8-2 depicts Alternative A.

# 8.4.2 Nonaviation Revenue Support Uses

The nonaviation land uses proposed under Alternative A are the same as the Proposed Project. However, the aviation development area in this alternative would be reduced by approximately 200 acres in Planning Area 1 and by about 175 acres in Planning Area 2 compared to the Proposed Project. These lands are assumed to be agricultural, horticultural, or passive open space uses.

# 8.4.3 Attainment of Project Objectives

This alternative would meet the general project objectives for base reuse except to enhance higher quality economic development. Alternative A would meet most of the aviation related objectives with the exception of meeting full international air traffic demand in Orange County, and achieving economic growth and business activities that would rely on full international aviation service. Since this alternative does not meet these project objectives, this Draft EIR proposes to reject this proposal.

# 8.4.4 Environmental Impacts of Alternative A

#### 8.4.4.1 Land Use

Compared to existing conditions, this alternative would have no significant land use impacts at JWA, but would have impacts at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the Proposed Project.

There would be a slight increase in aviation activity at JWA and a decrease in overall aviation activity at MCAS El Toro under Alternative A compared with the Proposed Project. The aviation and nonaviation revenue support land uses for Alternative A are essentially the same as the Proposed Project. The perimeter uses associated with the development of an airport on the El Toro site are generally comparable in intensity or less intense than the existing and planned adjacent off-site uses. Therefore, as with the Proposed Project, there are no significant land use conflicts associated with the proposed land uses under this alternative.

The on-site agricultural uses under this alternative will be preserved within an increasingly urbanized area. Agricultural management practices can be implemented to reduce potential impacts. As with the Proposed Project, the on-site agriculture uses will not have significant impacts on off-site or other on-site land uses, and the impacts that might occur can be controlled through agricultural management practices and through the terms of the County's lease agreements.

The Proposed Project includes approximately 65 acres of airport parking in Planning Area 5 north of Irvine Boulevard, 200 acres of aviation uses in Planning Area 1, and 175 acres of aviation uses in Planning Area 2, which would not be required for this alternative. These airport areas includes Prime Agricultural Soils; therefore, this alternative would reduce the loss of Prime Agricultural Soils by up to 440 acres compared to the ASMP. However, as with the Proposed Project, there would continue to be a loss of Prime Agricultural Soils compared to existing conditions.

The proposed airport use at MCAS El Toro under Alternative A would attract new development in nearby areas. There is a potential for undesirable land use development (such as sexually oriented businesses) in the vicinity of the site, unless the County and adjacent cities have adequate land use controls in place. Also, the design of future off-site development may adversely affect existing and planned development in the adjacent jurisdictions if appropriate design standards are not implemented by the local jurisdictions. This potential impact is the same under both Alternative A and the Proposed Project.

Although the JWA aviation activity under Alternative A is slightly higher than the Proposed Project, it is lower than existing conditions, and future improvements would be very limited within the boundaries of the airport. Therefore, Alternative A, with less commercial aviation activity than existing JWA, would not have significant land use impacts.

The impacts of Alternative A related to land use are comparable to the impacts under the Proposed Project. This alternative would not avoid or measurably lessen the impacts of the Proposed Project.

#### 8.4.4.2 General Plan Consistency

Compared to existing conditions, this alternative would have no significant impacts at JWA, but would have significant impacts at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance.

Alternative A introduces a civilian aviation use to MCAS El Toro; therefore, as with the Proposed Project, an amendment to the Airport Environs Land Use Plan (AELUP) is required. In addition, the General Plan Amendments required for the Proposed Project would also be required for this alternative. Alternative A includes land uses which conflict with the adopted City of Irvine General Plan for Planning Area 7 (City of Irvine Planning Area 30). An amendment to the City of Irvine General Plan would not be required since the site would be under the County's jurisdiction. The need for amendments to General Plans and the AELUP for Alternative A are comparable to those required for the Proposed Project, therefore; the General Plan consistency impacts for Alternative A are the same as for the Proposed Project.

# 8.4.4.3 Transportation and Circulation

Compared to existing conditions, this alternative would have no significant impacts at JWA, but would have significant impacts at the El Toro site similar to the Proposed Project. With the project mitigation measures identified for this alternative, the impacts of this alternative would be reduced to a level of insignificance.

The AM and PM peak hour and average daily traffic (ADT) generated by the aviation operations at JWA and OCX and by nonaviation revenue support land uses with build out of Alternative A are summarized in Table 8.4-1. Refer to Section 9.0 in the 1999 Traffic Analysis Technical Report for detailed information on the methodology applied to produce trip generation estimates for Alternative A. This alternative would generate an increase of 112,757 ADT at the El Toro site compared to an increase of 150,723 ADT for the Proposed Project over existing conditions. The alternative would generate 167,083 ADT less than the CRP. At JWA, this alternative would generate 11,176 ADT less than existing conditions. For a comparison of peak hour trip generation, see Table 4.3-8. In summary, the alternative would generate significantly fewer daily and peak hour trips than the Proposed Project.

Table 8.4-1
Trip Generation Summary - Alternative A

Project Component			lour				ADT	Existing ADT
Former MCAS El Toro site			Constant		Carles Sanda	and the second the	e e e e e e e e e e e e e e e e e e e	
OCX Commercial Air Terminal	2,552	1,699	4,251	2,571	2,530	5,101	84,328	
OCX Air Cargo Handling Facility	360	433	793	473	281	754	10,810	
Nonaviation Revenue Support Land Uses	3,692	728	4,420	989	3,573	4,562	43,019	***************************************
Sub-Total (El Toro Site)	6,604	2,860	9,464	4,033	6,384	10,417	138,157	25,400
JWA	1,088	729	1,817	1,433	1,438	2,871	36,274	47,450
TOTAL	7,692	3,589	11,281	5,466	7,822	13,288	174,431	72,850

The on-site and site access circulation plans anticipated for JWA and OCX in Alternative A are the same as those described earlier in Section 4.3 (Transportation and Circulation) for the Proposed Project with the exception that 2020 Alternative A conditions do not assume the Trabuco Road/ETC interchange option because of the reduced trip generation of the alternative. Peak hour levels of service with and without Alternative A were compared in order to identify locations on the existing plus committed circulation system that require improvements to mitigate traffic impacts of Alternative A and other foreseeable growth or development. Table 8.4-2 compares, in summary, the Alternative A highway impacts to the existing conditions and existing conditions plus Proposed Project. As discussed in Section 4.3.6.5, there is minimal comparison between the existing conditions plus Proposed Project versus the Alternative A impacts due to highway improvements recently completed and the effects of committed highway improvements. Section 9.0 in the 1999 Traffic Analysis Technical Report includes detailed summaries of the Alternative A build out traffic volumes and levels of service (LOS) and comparisons between existing plus committed conditions with and without Alternative A for intersections and arterial roadways within the traffic analysis study area, and Section 9.0 in the 2001 Traffic Analysis Technical Report Addendum includes comparable information for freeway/tollway mainline segments and freeway/tollway ramps within the traffic analysis study area).

Table 8.4-2
Summary Comparison of Traffic Impacts for Alternative A to
Existing Conditions and Existing Conditions Plus Project

Existing Conditions	Existing Conditions	Alternative A With Existing
Deficient Highway Facilities	Plus Proposed Project Impacts	Plus Committed Facilities
Location	Location	Location
INTERSECTIONS	IMPACTED INTERSECTIONS	IMPACTED INTERSECTIONS
Newport (NB) & Del Mar	Bake & Portola	ETC East Leg NB & Irvine
El Toro & SR-73 NB Ramps	Sand Canyon & Trabuco	Sand Canyon & Irvine
Campus & N. Bristol	Bake & I-5/I-405 SB Ramps	Sand Canyon & Trabuco
Jamboree (SB) & Walnut	Bake & Rockfield	Jeffrey & Irvine
Jamboree & I-405 NB Ramps	Jeffrey & Alton	Jeffrey & Trabuco
Jeffrey & I-405 NB Ramps	Jeffrey & I-405 NB Ramps	Sand Canyon & I-5 NB Ramps
Red Hill & MacArthur	Jeffrey & I-405 SB Ramps	Sand Canyon & I-5 SB Ramps
Irvine Center & Lake Forest	Jeffrey & Walnut/I-5 SB	Alicia & Paseo Valencia
Bake & Jeronimo	Sand Canyon & I-5 NB Ramps	El Toro & Rockfield
El Toro & Avd Carlota	Sand Canyon & I-5 SB Ramps	Alicia & Jeronimo
La Paz & Cabot/I-5 SB	Irvine Center & Lake Forest	Red Hill & I-5 NB Ramps
Los Alisos & Muirlands	Bake & Irvine/Trabuco	
Alicia & Jeronimo	Bake & Toledo	IMPACTED ARTERIAL ROADS
Alicia & Muirlands	Los Alisos & Muirlands	Irvine (ETC East Leg to PA-2 East Access Road)
La Paz & Muirlands/I-5 NB	Alicia & Jeronimo	Irvine (Jeffrey to Sand Canyon)
Red Hill & Edinger	Newport & Old Irvine	Laguna Canyon (south of El Toro)
Red Hill & Sycamore		
Red Hill & Walnut	IMPACTED ARTERIAL ROADS	IMPACTED FREEWAY/ TOLLWAY RAMPS
	Laguna Canyon (I-405 to SJHTC)	FTC (SR-241) at Portola East (NB Off-Ramp)
ARTERIAL ROADWAYS	Laguna Canyon (south of El Toro)	I-5 at Red Hill (SB On-Ramp)
Portola (Sand Canyon to Foothill Toll Road)	Culver (Bryan to Trabuco)	I-5 at La Paz (SB Off-Ramp)
Laguna Canyon (I-405 to SR-73)		I-405 at Sand Canyon (NB Direct On-Ramp)
Laguna Canyon (south of El Toro)	IMPACTED FREEWAY RAMPS	
Culver (Bryan to Trabuco)	I-5 at Culver (SB Off-Ramp)	IMPACTED FREEWAY/ TOLLWAY SEGMENTS
Michelson (Carlson to Harvard)	I-5 at Sand Canyon (NB On-Ramp)	FTC (Lake Forest to south of Portola East)
	I-5 at Sand Canyon (SB Off-Ramp)	I-5 (I-405 to north of SR-55)

Existing Conditions.	Alternative A With Existing
Phis Proposed Project Impacts	Plus Committed Facilities
I-405 at Sand Canyon (NB Direct On-Ramp)	I-405 (Jamboree to north of SR-55)
IMPACTED FREEWAY SEGMENTS	
I-5 (Jeffrey to north of SR-55)	
A	
	Phis Proposed Project Impacts I-405 at Sand Canyon (NB Direct On-Ramp)  IMPACTED FREEWAY SEGMENTS

Abbreviations: NB-northbound EB-eastbound SB-southbound WB-westbound

In addition, a comparison of the impacts of Alternative A may also be made to the Proposed Project's impacts during the phasing years. As discussed in detail in Section 4.3.6.6 of this Draft EIR No. 573, as supplemented, under the Proposed Project phasing years, four intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 1 conditions (2005), five intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 2 conditions (2010), and nine intersection locations, two arterial roadway segments, one continuous freeway mainline segment and two freeway ramps would be significantly impacted under Phase 3 conditions (2015). At Phase 4 build out the Proposed Project would result in significant impacts not previously identified to four freeway/tollway mainline segments and four freeway/tollway ramps. See Supplemental Analysis, Section 4.3.6.5. In each case, however, the identified impacts will be mitigated to a level below significant during the applicable phasing year (see Section 4.3.7.2, Table 4.3-20).

#### 8.4.4.4 Noise

Compared to existing conditions, this alternative would create no significant noise impacts at JWA (see Table 8.4-4). Table 8.4-3 shows a land use comparison with noise contours for 1998 military and year 2020 alternatives for El Toro. Also, see Figure 8-3, which depicts noise contours for Alternative A.

The Alternative A 65 CNEL contour line would include 6.6 square miles of land for OCX. For JWA, the numbers are the same as the Proposed Project. The 65 CNEL for the existing military aircraft operations at MCAS El Toro include 6.3 square miles of land. Therefore, Alternative A would increase the area affected by the 65 CNEL surrounding the El Toro site by 0.3 square miles, compared to an increase of 1.5 square miles for the Proposed Project.

The Proposed Project would increase noise sensitive land uses by three churches and one private school compared to existing conditions at the El Toro site (see Table 8.4-3). However, the alternative would avoid these impacts. Therefore, Alternative A would result in no increase in sensitive land uses affected by the 65 CNEL because: 1) County and City policies have restricted incompatible land uses within the much larger (28.81 square mile) MCAS El Toro AICUZ 65 CNEL, 2) the Alternative A 65 CNEL line does not exceed the AICUZ 65 CNEL boundary north of the El Toro site (and, therefore, avoids the impacts of the Proposed Project outside the AICUZ 65 CNEL line), and 3) land use restrictions and noise mitigation programs minimize land use conflicts at JWA.

EIR 563 concluded that a civilian airport at MCAS El Toro would result in significantly greater number of total operations compared to historical military levels of use, both throughout the day and during the nighttime hours. Although the Proposed Project and Alternative A would have significantly fewer operations than the Community Reuse Plan analyzed in EIR 563, the number of forecast civilian operations is still substantially greater at El Toro than the existing conditions level of military operations.

Table 8.4-3
Land Use Comparison with Noise Contours for 1998 Military and Year 2020 Alternatives for El Toro

Land Ose Comparison with Hoise Contours for										
rone				Year 2020 A	iternatives					
	Proposed			i i			No			
	110127	<b>A</b>	•	74		•	Project			
16.6	22	155	20.4	12.2	120	22.2	Λ			
Į.				1	1		0			
1		3					0			
3,0	3.9	2.7	3.0	1.9	2.3	3.6	v			
63	6.4	6	62			4.4	0			
	ŧ						0			
			1				0			
2.9	3.2	2.0	3	1.0	2.3	2.7	U			
0.3	0.7	0.5	0.2	0.15	0.2	1 2	0			
	1	1	l i	1	1		0			
1	i				i I		0			
	0	, o	U	0	U	U	U			
672	1927	1212	707	204	707	2.411	0			
ì	1			1			0			
	•	-					0			
0	U	U	U	U	U	U	U			
Δ.		0			•	2	۵			
	i •		_	I	3		0			
1	E .		_				0			
"	0	U	U	"		U	V			
1	3	2	2	,	,	2	0			
1 "	}			1			0			
	-	_		•	1		0			
	U	U	U	U	U	U	U			
0	<u> </u>	0	A	Λ.	0	0	Δ.			
1	1			I "			0			
-	1			_			0			
"	U	v	U	U	v	U	U			
۸	n	n	Λ .	0		^	0			
	•	_		-			0			
1	{	•		}	ŧ .	_	0			
	"	v	v	,	v	v	v			
12	13	13	10	g	12	10	0			
1	1		l .				0			
i .	i			_			0			
	1998 Military  16.6 6.3 3.0 6.3 4.8 2.9 0.3 0 0 672 0 0 1 on base 0 0 0 0 1 on base 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1998. Proposed Military Project  16.6	1998   Proposed   A	1998   Project   A   C	1998	1998   Proposed	1998   Proposed   A			

Table 8.4-4
Land Use Comparison with Noise Contours for 1998 and Year 2020 Alternatives for John Wayne Airport

•		gesti and the second			Year 2020 A		
10 mm	10 A S S S S S S S S S S S S S S S S S S	1985	Proposed				No
C	1998	Master Plan	Project	C :	F	G ·	Project
Square Miles Within Contour:				_			
- 60 to 65 CNEL Contour	2.2	4.13	1.8	2.77	2.76	7.4	2.49
- 65 to 70 CNEL Contour	0.75	1.22	0.8	1.07	1.08	3.28	0.98
- 70+ CNEL Contour	0.74	0.99	0.54	0.73	0.69	2.19	0.84
Square Miles of Residential:							
- 60 to 65 CNEL Contour	0.26	0.59	0.22	0.39	0.38	2.65	0.38
- 65 to 70 CNEL Contour	0.04	0.12	0.03	0.09	0.07	0.71	0.09
- inside 70 CNEL Contour	0	0	0	0	0	0.17	0.00
Number of Residences Inside Contour:							
- 60 to 65 CNEL Contour	682	1548	577	1023	997	6954	997
- 65 to 70 CNEL Contour	134	314	79	236	184	1863	236
- inside 70 CNEL Contour	0	l 0	0	0	0	446	0
Number of Public Schools Inside Contour:					_		
- 60 to 65 CNEL Contour	0	0	o	0	0	0	0
- 65 to 70 CNEL Contour	0	0	0	o	Ŏ	4	ő
- inside 70 CNEL Contour	٥	0	0	ő	Ŏ	ó	0
Number of Private Schools Inside Contour:	_		_	ľ	Ů	_	
- 60 to 65 CNEL Contour	lo	0	0	0	2	1	0
- 65 to 70 CNEL Contour	ŏ	0	Ö	0	0	3	0
- inside 70 CNEL Contour	ŏ	0	Ö	0	0	0	
Number of Colleges Inside Contour:	"		\ \ \	"	V	"	'
- 60 to 65 CNEL Contour	0	0	0	0	١ ,	0	_
- 65 to 70 CNEL Contour	0	0	1	Ĭ.	0	I -	0
- inside 70 CNEL Contour	0	0	0	0	0	0	0
Number of Hospitals Inside Contour:	U	l v	U	U	0	0	0
- 60 to 65 CNEL Contour					_		_
- 65 to 70 CNEL Contour	0	0	0	0	0	0	0
	0	0	0	0	0	0	0
- inside 70 CNEL Contour	0	0	0	0	0	0	0
Number of Churches Inside Contour:					_		
- 60 to 65 CNEL Contour	2	2	2	2	2	6	1
- 65 to 70 CNEL Contour	0	2	0	0	0	2	1
- inside 70 CNEL Contour	0	0	0	0	0	1	0

As discussed earlier, the CNEL calculation factors in the number of daily operations and assigns a "penalty weighting" to operations occurring during the nighttime hours (10 p.m. to 7 a.m.). However, the substantial increase in the number of operations, particularly during nighttime hours, may be considered a significant impact of Alternative A independent of the CNEL computation.

The noise levels identified for the Proposed Project as well as Alternative A will be considered an annoyance by some residents and nighttime events will cause some sleep disturbance regardless of the levels of significance prescribed by regulatory agencies. Therefore, a mitigation measure for sleep disturbance is proposed in Section 4.4. With this mitigation measure, Alternative A impacts are reduced but remain significant similar to the Proposed Project.

#### 8.4.4.5 Air Quality

Alternative A would result in new significant regional air quality impacts that would be greater in all phasing years than under the Proposed Project's development scenarios due to the failure of this alternative to meet local demand for air service. This alternative, as with the Proposed Project, may result in similar exceedances of the 1-hour standard for NO<sub>2</sub> projected at both OCX and JWA and the 24-hour PM<sub>10</sub> standard at OCX and JWA. Similar to the Proposed Project, Alternative A's construction emissions impacts would remain significant and unavoidable. Alternative A would also likely result in toxic air contaminant impacts similar to the Proposed Project.

#### Short-Term (Construction) Impacts

Under this alternative, no significant runway improvements would be made at JWA (ASMP Technical Report 6, Alternatives Definition Report, 1999). At the MCAS El Toro site, Runways 16L/34R and 7L/25R would be reconstructed to meet FAA parallel runway separation requirements for operations under visual conditions. Therefore, total construction emissions would be less than those of the Proposed Project; however, peak daily emissions, including both equipment exhaust and fugitive dust, would likely be similar to those of the Proposed Project. Therefore, this alternative would likely result in significant short-term construction impacts that cannot be mitigated below significance.

#### Operational Air Quality Impacts

#### **Emissions Inventories**

Project direct air pollutant emissions associated with airport operations, including aircraft, GSE, energy consumption, and vehicular trips, are shown below in Table 8.4-5 for this alternative. Regional air pollutant emissions, including airport operations at other airports in the region and VMT required for air travel passengers to get to these airports, are shown in Table 8.4-6 for this alternative. Compared to the Proposed Project, Alternative A would serve substantially less Orange County demand for aviation services; therefore, this alternative would result in a higher regional vehicle miles traveled on highways as passengers and cargo travel to other regional airports. This increase in VMT would result in higher regional air quality emissions for this alternative when compared to the Proposed Project. However, this alternative would generate less regional VMT and air quality emissions than the No Project/No Activity Alternative because this alternative would serve more locally generated demand in Orange County.

Table 8.4-5
Project Direct Air Pollutant Emissions (Pounds/Day) – Phase 4 Alternative A

	СО	NOx	ROC	SOx	PM <sub>10</sub>
Aircraft	10,976.22	10,574.57	912.07	720.08	121.75
OCX	4,864.63	8,579.53	607.54	559.26	89.30
JWA	6,111.59	1,995.04	304.53	160.82	32.45
GSE/APU	17,804.26	1,714.86	531.24	72.44	111.92
OCX	13,360.31	1,200.10	391.14	59.08	89.30
JWA	4,443.95	514.76	140.10	13.36	22.62
Fuel Storage/Dispensing			65.79		**
OCX			59.28		**
JWA	**		6.51	_	
Airport Roadways	429.59	85.98	19.59	5.67	5.61
OCX	345.90	76.03	16.73	4.71	4.78
JWA	83.69	9.95	2.86	0.96	0.83
Airport Parking	331.47	27.79	10.62	9.32	3.10
ocx	263.82	22.61	3.61	7.17	2.90
JWA	67.65	5.18	7.01	2.15	0.20
Energy Consumption	94.20	542.10	5.00	55.60	18.50
OCX	71.60	412.00	3.80	42.30	14.10
JWA	22.60	130.10	1.20	13.30	4.40
Vehicular Traffic	14,838	5,872	1,210	374	2,994
	13,266	5,158	1,900	<del>367</del>	2,611
OCX <sup>2</sup>	11,633	4,575	958	295	2,331
	10,061	4,575 3,861	848	288	1,948
JWA	3,205	1,297	252	79	663
Total	44,473.79	18,817.30	2,754.31	1,233.11	3,254.88
	42,902	<del>18,103</del>	2,644	1,230	2,872

Source: CH2M Hill and LSA Associates, Inc., 2001

<sup>1</sup> ROC emissions obtained by multiplying HC emissions reported by EDMS by a factor of 1.14.

<sup>&</sup>lt;sup>2</sup> SO<sub>X</sub> emissions are not reported by the URBEMIS7G model.

<sup>3</sup> Revised calculation of average trip length. This revision does not impact any of the significance determinations made in connection with the project.

Table 8.4-6
Regionwide Emissions Inventory Alternative A Phase 4
(Pounds/Day Unless Noted)

		CO	NOX	ROC	$SO_X$	PM <sub>10</sub>
Aircraft	El Toro	4,864.63	8,579.53	607.54	559.26	89.30
	JWA	6,111.59	1,995.04	304.53	160.82	32.45
	Other Airports	66,830.49	73,354.73	9,753.01	5,589.15	798.25
	Total Regional	77,806.71	83,929.30	10,665.08	6,309.23	920.00
GSE	El Toro	13,360.31	1,200.10	391.14	59.08	89.30
	JWA	4,443.95	514.76	140.10	13.36	22.62
	Other Airports	93,744.51	9,413.56	2,773.67	609.72	345.17
	Total Regional	111,548.77	11,128.42	3,304.91	682.16	457.09
Energy	El Toro	71.60	412.00	3.80	42.30	14.10
	JWA	22.60	130.10	1.20	13.30	4.40
	Others	579.00	3,331.00	31.00	340.90	114.00
	Total Regional	673.20	3,873.10	36.00	396.50	132.50
Fuel	El Toro	***	***	59.28	**	**
	JWA	****		6.51		
	Other Airports		~-	491.24		<del>-</del> -
	Total Regional			557.03		
Airport Roadways	El Toro	345.90	76.03	16.73	4.71	4.78
	JWA	83.69	9.95	2.86	0.96	0.83
	Other Airports	3,232.71	656.01	148.75	39.48	53.64
	Total Regional	3,662.30	741.99	168.34	45.15	59.25
Airport Parking	El Toro	263.82	22.61	3.61	7.17	2.90
	JWA	67.65	5.18	7.01	2.15	0.20
	Other Airports	2,020.74	580.44	27.89	53.66	21.18
	Total Regional	2,352.21	608.23	38.51	62.98	24.28
Roads	El Toro	11,633.00	4,575.00	958.00	295.00	2,331.00
		<del>10,061.00</del>	3,861.00	848.00	288,00	<del>1,948.00</del>
	JWA	3,205.00	1,297.00	252.00	79.00	663.00
	Others Airports2	<u>2,757,679.00</u>	490,576.00	70,624.00	48,634.00	<u>6,821.00</u>
		2,730,639.00	4 <del>85,975.00</del>	67,491.00	49,000.00	6,984.00
	Total Regional <sup>2</sup>	2,772,517.00	496,448.00	71,834.00	49,000.00	9,815.00
		2,743,905.00	491,133.00	68,591.00	49,367.00	9,995,00
TOTAL (pounds/o	lay)	2,968,560.19		86,603.87	56,496.02	11,408.12
		2,939,948.19	<del>591,414.04</del>	83,360.87	56,863.02	11,188.12
Change from 2020 N	o Project	(10,802.41)	(2,864.32)	(974.41)	(394.24)	(11.28)
(pounds/day)		(31,068.41)	(6,585.460)	(3,268.78)	(139.24)	(239.84)
SCAQMD Threshol (pounds/day)	d for Operation	550	55	55	150	150

Source: LSA Associates, Inc., 2001.

Revised calculation of average trip length. This revision does not impact any of the significance determinations made in connection with the project.

<sup>2</sup> Typographical correction.

#### Dispersion Analysis

No airport emissions dispersion analysis was conducted for this project alternative. Several local criteria pollutant hot spots for NO<sub>2</sub> and SO<sub>2</sub> were found under the Proposed Project. Although the Proposed Project has higher annual aircraft LTO operations, these local hot spots from aircraft exhaust emissions would also likely occur for Alternative A.

At intersections in the vicinity of the project sites, the CAL3QHC model was used to assess the CO concentrations. Tables 8.4-7 and 8.4-8 show that the 1-hour and 8-hour CO concentrations would be below the State and federal CO standards for Alternative A.

#### Toxic Air Contaminants

Under this alternative, toxic air contaminant impacts would likely be similar to those identified under the Proposed Project.

### 8.4.4.6 Topography

Compared to existing conditions, this alternative would have no significant impacts due to JWA operations, but would have impacts at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or lessen substantially the impacts of the Proposed Project.

Since development of MCAS El Toro under Alternative A is similar to that described for the Proposed Project, no significant impacts related to topography would occur.

Operation and development of JWA under Alternative A would be similar to the current usage, and would not entail expansion of the airport acreage. Therefore, Alternative A would not raise potential impacts related to topography. The impacts of Alternative A related to topography are slightly fewer than the impacts under the Proposed Project due to fewer grading requirements.

Table 8.4-7 Phase 4 Alternative A - Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

IMIN :		RECI		REC3	RECA	RECE		BECT.	BEC#	RECT.	RECIO"	RECIT"	BECIS
	CITY OF ORANGE®	3000											- X - B - Z - Lol - Z - Z
345	Jamboree & Chapman	7.0	7.0	7.2	7.0	7.0	7.2	6.8	7.2	6.8	6.8	6.7	6.7
	CITY OF SANTA ANA"												
154	MacArthur & Main	7.1	7.2	7.1	7.1	6.8	6.8	6.7	7.1	6.9	7,0	6.7	6.8
152	Main & Sunflower	7.0	7.1	6,7	7.1	6.6	7.0	6,5	6.5	6.7	6.8	7.1	6.8
90	Grand & Edinger	6.9	7.0	7.2	7.3	7.0	7.1	6.7	7.0	6.6	7.0	6.9	6.7
	CITY OF TUSTIN'S												
93	Newport & Edinger	7.2	7.1	6.8	7.2	6.7	6.7	6.6	6.7	6.7	7.0	6.6	6.9
115	Von Karman & Barranca	7.0	7.1	7.1	7.0	6.6	7.1	6.7	6.9	6.6	6.6	6.7	6.9
95	Tustin Ranch & Edinger	7.2	7,0	6.9	7. i	6.7	6.9	6.6	6.7	6.9	6.9	6.6	7.0
	CITY OF IRVINE"												
116	Jamboree & Barranca	5.8	6.0	5.6	5.6	5.2	5.5	5.3	5.5	5.5	5.4	5.5	5.8
156	Jamboree & Main	5.6	5.6	5.7	5.4	5.2	5.4	5.3	5,5	5.3	5.3	5.4	5.6
98	Culver & Irvine Center	5.6	5.6	5.7	5.7	5,3	5.4	5.4	5.5	5.3	5.5	5,4	5.5
134	Jamboree & Alton	5.6	5.6	5.7	5.6	5.2	5.5	5.2	5.3	5.3	5.4	5.4	5.7
175	Jamboree & Michelson	5.7	5.4	5.4	5.6	5.1	5.3	5.3	5.3	5.3	5.4	5.4	5.4
151	Red Hill & MacArthur	5.7	5.5	5.5	5.6	5.0	5.4	5.4	5.3	5.3	5.4	5.2	5.7
100	Jeffrey & Irvine Center	5.6	5.5	5.5	5.6	5.3	5.5	5.2	5.2	5.5	5.8	5,1	5,3
321	Access Rd. West & Irvine	5.2	5.2	5.2	5.4	\$.1	5.1	5.1	52	4.9	4.9	5.1	5.2
320	Perimeter Rd. & Irvine	5.3	5.2	5.2	5.1	4.9	4.9	5.0	5.2	4.8	4.8	5.1	5.1
153	Red Hill & Main	5.5	5.5	5.5	5.6	5.1	5.3	5.2	5.2	5.2	5.2	5.2	5.2
	CITY OF LAGUNA BEACH												
299	Moulton & El Toro	5.5	5.4	5.5	5.6	5.1	5.3	5.4	5.4	5,0	5.3	5.1	5.1
	CITY OF LAGUNA HILLS"												
280	El Toro & Avd. Carlota	5.4	5.4	5,4	5.3	5.0	5.2	5.0	5.0	5.2	5.2	5. i	5.3
	CITY OF LAKE FOREST'												
271	El Toro & Rockfield	5.6	5,5	5.6	5.5	5.2	5.2	5.2	5,4	5.2	5.3	5.3	5.4

\* - Concentrations are in parts per million (ppm)

<sup>1 -</sup> RECI SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> RECS W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient one-hour CO concentration, 6.1 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years 1996 to 2001, is added to the calculated one hour levels.

Table 8.4-8

Phase 4 Alternative A – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

NTV	Intersecting streets	RECI'	REC2	The second second	St. Wallet St. St.	RECS	Company of the Company of the Company	RECT	RECF	RECY	RECIO"	REC11"	REC12 <sup>11</sup>
22. (1.22)	CTTY OF ORANGE®	(	<u> </u>				32 Tr. 17 3 TG	<u> </u>	<i>**</i> **********************************	~8.46 × 80.17		X 10 00 100 AU	2022 1 12
145	Jamboree & Chapman	5,2	5.2	5.4	5.2	5.2	5.4	5.1	5.4	5.3	5.1	5.0	5.0
	CITY OF SANTA ANA"												
54	MacArthur & Main	5.3	5.4	5.3	5.3	5.1	5.1	5.0	5.3	5.2	5.2	5.0	5.1
52	Main & Sunflower	5.2	5.3	5.0	5.3	5.0	5.2	4.9	4.9	5.0	5.1	5.3	5.1
0	Grand & Edinger	5.2	5.2	5.4	5.4	5.2	5.3	5.0	5.2	5.0	5.2	5 2	5.0
	CITY OF TUSTIN"												
3	Newport & Edinger	5.4	5.3	5.1	5.4	5.0	5.0	5.0	5.0	5.0	5.2	5.0	5.2
15	Von Karman & Barranca	\$.2	5.3	5.3	5.2	5.0	5.3	5.0	5.2	5.0	5,0	5.0	5.2
)5	Tustin Ranch & Edinger	5.4	5.2	5.2	5.3	5.0	5.2	5.0	5.0	5.2	5 2	5.0	5,2
	CITY OF IRVINE												
16	Jamboree & Barranca	3.7	3,9	3.6	3.6	3.3	3.5	3.4	3.5	3.5	3.5	3.5	3.7
56	Jamboree & Main	3.6	3.6	3.7	3.5	3.3	3.5	3.4	3.5	3.4	3.4	3.5	3.6
8	Culver & Irvine Center	3.6	3.6	3.7	3.7	3,4	3.5	3.5	3.5	3.4	3.5	3.5	3.5
34	Jamboree & Alton	3.6	3.6	3.7	3.6	3.3	3.5	3.3	3,4	3.4	3.5	3.5	3.7
75	Jamboree & Michelson	3.7	3.5	3.5	3.6	3.3	3.4	3.4	3.4	3.4	3.5	3.5	3.5
51	Red Hill & MacArthur	3.7	3.5	3.5	3.6	3.2	3,5	3.5	3.4	3.4	3.5	3.3	3.7
100	Jeffrey & Irvine Center	3.6	3.5	3.5	3.6	3.4	3,5	3.3	3.3	3.5	3.7	3.3	3.4
321	Access Rd. West & Irvine	3.3	3.3	3.3	3.5	3.3	3.3	3.3	3.3	3.1	3.1	3,3	3.3
320	Perimeter Rd. & Irvine	3.4	3.3	3.3	3.3	3. i	3.1	3.2	3.3	3.0	3.0	3,3	3.3
153	Red Hill & Main	3.5	3.5	3.5	3.6	3.3	3.4	3.3	3.3	3.3	3.3	3.3	3.3
	CITY OF LAGUNA BEACH												
299	Moulton & El Toro	3.5	3.5	3.5	3.6	3.3	3.4	3.5	3.5	3.2	3.4	3.3	3.3
	CITY OF LAGUNA HILLS"												
280	El Toro & Avd. Carlota	3.5	3.5	3.5	3.4	3.2	3.3	3.2	3.2	3.3	3.3	3.3	3.4
	CITY OF LAKE FOREST"												
271	El Toro & Rockfield	3.6	3,5	3.6	3.5	3.3	3.3	3.3	3 5	3.3	3.4	3.4	3.5

Note: \* - Concentrations are in parts per million (ppm)

<sup>1 -</sup> REC1 SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> REC8 W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> RECIO S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient eight-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

<sup>14 -</sup> The ambient eight-hour CO concentration, 2.9 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7

#### 8.4.4.7 Soils, Geology and Seismicity

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to development at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance.

Development of MCAS El Toro under Alternative A is similar to that described for the Proposed Project, and does not significantly differ in its impacts related to soils or local geologic features. Alternative A also does not entail additional risk based on projected earthquake events beyond those discussed for the Proposed Project. The impacts of Alternative A related to seismicity are the same as the impacts under the Proposed Project.

Operation and development of JWA under Alternative A would be similar to the current usage, and would not entail expansion of the airport acreage. Therefore, Alternative A would not raise impacts related to soils, geologic features or seismicity.

The impacts of Alternative A related to soils, geology and seismicity would be the same as under the Proposed Project, and this alternative would not avoid or lessen substantially the impacts of the project.

#### 8.4.4.8 Hydrology and Water Quality

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to development at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the project, the impacts would be reduced to a level of insignificance. As discussed in Section 4.8 (Hydrology and Water Quality), the Proposed Project will not result in significant adverse impacts related to drainage and surface water quality. Because most issues related to drainage at the MCAS El Toro site can be addressed adequately through proper design and engineering, it is anticipated that Alternative A could also be developed for use as a civilian airport without significant adverse impacts related to drainage. Similarly, as discussed in Section 4.8, operations can be conducted and controls implemented to minimize potential adverse impacts related to surface water quality under Alternative A. Consequently, development of Alternative A would not result in significant impacts related to surface water quality.

No groundwater will be pumped from the MCAS El Toro site under this alternative so there will be no impacts to local groundwater levels or basin storage under this alternative. Groundwater quality impacts under this alternative will be the same as those discussed for the Proposed Project in Section 4.8. As with the Proposed Project, no hazardous waste remediation activities at the MCAS El Toro site are a component of this alternative. Therefore, this alternative will result in no significant adverse impacts related to groundwater.

Under this alternative, JWA will have a lower MAP level compared with current operations and therefore will require no major construction. Therefore, this alternative will not result in impacts related to hydrology and water quality beyond existing conditions at JWA.

In summary, the hydrology and water quality impacts of Alternative A will be similar to the level of impacts under the Proposed Project, and this alternative would not avoid or lessen substantially the impacts of the project.

#### 8.4.4.9 Biological Resources

When compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to development of the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. However, this alternative would not avoid or lessen the impacts of the Proposed Project.

The aviation land use and airfield component for the MCAS El Toro site and the corresponding nonaviation component under this alternative are very similar to those under the Proposed Project. The only substantive difference between Alternative A and the Proposed Project is that there are no plans for a runway extension under Alternative A. However, since the extension results in impacts to non-native or ruderal grassland, the difference in biological resource impacts are not significantly different for direct impacts (i.e. native plant communities, wildlife, wildlife dispersion corridors and special interest species). For indirect impacts, the biological resource impacts under Alternative A are not substantially different than for the Proposed Project. However, there is one identifiable indirect impact that is expected to be different and that is noise exposure to biological resources. For Alternative A, the CNEL noise contour is substantially shorter to the north. The CNEL noise contour differences to the east and the Habitat Reserve, and to the south and the San Joaquin Hills are not substantially different. SEL values are not expected to be substantially different from the Proposed Project. The shorter CNEL noise contours to the north reflect a lower average noise level from aircraft overflights at Siphon Ridge as compared to the Proposed Project. Although, this is an improvement from the Proposed Project, it is not anticipated to result in a substantially different level of biological productivity in the Siphon Ridge area. This alternative would have impacts similar to the Proposed Project on wetlands and Waters of the U.S. With the mitigation measures recommended for the project, the impacts of this alternative would be reduced to a level of insignificance.

This alternative will not result in significant adverse impacts to native plant communities, wildlife dispersion corridors, or special interest species at JWA or Upper Newport Bay. There are no substantive biological resources on the JWA site, and impacts to the Upper Newport Bay are limited to indirect impacts as a result of aircraft operations, which are less than the Proposed Project.

#### 8.4.4.10 Public Services and Utilities

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to development at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or lessen substantially the impacts of the Proposed Project.

Redevelopment of MCAS El Toro under Alternative A is similar to that of the Proposed Project, no significant unmitigated impacts related to public services would occur. The same conclusions are made for JWA, which will remain status quo for this alternative.

As described in Section 4.10 (Public Services and Utilities), the Proposed Project would not result in significant unmitigated adverse impacts related to utilities. Alternative A could be served with utilities without significant adverse impacts after mitigation, similar to conditions under the Proposed Project.

#### 8.4.4.11 Natural Resources and Energy

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have significant impacts at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance except for impacts to Agricultural Resources, which would remain significant after mitigation. This alternative would lessen the impacts on Prime Agricultural Soils by up to 440 acres compared to the Proposed Project. However, impacts would remain significant and unavoidable.

The land use/airfield component and nonaviation land use component for this alternative are similar to those of the Proposed Project. The primary exception is that no runway extensions are planned at the MCAS El Toro site under this alternative. As discussed in Section 4.11 (Natural Resources and Energy), the Proposed Project will not result in significant adverse impacts related to natural resources and energy, with the exception of unmitigatable significant impacts to agricultural resources on the MCAS El Toro site.

There are no agricultural resources existing at JWA; therefore, no impacts at JWA would result from this alternative.

# 8.4.4.12 Aesthetics, Light and Glare

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to construction and operations at the MCAS El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the aesthetic, light, and glare impacts of this alternative would be reduced

to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the Proposed Project.

The design of the facilities for Alternative A is similar to that of the Proposed Project, with an insignificant reduction of the size of the terminal, number of gates, and ancillary aviation support facilities. The Nonaviation Revenue Support uses would consist of the same facilities as proposed with the project, although the acreages for agricultural and passive open space would be larger by approximately 440 acres due to the reduced size of the airport facilities needed to serve 19 MAP. The overall appearance of the MCAS El Toro site, including the airport facilities (runways, terminal, cargo buildings, parking structures, etc.) and Nonaviation Revenue Support uses (regional park, golf courses, office/commercial and cultural and institutional buildings) would be similar to the appearance of the Proposed Project. Views of the MCAS El Toro site from the vantage points described in Section 4.12 would not differ substantially from the views created by the development of the Proposed Project. In that there is less development and more open space/green space on the site due to the less intense commercial passenger service plan under Alternative A, this alternative would have slightly less visual change than those of the Proposed Project; however, the reduction would be insignificant.

The impacts of light and glare at the MCAS El Toro site under this alternative would approximate those of the Proposed Project.

At JWA, Alternative A would slightly reduce the commercial service level from the existing level and, therefore, any visual changes caused by either the Proposed Project or Alternative A would be slightly less than existing conditions. Potential light and glare at JWA would be similar to that of the Proposed Project; no substantive change in this effect would occur.

#### 8.4.4.13 Cultural Resources

Compared to existing conditions, this alternative would have no effect on cultural resources at JWA, but would have impacts due to development at the MCAS El Toro site similar to the Proposed Project. This alternative would not avoid or substantially lessen the impacts of the Proposed Project.

The physical effects of this alternative on cultural resources at MCAS El Toro would be approximately the same as with the Proposed Project. As the cultural resources within the disturbance area (construction and operations) are not considered significant (SHPO concurrence pending), no significant impacts would be caused by this alternative, as with the Proposed Project.

Under Alternative A, there would be no additional or new effects on cultural resources at JWA since there is no known archaeological, paleontological or historic resources on the already developed airport property.

#### 8.4.4.14 Recreation

Compared to existing conditions, this alternative would have fewer impacts from operations at JWA, because the primary project development would take place at the MCAS El Toro site. Alternative A would have impacts due to development at the MCAS El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen impacts of the Proposed Project.

Alternative A would have approximately the same effects related to recreational resources in the MCAS El Toro area as the Proposed Project. The physical boundaries of construction with Alternative A would be, for all practical purposes, the same as that of the Proposed Project. Therefore, no adjacent off-road trails would be physically impacted with Alternative A. Development at the MCAS El Toro site under Alternative A would have the same effect regarding consistency with County and City General Plan Recreational policies, goals and objectives, in that nonaviation land uses would be included on-site similar to the Proposed Project. In addition, Alternative A would not differ from the project in that it would not exceed Thresholds of Significance 2 and 3 in Section 4.14.

Noise impacts at MCAS El Toro under Alternative A would be reduced from those of the Proposed Project because the noise contours would be reduced. Existing recreational facilities and planned future facilities in the 65 dB CNEL contour for Alternative A would include approximately the same facilities as the project contour, with the potential for the use of fewer facilities or smaller portions of the same facilities to be affected. The overall noise impact on the use of area recreational facilities would be similar to the impacts of the Proposed Project, given that the alternative calls for the same type of uses on the site, at a reduced intensity (28.8 MAP for the project, 19 MAP for Alternative A).

The physical effects on area recreational facilities in the JWA area under Alternative A would be approximately the same as under the Proposed Project. Similarly, no significant differences in noise effects on public use of area recreational facilities would occur in that the 65 dB CNEL noise contour for JWA in Alternative A would be approximately the same.

# 8.4.4.15 Public Health and Safety

Compared to existing conditions, the overall potential for accidents is greater with this alternative. The increase in accident potential is not deemed to be significant as an extraordinary risk is not created. This alternative would not avoid or substantially lessen impacts of the Proposed Project.

# Aviation Safety

Compared to the Proposed Project, there would be an increase of approximately 7,600 air carrier and air cargo operations and a decrease of approximately 2,000 general aviation

operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 11.3% to reflect the number of increasing aviation activity diverted from OCX to JWA and the potential accident risks for general aviation at JWA would slightly decrease by 0.6% correspondingly. At OCX, there would be an estimated decrease of 59,100 air carrier and air cargo operations and an estimated increase of 11,000 general aviation operations. Under this condition, the potential air carrier and air cargo accident risks at OCX would decrease by approximately 21.3% to reflect the fewer number of operations at OCX. The potential general aviation risks at OCX would increase by 50.0% correspondingly. Compared to the Proposed Project relative to on-airport and off-airport fatal accidents per million operations, there would be no significant adverse impacts related to aviation safety at the MCAS El Toro site or at JWA.

Compared to the existing conditions, there would be a decrease of approximately 14,908 air carrier and air cargo operations and an increase of approximately 29,376 general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would decrease by approximately 16.6% to reflect the number of decreasing operations and the potential general aviation accident risks would increase by 9.0% correspondingly. Compared to the existing conditions, there would be no significant adverse impacts related to aviation safety at JWA.

#### 8.4.4.16 Hazardous Materials and Hazardous Wastes

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to development at the El Toro site similar to the Proposed Project. With mitigation measures, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the Proposed Project.

Implementation of Alternative A would result in impacts related to hazardous wastes approximately the same as under the Proposed Project. This alternative would not alter remedial investigations, response actions or environmental risks associated with any hazardous waste sites on the MCAS El Toro and JWA sites.

Any use of hazardous materials and/or generation of hazardous waste under Alternative A would be regulated by applicable State law, federal law, and regulations pertaining to worker protection, hazardous materials storage and use, and hazardous waste generation and disposal. Implementation of these regulations will reduce potential impacts associated with the presence of these hazardous substances to below a level of significance.

The impacts of Alternative A related to hazardous materials and hazardous wastes are approximately the same as the impacts under the Proposed Project.

#### 8.4.4.17 Socioeconomics

Under this alternative, a total of 22,900 jobs would be generated, including 19,200 at MCAS El Toro and 3,700 at JWA, representing a net increase of 16,500 jobs at MCAS El Toro and 1,600 jobs at JWA over existing 1998 conditions. The distribution of jobs between MCAS El Toro and JWA differs under this alternative compared to the Proposed Project. There would be fewer jobs generated at the MCAS El Toro site under Alternative A than under the Proposed Project. Employment at JWA would be marginally higher under Alternative A than under the Proposed Project.

As with the Proposed Project, economic activity at the El Toro site and JWA site, as well as expenditures by visitors arriving by air through the two airports, would stimulate additional off-site job growth. Given the lower number of on-site jobs and air passengers served by this alternative, the number of off-site jobs stimulated by the airport system would be significantly lower than the level under the Proposed Project.

Given the lower number of jobs generated under Alternative A compared to the Proposed Project, the magnitude of impacts related to inducing substantial growth or concentration of employment, and demand for housing, including low and moderate income housing, would be lower than under the Proposed Project. The employment projections under Alternative A would also be inconsistent with the adopted regional forecasts, as under the Proposed Project.

In summary, this alternative would not avoid or substantially lessen the impacts of the Proposed Project.

# 8.4.4.18 Risk of Upset

Compared to existing conditions, this alternative would have no significant impacts due to operations at JWA, but would have impacts due to development at the El Toro site similar to the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

As discussed in Section 4.18 (Risk of Upset), the Proposed Project would not result in significant adverse impacts to public health and safety resulting from project-related risk of upset conditions. The ultimate build out and phased development of this alternative will entail a lower level of operations than the Proposed Project, with a commensurate lower level of risk of upset potential associated with jet fuel storage and delivery. Consequently, implementation of this alternative would not result in significant adverse impacts to public health and safety.

# 8.4.5 Feasibility

In summary, this alternative, which is very similar to the Proposed Project, would have the same development and environmental feasibility. However, this alternative would have slightly lower development cost due to the reduced extent of terminal and related aviation facilities, and lower revenues due to reduced aviation use.

#### 8.4.6 Conclusions

Alternative A would reduce the area affected by the 65 dB CNEL for OCX, and would reduce traffic, and loss of agricultural soils impacts compared to the Proposed Project. This alternative would have greater regional air quality impacts due to increased regional traffic to airports meeting the County's unmet air service demand. In conclusion, Alternative A:

- Does not meet the County's future demand for aviation services, especially international service. This would have an adverse impact on trade, business, tourism, jobs, and other economic activity in the County.
- Would result in higher regional VMT and regional air quality emissions as passengers and cargo travel from Orange County to LAX or other airports.
- Since the 65 dB CNEL for LAX and other airports (which would serve the County's unmet aviation demand) already include large numbers of noise sensitive populations/ developments, this alternative would increase the adverse effects of aviation noise on a regional basis.
- Would generate fewer on-site and off-site jobs than the Proposed Project.
- Would not result in a significant reduction in project impacts (after mitigation), and would not avoid project impacts that cannot be reduced to a level of insignificance, including short-term construction air quality impacts, local air quality impacts due to aircraft and associated operations, and toxic air contaminant impacts.

# 8.5 ALTERNATIVE C: JWA – SHORT-HAUL; OCX – MEDIUM-HAUL TO FULL INTERNATIONAL (LINKED) AIR SERVICE

This section presents the potential impacts of Alternative C as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

This alternative was selected for analysis because it has the potential to lessen aircraft noise, traffic, and local air quality emissions at OCX while still feasibly attaining most of the objectives of the Proposed Project. The Notice of Preparation (NOP) identified this alternative and the Proposed Project as preferred projects to be analyzed in the Draft EIR. However, the analysis of this option determined that this alternative is infeasible due to costs as described below. Therefore, this plan is analyzed as an alternative that was considered.

#### 8.5.1 Aviation Uses

Under this alternative, JWA and OCX would ultimately (year 2020) be linked by an airport-to-airport connector such as a light rail system that would allow the two airports to function as one for connecting passengers. Without this connector, market segmentation between the two airports is not feasible. The market roles of the two airports would include regulatory perimeter rules defining their respective permitted roles. Under Alternative C, OCX provides long-haul domestic and international air passenger service for an estimated 23.4 MAP, 22 percent (5.1 MAP) of which are passengers with connecting flights (45 percent of these connecting passengers transfer between JWA and OCX via the transit facility that is proposed to link the two airports). OCX is also forecast to annually handle approximately 0.84 million tons of international cargo and 1.18 million tons of domestic cargo. The 2020 service level at JWA under this alternative would be 10.1 MAP. This alternative includes a proposed on-airport 500-room hotel in the OCX terminal area. JWA would serve general aviation activity and short-haul passengers. The runway improvements at OCX would be the same as under the Proposed Project. Figure 8-4 depicts Alternative C.

# 8.5.2 Nonaviation Revenue Support Uses

Nonaviation land uses proposed under Alternative C are the same as assumed for the Proposed Project.

# 8.5.3 Attainment of Project Objectives

This alternative meets the general project objectives for base reuses except to optimize cost/revenues, as well as the aviation related objectives. However, the very high costs/passenger for the JWA/OCX transit connector would result in an infeasible project.

# 8.5.4 Environmental Impacts of Alternative C

#### 8.5.4.1 Land Use

Compared to existing conditions, this alternative would have greater land use impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project and additional mitigation related to noise impacted land uses around JWA, the impacts of this alternative could be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

Compared to the Proposed Project, the primary difference is that Alternative C includes an airport-to-airport transit connector. The selected option was a fully elevated fixed-guideway system from OCX along SR-133 to I-405, along the I-405 right-of-way to MacArthur Boulevard, and then to JWA. The land use impacts of this alternative were found to be low, with the assumption that a large portion of the system can be provided within existing right-of-way.

Uses along the perimeter of OCX are generally comparable in intensity or less intense than the existing and planned adjacent off-site uses. Therefore, as with the Proposed Project, there are no significant land use conflicts associated with the proposed land uses under this alternative. The elevated nature of the airport connector means that it will be visually prominent, and the connector would be expected to generate noise and vibration effects on adjacent land uses.

The on-site agricultural uses preserved under this alternative will be the same as is preserved with the Proposed Project. As with the Proposed Project, the agriculture impacts that might occur can be controlled through agricultural management practices and through the terms of the County's lease agreements.

The proposed airport use at MCAS El Toro under Alternative C will attract new development in nearby areas. There is a potential for undesirable land use development (such as sexually oriented businesses) in the vicinity of the site, unless the County and adjacent cities have adequate land use controls in place. Also, the design of future off-site development may adversely affect existing and planned development in the adjacent jurisdictions if appropriate design standards are not implemented by the local jurisdictions. This potential impact is the same under both Alternative C and the Proposed Project.

Although the JWA aviation activity under Alternative C is higher than the Proposed Project, major future aviation and terminal improvements would be limited since the existing facility was designed to accommodate 10.1 MAP. Also, future improvements under Alternative C would take place within the existing boundaries of the airport. This alternative would create a JWA 65 CNEL noise contour, which would be larger than the Proposed Project (see noise analysis below). With the current JWA noise mitigation program, these impacts would be reduced to a level of insignificance.

The impacts of Alternative C related to land use are generally comparable to the impacts under the Proposed Project at El Toro. The airport-to-airport connector proposed under Alternative C would have aesthetic, noise, and vibration impacts on adjacent uses. However, these impacts would be principally limited to the adjacent freeway or highway right-of-way. The land use impacts around JWA due to a larger 65 CNEL noise contour would be greater than the Proposed Project, which has a 65 CNEL contour smaller than that existing today.

#### 8.5.4.2 General Plan Consistency

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the Proposed Project impacts.

Alternative C introduces a civilian aviation use to MCAS El Toro and modifies existing aviation activity conditions at JWA; therefore, as with the Proposed Project, an amendment to the AELUP is required. An amendment to the Orange County General Plan Land Use Element map is needed for this alternative, to address the conflicts with proposed land uses in Planning Area 5, and to redesignate the Open Space portion at the south end of JWA to Public Facilities. The adopted 65 dB CNEL noise contour policy implementation line would change at El Toro as a result of this alternative, therefore, an amendment to the Orange County General Plan Noise Element would be required. Alternative C includes land uses which conflict with the adopted City of Irvine General Plan for Planning Area 7 (City of Irvine Planning Area 30). An amendment to the City of Irvine General Plan would not be required since the site would be owned by the County. The need for amendments to General Plans and the AELUP for Alternative C are comparable to those required for the Proposed Project, therefore; the General Plan consistency impacts for Alternative C are similar to the Proposed Project.

# 8.5.4.3 Transportation and Circulation

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the project alternative mitigation measures, the impacts of this alternative would be reduced to a level of

insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

The AM and PM peak hour and ADT trips generated by the aviation operations at JWA and OCX and by nonaviation revenue support land uses with build out of Alternative C are summarized in Table 8.5-1. Refer to Section 10.0 in the 1999 Traffic Analysis Technical Report for detailed information on the methodology used to produce trip generation estimates for Alternative C. This alternative would generate an increase of 126,873 ADT at the El Toro site compared to an increase of 150,723 ADT for the Proposed Project over existing conditions. The alternative would generate 152,967 ADT less than the CRP. At JWA, this alternative would generate 1,426 ADT more than existing conditions. For a comparison of peak hour trip generation, see Table 4.3-8. In summary, this alternative would substantially reduce the number of vehicle trips generated at the El Toro site compared to the Proposed Project. However, after mitigation measures are applied, the Proposed Project and this alternative would have no significant adverse impacts.

Table 8.5-1
Trip Generation Summary – Alternative C

	AM Peak Hour			PM Peak Hour				Existing	
Project Component	- In	Out	Total	∮ In	Out	Total	ADT	ADT	
Former MCAS El Toro Site									
OCX Commercial Air Terminal	2,771	1,847	4,618	2,786	2,747	5,533	91,751	***************************************	
OCX Air Cargo Handling Facility	565	707	1,272	790	513	1,303	17,503		
Nonaviation Revenue Support			***************************************				43,019		
Land Uses	3,692	728	4,420	989	3,573	4,562			
Sub-Total (El Toro Site)	7,028	3,282	10,310	4,565	6,833	11,398	152,273	25,400	
JWA	1,468	982	2,450	1,930	1,931	3,861	48,876	47,450	
TOTAL	8,496	4,264	12,760	6,495	8,764	15,259	201,149	72,850	

The on-site and site access circulation plans assumed for JWA and OCX in Alternative C are the same as those described in Section 4.3 (Transportation and Circulation) for the Proposed Project. Peak hour levels of service with and without Alternative C were compared in order to identify the locations on the existing plus committed circulation system that require project related improvements to mitigate the traffic impacts of Alternative C and other foreseeable growth or development. Table 8.5-2 compares, in summary, the Alternative C highway impacts to the existing conditions and existing conditions plus Proposed Project. As discussed in Section 4.3.6.5, there is minimal comparison between the existing conditions plus Proposed Project versus the Alternative C impacts due to highway improvements recently completed and the effects of committed highway improvements.

Table 8.5-2
Summary Comparison of Traffic Impacts for Alternative C to
Existing Conditions and Existing Conditions Plus Project

Existing Conditions	Existing Conditions	Alternative C With Existing
Deficient Highway Facilities	Plus Proposed Project Impacts	Plus Committed Facilities
Location	Location	Location
INTERSECTIONS	IMPACTED INTERSECTION	IMPACTED INTERSECTIONS
Newport (NB) & Del Mar	Bake & Portola	Main & Sunflower
El Toro & SR-73 NB Ramps	Sand Canyon & Trabuco	ETC East Leg NB & Irvine
Campus & N. Bristol	Bake & I-5/I-405 SB Ramps	Sand Canyon & Irvine
Jamboree (SB) & Walnut	Bake & Rockfield	Sand Canyon & Trabuco
Jamboree & I-405 NB Ramps	Jeffrey & Alton	Jeffrey & Irvine
Jeffrey & I-405 NB Ramps	Jeffrey & I-405 NB Ramps	Jeffrey & Trabuco
Red Hill & MacArthur	Jeffrey & I-405 SB Ramps	MacArthur & Main
Irvine Center & Lake Forest	Jeffrey & Walnut/I-5 SB	Red Hill & Main
Bake & Jeronimo	Sand Canyon & I-5 NB Ramps	Sand Canyon & I-5 NB Ramps
El Toro & Avd Carlota	Sand Canyon & I-5 SB Ramps	Sand Canyon & I-5 SB Ramps
La Paz & Cabot/I-5 SB	Irvine Center & Lake Forest	MacArthur & Campus
Los Alisos & Muirlands	Bake & Irvine/Trabuco	Alicia & Paseo Valencia
Alicia & Jeronimo	Bake & Toledo	El Toro & Rockfield
Alicia & Muirlands	Los Alisos & Muirlands	Alicia & Jeronimo
La Paz & Muirlands/I-5 NB	Alicia & Jeronimo	Red Hill & I-5 NB Ramps
Red Hill & Edinger	Newport & Old Irvine	
Red Hill & Sycamore		IMPACTED ARTERIAL ROADS
Red Hill & Walnut	IMPACTED ARTERIAL ROADS	Irvine (ETC East Leg to PA-2 West Access Road)
	Laguna Canyon (I-405 to SJHTC)	Irvine (Jeffrey to Sand Canyon)
ARTERIAL ROADWAYS	Laguna Canyon (south of El Toro)	Laguna Canyon (south of El Toro)
Portola (Sand Canyon to Foothill Toll Road)	Culver (Bryan to Trabuco)	
Laguna Canyon (I-405 to SR-73)		IMPACTED FREEWAY/ TOLLWAY RAMPS
Laguna Canyon (south of El Toro)	IMPACTED FREEWAY RAMPS	FTC (SR-241) at Portola East (NB Off-Ramp)
Culver (Bryan to Trabuco)	I-5 at Culver (SB Off-Ramp)	I-5 at Alton (NB Direct On-Ramp)
Michelson (Carlson to Harvard)	I-5 at Sand Canyon (NB On-Ramp)	I-5 at Jamboree (NB Off-Ramp)
77.	I-5 at Sand Canyon (SB Off-Ramp)	I-5 at La Paz (SB Off-Ramp)

Existing Conditions Deficient Highway Facilities	Existing Conditions Plus Proposed Project Impacts	Alternative C With Existing Plus Committed Facilities
FREEWAY RAMPS	I-405 at Sand Canyon (NB Direct On-Ramp)	I-5 at Red Hill (SB On-Ramp)
I-5 at Culver (SB Off-Ramp)		I-5 at Sand Canyon (SB Off-Ramp)
I-405 at Jamboree (SB Off-Ramp)	IMPACTED FREEWAY SEGMENTS	I-405 at Sand Canyon (NB Direct On-Ramp)
I-405 at MacArthur (SB On-Ramp)	I-5 (Jeffrey to north of SR-55)	
I-405 at MacArthur (NB On-Ramp)		IMPACTED FREEWAY/ TOLLWAY SEGMENTS
I-405 at MacArthur (NB Off-Ramp)		FTC (Alton to south of Portola East)
SR-55 at Dyer (SB On-Ramp)		I-5 (I-405 to north of SR-55)
SR-55 at Dyer (NB Off-Ramp)		I-405 (Jamboree to SR-55)
SR-55 at MacArthur (SB Direct On- Ramp)		
SR-55 at MacArthur (NB Direct On-Ramp)		
SR-55 at MacArthur (SB Off- Ramp)		
FREEWAY SEGMENTS		
I-5 (Culver to north of SR-55)		
I-5 (Alton to I-405)	<u> </u>	
I-5 (El Toro to La Paz)		
I-405 (MacArthur to SR-133)	-	
SR-55 (I-5 to SR-73)		

Abbreviations:

NB-northbound

EB-eastbound

SB-southbound

WB-westbound

Table 8.5-3 summarizes the intersection locations, arterial roads and freeway ramps which are significantly impacted by Alternative C at build out (refer to Section 10.0 in the Traffic Analysis Technical Report for detailed summaries of the Alternative C traffic volumes and level of service (LOS) and comparisons between existing plus committed conditions with and without Alternative C) for intersections and arterial roadways within the traffic analysis study area, and refer to Section 10.0 in the 2001 Traffic Analysis Technical Report Addendum for comparable information for freeway/tollway mainline segments and freeway/tollway ramps within the traffic analysis study area).

Table 8.5-3
Alternative C Impact Summary

Location	Jurisdiction	Location **	Jurisdiction 🕒
IMPACTED INTERSECTIONS			
Main & Sunflower	Costa Mesa/ Santa Ana	Sand Canyon & I-5 NB Ramps	Irvine
ETC East Leg NB & Irvine	County	Sand Canyon & I-5 SB Ramps	Irvine
Sand Canyon & Irvine	County	MacArthur & Campus	Irvine/ Newport Beach
Sand Canyon & Trabuco	County	Alicia & Paseo Valencia	Laguna Hills
Jeffrey & Irvine	County/Irvine	El Toro & Rockfield	Lake Forest
Jeffrey & Trabuco	County/Irvine	Alicia & Jeronimo	Mission Viejo
MacArthur & Main	Irvine	Red Hill & I-5 NB Ramps	Tustin
Red Hill & Main	Irvine		
IMPACTED ARTERIAL ROADS			
Irvine (ETC East Leg to PA-2 West Access Rd)	County	Laguna Canyon (south of El Toro)	County/ Laguna Beach
Irvine (Jeffrey to Sand Canyon)	County		
IMPACTED FREEWAY/TOLLW	AY SEGMENTS		
FTC (Alton to south of Portola East)	Caltrans/TCA	I-405 (Jamboree to SR-55)	Caltrans
I-5 (I-405 to north of SR-55)	Caltrans		
<b>IMPACTED FREEWAY/TOLLW</b>	AY RAMPS		
I-5 at Alton (NB Direct On-Ramp)	Caltrans/Irvine	I-5 at La Paz (SB Off-Ramp)	Caltrans/ Laguna Hills
I-5 at Jamboree (NB Off-Ramp)	Caltrans/Irvine	FTC (SR-241) at Portola East (NB Off-Ramp)	Caltrans/TCA/ Lake Forest
I-5 at Sand Canyon (SB Off-Ramp)	Caltrans/Irvine	I-5 at Red Hill (SB On-Ramp)	Caltrans/Tustin
I-405 at Sand Canyon (NB Direct On-Ramp)	Caltrans/Irvine		

A comparison of Alternative C to the Proposed Project during the phasing years may also be made. As discussed in detail in Section 4.3.6.6 of this Draft EIR No. 573, as supplemented, under the Proposed Project phasing years, four intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 1 conditions (2005), five intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 2 conditions (2010), and nine intersection

locations, two arterial roadway segments, one continuous freeway mainline segment and two freeway ramps would be significantly impacted under Phase 3 conditions (2015). At Phase 4 build out, the Proposed Project would result in significant impacts not previously identified to four freeway/tollway mainline segments and four freeway/tollway ramps. See Draft Supplemental Analysis, Section 4.3.6.5. In each case, however, the identified impacts will be mitigated to a level below significant during the applicable phasing year (see Section 4.3.7.2, Table 4.3-20).

#### 8.5.4.4 Noise

Compared to existing conditions, Alternative C would create a greater noise impact at JWA than currently exists because of the forecast increase in use of the airport under this alternative. Alternative C would increase the 60 and 65 CNEL John Wayne Airport contours somewhat but not to the extent where they exceed those of the 1985 Master Plan contours (EIR No. 508). Table 8.4-3 shows a land use comparisons between noise contours for 1998 military and year 2020 alternatives for El Toro, and Table 8.4-4 shows land use comparisons between noise contours for 1998 and year 2020 alternatives for John Wayne Airport. The number of residences inside the 60 to 65 dB CNEL contour at JWA is 1,023 compared to the Proposed Project level of 577 and the 1998 existing condition of 682. The number of residences inside the 65 dB CNEL contours for those three scenarios is 236, 79, and 134, respectively. Figure 8-5 illustrates noise contours for Alternative C at El Toro. This alternative would not avoid or substantially lessen the impacts of the project.

The Alternative C 65 CNEL contour line would include 9.2 square miles of land for OCX and 1.07 square miles of land for JWA. The 65 CNEL for the existing military aircraft operations at MCAS El Toro include 6.3 square miles of land and for JWA, the existing conditions include 0.75 square miles of land. Therefore, Alternative C would increase the area affected by the 65 CNEL surrounding the El Toro site by 2.9 square miles, compared to an increase of 3.5 square miles for the Proposed Project. At JWA, Alternative C would increase the area affected by the 65 CNEL by 0.32 square mile, compared to 0.05 square mile for the Proposed Project.

The Proposed Project would increase noise sensitive land uses within the OCX 65 CNEL by three churches and one private school compared to existing conditions. However, this alternative would avoid the impacts of the Proposed Project on the three churches, but the private school would still be affected by the 65 dB CNEL. In general, the 65 CNEL line for this alternative is located within the much larger (28.8 square mile) MCAS El Toro AICUZ 65 CNEL. However, the Alternative C 65 CNEL line does exceed the AICUZ 65 CNEL boundary north of the El Toro site and, therefore, Alternative C has the same impacts of the Proposed Project outside the AICUZ 65 CNEL line.

EIR No. 563 concluded that a civilian airport at MCAS El Toro would result in significantly greater number of total operations compared to historical military levels of use, both throughout the day and during the nighttime hours. Although the Proposed Project and

Alternative C would have significantly fewer operations than the Community Reuse Plan analyzed in EIR No. 563, the number of forecast civilian operations is still substantially greater at El Toro than the existing conditions level of military operations.

As discussed earlier, the CNEL calculation factors in the number of daily operations and assigns a "penalty weighting" to operations occurring during the nighttime hours (10 p.m. to 7 a.m.). However, the substantial increase in the number of operation, particularly during nighttime hours, may be considered a significant impact of Alternative C independent of the CNEL computation.

The noise levels identified for the Proposed Project as well as Alternative C will be considered an annoyance by some residents and nighttime events will cause some sleep disturbance regardless of the levels of significance prescribed by regulatory agencies. Therefore, a mitigation measure for sleep disturbance is proposed in Section 4.4. With this mitigation measure, Alternative C impacts are reduced but remain significant similar to the Proposed Project.

# 8.5.4.5 Air Quality

Alternative C would have greater impacts due to JWA operations, but would have fewer impacts at the El Toro site than under the Proposed Project. The local and regional impacts of this alternative would likely be similar to the Proposed Project. Construction impacts would also likely remain significant and unavoidable, similar to the air quality impacts identified for the Proposed Project. Air toxics impacts would also be similar to those under the Proposed Project. This alternative would not avoid or substantially lessen the air quality project impacts.

# Short-Term (Construction) Impacts

Total on-site construction emissions under Alternative C would be similar to those of the Proposed Project. Construction of an airport to airport connector system under this alternative would add to the total project construction emissions, but may not increase the total peak daily emissions depending on the construction scheduling. Nevertheless, Alternative C would result in significant unavoidable short-term construction emissions impacts similar to the Proposed Project.

# Operational Air Quality Impacts

#### **Emissions Inventories**

Direct air pollutant emissions associated with airport operations, including aircraft, GSE, energy consumption, and vehicular trips, are shown in Table 8.5-4 for this alternative. Air pollutant emissions under this project alternative are very similar to those under the

Proposed Project, due to similar total number of air travel passengers projected.<sup>1</sup> Although project site emissions at OCX are larger than the No Project/No Activity Alternative and the Nonaviation Plan Alternative, like the Proposed Project, this alternative would reduce regional VMT compared to No Project conditions because more of the demand would be serviced in Orange County. With lower regional VMT, this alternative would result in lower total regional emissions than the No Project or ETRPA Alternative. See Table 8.5-5.

Table 8.5-4
Phase 4 Alternative C – Project Direct Air Pollutant Emissions (Pounds/Day)

	CO	NOx	ROC <sup>I</sup>	SOx	PM <sub>10</sub>
Aircraft	12.457.43	14,964.43	1,249.39	966.37	157.78
OCX	5,294.64	12,038.99	773.56	690.50	102.62
JWA	7,162.79	2,925.44	475.83	275.87	55.16
GSE/APU	19,532.10	1,732.62	568.00	67.81	73.02
OCX	10,975.79	1,096.30	342.46	50.98	47.01
JWA	8,556.31	636.32	225.54	17.73	26.01
Fuel Storage/Dispensing	-		83.46		MA JOS
OCX			72.50	***	***
JWA			10.96		
Airport Roadways	661.56	141.48	213.35	8.67	8.99
OCX	540.79	126.57	209.14	8.02	7.76
JWA	120.77	14.91	4.21	0.65	1.23
Airport Parking	411.07	34.65	14.36	11.24	3.45
OCX	314.59	27.22	4.37	8.19	3.17
JWA	96.48	7.43	9.99	3.05	0.28
Energy Consumption	126.10	726.50	6.80	74.50	24.80
OCX	88.10	507.50	4.70	52.00	17.30
JWA	38.00	219.00	2.10	22.50	7.50
Vehicular Traffic3	17,145	6,802	1,392	446	3,470
	15,573	6,088	1,282	440	3,087
OCX3	12,826	5,055	1,052	339	2,576
_	11,254	4,341	942	339 333	2,193
JWA	4,319	1,747	340	107	894
Total	50,333.26	24,401.68	3,527.36	1,574.59	3,738.04
	48,761	21,955	3,417	1,569	3,355

Source: CH2M Hill and LSA Associates, Inc., 2001

County of Orange Final EIR No. 573

ROC emissions obtained by multiplying HC emissions reported by EDMS by a factor of 1.14

SO<sub>x</sub> emissions are not reported by the URBEMIS7G model.

Revised calculation of average trip length. This revision does not impact any of the significance determinations made in connection with the project.

For a more detailed emissions inventory discussion, please see the Proposed Project discussion in Chapter 2.0 of this supplemental analysis.

Table 8.5-5
Regionwide Emissions Inventory Alternative C Phase 4
(Pounds/Day Unless Noted)

		CO	$NO_X$	ROC	SOX	PM <sub>10</sub>
Aircraft	El Toro	5,294.64	12,038.99	773.56	690.50	102.62
	JWA	7,162.79	2,925.44	475.83	275.87	55.16
	Other Airports	66,392.26	72,068.29	9,570.54	5,484.71	788.94
	Total Regional	78,849.69	87,032.72	10,819.93	6,451.08	946.72
GSE	El Toro	10,975.79	1,096.30	342.46	50.08	47.01
	JWA	8,556.31	636.32	225.54	17.73	26.01
	Other Airports	91,932.50	9,231.57	2,720.06	597.93	338.51
	Total Regional	111,464.60	10,964.19	3,288.06	665.74	411.53
Energy	El Toro	88.10	507.50	4.70	52.00	17.30
	JWA	38.00	219.00	2.10	22.50	7.50
	Others	544.00	3,132.00	29.00	319.00	107.00
	Total Regional	670.10	3,858.50	35.80	393.50	131.80
Fuel	El Toro	**	++	72.50		
	JWA	***		10.96		***
	Other Airports			481.74		***
	Total Regional	***		565.20		**
Airport Roadways	El Toro	540.79	126.57	209.14	8.02	7.76
*	JWA	120.77	14.91	4.21	0.65	1.23
	Other Airports	3,170.27	643.34	145.87	68.30	52.60
	Total Regional	3,831.83	784.82	359.22	76.97	61.59
Airport Parking	El Toro	314.59	27.22	4.37	8.19	3.17
	JWA	96.48	7.43	9.99	3.05	0.28
	Other Airports	1,981.71	170.63	27.35	52.63	20.77
	Total Regional	2,392.78	205.28	41.71	63.87	24.22
Roads	El Torol	12,826.00	5,055.00	1,052.00	339.00	2,576.00
		11,254.00	4,341.00	942.00	333.00	2,193.00
	JWA	4,319.00	1,747.00	340.00	107.00	894.00
	Others Airports2	2,755,094.00	489,631.00	70,441.00	48,560.00	<u>6,344.00</u>
		2,722,511.00	483,968.00	66,692.00	48,996.00	<del>6,445.00</del>
	Total Regional <sup>2</sup>	2,772,239.00	496,433.00	71,833.00	48,996.00	<u>9,814.00</u>
		<del>2,738,084.00</del>	490,056.00	67,974.00	49,436.00	9,532.00
TOTAL (pounds/day	y)	2,969,448.00	599,278.51	86,942.92	56,647.16	11,389.86
		2,935,293.00	592,901.51	83,083.92	<del>57,087.16</del>	11,107.86
Change from 2020 No	Project	(9,914.60)	(314.85)	(635.36)	(243.10)	(29.54)
(pounds/day)		(35,723.60)	(5,097.99)	(3,545,73)	84.90	(320.11)
SCAQMD Threshold for Operation (pounds/day)		550	55	55	150	150

Source: LSA Associates, Inc., 2001.

Revised calculation of average trip length. This revision does not impact any of the significance determinations made in connection with the project.

<sup>2</sup> Typographical correction.

# Dispersion Analysis

The Proposed Project would result in several exceedances of the 1-hour standard for NO<sub>2</sub> at JWA and OCX and continue the exceedances of the State 24-hour standard for PM<sub>10</sub> at OCX and JWA. Although no airport dispersion analysis was conducted for this project alternative, these local criteria pollutant hot spots found under the Proposed Project may also occur under this alternative.

At intersections in the vicinity of the project sites, CAL3QHC model was used to assess the CO concentrations. Tables 8.5-6 and 8.5-7 show that the 1-hour and 8-hour CO concentrations would be below the State and federal CO standards of 9 ppm/20 ppm and 9 ppm/35 ppm, respectively. Therefore, similar to the Proposed Project, no CO hot spots would occur from project related vehicular traffic trips under this alternative.

# Toxic Air Contaminant Impacts

This alternative would avoid some of the impacts identified under the Proposed Project at MCAS El Toro but have greater impacts than under the Proposed Project at JWA. Therefore, air toxic impacts would likely be similar to those under the Proposed Project. Impacts would be reduced with the mitigation measures recommended for the project but are anticipated to remain significant after mitigation.

# 8.5.4.6 Topography

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

With the exception of the JWA/OCX connector, the facilities to be developed and constructed for this alternative are very similar to those of the Proposed Project. Since development of the MCAS El Toro site under Alternative C is similar to that described for the Proposed Project, no significant impacts related to topography would occur.

Operations and construction at JWA under Alternative C would be similar to the current usage, and would not entail expansion of the airport acreage although it would require some facilities improvements in previously developed or disturbed areas. Therefore, Alternative C would not raise potential impacts related to topography.

Table 8.5-6

Phase 4 Alternative C – Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

ENT#	INTERSECTING PIERF	a Reco	, REC	1462	Hick	143	SEC.	. HECT	a secs.	RECY	Breit	REC11	REC
	CITY OF ORANGE"												
45	Jamborce & Chapman	7.0	7.0	7.2	7.0	7.0	7.2	6.8	7.2	6.9	6.8	6.8	6.7
	CITY OF SANTA ANA <sup>D</sup>												
54	MacArthur & Main	7.1	7.2	7.1	7.2	6.9	6.9	6,8	7.1	6.9	7.0	6.7	6.8
52	Main & Sunflower	7.0	7.1	6.7	7.1	6.6	7.0	6.6	6.5	6.7	6.8	7.1	6.9
0	Grand & Edinger	6.9	7.0	7.0	7.3	7.0	7.1	6.7	7.0	6.6	7.0	6.9	6.7
	CITY OF TUSTIND												
3	Newport & Edinger	7.2	7.1	6.9	7.2	6.7	6.7	6.6	6.7	6,7	7.0	6,6	6.9
15	Yon Karman & Barrance	7.0	7.1	7.1	6.9	6.6	7.1	6.7	6.9	6.6	6.6	6.7	6.9
25	Tustin Ranch & Edinger	7.1	7.0	6.9	7.1	6.7	6.9	6.6	6.7	6,9	7.0	6.6	7.0
	CTTY OF IRVINE"												
16	Jamboree & Barranca	5,9	6.0	5.6	5.6	5.3	5.5	5.3	5.5	5,5	5.4	5.5	5.8
156	Jamboree & Main	5,6	5.6	5.7	5.4	5.2	5.4	5.3	5.4	5.3	5,3	5.4	5.5
8	Culver & Irvine Center	5.6	5,6	5.7	5.7	5.3	5.4	5,4	5,5	5.3	5,5	5,4	5.5
34	Jamboree & Alton	5.6	5.6	5.7	5.6	5.1	5,5	5,2	5.4	5.4	5.4	5.4	5.7
75	Jamboree & Michelson	5.7	5.4	5.4	5.6	5.1	5.3	5,3	5.3	5.3	5.4	5.4	5.4
51	Red Hill & MacArthur	5.7	5.4	5.5	5.6	5,1	5.4	5.4 5.2	5.3	5,3 5,4	5.4	5,4 5,1	5.7 5.3
00 21	Jeffrey & Irvine Center Access Rd. West & Irvine	5.6 5.2	5,5 5,2	5.5 5.2	5.7 5.4	5.4 5.1	5.6 5.1	3.2 3.1	5,3 5.2	4,9	5.7 4.9	5.1	5.2
53	Red Hill & Main	5.4	5.2 5.5	5.5	5.6	5.1	5,3	5.2	5.2 5.2	5.2	5.2	5,1	5.2
20	Perimeter & Irvine	5.3	5.3	5.3	5.1	5.0	4.9	5.0	5.2	4.8	4.8	5.1	5.1
	CITY OF LAGUNA BEAC	H <sup>14</sup>											
99	Moulton & El Toro	5,4	5,4	5,5	5.5	5,1	5.3	5.4	5.4	\$.0	5.3	5.1	5.1
	CITY OF LAGUNA HILL	g14											
80	El Toro & Avd. Carlota	5.4	5,4	5,4	5.3	5.0	5.2	5.0	5.0	5.2	5.2	5.1	5.3
	CITY OF LAKE FOREST	н											
271	El Toro & Rockfield	5,6	5.5	5.6	5.5	5.2	5.3	5.2	5.3	5.2	5.3	5,3	5.4
Note:	* - Concentrations are in part 1 - REC1 SW CORNER 2 - REC2 SE CORNER 3 - REC3 NE CORNER 4 - REC4 NW CORNER 5 - REC5 S. DEPARTURE 6 - REC6 N. APPROACH - 7 - REC7 E. DEPARTURE 8 - REC8 W. APPROACH - 9 - REC9 N. DEPARTURE 10 - REC10 S. APPROACH 11 - REC11 W. DEPARTURE 11 - REC11 W. DEPARTURE 12 - REC12 E. APPROACH	- MID BLOCK MID BLOCK - MID BLOCK - MID BLOCK - MID BLOCK - MID BLOCK I - MID BLOCK RE - MID BLOCK	c <b>k</b>										

<sup>13 -</sup> The ambient one-hour CO concentration, 6.10 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Central Grange County Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

Table 8.5-7

Phase 4 Alternative C – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

		Intersection											
INT#	intersecting street	rs Recu	RECY	HEC	RECT	RECS	PECS.	REC?	RECE	· Max	RECIO	REC13 <sup>11</sup>	REC12
A STATE OF THE PARTY OF THE PAR	CITY OF ORANGE®	E.E		e. ac tatan et austil		a ware a state of						AND DESCRIPTION OF THE PARTY OF	
345	Jamboree & Chapman	5.2	5.2	5.4	5.2	5.2	5.4	5.1	5.4	5.2	5.1	5.1	5.0
	CITY OF SANTA ANA"												
154	MacArthur & Main	5,3	5.4	5.3	5.4	5.2	5.2	5.1	5.3	5.2	5.2	5.0	5.1
152	Main & Sunflower	5.2	5.3	5.0	5.3	5.0	5.2	5.0	4.9	5.0	5.1	5.3	5.2
90	Grand & Edinger	5.2	5.2	5,2	5.4	5.2	5.3	5.0	5 2	5 0	5.2	5.2	5.0
	CITY OF TUSTIN"												
93	Newport & Edinger	5.4	5.3	5.2	5.4	5.0	5.0	5.0	5.0	5.0	5.2	5.0	5.2
115	Von Karman & Barranca	5.2	5,3	5.3	5.2	5.0	5,3	5.0	5.2	5.0	5.0	5.0	5.2
95	Tustin Ranch & Edinger	5.3	5.2	5.2	5.3	5.0	5.2	5.0	5.0	5.2	5.2	5.0	5.2
	CITY OF IRVINE"												
116	Jamboree & Barranca	3.8	3.9	3.6	3.6	3,4	3.5	3.4	3.5	3.5	3.5	3.5	3.7
156	Jamboree & Main	3.6	3.6	3.7	3.5	3.3	3.5	3.4	3.5	3.4	3.4	3.5	3.5
98	Culver & Irvine Center	3.6	3.6	3.7	3.7	3.4	3,5	3.5	3.5	3.4	3.5	3.5	3.5
134	Jamboree & Alton	3.6	3.6	3.7	3.6	3.3	3.5	3.3	3.5	3.5	3.5	3.5	3.7
175	Jamboree & Michelson	3.7	3.5	3.5	3.6	3.3	3.4	3.4	3.4	3.4	3.5	3.5	3,5
151	Red Hill & MacArthur	3.7	3,5	3.5	3.6	3.3	3.5	3.5	3.4	3.4	3,5	3.5	3.7
100	Jeffrey & Irvine Center	3.6	3,5	3.5	3.7	3.5	3.6	3.3	3.4	3.5	3.7	3.3	3.4
321	Access Rd. West & Irvine	3.3	3.3	3.3	3.5	3.3	3.3	3.3	3.3	3.1	3.1	3.3	3,3
153	Red Hill & Main	3.5	3.5	3.5	3.6	3.3	3.4	3.3	3.3	3.3	3.3	3.3	3.3
320	Perimeter & Irvine	3.4	3.4	3.4	3,3	3.2	3,1	3.2	3.3	3.0	3.0	3.3	3.3
	CITY OF LAGUNA BEAC	CH <sub>rt</sub>											
299	Moulton & El Toro	3.5	3.5	3.5	3.5	3.3	3.4	3.5	3.5	3.2	3.4	3.3	3,3
	CITY OF LAGUNA HILL	<b>5</b> "											
280	El Toro & Avd. Carlota	3.5	3.5	3.5	3.4	3.2	3.3	3.2	3.2	3.3	3.3	3.3	3,4
	CITY OF LAKE FOREST	P**											
271	El Toro & Rockfield	3.6	3,5	3.6	3.5	3.3	3.4	3.3	3.4	3.3	3,4	3.4	3.5
Note:	* - Concentrations are in par	rts per million (ppn	n)										
	1 - RECI SW CORNER												
	2 - REC2 SE CORNER												
	3 - REC3 NE CORNER												
	4 - REC4 NW CORNER												
	5 - REC5 S. DEPARTURE	- MID BLOCK											

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> RECS W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient eight-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Central Grange County Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

<sup>14 -</sup> The ambient eight-hour CO concentration, 2.9 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

# 8.5.4.7 Soils, Geology and Seismicity

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

With the exception of the JWA/OCX connector, the physical area to be developed/redeveloped and constructed for Alternative C is very similar to that under the Proposed Project. Development on the MCAS El Toro site under Alternative C would be very much like that assumed for the Proposed Project, and would not significantly differ in its potential impacts related to soils or local geologic features. Alternative C also does not entail additional risk based on projected earthquake events beyond those discussed for the Proposed Project.

Operations and construction at JWA under Alternative C would be similar to current usage, and would not entail expansion of the airport acreage. Therefore, Alternative C would not raise potential impacts related to soils, geologic features or seismicity.

# 8.5.4.8 Hydrology and Water Quality

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

With the exception of the JWA/OCX connector, the facilities to be developed and constructed for Alternative C are very similar to those of the Proposed Project. As discussed in Section 4.8 (Hydrology and Water Quality), the Proposed Project will not result in significant adverse impacts related to drainage and surface water quality. Because most issues related to drainage at the MCAS El Toro site can be addressed adequately through proper design and engineering, it is anticipated that Alternative C could be developed for use as a civilian airport without significant adverse impacts related to hydrology. Similarly, as discussed in Section 4.8, operations can be conducted and controls implemented to minimize potential project-related adverse impacts to surface water quality. Consequently, development of this alternative is unlikely to result in significant adverse impacts to surface water quality.

The JWA/OCX connector will have additional impacts related to both runoff and water quality associated with the connector corridor and facility between the two airports. The drainage impacts can be mitigated using proper engineering design and construction practices; similar to those assumed for the roads and runways under the Proposed Project. Impacts to surface water quality from construction and operation of this connector can be

mitigated, using Best Management Practices (BMPs) and other permit requirements to minimize adverse impacts related to water quality, similar to the Proposed Project. Therefore this alternative will not result in significant adverse drainage and surface water impacts after mitigation.

No groundwater will be pumped from the MCAS El Toro site under this alternative so there will be no impacts to local groundwater levels or basin storage under this alternative. Groundwater quality impacts under this alternative will be the same as those discussed for the Proposed Project in Section 4.8, associated with the base closure plan remediation. Therefore, this alternative will result in no significant adverse impacts after mitigation related to groundwater. As with the Proposed Project, no hazardous waste remediation activities at the MCAS El Toro site are included as a component of this alternative.

In summary, the hydrology and water quality impacts of Alternative C will be slightly greater than the level of impacts under the Proposed Project because of the connector and the increased aviation activities at JWA.

## 8.5.4.9 Biological Resources

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

The aviation land use and airfield component for the MCAS El Toro site and the corresponding nonaviation component are very similar to the Proposed Project. Alternative C project components that require construction occupy nearly the same areas and closely parallel functions identical to those under the Proposed Project. The primary difference between Alternative C and the Proposed Project is the airport-to-airport connector. However, the physical improvements that comprise Alternative C have nearly identical biological resource impacts as to the Proposed Project.

The direct impacts of Alternative C also include the loss of approximately 139 acres of agricultural land, which is the same acreage loss estimated for the Proposed Project. This impact results in reduced foraging opportunities for raptor species similar to the Proposed Project. Other direct impacts (i.e. native plant communities, wildlife, wildlife dispersion corridors and special interest species) are also very similar. There are some slight differences in potential impacts as a result of noise exposure and aircraft overflights, since the aircraft operations differ at both the MCAS El Toro and JWA sites. However, noise and overflight characteristics are not substantively different between Alternative C and the Proposed Project and are not expected to result in significant adverse biological resource impacts at Siphon Ridge, the Habitat Reserve, the San Joaquin Hills or Upper Newport Bay.

The CNEL and SEL values at these locations are discussed in detail in the Biological Resources Technical Report.

#### 8.5.4.10 Public Services and Utilities

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

The airport-to-airport light rail system that is unique to Alternative C would require some level of police security, emergency and medical service, and transit planning. Mitigation to implement the needed services would reduce any potential impacts of Alternative C on public services to below a level of significance. Alternative C is nearly identical to the Proposed Project in all other aspects so the provision of public services would not be impacted.

As described in Section 4.10 (Public Services and Utilities), the Proposed Project is not anticipated to result in significant adverse impacts related to utilities at El Toro site or JWA. It is anticipated that the utilities needs at El Toro and JWA under Alternative C could be served by existing or currently planned utilities, or extensions/expansions of existing utility infrastructure, without significant adverse impacts after mitigation, similar to the Proposed Project. Mitigation similar to that for the Proposed Project would reduce the potential adverse impacts of this alternative related to utilities infrastructure and services at El Toro and JWA to below a level of significance.

# 8.5.4.11 Natural Resources and Energy

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance with the exception of the unavoidable adverse impact to loss of Prime Agricultural Lands. This alternative would not avoid or substantially lessen the impacts of the project.

The land use/airfield, nonaviation land use and associated infrastructure components for this alternative are virtually identical to those of the Proposed Project. The primary exception is the light-rail, airport-to-airport link planned under Alternative C.

As discussed in Section 4.11 (Natural Resources and Energy), the Proposed Project will not result in significant adverse impacts related to natural resources and energy at either JWA or the MCAS El Toro site, with the exception of unmitigatable significant impacts to agricultural resources on the MCAS El Toro site. There are no agricultural resources at

JWA. The incremental increase in regional energy consumption associated with operation of the JWA/OCX connector would be minor. Consequently, although energy consumption would be greater for this alternative than for the Proposed Project, no significant adverse impacts to energy resources will occur with the implementation of this alternative.

This alternative and the Proposed Project would have the same level of significant adverse unmitigatable impacts associated with loss of agricultural resources at the MCAS El Toro site. There are no agricultural resources existing at JWA; therefore, no impacts at JWA would result from this alternative.

# 8.5.4.12 Aesthetics, Light and Glare

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the MCAS El Toro site than the Proposed Project. With mitigation measures, the impacts of this alternative would be reduced to a level of insignificance at both JWA and MCAS El Toro sites. This alternative would not avoid or substantially lessen the project impacts.

The visual effect of Alternative C at the MCAS El Toro site would be very similar to that of the Proposed Project. The primary differences between Alternative C and the Proposed Project are the addition of a people mover (APM) passenger and baggage transport system between JWA and OCX.

The JWA/OCX connector would be an elevated fixed guideway system along SR-133 to the I-405 corridor, then along the I-405 Freeway right-of-way to MacArthur Boulevard and then to JWA. Examples of "people mover" systems are shown in Technical Report No. 6, Alternatives Definition Report (OCAA, November, 1999), Figures 1-4 through 1-6. Provision of an elevated guideway system for passenger and baggage connection between the two airport sites would impact the existing visual setting along the freeway corridor by creating an upper level structure that currently does not exist. The new structure would block views from the freeway corridor. The terminal points of this system at OCX and JWA would be visible from existing roads and the I-405 Freeway.

The views from the majority of the vantage points described in Section 4.12 would not change substantially from those of the Proposed Project. Vantage Point 8 would show the connection of the elevated APM system as it enters the terminal. Light and glare effects of this alternative would be similar to those of the Proposed Project, with some potential additional lighting from the APM facility along the freeway route and at the OCX terminal.

No significant runway improvements would be made at JWA, and the terminal would be expanded by lengthening both concourses and increasing the size of the RON area; no expansion of the existing boundaries of JWA would be required. The APM connecting JWA with OCX would enter the terminal area at JWA by way of MacArthur Boulevard and terminate in a station adjacent to the existing terminal. The addition of this passenger

system would be visible from existing roads and the I-405 Freeway adjacent to JWA. No scenic vistas or views would be blocked or altered by the addition of this structure, as the area is urbanized; the system would further intensify the urbanized effect of the visual setting. Compared to the Proposed Project, additional lighting and potential glare would be generated by the APM system as it connects with the JWA terminal.

In conclusion, the aesthetic impact of Alternative C would not be less than those of the Proposed Project.

#### 8.5.4.13 Cultural Resources

Compared to existing conditions, this alternative would have the potential for greater impacts at JWA and slightly less impact at the MCAS El Toro site than the Proposed Project. With mitigation measures, the impacts of this alternative at either site would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the project impacts.

The effects of Alternative C on cultural resources at the MCAS El Toro site would be approximately the same as with the Proposed Project. As the cultural resources within the disturbance area (construction and operations) are not considered significant (SHPO concurrence to be included in the DOD's EIS), no significant impacts would be caused by this alternative, as with the Proposed Project.

Under Alternative C, improvements at JWA would be made within the boundaries of the existing airport site. For Alternative C, there would be no additional or new effects on cultural resources since there is no known archaeological, paleontological or historic resources on the already developed airport property.

#### 8.5.4.14 Recreation

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the MCAS El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project and additional mitigation related to noise impacted recreation uses around JWA, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the project.

Alternative C would have approximately the same effects related to recreational resources in the MCAS El Toro area as the Proposed Project. The area of construction with Alternative C would be, for all practical purposes, the same as that of the Proposed Project. Therefore, no adjacent off-road trails would be physically impacted with Alternative C similar to the Proposed Project. Development at the MCAS El Toro site with Alternative C would have the same effect regarding consistency with County and City General Plan Recreational policies, goals and objectives, in that nonaviation land uses would be included on-site

similar to the Proposed Project. In addition, Alternative C would not differ from the project in that it would not exceed Thresholds of Significance ii and iii in Section 4.14.

Given the alignment of the JWA/OCX connector, impacts to existing recreational facilities would be limited to temporary disruption of use of on-street Class II bikeways adjacent to OCX and JWA during construction of the connector facility. The temporary impact to on-street bikeways would be less than significant with implementation of standard construction detour measures. No off-road trails or site specific recreational areas would be physically impacted by the airport to airport connector.

Compared to the Proposed Project, Alternative C would have substantially the same impact on recreational facilities and planned future facilities within the 65 CNEL contour. The overall noise impact on the use of area recreational facilities would be similar to the impacts of the Proposed Project.

The airport to airport connector would be located within existing freeway right of way and along existing highways, such that recreational facilities such as off-street trails and parks would not be affected. Alternative C would result in slightly enlarged noise contours around JWA; however, this increase would be minimal, and not anticipated to include any additional recreational facilities within the 65 CNEL noise contour compared to existing conditions.

# 8.5.4.15 Public Health and Safety

Compared to existing conditions the overall potential for accidents is greater with this alternative. The increase in accident potential is not deemed to be significant as an extraordinary risk is not created. This alternative would not avoid or substantially lessen project impacts.

# Aviation Safety

Compared to the Proposed Project, there would be an increase of approximately 79,500 air carrier and air cargo operations and a decrease of approximately 42,000 general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 117.8% to reflect the number of increasing aviation activity diverted from OCX to JWA and the potential accident risks for general aviation at JWA would decrease by 11.7% correspondingly. At OCX, there would be an estimated decrease of 101,500 air carrier and air cargo operations and an estimated decrease of 7,000 general aviation operations. Under this condition, the potential air carrier and air cargo accident risks at OCX would decrease by approximately 36.6% to reflect the fewer number of operations at OCX and the potential general aviation accident risks would decrease by 31.8% correspondingly. Compared to the Proposed Project relative to onairport and off-airport fatal accidents per million operations, there would be no significant adverse impacts related to aviation safety at the MCAS El Toro site or at JWA.

Compared to the existing conditions, there would be an increase of approximately 56,992 air carrier and air cargo operations and a decrease of approximately 10,624 general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 63.3% to reflect the number of increasing operations and the potential general aviation accident risks would slightly decrease by 3.2% correspondingly. Compared to the existing conditions, there would be no significant adverse impacts related to aviation safety at JWA.

## 8.5.4.16 Hazardous Materials and Hazardous Wastes

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the MCAS El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the Proposed Project.

Construction of facilities required under both the Proposed Project and Alternative C would require ground-disturbing activities. Under Alternative C, the impacts of greatest concern, when dealing with soil and groundwater contamination, are human exposure and the spread of contaminants in the environment. Since the preferred approach to the transit connector is an elevated system, no significant excavation is expected.

If PCE levels in the groundwater exceed regulatory levels at the time of construction, treatment would be required before the extracted water could be discharged.

The impacts of Alternative C related to hazardous materials and hazardous wastes are the same as under the Proposed Project.

#### 8.5.4.17 Socioeconomics

Under this alternative, a total of 29,100 jobs would be created for the Airport System Master Plan, including almost 22,900 jobs at the MCAS El Toro site and 6,200 jobs at JWA in 2020, representing a net increase of 20,200 jobs at MCAS El Toro site, and 4,100 jobs at JWA, over existing 1998 conditions. The total number of jobs generated under this alternative is marginally lower than under the Proposed Project. However, there are differences in the distribution of jobs between the two alternatives. The number of jobs generated at the El Toro site under Alternative C will be lower than the number of jobs generated under the Proposed Project. However, a greater number of jobs would be generated at JWA under Alternative C than under the Proposed Project.

As with the Proposed Project, economic activity on the MCAS El Toro and JWA sites, as well as expenditures by visitors arriving by air through the two airports, would stimulate

additional off-site job growth. The number of off-site jobs stimulated by the airport system under Alternative C would be similar to the level under the Proposed Project.

Given the marginal difference in the total number of jobs generated under the Proposed Project and Alternative C, at 29,500 and 29,000 jobs respectively, the magnitude of impacts under Alternative C related to inducing substantial growth or concentration of employment, consistency with adopted regional forecasts, and increased demand for housing, including low and moderate income housing, would be similar to that of the Proposed Project. Therefore, the impacts of Alternative C will not be substantially different from the impacts of the Proposed Project. This alternative would not avoid or measurably lessen the impacts of the Proposed Project.

# 8.5.4.18 Risk of Upset

Compared to existing conditions, this alternative would have greater impacts at JWA and slightly less impact at the El Toro site than the Proposed Project. With the mitigation measures proposed for the Proposed Project, the impacts of this alternative would be reduced to a level of insignificance. This alternative would not avoid or substantially lessen the impacts of the Proposed Project.

As discussed in Section 4.18 (Risk of Upset), the Proposed Project would not result in significant adverse impacts to public health and safety resulting from project-related risk of upset conditions. The ultimate build out and phased development of this alternative will entail a level of operations similar to the Proposed Project, with similar levels of risk of upset potential associated with jet fuel storage and delivery requirements. Consequently, this alternative would not result in significant adverse impacts to public health and safety.

# 8.5.5 Feasibility

Technical Report 13, published March 18, 1999, provided a detailed analysis of the OCX-JWA connector system to evaluate the feasibility of Alternative C. In order to allow passengers to connect effectively between the short-haul flights at JWA and the longer haul flights at OCX, it would be necessary to build and operate a connector system between the two airports which would, in effect, allow the two facilities to function as a single airport. Without this connector, the market segmentation between the two airports is not feasible. Also, it is assumed that regulatory perimeter rules would define the roles of the two airports.

The costs of the connector were found to be unreasonable to the extent they would impose unnecessary burdens on the Orange County air traveling public and the airlines that serve them. The total costs per rider for a two-way connector trip were estimated to be between \$103 and \$110 (in 2020 dollars), assuming the connector would be implemented in Phase 4 when connector costs could be spread over a greater number of passengers. If the connector were to begin service in 2005, the cost per rider would be \$248 (in 2005 dollars) for a two-way connector trip. Most of these connector costs would be absorbed ultimately by the

passenger in the form of higher ticket prices, and such an increase would be unacceptable to the airlines and passengers.

## 8.5.6 Conclusions

For the reasons noted above, Alternative C is infeasible. In addition to infeasibility, the impact analysis demonstrates that this alternative would:

- Meet the general project objectives except to optimize project cost/revenues.
- Increase aviation noise impacts at JWA and regional air quality impacts.
- Decrease aviation noise impacts near OCX compared to the Proposed Project, but impacts on sleep disturbance and recreation uses would remain significant and unavoidable.

Have land use, General Plan consistency, traffic, sleep disturbance, noise impacts on recreation uses, local and regional air quality impacts, construction air quality impacts, toxic air contaminant impacts, soils, geology, seismicity, hydrology, water quality, biological, public services, natural resource, energy, aesthetics, light and glare, cultural, recreational, public health, safety, hazardous materials/wastes, socioeconomics, and risk of upset impacts the same or similar to the project.

# 8.6 ALTERNATIVE F: JWA – SHORT- TO LIMITED LONG-HAUL WITH LIMITED GENERAL AVIATION; NO AVIATION REUSE AT FORMER MCAS EL TORO

This section presents the potential impacts of Alternative F as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

This alternative was selected for analysis because it has the potential to avoid significant unavoidable aircraft noise and aircraft air quality emission impacts at the El Toro site while still feasibly attaining some of the objectives of the Proposed Project.

# 8.6.1 Aviation Uses

Under Alternative F, JWA would continue to provide short- and medium-haul domestic passenger service (with limited long-haul service), and there would be no aviation reuse at MCAS El Toro. JWA would also provide all-cargo service to short-, medium-, and limited long-haul destinations. JWA would not be constrained by existing limits on passengers or aircraft operations under this alternative. The airport would accommodate as much passenger demand as possible, estimated to be approximately 14 MAP in 2020, by expanding airport facilities to the extent possible within the existing airport property limits, approximately four percent (0.6 MAP) of which would be passengers with connecting flights. JWA is also forecast to annually handle approximately 180 thousand tons of domestic cargo. Alternative F would include 29 jet aircraft gates and 8 commuter aircraft gates, 19 Remain Overnight (RON) aircraft parking spaces, 13,820 vehicle parking spaces, and approximately 1.14 million square feet of terminal area. There would be minimal general aviation service at JWA, which would allow the airport to accommodate expanded commercial service. The general aviation runway would be closed. The main runway would be extended from 5,700 feet to 6,800 feet. General aviation activity would be displaced to private or municipal airports in Orange County or other counties. Figure 8-6 depicts Alternative F.

The environmental analysis of this alternative focuses on the impacts of the alternative at JWA. This alternative does not propose or include any physical changes at the El Toro site. However, if Alternative F were selected and implemented, it necessarily would result in the adoption of a nonaviation plan for the El Toro site, possibly one similar to the ETRPA Nonaviation Alternative. To understand the full impacts of Alternative F along with the

ETRPA Nonaviation Alternative, for example, the reader should review the impacts of both alternatives as addressed in this section.

# 8.6.2 Nonaviation Revenue Support Uses

Alternative F does not propose nonaviation uses at JWA and does not include any physical changes at the El Toro site. However, approval of Alternative F would lead to the adoption of a nonaviation plan for the El Toro site.

# 8.6.3 Attainment of Project Objectives

This alternative will not meet the general project objectives for reuse of MCAS El Toro. Alternative F will also not meet the general aviation, existing land use restrictions, and General Plan implementation objectives. It will have a major adverse impact on general aviation as the more than 500 general aviation aircraft now at JWA would have to be relocated. Alternative F also does not encourage growth of service opportunities, and it does not implement the two airport system. This alternative will partially further the other aviation related objectives.

# 8.6.4 Environmental Impacts of Alternative F

## 8.6.4.1 Land Use

This alternative would have no land use impacts at the El Toro site since all development would occur at JWA. However, this alternative would have greater adverse land use impacts at JWA than the Proposed Project. Based on this analysis, the alternative would not avoid or substantially lessen impacts compared to the project.

JWA under Alternative F will serve almost twice as many commercial air passengers as are currently served at JWA. This will require a runway extension and facilities expansion. The JWA site is surrounded by business parks, light industrial uses, and airport serving businesses, which are compatible with intensified airport use at JWA, therefore; the intensification of on-site land uses associated with Alternative F will not have significant impact on adjacent off-site land uses. However, as a result of a larger 65 CNEL noise contour, this alternative will have a significant effect on existing residential uses compared to no significant effect under the Proposed Project (see Section 8.4.4.4).

# 8.6.4.2 General Plan Consistency

This alternative would impact General Plan consistency issues at the El Toro site, although it would not raise General Plan issues with respect to JWA. Amendments to the County Noise Element and AELUP are not required for JWA because the new noise contours related to the increase in the aviation activity at JWA would be within the 1985 JWA Master Plan

contours. Although this alternative avoids aviation uses at the El Toro site, a nonaviation use at El Toro would require a County General Plan Amendment to replace Measure A policies designating the El Toro site for commercial airport development and amendments to reflect the absence of aviation noise and associated land use restrictions.

Under the Proposed Project, the Land Use, Noise, Public Services and Facilities, and Safety elements of the General Plan are proposed to be amended.

# 8.6.4.3 Transportation and Circulation

This alternative would have greater adverse traffic impacts at JWA than the Proposed Project. Additionally, since this alternative would meet less existing and future County aviation demand, the alternative would result in higher regional vehicle miles traveled (VMT) compared to the Proposed Project. This alternative would not avoid or lessen measurably the project impacts because JWA impacts would be increased and because the foreseeable development of the El Toro site with nonaviation uses would generate adverse impacts greater than the project due to higher regional VMTs.

The AM and PM peak hour and ADT traffic generated by JWA with build out of Alternative F is summarized in Table 8.6-1. Refer to Section 11.0 in the 1999 Traffic Analysis Technical Report for detailed information on the methodology applied to produce trip generation estimates for Alternative F.

Table 8.6-1
Trip Generation Summary – Alternative F

JWA	2,524	1,711	4,235	3,330	3,312	6,642	83,943	47,450
Project Component	Lin.	Out "	Total	·In	- Out	"Total;		ADT
					A STATE OF THE STA	The second second	ADT	- Existing

For the JWA site, no changes to the connections that currently provide access between JWA and the surrounding circulation system are envisioned under Alternative F. Primary access to the passenger terminal would be provided by the existing entryways from MacArthur Boulevard at the Michelson Drive and I-405 southbound ramp intersections and from SR-55 via the existing JWA direct connector ramps. The parking areas that replace the existing general aviation facilities in the southeast part of the airport would be accessed from Campus Drive via the existing Airport Way intersection. The parking areas that replace the existing general aviation facilities in the southwest part of the airport would be accessed via the existing general aviation entryway from Baker Street east of Red Hill Avenue.

Table 8.6-2 compares, in summary, the Alternative F highway impacts to the existing conditions and existing conditions plus Proposed Project. There is minimal comparison between the existing plus Proposed Project versus the Alternative F impacts due to the large differences between the scope of the project (two airports) and the alternative (one airport).

Table 8.6-2
Summary Comparison of Traffic Impacts for Alternative F to
Existing Conditions and Existing Conditions Plus Project

Existing Conditions  Deficient Highway Facilities	Existing Conditions. Plus Proposed Project Impacts	Alternative F With Existing Plus Committed Facilities
Location	Location	Location
INTERSECTIONS	IMPACTED INTERSECTION	IMPACTED INTERSECTIONS
Newport (NB) & Del Mar	Bake & Portola	Red Hill & Baker
El Toro & SR-73 NB Ramps	Sand Canyon & Trabuco	SR-55 NB Frontage & Baker
Campus & N. Bristol	Bake & 1-5/1-405 SB Ramps	SR-55 NB Frontage & Paularino
Jamboree (SB) & Walnut	Bake & Rockfield	Main & Sunflower
Jamboree & I-405 NB Ramps	Jeffrey & Alton	Campus/Irvine & South Bristol
Jeffrey & I-405 NB Ramps	Jeffrey & I-405 NB Ramps	Campus & North Bristol
Red Hill & MacArthur	Jeffrey & I-405 SB Ramps	Jamboree & Michelson
Irvine Center & Lake Forest	Jeffrey & Walnut/I-5 SB	MacArthur & Main
Bake & Jeronimo	Sand Canyon & I-5 NB Ramps	MacArthur & Michelson
El Toro & Avd Carlota	Sand Canyon & I-5 SB Ramps	Red Hill & MacArthur
La Paz & Cabot/I-5 SB	Irvine Center & Lake Forest	Red Hill & Main
Los Alisos & Muirlands	Bake & Irvine/Trabuco	Von Karman & Michelson
Alicia & Jeronimo	Bake & Toledo	MacArthur & Campus
Alicia & Muirlands	Los Alisos & Muirlands	-
La Paz & Muirlands/I-5 NB	Alicia & Jeronimo	IMPACTED FREEWAY RAMPS
Red Hill & Edinger	Newport & Old Irvine	I-405 at MacArthur (NB Off-Ramp
Red Hill & Sycamore		I-405 at MacArthur (SB On-Ramp
Red Hill & Walnut	IMPACTED ARTERIAL ROADS	I-405 at Sand Canyon (NB Direct On-Ramp)
	Laguna Canyon (I-405 to SJHTC)	
ARTERIAL ROADWAYS	Laguna Canyon (south of El Toro)	IMPACTED FREEWAY SEGMENTS
Portola (Sand Canyon to Foothill Toll Road)	Culver (Bryan to Trabuco)	SR-55 (MacArthur to I-405)
Laguna Canyon (I-405 to SR-73)		SR-55 (south of SR-73)
Laguna Canyon (south of El Toro)	IMPACTED FREEWAY RAMPS	
Culver (Bryan to Trabuco)	I-5 at Culver (SB Off-Ramp)	
Michelson (Carlson to Harvard)	I-5 at Sand Canyon (NB On-Ramp)	
	I-5 at Sand Canyon (SB Off-Ramp)	
FREEWAY RAMPS	I-405 at Sand Canyon (NB Direct On-Ramp)	
I-5 at Culver (SB Off-Ramp)		
I-405 at Jamboree (SB Off-Ramp)	IMPACTED FREEWAY SEGMENTS	
I-405 at MacArthur (SB On-Ramp)	I-5 (Jeffrey to north of SR-55)	

Existing Conditions Deficient Highway Facilities.	Existing Conditions Plus Proposed Project Impacts.	Alternative F With Existing  * Plus Committed Facilities
I-405 at MacArthur (NB On-Ramp)		
I-405 at MacArthur (NB Off-Ramp)		
SR-55 at Dyer (SB On-Ramp)		
SR-55 at Dyer (NB Off-Ramp)		
SR-55 at MacArthur (SB Direct On- Ramp)		
SR-55 at MacArthur (NB Direct On-Ramp)		
SR-55 at MacArthur (SB Off- Ramp)		
FREEWAY SEGMENTS		
I-5 (Culver to north of SR-55)	***************************************	
I-5 (Alton to I-405)		
I-5 (El Toro to La Paz)		
I-405 (MacArthur to SR-133)		
SR-55 (I-5 to SR-73)		

Abbreviations:

study area).

NB-northbound SB-southbound

EB-eastbound WB-westbound

However, Table 8.6-2 indicates that under Alternative F traffic impacts will occur in the vicinity of JWA, while under the Proposed Project impacts will result primarily in the El Toro vicinity (refer to Section 11.0 in the 1999 Traffic Analysis Technical Report for detailed summaries of the Alternative F traffic volumes and LOS, as well as comparisons between existing plus committed conditions with and without Alternative F for intersections and arterial roadways within the traffic analysis study area, and refer to Section 11.0 in the 2001 Traffic Analysis Technical Report Addendum for comparable information for freeway/tollway mainline segments and freeway/tollway ramps within the traffic analysis

A comparison of the impacts of Alternative F to the Proposed Project during the phasing years may also be made. As discussed in detail in Section 4.3.6.6 of this Draft EIR No. 573, as supplemented, under the Proposed Project phasing years, four intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 1 conditions (2005), five intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 2 conditions (2010), and nine intersection locations, two arterial roadway segments, one continuous freeway mainline segment and two freeway ramps would be significantly impacted under Phase 3 conditions (2015). At Phase 4 build out, the Proposed Project would result in significant impacts not previously identified to four freeway/tollway mainline segments and four freeway/tollway ramps. See Draft Supplemental Analysis Section 4.3.6.5. In each case, however, the identified impacts

will be mitigated to a level below significant during the applicable phasing year (see Section 4.3.7.2, Table 4.3-20).

In evaluating traffic impacts under Alternative F, the reader should keep in mind that approval of this alternative would lead to adoption of a nonaviation plan for the El Toro site such as the ETRPA Nonaviation Plan Alternative analyzed in Section 8.10.

#### 8.6.4.4 Noise

Compared to existing conditions, this alternative would have no aviation noise impacts at the El Toro site, but would have greater adverse noise impacts at JWA than existing operations and the Proposed Project. Alternative F would increase the 60 and 65 CNEL contours at JWA from the 1998 contours somewhat, but not to the extent where they exceed those of the 1985 Master Plan contours. The number of individual commercial aircraft events will also increase substantially under this alternative. This alternative assumes that these operations would all be accommodated during existing operations hours, and no increase in the number of nighttime operations is expected under this alternative at JWA. However, the substantial increase in the number of operations may be considered a significant impact of Alternative F independent of the CNEL computation, as it was for El Toro under the Proposed Project and various other alternatives. The number of affected residences inside the 60 and 65 dB CNEL contours is greater under Alternative F than either existing conditions or the Proposed Project (Table 8.4-4).

In conclusion, this Alternative would avoid aircraft noise impacts at the El Toro site. Also see Figure 8-7, which depicts noise contours for Alternative F.

Adoption of this alternative would probably lead to approval of a nonaviation land use plan for the El Toro site, such as the ETRPA Nonaviation Alternative. For analysis of the noise impacts of a nonaviation land use plan, refer to Section 8.3.

# 8.6.4.5 Air Quality

As described below, Alternative F would result in: 1) significant unavoidable short-term construction impacts greater than under the Proposed Project; 2) significant regional impacts greater than the Proposed Project under all development scenarios due to Orange County generated demand being serviced at other regional airports outside of the County similar to the No Project/No Alternative; and 3) significant local air quality impacts at JWA greater than the Proposed Project resulting from aircraft operations at JWA. This alternative, however, would avoid the significant local air quality impacts of the Proposed Project resulting from aircraft emissions at OCX. This alternative may, however, result in the adoption of a nonaviation plan for the El Toro site that could have local CO hot spot impacts greater than the Proposed Project.

## Short-Term (Construction) Impacts

Under this alternative, there would be a facility expansion and runway extension at JWA. Construction emissions would be greater than those of the Proposed Project at JWA. With respect to MCAS El Toro, this alternative may lead to the adoption of a nonaviation alternative similar to the ETRPA Nonaviation Alternative. Construction emission impacts at MCAS El Toro under this scenario could be greater than those of the Proposed Project due to higher density or intensity land uses being proposed. Therefore, this alternative would result in significant and unavoidable construction emissions that could be greater than the Proposed Project under all development scenarios and would not avoid or substantially lessen impacts compared to the Proposed Project.

# Operational Air Quality Impacts

#### **Emissions Inventories**

Under this alternative, total annual passenger and total annual aircraft LTO operations at JWA would be greater than those of the Proposed Project. Alternative F's direct air pollutant emissions associated with airport operations, including aircraft, GSE, energy consumption, and vehicular trips, are shown below in Table 8.6-3.

When compared to the direct air quality emissions associated with the Proposed Project at build out, Alternative F would have greater CO, NO<sub>X</sub>, and PM<sub>10</sub> emissions at JWA but lower ROC emissions.

Air pollutant emissions, including airport operations at other airports in the region and VMT required for air travel passengers to get to these airports, are shown in Table 8.6-4 for this alternative. The regional air quality impacts under this alternative would be significant and would be greater than under the Proposed Project. These regional air quality impacts, however, would be less than under the No Project Alternative.

Table 8.6-3
Phase 4 Alternative F – Project Direct Air Pollutant Emissions (Pounds/Day)

	CO	NOx -	ROC <sup>1</sup>	SOx	PM <sub>10</sub>
Aircraft	2,073.72	5,146.14	205.17	349.98	65.49
OCX					
JWA	2,073.72	5,146.14	205.17	349.98	65.49
GSE/APU	9,605.97	622.45	246.59	15.13	26.49
OCX					
JWA	9,605.97	622.45	246.59	15.13	26.49
Fuel Storage/Dispensing			4.76		
OCX					
JWA			4.76		
Airport Roadways	198.12	26.61	7.23	1.21	2.10
OCX					
JWA	198.12	26.61	7.23	1.21	2.10
Airport Parking	154.99	12.02	16.02	4.90	0.47
OCX					
JWA	154.99	12.02	16.02	4.90	0.47
Energy Consumption	52.80	303.60	2.80	31.10	10.40
OCX					
JWA	52.80	303.60	2.80	31.10	10.40
Vehicular Traffic	7,417	3,000	583	185	1,535
ocx					
JWA	7,417	3,000	583	185	1,535
Total	19,502	9,111	1,066	587	1,640

Source: CH2M Hill and LSA Associates, Inc., 2001

ROC emissions obtained by multiplying HC emissions reported by EDMS by a factor of 1.14.

Table 8.6-4
Regionwide Emissions Inventory Alternative F Phase 4
(Pounds/Day Unless Noted)

		СО	NOX	ROC	$SO_X$	PM <sub>10</sub>
Aircraft	El Toro		***			
	JWA	2,073.72	5,146.14	205.17	349.98	65.49
	Other Airports	78,722.06	80,454.54	10,959.04	6,126.26	887.12
	Total Regional	80,795.78	85,600.68	11,164.21	6,476.24	952.61
GSE	El Toro		-00 100		<b>**</b> *	
ACC.	JWA	9,605.97	622.45	246.59	15.13	26.49
	Other Airports	102,215.15	10,264.12	3,024.28	664.81	376.36
	Total Regional	111,821.12	10,886.57	3,270.87	679.94	402.85
Energy	El Toro				***	
	JWA	52.80	303.60	2.80	31.10	10.40
	Others	620.00	3,568.00	33.00	365.00	122.00
	Total Regional	672.80	3,871.60	35.80	396.10	132.40
Fuel	El Toro	***		***		Mr. com
	JWA	water stars		4.76	***	
	Other Airports			535.63	***	
	Total Regional		w <u>s</u> a.	540.39		
Airport Roadways	El Toro		##.##	*****		***
	JWA	198.12	26.61	7.23	1.21	2.10
	Other Airports	3,524.86	715.29	162.18	43.04	58.49
	Total Regional	3,722.98	741.90	169.41	44.25	60.59
Airport Parking	El Toro			· ·		
	JWA	154.99	12.02	16.02	4.90	0.47
	Other Airports	2,203.37	189.72	30.42	58.51	23.10
	Total Regional	2,358.36	201.74	46.44	63.41	23.57
Roads	El Toro					***
	JWA	7,417.00	3,000.00	583.00	185.00	1,535.00
	Others Airports1	<u>2,768,580.00</u>	493,807.00	71,290.00	48,891.00	<u>8,292.00</u>
		2,755,707.00	491,238.00	69,753.00	49,058.00	8,311.00
	Total Regionall	<u>2,775,997.00</u>	496,807.00	71,873.00	49,058.00	<u>9,827.00</u>
		<del>2,763,124.00</del>	494,238.00	70,336.00	49,243.00	9,846.00
TOTAL (pounds/	day)	2,975,368.04	<u>598,109.49</u>	87,100.12	56,717.94	11,399.02
		2,962,495.04	595,540.49	85,563.12	56,902.94	11,418.02
Change from 2020 N	lo Project	(3,994.56)	(1,483.87)	(478.16)	(172.32)	(20.38)
(pounds/day)		(8,521.56)	(2,459.01)	(1,006.53)	(99.32)	(9.95)
SCAQMD Thresho	ld for Operation					<u></u>
(pounds/day)		550	55	55	150	150

Source: LSA Associates, Inc., 2001.

I Typographical correction.

#### **Dispersion Modeling**

At intersections in the vicinity of the project sites, the CAL3QHC model was used to assess the CO concentrations for Alternative F. Tables 8.6-5 and 8.6-6 show that the 1-hour and 8-hour CO concentrations would be below the State and federal CO standards. No CO hot spots would occur from vehicular traffic trips caused by this alternative.

# Toxic Air Contaminant Impacts

This alternative would avoid impacts at the El Toro site but would increase impacts at the JWA site. Impacts would be reduced with the mitigation measures recommended for the project but are anticipated to remain significant after mitigation.

# 8.6.4.6 Topography

Under Alternative F the main runway at JWA (Runway 19R-1L) would be extended 1,100 feet to the north; however, no expansion of overall airport acreage is planned. Under Alternative F, terminal expansion would potentially require minor grading, but because the site is essentially flat, there would be no significant impact related to topography. Therefore, Alternative F would not be expected to result in adverse impacts related to topography at JWA, similar to the result found for the Proposed Project. As noted above, approval of this alternative would lead to the adoption of a nonaviation alternative for the El Toro site. Therefore, this alternative would not avoid or substantially lessen the impacts of the project.

# 8.6.4.7 Soils, Geology and Seismicity

This alternative would implement improvements at JWA within the existing developed area of the site. Therefore, no significant impacts related to soils, geology, or seismicity would occur, similar to the Proposed Project. Under Alternative F the main runway at JWA (Runway 19R-1L) would be extended 1,100 feet to the north; however, no expansion of overall airport acreage is planned. Therefore, Alternative F would not result in adverse impacts related to soils, geologic features or seismicity, similar to the Proposed Project. No aviation uses would be developed at the El Toro site. However, this alternative would lead to the adoption of a nonaviation plan for the El Toro site and would not, therefore, avoid or substantially lessen impacts of the project.

Table 8.6-5

Phase 4 Alternative F – Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

INT	INTERSECTING STREETS	REC!	REC2	DEC3	REC1	RECS	RECE	RECT	REC*	RECY	REC10	RECUI".	REC12
<u> </u>	CITY OF ORANGEO			A		**************************************			Taranta Santa	~		310.80.40	
345	Jamboree & Chapman	7.0	7.0	7.2	7.0	7.0	7.2	6.8	7.2	6,8	6.8	6.7	6.6
	CITY OF SANTA ANA"												
154	MacAnhur & Main	7.1	7.2	7,2	7.3	7.0	6.9	6.9	7.1	6.9	7.0	6.7	6.9
152	Main & Sunflower	7.0	7. t	6.8	7.1	6.7	7.0	6.6	6.5	6.7	6.8	7.1	6.9
90	Grand & Edinger	6.9	7.0	7.0	7.3	7.0	7,1	6.7	6.9	6.6	7.0	6.9	6.7
	CITY OF TUSTING												
93	Newport & Edinger	7.2	7.1	6.8	7.2	6.7	6.6	6.6	6.7	6.7	7.0	6.6	6.9
115	Von Karman & Barrance	7.0	7.1	7.1	6.9	6.6	7.1	6.7	6.9	6.6	6.5	6.7	7.0
95	Tustin Ranch & Edinger	7.2	7.0	6.9	7.1	6.7	6.9	6.6	6.7	6.9	6.9	6.6	7.0
	CITY OF IRVINE"												
116	Jamboree & Barranca	5.9	6.0	5.6	5.7	5.3	5.5	5.3	5.5	5.5	5.4	5.5	5.9
156	Jamboree & Main	5.7	5.7	5.7	5.4	5.2	5.4	5,3	5.4	5.3	5.4	5.4	5,7
134	Jamboree & Alton	5,6	5.6	5.7	5.5	5.2	5.5	5.2	5.4	5.4	5.4	5.4	5.7
98	Culver & Irvine Center	5.6	5.6	5.7	5.7	5.2	5.4	5.4	5.5	5.3	5.5	5.4	5.5
175	Jamboree & Michelson	5.7	5.4	5.4	5.6	5.1	5.3	5.3	5.3	5,3	5.4	5.4	5.4
151	Red Hill & MacArthur	5.7	5.5	5.5	5.6	5.0	5.4	5.4	5.4	5.3	5.4	5.3	5,7
155	Von Karman & Main	5.5	5.7	5.8	5.5	5.3	5.7	5.2	5,3	5.3	5.4	5,3	5.5
153	Red Hill & Main	5,5	5.6	5.5	5.6	5.2	5.4	5.2	5.3	5.2	5.2	5.2	5.2
174	Von Karman & Michelson	5.4	5.4	5.3	5.2	5.0	5. l	5. I	5.0	5.0	5.2	5.0	5.3
186	MacArthur & Campus	5.7	5.4	5.4	5.5	5.0	5.1	5.1	5.2	5.2	5.2	5.1	5.3
177	Culver & Michelson	5.4	5.5	5.2	5.2	5.0	5.2	5,1	5.0	5.1	5.1	5.3	5,4
	CITY OF LAGUNA BEACH												
299	Moulton & El Toro	5.4	5.4	5.4	5.5	5.1	5.3	5,2	5.4	5.0	5.3	5.1	5.1
	CTIY OF LAGUNA HILLS <sup>14</sup>												
280	El Toro & Avd. Carlota	5,4	5.4	5.4	5.3	5.0	5.3	5.0	5.1	5.2	5.2	5.1	5.3
Note:	* - Concentrations are in party per	r million (ne	em )										

Concentrations are in parts per million (ppm)

<sup>1 -</sup> RECI SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> REC8 W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient one-hour CO concentration, 6.1 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

Table 8.6-6

Phase 4 Alternative F – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

INTE	INTERSECTING STREET	, brci	. REC2	, BRGA	RECE	arcs.	RECE	EECT	RECE	1160	incle	RECII"	REC12"
	CITY OF ORANGE"												
345	Jamboree & Chapman	5.2	5.2	5.4	5.2	5,2	5.4	5.1	5.4	5.1	5.1	5.0	5.0
	CITY OF SANTA ANA"												
154	MacArthur & Main	5.3	5.4	5.4	5,4	5.2	5.2	5.2	5.3	5.2	5.2	5.0	5.2
152	Main & Sunflower	5.2	5.3	5.1	5,3	5.0	5.2	5.0	4.9	5.0	5.1	5,3	5.2
90	Grand & Edinger	5.2	5.2	5.2	5.4	5.2	5.3	5.0	5.2	5.0	5.2	5.2	5.0
	CITY OF TUSTING												
93	Newport & Edinger	5.4	5.3	5.1	5.4	5.0	5.0	5.0	5.0	5.0	5.2	5.0	5.2
115	Von Karman & Barranca	5.2	5.3	5.3	5.2	5.0	5.3	5.0	5.2	5.0	4.9	5.0	5.2
95	Tustin Ranch & Edinger	5.4	5.2	5.2	5.3	5.0	5.2	5.0	5,0	5.2	5.2	5.0	5.2
	CITY OF IRVENE"												
116	Jamboree & Barranca	3.8	3.9	3.6	3.7	3.4	3.5	3.4	3.5	3.5	3,5	3.5	3.8
156	Jamboree & Main	3.7	3.7	3.7	3.5	3.3	3.5	3,4	3.5	3.4	3.5	3.5	3.7
134	Jamboree & Alton	3.6	3.6	3.7	3.5	3.3	3.5	3.3	3.5	3.5	3.5	3.5	3.7
98	Culver & Irvine Center	3.6	3.6	3.7	3.7	3.3	3.5	3.5	3.5	3.4	3.5	3.5	3.5
175	Jamboree & Michelson	3.7	3.5	3.5	3.6	3.3	3.4	3.4	3.4	3.4	3.5	3.5	3.5
151	Red Hill & MacArthur	3.7	3.5	3.5	3.6	3.2	3.5	3.5	3.5	3.4	3.5	3.4	3.7
155	Von Karman & Main	3.5	3.7	3.7	3,5	3.4	3.7	3.3	3.4	3.4	3,5	3.4	3.5
153	Red Hill & Main	3.5	3.6	3.5	3.6	3,3	3.5	3,3	3.4	3.3	3.3	3,3	3.3
174	Von Karman & Michelson	3.5	3.5	3.4	3,3	3.2	3.3	3.3	3.2	3.2	3.3	3.2	3,4
186	MacArthur & Campus	3.7	3.5	3.5	3.5	3.2	3.3	3.3	3.3	3.3	3.3	3 3	3.4
177	Culver & Michelson	3.5	3.5	3.3	3,3	3.2	3.3	3.3	3.2	3,3	3.3	3.4	3.5
	CITY OF LAGUNA BEACH"	ı.											
299	Moulton & El Toro	3.5	3.5	3.5	3.5	3.3	3.4	3.3	3.5	3 2	3.4	3.3	3,3
	CITY OF LAGUNA HILLS"												
280	El Toro & Avd. Carlota	3.5	3.5	3.5	3.4	3.2	3.4	3.2	3.3	3.3	3.3	3.3	3.4
Note:	* - Concentrations are in parts p	er million (pp	am)										

<sup>1-</sup>RECL SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> RECS W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient eight-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

<sup>14 -</sup> The ambient eight-hour CO concentration, 2.9 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

# 8.6.4.8 Hydrology and Water Quality

Since all improvements would occur within the existing developed area of JWA, this alternative will incur hydrology and water quality impacts at JWA similar to the Proposed Project. No aviation uses would be developed at the El Toro site. However, this alternative would lead to adoption of a nonaviation plan for the El Toro site and would not, therefore, avoid or substantially lessen impacts of the project.

# 8.6.4.9 Biological Resources

Since the biological resource component at JWA is very limited, no direct or indirect impacts would be expected at JWA under Alternative F. For indirect impacts, the biological resource issues would not be substantially different from the Proposed Project. There would be some slight differences in impacts as a result of noise exposure and aircraft overflights since the aircraft operations differ at JWA. Noise and overflight characteristics are different between Alternative F and the Proposed Project, because the 60 CNEL contour is longer for Alternative F. However, this difference would not result in adverse impacts to biological resources in the Newport Back Bay. This alternative would not, therefore, avoid or substantially lessen the impacts of the project.

#### 8.6.4.10 Public Services and Utilities

This alternative would have greater adverse impacts at JWA than the Proposed Project. Due to the plan to expand JWA in Alternative F, a need for increased fire and emergency medical, police services, and transit to the area would likely arise. Like the Proposed Project, mitigation measures prescribed in section 4.10 (Public Services and Utilities) would reduce staffing impacts to below a level of significance.

As described in Section 4.10 (Public Services and Utilities), the Proposed Project is not anticipated to result in significant adverse impacts related to utilities. Alternative F could also be served with utilities without significant adverse impacts after mitigation, similar to conditions under the Proposed Project. Depending on the specific land uses and utility services and infrastructure needs associated with Alternative F, a utility infrastructure different from that anticipated under the Proposed Project may be necessary to most effectively provide utility services under this alternative. Mitigation similar to that for the Proposed Project would reduce adverse impacts of this alternative related to utilities infrastructure and services to below a level of significance. This alternative would not avoid or substantially lessen the impacts of the project.

# 8.6.4.11 Natural Resources and Energy

This alternative would have greater adverse impacts at JWA than the Proposed Project. As discussed in Section 4.11 (Natural Resources and Energy), the Proposed Project would not result in significant adverse impacts related to natural resources, with the exception of significant adverse impacts to agricultural resources on the El Toro site, which could not be mitigated to below a level of significance. This alternative could reduce or avoid the project impacts on agricultural soils depending on the reuse alternative selected for the El Toro site. Section 8.3 analyzes the impacts of a nonaviation alternative for the El Toro site. There are no natural or agricultural resources at JWA.

Implementation of the Proposed Project also would result in a less than significant increase in regional energy consumption, associated primarily with construction and operation of a new international airport at the MCAS El Toro site, as compared to existing conditions. Under this alternative, the temporary increase in energy consumption associated with construction activities at MCAS El Toro would be replaced by a lower level of effort to expand JWA. From a regional standpoint, this alternative also would realize lower energy consumption levels associated with airport operations compared with the two-airport system of the Proposed Project. This energy savings would be offset somewhat, however, by increased highway travel, as the shortfall in forecasted demand for air travel service forces air travelers to drive to other airports within the region than would be required with the Proposed Project.

The ultimate build out and phased development of Alternative F would require more intense construction efforts, and entail a higher level of operations at JWA than the Proposed Project. Therefore, the impacts related to natural resources and energy would be greater to JWA under Alternative F because of the higher level of operations and higher energy consumption. In conclusion, this alternative would not avoid or substantially lessen impacts compared to the project.

# 8.6.4.12 Aesthetics, Light and Glare

This alternative would have greater impacts at JWA compared to the Proposed Project. With mitigation measures, the impacts of this alternative would be reduced but would remain significant. This alternative would lead to adoption of a nonaviation reuse plan for the El Toro site similar to the ETRPA Nonaviation Alternative. Refer to Section 8.3 for an analysis of this alternative.

Physical changes to the JWA site under Alternative F include extension of the main commercial runway (Runway 1L-19R) from 5,700 feet to 6,800 feet, which is the maximum extension allowable within the existing property boundary of the airport. The existing terminal concourse would be lengthened by several hundred feet on the north and south ends to provide additional aircraft gates. An additional terminal would be created to the south and connected to the existing terminal. Additional parking would be provided in all

remaining areas around the new terminal, and additional long-term parking would be provided in the general aviation areas on the southeast and southwest of the airport. Additional off-site parking would likely be necessary for this alternative. Expansion of access roadways to JWA would be necessary to accommodate the expanded passenger service under Alternative F. Expansions would potentially include the existing direct access from SR-55, Campus Drive, and MacArthur Boulevard at Michelson Drive.

Alternative F would increase the intensity of development on and around the JWA site, whereas the Proposed Project would make no substantial aesthetic changes to the site (a slight reduction of commercial service is proposed). The potential effects of light and glare under this alternative would be greater than those of the Proposed Project due to the increased services at the airport, especially evening (the nighttime curfew is assumed to continue for commercial flights) aircraft light and glare impacts on nearby residential areas (e.g. Santa Ana Heights). Ground level light and glare impacts would be reduced to a level of insignificance with County Standard Condition of Approval LG 1. In conclusion, this alternative would not avoid or substantially lessen impacts compared to the project at the JWA site.

#### 8.6.4.13 Cultural Resources

Improvements at JWA under Alternative F would occur within the physical confines of the existing airport site. Like the Proposed Project, there would be no additional or new effects on cultural resources since there are no known archaeological, paleontological or historic resources on the already developed JWA airport property. Approval of this alternative would lead to adoption of a nonaviation plan for the El Toro site, which could potentially impact cultural resources more than the Proposed Project. Therefore, this alternative would not avoid or substantially lessen impacts compared to the project.

#### 8.6.4.14 Recreation

Under Alternative F, noise impacts on the use of area recreational facilities in the JWA area would increase due to the enlarged 65 dB CNEL contours resulting from the increased commercial aviation and cargo services under this alternative. This alternative would avoid aviation noise impacts on recreation uses at the El Toro site but would increase aviation impacts from aircraft noise on the use of trails, parks, and other recreational facilities at the JWA site. Approval of this alternative would lead to adoption of a nonaviation plan for the El Toro site similar to the ETRPA Nonaviation Alternative. Refer to Section 8.3 for an analysis of the nonaviation plan impacts.

# 8.6.4.15 Public Health and Safety

# Aviation Safety

Compared to the Proposed Project, under Alternative F there would be an increase of approximately 97,900 air carrier and air cargo operations and a decrease of approximately 352,400 general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 145.0% to reflect the number of increasing aviation activity diverted from OCX to JWA. The potential accident risks for general aviation at JWA would decrease by 98.2%. Since there is no aviation activity at OCX under this alternative, there would be zero aviation risks. Compared to the Proposed Project relative to on-airport and off-airport fatal accidents per million operations, there would be no significant adverse impacts under this alternative related to aviation safety at the MCAS El Toro site or at JWA.

Compared to the existing conditions, there would be an increase of approximately 75,392 air carrier and air cargo operations and a decrease of approximately 321,024 general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 83.8% to reflect the number of increasing operations and the potential general aviation accident risks would decrease by 98.0% correspondingly.

#### 8.6.4.16 Hazardous Materials and Hazardous Wastes

Under this alternative, aviation facilities would be expanded to the maximum available capacity within the existing airport property limits at JWA. Hazardous waste handling practices would remain unchanged at JWA; however, an increase in hazardous materials consumption, particularly jet fuel, commensurate with the expansion of aviation facilities would result.

Like the Proposed Project, any use of hazardous materials and/or generation of hazardous waste at JWA under Alternative F would be regulated by applicable State law, federal law, and regulations pertaining to worker protection, hazardous materials storage and use, and hazardous waste generation and disposal. Implementation of these regulations will reduce potential impacts associated with the presence of these hazardous substances to below a level of significance. This alternative would lead to adoption of a nonaviation alternative at the El Toro site similar to the ETRPA Nonaviation Alternative. Refer to Section 8.3 for an analysis of nonaviation impacts.

#### 8.6.4.17 Socioeconomics

Under this alternative, 8,500 jobs would be generated at JWA in 2020, which represents a net increase of 6,400 jobs at JWA over existing 1998 conditions. The number of jobs generated at JWA under this alternative would therefore be substantially greater, at approximately 5,200 jobs, than the number of jobs generated at JWA under the Proposed Project. Under this alternative, it is assumed that the MCAS El Toro site would develop with a range of nonaviation uses similar to those shown in the ETRPA Nonaviation Plan. This development would result in an estimated 50,700 jobs, 13,600 persons, and 5,900 housing units being located on the site in 2020. This level of activity is significantly higher than the level anticipated under the Proposed Project. In total, this alternative is projected to generate 59,000 jobs, 13,600 persons, and 5,900 housing units on the project site. This figure is significantly higher than the number of jobs, persons, and housing units expected under the Proposed Project.

As with the Proposed Project, economic activity on the JWA and MCAS El Toro sites, as well as expenditures by visitors arriving by air at JWA, would stimulate additional off-site job growth. Given the higher total number of on-site jobs for MCAS El Toro and JWA and the lower number of air passengers served by this alternative, the number of off-site jobs stimulated by the airport system could be similar to the Proposed Project.

Given the greater number of jobs generated under Alternative F, at 59,000 jobs versus 29,500 jobs under the Proposed Project, the magnitude of impacts related to induced growth or concentration of population and employment in the area, and increasing demand for housing, including low and moderate income housing, would be significantly higher than under the Proposed Project. However, the additional demand for housing created by project related employment would be partially offset by the housing to be developed on the MCAS El Toro site under this alternative. The employment, population, and housing projections under Alternative F would also be inconsistent with the adopted regional forecasts, as under the Proposed Project. In conclusion, this alternative would not avoid or substantially lessen Proposed Project impacts.

# 8.6.4.18 Risk of Upset

The ultimate build out and phased development of this alternative would entail no aviation reuse at the El Toro site and a market-driven increase in operations at JWA. General aviation operations would be eliminated and the main JWA runway would be extended to accommodate expanded commercial service.

This alternative would generate an increase in demand for jet fuel at JWA, as well as associated tank truck jet fuel transport operations since JWA (unlike OCX) is not served by pipelines. Additional requirements for fuel storage capacity on the JWA site also would be required. Under this circumstance, the risk of upset potential at JWA would be higher than under the Proposed Project. As discussed in Section 4.18 (Risk of Upset), the Proposed

Project would not result in significant adverse impacts to public health and safety resulting from project-related risk of upset conditions after mitigation. Therefore, this alternative would not avoid or substantially lessen impacts compared to the project at JWA, but would avoid aviation impacts at the El Toro site.

# 8.6.5 Feasibility

This alternative is feasible from a physical standpoint in that the existing main runway can be extended to serve the intended market role (short- and medium-haul and limited long-haul). The existing short general aviation runway can also be converted to a taxiway. The existing terminal building can be expanded to accommodate the projected demand. The present facilities can be expanded to meet projected demand in this alternative with some exceptions. Some vehicle parking would be located off-airport. In addition, facilities for inflight catering would be located off-airport. No space would be available at the airport for aircraft maintenance.

From an operational standpoint, the single runway for JWA is capable of supporting a limited long-haul market role. However, it is not feasible for the runway in this alternative to accommodate operations by unlimited long-haul or full international service. General aviation operations can be accommodated on a very limited basis, however, the more than 500 JWA based general aviation aircraft must be relocated to other general aviation airports in the region. Furthermore, since the airport would be reduced to a single runway, it could be subject to periods of closure if the runway was under repair or otherwise unusable.

From a market perspective, the alternative allows some growth in passenger service beyond today's passenger levels and some growth in all-cargo service. It does not accommodate a substantial portion of Orange County's long-term air travel needs, including general aviation demand.

The costs for Alternative F, described in ASMP Technical Report 6, Volume 2, Appendix D are described as "order of magnitude" because they were prepared without the benefit of a master plan. They can be used, however, in a general comparison with the capital costs of the Proposed Project. The order of magnitude cost for Alternative F was estimated at \$350 million. The net revenue for this alternative would be expected to be less than other aviation "build" alternatives due to the lower level of airport activity.

In conclusion, this alternative would be physically feasible, but would result in operational and development limitations, and would not meet the market objectives of the project.

#### 8.6.6 Conclusions

#### Alternative F does not:

- Encourage growth of air service opportunities;
- Implement two airport system;
- Enhance GA opportunities for O.C. residents; and
- Take advantage of existing land use restrictions.

In comparison to the Proposed Project, Alternative F would result in significant regional air quality impacts and short-term construction impacts greater than the Proposed Project, greater local air quality impacts at JWA compared to the Proposed Project, and additional land use impacts, traffic impacts, noise impacts, public services and utilities impacts, natural resources and energy impacts, and aesthetics, light and glare impacts at JWA than under the Proposed Project.

This alternative would lead to adoption of a nonaviation alternative for the El Toro site similar to the ETRPA Nonaviation Alternative. Refer to Section 8.3 for conclusions regarding nonaviation uses.

# 8.7 ALTERNATIVE G: JWA – LIMITED INTERNATIONAL; NO AVIATION REUSE AT FORMER MCAS EL TORO

This section presents the potential impacts of Alternative G as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

This alternative was selected for analysis because it has the potential to avoid aircraft noise and aircraft air quality emission impacts at the El Toro site while still feasibly attaining some of the objectives of the Proposed Project.

#### 8.7.1 Aviation Uses

In Alternative G, the former MCAS El Toro is assumed to be a nonaviation use and JWA provides short, medium, and long-haul domestic and international air passenger service for an estimated 25.0 MAP, nine percent (2.2 MAP) of which are passengers with connecting flights. JWA is also forecast to annually handle approximately 40.0 thousand tons of international cargo and 1.23 million tons of domestic cargo. JWA would not be constrained by existing limits on passengers or aircraft operations under this alternative, and it is assumed that the airport would accommodate all of the demand in these categories projected for the airport beyond 2005 (estimated to be approximately 25 MAP in 2020 as described in the ASMP). To enable the airport to handle this demand, a major program for the acquisition of property would be required. Property to be acquired would include existing developed property north of JWA, extending the airport boundary west to SR-55, and a triangular shaped area to the east bound by Campus Drive, MacArthur Boulevard, Bristol Street, and Jamboree Road. New runway facilities, terminal facilities, parking, cargo facilities, and support facilities would be necessary. The closure of the general aviation runway, a 2,300-foot extension to the main runway, and a new 6,700-foot runway are envisioned in this alternative to accommodate the projected demand. Alternative G would include 52 jet aircraft gates, 11 commuter aircraft gates, 29 RON aircraft parking spaces, 21,500 vehicle parking spaces, and 1.84 million square feet of terminal area (Figure 8-8 depicts Alternative G). There would be no aviation reuse of MCAS El Toro.

Under this alternative, there would be no aviation reuse of MCAS El Toro. Accordingly, the environmental analysis of this alternative focuses on the impacts of the alternative at JWA. However, if Alternative G were selected and implemented, it would lead to adoption of a nonaviation plan for the El Toro site, possibly one similar to the ETRPA Nonaviation Alternative analyzed in Section 8.3. To understand the full impacts of Alternative G along

with the ETRPA Nonaviation Alternative, for example, the reader should review the impacts of both alternatives as addressed in this section.

# 8.7.2 Nonaviation Revenue Support Uses

Alternative G does not propose nonaviation uses at JWA and does not include physical changes at El Toro. However, as noted above, approval of Alternative G lead to the adoption of nonaviation uses at El Toro, possibly similar to the ETRPA Nonaviation Plan Alternative addressed in Section 8.3.

# 8.7.3 Attainment of Project Objectives

This alternative would not meet the general project objectives for reuse of MCAS El Toro. Alternative G will further most of the aviation related objectives, but not to the same extent as the Proposed Project. This alternative does not meet the general aviation objectives. It will have a major impact on general aviation as the more than 500 general aviation aircraft now at JWA would have to be relocated. This alternative will not meet the existing land use or General Plan implementation aviation objectives. This alternative does not encourage service opportunities such as international service, and this alternative does not implement the two airport system to avoid impacts of a single system.

# 8.7.4 Environmental Impacts of Alternative G

#### 8.7.4.1 Land Use

This alternative would have no land use impacts at the MCAS El Toro site since all development would occur at JWA. This alternative would have significant adverse land use impacts at JWA.

Under Alternative G, JWA would be expanded to accommodate 25 MAP, requiring the acquisition of a considerable amount of developed land outside the current property boundary. The land acquisition would occur primarily to the west, extending the JWA boundary to SR 55, and to the southeast, adding a large triangular shaped area south of Campus Drive and MacArthur Boulevard.

There are residential land uses west of SR 55, the western boundary of JWA under Alternative G. Alternative G proposes 60 acres of parking at JWA along SR 55. Parking lot lighting has the potential to be a significant impact to the nearby residential uses, if not appropriately mitigated. The residences would not be directly impacted by the airport expansion, as the existing freeway separates the two uses.

JWA acquisition to the southeast under this alternative would include the addition of a triangular shaped area of land bound by Campus Drive, MacArthur Boulevard, and Bristol

Street. A small portion of this area would front on Jamboree Road. The existing land uses in this triangular area are primarily commercial, offices and light industrial. A large number of businesses, primarily commercial offices, would be displaced. Campus Drive and Birch Street would no longer extend through this area under Alternative G.

The existing JWA boundary along Campus Drive south of the MacArthur intersection places the existing airport facilities across the street right-of-way from business park uses, generally office and light industrial. The acquisition of the additional area east of Campus Drive would result in a similar situation, with business park uses across from the MacArthur Boulevard boundary of the acquisition area. Under this alternative, the SR 73/Bristol Street boundary of JWA is extended further south. The Bristol Street right-of-way forms a large barrier between the airport uses in the acquisition area and the Bristol Street commercial frontage and residential uses behind the frontage.

The extended runways needed under Alternative G will result in an extension of the ROFA area to the south and the tunneling of SR 73 under the ROFA. The significant transportation improvements needed in this area may result in the acquisition of additional properties along Bristol Street, and potentially impact the business frontage, the existing golf course, and possibly the residential areas behind the gold course. The extent of these impacts is not known at this time, but they are potentially significant.

The existing long-term parking to the north of JWA and the 405 freeway would remain. Some additional property acquisition for parking could be required. The primary potential off-site effects of the long-term parking use are nighttime lighting and vehicular noise, which do not conflict with the typical daytime use of nearby office buildings; therefore, the long-term parking use is compatible in the business park setting where it is located.

The land acquisition needed for the extension of the ROFA to the south and related transportation improvements under this alternative would involve the acquisition of property south of Bristol Street. In addition, two major JWA acquisition areas, west to SR 55 and southeast to SR 73, involve the displacement and relocation of a large number of existing businesses. The extent of the acquisition south of Bristol and the related impacts are not known at this time. The two major acquisitions to the west and southeast will disrupt businesses, but will not disrupt or divide residential communities, because there are no residences in these areas and, therefore, no established residential neighborhoods.

The new boundaries of JWA under this alternative are major streets or highways, including SR 55 on the west and SR 73/Bristol Street and MacArthur Boulevard on the east. These major streets create a physical separation between the JWA acquisition areas and adjacent uses. The adjacent uses across MacArthur Boulevard are primarily office and light industrial that would not be significantly affected by operational characteristics of the airport, such as stationary noise sources, periodic emissions of fumes or odors related to engine startups or testing, and lighting for nighttime activities. The land uses across SR73/Bristol are primarily community commercial uses with residential neighborhoods behind the Bristol Street frontage. The land use impacts associated with this alternative, such as stationary

noise, fumes, and lighting, may exceed the threshold of significance by creating substantial incompatibilities between this alternative's land uses and adjacent existing and planned land uses. These impacts may be mitigatable through such measures as screen walls, light fixture hoods and/or automatic timers, the careful placement of buildings and building openings, and other measures.

Overall, the land use impacts of Alternative G on the JWA area are greater than those of the Proposed Project. With respect to El Toro, as noted above, this alternative would lead to adoption of a nonaviation plan for the El Toro site. If for example, the ETRPA Nonaviation Alternative were adopted, this would result in land use impacts at the El Toro site greater than the Proposed Project. Therefore, this alternative would not avoid or substantially lessen Proposed Project land use impacts.

### 8.7.4.2 General Plan Consistency

Compared to existing conditions, this alternative would have significant general plan consistency impacts at the El Toro site and at JWA.

Alternative G involves significant changes to JWA that will result in new runways and expanded noise contours, among other changes. These modifications necessitate amendments to the AELUP and the Noise, Land Use, Safety, and Public Services and Facilities Elements of the County General Plan. The change in airport boundaries will require changes to the Land Use, Noise, and related general plan elements of adjacent jurisdictions, including the Cities of Costa Mesa, Newport Beach, Irvine, Santa Ana, and Tustin.

With respect to the El Toro site, Alternative G is inconsistent with the Measure A provisions of the County General Plan and would require more extensive amendments to adopted plans than the Proposed Project. Therefore, the impacts of Alternative G related to general plan consistency are greater than the impacts projected to occur under the Proposed Project. Adoption of the ETRPA Nonaviation Alternative for the El Toro site also would be inconsistent with the County General Plan. Therefore, this alternative would not avoid or substantially lessen Proposed Project general plan consistency impacts.

# 8.7.4.3 Transportation and Circulation

Compared to existing conditions, this alternative would have significant adverse transportation and circulation impacts at JWA. This alternative would result in adoption of a nonaviation plan for the El Toro site likely similar to the ETRPA Nonaviation Alternative (Section 4.3), which would have significant unavoidable adverse impacts. Therefore, this Alternative would not avoid or substantially lessen project impacts.

The AM and PM peak hour and ADT traffic generated by JWA with build out of Alternative G is summarized in Table 8.7-1. Refer to Section 12.0 in the 1999 Traffic Analysis Technical Report for detailed information on the methodology applied to produce trip generation estimates for Alternative G.

Table 8.7-1
Trip Generation Summary - Alternative G

	AN	l Peak Ho	ur	P	M Peak H	lour	ADT	Existing	
Project Component	ln	Out	Total	In	Out	Total		ADT	
JWA	3,533	2,575	6,108	3,579	3,422	7,001	116,424	47,450	

Table 8.7-2 compares, in summary, the Alternative G highway impacts to the existing conditions and existing conditions plus Proposed Project. There is minimal comparison between the existing plus project versus the Alternative G impacts due to the large differences between the scope of the project (two airports) and the alternative (one airport). In addition, Alternative G would require the removal of a large portion of the existing development surrounding JWA, along with the removal of existing trip generators.

In addition, a comparison of the impacts of Alternative G to the impacts of the Proposed Project during the phasing years may also be made. As discussed in detail in Section 4.3.6.6 of this Draft-EIR No. 573, as supplemented, under the Proposed Project phasing years, four intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 1 conditions (2005), five intersection locations, two arterial roadway segments, one continuous freeway mainline segment and one freeway ramp would be significantly impacted under Phase 2 conditions (2010), and nine intersection locations, two arterial roadway segments, one continuous freeway mainline segment and two freeway ramps would be significantly impacted under Phase 3 conditions (2015). At Phase 4 build out, the Proposed Project would result in significant impacts not previously identified to four freeway/tollway mainline segments and four freeway/tollway ramps. See Draft Supplemental Analysis, Section 4.3.6.5. In each case, however, the identified impacts will be mitigated to a level below significant during the applicable phasing year (see Section 4.3.7.2, Table 4.3-20).

Table 8.7-2
Summary Comparison of Traffic Impacts for Alternative G to Existing Conditions and Existing Conditions Plus Project

Existing Conditions	Existing Conditions	Alternative G With Existing
Deficient Highway Facilities	Plus Proposed Project Impacts	Plus Committed Facilities
Location	Location	Location
INTERSECTIONS	IMPACTED INTERSECTION	IMPACTED INTERSECTIONS
Newport (NB) & Del Mar	Bake & Portola	Red Hill & Main
El Toro & SR-73 NB Ramps	Sand Canyon & Trabuco	MacArthur & Jamboree
Campus & N. Bristol	Bake & I-5/I-405 SB Ramps	MacArthur & Von Karman
Jamboree (SB) & Walnut	Bake & Rockfield	
Jamboree & I-405 NB Ramps	Jeffrey & Alton	IMPACTED FREEWAY RAMPS
Jeffrey & I-405 NB Ramps	Jeffrey & I-405 NB Ramps	I-405 at Jamboree (SB Off-Ramp)
Red Hill & MacArthur	Jeffrey & I-405 SB Ramps	I-405 at MacArthur (NB On-Ramp)
Irvine Center & Lake Forest	Jeffrey & Walnut/I-5 SB	I-405 at MacArthur (NB Off-Ramp)
Bake & Jeronimo	Sand Canyon & I-5 NB Ramps	I-405 at Sand Canyon (NB Direct On- Ramp)
El Toro & Avd Carlota	Sand Canyon & I-5 SB Ramps	
La Paz & Cabot/I-5 SB	Irvine Center & Lake Forest	IMPACTED FREEWAY SEGMENTS
Los Alisos & Muirlands	Bake & Irvine/Trabuco	SR-55 (I-5 to south of SR-73)
Alicia & Jeronimo	Bake & Toledo	
Alicia & Muirlands	Los Alisos & Muirlands	
La Paz & Muirlands/I-5 NB	Alicia & Jeronimo	
Red Hill & Edinger	Newport & Old Irvine	
Red Hill & Sycamore		
Red Hill & Walnut	IMPACTED ARTERIAL ROADS	
	Laguna Canyon (I-405 to SJHTC)	
ARTERIAL ROADWAYS	Laguna Canyon (south of El Toro)	***************************************
Portola (Sand Canyon to Foothill Toll Road)	Culver (Bryan to Trabuco)	
Laguna Canyon (I-405 to SR-73)		
Laguna Canyon (south of El Toro)	IMPACTED FREEWAY RAMPS	
Culver (Bryan to Trabuco)	I-5 at Culver (SB Off-Ramp)	
Michelson (Carlson to Harvard)	I-5 at Sand Canyon (NB On-Ramp)	
	I-5 at Sand Canyon (SB Off-Ramp)	
FREEWAY RAMPS	I-405 at Sand Canyon (NB Direct On-Ramp)	
I-5 at Culver (SB Off-Ramp)		
I-405 at Jamboree (SB Off-Ramp)	IMPACTED FREEWAY SEGMENTS	
I-405 at MacArthur (SB On-Ramp)	I-5 (Jeffrey to north of SR-55)	
I-405 at MacArthur (NB On-Ramp)		•
I-405 at MacArthur (NB Off-Ramp)		
SR-55 at Dyer (SB On-Ramp)		
SR-55 at Dyer (NB Off-Ramp)		
SR-55 at MacArthur (SB Direct		
On-Ramp)		
SR-55 at MacArthur (NB Direct		
On-Ramp)		
SR-55 at MacArthur (SB Off-Ramp)	<b>1</b>	

Existing Conditions Deficient Highway Facilities	Plus Proposed Project	tions 4
Location	Location	Location
FREEWAY SEGMENTS		
I-5 (Culver to north of SR-55)		
I-5 (Alton to I-405)		
I-5 (El Toro to La Paz)		
I-405 (MacArthur to SR-133)		
SR-55 (I-5 to SR-73)		

Abbreviations:

NB-northbound SB-southbound EB-eastbound WB-westbound The acquisition of a substantial amount of developed property and roadway infrastructure around JWA (i.e., outside the existing JWA property boundary) in the Cities of Costa Mesa and Newport Beach would be required to accommodate the runway extension and airport facility requirements for expanding JWA under Alternative G. The elimination of existing and planned development in the Costa Mesa and Newport Beach areas that would need to be acquired would result in a reduction of approximately 75,000 average daily trips, 6,200 AM peak hour trips and 6,900 PM peak hour trips in the immediate vicinity of JWA under year 2020 conditions. The JWA access concept anticipated in the analysis of Alternative G takes into consideration the effects of the existing roadway infrastructure that would be acquired, as well as the impact that the runway extension required under Alternative G would have on the Bristol Street/SR73 corridor immediately south of JWA. The JWA site access/roadway reconfiguration plan for Alternative G is described as follows for three general areas surrounding JWA.

# Northeast (I-405 Freeway/MacArthur Boulevard)

Under Alternative G, the JWA airport terminal area would be expanded along the west side of MacArthur Boulevard to a point south of the existing MacArthur Boulevard/Von Karman Avenue intersection. Such an expansion of the air terminal and other associated airport facilities would eliminate the existing City of Newport Beach land uses and roadway system in the area bounded by MacArthur Boulevard, Campus Drive, North Bristol Street, and Jamboree Road. Master Plan of Arterial Highway (MPAH) facilities that would be eliminated in this area include Campus Drive and Birch Street between MacArthur Boulevard and North Bristol Street. The existing direct connector ramps between SR-55 and the terminal would be retained as would the terminal access provided from MacArthur Boulevard at the Michelson Drive and I-405 southbound ramp intersections. In addition, terminal entryways would be provided from MacArthur Boulevard at the Campus Drive and Von Karman Avenue intersections.

# South (Bristol Street/SR-73)

In addition to being affected by the JWA terminal expansion mentioned previously, the Bristol Street/SR-73 corridor south of JWA would be impacted by Runway Object Free Area (ROFA) requirements associated with the extended JWA runways needed under Alternative G. The ROFA area would extend across SR-73 to a point south of existing South Bristol Street. It is assumed that SR-73 would tunnel under the ROFA area in its existing alignment and that North Bristol Street and South Bristol Street would be reconstructed south of the ROFA as a standard two-way primary arterial road from Red Hill Avenue to Birch Street and that Irvine Avenue would intersect the realigned Bristol Street from the south but would not extend north of Bristol Street. North Bristol Street and South Bristol Street east of Birch Street are assumed to remain in place as they are currently constructed. The elimination of Campus Drive and the realignment of Bristol Street would also result in the elimination of the existing SR-73 on- and off-ramps northwest of Campus Drive. It is anticipated that a new northbound SR-73 on-ramp would be constructed from North Bristol Street south of

Birch Street but that a new southbound SR-73 off-ramp would not be constructed since an off-ramp to South Bristol Street currently exists south of Birch Street.

#### West (Red Hill Avenue/SR-55)

In Alternative G, the City of Costa Mesa area bounded by the current JWA property boundary and I-405, SR-55 and SR-73 would be acquired and utilized for air cargo operations, various airport support facilities and public parking areas. It is anticipated that Red Hill Avenue would be retained, although realigned, as a four-lane arterial through this area, as would the connections of Paularino Avenue and Baker Street between Red Hill Avenue and the existing SR-55 collector/distributor roadway system. Entryways to the public parking and air cargo areas would be provided from Red Hill Avenue at the Paularino Avenue and Baker Street intersections.

Approval of this alternative would lead to adoption of a nonaviation plan for the El Toro site such as the ETRPA Nonaviation Plan Alternative analyzed in Section 8.3. The traffic impact analysis for the ERTPA Nonaviation Plan Alternative concludes that traffic impacts cannot be mitigated to below a level of significance. Therefore, a combination of Alternative G and a nonaviation plan for the El Toro site would result in a total traffic impact that could not be mitigated to acceptable levels.

#### 8.7.4.4 Noise

Compared to existing conditions, this alternative would have significant adverse noise impacts at JWA because of the increased use of this airport. Under Alternative G, a large increase in the land area affected by the 60 and 65 CNEL noise contours would occur in comparison both to the 1998 and the 1985 Master Plan airport noise contours. The total number of daily jet carrier operations (arrivals and departures) would be more than 900 per day under this alternative (see Table 8.1-1). Under this alternative, the forecast number of commercial aircraft operations could not be accommodated unless the existing nighttime noise restrictions at JWA were removed. As discussed earlier, the CNEL calculation factors in the number of daily operations and assigns a "penalty weighting" to operations occurring during the nighttime hours (10 p.m. to 7 a.m.). However, the substantial increase in the number of operations, particularly during nighttime hours, is considered a significant impact of Alternative G independent of the CNEL computation. See Table 8.4-4, which shows the land use comparison with noise contours for 1998 and year 2020 alternatives for JWA. Also see Figure 8-9, which depicts noise contours for Alternative G.

Adoption of this alternative would probably lead to approval of a nonaviation land use plan such as the ETRPA Nonaviation Alternative analyzed herein for the El Toro site. For analysis of the noise impacts of a nonaviation land use plan, refer to Section 8.3. In conclusion, compared to the Proposed Project, this alternative would avoid aircraft noise impacts at the El Toro site and surrounding areas. However, this alternative would generate

substantially greater impacts than the Proposed Project in the vicinity of JWA, including sleep disturbance.

# 8.7.4.5 Air Quality

Compared to existing physical conditions (1998), Alternatives would have significant construction, regional and local air quality impacts. This alternative would not avoid or substantially lessen the Proposed Project's impacts because: a) local impacts at JWA would be increased, and b) this alternative would lead to adoption of a nonaviation pan for the El Toro site, which could have local CO hot spot impacts not found under the Proposed Project and regional impacts that are greater than the Proposed Project in light of the increase in VMT caused by demand being met at other regional airports outside Orange County. See Section 4.2 for an analysis of air quality impacts of a potential nonaviation plan for the El Toro site.

# Short-Term (Construction) Impacts

Under this alternative, new runway facilities, terminal facilities, parking, cargo facilities, and support facilities at JWA would be necessary. Construction emissions would be greater than those of the Proposed Project at JWA. Under this alternative, there would be no aviation reuse at MCAS El Toro. Therefore, construction emissions would likely be similar at this site to those associated with the Nonaviation Alternative. When the construction impacts for the Nonaviation Alternative are added to construction emissions at JWA, the total construction emissions could exceed the Proposed Project and would be significant and unavoidable.

# Operational Air Quality Impacts

#### **Emissions Inventors**

Under this project alternative, JWA will serve up to 25 MAP at build out. Total annual passengers and total annual aircraft LTO operations are less than those of the Proposed Project. Direct air pollutant emissions associated with airport operations, including aircraft, GSE, energy consumption, and vehicular trips, are shown below in Table 8.7.3 for this alternative.

Table 8.7-3
2020 Alternative G – Project Direct Air Pollutant Emissions (pounds/day)

, 54	CO	NOx	ROC <sup>1</sup>	SOx	PM <sub>10</sub>
Aircraft	4,021.51	10,622.78	368.32	672.44	125.82
ocx					
JWA	4,021.51	10,622.78	368.32	672.44	125.82
GSE/APU	17,588.19	1,203.18	462.52	27.61	52.78
ocx					
JWA	17,588.19	1,203.18	462.52	27.61	52.78
Fuel Storage/Dispensing	**		27.15	**	
ocx					
JWA			27.15		
Airport Roadways	344.51	68.41	15.64	4.07	4.48
OCX	-	400 1004			
JWA	344.51	68.41	15.64	4.07	4.48
Airport Parking	249.31	20.21	25.27	8.10	0.91
OCX				***	
JWA	249.31	20.21	25.27	8.10	0.91
Energy Consumption	94.20	542.20	5.00	55.60	18.50
ocx		***			
JWA	94.20	542.20	5.00	55.60	18.50
Vehicular Traffic	10,288	4,161	809	268	2,129
ocx		<b>-</b> ≠	~~	***	
JWA	10,288	4,161	809	268	2,129
Total	32,586	16,608	1,703	1,036	2,331

Source: CH2M Hill and LSA Associates, Inc., 2001

Regional air pollutant emissions, including airport operations at other airports in the region and VMT required for air travel passengers to get to these airports, are shown in Table 4.7-2 for this alternative. Regional vehicle miles traveled for this alternative would be higher than existing conditions (1998) and the Proposed Project (Phase 4) because of the regional diversion issue. Therefore, this alternative would result in significant regional emissions that are greater than under the Proposed Project. These regional emissions, however, would be less than under the No Project Alternative. See Table 8.7-4.

<sup>&</sup>lt;sup>1</sup> ROC emissions were obtained by multiplying HC emissions reported by EDMS by a factor of 1.14.

<sup>&</sup>lt;sup>2</sup> SO<sub>x</sub> emissions are not reported by the URBEMIS7G model.

Table 8.7-4
Regionwide Emissions Inventory Alternative G Phase 4
(Pounds/Day Unless Noted)

		co	NOX	ROC	SOX	PM <sub>10</sub>
Aircraft	El Toro	**	~-			No sin
	JWA	4,021.51	10,622.78	368.32	672.44	125.82
	Other Airports	71,705.90	73,399.17	9,966.37	5,590.88	809.05
	Total Regional	75,727.41	84,021.95	10,334.69	6,263.32	934.87
GSE	El Toro	**				m m.
	JWA	17,588.19	1,203.18	462.52	27.61	52.78
	Other Airports	93,222.50	9,361.15	2,759.22	606.34	343.25
	Total Regional	110,810.69	10,564.33	3,221.74	633.95	396.03
Energy	El Toro					
	JWA	94.20	542.20	5.00	55.60	18.50
	Others	579.00	3,331.00	31.00	341.00	114.00
	Total Regional	673.20	3,873.20	36.00	396.60	132.50
Fuel	El Toro	***				
4	JWA	***		27.15		
	Other Airports			488.50		
	Total Regional	***		515.65		«···
Airport Roadways	El Toro	**	m-m			**
	JWA	344.51	68.41	15.64	4.07	4.48
	Other Airports	3,214.75	652.36	147.92	39.26	53.34
	Total Regional	3,559.26	720.77	163.56	43.33	57.82
Airport Parking	El Toro	ann Agus	my va-		**	***
	JWA	249.31	20.21	25.27	8.10	0.91
	Other Airports	3,408.00	293.44	47.04	90.52	35.72
	Total Regional	3,657.31	313.65	72.31	98.62	36.63
Roads	El Toro	***				**
	JWA	10,288.00	4,161.00	809.00	268.00	2,129.00
	Others Airports!	2,763,687.00	492,434.00	71,041.00	48,776.00	<u>7,691.00</u>
		2,745,343.00	489,026.00	68,947.00	49,025.00	7,761.00
	Total Regional!	<u>2,773,975.00</u>	496,595.00	71,850.00	49,024.00	9,820.00
		2,755,631.00	493,187.00	69,756.00	49,293.00	<del>9,890.00</del>
TOTAL (pounds/e	day)	2,968,402.87	596,088.90	86,193.95	56,459.82	11,377.85
		290,058.87	592,680.00	84,099.95	56,728.82	11,447.85
Change from 2020 N	lo Project	(10,959.73)	(3,504.46)	(1,384.33)	(430.44)	(41.55)
(pounds/day)		(20,957.73)	(5,318.60)	(2,529.70)	(273.44)	19.88
SCAQMD Thresho (pounds/day)	ld for Operation	550	55	55	150	150

Source: LSA Associates, Inc., 2001.

<sup>&</sup>lt;sup>1</sup> Typographical correction.

The CAL3QHC model was used to assess the CO concentrations at intersections in the vicinity of the project sites. Tables 8.7-5 and 8.7-6 show that the 1-hour and 8-hour CO concentrations would be below the State and federal CO standards. Similar to the Proposed Project, no CO hot spots at JWA would occur from project related vehicular traffic trips under this alternative. Local CO hot spots, however would likely occur at the MCAS El Toro site similar to those under the Nonaviation Alternative.

# Toxic Air Contaminant Impacts

This alternative would avoid toxic air contaminant aviation impacts at the El Toro site but would increase impacts at the JWA site. Toxic air contaminant impacts would also result from development of the MCAS El Toro site as nonaviation.

# 8.7.4.6 Topography

Under Alternative G, the acreage of JWA would be expanded to meet a substantially increased volume of passenger traffic. In order to accomplish this, additional land surrounding the airport would be purchased. Expansion of the existing JWA runway would result in significant impacts to topography due to grading in order to extend the runway south by approximately 1,000 feet and north by approximately 1,300 feet. In addition, this alternative requires the addition of a runway to the JWA site, which would also result in significant impacts related to topography due to the grading necessary to create the addition. Based on these identified impacts, this alternative would result in a greater level of impacts related to topography than that identified under the Proposed Project. Therefore, this alternative would not avoid or substantially lessen impacts compared to the project.

# 8.7.4.7 Soils, Geology and Seismicity

This alternative would have no significant impacts at the El Toro site but would have significant adverse impacts at JWA.

Under Alternative G, the acreage of JWA would be expanded to meet the increased volume of passenger traffic. Expansion of the existing JWA runway would result in significant impacts related to soils, geology, and seismicity because of the necessity for runway extensions south by approximately 1,000 feet and north by approximately 1,300 feet. Another aspect of this alternative is the development of an additional parallel runway. The soils in the northern part of the JWA site are classified as part of the Omni soil association and are potentially highly expansive. The northern expansion area would be into a flood prone and high liquefaction area. Although it is anticipated that construction design would include mitigation measures, the impacts related to soils, geology, and seismicity would be greater under this alternative than those identified under the Proposed Project.

Table 8.7-5 Phase 4 Alternative G - Predicted One Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

		111101	accriona	44 1 7 2 2	rue ii	Rucar	AOIGIN	C ANG	** UI 31	TICACI			
inty	INTERSECTING STREETS	RECI	. REC1	RECY	RECE	RECS	RECE	NECT	RECF	RECY	RECIG	RECII	REC12
	CITY OF ORANGE												
45	Jamboree & Chapman	7.0	7.0	7.2	7.0	7.0	7.2	6.8	7,2	6.B	6.8	6.7	6.6
	CITY OF SANTA ANAII												
54	MacArthur & Main	7.0	7.1	7.1	7.2	6.8	6.8	6.6	7.0	6.9	7.0	6.7	6.8
52	Main & Sunflower	7.0	7.1	6.7	7.1	6.6	7.0	6.5	6.5	6.7	6.8	7.2	70
ю	Grand & Edinger	7.0	7.0	7.0	7.2	6,9	7.0	6.7	7.1	6.6	6.9	5.8	6.8
	CITY OF TUSTING												
3	Newport & Edinger	7.2	7.1	6.9	7.2	6.7	6.7	6.6	6.7	6.7	7.0	6.6	6.9
5	Tustin Ranch & Edinger	7.2	7.0	6.9	7.1	6.7	6.9	6.6	6.7	6.9	6.9	6.6	7.0
15	Von Karman & Barranca	7.0	7.1	7.1	6.9	6.6	7.1	6.7	6.9	6.6	6.5	6.8	7.0
	CITY OF IRVINE"												
16	Jamboree & Barranca	5.9	6.0	5.6	5.6	5.3	5.5	5.3	5.5	5.5	5.4	5.5	5.9
56	Jamboree & Main	5.6	5.5	5.7	5.4	5.2	5.4	5.3	5.4	5.3	5.3	5.4	5.5
34	Jamboree & Alton	5.6	5.6	5.7	5.7	5.2	5.5	5.2	5.3	5.4	5.4	5.4	5.7
8	Cuiver & Irvine Center	5.6	5.6	5.7	5.7	5.3	5.4	5.4	5.5	5.3	5.5	5.4	5.5
75	Jamboree & Michelson	5.6	5.4	5.4	5.6	5.3	5.3	5.3	5.3	5.3	5.4	5.4	5.4
51	Red Hill & MacArthur	5.7	5.6	5.5	5.6	5.1	5.4	5.3	5.3	5.3	5.3	5.3	5.7
00	Jeffrey & Irvine Center	5.6	5.5	5.5	5.6	5.3	5.5	5.2	5.2	5.4	5.7	5.1	5.3
53	Red Hill & Main	5.4	5.5	5.5	5.5	5.2	5.4	5.1	5.2	5.2	5.2	5.1	5.3
95	MacArthur & Jamboree	5.5	5.5	5.5	5.3	5.1	5,2	5.1	5.3	5,4	5.3	5.2	5.4
74	Von Karman & Michelson	5,4	5.4	5.3	5.2	5,0	5.1	5.2	5.1	5.1	5.2	4,9	5,3
77	Culver & Michelson	5.4	5.5	5.3	5.2	5.0	5.2	5.1	5.0	5.1	5.1	5,3	5.4
	CTTY OF LAGUNA BEACH												
299	Moulton & El Toro	5.4	5.4	5.5	5.5	5.1	5.3	5.4	5.4	5.0	5.3	5.1	5.1
	CITY OF LAGUNA HILLS												
280	El Toro & Avd. Carlota	5.4	5.4	5.4	5.3	5.0	5.3	5.0	5.1	5.2	5,2	5.1	5.3
Note:	* - Concentrations are in parts ;	er million (pr	om)										

<sup>1 -</sup> REC1 SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> REC5 S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> REC8 W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient one-hour CO concentration, 6.1 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

<sup>14 -</sup> The ambient one-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest one-hour CO concentration at the nearest air monitoring station, Saddleback Valley Air Monitoring Station between the years 1996 to 2000, is added to the calculated one hour levels.

Table 8.7-6

Phase 4 Alternative G – Predicted Eight Hour Ambient Carbon Monoxide Concentration for Intersections with the Highest Volume and Worst Level of Service (LOS)

INTE	MTERSECTING STREETS.	NKC1	REC?	RECT	III III CA	RICE	RLCS	REC7	8843	- RDC	BEC10	PECIF	REC12*
	CITY OF ORANGE®												
345	Jamborce & Chapman	5.2	5.2	5.4	5.2	5.2	5.4	5.1	5.4	5.1	5.1	5.0	5.0
	CITY OF SANTA ANAD												
154	MacArthur & Main	5.2	5.3	5.3	5.4	5.1	5.1	5.0	5.2	5,2	5.2	5.0	5.1
152	Main & Sunflower	5,2	5.3	5.0	5.3	5.0	5.2	4.9	4.9	5.0	5.1	5.4	5.2
90	Grand & Edinger	5.2	5.2	5.2	5.4	5.2	5.2	5.0	5.3	5.0	5.2	5.1	5.1
	CITY OF TUSTIN"												
93	Newport & Edinger	5.4	5.3	5.2	5.4	5.0	5.0	5.0	5.0	5.0	5.2	5.0	5.2
95	Tustin Ranch & Edinger	5,4	5.2	5.2	5.3	5.0	5.2	5.0	5.0	5.2	5.2	5.0	5.2
115	Von Karman & Barranca	5.2	5.3	5.3	5.2	5.0	5.3	5.0	5.2	5.0	4.9	5.1	5.2
	CITY OF IRVINE"												
116	Jamboree & Barranca	3.8	3.9	3.6	3.6	3.4	3.5	3.4	3.5	3.5	3.5	3.5	3.8
156	Jamborce & Main	3.6	3.5	3.7	3.5	3.3	3.5	3.4	3.5	3.4	3.4	3.5	3.5
134	Jamboree & Alton	3.6	3.6	3.7	3.7	3.3	3.5	3.3	3.4	3.5	3.5	3,5	3.7
92	Culver & Irvine Center	3.6	3.6	3.7	3.7	3.4	3.5	3.5	3.5	3.4	3,5	3.5	3.5
175	Jamboree & Michelson	3.6	3.5	3.5	3,6	3.3	3.4	3.4	3.4	3,4	3,5	3,5	3.5
151	Red Hill & MacArthur	3.7	3.6	3.5	3.6	3.3	3.5	3.4	3.4	3.4	3.4	3.4	3.7
100	Jeffrey & Irvine Center	3.6	3.5	3.5	3.6	3.4	3.5	3.3	3.3	3.5	3.7	3.3	3.4
153	Red Hill & Main	3.5	3.5	3.5	3.5	3.3	3.5	3.3	3.3	3.3	3.3	3.3	3.4
195	MacArthur & Jamboree	3.5	3.5	3.5	3.4	3.3	3.3	3.3	3.4	3,5	3.4	3.3	3.5
174	Von Karman & Michelson	3.5	3.5	3.4	3.3	3.2	3.3	3.3	3,3	3.3	3.3	3.1	3.4
177	Culver & Michelson	3.5	3.5	3.4	3.3	3.2	3.3	3.3	3.2	3.3	3.3	3.4	3.5
	CITY OF LAGUNA BEACH"												
299	Moulton & El Toro	3.5	3.5	3.5	3.5	3.3	3.4	3.5	3.5	3.2	3,4	3.3	3,3
	CITY OF LAGUNA HILLS"												
280	El Toro & Avd. Carlota	3.5	3.5	3.5	3.4	3.2	3.4	3.2	3.3	3.3	3.3	3.3	3.4

Note: \* - Concentrations are in parts per million (ppm)

<sup>1 -</sup> RECL SW CORNER

<sup>2 -</sup> REC2 SE CORNER

<sup>3 -</sup> REC3 NE CORNER

<sup>4 -</sup> REC4 NW CORNER

<sup>5 -</sup> RECS S. DEPARTURE - MID BLOCK

<sup>6 -</sup> REC6 N. APPROACH - MID BLOCK

<sup>7 -</sup> REC7 E. DEPARTURE - MID BLOCK

<sup>8 -</sup> RECS W. APPROACH - MID BLOCK

<sup>9 -</sup> REC9 N. DEPARTURE - MID BLOCK

<sup>10 -</sup> REC10 S. APPROACH - MID BLOCK

<sup>11 -</sup> REC11 W. DEPARTURE - MID BLOCK

<sup>12 -</sup> REC12 E. APPROACH - MID BLOCK

<sup>13 -</sup> The ambient eight-hour CO concentration, 4.6 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Central Orange County Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

<sup>14 -</sup> The ambient eight-hour CO concentration, 2.9 ppm, obtained by multiplying a rollback factor to the second highest eight-hour concentration at the nearest air monitoring station, Saddleback Vailey Air Monitoring Station between the years of 1996 to 2000, is added to the product of the calculated one-hour levels multiplied by a persistent factor of 0.7.

With regard to the MCAS El Toro site, the impacts related to soils, geology, and seismicity would be similar to those identified in Section 8.3 for the ETRPA Nonaviation Alternative. This alternative would not, therefore, avoid or substantially lessen impacts compared to the project.

# 8.7.4.8 Hydrology and Water Quality

This alternative would have significant adverse impacts at JWA. The hydrology and water quality impacts of Alternative G at JWA would be greater than the level of impacts under the Proposed Project due to the increased aviation activities. These impacts could be mitigated using proper engineering design and construction practices, similar to those described under the Proposed Project. With regard to the El Toro site, the impacts related to hydrology and water quality would be similar to those identified in Section 8.3 for the ETRPA Nonaviation Alternative. This alternative, therefore, would not avoid or substantially lessen impacts compared to the project.

# 8.7.4.9 Biological Resources

This alternative would have significant adverse impacts at JWA. At JWA there are substantial new facility improvements required to accommodate the increase in aviation operations. Compared to the Proposed Project, the physical improvements for Alternative G would impact additional non-native, ornamental vegetation but would not likely impact native plant communities. The most substantive change in the environment would be associated with the increase in aircraft activity and commensurate increases in noise exposure and overflights in Newport Back Bay. These increases do represent a substantial adverse change from the Proposed Project as well as from existing conditions. CNEL values in the Upper Newport Bay would range from in excess of 65 dB CNEL to over 70 dB CNEL. Depending upon the receptor location, this represents a CNEL increase of 5 to 10 dB. Alternative G ranges between 3 and 9 dB CNEL greater than for existing conditions, and between 4 and 10 dB CNEL greater than the Proposed Project conditions.

Due to the increase in noise exposure, adverse impacts to nesting behavior as a result of the CNEL increase are anticipated. Local listed species that reside in the Upper Newport Bay include California gnatcatcher, California least tern, Belding's savannah sparrow, California brown pelican, and clapper rail. However, the SEL impacts are not anticipated to change because of the similarity of the aircraft fleet mix under Alternative G and the Proposed Project. Therefore, no increase in the startle response is anticipated. Regardless, the increase in overflights, coupled with the higher average noise exposure could result in additional adverse impacts to biological productivity in the Upper Newport Bay. Regarding the El Toro site, this alternative would result in impacts similar to the ETRPA Nonaviation Alternative. In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

#### 8.7.4.10 Public Services and Utilities

This alternative would have significant adverse impacts at JWA. Alternative G would expand JWA beyond its current property limits and would substantially increase the MAP served, which would generate the need for increased fire and emergency medical, police services, and transit service in the area. Mitigation measures prescribed in Section 4.10 (Public Services and Utilities) would be applied, which would reduce the impacts of increased service needs. With demand for additional public services at both MCAS El Toro and JWA areas, Alternative G would have greater impacts to public service providers and facilities than the Proposed Project.

As described in Section 4.10 (Public Services and Utilities), the Proposed Project is not anticipated to result in significant adverse impacts related to utilities at the El Toro site or at JWA. It is anticipated that the utilities needs at El Toro under Alternative G would be similar to the anticipated needs under the ETRPA Nonaviation Alternative. Therefore, this alternative would not avoid or substantially lessen impacts compared to the project.

# 8.7.4.11 Natural Resources and Energy

As noted in Section 4.11 (Natural Resources and Energy), the Proposed Project would not result in significant adverse impacts to natural resources and energy, with the exception of impacts to agricultural resources at MCAS El Toro, which could not be mitigated to below a level of significance. This alternative could reduce or avoid the project impacts on agricultural soils depending on the reuse plan selected for the El Toro site. However, a nonaviation plan such as the ETRPA Nonaviation Alternative (Section 8.3) would have greater impacts than the project on agricultural soils. There are no natural or agricultural resources at JWA.

Implementation of the Proposed Project also would result in a less than significant increase in regional energy consumption, associated primarily with construction and operation of a new international airport at the MCAS El Toro site, as compared to existing conditions. Alternative G also would realize lower energy consumption levels associated with airport operations. This energy savings would be offset, however, by increased highway miles traveled, as the shortfall in forecasted demand for air travel service forces air travelers to drive to other airports within the region than would be required with the Proposed Project. In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

# 8.7.4.12 Aesthetics, Light and Glare

This alternative would have greater impacts at JWA compared to the Proposed Project. Because significant expansion of the JWA site would be required for Alternative G, substantial alteration of the existing visual setting would take place under this alternative, whereas the Proposed Project would not substantially modify the existing visual appearance

of JWA. Visual impacts of this alternative could be reduced to a level of insignificance with adequate mitigation; however, the impact of aircraft light and glare on nearby residential uses, mainly Santa Ana Heights, would be unavoidably significant.

Alternative G would require property acquisition of the developed area east of the airport between Campus Drive, MacArthur Boulevard, Bristol Street (SR-73), and Jamboree Road. In addition, property to the west of the airport beyond Red Hill Avenue would be needed for a new runway and aviation support uses. At the JWA site, Alternative G would have a significantly greater visual impact than that of the Proposed Project, which would not alter the existing visual effect on the surrounding land uses. Since the existing setting is one of intensive urban development, the expansion of facilities under Alternative G would not have a substantial adverse effect on a scenic vista, or damage scenic resources such as rock outcroppings, trees, historic buildings, or a scenic highway; such scenic resources are not present in the immediate JWA area. The effects of light and/or glare at the JWA site under this alternative would be more adverse than the existing setting and the Proposed Project. With the County Standard Condition of Approval LG 1 (Appendix L), the effects of the increased ground-level light and glare would be reduced to below the level of significance. However, the added nighttime aircraft light and glare impacts would be significant after mitigation. In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

#### 8.7.4.13 Cultural Resources

This alternative would have impacts at JWA similar to the Proposed Project. The records search referenced in Section 4.13, Cultural Resources, included the expansion area between MacArthur Boulevard, Campus Drive, and Bristol Street that would be required for development of Alternative G at JWA. The majority of the expansion area southeast of JWA has not been surveyed for historic or prehistoric archaeological resources because the area is intensely developed with office and commercial buildings. The surveys that have been conducted were at locations at the periphery of the expansion area. Similar to Proposed Project conditions, no prehistoric or historic archaeological sites or properties of historic significance were found in the few surveys at the periphery of the area.

In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

#### 8.7.4.14 Recreation

This alternative would have greater impacts at JWA than those under the Proposed Project. Alternative G would expand the physical area of JWA and interrupt trail use and/or encroach into the golf courses south of JWA. On-street bikeways would be impacted by enlargement of JWA and required realignment of area streets such as Redhill Avenue. The extent of encroachment or interruption of use would be determined with a more precise level of planning for Alternative G. In addition, noise impacts on recreational use of trails and parks

in the area would increase from existing conditions and the Proposed Project. Significant noise related recreational impacts would occur under Alternative G since the 65 dB CNEL contour would be expanded to areas beyond the existing 1998 65 dB CNEL contour.

In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

# 8.7.4.15 Public Health and Safety

Compared to existing conditions, the overall potential for accidents under this alternative is less than the Proposed Project. This alternative would avoid aviation accident impacts at the El Toro site and lessen the aviation accident potential at JWA due to a reduction in GA operations.

# Aviation Safety

Compared to the Proposed Project, there would be an increase of approximately 234,700 air carrier and air cargo operations and a decrease of approximately 313,700 general aviation operations at JWA under this alternative. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 347.7% to reflect the number of increasing aviation activity and the potential accident risks for general aviation at JWA would decrease by 87.4% correspondingly. Since there is no aviation activity at OCX, under this alternative, there would be zero aviation risks. Compared to the Proposed Project, relative to on-airport and off-airport fatal accidents per million operations, there would be no significant adverse impacts related to aviation safety at the MCAS El Toro site or at JWA.

Compared to the existing conditions, there would be an increase of approximately 212,192 air carrier and air cargo operations and a decrease of approximately 282,324 general aviation operations at JWA. Under this scenario, the potential air carrier and air cargo accident risks at JWA would increase by approximately 235.7% to reflect the number of increasing operations and the potential general aviation accident risks would decrease by 86.2% correspondingly. Compared to the existing conditions, there would be no significant adverse impacts related to aviation safety at JWA.

#### 8.7.4.16 Hazardous Materials and Hazardous Wastes

Under this alternative, expansion of runway facilities at JWA would have a moderate potential to result in excavation of contaminated soils. Although it is not presently known if subsurface contamination exists within runway expansion areas, there is a possibility that petroleum hydrocarbon contamination may be encountered from leaking underground storage tank sites in the vicinity. However, construction activities would be required by state and federal law to ensure that any hazardous waste contamination encountered during construction is reported and handled to the satisfaction of the appropriate local agencies. Therefore, with the application of existing laws governing hazardous waste remediation, the

impacts of Alternative G related to soil and groundwater contamination would be anticipated to be less than significant. The impacts of Alternative G related to hazardous materials and hazardous wastes at JWA, however, would be greater than under the Proposed Project.

Any use of hazardous materials and/or generation of hazardous waste under Alternative G would be regulated by applicable State law, federal law, and regulations pertaining to worker protection, hazardous materials storage and use, and hazardous waste generation and disposal. Implementation of these regulations would reduce potential impacts associated with the presence of these hazardous substances to below a level of significance. In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

#### 8.7.4.17 Socioeconomics

Under this alternative, approximately 700 acres of surrounding land would have to be acquired to accommodate the projected aviation activities. Much of this land is currently developed with commercial/industrial uses, which would have to be displaced and relocated. Therefore, under the threshold of significance related to displacement of a large number of persons, this alternative would have a significant adverse impact that is not anticipated under the Proposed Project. No existing housing units would be displaced under this alternative.

Under this alternative, an estimated 17,500 jobs would be provided at JWA in 2020, representing a net increase of 15,400 jobs over existing 1998 conditions. The total number of jobs generated under this alternative would be substantially lower than under the Proposed Project. However, the number of jobs at JWA under Alternative G substantially exceeds the jobs projected at JWA under the Proposed Project. Under this alternative, it is assumed that the MCAS El Toro site would develop with a range of nonaviation uses similar to those shown in the ETRPA Nonaviation Plan. This development would result in an estimated 50,700 jobs, 13,600 persons, and 5,900 housing units being located on the site in 2020. This level of activity is significantly higher than the level anticipated under the Proposed Project. In total, this alternative supports 68,200 jobs, 13,600 persons, and 5,900 housing units on the project site. This figure is significantly higher than the number of jobs, persons, and housing units expected under the Proposed Project.

As with the Proposed Project, economic activity occurring at the JWA and El Toro sites, as well as expenditures by visitors arriving by air through JWA, would stimulate additional offsite job growth. Given the higher number of on-site jobs and fewer air passengers served by this alternative, the number of off-site jobs under Alternative G would be similar to the Proposed Project.

Given the greater number of jobs generated under Alternative G, at 68,200 jobs versus 29,500 jobs under the Proposed Project, the magnitude of impacts related to induced growth or concentration of population and employment, and increasing demand for housing, including low and moderate income housing, would be greater than under the Proposed

Project. The additional demand for housing created by project related employment would be partially, but not completely, offset by the housing to be developed on the El Toro site under this alternative.

In summary, this alternative would not avoid or substantially lessen impacts compared to the project.

# 8.7.4.18 Risk of Upset

The ultimate build out and phased development of this alternative would entail no aviation reuse at the El Toro site and a market-driven increase in operations at JWA. An increase in operations under this alternative would entail an approximate three-fold increase in ultimate commercial passenger service levels at JWA over the existing cap. This growth in passenger service would also generate a substantial increase in demand for jet fuel at JWA, as well as associated diesel-fueled tank truck jet fuel transport operations since JWA (unlike OCX) is not served by pipelines. Additional requirements for fuel storage capacity on the JWA also could be required. Under this circumstance, the risk of upset potential at JWA would be higher than that under the Proposed Project.

As discussed in Section 4.18 (Risk of Upset), the Proposed Project would not result in significant adverse impacts to public health and safety resulting from project related risk of upset conditions. Therefore, this alternative would not avoid or substantially lessen impacts at JWA, but would avoid avoid aviation impacts at the El Toro site.

# 8.7.5 Feasibility

This alternative is feasible from a physical standpoint in that the existing main runway can be extended to serve an expanded market role (limited international). However, significant acquisition of developed property adjacent to the airport is required in order to provide space for additional airfield development and required terminal facilities.

From an operational standpoint, the primary runway for JWA is capable of supporting a limited international market role. However, it is not feasible for the runway in this alternative to serve operations by all commercial aircraft types. The runway length would not be capable of supporting full international service. This alternative provides a limited amount of space for general aviation. As such, general aviation operations can be served on a very limited basis, and the more than 500 JWA based aircraft must be relocated to other general aviation airports in the region.

From a market perspective, this alternative provides for substantial growth in passenger and cargo service beyond today's levels. It does not however meet all of the project market objectives.

From a fiscal perspective, the order of magnitude cost for Alternative G is estimated at \$4.3 billion as described in ASMP Technical Report 6. These are identified as "order of magnitude" costs since they have been prepared without the benefit of a master plan (\$4.3 billion is 54 percent higher than the Proposed Project's cost).

#### 8.7.6 Conclusions

- This alternative would increase the adverse effects of noise compared to the Proposed Project, No Project, and all other alternatives. It is the only one of the alternatives evaluated with existing residences (446 homes) inside the highest impact 70 CNEL noise contour. It has 6,954 residences inside the 60 CNEL contour, 4,540 more than the Proposed Project.
- The feasibility of this alternative is questionable from a financial standpoint. The "order of magnitude" capital cost estimate is \$4.3 billion, 54 percent higher than the Proposed Project. Compared to the Proposed Project, the reserve potential is much less due to 27 percent fewer passengers and fewer revenue generating airport compatible uses.
- This alternative fails to meet a major objective of satisfying, a substantial portion of Orange County's general aviation demand.
- The alternative does not fulfill the LRA's objective of implementing a two commercial airport system.
- The alternative does not take advantage of existing land use restrictions within the Policy Implementation Line (PIL).
- The alternative does not meet as much of the Orange County commercial aviation demand as the Proposed Project alternative. The runway length would not be capable of supporting full international service. Therefore, the alternative does not meet all of the project market objectives.
- This alternative would result in significant local and regional air quality impacts and air quality impacts related to construction greater than the Proposed Project. Toxic air contaminant health risk impacts would also be significant under this alternative.

# 8.8 ALTERNATIVE J: JWA – STATUS QUO AVIATION ROLES; OCX FULL INTERNATIONAL WITH WIDELY-SPACED RUNWAYS

This section presents the potential impacts of Alternative J as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020 build out, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

This alternative was selected for analysis because it has the potential to avoid the impacts of easterly departures (and the potential for westerly departures) while still obtaining most of the objectives of the project.

#### 8.8.1 Aviation Uses

The airport roles and expected aviation activity levels for Alternative J would be the same as for the Proposed Project. Under this alternative, however, MCAS El Toro would be developed with two parallel north-south runways, with a centerline separation of 3,000 feet. This would provide greater separation of the arrival and departure streams of aircraft to increase the aircraft arrival rate under instrument weather conditions. It would also create a large "infield" area between the runways for the development of terminal or other aviation related facilities. Figure 8-10 depicts Alternative J.

# 8.8.2 Nonaviation Revenue Support Uses

The nonaviation land uses proposed under Alternative J are the same as assumed for the Proposed Project.

# 8.8.3 Attainment of Project Objectives

This alternative meets the general project objectives for reuse of the base except special planning opportunities and minimize environmental impacts. Alternative J also meets the aviation related objectives, with the exception of existing land use restrictions. However, the greater separation of the runways will subject large areas of existing and planned noise sensitive uses to aviation noise impacts exceeding 65 CNEL. For this reason, this Draft EIR proposes to reject this alternative.

# 8.8.4 Environmental Impacts of Alternative J

The airport role and MAP levels would be the same as with the Proposed Project, therefore most of the impacts would be identical or similar to those addressed for the Proposed Project, except that aviation noise impacts would be much more adverse than with the Proposed Project. Because this alternative would have the identical or similar impacts compared to the Proposed Project, the following analysis focuses on the topical areas where there are measurable differences between the alternative and the Proposed Project.

#### 8.8.4.1 Noise

Compared to existing conditions, this alternative would have the same impacts at JWA as the project; however, the impacts surrounding the El Toro site would be significantly worse than the Proposed Project. This alternative would not, therefore, avoid or substantially lessen the project impacts.

The Alternative J 65 CNEL contour line would include 13.97 square miles of land for OCX and 1.49 square miles of land for JWA. The 65 CNEL for the existing military aircraft operations at MCAS El Toro include 8.0 square miles of land and, for JWA, the existing conditions include 1.49 square miles of land. Therefore, Alternative J would increase the area affected by the 65 CNEL surrounding the El Toro site by 5.97 miles compared to an increase of 5.7 square miles for the Proposed Project. At JWA, Alternative J would impact the same area affected by the 65 CNEL.

The Proposed Project would increase noise sensitive land uses by three churches and one private school compared to existing conditions at the El Toro site (see Table 8.2-3). This alternative would result in 525 residences in the vicinity of El Toro located in areas subject to aviation noise levels of 65 to 70 CNEL compared to zero residences in the vicinity of El Toro impacted by the 65 to 70 CNEL for the Proposed Project and existing conditions (see Table 8.2-4). This represents a significant adverse impact that could not be reduced through mitigation measures. This is due to the proposed addition of the westerly runway complex, which would result in aircraft approaches over existing homes south of the El Toro site. Departures from this new westerly runway complex would fly close to existing homes in the Northwood Pointe area of Irvine and over planned residences in north and northeast Irvine. Figure 8-11 illustrates the year 2020 dB CNEL contours for Alternative J.

As discussed earlier, the CNEL calculation factors in the number of daily operations and assigns a "penalty weighting" to operations occurring during the nighttime hours (10 p.m. to 7 a.m.). However, the substantial increase in the number of operations, particularly during nighttime hours, may be considered a significant impact of Alternative J independent of the CNEL computation for the same reasons identified for the Proposed Project and other relevant alternatives.

#### 8.8.4.2 Biological Resources

Compared to existing conditions, this alternative would have no significant impact at JWA, but would have significant adverse impacts at the El Toro site similar to the Proposed Project except as noted below. The mitigation measures proposed for the Proposed Project would reduce any impacts of this alternative to a level of insignificance.

The elimination of the east-west runway would reduce aircraft noise exposure in the federal Habitat Reserve. For Alternative J, the 60, 65 and 70 CNEL contours do not overlay any of the Habitat Reserve. This would be a beneficial impact of Alternative J as the noise exposure from aircraft overflights is substantially reduced. However, the habitat in the preserve has included relatively high densities of gnatcatcher occupation. Even during the military aircraft utilization of the east-west runway, and the corresponding noise exposure in the preserve, there were relatively high densities of California gnatcatcher. Therefore, the reduction in noise exposure within the preserve boundary would not result in a corresponding increase in gnatcatcher density. The quality and extent of the habitat in the preserve contributes to the density of occupation, rather than indirect influences resulting from noise exposure.

The north-south runway, being separated by 3,000 feet, results in an increase in the width of the CNEL contours at Siphon Ridge to the north. The noise is essentially spread out over a wider geographic area, although the closure point of the CNEL contours (e.g. 65, 70) is not substantially changed from the Proposed Project. This condition is consistent for both the 70 CNEL contour as well as for the 65 CNEL contour. The closure point for the 65 CNEL contour extends further to the north well beyond the south facing slopes of Loma Ridge, into the Santiago Hills. The most significant biological resource at Siphon Ridge is coastal sage scrub habitat and the California gnatcatcher. However, since the proposed federal habitat area has one of the densest occupations by California gnatcatchers in Orange County in an area subject to decades of very high military aircraft noise, there appears to be no correlation between aircraft noise and adverse habitat impacts. Therefore, Alternative J would not have a significant adverse impact on the Siphon Ridge coastal sage scrub habitat area.

# 8.8.4.3 Public Health and Safety

Compared to the Proposed Project, the overall aviation activity levels and aircraft accident risks at JWA and the MCAS El Toro site would remain the same because the level of aircraft operations would be the same. The change in runway configuration would not significantly affect aviation accident risk.

# 8.8.5 Feasibility

This alternative requires significant land acquisition for runway construction at OCX and requires major modifications to SR133, the Eastern Transportation Corridor (including the construction of a bridge to allow a new parallel runway and associated taxiway to pass over SR133), and potentially modifications to Irvine Boulevard.

From an operational standpoint, the runway configuration at OCX does not allow departures to the east. This increases airspace interactions with JWA and other traffic to the north, and does not take full advantage of the existing Policy Implementation Line (PIL). This alternative would create new noise impact areas north and south of OCX outside the PIL.

From a market perspective, this alternative provides for substantial growth in passenger and cargo service beyond today's levels, and also accommodates Orange County's general aviation demand.

#### 8.8.6 Conclusions

Conclusions regarding Alternative J are as follows:

- This alternative increases airspace intersections by directing more air traffic towards the JWA primary approach corridor, towards a VFR flyway between Corona and Los Alamitos (V-8-21), and towards airspace used by Ontario Airport departing aircraft. It fails to take advantage of less active airspace to the east of El Toro.
- This alternative requires significant land acquisition and requires significant major Modifications to SR 133, the Eastern Transportation Corridor, and potential modifications to Irvine Boulevard.
- This alternative does not take full advantage of existing land use restrictions inside the Policy Implementation Line (PIL) and creates new noise impacts areas north and south of OCX. It causes 5425 residences to be within the OCX 65 CNEL contour, and 3,411 residences to be within the OCX 60 CNEL contour.

# 8.9 ALTERNATIVE OCX AIRPORT RUNWAY LAYOUT (WILDLANDS RANCH ALTERNATIVE)

This section presents the potential impacts of the Wildlands Ranch Alternative as measured against the existing setting, as well as a comparison of the alternative's impacts to those of the Proposed Project at build out. In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing of the project, i.e., in those instances in which the impacts of the Proposed Project during the phasing years are materially different from those impacts at year 2020 build out, a comparison of the alternative's impacts to those of the Proposed Project for the applicable phasing year is also provided.

In April 1998, a proposed El Toro runway layout consisting of a "V" configuration was submitted by Mr. Charles E. Griffin to the Orange County Register. Throughout 1998, regular updates of the concept were distributed by Mr. Griffin, including submittals to the Orange County Board of Supervisors. On November 24, 1998, and December 1, 1998, Mr. Griffin submitted responses to the Notice of Preparation of Draft Environmental Impact Report No. 573. These included conceptual sketches of an alternative airport runway layout. On April 7, 1999, Mr. Griffin submitted his then latest iteration of the proposed alternative to the County of Orange. A review of this submittal was conducted as part of this EIR. Subsequently, a later concept was submitted to the County on October 5, 1999. This submittal contained significant modifications to the April 7 concept. This concept was also reviewed. The description of the alternative presented herein is based on the information contained in both the April 7 and October 5, 1999, submittals. Each submittal is addressed separately in this EIR subsection.

# 8.9.1 Aviation Uses

The following is a chronology of materials submitted by Mr. Griffin that have been received by the County of Orange El Toro Master Development Program and their aviation planning consultants.

- (i) April 26, 1998 Submitted to the Orange County Register. The submittal offers a concept of reorienting runways so flight paths are over non-residential areas. The concept is based on a 3 degree glide slope to Runway 16.
- (ii) May 20, 1998 Submitted to the Orange County Register. The concept is similar to the April 26 submittal, except the concept is based on a glide slope greater than 3 degrees.
- (iii) July 21, 1998 Submitted to the Orange County Board of Supervisors. The submittal provides further documentation of the proposed concept, including additional documentation of TERPS issues. The concept represented in this submittal is based on a 3 degree glide slope to Runway 16 and Runway 2.

- (iv) July 28, 1998 Submitted to the Orange County Board of Supervisors. The submittal expresses an opinion on an issue that the separation of parallel runways should be at least 2,500 feet and preferably 5,000 feet for simultaneous IFR operations, per FAA Advisory Circular 150/5300-13.
- (v) August 11, 1998 Submitted to the Orange County Board of Supervisors. The submittal expresses the opinion that FAA should not approve an Airport Layout Plan that includes parallel runways with centerline spacings of less than 2,500 feet and therefore will also not approve funding of runway improvements proposed by alternatives considered by Orange County.
- (vi) September 1, 1998 Submitted to the Orange County Board of Supervisors. The submittal was in response to proposed modifications of Airport Community Concepts B and C. The submittal contains a conceptual runway layout entitled "Alternative Airport and Open Space Plan Year 2020 Concept C As Recommended by the FAA in Advisory Circular AC 150/5300-13". The runway layout depicts extension of the north-south runway on the south end to Bake Parkway.
- (vii) November 24, 1998 Comments submitted on the Notice of Preparation of Draft EIR No. 573.
- (viii) December 1, 1998 Comments submitted on the Notice of Preparation of Draft EIR No. 573. The submittal proposes a standard approach angle of 3 degrees to Runway 16.
- (ix) April 7, 1999 Submitted to the El Toro Master Development Program in response to the Notice of Availability of Final EIR No. 563 Draft Supplemental Analysis. The submittal contains a concept plan entitled, "The Airport and Wildlands Ranch Plan Year 2020 Concept V". The concept plan is dated 1/23/99. The concept is based on an approach angle of 3.1 degrees to Runways 16 and Runway 01.
- (x) May 7, 1999 Submitted to the El Toro Master Development Program Office to offer comments on the "Green Airport Plan" dated April 29, 1999 and addresses water quality issues related to San Diego Creek and Serrano Creek. The submittal indicates an approach angle of 3.1 degrees is possible to Runway 16.
- (xi) October 5, 1999 A document entitled "The Alternative Airport Runway Layout Long V and Short V FAA TERPS Analysis Feasibility Study". The submittal includes an alternative layout in which the north-south runway is maintained north of the AT&SF Metrolink railroad tracks. A 3.3 degree glide slope is indicated for approaches to Runway 16. The submittal also includes the original runway concept in which the north-south runway is extended to Bake Parkway with approaches to Runway 16 using a 3.1 degree glide slope.

#### 8.9.1.1 General Features of the Wildlands Ranch Alternative

The description of the concept provided by the proponent provides general information related to the proposed runway configuration. The alternative proposed did not address all aspects of the project, so assumptions were made with respect to the following information:

- (i) Information describing the role and the design demand level of OCX.
- (ii) Information regarding the role and design demand level of JWA.
- (iii) Information regarding the extent and specific location of airfield, terminal, landside and access facilities.

The "V" runway configuration utilizes the existing Runway 16L-34R and a new Runway 1-19. In the April 7 submittal, the south end of the existing Runway 16L-34R is proposed to be extended 7,000 feet. Since a total runway length of 18,000 feet is proposed and the existing runway length is 10,000 feet, it is inferred that the north end of the runway is extended 1,000 feet. A new 12,000 foot Runway 1-19 is proposed, and is generally aligned with the SR-133 Freeway. In the October 5, 1999 submittal, Runway 16L-34R is not extended as far to the south. The Runway 34R threshold is located within the "Measure A" boundary and does not extend into the "south panhandle" of the base property.

The rationale for the alternative runway configuration is to redirect air traffic patterns over vacant land southwest of MCAS El Toro, to permit the use of Runway 16 for arrival operations, and to eliminate the use of the existing east-west runways.

The first phase of this alternative would operate with no changes to the current Runways 16L and 16R. Runway 16L would initially be used for landing from the north, and 16R would be used for takeoffs to the south. Runway 34L would be used during Santa Ana wind conditions for VFR circle approach from the southwest, and 34R would be used for takeoffs to the north. Runways 7R/25L and 7L/25R are used only as a concourse for gateways to temporary terminals for initial operations, until the new terminal is constructed. The new Runway 1/19 would be constructed to allow simultaneous operations as FAA budget and grant funds are allocated. The new runway would be aligned with SR 133, and a three mile wide corridor of undeveloped land (wildlands) would become public land south of the Irvine Spectrum complex and extending to Crystal Cove State Park. This alternative also calls for Runway 16L/34R to be extended across the railroad tracks to Bake Parkway as FAA funds become available.

The proposed initial phasing of this alternative would require an approach glide slope of 3.3 degrees, which would preclude precision instrument approaches by all aircraft with an approach speed of 141 knots or more.

# 8.9.1.2 Attainment of Project Objectives

This alternative limits operations by certain aircraft types that could use MCAS El Toro. Aircraft with approach speeds of 141 knots or more (termed as Approach Category D) would not be provided a precision instrument approach to the primary arrival runway and, therefore, would not be accommodated by this alternative. The limitation on Approach Category D aircraft does not encourage the growth of air service opportunities such as international, domestic long haul, and cargo. As such, while this alternative allows a greater portion of locally generated air traffic demand to be served, it is not to the same degree as the Proposed Project. Certain business jet models are also affected by the Category D limitation and, therefore, general aviation opportunities are not fully enhanced. This is described in more detail later in this section in the evaluation of the short "V" alternative.

This alternative does not meet the objective of taking maximum advantage of the historical compatible land use regulation around MCAS El Toro, because the runway configuration, and hence noise contours, are substantially different from the military operation. While the alternative is intended to direct flights over vacant areas, these areas are planned for 3,000 to 5,000 homes and other noise sensitive uses in the City of Irvine and County General Plan.

# 8.9.1.3 OCX Facility Improvements for the Wildlands Ranch Alternative (April 7, 1999, Submittal)

Figure 8.12 summarizes the basic features of this alternative for OCX described below.

#### Airfield

Runway 16 is proposed as the primary arrival runway and the landing threshold of Runway 16 is assumed to be displaced approximately 7,100 feet, based on interpretation of a drawing dated January 23, 1999, and included in the proponent's submittal of April 7, 1999. The landing threshold of Runway 34 is also shown to be displaced an equivalent distance. Based on the configuration of the Inner Safety Zone depicted by the proponent's submittal, approximately 8,300 feet of runway length is available for takeoff on Runway 16. This is considerably less than the existing runway length of 10,000 feet and runway lengths included in other alternatives (i.e., more than 12,000 feet). It appears that the right-turn after takeoff proposed in this alternative would start too soon, and protection zones should actually be slightly farther south.

A new runway oriented in a northeast-southwest direction, and designated as Runway 1-19 is proposed along the northwest side of the base boundary, generally aligned parallel with SR 133. It is planned to be 12,000 feet long. The landing threshold of Runway 1 is shown to be displaced by approximately 1,100 feet.

It is proposed to use Runways 16 and 1 for landings, and Runways 19, 16 and 34 for takeoffs under this alternative. The submittal specifies a 3.1 degree glide slope for the

proposed precision instrument approaches for Runways 16 and 1. This is considered non-standard as it exceeds a standard glide slope of 3 degrees. Almost all commercial airports in the United States are equipped for precision instrument approaches with standard 3 degree glide slopes.

The runway centerline to parallel taxiway centerline separation scales to a dimension of 400 feet.

#### **Evaluation of Proposed Airfield**

Several aspects of the proposed airfield are discussed here. These include approach slope, capacity, runway length available for landing, runway markings, airfield geometrics, taxiway system and construction issues.

#### Approach Slope

The Alternative provides a non-standard approach angle of 3.1 degrees. Very few major commercial airports have non-standard glide slopes, and the other airfield alternatives considered in this EIR are based on a standard approach angle of 3 degrees.

#### Capacity

The capacity of this alternative is limited due to the fact that only one runway is available for arrivals. Substantial delays will result during peak arrival periods at 2020 demand levels. The Proposed Project includes two runways that can be used for arrivals during prevailing weather conditions. While one runway can accommodate off-peak arrival periods, the second runway provides the ability to reduce delays during peak arrival periods.

Data contained in FAA guidelines (FAA Advisory Circular 150/5060-5, Airport Capacity and Delay) indicate that the Proposed Project provides 25 to 30 percent more runway capacity than the "V" configuration. Therefore, the "V" runway configuration will not accommodate traffic as efficiently as the Proposed Project and will be subject to escalating delays during peak periods.

#### Runways

Analysis of precision instrument approaches from the north in accordance with TERPS was conducted by the aviation planning team. The TERPS analysis concluded that the required location of the landing threshold for a 3.1 degree glide slope is approximately 8,200 feet south of the end of the existing Runway 16L. Therefore, the landing threshold for Runway 16 needs to be located approximately 2,000 feet further south than depicted under this alternative. The maximum amount of runway extension to the south (towards Bake Parkway) that is usable for landings on Runway 16 is approximately 6,200 feet. Such an extension would provide a landing distance of approximately 8,050 feet. This landing distance is significantly less than landing distances provided by other alternatives

(approximately 9,900 to 10,400 feet). This is a marginal runway requirement to be used for the planning of the primary arrival runway at a new commercial airport with operations and role as forecast for the Proposed Project.

While the landing distances provided by other alternatives are greater than this alternative, it is also noted that other alternatives are based on a standard, 3-degree glide slope. The runway configuration in this alternative is operationally inferior to the other alternatives due to the combination of steeper approach angle and reduced landing length. Compared to other alternatives, the available landing length is unacceptable.

#### **Runway Markings**

The placement of Runway Protection Zones under this alternative (Figure 8.12) suggests that Runway 34 will be retained for precision instrument approaches. Therefore, it will be necessary to provide precision runway markings for both Runways 16 and 34. This will require relocating the landing threshold of Runway 34 approximately 4,200 feet to the south to accommodate runway markings. This may impact precision instrument approaches from the south. It would also displace arrival SENEL contours by an equivalent amount to the south, increasing single event noise levels in some existing residential areas.

#### Airfield Geometrics

The runway centerline to taxiway centerline separation of 400 feet is not adequate to accommodate future New Large Airplane (NLA) models that may operate at the airport. A separation distance of 600 feet is required. The airfield geometrics of the Proposed Project are such that NLA operations can be accommodated.

#### **Taxiway Operations**

The separation of the runway centerline to the face of the terminal concourse that is parallel to Runway 16-34 is approximately 1,000 feet as shown in Figure 8.12. This precludes development of dual parallel taxiways along the terminal, which suggests a high potential for bottlenecks and aircraft delays on the ground due to pushbacks and maneuvering to and from parking positions. Parallel taxiways for Aircraft Design Group V are possible under this alternative but would limit the size of aircraft at concourse parking positions to aircraft with fuselage lengths of approximately 125 feet, rendering the gates unusable by the majority of the commercial aircraft fleet.

The extension of Runway 16-34 to the south as proposed in this alternative results in long taxi distances for aircraft arriving on the primary arrival runway, on the order of two miles.

#### Construction Issues

This airfield concept requires considerable fill on the south end. The construction of runways under this alternative requires that several roads and ground access infrastructure be tunneled under runways. These include:

- i) Main airport entrance road
- ii) Airport perimeter road
- iii) Alton Parkway
- iv) AT&SFMetrolink railroad tracks and Borrego Canyon Wash
- v) Irvine Boulevard

The south end of the airfield will require significant fill. It is through this area of fill that Alton Parkway, the airport perimeter road and the railroad would be tunneled. The south end of the extension of Runway 16-34 is located in the City of Irvine, and pursuant to the State Aeronautics Act, City approval may be required. However, City approval of any airport facility would be inconsistent with the City's position on the Reuse Plan (see City Council Resolution 99-01 and Measure D, November, 1998).

# Terminal and RON Parking

A linear terminal complex is proposed between the "V" shaped runways. Details on the terminal have not been documented by the proponent, however, based on the concept drawing the terminal provides approximately 9,000 linear feet of ramp frontage to accommodate aircraft contact gates at the terminal. This is considerably less ramp frontage and fewer gates than that provided by the terminal concept under the Proposed Project which provides over 11,500 feet of ramp frontage.

The face of the concourse that is parallel to Runway 16-34 is approximately 1,000 feet from the runway centerline. As previously indicated, this prevents the development of dual parallel taxiway capability which is important for efficient airfield operations. The terminal setback from the runway also affects the size of aircraft in terms of tail heights that can park at the terminal (in accordance with obstruction criteria specified in Federal Aviation Regulations Part 77, Objects Affecting Navigable Airspace). Aircraft parking areas need to be sufficiently separated from the runway so that parked aircraft are not obstructions as defined in FAR Part 77. If an aircraft parking limit line is assumed at a distance of 300 feet from the face of the terminal concourse in this alternative, the maximum tail height permitted at this line is 28.6 feet above the elevation of the nearest point of the runway centerline. Tail heights of the MD-11 and B747 exceed 57 and 64 feet, respectively, and would be precluded from parking at the concourse, as well as other aircraft that would be obstructions under FAR Part 77.

Remain Overnight Parking (RON) is not indicated on the concept drawing included in the proponent's submittal of April 7, 1999 (Figure 8-12).

### Access and Parking

#### **Terminal Access**

Primary access to the terminal complex would be provided at Trabuco Road and Barranca/Alton Parkways near the Irvine Transportation Center (ITC). Trabuco Road would cross under the proposed Runway 1-19 and connect directly to the terminal loop road. A new access road from Barranca/Alton Parkways would cross over the railroad tracks and connect to the terminal loop road north of the proposed Airport Transportation Center (ATC). Secondary terminal access would be provided by Marine Way which would be realigned at the south end of Runway 1-19.

It is assumed that the terminal loop road is at-grade (as there is no indication or provision for ramps in the proponent's submittal). Vehicles would circulate counterclockwise along the terminal loop road.

#### Assessment

The connection of Trabuco Road to the terminal loop road poses some technical and safety concerns.

The technical aspect pertains to the need to provide adequate distance from edge of the runway area for vertical gradient. A gradient of six percent, the standard used in other alternatives, could be accommodated over a distance of approximately 330 feet. The Alternative does not provide such distance. Thus, a steeper, substandard gradient would have to be used.

The safety aspect pertains to connecting the depressed segment of Trabuco Road directly to the terminal loop road immediately at the end of the tunnel. This intersection would have to be signalized. Due to the depression, drivers could not readily see the oncoming intersection, making it prone to accidents. Warning signals would need to be installed in the tunnel to alert drivers of the signal ahead. Furthermore, vehicles would be queued in the tunnel on Trabuco Road due to signalization. This could result in unsafe conditions since drivers of vehicles entering the tunnel could not immediately see the end of the queue. Vehicle queuing within the tunnel is also expected to result in an unhealthful concentration of vehicle fumes.

Angular bends on the terminal loop road, particularly the ones in front of the north and south terminal buildings, could result in traffic bottlenecks due to vehicles slowing down at bends. A single level terminal road suggests the potential for congestion due to the mixing of arrival and departure traffic which is separated by multi-level terminal roads in the Proposed Project.

#### **Parking**

This alternative provides three parking areas on a total of approximately 125 acres. All parking areas are inside the terminal loop road. It is assumed that these areas would accommodate short term, long term and employee parking as well as rental car facilities.

#### Assessment

The total parking area provided in this alternative is roughly equal to the on-site short term, long term, employee and rental car areas provided in the Proposed Project. If this alternative is to accommodate the same air passenger level as forecast for the Proposed Project, additional remote areas would have to be provided. In the Proposed Project, remote parking areas are conveniently located in the Northern Panhandle. This alternative does not have such area that could be easily accessed from the terminal. The parking area shown on the east-side of Runway 16-34 is not ideally suited for remote public parking.

Furthermore, since all parking areas in this alternative are inside the terminal loop road, employee and rental car traffic would have to mix with terminal (short term, long term and curbside) traffic. This would substantially add to the volume of traffic using the terminal loop road and, together with the effect of roadway bends discussed above, could lead to severe traffic congestion on the terminal loop road.

#### Non-Terminal Roadway Access

Primary access to the air cargo area would be provided by Alton Parkway. Although there are existing I-5 interchanges at Alton Parkway and Bake Parkway, their location would force air cargo truck traffic to use local streets.

The tunnel section of Irvine Boulevard under Runway 1-19 raises similar safety and air quality concerns as discussed for Trabuco Road. Westbound traffic on the Irvine Boulevard tunnel would immediately encounter the signalized intersection at the SR 133 northbound on/off ramps on exiting the tunnel. Also, traffic would be queued under the tunnel on the westbound approach to the intersection, which could result in unsafe driving and air quality conditions.

# 8.9.2 OCX Facility Improvements for the Wildlands Ranch Alternative (October 5, 1999, Submittal)

Figure 8-12A summarizes the basic features of this alternative (Wildlands Ranch Plan Alternative 1) for OCX described below. Figure 8-12B (Wildlands Ranch Plan Alternative 2) presents a variation of the alternative.

#### 8.9.2.1 Airfield

The major difference between this concept and that reflected in the April 7 submittal is that the north-south runway (16-34) is not extended as far south in order to locate runways entirely within the "Measure A" boundary. This results in a potential runway length of 12,000 feet which is achieved by constructing 1,000 foot extensions on each end of the existing Runway 16L-34R. Runway 16 would be the primary arrival runway. The runway would be equipped for precision instrument approaches with a specified glide path angle of 3.3 degrees. The proponent states that 8,000 feet of runway would be available for landing which suggests the landing threshold is displaced 4,000 feet.

The primary departure runway, Runway 19, is proposed in the concept at a length of 10,000 feet, and expandable to a length of 12,000 feet. The ultimate length is achieved by constructing a 2,000-foot extension on the end of Runway 19. Runway 1 is also equipped for precision instrument approaches. The submittal specifies a 3.1 degree glide slope for Runway 1. The runway is proposed for arrivals during Santa Ana wind conditions and also to accommodate arrivals of Approach Category D aircraft. The elevation of the end of Runway 1 is indicated at 300 feet MSL. This is approximately 60 feet more than the existing ground elevation and will require significant fill.

The proposed airfield of the October 5, 1999, submittal (Alternative 1) is assessed below with respect to those factors considered for the April 7, 1999 submittal.

## Approach Slope

A non-standard approach angle of 3.3 degrees is specified for the primary arrival runway, Runway 16. This precludes use of the runway for arrival operations by aircraft with approach speeds of at least 141 knots. This includes the following aircraft commercial transport aircraft: L-1011, DC-10-30, DC-10-40, DC-8-61, DC-8-63, MD-11, B747 (all models), B777, B767-300, B757-300 and the B737-800. It also precludes approaches by Gulfstream II, IV and V, and Lear 35 business jets. The submittal suggests a strategy for accommodating Approach Category D aircraft whereby Runway 1 is used for arrivals by these aircraft. As explained herein, this would be a "counter-flow" or head-to-head operation and raises safety and capacity concerns.

A 3.3 degree glide slope would have landing minimums of 250 foot decision height and visibility of ¾ miles. These are slightly higher than standard Category I minimums of 2001/½.

#### Runway 16 Approach

The development of the instrument approach procedure for Runway 16 contained in the submittal does not include adverse obstacle assumptions and allowance for precipitous terrain. These factors are typically applied by FAA in the development of instrument

procedures and most likely would be applied for instrument procedures at El Toro. These were applied in the previous assessment of instrument approach procedures conducted as part of the Airport System Master Plan and are prudent for the planning of a new airport.<sup>1</sup> Considering these allowances and the proposed approach procedure for Runway 16, it is found that a 3.3 degree glide slope is not possible for the proposed landing threshold. In order to provide a 3.3 degree glide slope the landing threshold would have to be displaced an additional 1,900 feet to the south. If the proposed landing threshold is retained, a glide slope angle of 3.6 degrees would be required to provide adequate obstacle clearance.

The options of additional displacement of the landing threshold or a steeper (3.6 degree) glide slope required to implement an acceptable approach procedure, render the proposed concept of Runway 16 arrivals not feasible for a new commercial airport. Even if a 3.3 degree approach procedure could be developed in accordance with standard FAA practices, it is not a desirable basis for planning the main arrival runway at a commercial airport.

#### Runway 1 Approach

An analysis of the proposed approach for Runway 1 concluded that a glide slope of 3.1 degrees is possible for the proposed landing threshold location and elevation.

#### **Runway 19 Departures**

Departures on Runway 19 will require a minimum climb rate of 300 feet per nautical mile to an altitude of 1,600 feet MSL before resuming a standard rate of climb.

# Approach Category D Aircraft Limitations

The steep glide slope proposed for Runway 16 prevents the use of the runway for arrival purposes by aircraft with approach speeds of 141 knots and greater. It is estimated that this restriction would affect 29,700 aircraft arrivals projected for the Proposed Project in 2020 (see Table 8.9-1), or 21 percent of all commercial aircraft arrivals forecast for OCX in 2020.

As may be noted in Table 8.9-2, several market segments are particularly impacted, including passenger arrivals on Asia-Pacific routes (100 percent of arrivals), Atlantic routes (72 percent), and domestic long-haul routes (33 percent of arrivals), as well as all-cargo arrivals on international routes (94 percent of arrivals) and domestic routes (32 percent of arrivals). If these operations cannot be accommodated at the airport, the ability of the airport to provide the range of services envisioned under the Proposed Project would be severely limited, particularly for international passenger and cargo markets.

<sup>&</sup>lt;sup>1</sup> Instrument Flight Procedures Analysis Final Report. K&M Consultants. May 1998.

Table 8.9-1
Summary of Arrivals by Type of Aircraft Impacted by Short "V"
Alternative Airport Runway Layout at OCX in 2020

Aircrast Type	Augual Arrivals	Percent Total
A340	2,823	10%
A3XX/B7XX	1,583	5%
DC10	631	2%
MD11	4,853	16%
737-800	1,599	5%
747	4,894	16%
757-300	1,064	4%
767-300/400 (some)	5,695	19%
777	6,558	22%
Total	29,698	100%

Source: P&D Aviation

NOTE: [1] Number of Category D aircraft operations based on the projected fleet mix reflected

in the Proposed Project.

Table 8.9-2
Summary of Impact Short "V" Alternative Airport Runway Layout on Projected
Aircraft Arrivals at OCX in 2020

	: Annual C	perations.	% Arrivels
	Alt. B	Impacted [2]	Impacted
	Passenger 2	terivals, 🦖 .	
Domestic			
Regional/Commuter	19,950	0	0%
Short Haul	36,400	543	1%
Medium Haul	27,650	3,544	13%
Long Haul	23,150	7,635	33%
Subtotal-Domestic	107,150	11,721	11%
International			
Latin America	5,650	867	15%
Atlantic	2,900	2,091	72%
Asia-Pacific	8,850	8,850	100%
Canada/Other	1,000	111	11%
Subtotal-International	18,400	11,918	65%
Total Passenger Operations	125,550	23,639	19%
	All-Ça	rgo	100
Domestic	10,350	3,286	32%
International	2,950	2,774	94%
Total-All Cargo Operations	13,300	6,060	46%
Territoria de la Companya del Companya de la Companya del Companya de la Companya	tal Commercia	al-Operations	
Total Commercial Operations	138,850	29,699	21%

Source: P&D Aviation

NOTE: [1] Number of Category D aircraft operations based on the projected fleet mix reflected in the Proposed Project.

Alternative Category C equipment providing similar lift capacity is available to substitute for some of the aircraft impacted by the restriction. Examples of alternative aircraft include the A320/321 and B737-400 for the B737-800, and the A300 and A330 for the B757-300 and B767-300/400. However, there are effectively no alternative non-Category D aircraft available to replace the larger aircraft affected by the restriction.

The fleet mix of individual airlines that choose to provide service at the airport will dictate the availability of these particular aircraft types. Since most domestic airlines and many international airlines have at least some of the impacted aircraft types in their fleet, it is reasonable to expect that a constraint at OCX that would restrict the flexibility of the airlines to assign equipment to the airport would place OCX at a competitive disadvantage relative to other airports in the region without such restrictions. This would further limit the ability of OCX to provide the range of services anticipated under the Proposed Project.

The submittal suggests a strategy for accommodating Category D arrival operations. This is achieved by using Runway 1 for arrivals. The proposed strategy involves a counterflow operating configuration in which departures from Runway 16 would track outbound in the opposite direction of arriving aircraft. Such an operating mode raises safety and capacity concerns.

### Capacity

The assessment of capacity issues for the April 7 submittal also applies to the October 5, 1999, submittal. However, the concept is less efficient than the April 7 submittal from the standpoint that it suggests an operating mode that promotes counterflow operations. This operating mode is proposed as a means of overcoming limitations on arrivals by Approach Category D aircraft that are inherent in the alternative.

The assessment of Category D limitations concluded that approximately 21 percent of the forecast fleet mix cannot use Runway 16 (the primary arrival runway) and must use Runway 1 for arrivals. Use of Runway 1 for arrivals will close down the primary departure runway (Runway 19) during these periods. Considering that Category D arrivals can be expected on a regular basis throughout the day suggests that inefficiencies at best can be expected due to the frequent need to change operating configuration of the runways.

# Flight Tracks

Proposed flight tracks are basically "straight-in" and "straight-out" for arrivals and departures with the exception of departures on Runway 16. The proposed departure specifies a right turn be commenced shortly after takeoff, followed by a left turn so that the departure track parallels that of Runway 19. The turning departure is intended to avoid residential communities of Laguna Village, Leisure World and Laguna Hills. The ability to avoid these noise sensitive areas and follow the depicted flight tracks will be controlled by the point at which turns actually occur and the radius of turn that is executed.

The following comments are based on data depicted on an exhibit entitled "Proposed Flight Paths The Alternative Airport" contained in the October 5, 1999, submittal. The submittal indicates that the right turn out on departure will be initiated approximately 5,000 feet from brake release with a turn radius of approximately 1 nautical mile. After turning approximately 50 degrees from the runway heading, a left turn is executed to parallel the departure track of Runway 19.

This is not representative of the flight tracks that could be expected by the proposed operation. First, many aircraft would commence the proposed initial right turn out farther south than depicted. A distance of 8,000 feet from brake release is reasonable, and it is understood that some aircraft would require longer distances. Taking this into account, the proposed departure track would then over fly the northern edge of the noise sensitive area. Secondly, larger turn radii than that which has been assumed would direct the departure track farther south such that a greater portion of residential area is over flown. The flight track as shown does not reflect the normal range of ground tracks that can be expected due to dispersion. Lastly, the flight track is based on the assumption that a departure procedure involving the proposed double turns will be implemented. If a procedure based on only a right turn is employed, this would result in over flight of the greatest portion of the Laguna Village area.

### Runway Length Available for Takeoff

Geodetic computations of the runway end coordinates given for Runway 1-19 indicate a length of 9,333 feet (versus a length of 10,000 feet depicted in the submittal). Taking into account the need to provide required runway object free area, a takeoff length of approximately 9,000 feet is available on Runway 19 with the "Short V" Alternative Runway Layout presented in the submittal. This takeoff distance is significantly less than the takeoff distances provided by other alternatives (takeoff lengths greater than 11,000 feet are available under the Proposed Project).

# Runway Length Available for Landing

The review of the Runway 16 approach procedure concluded that the proposed landing threshold location cannot be assumed to provide a 3.3 degree approach slope. A 3.6 degree glide slope is possible for the proposed landing threshold location. In order to provide a 3.3 degree glide slope the landing threshold would need to be displaced an additional 1,900 feet to the south. Considering the need to provide 1,000 feet of Object Free Area beyond the stop end of the runway results in an available landing distance of approximately 5,800 feet. This landing distance is significantly less than the landing distances provided by other alternatives (as previously indicated to be approximately 9,900 to 10,400 feet).

It was previously stated that geodetic computations of the runway end coordinates given for Runway 1-19 indicate a length of 9,333 feet. When requirements for runway object free area are taken into account this results in an available landing distance of approximately 8,400 feet for the Short V configuration. This landing distance is significantly less than the landing distances provided by other alternatives.

#### Airfield Geometrics

The conclusions stated for the April 7 submittal apply to this alternative airport configuration.

## Taxiway Operations

Previously stated inefficiencies of the April 7 submittal due to the inability to provide dual parallel taxiways in the terminal area are also inherent in this concept.

#### 8.9.2.2 Terminal and RON Parking

A linear terminal complex is proposed and is located in between the "V" shaped runways. A difference between this concept and the April 7 submittal is that retail uses previously proposed in the terminal building appear to have been eliminated with the building area used for aircraft gates. This would provide an additional 1,100 feet of ramp frontage (approximately 10,100 feet overall). This is less than that provided by the preferred terminal concept (which provides over 11,500 feet of ramp frontage).

As previously described for the April 7 submittal, the setback of the terminal from the runway will limit the heights of aircraft that can park along the face of the building. RON parking is not included in the concept.

# 8.9.3 Nonaviation Revenue Support Uses

Although not specifically included in the proponent's information about this alternative, it is assumed that nonaviation uses similar to those included in the Proposed Project would be included. Because of the runway configuration, this alternative would not have the same space available for the wildlife habitat area and preservation of existing agricultural uses as the Proposed Project.

# 8.9.4 Environmental Impacts of Wildlands Ranch Alternative

### 8.9.4.1 General Impacts

As previously discussed, this alternative presents serious technical and feasibility concerns. Due to these unresolved issues, the alternative could not feasibly attain most of the project basic objectives. With respect to environmental impacts, while some impacts under this alternative would be similar to those of the Proposed Project, this alternative would not avoid or substantially lessen impacts projected to result from the Proposed Project. In fact, this alternative would result in a number of impacts that would be greater than the Proposed Project, including noise and biological resources, as discussed below.

#### 8.9.4.2 Noise

Review of this alternative indicates it may produce the greatest noise impacts of any alternative considered in the El Toro Master Planning Process.

This alternative, as presented by the Proponent, includes several noise footprints which are referenced as produced by the County's consultants. In fact, none of the noise contours included in the proponent's submittal were generated or reviewed by the County except for those that are direct reproductions of contours provided by the County consultants for the Proposed Project. Noise contours which are purported to reflect SENEL and CNEL contours for this alternative were not produced by the County consultants and are not credible representations of the noise footprints that would be created by this alternative.

This alternative assumes departures to the south on Runways 19 and 16. The Runway 19 departures would not expose existing residential uses to noise levels in excess of 65 CNEL. However, the track from Runway 19 would overfly a portion of the Irvine Meadows Amphitheater and near the Irvine Medical Center. The associated noise levels would probably exceed FAA Part 150 recommended noise levels for such uses as well as exceed County of Orange and City of Irvine noise standards for such uses.

The Runway 16 departures assume a right turn "as soon as possible." It is these right turns that cause the greatest noise impact on existing noise sensitive uses. If the turns do not occur very early, the 65 dB CNEL contour would likely impact most of the residential areas of Laguna Village, part of Laguna Hills and potentially Laguna Woods. Should residential uses be allowed in the Irvine Spectrum area by the City of Irvine, then these residential areas would also be impacted by planes taking off of Runways 19 and 16. The extent of the noise impact will depend on where aircraft departing Runway 16 make their turns and how small a radius is used for those turns. For the CNEL contour to impact no residential uses, the turns would have to be completed prior to reaching the present end of Runway 16, not the proposed relocated end of Runway 16. This would require a very sharp turn very early in

the departure procedure. This turn would be highly unlikely for heavy aircraft and subsequent late turns by heavy aircraft would impact Laguna Village, Laguna Hills and Leisure World. Given the typical dispersion associated with this right turn, the expected 65 dB CNEL contour would impact more residential uses than any other Alternative considered. Other than this alternative, only Alternative J impacts residential uses with noise in excess of 65 dB CNEL.

These conclusions were based on the size and shape of the CNEL contours shown for Alternative A but applied to the runway configuration under this alternative. The Alternative A contours were assumed because the runway capacity of this alternative is less than the Proposed Project. If noise contours for the Proposed Project are applied, the impacts would be greater than those indicated above.

The 65 CNEL for this alternative would affect seven residential planning areas in the City of Irvine and the City's Sphere of Influence. While there are few, if any, residential units currently constructed in these planning areas, all of them include major future residential planned communities or villages that would be affected by the 65 CNEL. The departure corridor would result in a 65 CNEL affecting approximately 1,200 dwelling units proposed in Planning Area (PA) 33, 2,030 units allocated by the City General Plan to PA 17, and 750 units allocated to PA 18 for a total of 3,980 dwelling units. In addition to these units, the 65 CNEL could affect dwelling units in PA 22 and the Laguna Laurel Planned Community depending on the extent of the 65 CNEL and the final development plan for the Laguna Laurel Planned Community (this planned community would permit up to 2,042 dwelling units).

The arrival corridor 65 CNEL for this alternative would affect PAs 2, 5, and 9 in the City's Sphere of Influence, which also include planned residential units. However, most of the area affected by the 65 CNEL has been enrolled in the NCCP Program. The approach corridor 65 CNEL would also affect residential development planned in the East Orange General Plan. However, no zoning or development plans have been proposed for these future residential areas, so the impact cannot be estimated reliably. Based on plans available, the alternative is expected to impact 3,000 to 6,000 future homes.

Mitigation of the 65 CNEL impacts on existing and future residential development is not feasible. There is no mitigation for exterior noise levels of 65 CNEL or higher that would reduce the effects of aircraft noise on existing housing. In addition, amendments to the City of Irvine and County General Plans to eliminate or relocate thousands of future residential units is infeasible because there are no alternative locations for the 3,000 to 6,000 future units affected by the 65 CNEL. Most of the remaining unentitled lands (30,000+ acres) in central Orange County have been enrolled in the NCCP Program. In addition, amendment to the City's Conservation/Open Space Plan to accommodate transfer of dwelling units is not considered feasible and may require a City ballot measure. Purchase of the development rights for these units would be prohibitively high.

For these reasons, this alternative would have a significant adverse noise impact, which would not be reduced by feasible mitigation measures. Therefore, this alternative would have the greatest adverse noise impact of any of the alternatives analyzed herein.

## 8.9.4.3 Biological Resources

This alternative is substantially different from any of the other alternatives considered in several ways. One of the differences is how the concept accommodates the federal Habitat Reserve. The nonaviation land use component does accommodate a wildlife habitat area, although the southerly portion of this area is substantially narrower than under Proposed Project.

Under this alternative, the extension of existing Runway 16L-34R restricts available open space to the extreme east side of the MCAS El Toro site. This more narrow, constricted area requires the realignment and reconfiguration of the federal Habitat Reserve. South of the proposed Jeronimo Road extension there is a narrow area reserved for wildlife use. The area is constrained by the runway extension on the west and the airport property to the east. This narrow segment of the wildlife area extends an estimated 7,000 feet before a new wildlife corridor underpass is provided at Bake Parkway. The width of the wildlife area is estimated at 500 feet, with the runway and manicured aviation land to the west and an industrial park immediately adjacent to the east. This alternative relocates the wildlife underpass at I-5 to the San Diego Creek outlet, rather than at Serrano Creek. This relocation lengthens the wildlife habitat area on the MCAS El Toro site. To accommodate this area, the Alternative requires the realignment of Serrano Creek to join with San Diego Creek, south of Bake Parkway. This realignment represents a new impact to soft bottom habitat not reflected in the Proposed Project.

The most substantive difference between this alternative and the Proposed Project is the wildlife habitat area configuration. In addition to the changes discussed on the southern portion of the area, there are some modifications to the north. The access into the Habitat Reserve occurs further to the east and accommodates a new equestrian use.

This alternative does not share some of the project components discussed for the Proposed Project, or they vary in their locations. For example, there is no provision for Astor Road. The Alton/Barranca intersection is relocated and reconfigured. There is no provision for the future Rockfield alignment, which is inconsistent with the Master Plan of Arterial Highways. Serrano Creek is realigned to the east with a new culvert requirement at Bake Parkway to transition into San Diego Creek. These modifications constrain and lengthen the Wildlife Habitat area, subjecting this area to additional nighttime illumination and to more contiguous aircraft activity, and reduce golf and agricultural open space buffers relative to the Proposed Project. The alternative is not expected to be as functional for wildlife movement as that proposed as part of the Proposed Project due to the narrower width and proximity to more indirect aviation activities along the runway extension.

With the exception of differences in the wildlife habitat area, this alternative is not expected to generate substantially different direct impacts on biological resources than under the Proposed Project. Direct impacts to native plant communities, wildlife, wildlife dispersion corridors and special interest species are very similar to the Proposed Project. Indirect impacts resulting from aircraft overflights are discussed below.

The flight tracks under this alternative are also substantially different from the Proposed Project. Similar to Alternative J, there are no direct overflights associated with the federal Habitat Reserve due to the elimination of the east-west runway. The aircraft overflights are generally reversed and aircraft generally land from the north and depart to the south. One of the purposes of this runway configuration is to direct overflights into a different area of the San Joaquin Hills, rather than having overflights over the populated Aliso Viejo area. These overflights would occur in the Shady, Bommer, Moro Canyon areas, that are a part of the NCCP reserve. Similar to the Proposed Project, these overflights are not expected to result in substantially new biological resource impacts.

#### 8.9.5 Conclusions

The alternative limits operations by certain aircraft types. The limitation does not encourage the growth of air service and general aviation opportunities. Service opportunities such as international, domestic long haul, and cargo are not accommodated by the alternative. Thus, aviation demand is not served, and economic growth is not enhanced to the same degree as the Proposed Project. The alternative is technically inferior to the Proposed Project and other alternatives with respect to several airport planning issues. The alternative also impacts residential land uses to noise in excess of 65 dB CNEL.

# 8.10 LAND USE ALTERNATIVES AT FORMER MCAS EL TORO

This section evaluates possible alternatives to the nonaviation uses on the MCAS El Toro site under the Proposed Project for the purpose of reducing any significant adverse impacts pertaining to the nonaviation uses to below the level of significance. In both alternatives, the aviation-related land uses of the Proposed Project would remain the same as described in Chapter 3.0. These alternatives to the nonaviation uses do not affect the ability of the Proposed Project to meet the general project and aviation related objectives.

# 8.10.1 Nonaviation Revenue Support Area Alternative 1

In summary, this alternative was selected for analysis because it has the potential to lessen the project impacts on Prime Agricultural Soils, traffic, and traffic related impacts while still obtaining most of the objectives of the project.

Figure 8-13 shows the proposed Nonaviation Revenue Support Area Alternative 1 land use plan and Table 8.10-1 shows the proposed land uses and acreages by parcel. In summary, in comparison to the Proposed Project, this alternative would:

- 1. Delete the Business Park uses in Planning Area 7 and retain the existing agricultural uses to reduce the loss of Prime Agricultural Soils.
- 2. Relocate the Regulation Golf Course from the southerly portion of Planning Area 3 (thus retaining approximately 120 acres of Prime Agricultural Soils to the northerly portion of Planning Area 2 replacing (a) Regional Park uses and (b) Cultural/Institutional uses planned in the Proposed Project (formerly military housing areas). Cultural/Institutional uses would be reduced from approximately 80 acres to approximately 40 acres in Planning Area 2. Approximately 160 acres of Regional Park uses would be reclassified to Golf Course uses. The equestrian stable area in the Proposed Project (Parcel 2-2) would remain unchanged (approximately 36 acres).
- 3. Relocate the Vehicle Maintenance Yard and the State Department of Education Warehouse from Planning Area 5 to Planning Area 3, deleting an equal area of Regional Park uses in the high aircraft noise portion of Planning Area 3. This change from the Proposed Project would reduce the loss of farm lands by about 64 acres in Planning Area 5.
- 4. Relocate the proposed Executive Golf Course from Planning Area 7 to the location of the existing (former military) golf course in Planning Area 3. This change would reduce the loss of Prime Agricultural Soils in Planning Area 7 by approximately 98 acres.

Table 8.10-1 Nonaviation Revenue Support Area Alternative 1

LAND USE	LU_CODE	ACRES
Agriculture	5-1a	102.76
Agriculture	3-1b	45.07
Agriculture	3-1c	13.03
Agriculture	3-1d	166.58
Agriculture	3-1e	78.98
Agriculture	3-1f	18.57
Agriculture	7-1g	22.22
Agriculture	7-1h	70.35
Agriculture	7-1i	6.24
Agriculture	7-1j	181.40
Agriculture	7-1k	33.05
Agriculture To	<b>id</b> io Age de l'	738.25
Park Uses	2-2	35.94
Park Uses Tot		35.94
7 2 2 2 2 2		7///
Golf Course (North)	2-3a	197.26
Golf Course (North)	2-3b	41.24
Golf Total		238.50
Habitat Reserve	6-5a	970.45
Habitat Reserve	3-5b	20.64
Habitat Reserve	Cotal	991.09
Restricted Open Space	8-6a	12.98
Restricted Open Space	8-6b	4.64
Restricted Open Space	8-6c	68.83
Restricted Open Space	8-6d	32.45
Restricted Open Space	8-6c	11.42
Restricted Open Space	8-6f	16.29
Restricted Open Space	7-6g	30.16
Restricted Open Space	7-6b	9.14
Restricted Open Spa		185.92
resurere open spa	CE A CHARL	1000
Marshburn Retarding Basin	5-7	38.87
	Total	38.87
Airfield Total	8-8	1,120.03
Airlield Total		1,120.03
	1 - 9 - 19 - 19 - 19 - 19 - 19 - 19 - 1	
Parking	5-9a	63.65
Terminal and Parking	1-9b	481.10
Terminal and Parkin	g Total	544,75
Airport Shuttle Rail Yard	4-10a	12.53
Airport Transportation Center	4-10a 4-10b	9.34
Amport Transportation Center  Transportation Facili	1	21.86
i anopul muvi facili	· J AULAL	41,00

4-11a 3-11b 2-12a 2-12b	. 193.67
2-12a	
2-12a	1 <b>93.67</b> 17.07
	17.07
2-12b	
	99.49
2-12c	4.00
2-12d	54.28
2-12e	4.00
3-12f	25.71
3-12g	46.41
4-12h	14.00
4-12i	14.30
4-12j	5.45
tal	. 284.72
3-13a	57.40
3-13b	6.18
3-13c	5.73
3-13d	1.20
3-13e	25.36
3-13f	9.00
4-13g	28.29
4-13h	1.00
alt d'ann	134.16
2-16	8.00
otal	8.00
	156.62
Cotal .	156.62
	4,692.38
	3-12f 3-12g 4-12h 4-12i 4-12j 4-12j  tal  3-13a 3-13b 3-13c 3-13d 3-13c 3-13f 4-13g 4-13h  2-16  Total

# 8.10.2 Environmental Impacts of Nonaviation Revenue Support Area Alternative 1

Compared to the Proposed Project, this alternative is intended to reduce the project impacts on traffic (and traffic related impacts such as air quality and highway noise) and on Prime Farm Lands. This alternative would result in a net reduction in Regional Park acreage, a net increase in farm lands, and elimination of the Business Park as further described below. In addition, this alternative reduces the aircraft noise impacts on recreational uses by reducing the planned recreation areas in the high noise impact areas (i.e., 70+ CNEL).

The alternative would also reduce the development costs for Regional Park uses and Cultural/Institutional uses, and eliminate development costs for the Business Park. The alternative would slightly increase revenues from farm land leases, reduce revenues from Cultural/Institutional and Regional Park uses, and eliminate revenue from the Business Park.

#### Land Use

Compared to the Proposed Project, this alternative would result in a reduction in the lands planned for the Business Park and Cultural/Institutional uses, an increase of approximately 379 acres of farm lands, and relocations of the recreational uses, the Vehicle Maintenance Yard, and the State Department of Education Warehouse uses. No changes are proposed to aviation uses, aviation support uses, habitat uses, or other public facilities (e.g., homeless providers, IRWD, OCTA rail yard and ANG). As with the Proposed Project, there would be no significant impact related to land use compatibility.

# General Plan Consistency

The adjustments to nonaviation revenue support use do not affect the need for General Plan Amendments compared with the Proposed Project.

# Transportation and Circulation

This alternative would reduce the trips generated by the Project as follows:

CHANGE I	IN T	RIP GE	NERA	TION:
----------	------	--------	------	-------

LAND USE	AM PEAK	PM PEAK	TOTAL	% CHANGE
Business Park	-1,735	-1,566	-15,775	(-100%)
Cultural/Institutional	-390	-350	-3,920	(-50%)
Regional Park	-149	-179	-3732	(-81%)
Agricultural Uses	+4	+4	+798	(+480%)
Total	-2,270	-2,091	-22,629	(N/A)

In summary, this alternative would reduce the Proposed Project trip generation as follows:

PROJECT CASES:	TRIP G	GENERATION FOR OCX	AREA:
	AM PEAK	PM PEAK	TOTAL
Proposed Project:	11,498	12,796	176,123
Alternative:	9,228 (-20%)	10,705 (-16%)	153,494 (-13%)
Existing Conditions:	2,200	2,300	25,400

This alternative would reduce the peak hour trip generation by 20 percent in the morning peak hour and 16 percent in the afternoon peak hour, which is a significant reduction. This alternative could result in a measurable reduction in peak hour conditions east and southeast of the OCX project area.

Elimination of the Business Park and relocation of the Executive Golf Course would have a measurable reduction in trips on Rockfield Drive, Alton Parkway, Bake Parkway, Barranca Parkway, and the I-5 Freeway ramps in the vicinity of the Business Park site. The relocation of the Vehicle Maintenance Yard and State Warehouse would reduce trips on Portola Parkway between Sand Canyon and the Foothill Tollroad and increase by an equal amount the trips on Irvine Boulevard between Sand Canyon and Bake Parkway. This increase would be partially offset by reductions in trip generation for Cultural/Institutional uses and regional park uses. The increase in agricultural trips and relocation of Regulation Golf Course trips would have insignificant effects.

The traffic impacts of the Proposed Project are reduced by mitigation measures to a level of insignificance. Therefore, this alternative would not result in a different conclusion regarding project impacts. However, this alternative would reduce the extent and costs of required mitigation for traffic impacts, and would reduce traffic impacts in congested areas to the east and south of the project site, especially in the Spectrum and Lake Forest areas.

# Aviation Compatibility

The alternative would virtually eliminate recreational uses (i.e., golf course areas) within the Safety Zones for the Proposed Project and retain most of the existing agricultural uses located in these Zones. In addition, this alternative would virtually eliminate Regional Parktype uses within high aviation noise areas (i.e., 70+ CNEL).

Since Golf Courses and Regional Park-type uses proposed in the Safety Zones and high aviation noise areas are considered compatible land uses, the Proposed Project would not result in recreational uses being incompatible with aviation activities. Therefore, this alternative would not change the conclusions of the Proposed Project impact analysis.

# Air Quality and Highway Noise

This alternative would reduce the highway traffic (mobile source) air quality emissions by about 13 percent, and total on-site generated project emissions by approximately 5 percent.

Highway noise impacts in the project area would be reduced significantly compared to the project case on Rockfield Drive in and near the site, and to a lesser degree on Bake Parkway, Alton Parkway, and Barranca Parkway. Highway noise would be reduced on Portola Parkway near the project site to a less than significant level. However, highway noise would be increased on Irvine Boulevard in and near the project site to a less than significant level.

This alternative would not result in a different conclusion regarding project impacts. However, this alternative would measurably reduce the total project air quality emissions due to mobile sources, and would measurably reduce highway noise impacts in some areas near the project site.

#### Recreation

This alternative would reduce the amount of active and passive types of recreational uses included in the proposed regional park area, retain the project proposals for equestrian stables, and Executive Golf Course and a Regulation Golf Course, and increase farm land for an overall insignificant change in total open space. Therefore, this alternative would have approximately the same effects as the Proposed Project in providing open space in the rapidly urbanizing central and southern Orange County area. This alternative would preserve less open space, however, than would be preserved by the No Project/No Activity Alternative, but the alternative would provide a larger amount of active recreational uses (i.e., two golf courses) compared to the No Project case.

However, this alternative would result in a large reduction in Regional Park-type uses compared to the Proposed Project. The demand for active (e.g., ballfields) and passive (e.g., picnicking) Regional Park-type uses in the project area is significant, and recreational surveys for the project and studies by the County and nearby cities demonstrate a significant unmet need. This alternative would reduce or eliminate the opportunity to meet this demand. Therefore, this alternative would be inferior to the Proposed Project in meeting the demand for Regional Park-type recreational uses.

# Natural Resources and Energy

This alternative would increase the amount of agricultural lands from approximately 139 acres under the Proposed Project to approximately 517 acres. Per Section 4.1.6 of the Draft EIR, the Department of Navy leased approximately 1,040 acres for agricultural uses at the OCX site, of which 726 acres have been classified as "Prime Farmland" and 92 acres have been classified "Farmland of Statewide Importance" by the U.S. Department of Agriculture, Natural Resources Conservation Service.

According to Section 4.11.1.1, P. 4-453 and Figure 4-91 of FEIR 563, all of Planning Area 5 (approx. 269 acres), portions of Planning Area 3 (approx. 175 acres), portions of Planning Area 7 (approximately 245 acres) and a portion of Planning Area 8 (approx. 40 acres) are classified Prime Farmlands. Note, however, that since the U.S. Department of Agriculture's

classification, Marshburn Detention Basin has been constructed, reducing the lands classified as Prime Farmland by approx. 39 acres to a total of 230 acres in Planning Area 5.

Of these Prime Farmlands, 166 acres (72 percent) would be retained in Planning Area 5, 131 acres (75 percent) would be retained in Planning Area 3, and 203 acres (83 percent) would be retained in Planning Area 7. All of the Prime Farmland in Planning Area 8 would be lost due to the ROFA and RPZ proposed for OCX. In total, of the 726 acres classified Prime Farmland (687 acres after construction of Marshburn Basin), approximately 500 acres would be retained by this alternative. Note that the construction of the Rockfield Drive extension in Planning Area 7, although not required for this alternative, would remove approximately seven acres of Prime Farm Land.

Figure 4-453 also shows that portions of Planning Area 7 (approximately 82 acres) and a portion of Planning Area 3 (approximately ten acres) are classified Farmland of Statewide Importance. Virtually all of this land is located in the Runway Obstacle Free Area (ROFA) or the Runway Protection Zone (RPZ) proposed for OCX. According to FAA Advisory Circular AC 150/5200-33, the FAA recommends that no agricultural activities be conducted in the ROFA and related zones to ensure safe, efficient aircraft operations. Therefore, the ROFA and RPZ for the Proposed Project and this alternative would result in the loss of approximately 67 acres of Farmland of Statewide Importance. In addition, the proposed location of the IRWD Aqueous Waste Treatment Plant in Planning Area 7 would remove another nine acres of these Farmlands for a total loss of approximately 76 acres.

In conclusion, the Nonaviation Revenue Support Area Alternative 1 would significantly reduce the loss of farmland compared to the Proposed Project. The alternative would, however, result in a significant loss of farmland compared to the No Project/No Activity Alternative.

In regard to Farmlands of Statewide Importance, the ASMP includes an alternative that would locate all of the ROFA and RPZ on the north side of the AT&SF Metrolink railroad in the unincorporated area. This alternative could reduce the loss of Farmlands of Statewide Importance from 76 acres to approximately 10 acres of loss, which would reduce the impacts to a level of insignificance.

#### Cultural/Institutional Uses

This alternative would reduce the Cultural/Institutional acreage by over 50 percent from the Proposed Project, which could potentially result in the site being unsuitable in size to accommodate the proposed branch university. However, this site would still be large enough to accommodate the remaining Cultural/Institutional uses proposed by the project (e.g., Sheriff's education center, etc.). Therefore, this alternative would be expected to have an adverse impact on accommodating a portion of the demand for a branch university in southern Orange County.

## Feasibility

This alternative would reduce development costs for nonaviation uses, so economic feasibility is enhanced.

#### Conclusion

Under this alternative, the level of development at MCAS El Toro would be less intense than with the Proposed Project. This would result in slight differences in effects from the Proposed Project, for example, fewer trips and fewer jobs created. Most of the impacts would be similar to, or slightly less than, the impacts of the Proposed Project. The primary difference in effects is that more agricultural land would be preserved although the impacts would remain significant under this alternative. However, for many of the impact categories for which this alternative results in slightly less than the Proposed Project, the Proposed Project does not result in significant unmitigated impacts; thus implementation of this alternative would not reduce any significant impacts. For these reasons, this Draft EIR proposes to reject this alternative.

# 8.10.3 Nonaviation Revenue Support Area Alternative 2

In summary, this alternative was selected for analysis because it has the potential to avoid project impacts on Prime Agricultural Soils and lessen impacts on traffic and traffic related impact while still attaining the objectives of the project.

Under Nonaviation Area Alternative 2, nonaviation uses proposed are shown in Table 8.10-2 and Figure 8-13. Compared to the Proposed Project, this alternative is intended to reduce the project impacts on traffic (and traffic related impacts such as air quality and highway noise) and on Prime Farm Lands. Compared to the Proposed Project, this alternative would result in a net reduction in:

- a. Regional Park acreage
- b. Golf Course acreage
- Proposed County Wildlife Habitat area
- d. Business Park area
- e. Cultural/Institutional uses

This alternative would result in a net increase in farm lands compared to the Proposed Project. In addition, this alternative reduces the aircraft noise impacts on recreational uses by reducing the planned recreation areas in high noise impact areas.

The alternative would also reduce the development costs for Regional Park uses and the Cultural/Institutional uses and eliminate development costs for the Business Park, the

Table 8.10-2 Nonaviation Revenue Support Area Alternative 2

Agriculture Agriculture Agriculture Agriculture	5-1a 3-1b 3-1c	102.76 27.01
Agriculture Agriculture Agriculture		
Agriculture Agriculture	3-1c	
Agriculture		13.03
	3-1d	0.37
Agriculture	3-1e	119.94
Agriculture	7-1f	61.35
Agriculture	7-1g	6.24
Agriculture	7-1h	53.62
Agriculture	7-1i	99.81
Agriculture	7-1j	33.05
Agriculture Tot		517.18
Park Uses	2-2	35.94
Park Uses Tota		35.94
Golf Course (North)	2-3a	165.16
Golf Course (North)	2-3a 2-3b	41.24
Golf Course (South)	3-3c	15.32
Golf Course (South)	3-3d	47.04
Golf Total		268.76
GOLI GOL	) 	200.70
Open Space	3-4a	0.78
Open Space	3-4a 3-4b	0.78
Open Space Total		0.33 131
Open Space 100	ol:	الال
Habitat Reserve	6-5a	970.45
Habitat Reserve	3-5b	20.64
Habitat Reserve T		
Mapriat Reserve I	VIZI	. 991.09
Restricted Open Space	8-6a	12.98
Restricted Open Space	8-6b	4.64
Restricted Open Space	8-6c	68.83
Restricted Open Space	8-6d	32.45
Restricted Open Space	8-6e	11.42
	·	
Restricted Open Space	8-6f	16.29 38.02
Restricted Open Space	7-6g 7-6h	9.14
Restricted Open Space	,	9.14 <b>193.78</b>
Restricted Open Space Total		193./8
		20.05
Marshburn Retarding Basin	5-7	38.87
Retarding Basin T	otal	38.87
Airfield Total	8-8	1,120.03
Airfield Total		1,120.03
May Kirking Carlot Carl		
Parking	5-9a	63.65
Parking Terminal and Parking	5-9a 1-9b	481.10
Terminal and Parking  Terminal and Parking		481.10 <b>544.75</b>
PETHODE BUG EAR KIN	g roun	544./5
Aimed Chuttle Pail Vand	4-10a	13.52
Airport Shuttle Rail Yard		12.53
Airport Transportation Center	4-10b	9.34
Transportation Facilit	y 1.0021	21.86

LAND USE	LU_CODE	- ACRES
Cargo (west)	4-11a	137.59
Cargo (east)	3-11b	56.08
Cargo Total		193.67
In-flight Catering	2-12a	17.07
Aviation/Industrial Related	2-12b	99.49
Control Tower	2-12c	4.00
FBO/Corporate Aviation	2-12d	54.28
ARFF	2-12e	4.00
Airport Maintenance	3-12f	25.71
Airline Maintenance	3-12g	46.41
Ground Service Equipment	4-12h	14.00
Fuel Storage	4-12i	14.30
Aviation Support	4-12j	5.45
Aviation Support To	(al	279:27
Vehicle Maintenance Yard	3-13a	57.40
Food Distribution Warehouse	3-13b	6.18
IRWD Reservoir and Pumping	3-13c	5.73
Fire Station	3-13d	1.20
California Air National Guard	3-13e	24.05
Homeless Service Providers	4-13f	28.29
Fire Station	4-13h	1.00
IRWD Facility	7-13g	9.00
Public Facilities To	al	132,85
Wildlife Habitat Area	3-151	17.57
Wildlife Habitat Area	3-15b	20.09
Wildlife Habitat Area	3-15c	42.96
Wildlife Habitat Area	3-15d	18.57
Wildlife Habitat Area	7-15e	14.15
Wildlife Habitat Area	7-15f	27.98
Wildlife Habitat Area	l'otal .	150.32
Cultural Institutional	2-16	40.10
Cultural Institutional	otal:	40:10
Roads and Easements		157.16
Roads and Easements	Patel	157.16
AOROS RIIG EASCIDERS	.val	13/.10
GRANDITOTAL		4;692.38
1- Planning Area		

Executive Golf Course, and the proposed Wildlife Habitat Area. The alternative would slightly increase revenues from farm land leases, reduce revenues from Cultural/Institutional and regional park uses, and eliminate revenue from the Business Park. In summary, this alternative would:

- 1. Delete the Business Park (87 acres), Executive Golf Course (98 acres), and County Wildlife Habitat Area (40 acres) in Planning Area 7 and retain the existing agricultural uses, to reduce the loss of existing farm lands by approximately 283 acres in Planning Area 7.
- 2. Relocate the Regulation Golf Course from the southerly portion of Planning Area 3 (thus retaining approximately 120 acres of farm lands in Parcel 3-1e) to the northerly portion of Planning Area 3 replacing (a) regional park uses and (b) Cultural/ Institutional uses planned in the Proposed Project (formerly military housing areas). Relocation of the golf course would permit conversion of the existing (former military) golf course (a net area of 62.36 acres after reductions for proposed aviation and other uses) to the proposed Irvine Ranch Water District (IRWD) Aqueous Waste Treatment Plant site (9 acres) and to agricultural uses (approximately 53 acres). This agricultural area could accommodate the relocation of nursery uses from Planning Area 5 required by the proposed parking area in Parcel 5-9a. The proposed County Wildlife Habitat Area in Planning Area 3 (approximately 104 acres) would be eliminated and the existing agricultural uses would be retained.
- 3. Cultural/Institutional uses would be reduced from approximately 80 acres to approximately 8 acres in Planning Area 2, and 72 acres would be reclassified to golf course use for the Regulation Golf Course. Approximately 195 acres of regional park uses would be reclassified to golf course uses. The equestrian stable area in the Proposed Project would remain unchanged (approximately 36 acres).
- 4. Relocate the Vehicle Maintenance Yard (57.4 acres) and the State Department of Education Warehouse (6.18 acres) from Planning Area 5 to Planning Area 3. This change from the project plan would reduce the loss of farm lands by 64 acres in Planning Area 5.

# 8.10.4 Environmental Impacts of Nonaviation Revenue Support Area Alternative 2

#### Land Use

Compared to the Proposed Project, this alternative would result in a reduction in the lands planned for the Business Park, Golf, Regional Park-type uses, proposed County Wildlife Habitat Area, and Cultural/Institutional uses. This alternative would increase the amount of existing agricultural uses retained by the project from 139 acres to 738 acres, an increase of approximately 600 acres. Finally, this alternative would relocate the Regulation Golf

Course, the Vehicle Maintenance Yard, and the State Department of Education Warehouse uses compared to the Proposed Project. No changes are proposed to aviation uses, aviation support uses, or other public facilities (e.g., homeless providers, IRWD, OCTA rail yard, and ANG).

This alternative would reduce the Cultural/Institutional acreage by 90 percent, which would result in the site being unsuitable in size to accommodate the proposed branch university and virtually all the other Cultural/Institutional uses. However, this site would still be large enough to accommodate the small Cultural/Institutional uses.

## General Plan Consistency

The adjustments to nonaviation revenue support use do not affect the need for General Plan Amendments compared with the Proposed Project.

### Transportation and Circulation

This alternative would reduce the trips generated by the Proposed Project as follows:

CHANGE IN TRIP GENERATION:									
LAND USE	AM PEAK	PM PEAK	TOTAL	% CHANGE					
Business Park	-1,735	-1,566	-15,775	(-100%)					
Cultural/Institutional	-796	-716	-7,956	(-90%)					
Regional Park	-149	<b>-179</b>	-3732	(-81%)					
Agricultural Uses	+6	+6	+1,200	(+736%)					
Total	-2,674	-2,455	-26,263	(N/A)					

In summary, this alternative would reduce the Proposed Project trip generation as follows:

PROJECT CASES:	TRIP GENERATION FOR OCX AREA:						
	AM PEAK	PM PEAK	TOTAL				
Proposed Project:	11,498	12,796	176,123				
Alternative:	8,824 (-23%)	10,341 (-19%)	150,222 (-15%)				
Existing Conditions:	2,200	2,300	25,400				

This alternative would reduce the peak hour trip generation by 23 percent in the morning peak hour and 19 percent in the afternoon peak hour, which is a significant reduction. This alternative could result in a measurable reduction in peak hour conditions east and south east of the OCX project area.

Elimination of the Business Park and the Executive Golf Course would have a measurable reduction in trips on Rockfield Drive, Alton Parkway, Bake Parkway, Barranca Parkway, and the I-5 Freeway ramps in the vicinity of the Business Park site. The relocation of the Vehicle Maintenance Yard and State Warehouse would reduce trips on Portola Parkway between Sand Canyon and the Foothill Tollroad and increase by an equal amount the trips on Irvine Boulevard between Sand Canyon and Bake Parkway. This increase would be

almost entirely offset by reductions in trip generation for Cultural/Institutional uses and regional park uses. The increase in agricultural trips and relocation of Regulation Golf Course trips would have insignificant effects.

The traffic impacts of the Proposed Project are reduced by mitigation measures to a level of insignificance. Therefore, this alternative would not result in a different conclusion regarding project impacts. However, this alternative would reduce the extent and costs of required mitigation for traffic impacts, and would reduce traffic impacts in congested areas to the east and south of the project site, especially in the Spectrum and Lake Forest areas.

## Aviation Compatibility

Compared to the Proposed Project, the alternative would eliminate recreational uses (i.e., golf course areas) and proposed County Wildlife Habitat Areas within the Safety Zones for the Proposed Project and retain almost all of the existing agricultural uses located in these Zones. The ROFA and RPZ would remove small amounts of existing agricultural uses. In addition, this alternative would eliminate Regional Park-type uses within high aviation noise areas (i.e., 70+ CNEL).

Since Golf Courses, Regional Park-type uses, and proposed County Wildlife Habitat areas proposed in the Safety Zones and high aviation noise areas are considered compatible land uses, the Proposed Project would not result in open space uses being incompatible with aviation activities. Therefore, this alternative would not change the conclusions of the Proposed Project impact analysis.

# Air Quality and Highway Noise

Compared to the Proposed Project, this alternative would reduce the highway traffic (mobile source) air quality emissions by about 15 percent, and total on-site generated project emissions by approximately 7 percent. Highway noise impacts in the project area would be reduced significantly compared to the project case on Rockfield Drive in and near the site, and to a lesser degree on Bake Parkway, Alton Parkway, and Barranca Parkway. Highway noise would be reduced on Portola Parkway near the project site to a less than significant level. However, highway noise would be increased on Irvine Boulevard in and near the project site to a less than significant level.

The local and regional air quality impacts of the Proposed Project are reduced by mitigation measures to a level of insignificance. Therefore, this alternative would not result in a different conclusion regarding project impacts. However, this alternative would measurably reduce the total project air quality emissions due to mobile sources, and would measurably reduce highway noise impacts in some areas near the project site.

#### Recreation

Compared to the Proposed Project, this alternative would eliminate the amount of active and passive types of recreational uses included in the proposed Regional Park area and in the Executive Golf Course. However, this alternative would retain the project proposals for Equestrian Stables (approximately 36 acres) and a Regulation Golf Course (238.5 acres), and increase farm land for an overall insignificant change in total open space. Therefore, this alternative would have approximately the same effects as the Proposed Project in providing open space in the rapidly urbanizing central and southern Orange County area. This alternative would preserve less open space, however, than would be preserved by the No Project/No Activity Alternative, but the alternative would provide a larger amount of active recreational uses (i.e., golf course) compared to the No Project case.

This alternative would result in a large reduction in regional park-type uses. The demand for active (e.g., ballfields) and passive (e.g., picnicking) regional park-type uses in the project area is significant, and recreational surveys for the project and studies by the County and nearby cities demonstrate a significant unmet need. This alternative would reduce or eliminate the opportunity to meet this demand. Therefore, this alternative would be inferior to the Proposed Project in meeting the demand for regional park-type recreational uses.

#### Natural Resources and Energy

This alternative would increase the amount of farm lands from approximately 139 acres under the Proposed Project to approximately 738 acres. Per Section 4.1.6 of the Draft EIR, the Department of Navy leased approximately 1,040 acres for agricultural uses at the OCX site, of which 726 acres have been classified as "Prime Farmland" and 92 acres have been classified "Farmland of Statewide Importance" by the U.S. Department of Agriculture, Natural Resources Conservation Service.

According to Section 4.11.1.1, P. 4-453 and Figure 4-91 of FEIR 563, all of Planning Area 5 (approximately 269 acres), portions of Planning Area 3 (approximately 175 acres), portions of Planning Area 7 (approximately 245 acres), and a portion of Planning Area 8 (approximately 40 acres) are classified Prime Farmlands. Note, however, that since the U.S. Department of Agriculture's classification, Marshburn Detention Basin has been constructed, reducing the lands classified as Prime Farmland by approximately 39 acres to a total of 230 acres in Planning Area 5.

Of these Prime Farmlands, this alternative would retain in agricultural use 166 acres (72 percent) in Planning Area 5, 175 acres (100 percent) in Planning Area 3, and 245 acres (100 percent) in Planning Area 7. However, all of the Prime Farmland in Planning Area 8 (approximately 40 acres) would be lost due to conflicts with the ROFA and RPZ for OCX.

In total, of the 726 acres classified Prime Farmland (689 acres after construction of Marshburn Basin), 586 acres of Prime Farm Land would be retained by this alternative.

Note that the extension of Rockfield Drive, although not required for this land use alternative, would remove approximately seven acres of agricultural lands in Planning Area 7 when constructed to implement the Master Plan of Arterial Highways.

Figure 4-453 also shows that portions of Planning Area 7 (approximately 82 acres) and a portion of Planning Area 3 (approximately ten acres) are classified Farmland of Statewide Importance. Virtually all of this land is located in the Runway Obstacle Free Area (ROFA) or the Runway Protection Zone (RPZ) proposed for OCX. According to FAA Advisory Circular AC 150/5200-33, the FAA recommends that no agricultural activities be conducted in the ROFA and related zones to ensure safe, efficient aircraft operations. Therefore, the ROFA and RPZ for the Proposed Project and this alternative would result in the loss of approximately 67 acres of Farmland of Statewide Importance.

In conclusion, the Nonaviation Revenue Support Area Alternative 2 significantly reduce the loss of farmland compared to the Proposed Project. The alternative would, however, result in a significant loss of farm land compared to the No Project/No Activity Alternative.

In regard to Farmlands of Statewide Importance, the ASMP includes an alternative which would locate all of the ROFA and RPZ on the north side of the AT&SF Metrolink railroad in the unincorporated area. This alternative could reduce the loss of Farmlands of Statewide Importance from 67 acres to approximately 10 acres of loss, which could reduce the impacts to a level of insignificance.

Final EIR 563/FSA EIR 563 concluded that the CRP would not have a significant adverse impact on prime farm lands. Therefore, this alternative would not result in a different conclusion regarding project impacts.

# Feasibility

Under this alternative, the development costs for nonaviation development are reduced, which increases the feasibility of the alternative.

#### **Conclusions**

Under this alternative, the level of development at MCAS El Toro would be less intense than with the Proposed Project. This would result in slight differences in effects from the Proposed Project, for example, fewer trips and fewer jobs created. Most of the impacts would be similar to, or slightly less than, the impacts of the Proposed Project. The primary difference in effects is that more agricultural land would be preserved although the impacts would remain significant under this alternative. However, for many of the impact categories for which this alternative results in slightly less than the Proposed Project, the Proposed Project does not result in significant unmitigated impacts; thus implementation of this alternative would not reduce any significant impacts. The Draft EIR proposes to reject this alternative because it eliminates active recreational uses such as ballfields, which are needed to meet demand in South County.

# 8.11 ALTERNATIVE K: JWA – STATUS QUO AVIATION ROLES; ALTERNATIVE AIRPORT SITEFULL DOMESTIC TO FULL INTERNATIONAL; NO AVIATION REUSE AT MCAS EL TORO

# 8.11.1 CEQA Requirements for Alternative Sites

Section 15126.6 (Consideration and Discussion of Alternatives to the Proposed Project) of the CEQA Guidelines specifically describes the types of alternatives to a Proposed Project that should be evaluated in an EIR. Section 15126.6(f)(2) provides the following guidance on identifying and considering alternative sites for Proposed Projects:

- (A) Key question. The key question and first step in analysis is whether any of the significant effects of the project would be avoided or substantially lessened by putting the project in another location. Only locations that would avoid or substantially lessen any of the significant effects of the project need be considered for inclusion in the EIR.
- (B) None feasible. If the lead agency concludes that no feasible alternative locations exist, it must disclose the reasons for this conclusion, and should include the reasons in the EIR. For example, there may be no feasible alternative locations for a geothermal plant or mining project, which must be in close proximity to natural resources at a given location.
- (C) Limited new analysis required. Where a previous document has sufficiently analyzed a range of reasonable alternative locations and environmental impacts for projects with the same basic purpose, the lead agency should review the previous document. The EIR may rely on the previous document to help it assess the feasibility of potential project alternatives to the extent the circumstances remain substantially the same as they relate to the alternative.

# 8.11.2 Previous Studies of Alternative Airport Sites

Over the last approximately 30 years, a number of studies have been conducted regarding the siting of an airport in addition to JWA to serve all or some of the anticipated increase in demand for air travel in Orange County. These prior studies considered a wide range of possible sites and evaluated these candidate locations based on a number of characteristics, including suitability for aviation uses, ground transportation, physical site constraints and/or environmental considerations. Given the large number of sites that have been considered for an additional airport in Orange County, a detailed description of the site evaluation and history of environmental documentation is contained in Appendix J. A table summarizing the potential environmental impacts and other constraints of each alternative site (Table J-A) is included in Appendix J. A brief summary of the prior studies follows.

EIR No. 102, Orange County Airport Alternative Futures (DMJM 1978), identified several alternative airport sites to assist JWA in serving aviation demand in Orange County. EIR No. 102 did not evaluate either the potential environmental impacts of these alternative airport sites or the ability of these sites to accommodate a civilian airport. The following eight alternative airport sites were identified in EIR No. 102, as discussed in detail in Appendix J:

- (i) Mesa de Colorado (Rancho California)
- (ii) Ontario International Airport
- (iii) Naval Air Station (NAS) Los Alamitos
- (iv) Camp Pendleton
- (v) Chino Hills
- (vi) Long Beach Airport
- (vii) Bell Canyon

EIR No. 508 (County of Orange, 1985), prepared in support of the JWA Master Plan and Santa Ana Heights Land Use Compatibility Program, evaluated a number of alternative sites for airports which would have accommodated some or all of the expanded flight activity planned for JWA. These sites, which are discussed in detail in Appendix J, are:

- i) Camp Pendleton
- ii) Armed Forces Reserve Center (AFRC), Los Alamitos
- iii) Ontario and Los Angeles International Airports and Long Beach Airport (Combined Alternative)
- iv) Santiago Canyon
- v) San Pedro Bay/Long Beach Harbor
- vi) Chino Hills

A relatively recent study related to alternative airport sites was the <u>Airport Site Consensus Team Final Report</u> (The Planning Center 1990). That study identified a wide range of possible sites and considered potential advantages and disadvantages of each site. The report evaluated 20 sites and identified four that were considered potentially able to serve Orange County air service demand. These sites, which are discussed in detail in Appendix J, are:

- (i) Potrero Los Piños
- (ii) South Camp Pendleton
- (iii) Cristianitos Canyon
- (iv) March Air Force Base (AFB)

On June 12, 1990, the Orange County Board of Supervisors approved a motion to find that none of the four sites recommended by the Airport Site Coalition Consensus Team were appropriate for master planning at that time. On December 4, 1990, the Board of Supervisors voted to support planning efforts for a commercial airport at George Air Force Base, including a rail linkage.

Final EIR No. 563 (P&D 1996), prepared in support of the reuse of MCAS EI Toro, considered the four possible alternative sites identified in the 1990 Consensus Report. Technical Report 6 to EIR No. 563, Alternatives Definition Report for the MCAS El Toro Master Development Program, identified three possible alternative sites for an airport to serve Orange County demand: the AFRC Los Alamitos, Cristianitos Canyon and Camp Pendleton. The sites considered in EIR No. 563 and Technical Report 6 are discussed in detail in Appendix J. FEIR No. 563 concluded that none of the sites was feasible for the CRP project, as explained in Appendix J.

#### 8.11.3 Alternative Sites Evaluated for EIR No. 573

As part of EIR No. 573, no additional feasible sites were found in Orange County, and none that would satisfy the project objectives or that would avoid or substantially lessen the potential adverse impacts of the Proposed Project. Therefore, pursuant to CEQA Guidelines Section 15126.6(f), there are no additional sites to be considered for the Proposed Project. For further information, see Section 8.14.1.5, New Airport Site Only.

In addition, to the extent that increased use of other existing airports in the region to accommodate Orange County demand which would otherwise be served in Orange County under the Proposed Project could be considered an "off-site" alternative, the environmental effects of such a scenario are summarized, to the extent practicable, in the No Project/No Activity Alternative (see particularly the relevant discussion in Section 8.2.4 of this EIR).

#### 8.12 ENVIRONMENTALLY SUPERIOR ALTERNATIVE

#### 8.12.1 Introduction

Section 15126.6 of the State CEQA Guidelines requires an analysis of a range of reasonable alternatives to the Proposed Project. In particular, subsections (1) and (e)(2) in relevant part state:

(1) "The specific alternative of "no project" shall also be evaluated along with its impact: (2) If the environmentally superior alternative is the "no project" alternative, the EIR shall also identify an environmentally superior alternative among the other alternatives."

Section 8.2 summarizes the impacts of the No Project/No Activity Alternative in comparison to the Proposed Project in detail.

# 8.12.2 No Project/No Activity Impacts Summary

As analyzed in Section 8.2 and Table 8.13-1, the No Project/No Activity Alternative would not be the environmentally superior alternative because it would have greater adverse environmental impacts than the Proposed Project, in summary, as follows:

- The aviation alternatives, the ETRPA Nonaviation Alternative, and the project would generate less regional VMT.
- The aviation alternatives (except Alternative F) and the project would generate less regional air quality emissions than the No Project/No Activity Alternative.
- While the No Project/No Activity Alternative would avoid increased aviation operations
  and sleep disturbance impacts near the El Toro site, this alternative would increase
  operations and sleep disturbances at JWA and regional airports where the impacts would
  be worse due to the large number of noise sensitive uses within the 65 CNEL at regional
  airports.

For these reasons, the No Project/No Activity Alternative would not be the environmentally superior alternative.

# 8.12.3 Environmentally Superior Alternative

Based upon the comparisons in Table 8.13-1, Alternative A, because it creates less noise, transportation, local air quality emissions, toxic air contaminants due to aircraft, natural resources and energy impacts than the Proposed Project, although creating greater regional VMT and air quality impacts, is the environmentally superior alternative. For the nonaviation revenue support land uses. Alternative LU-2 is the environmentally superior alternative because it substantially reduces the loss of agricultural soils, trip generation, and local air quality emissions while retaining substantially the benefits of proposed recreation uses and public facilities.

# 8.13 COMPARISON OF THE ENVIRONMENTAL IMPACTS OF THE PROJECT ALTERNATIVES

A comparison of the key impacts of each of the alternatives analyzed in this EIR is provided in Table 8.13-1.

Table 8.13-1
Comparison of Key Impacts of Alternatives Relative to the Proposed Project

	Project	<b>\</b>	٥	<b>D</b>	<b>U</b>	1	Land Use 1	Land Use 2	No Project No Activity	Wildlands Ranch	ETRPA Non-Aviation
Land Use	I	****		S+	S+	S+		-	S+	S+	=
G.P. Consistency	I	==	==	S+	S+	S+	<del> </del>	==	S+	S+	=
Transportation	I	-	=	S+	S+	=	_	-	_	=	S+
Noise	S	S-	<b> </b>	S-	S+	S+	S-	==	+	S+	S+
Air Quality, Local	<u>[**</u>	-		S+	S+	::::	-	_	-	*	S+
Air Quality, Regional	I	S+	*****	S+	S+			=	S+	S+	S+
Air Quality, Toxics	S	S-	=	S	S	=	=	=	**	S-	·*
Topography	I	===	=		S+	=	=	_	=	=	
Soils, Geology	I	=	=	==	=	=	=	===	=		****
Hydrology	I	=	=	===	_	=		=	S+	===	****
Biology	I	==	:==	+	+	=	+	=	+	==	+
Public Services	I	3115	***	=	=	***	122	=	+	===	=
Nat. Resources/Energy	S	S-	=	S+	S+	=	-	-	***	S+	S+
Aesthetics	1	and the same of th	=	-	S+	=	-	-	S	=	
Cultural Resources	I	=	=	***	=	=	=	=	=	=	
Recreation	S	==	=	-	S	=	+	+	S	3000	-
Health, Safety	I	=		=	=	****	=	=	-	===	=
Hazardous Waste	Ī			=	=	=	==	=	S	==	S
Socioeconomics	S	-	=	+	+	=	=	==	+	=	S
Risk of Upset	I	=	=	+	+	=	=	==	-	=	*
Cumulative	S	==	=	+	+	=	==	=	+	=	+

<sup>\*</sup> In those instances in which the comparison of the alternative to the Proposed Project is materially affected by the phasing, a footnote has been added to identify those differences.

# \*\* Significant local air quality impacts will not occur until sometime after 2010. These impacts will be mitigated to a level below the level of significance.

Legend: I = Impacts are insignificant after mitigation.

S = Impacts are significant after mitigation.

+ = Impacts are substantially greater than the Proposed Project.

= Impacts are substantially less than the Proposed Project.

= Impacts are similar to the Proposed Project

NA = Not Applicable.

#### 8.14 ALTERNATIVES CONSIDERED BUT REJECTED

This section discusses alternatives to the Proposed Project that have been considered but rejected from further consideration. The alternatives that have not been carried forward for further consideration were eliminated, generally, if screening analyses indicated that the scenario:

- (i) had a serious operational or environmental deficiency and/or was clearly inferior to one or more scenarios retained for further study;
- (ii) was infeasible or would have failed to meet significant and substantial goals and criteria, as established in the Community Reuse Plan and further refined for the ASMP;
- (iii) did not offer significant advantages over another scenario retained for further study, or;
- (iv) was closely bracketed by two other scenarios that were carried forward for further consideration, or was a closely-related variant of a scenario that had been carried forward.

The following subsections describe these alternatives and the specific reasons for eliminating them from further consideration in this EIR.

# 8.14.1 One-Airport Scenarios Not Carried Forward

# 8.14.1.1 JWA Only - Status Quo with Runway Extension

Under this alternative, the role of JWA would remain unchanged, but the main runway would be extended 1,100 feet to the north for a total runway length of 6,800 feet. The northerly extension would allow some commercial aircraft operating at JWA to have greater takeoff weights, enabling them to travel farther and/or carry more passengers. It would also provide an added margin of safety. In order to accommodate a significant amount of commercial passenger demand beyond 8.4 MAP, general aviation would be eliminated at JWA and the short runway closed. This alternative was eliminated from further consideration, because it was concluded that this scenario offered no significant advantage over Alternative F, while having the same failures to achieve project objectives as set forth in Alternative F in Section 8.4.3.

### 8.14.1.2 JWA Only – Enhanced Service

Under this alternative, the role of JWA would include additional long haul service. The improvements needed, the elimination of general aviation, and the conclusions for this alternative are the same as for Alternative 8.14.1.1, above.

# 8.14.1.3 JWA Only – Enhanced Service with Reduced General Aviation

Under this alternative, the role of JWA would be expanded to include full long-haul service. The main runway would be extended 1,100 feet to the north for a total runway length of 6,800 feet. This alternative was eliminated from further consideration, because it was concluded that this scenario offered no significant advantage over Alternative G, while failing to attain the same project objectives as discussed in Section 8.5.3.

# 8.14.1.4 MCAS El Toro Only

Ten alternatives were evaluated which involved closing JWA to all aircraft operations and developing MCAS El Toro to varying degrees of passenger and other service. None of the single-airport scenarios that would close JWA has been considered further, because the general aviation facilities at JWA would become unavailable to Orange County general aviation users or would need to be replaced at MCAS El Toro. If all JWA general aviation activity were relocated to MCAS El Toro, it would significantly affect the ability of MCAS El Toro to accommodate growth in commercial air passenger and cargo needs. The project objectives to follow the County's General Plan and board direction to utilize a two airport system would also not be attained.

# 8.14.1.5 New Airport Site Only

Under this alternative, a new airport site would have been developed to serve Orange County. JWA would be closed to all aircraft operations, and MCAS El Toro would have been reused for nonaviation purposes. Three potential new airport sites were identified for this analysis based on prior airport site selection studies in Orange County (AFRC Los Alamitos and a Cristianitos Canyon site in Orange County, and Camp Pendleton in San Diego County). No determination has been made here of the suitability of any of the three sites for a commercial airport with full domestic to full international service (see Section 8.12).

This scenario has not been carried forward because a guideline established by the Board of Supervisors in adopting the CRP was that a system of two airports, including commercial service at JWA, is favored. Alternative K consists of a two-airport system, with commercial service at JWA and a new commercial airport site serving Orange County. Furthermore, there is considerable uncertainty and speculation regarding the feasibility of both of the Orange County sites suggested. With the Camp Pendleton site about ten miles from the

Orange County border, it would not be a suitable replacement for JWA. The County has no access to the ownership of the Pendleton and Los Alamitos sites, and they remain under military ownership and control. There are no plans pending to close either of those bases. As set forth in Section 11.3.3.3 on page 11-56 of EIR No. 563, the Cristianitos Canyon site was rejected because it would be only a medium haul facility (maximum runway length 6,800 feet), is within the ten mile Emergency Planning Zone for the San Onofre Nuclear Power Plant, and it would result in greater environmental impacts than the Community Reuse Plan. See also the discussion in Section 8.11, Appendix J to this EIR, and the studies referenced in Appendix J.

# 8.14.2 Unlinked Two-Airport Scenarios Not Carried Forward

#### 8.14.2.1 Alternative D

Under this alternative, MCAS El Toro would be developed to provide full domestic and international passenger and cargo service, and general aviation service, as in the Proposed Project. However, JWA would serve only general aviation. No major facility improvements would be needed at JWA. Runway improvements at MCAS El Toro would be the same as the Proposed Project and Alternative C. Although this alternative was the closest refined alternative to CRP Alternative A, this alternative was rejected because it does not meet the project objective of a two airport system, and because it causes greater environmental impacts than the Proposed Project.

# 8.14.2.2 Alternative H: JWA - Status Quo; MCAS El Toro - Full Domestic with 10 MAP Limit

The airport roles and airfield improvements for this alternative would be the same as for Alternative A, except that MCAS El Toro would be constrained to 10 MAP. The limited service at OCX would result in twice as many passengers being served at JWA in 2020 compared with the Proposed Project (10.8 MAP compared with 5.4 MAP). This alternative has been rejected because it does not meet the project objectives. Less than two-thirds of the County's air passenger service demand would be served by Alternative H. The regional air quality benefits of serving nearly all the demand in Orange County would be lost. Noise impacts around JWA would increase. Figure 8-9 depicts Alternative H.

# 8.14.2.3 Alternative I: JWA – Status Quo; MCAS El Toro - Full Domestic with 15 MAP Limit

#### Aviation Uses

The airport roles and airfield improvements for this alternative would be the same as for Alternative A, except that MCAS El Toro would be constrained to 15 MAP. Because of this

limitation, JWA would serve slightly more passengers in 2020 than under the Proposed Project, 7 MAP compared with 5.4 MAP. JWA currently (1998) serves approximately 7.5 MAP. This alternative has been rejected because it does not meet the project objectives. Alternative I would serve less than two-thirds of the County's aviation passenger demand, will reduce the potential economic benefits of the project, would not take full advantage of the noise buffer around El Toro, and would not reduce regional air quality impacts to the extent feasible under the Proposed Project.

## 8.14.2.4 JWA -- Status Quo/MCAS El Toro -- General Aviation and Cargo

This scenario was the Community Reuse Plan (CRP) Alternative B. JWA would have retained its existing role, while MCAS El Toro would have been devoted to general aviation and cargo use only. The airfield at MCAS El Toro would consist of two closely-spaced parallel runways, the existing Runway 16R-34L and a new 4,200-foot runway with a runway centerline separation of 700 feet. The CRP and associated Environmental Impact Report No. 563 concluded that the CRP Alternative A was superior to the CRP Alternative B. For the reasons stated in EIR No. 563, this scenario was not carried forward for further study.

#### 8.14.2.5 JWA -- Status Quo/MCAS El Toro -- Limited International

Under this alternative, JWA would have continued its existing role. MCAS El Toro would have provided limited international service, including limited service to overseas destinations in addition to service to North and Central America. The runway configuration at JWA would have been unchanged. MCAS El Toro would have had the same runway improvements as Alternative A. The role of MCAS El Toro in this scenario would have been between that of Alternatives A (MCAS El Toro full domestic, including international to North and Central America) and B (MCAS El Toro full international). The effects of this alternative are adequately tested by Alternatives A and B, so it was not be carried forward for further analysis.

## 8.14.2.6 JWA (North Flow) -- Status Quo/MCAS El Toro -- Full International with Wide Parallel Runways

This alternative is similar to Alternative J, except JWA would have operated in predominantly north-flow (aircraft landing and departing to the north), rather than the existing south-flow operations. North flow operations at JWA have been examined in the airspace analysis of Alternative J. Therefore, further study under a separate scenario was not needed.

## 8.14.2.7 JWA -- Status Quo/MCAS El Toro -- Full International with Realigned Runways 13/31

Under this alternative, JWA would have operated as status quo, and one or more of the runways at MCAS El Toro would have been realigned to a northwest-southeast (Runway 13/31) direction. The purpose of the realignment would have been to minimize the presence of high terrain in the approach and departure paths. This scenario would be similar to Alternative B, except for the MCAS El Toro runway configuration. The realignment of runways would not take full advantage of the non-residential areas within the County Policy Implementation Line (PIL) and, in fact, exposes new noise sensitive areas to aircraft noise impacts. For this reason, a separate alternative with realigned runways was not carried forward.

## 8.14.2.8 JWA -- Expanded Role/MCAS El Toro -- General Aviation and Cargo

Two alternatives would have expanded passenger service roles for JWA (ranging from full domestic to limited international), with only general aviation and cargo at MCAS El Toro. Runways at JWA would have been extended to 6,800 to 8,000 feet, while MCAS El Toro would have had the same runway two-runway configuration discussed previously. In a two-airport system with MCAS El Toro, long-haul and international service is more suited to MCAS El Toro due to its longer runways and ample space for the necessary terminal facilities. Runways of 6,800 to 8,000 feet would not efficiently provide full domestic and limited international service, respectively at JWA. The project objectives to serve Orange County's aviation demand and take advantage of economic and land use opportunities prescribed by the availability of El Toro would not be met. Thus, these scenarios were not considered further.

## 8.14.2.9 JWA -- Expanded Cargo or Passenger Roles/MCAS El Toro -- Short- and Medium-Haul

Three alternatives would have expanded the role of JWA (either all-cargo service or full domestic or limited international passenger service), while MCAS El Toro would have assumed JWA's current role. These alternatives all would have forced JWA to assume a role that requires longer runways and more space than presently exists at the airport. Although, the main runway at JWA would have been extended from 6,800 to 8,000 feet, it would not have adequately accommodated the roles envisioned in these alternatives. On the other hand, the long runways at MCAS El Toro would have been underutilized by the short- and medium-haul role there. Alternatives with JWA fulfilling the primarily short- and medium-haul role and MCAS El Toro serving longer flights (such as Alternatives A, B, H, I and J) provide a better balance considering existing facilities and available space at the two airports. The studied alternatives take better advantage of the economic and land use

opportunities with El Toro to meet the project objectives to serve Orange County's aviation demand. Therefore, these three scenarios were not carried forward for further study.

## 8.14.2.10 JWA -- General Aviation and Cargo/MCAS El Toro -- Long-Haul or Limited International

Two alternatives envisioned JWA with general aviation and all-cargo service, and MCAS El Toro with long-haul to limited international passenger service. The main runway at JWA would have been extended to 6,800 feet. MCAS El Toro would have had intersecting pairs of close parallel runways. The 6,800-foot runway length at JWA would not have adequately accommodated expanded all-cargo service. Furthermore, there are no facilities and little space to support an all-cargo role there. The longer runways at MCAS El Toro make that airport more suited to the all-cargo role. Thus, this alternative fails to meet the project objectives to take advantage of economic and land use opportunities at El Toro to meet Orange County's aviation demand.

## 8.14.2.11 JWA -- General Aviation and Short-Haul/MCAS El Toro -- Limited or Full International

Here, JWA would have had a general aviation and short-haul role, while MCAS El Toro would have provided limited to full international passenger service. The runway configuration at JWA would have remained unchanged. MCAS El Toro would have had intersecting pairs of close parallel runways. These alternatives are very similar to Alternative B, the primary difference being medium-haul service at JWA in Alternative B. Demand forecasts have shown that with short- and medium haul service at JWA it would attract fewer passengers with Alternatives A and B than served today. Therefore, it is more appropriate to consider medium-haul service at JWA, as in Alternatives A and B. Moreover, Alternative C tests the general aviation and short-haul role at JWA. For these reasons, these two scenarios were not carried forward.

#### 8.14.2.12 JWA -- General Aviation/MCAS El Toro -- Full Domestic

In this scenario, JWA would have served only general aviation, while MCAS El Toro would have had a full domestic role. This scenario would be similar to Alternative D, except MCAS El Toro would not provide international service. It is concluded that Alternative D will adequately test the limits of effects under this scenario, and therefore this scenario was eliminated from further consideration.

## 8.14.2.13 JWA -- General Aviation/MCAS El Toro -- Full International

This alternative is CRP Alternative A: general aviation at JWA and all-cargo and full international passenger service at MCAS El Toro. Under this alternative, JWA would have retained its existing runway configuration. MCAS El Toro would have had the same airfield configuration as Alternative A. The more refined and detailed Alternative D would have had the same roles as this alternative, but would have had extended runways at MCAS El Toro (within the existing MCAS El Toro property). The longer runways at MCAS El Toro would have provided the greater takeoff length needed for intercontinental flights. This scenario does not appear to have a significant advantage over Alternative D, and thus was not carried forward for further study in its original form. For the reasons stated in Section 8.14.2.1, the CRP has evolved in its more refined form, and with Board direction for a two airport system, into the Proposed Project, Alternative B.

## 8.14.2.14 General Aviation/MCAS El Toro -- Full International with Parallel Runways Separated by 2,500 Feet

This alternative would have been similar to Alternative J, but with a narrower runway separation to reduce impacts to SR133. The 3,000-foot separation of Alternative J could potentially permit simultaneous landings or departures on the two runways during IFR conditions. While a separation of 2,500 feet allows a simultaneous departure and arrival, it does not allow simultaneous arrivals or departures. Furthermore, the area between the runways for terminal development would be smaller than desirable for terminal development. This scenario offers no significant advantage over Alternative J, and was not carried forward for further consideration.

## 8.14.3 Linked Two-Airport Scenarios Not Carried Forward

### 8.14.3.1 JWA - Short- to Medium-Haul/MCAS El Toro - Long-Haul to Full International

Alternative C examines a linked two-airport system with short-haul service at JWA and medium-haul to full international service at MCAS El Toro. This alternative is a variation of that concept, whereby JWA would assume a greater role in the split of activity. Demand forecast studies have shown that JWA would serve 10.1 MAP by 2020 with short-haul service only (Alternative C). Because of the lack of terminal capacity beyond the 10 MAP range at JWA, and because the longer runways at MCAS El Toro would be more suited to the longer-haul operations, this scenario would offer no significant advantage over Alternative C, in the context of project objectives to take advantage of economic and land use opportunities at El Toro to serve Orange County's aviation demand. Further, the expense of the people mover system connecting the two airports is infeasible.

### 8.14.3.2 JWA - Full Domestic/MCAS El Toro - International Only

The reasons for not carrying forward this alternative are similar to those for the preceding alternative. In this case, however, the full domestic role is significantly less suitable for JWA than the short-haul role of Alternative C, because of the limited capability to extend the main runway within airport property. Furthermore, the split of activity between the two airports in this alternative would place the majority of passengers at JWA, which does not have the space on-airport to accommodate it. The scenario would also require a very large number of passengers to connect between the two airports on the people mover system, the cost of which is infeasible.

# 9.0 INVENTORY OF MITIGATION MEASURES AND STANDARD CONDITIONS

#### 9.1 INTRODUCTION

Public Resources Code (PRC) Section 21081.6 requires state and local agencies to adopt programs for the monitoring and reporting of the implementation of mitigation measures addressing the significant adverse environmental impacts of projects agencies approve subject to the California Environmental Quality Act (CEQA). As part of this Final Environmental Impact Report (EIR), the County Board of Supervisors will adopt a Mitigation Monitoring Program (MMP), consistent with the requirements of PRC Section 21081.6 and the State CEQA Guidelines. The MMP will be used by the County of Orange for monitoring the implementation of the mitigation measures applicable to the selected reuse alternative.

The mitigation measures identified earlier in Sections 4.1 to 4.18 are listed in this section, by environmental parameter. These measures may be refined or deleted, or new measures added, by the Board of Supervisors in the Final EIR and the MMP.

## 9.1.1 Standard Conditions of Approval

The Proposed Project includes a number of County of Orange Standard Conditions of Approval (COAs) (see listing at end of chapter). These COAs, listed in Appendix L, serve to implement various County regulations and policies, and to provide protection to the environment. These COAs are part of the Proposed Project and, as a result, are themselves project features which can avoid or substantially reduce potential adverse impacts of the Proposed Project on the environment. The County will ensure that each COA incorporated in the Proposed Project is properly implemented as an integral part of the land use review compliance process.

### 9.2 MITIGATION RELATED TO LAND USE

### 9.2.1 Final EIR No. 563 Mitigation Measures

The findings for EIR No. 563 included three land use mitigation measures concerning subsequent amendments to the Orange County General Plan and the need for surrounding cities to consider adopting proposed mitigation. These mitigation measures are as follows:

LU-1 Subsequent to the adoption of the MCAS El Toro Community Reuse Plan and concurrent with the preparation of an airport master plan or other development plans for the MCAS El Toro site, the County of Orange shall prepare amendments to the Orange County General Plan for consideration by the Board of Supervisors to ensure consistency between the adopted Community Reuse Plan and the Orange County General Plan. These amendments would be subject to the discretion of the Board of Supervisors. Further, the Board of Supervisors shall consider zone changes to allow the variety of uses

under the Reuse Plan. These zone changes will be considered at subsequent stages of the planning process, when appropriate, such as at the Master Plan level.

Relationship to EIR No. 573. General Plan amendments are proposed as part of the project. No zone changes are required for the Proposed Project due to the County's exemption from its own zoning, enacted after EIR No. 563. Thus, LU-1 is still applicable and will be implemented upon approval of the Proposed Project.

The County of Orange does not have land use authority in adjacent jurisdictions; therefore, the County included Measures LU-2 and LU-3 for consideration by the adjacent jurisdictions.

- LU-2 The County of Orange shall coordinate any General Plan Amendments, when deemed necessary, with adjacent jurisdictions and adjacent landowners. The person responsible for this coordination shall be the County of Orange Director of Planning and Development Services or his designee.
- LU-3 Surrounding cities should also consider specific impacts resulting from the proposed project, and adopt proposed mitigation, as feasible, in accordance with their local land use jurisdiction.

Relationship to EIR No. 573. Mitigation Measures LU-2 and LU-3 are within the responsibility and jurisdiction of another public agency, and not the County of Orange. See CEQA Guideline Section 15091(a)(2). Mitigation Measures LU-2 and LU-3 remain applicable to the Proposed Project.

## 9.2.2 Additional Precautionary Mitigation Measures

- LU-4 Prior to the execution of leases, the County will review compatibility with adjacent and nearby uses and include requirements that minimize off-site impacts. These additional requirements will include compliance with all hazardous waste disposal protocols, the provision of a solid perimeter screen wall or fence, hoods for light fixtures to direct rays on site, and other site improvements or measures determined to be necessary to protect nearby property owners and lessees.
- LU-5 Prior to execution of leases for agricultural uses, the County will review compatibility with adjacent nearby uses, and include the following requirements where appropriate:
  - Restricting the agriculture use to row crops;
  - Prohibiting livestock;
  - Requiring low volume watering, as applicable;
  - Containing runoff on site or directing it to adequately sized public storm sewers;
  - Prohibiting aerial spraying of pesticides;
  - Requiring the application of integrated pest management practices;

- Requiring watering before and after any activities that disturb soil to reduce dust emissions:
- Conducting soil moisture monitoring;
- Maintaining farm equipment in good operating condition;
- Requiring the use of mulch, cover crops, crop residue management, surface roughening, minimum tillage, or other methods to control dust when the land is fallow;
- Requiring the use of gravel or other surface material on highly traveled farm roads to reduce dust emissions; and
- Restricting vehicle speeds and public access to unpaved farm roads.
- LU-6 The County will use its best efforts to ensure that nearby cities adopt and/or maintain existing restrictions on undesirable land uses or developments.

## 9.3 MITIGATION RELATED TO GENERAL PLAN CONSISTENCY

## 9.3.1 Final EIR No. 563 Mitigation Measures

LU-1 Subsequent to the adoption of the MCAS El Toro Community Reuse Plan and concurrent with the preparation of an airport master plan or other development plans for the MCAS El Toro site, the County of Orange shall prepare amendments to the Orange County General Plan for consideration by the Board of Supervisors to ensure consistency between the adopted Community Reuse Plan and the Orange County General Plan. These amendments would be subject to the discretion of the Board of Supervisors. Further, the Board of Supervisors shall consider zone changes to allow the variety of uses under the Reuse Plan. These zone changes will be considered at subsequent stages of the planning process, when appropriate, such as at the Master Plan level.

Relationship to EIR No. 573. The Proposed Project includes the General Plan Amendment. There is no longer a need for a zone change due to the County's exemption from its Zoning Code. Thus, LU-1 is still applicable and will be implemented upon approval of the Proposed Project.

- LU-2 The County of Orange shall coordinate any General Plan Amendments, when deemed necessary, with adjacent jurisdictions and adjacent landowners. The person responsible for this coordination shall be the County of Orange Director of Planning and Development Services or his designee.
- LU-3 Surrounding cities should also consider specific impacts resulting from the Proposed Project, and adopt proposed mitigation, as feasible, in accordance with their local land use jurisdiction.

Relationship to EIR No. 573. Mitigation Measures LU-2 and LU-3 are within the responsibility and jurisdiction of another public agency, and not the County of Orange. See CEQA Guidelines Section 15091(a)(2). Mitigation Measures LU-2 and LU-3 remain applicable to the Proposed Project.

## 9.3.2 Additional Mitigation Measures<sup>1</sup>

- GPC-1 The County of Orange will submit the Board of Supervisors approved Airport System Master Plan (ASMP) to the Airport Land Use Commission (ALUC) within 30 days of Board approval of the ASMP. The County will use its best efforts to obtain ALUC approval of the ASMP as the 20 year plan for both airports, and ALUC adoption of necessary amendments to the AELUP.
- GPC-2 Prior to any landowner/developer application for a proposed zone change within the Proposed Project's 65 dB CNEL contour, the County of Orange shall negotiate with affected landowners to reallocate land uses, densities/intensities, development standards, and development allocations/forecasts to be compatible with the 65 dB CNEL contours.
- GPC-3 The City of Irvine can and should amend its General Plan to reflect the Proposed El Toro Project, including the General Plan designations for the portion of the site in the City of Irvine but under the jurisdiction of the LRA (Proposed Project Planning Area 7), and changes to the General Plan designations in the City's Sphere of Influence where inconsistent with the Proposed Project CNEL contours.
- GPC-4 Subsequent to adoption of the ASMP, the County shall use its best efforts to obtain SCAQMD's adoption of an updated AQMP, which replaces the obsolete military emissions budget for El Toro with an emissions budget based on SCAG's 2001 RTP as it relates to the Proposed Project.

Note: Pursuant to CEQA Guidelines Section 15091 (a)(2), this these measures are is within the responsibility and jurisdiction of another public agency, and not the County of Orange.

## 9.4 MITIGATION RELATED TO TRANSPORTATION AND CIRCULATION

# 9.4.1 Final EIR No. 563 and EIR No. 563 Supplemental Analysis Mitigation Measures

T-1 Before the County Board of Supervisors adopts an amendment to the Land Use Element or Transportation Element of the General Plan related to implementation of the Community Reuse Plan, or approves an airport master plan for an airport on the MCAS

,

Land use compatibility mitigation measures in Section 4.1 of this EIR are designed with "LU;" therefore, the General Plan Consistency mitigation measures are designated "GPC" in this EIR.

El Toro site, whichever comes first, the specific levels of capacity augmentation that would be applicable to the study area circulation system and other infrastructure will be identified by the Director of the Public Facilities and Resources Department in consultation with other responsible agencies for the selected reuse alternative, based on the best available socioeconomic data (i.e., OCP-96).

Relationship to DEIR No. 573: The methodology and findings of this DEIR No. 573 transportation and circulation study fulfill the requirements of Mitigation Measure T-1. Therefore, Mitigation Measure T-1 is no longer applicable.

- T-2 Before the County Board of Supervisors adopts an amendment to the Land Use Element or Transportation Element of the General Plan, related to implementation of the Community Reuse Plan, or approves an airport master plan for an airport on the MCAS El Toro site, whichever comes first, the Director of the Public Facilities and Resources Department will prepare an Area Traffic Improvement Action Plan ("Action Plan"). The Action Plan will contain the following:
  - i) A defined land use and circulation phasing plan;
  - ii) An analysis of the land use phasing plan demonstrating that, in compliance with the County's GMP and CMP, circulation improvements are implemented commensurate with development milestones;
  - iii) A transit impacts analysis;
  - iv) A discussion which identifies and assigns the project's obligation (full share or fair share) to fund road phasing improvements and development milestones, thus assuring that GMP and CMP LOS standards are met;
  - v) An updated traffic study, prepared in conjunction with the Transportation Corridor Agencies (TCA), the California Department of Transportation (Caltrans), the Orange County Transit Authority (OCTA) and affected cities and major land owners, which uses the land use and circulation system capacity assumptions contained in the County General Plan (and, where applicable, the assumptions contained in the General Plans of adjacent jurisdictions);

The traffic study will be conducted pursuant to the provisions of the County's Growth Management Plan (GMP) and Congestion Management Plan (CMP), and will include both an ADT and a peak hour analysis of the proposed project;

The traffic study will also be governed by the County's Traffic Implementation Manual (TIM), including the General Plan requirement that, within the time frames established by the County's "traffic level of service policy," the necessary improvements to arterial highways, to which the project contributes measurable traffic, are constructed and completed to attain the Level of Service (LOS) D, as provided in the TIM. Intersections exempt from the County's "traffic level of service policy" include facilities under the jurisdiction of a city or the State or those included on the County's Deficient Intersection

List established pursuant to the Growth Management Element. However, where the Community Reuse Plan contributes measurable traffic to intersections on the Deficient Intersection List, the project will contribute on a pro-rata basis to the Deficient Intersection Fund;

- vi) A mitigation feasibility analysis and cost estimate. This analysis shall identify potential constraints in implementing any recommended traffic mitigation measures for roadways and intersections, so that the feasibility of implementing a particular traffic-related mitigation measure is assessed in conjunction with the feasibility analysis to be undertaken at the project-specific implementation phase of the Community Reuse Plan; and
- vii) A mitigation implementation plan.

Relationship to DEIR No. 573: The methodology and findings of this DEIR No. 573 transportation and circulation study and the ASMP and its supporting technical reports fulfill each of the individual requirements specified in Mitigation Measure T-2 for the preparation of an Area Traffic Improvement Action Plan. Therefore, Mitigation Measure T-2 is no longer applicable.

Prior to issuing any building permits, the Director of the Public Facilities and Resources
Planning and Development Services Department will—shall prepare a Transportation
Demand Management (TDM) Plan pursuant to the County Transportation Demand
Management Ordinance. The TDM Plan will address such things as vanpooling, fleet
pooling, ridesharing, public transit, alternative work hours, bikeways and other measures
related to the mitigation of traffic through demand management. The TDM Plan also will
establish a program to implement the following air quality mitigation measures which are
components of the TDM Plan: AQ-15i, AQ-23, AQ-37, AQ-41, and AQ-42.
Annual monitoring reports will be required. These reports will assess the implementation
status of the required TDM Plan features. In addition, a transportation coordinator will
be provided to support the TDM Plan.

Relationship to DEIR No. 573: Mitigation Measure T-3 remains applicable to the Proposed Project.

T-4 For highway facilities requiring mitigation for the second-tier Community Reuse Plan process, but that are not under the sole control of the County, mitigation will be coordinated with the affected jurisdictions consistent with Measure M countywide GMP, which requires the County to participate in the Inter-Jurisdictional Planning Forums (IJPFs) at the Growth Management Area (GMA) level to examine regional improvements needed within the affected GMA. Each IJPF develops an annual prioritized list of transportation improvement projects, which is approved by the IJPF's elected officials and submitted to the Orange County Transportation Authority for funding consideration. The County will participate in development of the annual GMA Transportation Improvement Project List for each GMA affected by the project.

For intersections and highway links not under the sole control of the County to which the Project contributes measurable traffic, the County will identify improvements which will mitigate the Project impacts as governed by the applicable standards of the jurisdiction in which the intersection or highway is located, and the County will be responsible for the Project fair share of the mitigation costs. If a mitigation measure is unacceptable to the applicable jurisdiction, and the County or the jurisdiction identifies an alternative mitigation measure which is feasible and acceptable to the jurisdiction and the County, the County will be responsible for the Project fair share of the cost of the alternative improvement; or at the election of the jurisdiction, the County will provide funds to the jurisdiction equal to the fair share mitigation costs to permit the jurisdiction to make alternative improvements acceptable to the jurisdiction.

Relationship to DEIR No. 573: Mitigation Measure T-4 remains applicable to the Proposed Project. A modified version that refers specifically to improvement locations which are not under the sole control of the County is therefore included in the additional transportation and circulation mitigation measures summarized at the end of this section.

T-5 Before the County Board of Supervisors adopts an amendment to the Land Use Element or Transportation Element of the General Plan related to implementation of the Community Reuse Plan, or approves an Airport Master Plan for an airport on the MCAS El Toro site, the Director of the Public Facilities and Resources Department will prepare a Comprehensive Phasing Plan (CPP) consistent with the County's Growth Management Element. With regard to road improvements, the CPP will identify road and infrastructure improvements and a financing plan, and include level of service requirements and take into account measurable traffic impacts on the circulation system within the MCAS El Toro Community Reuse Plan Study Area.

Relationship to DEIR No. 573: Mitigation Measure T-5 remains applicable to the Proposed Project. A modified version that refers to the development of a CPP based on specific project related improvements is therefore included in the additional transportation and circulation mitigation measures summarized at the end of this section.

T-6 In conjunction with Mitigation Measure T-5, the adopted CPP shall include a Development Phasing Plan, consistent with the County's Growth Management Element, to ensure that necessary road improvements or fair share funding are conditions of relevant phases of project development.

Relationship to DEIR No. 573: Mitigation Measure T-6 remains applicable to the Proposed Project. A modified version that refers to the development of a CPP based on the phasing of specific project related improvements is therefore included in the transportation and circulation mitigation measures summarized at the end of this section.

T-7 Within one year after the approval of an amendment to the Land Use Element or Transportation Element of the General Plan related to implementation of the Community Reuse Plan, or approval of an Airport Master Plan for an airport on the MCAS El Toro site, annual monitoring reports will be required to be prepared by the Director of Planning and Development Services for the project pursuant to the County's Performance Monitoring Program contained in the Growth Management Element of the General Plan.

Relationship to DEIR No. 573: Mitigation Measure T-7 remains applicable to the Proposed Project

### 9.4.2 DEIR No. 573 Additional Mitigation Measures

- T-8 Prior to issuance of building permits for the initial phase of the ASMP, the County will prepare a Comprehensive Phasing Plan (CPP) consistent with the County's Growth Management Element. The adopted CPP shall include a Development Phasing Plan. With regard to road improvements, the CPP will identify road and infrastructure improvements and a financing plan based on the circulation improvements listed in Tables 4.3-19 and 4.3-20 and their associated implementation dates and project funding obligations, thereby ensuring that necessary road improvements or fair share funding are conditions of relevant phases of project development. All improvements listed in Tables 4.3-19 and 4.3-20 that are within the sole control of the County and are to be fully funded during project development must be constructed prior to or concurrent with issuance of occupancy permits for the project development phase that requires the improvements.
- T-9 Highway improvements listed in Tables 4.3-19 and 4.3-20 that are not under the sole control of the County are subject to the improvement identification and implementation procedures described in Mitigation Measure T-4. Prior to issuing any building permits for the initial phase of the ASMP, the County will enter into a cooperative agreement with the affected local jurisdictions and landowners outlining a course of action to participate on a fair share basis in the implementation of the planned circulation improvements listed in Table 4.3-20. Such an agreement will include conditions ensuring that the improvements listed in Table 4.3-20 would be constructed to the standards of the affected local jurisdictions. If in the event a cooperative agreement cannot be reached with local jurisdictions and adjacent landowners, County will request that OCTA mediate any outstanding issues and adopt a course of action to ensure implementation and funding mechanisms for the Master Plan of Arterial Highways. The County does commit to participate in an implementation and funding program based on the circulation improvements and project fair share percentages identified in Table 4.3-20. implementation and funding program will be developed in concert with local jurisdictions and affected landowners, with participation by OCTA.
- T-10 Prior to final engineering design, for any project facility connecting to, or located within the right-of-way or property of Caltrans, the Transportation Corridor Agencies, and/or local jurisdictions, the County will enter into an agreement with affected agencies governing the design, operation and construction of the facility to ensure compatibility with the agency's facilities and operations. For project facilities connecting to toll roads,

the agreement with Caltrans and the Transportation Corridor Agencies shall address the project effect on toll revenues to ensure that revenues to the TCA are not reduced from their assumptions or forecasts effective at certification of EIR No. 573.

T-11 Prior to the final engineering design for ASMP facilities or approval of leases for non-ASMP uses, the County will ensure that parking and loading facilities are incorporated to meet or exceed the County Zoning Code parking regulations. The County will ensure through construction design or lease implementation that parking and loading facilities are provided and maintained per the County Zoning Code parking regulations.

Mitigation Measures T-12 and T-13, which are included in Section 5.4.3.2, address the cumulative transportation and circulation impacts of the project.

Prior to issuing any building permits or construction contracts (whichever occurs first) for each phase of the ASMP, the County shall make every reasonable effort to enter into an agreement with the California Department of Transportation (Caltrans) in order to coordinate each project phase with appropriate improvements identified in the Caltrans Traffic Operations Strategies (TOPS) program relative to freeway/tollway mainline impacts identified as a direct result of the proposed project. The TOPS program includes, as necessary, 1a) implementation of "intelligent infrastructure" improvements such as system-wide adaptive ramp metering, changeable message signs, traffic management centers, incident response systems, advanced traveler information systems, and real-time performance measurement systems; 1b) implementation of physical operational improvements such as the construction of freeway auxiliary lanes (merge lanes provided before and after on-ramps), the modification of ramp/city street access, and the addition of short passing lanes and truck climbing lanes; 2) construction of HOV drop ramps and freeway to freeway HOV connectors, as well as closing HOV gaps that exist within corridors; 3) construction of improvements to freeway-to-freeway interchanges to minimize traffic flow disruptions and resulting congestion.

The County commits to participate in the Caltrans TOPS program, including funding to the extent required by law, based on the project fair share percentages identified in Table 4.3-20. Pursuant to CEQA Guidelines Section 15091 (a)(2), implementation of the TOPS program is within the responsibility and jurisdiction of another public agency and not the County of Orange.

T-15 Prior to issuance of building permits for the initial phase of the ASMP, the County shall incorporate the project site into the Irvine Spectrum Transportation Management Association (TMA) ("Spectrumotion") in accordance with Article X of the recorded Declaration of Covenants, Conditions and Restrictions (CC&Rs) for the Irvine Spectrum TMA, including any supplementary and amended CC&Rs. The primary goal of the TMA is to reduce morning (6:00-10:00 AM) and evening (3:00-6:00 PM) peak period vehicular traffic within the Spectrum Area. work with the Irvine Spectrum Transportation Management Association and make every reasonable effort to formulate and implement a joint plan, consistent with Mitigation Measure T-3, to reduce morning (6:00 - 10:00 AM) and evening (3:00 - 6:00 PM) peak period vehicular traffic within the Spectrum Area.

- T-16 The development of each phase of the ASMP shall be subject to the trip generation restrictions identified in Tables 4.3-8, 4.3-12, 4.3-13, 4.3-15, and 4.3-17. Prior to issuance of building permits for each phase of the project, the County shall prepare a trip monitoring report demonstrating that the project's cumulative level of daily, peak hour and truck traffic is within the trip generation restrictions. A comprehensive traffic analysis shall be conducted prior to approval of any project development phase generating trips in excess of the trip generation limits listed in Tables 4.3-8, 4.3-12, 4.3-13, 4.3-15, and 4.3-17. This traffic analysis shall include identification of appropriate transportation mitigation measures for the increased project trip generation in accordance with County and applicable jurisdictional guidelines.
- T-17 Prior to the commencement of construction of the off-site highway improvements listed in Table 4.3-20, the County shall prepare a Construction Management Plan. The Construction Management Plan shall set forth appropriate construction practices necessary to minimize the potential disruption to properties, pedestrians, bicyclists, and motorists during construction of off-site roadway improvements. The following items shall be included in the Plan:
  - Public Notice: Signs providing advance notice of work to be done on a particular segment shall be posted for a period of two weeks prior to construction. Notification in a local newspaper shall be published two weeks prior to construction. Adjacent property owners and public service providers and utility companies shall also be notified two weeks prior to construction.
  - Traffic Routing: Signs shall be provided to route vehicular and bicycle traffic through segments under construction. In addition, signs that suggest possible alternate routes shall be posted. Construction vehicle access to construction sites shall occur at off-peak traffic hours. Construction vehicle access routes shall be directed around residential areas. In addition, traffic control personnel shall be provided as necessary to mitigate traffic congestion and to mitigate the impact to arterial service levels during construction in accordance with local, State, and federal standards.
  - Construction Staging: Storage of construction vehicles, equipment, and materials shall not occur in the immediate vicinity of residential areas and retail establishments.
  - Hours of Construction: Hours of construction shall conform to established County policy unless otherwise approved by the County.
  - Access to Properties: Construction activities shall be arranged so that access to properties will be maintained.

Note: Pursuant to CEQA Guidelines Section 15091(a)(2), traffic mitigation measures may be within the responsibility and jurisdiction of another public agency and not the County of Orange.

### 9.5 MITIGATION RELATED TO NOISE

## 9.5.1 Final EIR No. 563 Mitigation Measures

- N-1 For Existing Residential and Other Existing Noise Sensitive Uses. CRP projected noise contours for the year 2020 show that there may be residential areas exposed to noise levels in excess of 65 dB CNEL. Other mitigation discussions in this report have described many noise abatement and noise mitigation measures that would reduce noise impacts associated with aviation operations associated with the CRP. The feasibility of implementing one or more of these measures will depend on numerous factors, not all of which are under the control of the County of Orange, and many of which cannot be known or predicted until the master plan level of planning occurs. Until a detailed master plan assessment of the need for and an assessment of the feasibility of implementing each alternative measure can be evaluated in terms of statutory authority and cost benefit, specific commitments to any noise control measure cannot be made at this time. Therefore, the detailed evaluation and selection of noise mitigation measures will be done as part of an Airport Master Plan environmental review process. However, the objective should be to identify operational measures which could reasonably reduce noise levels, in a manner which does not unreasonably or unnecessarily impair the utility of the facility as a commercial airport serving the Orange County community, preferably (and in some cases, necessarily) with the concurrence and cooperation of the airlines, other airport users, and the FAA (See the Airport Noise and Capacity Act of 1990 and Federal Aviation Regulation, Part 161.)
- N-2 For Undeveloped Land Offsite. The County of Orange has a long history of enforcing the 1981 AICUZ boundary as a Policy Implementation Line as part of its General Plan Noise Element. Given the uncertainties both in the evolving FAA position and guidelines on noise compatibility, the inherent uncertainties in the planning process until a master plan is prepared for a civilian airport, and possible future military activity on the site until the military completes the base closure process, the fairest and most prudent course of action would appear to be to maintain the current noise requirements of the County's General Plan. However, an appropriate noise element amendment should be part of any Airport master planning process undertaken to implement the selected reuse plan.
- N-3 For Onsite Noise Sensitive Uses. The present review and design criteria used by the County of Orange as reflected in the County of Orange standard conditions of approval are adequate for any onsite development. Reuse of residential land uses or construction of new residential uses shall be restricted until the base closure is complete and a civilian master plan process is complete.

- N-4 Measure for Traffic Related Noise. Offsite road noise associated with each of the reuse alternatives may result in noise impacts in the residential areas of Lake Forest. A detailed evaluation of road noise impacts in outdoor residential areas, and development of noise barrier designs or other programs which reduce the outdoor living areas to noise exposures of 65 dB CNEL or less, shall be linked to the reuse of the project site and used to mitigate these road noise impacts.
- N-5 The County will, as part of the airport master planning process, identify specific noise control and mitigation actions which feasibly and effectively can be implemented in connection with the commencement of any civil aviation operations at El Toro, with the objectives being to identify and implement an operational scenario for a civilian airport at El Toro which would result in CNEL contours which are no larger than, and if feasible, smaller than, the contours projected in Figure 4-55 of Final EIR No. 563 (See Figure 4.1). The County will also identify and incorporate a feasible, effective and legally enforceable mechanism to implement that operational scenario.

## 9.5.2 Proposed Mitigation Measures

- N-6 The County will participate in a fair share program to construct or extend existing sound walls at relevant portions of Trabuco Road between Yale and Jeffrey Road, and along Jeffrey Road just south of Irvine Boulevard, to reduce project caused traffic noise levels in any residential property to a level at or below 65 CNEL.
- N-7 For all project construction, the County will adhere to the County Noise Ordinance and, for off-site improvements in any incorporated areas, any existing applicable local jurisdiction noise ordinances regulating construction noise.
- N-8 The County will construct and implement a noise monitoring system at OCX prior to the opening for public use of the passenger terminal facilities contemplated under the Proposed Project to be completed by the end of Phase 1 of the Proposed Project (2005). This noise monitoring system will include remote microphone stations at locations substantially as shown in Figure 4.4-52 of EIR 573. In addition, upon completion and County acceptance of the noise monitoring system, the County will implement a noise complaint and administration program and staffing comparable to its existing program at JWA.
- N-9 Prior to the initiation of scheduled commercial operations at El Toro, the County will implement an 86 dB SENEL noise mitigation restriction for El Toro during the hours of 11:00 p.m. and 7:00 a.m. The noise level limit will be based upon quarterly energy average SENEL values for scheduled commercial operators by operator and aircraft type and will be enforced at the remote microphone stations installed and monitored pursuant to mitigation measure N-8.

### 9.6 MITIGATION RELATED TO AIR QUALITY

### 9.6.1 Final EIR No. 563 Mitigation Measures

AQ-1 Prior to adoption of a construction level plan for the MCAS El Toro site, the County of Orange will include mitigation measures related to construction based on the recommendations of the South Coast Air Quality Management District.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-11.

AQ-2 The County of Orange will identify, as necessary, the need for a Transportation Demand Management (TDM) program for the MCAS El Toro site, as part of the construction level environmental documentation for the project.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-12.

AQ-3 In order to maintain consistency with the 1994 AQMP, the County of Orange will ensure that construction level environmental documentation for the project incorporates the appropriate mitigation measures identified in the 1994 AQMP that are applicable to airports.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-13.

AQ-4 The County of Orange will conduct additional traffic studies concurrently with construction level environmental documentation, to determine how to improve the capacity and minimize congestion at the intersection of Bake Parkway and Trabuco Road. If additional intersections are found in these detailed traffic studies to be adversely impacted, additional analysis to improve those intersections will also be conducted.

Relationship to DEIR No. 573: The methodology and findings of this DEIR No. 573 transportation and circulation study fulfill the requirements of Mitigation Measure AQ-4. Therefore, Mitigation Measure AQ-4 is no longer applicable.

AQ-5 During design of the aviation uses on the site, the County of Orange will consider including electrical power outlets for landside passenger shuttles, including incorporation of electrical power outlets in the terminal, and parking lot design to accommodate electric shuttle vehicles. This would include providing electrical outlets for battery charging for passenger shuttles that serve hotels, rental car agencies, and other businesses.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-14.

AQ-6 During design of the aviation uses on the site, the County of Orange will consider encouraging the use of alternative fuel vehicles powered by natural gas, propane, and/or other alternative fuels, and providing fuel storage facilities for these alternative fuels.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-15.

AQ-7 During the preparation of construction level environmental documentation for the project, the County of Orange or the airport operator will require that plans and procedures be prepared that includes the requirement for final design studies to minimize taxi-in and taxi-out times and reduce aircraft queuing times. These may include, but are not limited to, design features and specific operations procedures.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-16.

AQ-8 During design of the aviation uses on the site, the County of Orange will consider including electrical power and preconditioned air in the design of the terminal gates (jetways), to reduce emissions from operating aircraft engines at the gates.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-17.

AQ-9 During design of the aviation uses on the site, the County of Orange will consider including electrical power outlets for electric ramp vehicles, and for battery charging for ground support equipment.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-18.

AQ-10 During design of the aviation uses on the site, the County of Orange will consider incorporating hydrant fueling systems for commercial jet aircraft operations.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AQ-19.

# 9.6.2 Mitigation Measures Updating Final EIR No. 563 Mitigation

#### Construction Emissions

#### AQ-11a: Dust Suppression

During construction of the Proposed Project, the County of Orange and its contractors will be required to comply with regional rules, which would assist in reducing short-term air pollutant emissions. SCAQMD Rule 402 requires implementation of dust suppression techniques to prevent fugitive dust from creating a nuisance off-site. SCAQMD Rule 403 requires that fugitive dust be controlled with the best available control measures so that the presence of such dust does not remain visible in the atmosphere beyond the property line of the emission source. Applicable dust suppression techniques from Rule 403 are summarized below. Additional dust suppression measures in the SCAQMD CEQA Air Quality Handbook are included as part of the project's mitigation. Implementation of these dust suppression techniques can reduce the fugitive dust generation (and thus the PM<sub>10</sub> component) from grading by 50 percent.

#### Applicable Rule 403 Measures:

- Apply non-toxic chemical soil stabilizers according to manufacturers' specifications, to all inactive construction areas (previously graded areas inactive for ten days or more).
- Water active sites at least two times daily. (Locations where grading is to occur will be thoroughly watered prior to earth moving).
- All trucks hauling dirt, sand, soil, or other loose materials are to be covered, or should maintain at least two feet of freeboard in accordance with the requirements of California Vehicle Code (CVC) section 23114 (freeboard means vertical space between the top of the load and top of the trailer).
- Pave construction access roads at least 100 feet onto the site from main road.
- Traffic speeds on all unpaved roads shall be reduced to 15 mph or less.

#### Additional SCAQMD CEQA Air Quality Handbook Dust Measures:

- Revegetate disturbed areas as quickly as possible.
- All excavating and grading operations shall be suspended when wind speeds (as instantaneous gusts) exceed 25 mph.
- All streets shall be swept once a day if visible soil materials are carried to adjacent streets (recommend water sweepers with reclaimed water).
- Install wheel washers where vehicles enter and exit unpaved roads onto paved roads, or wash trucks and any equipment leaving the site each trip.

The following mitigation measures shall be implemented by the County and its contractors during all phases of construction of the project to supplement the above standard SCAQMD measures to further reduce the fugitive dust air quality impacts:

#### AQ-11b: Dust Suppression

Apply Penzsuppress or comparable soil binders consistent with manufacturer's specifications
to all unpaved parking and staging areas and unpaved road surfaces. Penzsuppress and other
similar materials are capable of binding unpaved roads and surfaces, reducing fugitive dust
emissions generated by vehicles traveling over the unpaved surfaces by 90%.

#### AQ-11c: Dust Suppression

All locations where scrapers, dozers and compactors will be traveling on exposed earth shall
be watered four times per day and soil binders shall be used daily as necessary, consistent
with manufacturers directions. When combined with mitigation measure AQ-11a this will
reduce fugitive dust emissions from equipment travel on exposed earth by 80%.

#### AQ-11d: Dust Suppression

• All demolition materials shall be wet crushed. By wetting all demolition materials before crushing, the fugitive dust emissions from demolition are reduced by approximately 88%.

#### AQ-11e: Dust Suppression

Increase watering from twice a day to four times daily during initial storage pile placement
and maximize application of non-toxic soil binders according to manufacturer's
specifications to exposed stock piles (i.e., gravel, sand, dirt) with five percent or greater silt
content. This mitigation measure will reduce fugitive dust from storage piles by 90%.

#### AQ-11f: Dust Suppression

 All grading equipment will be mounted with TrueFog dust suppression technology or comparable technology. This technology sprays a very fine mist of water around the construction equipment. This mist combines with the fugitive dust in the air causing it to fall back to the ground. When combined with mitigation measure AQ-11a this will reduce fugitive dust emissions from grading by 75%.

#### AQ-11g: Exhaust Emission Reduction

The SCAQMD CEQA Air Quality Handbook outlines several mitigation measures that will be utilized by the County and its contractors to minimize the potential impact from equipment exhaust emissions during all construction phases. These mitigation measures are listed below:

- Configure construction parking to minimize traffic interference.
- Provide temporary traffic control during all phases of construction activities to improve traffic flow (e.g., flagperson).

- Schedule construction activities that affect traffic flow to off-peak hours (e.g., between 7:00 p.m. and 6:00 a.m. and between 10:00 a.m. and 3:00 p.m.).
- Develop a construction traffic management plan that includes, but is not limited to:
  - Rerouting construction trucks off congested streets.
  - Consolidating truck deliveries.
  - Providing dedicated turn lanes for movement of construction trucks and equipment on-site and off-site.
- Suspend all construction equipment operations during second stage smog alerts. For daily forecast, to identify second stage smog alerts, the following number should be called: (800) 242-4022 1-800-288-7664 (1-800-CUT-SMOG) (Los Angeles and Orange counties).
- Use electricity from power poles rather than temporary diesel power generators whenever feasible.
- Use electricity from power poles rather than temporary gasoline power generators whenever feasible.
- Develop a construction management plan to minimize the use of construction fleet vehicles and equipment.
- Prevent utility and delivery trucks from idling on-site longer than two minutes.

The following additional mitigation measures to reduce vehicle/equipment emissions shall be implemented by the County and its contractors to supplement the above standard SCAQMD measures to reduce the construction air quality impacts during all construction phases (unless otherwise noted):

#### AO-11h: Exhaust Emission Reduction

• All on-site mechanic and foreman trucks and vehicles will be required to meet Super Ultra Low Emission Vehicle (SULEV) or Zero Emission Vehicle (ZEV) emission standards.

#### AO-11i: Exhaust Emission Reduction

• Off-road construction equipment shall comply with the requirements 40 CFR (9,86, and 89) Tier 2 emission requirements, which provide for strict emission limit requirements for construction vehicles. Federal regulations require that all new off-road construction equipment meet the Tier 2 emission requirements by Phase 1 (The date of implementation is dependent upon the horsepower rating of the engine.) All equipment used for the Proposed Project will be required to meet the regulations beginning on the first day of construction.

#### AQ-11j: Exhaust Emission Reduction

 All off-road construction equipment shall comply with the requirements of AQMP Measures M9 and M10 limiting brake horsepower/per hour emissions to 2.5 grams of NOx. All new off-road construction equipment will be required to meet the CARB's 2.5 g/bhp-h emissions threshold by Phase 1. All equipment used for the Proposed Project will be required to meet the regulations beginning on the first day of construction.

#### AQ-11k: Exhaust Emission Reduction

• The use of selective catalytic reduction (SCR) or comparable technology shall be used for all on-site construction equipment. SCRs and other emission reduction technologies capable of reducing NOx emissions by at least 70% will be available, and must be installed on all offroad construction equipment, beginning on the first day of construction.

#### AQ-111: Exhaust Emission Reduction

 All off-road construction equipment shall be fitted with a particulate trap with a minimum 80% PM<sub>10</sub> reduction efficiency, which will substantially reduce soot from diesel engine exhaust.

#### AQ-11m: Exhaust Emission Reduction

All on-site construction equipment and all construction material delivery trucks shall be
required to use low sulfur diesel fuel (15 ppm). Low sulfur fuel contains up to 97% less
sulfur than standard diesel fuel, which allows for the use of exhaust control devices that
substantially reduce air pollution from construction vehicles. Several of the new
technologies, including SCR and the new on-road diesel engines, require the use of low
sulfur diesel fuel to function properly and to avoid damage.

#### AQ-11n: Exhaust Emission Reduction

All construction material delivery trucks shall be required to meet the 2000 EPA on-road diesel regulations. Effective in Phase 1, all new on-road diesel truck will be required to meet strict emission regulations that will dramatically reduce NOx, PM<sub>10</sub>, and SOx emissions. All on-road material delivery trucks serving the Proposed Project will be required to meet the new regulations before any material deliveries or debris hauling can be performed.

#### AO-11o: Exhaust Emission Reduction

 To the maximum extent permitted by law and regulations, the County and its contractors shall require that construction workers be housed (Monday through Friday) on-site in trailers/mobile homes/RVs or reused military housing, and shall provide rail/bus/metro passes or clean vehicle shuttle service for those construction workers that will not be housed on-site.

#### AQ-11p: Exhaust Emission Reduction

The County and its contractors shall provide clean-fleet shuttles to major rail transit stations
and multi-modal centers during the construction phases of the project. The Irvine
Transportation Center is located adjacent to the OCX project site.

#### AQ-11q: Exhaust Emission Reduction

• The County and its contractors shall develop an on-site rock crushing facility to recycle and reuse demolition debris (rock/concrete) to reduce material delivery and debris hauling trips. This will reduce material delivery and debris hauling emissions by approximately 6%.

#### AQ-IIr

• Construction schedules shall be adjusted consistent with the market availability of technology required for construction equipment.

#### AQ-11s: Asphalt Paving

 The construction contractors shall utilize emulsified asphalts that do not contain volatile hydrocarbons in lieu of cutback asphalts to avoid VOC/ROC emissions associated with cutback asphalts.

#### AQ-11t: Architectural Coatings

• The County and its contractors shall minimize the amount of architectural coatings by utilizing spray equipment that have high transfer efficiencies, such as the electrostatic spray gun and manual paint applicators (rollers, brushes, sponge).

#### AQ-11u: Architectural Coatings

• The construction contractors shall maximize, to the extent feasible, the use of precoated materials or materials that have natural surfaces (marble, glass) that do not require the application of architectural coatings. The building surface areas used for the project shall be at least 70 percent precoated or composed of natural surfaces.

#### AQ-11v: Architectural Coatings

 The County shall use low or zero VOC content paints wherever feasible to reduce VOC emissions from architectural coatings.

#### AQ-11w: Asbestos Removal

All asbestos removal activities shall comply with the requirements of SCAQMD Rule 1403.

### Operational Emissions

The significant operational emissions impacts identified for the Proposed Project would be mitigated by the following feasible mitigation measures:

AQ-12 Consistent with Mitigation Measure T-3 in Section 4.3 of Draft EIR No. 573, prior to issuing any building permits, the Director of the <u>Planning and Development Services</u>

Public Facilities and Resources Department shall prepare a Transportation Demand Management (TDM) Plan for the MCAS El Toro site pursuant to the County

Transportation Demand Management Ordinance. The TDM Plan will address such things as vanpooling, fleet pooling, ridesharing, public transit, alternative work hours, bikeways and other measures related to the mitigation of traffic through demand management. The TDM Plan also will establish a program to implement the following air quality mitigation measures which are components of the TDM Plan: AQ-15i, AQ-23, AQ-32, AQ-37, AQ-41, and AQ-42. Annual monitoring reports will be required. These reports will assess the implementation status of the required TDM Plan features. In addition, a transportation coordinator will be provided to support the TDM Plan.

- AQ-13 Prior to commencement of construction, the County and its contractors shall incorporate into construction programs the appropriate mitigation measures applicable to airports identified in the 1994 and 1997 AQMPs. The 1997 AQMP contains one "Off-Road Mobile Source Control Measure" for aircraft (1997 AQMP Number M15). M15 reflects U.S. EPA adoption of nationwide emission standards for new aircraft engines. The focus of M15 is on the U.S. EPA since the federal government has sole authority over these emission sources. M15 is a mitigation measure that is to be implemented by and within the control of agencies other than the County of Orange. Other AQMP mitigation measures applicable to the non-aircraft portion of an airport are also included in these mitigation measures.
- AQ-14 The County of Orange shall include electrical power outlets for landside passenger shuttles, including incorporation of electrical power outlets in the terminal and parking lot designed to accommodate electric shuttle vehicles no later than the opening of the new terminal, contemplated to occur at the end of Phase 1. This would include providing electrical outlets for battery charging for passenger shuttles that serve hotels, rental car agencies, and other businesses associated with airport operations.
- AQ-15 Prior to opening Phase 1, the County of Orange shall require the following mitigation measures applicable to alternative fuels:
  - a) For airport owned and operated (directly or by contract) vehicles, the County shall require County owned airport support vehicles to use conventional or alternative fueled vehicles that achieve exhaust emissions equivalent to a SULEV or better. The type of alternative fuel vehicles selected shall reflect the appropriate power technology available at the time of vehicle acquisition. The County shall design the terminal area with the necessary infrastructure to fuel these vehicles.
  - b) The County shall require third party vehicles (such as shuttles, trucks and vans) using terminal areas to use low emission alternative fuels (comparable to SULEV emissions or better) and will provide all reasonably necessary facilities to fuel these vehicles. These facilities shall be provided no later than the opening of the new terminal, contemplated to occur at the end of Phase 1.

- c) The County shall provide preferential parking for alternative fueled vehicles in connection with all phases of the Proposed Project sufficient to encourage the increased use of alternative fuel vehicles in the fleet mix for the Basin.
- d) The County shall include in its bid proposals for rental vehicles and taxicab services in connection with operations in all phases of the Proposed Project a provision that requires the use of conventional or alternative fueled vehicles at the airport that achieve emissions equivalent to a SULEV or better, in compliance with SCAQMD Rule 1194.
- e) The County shall use incentives to encourage the use of alternative fueled rental cars using on-airport RAC facilities.
- f) The County shall use solar and/or low emission water heaters.
- g) The County shall use central water heating systems.
- h) The County shall use incentives to encourage the use of alternative fueled vehicles or engines in commercial vehicles using the terminal areas, in cargo vehicles entering the airport. The County shall require all leaseholders with diesel vehicles in their fleets to commit to 90% clean diesel fueled vehicles by Phase 1. The County shall also require that diesel fueled delivery and service trucks coming to the site to serve leaseholders be clean diesel fueled by completion of Phase 1. Clean diesel fueled vehicles are those that comply with the final federal rule regarding on-road diesel emissions issued in December, 2000, 40 CFR Parts 69, 80, and 86.
- i) The County shall require an average vehicle ridership (AVR) of 1.5 passengers for project employees, ensure that employees for both the airport and non-aviation land uses are provided information on carpooling and mass transit systems in an effort to increase the average vehicle ridership (AVR) and reduce traffic congestion and air pollutant emissions; and make available to the public information on mass transit systems through the use of signage, pamphlets, or information kiosks, or the County's website to promote the usage of mass transit and reduce traffic congestion as well as air and noise pollution.
- AQ-16 Prior to final design, the County shall ensure that final design for each phase of the project will maintain the ASMP features to minimize taxi-in and taxi-out times and reduce aircraft queuing times. These features incorporate airfield and terminal design and location of the terminal (selected to facilitate ready access to the airfield).
- AQ-17 The County of Orange shall include electrical power and preconditioned air in the design and construction of the OCX terminal gates (jetways) during each phase of the project, to reduce emissions from operating aircraft engines at the gates. These facilities allow airlines to avoid emissions associated with running the aircraft's auxiliary power unit while it is parked at the gate. The County of Orange shall advise

the tenant airlines in writing that the County supports the use of electrical power and preconditioned air at the OCX and JWA terminal gates (jetways) in order to reduce emissions from operating aircraft engines at the gates. This measure is partly within the responsibility and jurisdiction of the airlines.

- AQ-18 The County of Orange shall include electrical power outlets at OCX for electric ramp vehicles, and for battery charging for ground support equipment. The facilities will allow the airlines to convert their GSE to clean fuels. The facilities will be provided no later than the opening of the new terminal, contemplated to occur at the end of Phase 1.
- AQ-19 The County of Orange shall incorporate hydrant fueling systems for commercial jet aircraft operations at OCX. In addition, all new fuel handling and storage facilities will comply with the latest emission reduction regulations. These facilities will be provided consistent with the contemplated phasing of these facilities.
- AQ-20 The County of Orange shall advise tenant airlines in writing that the County supports the use of single or reduced engine taxiing to the extent that it would provide identifiable air quality benefits to the local community and the region, and if and to the extent, that it is determined by the FAA and the airlines to be a safe and efficient operational procedure for air carrier aircraft at OCX and JWA. This measure is within the responsibility and jurisdiction of the FAA and airlines.

## 9.6.3 Additional Mitigation Measures

Mitigation Measure HM-2, in Section 4.16, Hazardous Wastes and Hazardous Materials Use, addresses compliance with SCAQMD Rule 1403, Asbestos Emissions From Demolition/Renovation Activities during the project construction period.

Additional mitigation measures were also identified for this air quality analysis that were not considered applicable at the time EIR No. 563 was prepared. All of these mitigation measures will be implemented no later than the opening of the new terminal, contemplated to occur at the end of Phase 1, unless otherwise specified.

- AQ-21 The County of Orange will use its best efforts, in cooperation with airport users and FAA, to achieve a percentage of Runway 34 departures that would be sufficiently greater than fifty (50) percent of such departures occurring on Runway 34R from and after the end of Phase 2 of the project, after the year 2010, to result in a one hour an annual arithmetic mean (AAM) NO<sub>2</sub> value at the Irvine Transportation Center less than 0.25 0.0534 ppm. This mitigation measure is within the responsibility and jurisdiction of the FAA and the FAA can and should approve this measure.
- AQ-22 The County of Orange shall implement a program at the Proposed Project for the use or conversion of at least 64 percent of ground support equipment (GSE) to electric powered by Phase 2 and 83.5 percent by Phase 4.

- AQ-23 The County of Orange shall provide free shuttles for passengers and employees from the Airport Transportation Center to the terminal.
- AQ-24 The County of Orange shall install a monitoring system to ensure compliance with alternative fuel measures for airport shuttles and taxis at OCX.
- AQ-25 The County of Orange shall use lighting controls and energy efficient lighting at OCX in the design of airport facilities.
- AQ-26 The County of Orange shall use energy efficient low pressure sodium parking lot lights at OCX.
- AQ-27 The County shall purchase electricity for OCX and JWA from "green," non-polluting sources. Energy efficient building design shall be employed for the airport terminal to reduce consumption of electricity and gas. The energy efficient design shall include, but not be limited to, the following measures: orient terminal buildings design to the north for natural cooling and include passive solar design (e.g., daylighting); use fuel cells to produce heat and electricity; use light colored roof material to reflect heat; and use energy efficient or low emission appliances.
- AQ-28 The County of Orange shall use energy efficient and automated controls for air conditioners at OCX.
- AQ-29 The County of Orange shall provide preferential parking spaces for carpools and vanpools and provide 7 ft. 2 inches minimum vertical clearance in parking facilities for vanpool access at OCX.
- AQ-30 The County of Orange shall implement an on-site circulation plan in parking lots to reduce vehicle queuing at OCX.
- AQ-31 The County of Orange shall provide for on-site employee services such as food services, financial services, etc., at OCX.
- AQ-32 The County of Orange shall construct on-site bus turnouts and passenger shelters at OCX.
- AQ-33 The County of Orange shall require all construction deliveries to be made with clean fuel vehicles.
- AQ-34 The County shall promote the use of clean-fueled cargo vehicles through on-airport access pricing at cargo facilities for clean fuel trucks (CNG, LNG).
- AQ-35 The County shall design the parking facilities to encourage pay-on-foot (before getting into car) to minimize idle time.

- AQ-36 The County shall provide free parking, charging stations, and preferential parking locations for electric vehicles in all (including employee) lots.
- AQ-37 The County shall establish airport shuttle bus services from significant trip origin locations in the region.
- AQ-38 The County shall manage curbside drop-off by controlling traffic flows and providing coupon incentives for passengers to use shuttle van services.
- AQ-39 The County shall use clear signing and information systems on roadways to direct traffic through airport.
- AQ-40 The County shall require that contractors or lessees implement a home dispatching system where airport employees receive routing bus schedule by phone or electronically instead of driving to work at airports.
- AQ-41 The County shall include bicycle parking facilities, such as bicycle lockers and racks.
- AQ-42 The County shall include showers for pedestrian and bicycling employees' use.
- AQ-43 The County of Orange will attempt to enter into a Memorandum of Agreement with the South Coast Air Quality Management District and the California Air Resources Board to conduct toxic emission concentration measurements in the vicinity of JWA and OCX and provide the Orange County Board of Supervisors with annual reports regarding these measurements. This monitoring effort will allow the County to correlate the actual measured concentrations with those predicted in this SA. The County anticipates that this monitoring effort will support the conclusion that the modeling results of this SA incorporate worst case assumptions and are conservative similar to what was found at SeaTac Airport in connection with its monitoring efforts. This mitigation measure is partially within the control of SCAQMD and CARB and they can and should approve and implement this measure. This mitigation measure is partially within the control of SCAQMD and CARB and they can and should approve and implement this measure.
- AQ-44 Prior to commencement of construction, the County shall use its best efforts to obtain RECLAIM Program Credits allocated to El Toro from the federal government as a part of the BRAC process in order to offset the NO<sub>X</sub> and SO<sub>X</sub> emissions of the project. This mitigation measure is within the jurisdiction of the federal government, which can and should implement this measure.
- AQ-45 The County shall centralize the airport rental car facilities and provide a consolidated shuttle service.

Note: Mitigation Measures numbered AQ-13, AQ-20, AQ-21, AQ-43, and AQ-44 are within the jurisdiction and control, in whole or part, of public agencies or entities other than the County of

Orange. Therefore, these mitigation measures are not relied upon in this Supplemental Analysis to determine significance of adverse impacts after mitigation.

The following mitigation measures were considered to reduce the health risk impacts associated with the proposed project.

- TAC-1 The County shall require that a minimum of 70-64 percent of all commercial GSE will be powered with electric power by Phase 1, and 90-83.5 percent by Phase 2 (see Mitigation Measure AQ-22).
- TAC-2 The County shall require that 400 Hz power and preconditioned air be provided at each commercial aircraft gate, eliminating the need for jet aircraft to use APUs (see Mitigation Measure AQ-17).
- TAC-3 The County shall require that all light/medium/heavy diesel vehicles transporting cargo or Jet-A fuel on OCX or JWA be in compliance with the new EPA and CARB rules for reducing diesel PM10 from exhaust. A reduction of 80% of diesel exhaust PM10 from these sources is assumed with this measure (see Mitigation Measure AQ-15).
- TAC-4 The County of Orange will use its best efforts to enter into a Memorandum of Agreement with the South Coast Air Quality Management District and the California Air Resources Board to conduct toxic air contaminant measurements. to the extent possible, in the vicinity of JWA and OCX and to provide annual reports to the Orange County Board of Supervisors regarding the results of these monitoring efforts. This monitoring effort should allow the County to correlate the actual measured concentrations with those predicted in this SA. The County anticipates that this monitoring effort will support the conclusion that the TAC modeling results of this SA incorporate worst case assumptions and are conservative. If the actual measurements result in exceedances of established threshold levels, the County shall aggressively pursue the implementation of any new technological advances that may be available at that time to mitigate any exceedances. These measures may include, but not be limited to, aircraft engine emissions reduction technologies and low emission fuel replacements, both of which are currently in the development process by FAA and other private and public entities.

## 9.7 MITIGATION RELATED TO SOILS, GEOLOGY AND SEISMICITY

### 9.7.1 Final EIR No. 563 Mitigation Measures

- G-1 Mitigation Measure for Soils: Prior to acquiring a grading permit for project construction, the County of Orange will require preparation of a Geology Report and Preliminary and Final Soils Reports, to specifically assess the following:
  - i) the shrink-swell potential of potentially expansive soils on the site and specifically address this issue with appropriate recommendations for soil treatments, grading procedures and/or foundation designs, as appropriate, for the planned land uses on the site.
  - ii) the issues of possible hydroconsolidation and settlement of soils through the addition of irrigation water to the site and variations in groundwater levels in any possibly collapsible soils on the site.
  - iii) the issues of soils removal and recompaction where necessary and the suitability of the soils on the site that will be excavated for use as structural fills on the MCAS El Toro site.

The recommendations from the Geology and Soils report will be incorporated into the grading plan for the project to be followed by the County.

G-2 Mitigation Measure for Fugitive Dust: Prior to obtaining a grading permit for project construction, the County of Orange will require the Geology Report and the Preliminary and Final Soils Reports to specifically assess grading control with special emphasis on controlling fugitive dust which could be generated during site preparation, grading and construction. These reports will specifically provide for establishing procedures for dust control and monitoring so that unacceptable levels of dust do not escape from the site. These dust control measures will be coordinated with the dust control measures described earlier in Section 4.5 (Air Quality).

The standards and procedures developed in the reports will be incorporated into the grading plan to be followed by the County.

G-3 Mitigation Measure for Seismicity: With regard to the potential for severe seismic shaking due to major earthquakes on any of the several active faults in southern California, the County of Orange will ensure that all structures on the MCAS El Toro site will be designed in accordance with the seismic design provisions in the Geotechnical Report and of the Uniform Building Code to promote safety in the event of such an earthquake.

G-4 Prior to obtaining a grading permit for project construction, the County of Orange will require that detailed geotechnical and hydrological reports be prepared specifically addressing any needed modifications to the existing drainages on the MCAS El Toro site. These reports will also specifically address the issue of erosion control for the construction phase and long term operations of the planned land uses on the site. These geotechnical and hydrologic reports will also address surface runoff from the MCAS El Toro site into the channels during construction and/or modifications of the channels and during grading for the planned land uses on the site, with specific recommendations to prevent soil erosion, siltation and debris influx into the drainage system. The reports will also address the ecological sensitivity of the down drainage areas such as San Diego Creek and Upper Newport Bay to increased runoff and siltation. Recommendations and procedures will be developed to ensure these sensitive areas are not subjected to increased runoff, siltation and debris influx as a result of the planned land uses on the MCAS El Toro site.

The standards and procedures developed in the reports will be incorporated into the grading plan to be followed by the County.

## 9.8 MITIGATION RELATED TO HYDROLOGY AND WATER QUALITY

## 9.8.1 Mitigation Measures Identified in Final EIR No. 563

### 9.8.1.1 Hydrology

Final EIR No. 563 determined that no significant hydrologic impacts would result from the CRP and, therefore, no hydrologic mitigation measures were identified.

## 9.8.1.2 Water Quality

Final EIR No. 563 determined that the long-term water quality impacts associated with the CRP were potentially significant and adverse and, on that basis, identified the following mitigation measures to reduce potential impacts to a level below significant:

WQ-1 The County of Orange will require project developers to implement the appropriate measures, or participate in a fee program to finance such measures, for implementation of the County's sediment control program known as the "208 Plan" as documented in the Flood Control Master Plan for San Diego Creek.

Relationship to DEIR No. 573: Mitigation Measure WQ-1 is specifically directed at the CRP, which contemplated that private parties would develop the MCAS El Toro site. Under the Proposed Project, the County will be the primary developer. As such, the County will ensure compliance with all 208 Plan-related directives, as appropriate.

WQ-2 Prior to issuance of grading permits and prior to any grading on the site, the project developers and/or the County of Orange will obtain appropriate 401 (Water Quality Certification), 402 (National Pollutant Discharge Elimination System), 404 (dredge and fill) and 1601/1603 (streambed alteration) permits.

Relationship to DEIR No. 573: Mitigation Measure WQ-2 remains applicable to the Proposed Project. See EIR No. 573 Mitigation Measure WQ-4.

WQ-3 The County will assess any need for revisions to the San Diego Creek Flood Control Master Plan and require the project to pay its fair share of improvements to implement the San Diego Creek Flood Control Master Plan.

Relationship to DEIR No. 573: The improvements to the MCAS El Toro drainage system that are part of the Proposed Project are consistent with and in furtherance of the San Diego Creek Flood Control Master Plan. Throughout project implementation, the County will continue to assess the Master Plan as it applies to the Proposed Project and pay its fair share for improvements, as appropriate.

# 9.8.2 Mitigation Measures Identified in Final EIR No. 563, Final Supplemental Analysis

### 9.8.2.1 Hydrology

Seven mitigation measures related to hydrology impacts were recommended in the Supplemental Analysis to Final EIR No. 563. Four of the mitigation measures (HY-3 through HY-6) were directed toward impacts not caused by any of the reuse alternatives, including the CRP. Rather, these mitigation measures were deemed to be beneficial in reducing substantial regional flooding caused by previously existing off-base flows and to reduce excess Marshburn Channel outflow caused only in part by the CRP or other reuse alternatives.

The following mitigation measures were recommended as a means to reduce flooding conditions related to the Marshburn Channel to non-significant levels. It also was noted the completion of the Marshburn Retarding Basin would enhance water quality significantly by reducing sediment loads within the channel.

HY-1 During the airport system master planning process (and by a date not later than the issuance of grading permits), the County of Orange will reconstruct and expand, or cause the appropriate entity to reconstruct and expand, the Marshburn Channel to a level capable of accommodating project-related flows (in the event such improvements have not already been completed prior to that time); and

Relationship to DEIR No. 573: Interim improvements completed in 1997 to the Marshburn Channel (the expansion of the Marshburn Channel between Interstate 5 and Trabuco Road) constitute the effective implementation of Mitigation Measure HY-1.

HY-2 During the airport master planning process (and by a date no later than the issuance of grading permits), the County of Orange will construct, or cause the appropriate entity to construct, the Marshburn Retarding Basin to a level capable of accommodating project-related flows (in the event such improvements have not already been completed prior to that time).

Relationship to DEIR No. 573: Interim improvements completed in 1998 to the Marshburn Retarding Basin constitutes the effective implementation of Mitigation Measure HY-2.

The following mitigation measures to the Bee Canyon Channel were identified as a means to substantially reduce flooding at the southern base boundary and to substantially reduce regional flooding upstream of the base (both impacts not caused by any of the reuse alternatives). It also was noted that the completion of the Bee Canyon Channel would enhance water quality significantly by reducing sediment loads within the channel.

HY-3 During the airport master planning process (and by a date no later than the issuance of grading permits), the County of Orange will reconstruct the Bee Canyon Channel in a manner which diverts excess flows into the Marshburn Retarding Basin at a level capable of accommodating project flows (in the event such improvements have not already been completed prior to that time).

Relationship to DEIR No. 573: Interim diversion of excess flows from the Bee Canyon Channel into the Marshburn Retarding Basin completed in 1998 constitutes the effective implementation of Mitigation Measure HY-3.

HY-4 During the airport master planning process (and by a date not later than the issuance of grading permits), the County of Orange will modify, or cause the appropriate entity to modify, the Bee Canyon Channel at Lambert Road to a level that will accommodate project-related flows (in the event such improvements have not already been completed at that time).

Relationship to DEIR No. 573: Interim channel modifications at Lambert Road in 1998 constitute the effective implementation of Mitigation Measure HY-4.

HY-5 During the airport master planning process (and by a date not later than the issuance of grading permits), the County of Orange will reconstruct, or cause to reconstruct, the Bee Canyon Channel stilling basin and related tailworks near the southern base boundary to a level capable of accommodating project-related flows.

Relationship to DEIR No. 573: As noted in the Final Supplemental Analysis and related findings adopted by the County, Mitigation Measure HY-5 (as well as Mitigation Measure HY-7) is directed at impacts not caused by the CRP. While CEQA does not require that such impacts be mitigated, the County will, as is its custom, work cooperatively with any governmental or private entities undertaking or responsible for any improvements recommended by Mitigation Measure HY-5 (and HY-7) during project implementation. In any event, as discussed in Section 4.8.6.1, Bee Canyon flows exiting the base under the Proposed Project utilize only 75% of the channel's capacity, and mitigation is not required.

The following mitigation measure related to the Agua Chinon Channel was identified as a means to substantially reduce regional flooding upstream of the base (an impact not caused by any of the reuse alternatives) and as a means to reduce excess outflow caused in part by any of the reuse alternatives. It was also noted that the completion of the Agua Chinon Retarding Basin would significantly enhance water quality by reducing sediment loads within the channel.

HY-6 During the airport master planning process (and by a date no later than the issuance of grading permits), the County of Orange will construct, or cause the appropriate entity to construct, the Agua Chinon Retarding Basin north of MCAS El Toro to a level capable of accommodating project-related flows (in the event such improvements have not already been completed by that time).

Relationship to DEIR No. 573: Completion of construction of the Agua Chinon Retarding Basin in 1997 constitutes the effective implementation of Mitigation Measure HY-6.

The following mitigation measure, related to the Borrego Canyon Channel, was identified as a means to permit existing regional discharges, while limiting potential damage to the railroad, to non-significant levels:

HY-7 During the airport master planning process (and by a date not later than the issuance of grading permits), the County of Orange will: (i) construct, or cause the appropriate entity to construct, a concrete-lined section or stilling basin in the Borrego Canyon Channel to a level capable of accommodating project-related flows; and, (ii) construct, or cause the appropriate entity to construct, improved railroad bridge supports and abutments at the juncture of the Borrego Canyon Channel and the OCTA bridge to a level capable of accommodating project-related flows.

Relationship to DEIR No. 573: As noted in the Final Supplemental Analysis and related findings adopted by the County, Mitigation Measure HY-7 (as well as Mitigation Measure HY-5) is directed at impacts not caused by the CRP. While CEQA does not require that such impacts be mitigated, the County will, as is its custom, work cooperatively with any governmental or private entities undertaking or responsible for any improvements recommended by Mitigation Measure HY-7 (and HY-5) during project implementation. In any event, as discussed in Section 4.8.6.1, Borrego Channel flows exiting the base under the Proposed Project are approximately 1% over existing conditions, utilizing only 75% of the channel's capacity, and mitigation is not required.

Implementation of Mitigation Measures HY-5 and HY-7 was not recommended since these measures would improve conditions unrelated to the CRP.

### 9.8.2.2 Water Quality

The Supplemental Analysis determined that, under the CRP, water quality would not be significantly impacted as on-base sediment discharges would either decrease over existing levels (Marshburn, Bee Canyon Channels) or not be impacted (Agua Chinon, Borrego Canyon Channels). For this reason, the Supplemental Analysis did not propose mitigation measures related to water quality.

## 9.9 MITIGATION RELATED TO BIOLOGICAL RESOURCES

### 9.9.1 Final EIR No. 563 Mitigation Measures

Final EIR No. 563 did not include any mitigation measures for biological resources.

### 9.9.2 Additional Mitigation Measures

Although there are no identified direct significant impacts anticipated to result from implementation of the Proposed Project, the following mitigation program for biological resources includes precautionary or preventative measures to ensure that activities during construction do not cause nest abandonment by listed species such as the California gnatcatcher or raptor species such as nesting hawks. These measures are proposed because of the anticipated long-term construction activities necessary to implement the Proposed Project and the potential for indirect impacts that might occur in the areas closest to the Habitat Reserve during construction, such as noise, motion, and startle effects resulting from construction workers, equipment operation, and other activities.

Direct impacts during construction are also addressed in this mitigation program. Precautionary measures are included to prevent impacts to streambed resources. Finally, mitigation measures that address general biological resources protection are provided. These general measures apply to all the protected species during construction of the Proposed Project.

A Project Biologist will be identified to work with the County on the implementation and oversight of these mitigation measures.

## 9.9.2.1 Mitigation Measures for the Coastal California Gnatcatcher/Listed Species

#### Pre-Construction Measures

The following measures will assist in avoiding and minimizing indirect construction related impacts on the California gnateatcher.

- B-1 Prior to issuance of each grading permit for any site preparation, clearing, grubbing, or grading within 500 feet of potential gnatcatcher habitat, the County will provide for preconstruction gnatcatcher surveys conducted by a qualified biologist to update their presence/absence status and location in relationship to the construction limits and activities.
- B-2 Prior to issuance of the applicable grading permit, the County will provide for the mapping of special interest species and habitats within 500 feet of the project limits on the grading plans by a qualified biologist. California gnatcatcher habitat and other native habitats outside the construction limits will be designated on the grading plans and construction documents as Environmentally Restricted Areas (ERAs). ERAs are defined as areas that are permanently restricted to any unauthorized access or activities during construction. These ERAs will be marked on all construction documents.
- B-3 Prior to issuance of the applicable grading permit for areas in proximity to the Habitat Reserve and the Wildlife Habitat Area, especially where adjacent to gnatcatcher habitat, the County will prepare landscape plans/guidelines to avoid any potential indirect impacts associated with invasive weeds and plant material. These plans/guidelines will be reviewed by the Director of the Orange County Nature Reserve.
- B-4 Prior to issuance of each grading permit for any site preparation, clearing, grubbing, grading, building demolition or phase of the project, focused supplemental surveys will be conducted for listed species, including those not presently identified in this EIR, that may potentially occur within areas impacted by the project. Listed species are those designated as endangered or threatened by state and federal resource agencies. The focused, pre-construction surveys will be conducted by qualified biologists who possess any required permits, as designated by the resource agencies.

If the presence of such species is documented during the pre-construction surveys, and potential project impacts are considered to be significant, a mitigation plan will be developed. The mitigation plan will incorporate measures that will substantially reduce project impacts on occupied habitat through such methods as avoidance of occupied habitat or project phasing to avoid breeding periods. A monitoring plan may also be designed to ensure successful implementation of mitigation measures.

#### Construction Measures

The following mitigation measures will avoid or substantially minimize direct and indirect construction related impacts on the California gnatcatcher.

- B-5 If active nest sites are found during the preconstruction surveys, the County will require all construction contractors to phase all site preparation, grading, clearing, grubbing, and construction activities within 500 feet of habitat occupied by nesting gnatcatchers to allow for the completion of nesting and breeding activities (approximately mid-February through mid-August). The implementation of this measure will be overseen and conducted by a qualified biologist.
- B-6 During all site preparation, grading, clearing, grubbing, and construction activities, the County will require all construction contractors to limit movement related to construction activity, including ingress and egress of equipment and personnel, to the designated construction limits in areas within 500 feet of habitat occupied by the California gnatcatcher.

#### 9.9.2.2 Mitigation Measures for Raptors

#### Pre-Construction Measures

The following measures will assist in avoiding and minimizing direct and indirect construction related impacts on any nesting raptor species.

- B-7 Prior to any site preparation, clearing, grubbing, or grading within 1,000 feet of potential raptor nest sites, the County will require that raptor nesting surveys be conducted by a qualified biologist to determine the potential presence and location of active raptor nests in relationship to the construction limits and activities.
- B-8 The County will require that all active raptor nests within 1,000 feet of the construction area limits be mapped on construction documents by a qualified biologist and designated as Environmentally Sensitive Areas (ESAs) prior to any site preparation, clearing, grubbing, or grading. An ESA is defined as an area temporarily restricted from any unauthorized construction access or activities.

#### Construction Measures

B-9 As a condition of the ESA, the County will require all construction contractors to temporarily restrict/prohibit all construction related activities within 1,000 feet of active nests to avoid potential impacts on raptor nesting activities. The majority of raptor breeding activity typically occurs between February 15 and July 15. The County will require that nesting sites be resurveyed toward the end of the breeding season to verify completion of the breeding cycle. The ESA designation will be removed by the County after the nesting activities are complete.

#### 9.9.2.3 Mitigation Measure for Streambeds

- B-10 During final design, the County will provide for the mapping of jurisdictional waters and jurisdictional wetlands for San Diego Creek, Serrano Creek, Borrego Wash, and portions of Agua Chinon Wash designed to be retained. These jurisdictional limits will be designated on the grading plans and construction documents as Environmentally Restricted Areas (ERAs). ERAs are defined as areas that are permanently restricted to any unauthorized access or activities during construction. These ERAs will be marked on all construction documents.
- B-11 During final design, the County will modify the Runway 16L plan to shorten the "declared distance" by an estimated 200 feet. This shortening of the declared distance will allow the existing Borrego Wash and channel to be retained at its current location and configuration. This action will also allow the perimeter road to be retained such that there are no adverse impacts to Borrego Wash or improved portions of the channel.
- B-12 During final design, the County will modify the grading limits near the Agua Chinon channel, south of Runway 7R/25L and retain the southerly portion for an estimated length of 400 feet. The portion of the channel to be retained will include the cross section of the ordinary high water mark within the Army Corps of Engineers jurisdiction.
- B-13 During final design of the golf course and the Wildlife Habitat Area, the County will incorporate Serrano Creek and its tributaries into the Wildlife Habitat Area and golf course design and designate them as ERAs, as referenced in Mitigation Measure B-10.

## 9.9.2.4 General Biological Resources Mitigation Measures

- B-14 The County will show and define all designated ERAs and ESAs on project construction documents. Restrictions related to construction activities near ERAs will be described. Restrictions related to when construction can and cannot be conducted in and near ESAs will be described. The County will require all construction contractors to comply with all ESA and ERA designations during all site preparation, grading, clearing, demolition, hazardous materials remediation, grubbing, and construction activities.
- B-15 Prior to any hazardous materials remediation, site preparation, grading or construction, the County will conduct orientation meetings for all construction personnel. These meetings will identify protected species in the project area, any defined ESAs and ERAs and restrictions related to them, and other information relative on the avoidance of impacts to protected species during construction.
- B-16 The County will require all project contractors to maintain and operate all construction equipment in good working order and consistent with the manufacturer's specifications for noise control and management. The County will require all project contractors to document that all equipment used for site preparation, grading, demolition, clearing, grubbing, and hazardous materials remediation meets the manufacturers' noise specifications.

### 9.9.2.5 Cumulative Biological Resources Mitigation Measures

Based on the Cumulative Impact Analysis (Section 5.4.9), the following mitigation measure will avoid or substantially minimize cumulative impacts for the Alton Parkway Extension:

B-17 The County is the lead agency for a redesign of Alton Parkway. It will redesign the extension to minimize biological impacts to one acre or less of Waters of the U.S. and jurisdictional wetlands impacts.

## 9.10 MITIGATION RELATED TO PUBLIC SERVICES AND UTILITIES

## 9.10.1 Final EIR No. 563 Mitigation Measures

PSU-1 Concurrent with construction level environmental documentation, the County of Orange will coordinate with all utility and service providers including but not limited to, those identified below regarding the provision of needed services and utilities for the project. Each of these utility and service providers is subject to applicable internal regulations and local, state and, in some cases, federal statutes and regulations. In compliance with these regulations, arrangements will be made with each provider for needed services and utilities prior to implementation of the relevant phases of the project, thus avoiding any significant adverse project impacts.

Orange County Sheriff's Department
Orange County Fire Authority
Orange County Public Library System
Orange County Transportation Authority
Cable Television Services
Irvine Unified School District
Integrated Waste Management Department
Pacific Telephone
Southern California Edison
Southern California Gas Company
Irvine Ranch Water District

Relationship to DEIR No. 573: This mitigation measure was partially satisfied in the preparation of Section 4.10 of EIR No. 573 (coordination regarding provision of services and utilities). The remainder of the measure (arrangements prior to implementation) has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measures PS-1 through PS-5 and U-1 through U-4. Therefore, Mitigation Measure PSU-1 is no longer applicable.

## 9.10.2 Additional Mitigation Measures

- PS-1 The County of Orange will provide an area within the OCX terminal for a County Sheriff facility.
- PS-2 The County of Orange will comply with CCR, Title 27, by considering any potential effect of birds on aircraft in planning for the Proposed Project, as described in Section 4.15, Public Safety.
- PS-3 The County of Orange will develop a waste reduction plan for the waste generated from demolition and construction of new facilities and operation of the Proposed Project in order to comply with State law ABA 939 prior to the commencement of construction for each phase.
- U-1 During final design, existing utilities will be marked on the project plans as to whether each facility will be protected in place, removed or expanded/modified during construction. This will include all cable television, communications services, electrical, fuel, natural gas, domestic water, recycled water and waste water utility facilities on or immediately adjacent to the project site.
- U-2 Prior to any site preparation, grading or construction, the County will require all project contractors to contact a universal service alert organization. The purpose of this coordination will be to identify/verify the depth, height and location of existing utility facilities and notification procedures in the event of either a temporary disruption of service or accidental damage to a facility. This will include providers of cable television, communications services, electrical, fuel, natural gas, domestic water, recycled water and waste water facilities and services on and immediately adjacent to the project site.
- U-3 During final design, the County will coordinate with the appropriate service providers to ensure that planned project connections to the off-site distribution/collection systems are acceptable to the providers. This coordination will specifically address cable television, communications, electricity, natural gas, domestic water, recycled water and waste water service providers.
- U-4 During final design, the County will coordinate with The Irvine Company on identifying options to protect in place/relocate the existing wells and pipelines owned and operated by The Irvine Company on the project site. The requirements agreed to for the protection, relocation and/or reconstruction of these facilities will be documented in the project construction plans.
- PS-4 During final design for each phase of construction, the County will review the proposed transit facilities in Table 4.10-6 with OCTA to coordinate implementation of transit facilities with bus route programming and rail services. The County will obtain an agreement with OCTA, City of Irvine and rail agencies for the design of the proposed connection to ITC prior to the commencement of construction.

PS-5 Prior to approval of the Airport System Master Plan, the County will prepare a Fiscal Impact Report (FIR) analyzing the public facilities and services costs and revenues of the Proposed Project, including the effects on fire protection and emergency services. The FIR will be prepared in conformance with the County's FIR standards and policies. If the FIR concludes that there are negative fiscal impacts after mitigation on any facility or service, the FIR will recommend financing mechanisms, including but not limited to developer fees, assessment district financing, and/or tax increment financing (in the event of a redevelopment project area being created), for approval by the Board of Supervisors.

## 9.11 MITIGATION RELATED TO NATURAL RESOURCES AND ENERGY

## 9.11.1 Final EIR No. 563 Mitigation Measures

NRE-1 The California Code of Regulations (CCR) requires developers to incorporate features in building that are designed to be energy efficient, consistent with the recommendations of the affected utility companies and the specific requirements of Title 24 CCR "Energy Building Regulations." These regulations would apply to the development on the MCAS El Toro site, and would reduce all potential impacts of the reuse alternatives related to energy resources to below a level of significance.

Relationship to EIR No. 573: The requirements of Mitigation Measure NRE-1 are fulfilled by the Proposed Project Airport System Master Plan which requires that all buildings are designed to be energy efficient, consistent with the recommendations of the affected utility companies and the specific requirements of California Code Regulations Title 24 (24 CCR), "Energy Building Regulations."

## 9.11.2 Final EIR No. 563 Supplemental Analysis Mitigation Measures

Of the eight mitigation measures analyzed in the Final EIR No. 563 SA, two were determined to be feasible in terms of reducing the conversion of prime agricultural land to urban uses with implementation of the CRP. The implementation of these mitigation measures would not reduce the CRP significant impact on agricultural lands to less than a significant level. Therefore, the CRP impact upon prime agricultural land would remain an unavoidable significant impact.

The following mitigation measures were identified in Final EIR SA to provide partial reduction of the CRP impact to converting prime agricultural land to urban uses:

NRE AG-1 The County, acting as the LRA for MCAS El Toro, shall use its best efforts to secure the conveyance of 40 acres of existing prime agricultural land on MCAS El Toro from the Department of Navy for the benefit of the Orange County Sheriff-Coroner. The 40-acre conveyance will be used by the Orange County Sheriff-Coroner for permanent agricultural use in conjunction conduction with the

County's existing jail agricultural program at the James A. Musick Jail Facility. As a direct result of this mitigation, the conversion of prime agricultural land resulting from implementation of the Community Reuse Plan would be reduced from approximately 818 acres to 778 acres.

NRE AG-2 As part of the interim reuse phasing strategy, the County, acting as the LRA for MCAS El Toro, will use its best efforts to continue to make available for lease the acreage currently in agricultural use on the MCAS El Toro site. This action will help compensate for the adverse economic impact resulting from base closure, and provide a revenue stream to maintain the base property until project buildout. The County will continue to make agricultural lands available for lease as long as the interim uses are consistent with the uses contemplated in the Community Reuse Plan, and as long as those uses do not compromise the County's ability to implement the Community Reuse Plan in a timely manner. However, interim agricultural use would not be allowed for a period longer than 10 years.

Relationship to EIR No. 573: Final EIR No. 563 Supplemental Analysis Mitigation Measures NRE AG-1 and NRE-AG-2 will be implemented on an ongoing basis with implementation of the Proposed Project.

## 9.12 MITIGATION RELATED TO AESTHETICS, LIGHT AND GLARE

## 9.12.1 Final EIR No. 563 Mitigation Measures

Although there are no identified significant aesthetic impacts resulting from implementation of the Proposed Project, the following precautionary mitigation measures identified in FEIR No. 563, as amended, are appropriate to be included as precautionary measures.

AES-1 As appropriate, all buildings will be designed and constructed using materials to reduce glare and, when possible, will be painted in warm earth tone colors, grays, blues and similar colors which reduce glare.

Relationship to DEIR No. 573: This measure has been updated to reflect the current stage of planning. As revised, the measure is replaced by Mitigation Measure AES-3.

AES-2 Prior to approval of final site design, the County of Orange will ensure that all exterior light fixtures on the MCAS El Toro site will be the direct cut-off type of fixture with shields, non-glare bulbs, and frosted light shields, as necessary and appropriate, to reduce light spillage on and off the development site, particularly for uses adjacent to the Wildlife Habitat Area.

#### 9.13 MITIGATION RELATED TO CULTURAL RESOURCES

### 9.13.1 Final EIR No. 563 Mitigation Measures

CR-1 Concurrent with, or prior to, construction level environmental documentation, the County will ensure that a cultural resources inventory of the MCAS El Toro site will be conducted to determine whether there are any potentially NRHP eligible Cold War era structures and/or supporting features on the site and if any structures are architecturally significant. The survey will include recommendations regarding the management of any documented NRHP eligible structures.

This mitigation measure has been implemented, and no further action is required regarding the cultural resources inventory. None of the buildings or structures at MCAS El Toro qualify for listing on the National Register.

CR-2 Concurrent with, or prior to, construction level environmental documentation, the County will ensure that a report of the literature and records search and the field survey will be prepared. Mitigation may be required depending on the recommendations of this report. The report recommendations may include further testing, artifact collection, pre-grading salvage, and/or monitoring during soil disturbance. The report recommendations would be implemented during final design and grading, as appropriate.

This mitigation measure has been implemented, and no further action is required regarding a literature and records search and field work.

CR-3 Concurrent with, or prior to, construction level environmental documentation, the County will determine the need for a paleontological resources report for the MCAS El Toro site. The report recommendations would be implemented during final design and grading, as appropriate.

The County has determined that further study of the paleontological resources on the site is not required at this time and that implementation of County Standard Condition of Approval No. A7 will be recommended for approval as a precautionary measure.

#### 9.14 MITIGATION RELATED TO RECREATION

## 9.14.1 Final EIR No. 563 Mitigation Measures

FEIR No. 563 did not prescribe any specific mitigation measures for recreation. Mitigation Measure LU-1 in the Land Use Section was cross referenced for mitigation of project consistency with the County of Orange General Plan Resources Element.

## 9.14.2 Additional Mitigation Measures

- R-1 Prior to the approval of design plans, specifications, and estimates (PS&E) for off-site roadway improvements, the County shall prepare a Construction Action Plan. The Construction Action Plan shall set forth appropriate construction practices necessary to minimize the potential disruption to properties, pedestrians, bicyclists, and motorists. The following items shall be included in the Plan:
  - Public Notice: Signs providing advance notice of work to be done on a particular segment shall be posted for a period of two weeks prior to construction. Notification in a local newspaper shall be published two weeks prior to construction. Adjacent property owners and public service providers and utility companies shall also be notified two weeks prior to construction.
  - Traffic Routing: Signs shall be provided to route vehicular and bicycle traffic
    through segments under construction. In addition, signs that suggest possible
    alternate routes shall be posted. Construction vehicle access to construction sites
    shall occur at off-peak traffic hours. Construction vehicle access routes shall be
    directed around residential areas. In addition, traffic control personnel shall be
    provided as necessary to mitigate traffic congestion and to mitigate the impact to
    arterial service levels during construction in accordance with local, State, and
    federal standards.
  - Construction Staging: Storage of construction vehicles, equipment, and materials shall not occur in the immediate vicinity of residential areas and retail establishments.
  - Hours of Construction: Hours of construction shall conform to established County policy unless otherwise approved by the County.
  - Access to properties: Construction activities shall be arranged so that access to properties will be maintained.

#### 9.15 MITIGATION RELATED TO PUBLIC SAFETY

Mitigation Measure GPC-1 already proposes that the Airport Land Use Commission adopt an AELUP for the ASMP project at El Toro. The revisions to the AELUP should reflect the standards for civilian aviation use of the MCAS El Toro property, which will result in different accident risk and building height limitations in the surrounding communities than experienced with military activity.

AELUP-consistent General Plan Amendments are proposed as part of the Proposed Project. Surrounding cities with jurisdiction over areas included in the OCX AELUP study area should also adopt the land use guidelines described here in their General Plans in order to achieve the required statutory consistency.

## 9.16 MITIGATION RELATED TO HAZARDOUS WASTES AND HAZARDOUS MATERIALS USE

## 9.16.1 Final EIR No. 563 Mitigation Measures

- HM-1 Prior to accepting the MCAS El Toro property on which the presence of asbestos containing materials (ACMs) has been identified, the County of Orange or other transferee must ensure that all available information concerning ACMs has been provided, including the following:
  - i) information regarding the type, location, and condition of ACMs
  - ii) the results of any asbestos testing
  - iii) description of asbestos control measures taken, if any
  - iv) information regarding the costs or time necessary to remove existing ACMs
  - v) the results of any site-specific asbestos inventory updates
- Prior to major renovations and/or the demolition of any structure(s) known to contain ACMs, the County of Orange or other transferee must ensure that all asbestos is removed and disposed of in accordance with applicable federal, state and local regulatory requirements. Of primary concern is South Coast Air Quality Management District (SCAQMD) Rule 1403 (Asbestos Emissions from Demolition/Renovation Activities).
- Prior to the renovation or demolition of any structure constructed prior to October 1988, an asbestos survey must be conducted if the presence of ACMs is unknown. Asbestos in structures that will be retained for future uses must be managed according to an Operations and Maintenance (O&M) Plan to reduce the potential for fiber release and possible human health hazards.

HM-4 Prior to accepting the Wherry Housing and Saddleback Terrace/Vista Terrace Housing communities from the Navy, the County of Orange or other transferee shall ensure that all lead hazard abatement has been conducted. All abatement activities must be conducted in accordance with applicable federal, state, and local regulatory requirements.

The Proposed Project does not contemplate any residential uses on the site or reuse of any residential units.

Prior to accepting any other residential property from the Navy, the County of Orange or other transferee shall ensure that lead-based paint and lead hazard surveys have been conducted for those properties. For properties that have been surveyed, the County of Orange or other transferee shall obtain all inspection and survey results and shall ensure that the survey reports contain sufficient data to adequately assess potential lead hazards. The County of Orange or other transferee must ensure that the appropriate lead-hazard information has been received, and that the property transfer contract contains a lead warning statement.

The Proposed Project does not contemplate any residential uses on the site or reuse of any residential units.

HM-6 If the County of Orange or other transferee agrees to conduct all required renovation and/or lead-based paint abatement activities on residential units at MCAS El Toro, these activities must be conducted in accordance with all applicable federal, state, and local regulatory requirements.

The Proposed Project does not contemplate any residential uses on the site or reuse of any residential units.

HM-7 Concurrent with construction-level environmental documentation, the County of Orange will ensure that mitigation is identified, as appropriate, related to the requirements of the National Pollutant Discharge Elimination System (NPDES) General Construction Activities Stormwater permit administered by the Regional Water Quality Control Board, Santa Ana Region. Permit coverage must be obtained prior to beginning construction.

Relationship to EIR No. 573: This mitigation is replaced by Mitigation Measure WQ-4 in Section 4.8 Hydrology and Water Quality. This permit requirement is carried forward to the construction phase of the project.

## 9.16.2 Additional Mitigation Measures

Due to the nature of hazardous material and hazardous waste regulations, proposed land uses that involve the use, storage, handling, and/or disposal of hazardous materials or waste will be subject to all applicable federal, state, and local environmental regulations. These regulations are implemented by numerous government agencies as previously discussed. Each agency has established regulations regarding the proper management of hazardous materials and hazardous waste for specific operations and activities. The following mitigation is proposed to ensure compliance with these regulations and to reduce potential adverse impacts of the Proposed Project to below a level of significance.

- HM-8 The County shall require that monitoring for the presence of combustible gases, such as methane, shall be conducted by operators during all underground work conducted within 1,000 feet of the landfill refuse boundary and that, if gas levels exceed federal, state, or local standards, such a work shall cease until the gas levels have been remediated by the responsible agency or operators.
- HM-9 Prior to approval of permits for construction of new underground storage tanks, oil/water separators and associated piping systems, the County shall ensure that the facility's design complies with applicable requirements pertaining to underground storage tank systems. At a minimum, these tanks shall be double walled and have overflow protection and leak detection systems with remote leak alarm monitoring.
- HM-10 Prior to issuance of permits for aboveground storage tanks, County shall require all AST operators to ensure that all aboveground tank system design includes appropriate secondary containment with no less than 110 percent of the tank capacity.
- HM-11 Prior to issuance of permits for fuel storage areas, including aboveground tanks, County shall require all AST operators to ensure that runoff from all fuel handling facilities drains to an oil/water separator with stormwater samplers and monitors.
- HM-12 Prior to issuance of permits for aboveground storage tanks, County shall require all AST operators to ensure that all aboveground tank systems comply, at a minimum, with the standards of the American Petroleum Institute (API).
- HM-13 Prior to the issuance of permits for fuel hydrant piping systems, County shall require all AST operators to ensure that these systems have double wall construction, a leak detection system, and remote leak alarm monitoring as required by existing regulations.
- HM-14 Prior to issuance of permits for future fuel farm facilities at the MCAS El Toro site, the County shall ensure that these facilities have a self-contained fire suppression system, including such features as foam cannons and above and below tank foam injection systems as required by existing regulations.
- HM-15 Prior to installation of USTs, the County shall ensure that all UST operators obtain proper permits for all future USTs from the Orange County Health Care Agency, RWQCB, SCAQMD, and Orange County Fire Authority.

- HM-16 The County shall ensure that all fuel operators provide documentation of all fueling and maintenance facilities and activities involving hazardous materials to the Orange County Fire Authority prior to their construction and operation.
- HM-17 The County will require all operators (or operating agencies) of facilities storing fuels and hazardous materials to prepare Emergency Response Plans and submit Hazardous Materials Disclosure Forms to the Orange County Fire Authority as required by the agency, prior the issuance of permits for these facilities.
- HM-18 Prior to commencement of uses, all BTP tenants shall be required by the County to implement any of the Mitigation Measures HM-1 through HM-17 that are applicable to their use of leased property on the base, unless by agreement, the County or DON or other agency otherwise ensures compliance.
- HM-19 Prior to issuance of grading permits, the County shall incorporate final design and site plan features which ensure that any site runoff will be directed away from IRP landfill Sites 3 and 5.
- HM-20 The County shall require all operators (or operating agencies) of facilities using and/or storing hazardous materials to maintain a current hazardous waste generator number for the facility as a term of lease.
- HM-21 The County shall not allow any permanent storage of hazardous waste on the site by any operator (or operating agency) as a term and condition of their lease.
- HM-22 The County shall require all operators (or operating agencies) of facilities using and/or storing hazardous materials to provide to the County disclosures of hazardous materials inventories, and proof of acquisition of necessary permits and compliance with applicable plans pertaining to hazardous materials storage and use and hazardous waste generation and disposal as a term and condition of their lease.
- HM-23 The County shall require all operators (or operating agencies) of facilities storing fuels and conducting fueling activities to receive FAA-approved training from the fire department.
- HM-24 The County shall ensure that all activities proposing to use hazardous materials or generate and dispose of hazardous waste provide all applicable information concerning the nature of these activities to the Certified Unified Program Authority (CUPA) through the use of the Unified Program Consolidated Form.
- HM-25 The County shall ensure that no development on IRP Sites 3, 5, 8, 11, and 12 or development within 1,000 feet of Sites 3 and 5 occurs until appropriate remedial action has been implemented. The County will also ensure that no construction occurs on Site 16 until appropriate remediation has occurred. Finally, if FFA signatory review determines potentially significant impacts to the project could occur from Site 7 or Site 14, the County will ensure that no construction occurs on the site until appropriate remedial action has been implemented.

- HM-26 The County shall provide notification and disclosure regarding existing hazardous waste conditions and the presence of regulated materials to all leasehold tenants at OCX.
- HM-27 The County shall require all leasehold tenants to submit hazardous waste manifests annually to OCHA for review to ensure that materials have been taken off-site and recycled properly.
- HM-28 The County shall ensure that all monitoring wells are accommodated through grade adjustment of the wellheads.
- HM-29 The County shall ensure, through lease agreements, that all leasehold tenants use registered persons/firms for application of fertilizers and pesticides.
- HM-30 The United States Department of the Navy should fully implement its Base Cleanup Plan and comply fully with its statutory obligations to remediate the MCAS El Toro site.

#### 9.17 MITIGATION RELATED TO SOCIOECONOMICS

## 9.17.1 Final EIR No. 563 Mitigation Measures

SE-1 If the Proposed Project is adopted, the County of Orange will submit updated employment, population, and housing growth forecasts for the project site to SCAG for their next scheduled update of regional growth forecasts that reflect the level of activity anticipated under the Proposed Project.

#### 9.18 MITIGATION RELATED TO RISK OF UPSET

## 9.18.1 Additional Mitigation Measures

- RU-1 Prior to commencement of aviation flight operations, the County shall make every reasonable effort to lease or otherwise obtain appropriate agreement and/or approval for the use of the Norwalk Pipeline and Santa Fe Pacific Pipeline Line Section 126 (LS-126) for the purpose of conveying all jet fuel to the MCAS El Toro site. The objective of this measure is to obviate the need for and eliminate the adverse impacts associated with highway truck transport of jet fuel.
- RU-2 At any such time that the County becomes the owner of the Norwalk pipeline, the

  County or its agents will inspect the pipeline on a regular basis to conform to the
  standard practice in the industry, will take corrective action to remedy any leaks,
  ruptures, or other hazards caused by the pipeline, and will pay any damages as required
  by law that are deemed by the appropriate authority to be the liability of the County.

#### 9.19 MITIGATION RELATED TO CUMULATIVE IMPACTS

- T-12 Prior to issuance of the first building permits for each phase of the Airport System Master Plan development, the County will determine if the phase's traffic plus cumulative development traffic requires any improvement listed in Tables 5.4-2 and 5.4-3. If any improvement is required, the County will enter into a cooperative agreement with the lead local jurisdictions responsible for the cumulative development impacts to participate on a fair share basis in the implementation of the planned and unplanned long-range circulation improvements listed in Table 5.4-2 and 5.4-3. The agreements will commit the County to participate in the implementation process (i.e., the establishment of funding mechanisms, the preparation of design plans, the performance of feasibility assessments, etc.) based on the project fair share percentages identified in Tables 5.4-2 and 5.4-3. Where the County is the lead, the County will prepare cooperative agreements and use its best efforts to obtain fair share participation by non-lead local jurisdictions.
- T-13 Prior to issuance of the first building permits for the Airport System Master Plan Development, the County will use its best efforts to obtain a cooperative agreement with the applicable lead jurisdictions to process amendments to the Orange County Master Plan of Arterial Highways (MPAH) as appropriate through the Orange County Transportation Authority (OCTA) for the unplanned long-range circulation improvements listed in Table 5.4-3 [i.e., Moulton Parkway and Laguna Canyon Road] with the County's participation in the MPAH process being based on the project fair share percentages identified in Table 5.4-3.

Note: Pursuant to CEQA Guidelines Section 15091(a)(2), portions of these measures are within the responsibility and jurisdiction of another public agency, and not the County of Orange.

Mitigation Measure B-17, listed earlier in Biological Resources, applies to cumulative impacts.

#### 9.20 STANDARD CONDITIONS

The following standard conditions are incorporated in the Proposed Project:

A4	Archaeology Grading Observation and Salvage
A7	Paleontology Resource Surveillance
D1	Drainage Study
D2	Drainage Improvements
D3	Off-Site Drainage
D4	Master Plan of Drainage
D5	Elevation Certificates and Finished Floor Elevations for Floodplains
D6	Subordination of Easements
D7	Regional Facility Improvements
D8	Firm Map Revisions for Floodplains
D9	Floodplain Easements

EP1	Federal Endangered Species Act
FD1	Underground Storage Tanks
FD2	Uniform Fire Code Implementation
FD3	Waste Disposal
FF4	General County Facilities
FP1	Fire Hydrants
FP2	Water Availability
FP4	Fire Access Roads
FP5	Street Markings
FP6	Traffic Signal Preemption Devices
FP7	Fire Hazard Notification
FP8	Fuel Modification
FP9	Access Gates
FP10	Combustible Construction Letter
FP11	Hazardous Materials
FP12	Combustible Gas Mitigation
FP13	Building Use Letter
FP14	Architectural Building Plans
FP15	Fire Suppression System
FP16	Fire Alarm System
FP17	Storage Tanks
G1	Geology Report
G2	Grading Deviation
G3	Financing/Conveyance Tentative Maps
<del>G</del> 4	Off-Site and Cross-Lot Grading/Drainage
G5	Vector Control Measures
<del>G6</del>	Preliminary Soils Report
<del>G7</del>	Final Soils Report
HM1	Hazardous Materials
N1	Residential Noise
N2	Non-Residential Noise
N3	Overflight Notification
N4	Department of Real Estate Report Information
N5	Aviation Easement
N6	Aircraft Noise Signs
N7	Noise Generating Equipment
N8	Multi-Family Dwelling Units
N9	Construction Noise
N10	Transportation Corridor Notification
RC1	Uniform Fire Code Implementation
RC2	Waste Disposal
SW1	Solid Waste Collection Areas
WQ1	Pollutant Runoff
WQ2	Pollutant Runoff
WQ3	Chemical Management
WQ4	NPDES General Stormwater Permit

· ·		

# 10.0 ENVIRONMENTAL EFFECTS FOUND NOT TO BE SIGNIFICANT

CEQA Guidelines Section 15128 states an EIR "shall contain a statement briefly indicating the reasons that various possible significant effects of a project were determined not to be significant and were therefore not discussed in detail in the EIR. Such a statement may be contained in an attached copy of an initial study."

The EIR addresses a full range of factors and issues. No topics or factors were eliminated from discussion in the EIR as a result of an Initial Study or other analysis.

### 11.0 SUPPLEMENTAL PHASING ANALYSIS

#### 11.1 INTRODUCTION

The Proposed Project analyzed in Draft EIR No. 573 is anticipated to be built out over a twenty (20) year period, with projected completion by approximately 2020. Consistent with traditional airport master planning practice, Draft EIR No. 573 defines and analyzes the Proposed Project as occurring in incremental five-year "phases," defined as follows:

Phase 1 - EIR Certification (2000) to 2005

Phase 2 - 2006 to 2010

Phase 3 - 2011 to 2015

Phase 4 - 2016 to 2020

[Note: The actual calendar year during which each phase would occur is dependent upon the actual start date of airport related operations and/or construction.]

At the Phase 2 level of completion, the Proposed Project would serve a forecast commercial passenger activity level of 18.8 million annual passengers ("MAP"); at Phase 4 build out, the Proposed Project would serve a forecast commercial passenger activity level of 28.8 MAP. Draft EIR No. 573 identifies those impacts associated with build out of the Proposed Project to the Phase 4 level, as well as those impacts anticipated to occur with each phase of the Proposed Project development. One of the specific purposes of Draft EIR No. 573 in providing this phasing analysis is to afford maximum flexibility to the Board of Supervisors in the scope of project implementation and approval. (Draft EIR No. 573 p. 3-2.)<sup>1</sup>

Consistent with that purpose, this supplemental phasing analysis, in conjunction with the information presented in Draft EIR No. 573, as supplemented, provides the information necessary for the Board of Supervisors and the public to fully understand the Proposed Project both at the Phase 4 level and at the Phase 2 level.

This supplemental analysis, which is to be read in conjunction with the information previously provided in Draft EIR No. 573, presents a summary of the information provided in Draft EIR No. 573 relative to a description of the project and its impacts at the Phase 2 level of implementation, supplemented by additional information where applicable.

\*\*

All page references are to Draft EIR No. 573 (December 1999), unless otherwise noted.

#### 11.2 DESCRIPTION OF THE PROJECT AT PHASE 2

A description of the Draft EIR No. 573 Proposed Project is provided in Chapter 3.0 at pages 3-1 through 3-68. The Proposed Project consists of various actions related to the approval, adoption, and implementation (including construction and operation) of two plans: (1) an Airport System Master Plan for Orange County (ASMP) (published separately as, Draft, Airport System Master Plan for John Wayne Airport and Proposed Orange County International Airport (December 1999)), and (2) a Base Transition Plan (BTP) for former Marine Corps Air Station (MCAS) El Toro ("El Toro" or "the base") (published separately as, Draft, Base Transition Plan (November 9, 1999). (3-1.)

The BTP allows for some interim reuse of existing buildings and facilities at El Toro until implementation of the ASMP reaches a phase that requires the demolition of a specific facility, or its conversion to the permanent reuse contemplated by the ASMP. Under the BTP, a limited number of uses is scheduled to continue through Phases 3 and 4.

The Airport System Master Plan consists of an Airport Master Plan for a civilian commercial airport at El Toro (El Toro Airport Master Plan) and an Airport Master Plan for John Wayne Airport (JWA) (JWA Master Plan). A detailed list of the anticipated project improvements both at El Toro and JWA is contained in Draft EIR No. 573 Table 3-9.

The JWA Master Plan updates the County's 1985 Airport Master Plan for JWA. Under the Proposed Project, the Phase 4 forecast level of commercial passenger activity at JWA is 5.4 MAP, which is less than is currently served at JWA. The Master Plan Update for JWA does not contemplate any expansion of existing commercial aviation facilities at JWA. Rather, the Master Plan Update projects are intended to expand the general aviation facilities and capacity at the airport. All of the JWA related project components are assumed to be completed by the end of Phase 2 of the Proposed Project. (3-3.)

The El Toro ASMP presented in Draft EIR No. 573 Chapter 3.0 is composed of a large number of phased project components, classified as either aviation or nonaviation uses. The aviation use passenger related facilities contemplated by the Proposed Project are designed to serve a forecast commercial passenger activity level of 28.8 MAP by Phase 4. (3-3.) These facilities include a new terminal building, parking facilities, access roadways, airfield improvements, including aprons and the lengthening of all four runways at El Toro and the relocation of two of the four runways, as well as numerous airport support facilities. The various nonaviation uses planned for portions of the base are compatible with the planned commercial aviation use and would, in some cases, provide revenue support to the civilian redevelopment of the base and the development and operation of a commercial airport at El Toro. The general categories of the various project components, land uses, and improvements contemplated under the El Toro ASMP are summarized in Draft EIR No. 573 Table 3-1 (and listed in Draft EIR No. 573 Table 3-9 in alphabetically designated categories "A" through "Y").

At Phase 2 of the project, El Toro would serve a forecast commercial passenger activity level of 18.8 MAP and 1.18 million tons of cargo. To serve this level of passenger activity, Phase 2 of the project is composed of aviation components similar to those included in Phase 4 of the Proposed Project, although "downsized" to accommodate the reduced MAP levels. Specifically, at the Phase 2 level, the project includes a new terminal building with 44 jet gates, cargo (west) area, parking facilities, access roadways, numerous airport support facilities, and airfield improvements, including aprons and the reconstruction and lengthening of two runways at El Toro and the relocation (construction) of a third runway. (See Supplemental Phasing Analysis Table 3-9A, Summary List of Improvements At OCX And JWA Recommended for Phase 2 of the Airport System Master Plan.)

Thus, Phase 2 of the project differs from the Proposed Project at build out (Phase 4) in three significant respects. First, the Terminal Building will be 1.4 million square feet in size at Phase 2 compared to the Phase 4 terminal of 2.3 million square feet, and will include 44 jet gates versus 62 jet gates. Second, the fourth runway, Runway 7L-25R, would not be constructed until Phase 43. Third, cargo facilities in Phase 2 will be smaller than Phase 4 with a corresponding reduction in cargo handling capacity from 2.01 million tons in Phase 4 to 1.18 million tons in Phase 2. Full build out of the Terminal Building, as well as construction of Runway 7L-25R and additional cargo facilities, is scheduled to occur during Phases 3 and 4 of the Proposed Project. The land area required for aviation uses in Phase 2 is approximately 355 acres smaller than the land area required for aviation uses in Phase 4 (compare Draft EIR No. 573, Table 3-1 with Supplemental Analysis Table 3-1A). At Phase 2, these ±355 acres would not be developed for aviation uses, but rather would continue to be used for ongoing nonaviation uses, including interim uses of existing buildings, facilities, grounds, and agricultural leases. In addition to agricultural leases and golf course lands, Phase 2 includes interim reuse of existing buildings and facilities, including 80,000 sq. ft. of business park and 280,000 sq. ft. of warehouse floor area (Draft EIR No. 573 Table 7-4 Appendix D, Part 1).

With respect to nonaviation uses, at Phase 2 of the Proposed Project the proposed Regional Park and Wildlife Habitat Area are not yet fully developed. In all other respects, nonaviation revenue support uses are fully developed at the conclusion of Phase 2.

Draft EIR No. 573, Figure 3-4 illustrates the proposed development of OCX by phases. Specifically, Figure 3-4 depicts the construction and development scheduled to occur during each phase by means of color coding. In this manner, Table 3-4 provides the reader with a view of the project at the Phase 2 level of implementation.

Table 3-1A

Proposed P	roiect at Phas	e 2 of Develo	pment – Uses/Acreages B	v Planning Area	
LAND USE	PA/PARCEL	ACRES	LAND USE	PA/PARCEL	ACRES
	<b>.</b> .				
Agriculture	5-1a	37.17	Cargo (west)	4-11a	92.38
Agriculture	3-1b	27.01	Cargo Total		92.83
Agriculture	3-1c	13.03	1. 0:1. 0	0.10	0.01
Agriculture	3-1d	0.37	In-flight Catering	2-12a	8.91
Agriculture	7-1e	61.35	Aviation/Industrial Related	2-12b	43.62
Agriculture Total		138.93	Control Tower	2-12c	4.00
G 160 OT 11)			FBO/Corporate Aviation	2-12d	46.27
Golf Course (North)	3-3a	135.26	ARFF	2-12e	4.00
Golf Course (North)	3-3b	51.25	Airline Maintenance	3-12e g	14.01
Golf Course (South)	7-3c	43.18	Ground Services Equipment	4-12h	3.74
Golf Course (South)	7-3d	28.44	Fuel Storage	4-12i	7.80
Golf Course (South)	7-3e	8.80	Aviation Support 7	Total .	132.35
Golf Course (South)	7-3f	17.24			
Golf Total		284.17	Vehicle Maintenance Yard	5-13a	53.84
			Food Distribution Warehouse	5-13b	11.75
Open Space	3-4a	0.78	1RWD Reservoir and Pumping	3-13c	5.73
Open Space	3-4b	0.53	Fire Station	3-13d	1.20
Open Space Total		1.31	California Air National Guard	3-13e	24.05
•			Homeless Service Providers	4-13f	28.29
County Habitat Reserve	3-5b	20.64	IRWD Facility	7-13g	9.00
Habitat Reserve Total		20.64	Fire Station	4-13h	1.00
			Public Facilities To		134.86
Restricted Open Space	8-6a	12.98			
Restricted Open Space	8-6b	4.64	Business Park	7-14a	6.24
Restricted Open Space	8-6c	68.83	Business Park	7-14b	13.41
Restricted Open Space	8-6d	32.45	Business Park	7-14c	34.07
Restricted Open Space	8-6e	11.42	Business Park	7-14d	33.05
Restricted Open Space	8-6f	16.29	Business Park To		<b>86.77</b>
Restricted Open Space	7-6g	38.02	DESILUDIA AS R. A.V.	····	00.77
Restricted Open Space	7-6h	9.14	Roads and Easements		167.30
Restricted Open Space Total	7*011	193.77	Roads and Easements	167.30	107.50
Restricted Open Space Total		173.77	Roads and Lasements	107.50	
Marshburn Retarding Basin	5-7	38.87	Ongoing Uses	2-2a	191.79
Retarding Basin Total		38.87	Ongoing Uses	2-2b	73.34
•			Ongoing Uses	2-16	81.53
Airfield Total	8-8	1,120.49	Ongoing Uses	3-15a	13.09
Airfield Total		1,120.49	Ongoing Uses	3-15b	29.09
		ŕ	Ongoing Uses	3-15e	42.96
Parking	5-9a	20.04	Ongoing Uses	3-15d	18.57
Terminal and Parking	1-9b	422.26	Ongoing Uses	7-15e	14.15
Terminal and Parking Total		442.30	Ongoing Uses	7-15f	20.22
			Ongoing Uses	4-12h	9.8
Airport Shuttle Bus Yard	4-10a	12.53	Ongoing Uses	4-12j	5.45
Airport Transportation Center	4-10b	9.34	Ongoing Uses	4-12i	6.5
Transportation Facility Total	7-100	21.87	Ongoing Uses	3-11b	56.08
Transportation Pacinty Total		21.01	Ongoing Uses	3-110 3-12e	32.40
			Ongoing Uses	3-12f	25.71
			~ <b>~</b>		8.16
			Ongoing Uses	2-12a	
			Ongoing Uses	2-12b	55.87
			Ongoing Uses	2-12d	8.01
			Ongoing Uses	5-9a	43.61
			Ongoing Uses	1-9b	58.84
			Ongoing Uses Tot	181	355.09
			GRAND TOTAL		3,721.95
Note: Code referen		Dian Signer			

Note: Code reference is to Land Use Plan figure.

Original Source: The Planning Center Number of acres shown is approximate. County of Orange Final EIR No. 573

Supplemental Phasing Analysis

# 11.2.1 Proposed Project at Phase 2 - OCX Aviation Facility Improvements

Draft EIR No. 573 presents the OCX aviation facility improvements that comprise the ASMP Proposed Project at Section 3.3.1, pages 3-6 through pages 3-16. To the extent such improvements under Phase 2 of the project differ from those described in Draft EIR No. 573 for Phase 4 of the project, such distinctions are illustrated below in an <u>double underline/italicized</u> strikaout format – <u>double underlined</u> text indicates additions; strikaouts indicate deleted text.

The Phase 2 project aviation component includes three four-runways, a passenger terminal complex, and supporting facilities for domestic and international passenger service, air cargo, general aviation, and aviation related uses. The passenger terminal complex will be located in Planning Area 1, the northeast quadrant of the airport site, with the primary entry [off SR-133 and the minor entry] at Sand Canyon Avenue and Trabuco Road (Figure 3-6) [unchanged from the original project description]. In addition to the multi-level terminal, this area will include a parking structure, hotel, and other passenger serving facilities, such as car rental facilities.

Phase 2 of the Master Plan improvements at OCX include the following:

- (i) Aviation Facilities and Infrastructure
  - (a) Runway Protection Zone (RPZ) Easements
  - (b) Backbone utility systems
  - (c) Airfield
  - (d) Lighting and navigational aids
  - (e) Terminal apron and remain overnight (RON) aircraft parking
  - (f) Terminal building area
  - (g) Terminal vehicle parking and support areas
  - (h) Terminal access and roadways
  - (i) Non-terminal roadways
  - (i) West cargo area
  - (k) East cargo area
  - (1) General aviation area
  - (m) In-flight catering area
  - (n) Air traffic control tower (ATCT)
  - (e) Airport rescue and fire fighting (ARFF) facility
  - (p) Aviation industrial and related area
  - (a) Airport maintenance facilities
  - (r) Aircraft maintenance area
  - (s) Airport transportation center
  - (t) Airport shuttle bus maintenance facility
  - (u) Fuel delivery, storage and distribution facility
  - (v) Ground support equipment (GSE)
  - (w) Environmental facilities
  - (x) Support systems, facilities and equipment

Phase 2 improvements are listed in Table 3-9A of this Supplemental Analysis, which provides the Planning Area location, phase of development, and the entity responsible for the development. A detailed description of Phase 2 and Phase 4 of the Proposed Project facility improvements is contained in the Airport System Master Plan for John Wayne Airport and Proposed Orange County International Airport (ASMP), December 1999. The ASMP also illustrates the locations and proposed configurations of the facility improvements.

#### 11.2.1.1 Airfield

Phase 2 of the Proposed Project will require the reuse and reconstruction of two existing MCAS El Toro runways (16R/34L and 7R/25L) and the construction of one two new runways (16L/34R and 7L/25R), each of which will parallel its existing counterpart 16R/34L. Construction of a new runways is required because the centerline separation between the both pairs of existing parallel runways is 500 feet, which is 200300 feet less than the FAA standard for conducting simultaneous visual operations by large aircraft (FAA 1989). Because the separation distance between the existing MCAS El Toro parallel runway pairs does not meet current civilian airport standards for independent simultaneous aircraft operations in clearweather conditions, the each new runway will be constructed at an offset distance from its parallel twin, sufficient to meet FAA requirements for operation under these conditions. The proposed airfield improvements at OCX would include the following:

- Runway 16R/24L will be reconstructed and extended at the north end of existing Runway 16R/34L will be extended by approximately 1,500 feet, and the south end of the runway by approximately 900 feet, to an overall length of approximately 12,400 feet.
- New Runway 16L/34R will be constructed at a centerline separation of 800 feet from existing Runway 16R/34L. The overall length of the new runway will be approximately 12,100 feet.
- To meet FAA criteria for new runway longitudinal gradient (slope), the central portion of existing Runway 7R/25L will be overlain with a new surface, its eastern end will be lowered, and its western end raised. In addition, the western end of 7R/25L will be extended by approximately 1,150 feet to a total length of 9,150 feet.
- A new Runway 7L/25R, 9,150 feet in length, will be constructed at a centerline separation
  of 700 feet from existing Runway 7R/25L, with a gradient similar to that of reconstructed
  Runway 7R/25L.
- For each runway, new parallel taxiways and associated exit taxiways will be constructed.

### 11.2.1.2 Terminal Building Area

The proposed terminal development will consist of a three-level main terminal building with four six-two-level pier concourses. The terminal will be served by a three-level road and curb system and a multi-level parking garage connected by pedestrian bridges and underpasses to the terminal.

Figure 3-7 illustrates the proposed terminal development. A conceptual rendering of the terminal area is shown in Figure 3-8. A conceptual cross section shown on Figure 3-9 illustrates the three-level terminal building and road system. The pedestrian bridge and underpass connection between the terminal and multi-level parking garage are also shown.

#### 11.2.1.3 Cargo Facilities

One Two-sites at OCX (one in Planning Area 3 and one in Planning Area 4), totaling 79 180 acres, has have been identified for air cargo use. Together The sites will accommodate 21 42 aircraft parking positions, 1,040,000 2,040,000 square feet of cargo buildings, 163 290 truck docks, 106 190 truck staging spaces, and 950 1,900 automobile parking spaces at Phase 2 completion. The proposed cargo facility arrangement offers extreme flexibility. The planned design features a contiguous assembly of cargo buildings in the west cargo area and two smaller buildings in the east cargo area. The modularized buildings will accommodate the great diversity of space needs of express, general cargo, and freight forwarding activity. The space will also accommodate United States Customs and the United States Postal Service offices. The building structures may be one- or two-story designs dependent on specific tenant needs for processing and storage areas.

## 11.2.1.4 In-Flight Catering Area

The site identified for in-flight catering at OCX (refer to [Draft EIR No. 573] Figure 3-1) is located at the northern corner of Planning Area 2. This site will be served by the East Access Road, which will intersect Irvine Boulevard near the site. It will also have a short and direct access route to the terminal area (on the service road) for catering trucks. The in-flight catering site will accommodate two three catering firms, with relatively large-scale operations typical of an international airport. Each site will be approximately 4.7 acres and can accommodate new flight kitchens as large as 100,000 square feet in two levels. Catering facilities will be developed by each of the tenants.

## 11.2.1.5 Airport Rescue and Fire Fighting (ARFF) Facility

The facility formerly housing the crash/fire/rescue operations at MCAS El Toro is located in Planning Area 4 near the intersection of the runways. The building encompasses 11,440 square feet, has six bays, and was built in 1957. The building is in good condition and will could serve as the OCX ARFF station through Phase 2 of project development. In Phase 3, a new ARFF facility will be constructed in Planning Area 2 on a four acre site at the intersection of the runways (refer to Figure 3-1). This new facility will have vehicle access from Irvine Boulevard

by way of the future East Access Road and secondary access streets. The ARFF station will be adjacent to the airport service road and will have direct access to all airfield areas, the passenger terminal, and other airport quadrants via two fire access roads and the taxiway system. The new ARFF facilities will include a 15,700 square foot building.

#### 11.2.1.6 Aviation Industrial and Related Area

The site identified for aviation industrial and related facilities at OCX is located south of the inflight catering area in Planning Area 2. The 43 102-acre area will accommodate RON needs through Phase 2 (5 positions 16 acres), a ten-acre aircraft maintenance site, the ATCT site (four acres), and a variety of potential aviation industrial uses. (72 acres).

#### 11.2.1.7 Airport Maintenance Facilities

The long-term site for airport maintenance and facilities management will be at the east end of Planning Area 3. Existing office buildings and shops in Planning Area 2 will be used for airport maintenance during Phases 1 and 2.

#### 11.2.1.8 Aircraft Maintenance Area

The primary site identified for aircraft maintenance facilities at OCX (14 45 acres) is located west of the airport maintenance area in Planning Area 3. A second smaller site (10 acres) is located in the aviation industrial and related area at the north end of Planning Area 2. The proposed primary aircraft maintenance site will accommodate one facility through Phase 2, facilities, and the smaller site will accommodate one facility through Phase 2.

## 11.2.1.9 Airport Transportation Center

The Airport Transportation Center (ATC) will be constructed on seven acres in the southeast corner of Planning Area 4. This site is adjacent to the Southern California Regional Rail Authority (SCRRA) tracks, diagonally across from the Irvine Transportation Center (ITC). The ITC is a multi-modal transit facility providing an interface between rail and bus patrons. Air passengers arriving at the ITC (via public transit, Amtrak or Metrolink) will cross the railroad tracks by means of a pedestrian overcrossing. Shuttle buses operating approximately every five minutes will transport passengers between the ATC and the terminal complex.

## 11.2.1.10 Fuel Delivery, Storage and Distribution Facility

A bulk fuel storage facility will be located on an 7.7 44 acre site in Planning Area 4. The site will have a truck unloading area with 6 44 unloading positions (plus additional space for truck staging and spare positions as needed). This fuel storage site would also include three pumps to distribute fuel from the truck unloading positions to the bulk storage tanks, an operations/pump control building, and a building for spill response and supplies. An additional pump and appurtenant pipelines would also be provided for spare pumping capacity as needed. Since fuel delivery by truck would operate on a daily basis, a four to five day fuel supply will

be needed. Three Six-100-foot diameter, 30-foot high, aboveground storage tanks will be provided for a five day supply of fuel in Phase 2.

#### 11.2.1.11 Ground Support Equipment (GSE)

The 4.6 14-acre site identified for ground handling facilities is at the western end of Planning Area 4 adjacent to the parallel taxiway serving Runway 16L/34R. The site will accommodate one three ground support equipment (GSE) facility facilities. West of this site is a proposed location for an electrical substation to serve the GSE. To enhance air quality, GSE operators will be provided with facilities that accommodate vehicles powered by alternative fuels, such as electricity or natural gas.

## 11.2.2 Nonaviation Revenue Support Uses at Phase 2

Portions of the base are planned for various nonaviation uses that have been selected because, in addition to providing economic, social, and recreational benefits and opportunities to Orange County, they are compatible with the planned commercial aviation use. In some cases, these nonaviation uses provide revenue support to the civilian redevelopment of the base and the development and operation of a commercial airport at El Toro. The nonaviation revenue support uses developed through Phase 4 build out of the project are described in Draft EIR No. 573, Section 3.3.2.

At Phase 2 of the project, the nonaviation revenue support component of the plan includes approximately 1,000 acres on the airport site to be used primarily for a variety of open space, recreation, and agricultural uses. Various public use facilities and a small business park are also included. All of these uses are described in Draft EIR No. 573, Section 3.3.2. A specific listing of the Phase 2 aviation and nonaviation land uses and acreages associated with these uses is included in Table 3-1A.

With respect to nonaviation revenue support uses, there are two primary differences between the Phase 2 and Phase 4 development levels of the Proposed Project. First, at Phase 2 of the Proposed Project, the Regional Park is not yet fully developed; construction and development of the park are scheduled to be completed in Phase 4 (compare Draft EIR No. 573 Table 3-9 and Supplemental Analysis Table 3-9A). The second primary difference is that at the Proposed Project Phase 2 level, the Wildlife Habitat Area is not yet developed. Like the Regional Park, the Wildlife Habitat Area is scheduled for development in Phase 4 (compare Draft EIR No. 573 Table 3-9 and Supplemental Analysis Table 3-9A).

With these two exceptions, the description of nonaviation revenue support uses at Phase 2 is substantially similar to the Proposed Project at build out.

#### 11.2.3 Base Transition Plan at Phase 2

The BTP allows for some interim reuse of existing buildings and facilities at El Toro until implementation of the ASMP reaches a phase that requires the demolition of a specific facility, or its conversion to the permanent reuse contemplated by the ASMP. The buildings that will be reused are within PAs 1 through 4, primarily for interim business park or warehouse use. In PAs 3, 5, and 7, agriculture will also continue as an interim use on 586 acres. The BTP does not provide for interim leasing of buildings for residential uses. The Base Transition Plan, as developed through Phase 4 build out of the project, is described in Draft EIR No. 573, Section 3.3.3.

Draft EIR No. 573 Table 3-2 lists the nonaviation buildings/facilities that the BTP has identified as suitable for interim nonaviation reuse and indicates the planning phase in which each facility would continue as an interim use. The majority of these facilities is located in Planning Area 1 and would be discontinued during Phase 1 of the ASMP. A small number of facilities would continue as interim uses into Phases 3 and 4 of the Project until required for ASMP implementation. At Phase 2 of the project, those BTP uses planned for Phases 3 and 4 would not yet be developed.

#### 11.2.4 Interim Aviation Uses at Phase 2

Existing buildings that are suitable for civilian aviation use are listed in Draft EIR No. 573 Table 3-3. While some are available only for the short-term period before Phase 1 construction is completed, many could be used throughout much of the 20-year planning period. These interim aviation uses would be unaffected at Phase 2 of the project.

The initiation of commercial passenger service, air cargo service, and/or general aviation could occur during Phase 1 but before the Phase 1 improvements are fully completed. During this interim period, existing facilities would be used, perhaps with installation of some temporary facilities.

## 11.2.5 JWA Facility Improvements for the Proposed Project at Phase 2

Draft EIR No. 573 Section 3.3.5 describes the JWA Facility Improvements for the Proposed Project. As indicated by Table 3-9A, all JWA improvements are scheduled for construction and development during project Phases 1 and 2. Therefore, at Phase 2 all JWA facility improvements scheduled to occur under the Proposed Project will be completed.

TABLE 3-9A
"EMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

Description of  posed Project Improvements  (Through Phase 2)	Phase 1	Phase 2	Phase 3	Phase 4	Responsibility	Planning Area
OCX IMPROV	EMENTS	1	PARAMETER STATE OF THE STATE OF	1		3
xisting Base	All	:	***************************************	·	County	
xisting Base	All	1	<u> </u>		County	**
n collector s for PA2	22,500 linear feet of main	Additional 22,400 linear feet collector pipelines & sewer	and the second s		County/Utility Provider	1-4
**************************************	collector system  36,100 linear feet of main	laterals for PA2 and PA3	1		County/Utility Provider	1-4
o IRWD line and	to Irvine Ranch Water District	Additional 15,000 linear feet of distribution pipelines and 21 fire hydrants	1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	:	ı 	
ions, and 1 TIC	31,800 linear feet of main distribution pipelines, 41 control stations, and TIC well	Additional 23,800 linear feet of supply pipelines and 35 control stations	700	ı	County/Utility Provider	1-4
n and stem, and s	35,400 linear feet of main and secondary distribution lines and gas main relocation	Additional 19,400 linear feet of main and secondary distribution pipeline		1	County/Utility Provider	1-4

## TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

Improvament	Map Reference	Description of Proposed Project Improvements (Through Pieses 2)	Phone 1	Phoe 3	Plane 3	Plane 4	Steepe and Sity	
			,28,900 linear fact of main	Additional 20,300 linear feet of		!	County/Litting	4
•		49,200 linear of main backbone liters, new		backboar Times, substation		į	Presiden	
	1		GSE area, and 24 of 400Hz power					
3. Electrical Power Backbook System [c]	,	bowe systems	rystem	400ffz power systems		: _	1	_
							County/Littley	
<ol> <li>Communications Backbone system/Wide Area Network</li> </ol>		<ol> <li>160 linear feet of backbone supply lines and new MDF facility.</li> </ol>		Additional 20,400 linear feet of		i	Provider	1
VSE MEANE		128,330 feet of storm drains, 7,240 trench	supply times and new MDF facility	Deckerme supply trace			County/Dality	÷
		drains, 4 oil-water separations, 43 junction	Ĭ	57,133 front of storm drains.			Provider	i
i i		structures, 268 membels and 117 catch	71,197 feet of storm drains, 2,350			i .	Tuch Kites	-
		having for the Marshburg Changel, Bee	prench drains, I pil-water separator				1	
•		Carryon, Agus Chinon and Borress		structures. 95 manholes and 191		•		
7. Storm Dramage Backbree system		Symmu		catch basins			1	ì
	i-		21,500 liscar feet of fiber agtics			:	Coursey/Utility	-
	'	37,500 linear feet of fiber optics that	feeder and consist cable backbone	Additional 16,000 linear fact of			Provider	1
8. Catric Television Backbreis system	-	couxiel cable backtone lines	lines	couxin) cable buckbone fines			1	1
C. AIRFIELD								
1. Site Demosition		\$77,322 cubic feet of buildings/		513,524 cubic feet of			Country	1
•		structures; 994,574 square yards		building/structures; 191,236			1	
		of AC pavenents and 659,601	square yards of AC parespools	square yards of AC pavements			1	
		square yards of PCC peveracets	and 437,710 square yunits of PCC	and 221,891 square years of			Į.	1
		for Rumways, Taxiways, Aprosis.	per-centure.	PCC pevernesss.	4 warmen			۰. ا
2. Site Grading		Cut 2,569,445 yurds; Filt 6,901,921 cubic		Cut 2,486,243 cubic yards; Fill ; r5.756,647 cubic yards			County	
3. Site Difficies (d)		yards No further description	1,145,234 cubic yants Partial	Partiel			Constv	ý
-3, Jac transmis (4)	4-4	SAN INTERNATIONAL CONTRACTOR	(#U4)	raus ;			; CAMINEY	

[c] Includes 400 Hz for cargo and RON areas. Terminal 400 Hz power systems are included under Terminal Building Area

[d] Includes fire hydrants.

## TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN(s)

Emprovement	Map Reference	(Through Photo 2)	Photo 1	Phon 2	Phase 3	Place: 4	Respondently.	
4 Reconstruction and Langthening of Russiay 16R-34L	1-4	12,460 linear feet; 3,247 loss of paying overlay	Reconstruct existing 10,000 feet; 3,247 tons of paying overlay	Construct 2,400 feet extresion.			County	
Reconstruction and Lengthening of Rusway 7R-251.,     Statening the     Condicat	, H	9,150 linear feet		All			County	1
6. New Ranway 18134R		12,100 linear feet, 800 feet separation cast of Russeav 168-24L		AU	****	-	County	i
7 New Parallel N-5 Taxiway on Wast Side of Airfield (Taxiway A)		137,414 square yards of PCC paving: 11,817 linear feet lighting	112,182 square yards of paving and 9,687 linear feet lighting	25,232 square yards of paving and 2,130 linear feet lighting	***************************************		County	·
New Parallel N.S Taxioray on East Side of Airfield (Taxioray C)	8-8	106,526 square yards of PCC paying		106,526 square yards of paving and 11,963 Sacur free lighting		·	County	<u>.</u> I
9. New Parallel E-W Taxiway on North Side of Airtheld (Taxiway B)	14	81,997 square yards of PCC paving and 8,778 linear feet lighting		\$1,997 square yards of PCC paving and \$,775 limins fort lighting		1	County	
10. Stant Pads - Bustways 34,16,7,28	M	88,889 square yants Apphalt Courtese	24,889 square yards Asphalt Concrete Paying	44,444 square yards Asphalt Concrete Paving			County	
II. Exit Textiveys for Russways 16-34	A	190,724 square yearts of PCC puving	MAJIS square yerds of PCC preving	151,280 square yards of PCC			County	Ī
12. Exit Taxiways for Ramways 7-25	3-8	37.217 square yards of PCC paving		37,217 square yards of PCC			County	<b> </b>
13. Airport Service Road, highling, signing & striping		37,500 linear feet of restricted scores read	17,800 linear feet of restricted access road	19,700 linear feet of restricted			County	-
14 Blass Fences	Li	2,044 Instar feet Russway 16R-34L; 2,793 Instar feet Russway 16L-34R	2,044 linear feet Runway 16R-34L	2,793 linear feet Runway 16L-34R			County	-

#### Table 3-9A Summary list of improvements at OCX and JWA recommended for Phase 2 of Africat System master Plan[a]

	Engrerotaux.	Map Reference [h]	Description of Proposed Project Improvements (Through Phase 2)	Phon 1	Pier 2	Phon 3	Page 4	Responsibility	
D. LIGH	TING AND NAVIGATIONAL AIDS			······································					
	1. Site Demolitina	1.2	included in Airfield	Included in Airfield	included in Airfield		:	County	Ι.
	2 Site Grading	1-1	Included in Airfield	Included in Airfield	Included in Airfield			County	i.
	3. Sinc Unificier	1-8	Included in Airfield	Included in Airlieid	included in Airlield		,	County	-
	4. High intensity Edge Lighting for Runway 16R-14L		12,270 linear fees - both sides of nursusy	9,853 linear foot - both sides of runway	2,417 linear feet - both sides of narway		i	County	-
	5. High intensity Edge Lighting for Runway 161-34R		11,675 linear feet - both sides of runway		11,675 linear feet - both sides of runway			County	-
	6 High Intensity Edge Lighting for Runney 78-25L	1-4	9,156 linear feet - both sides of runway	:	9,150 linear feet - both sides of narway		;	County	i
	7. Medium Intensity Edge Lighting for all Transveys	1-1	51,374 linear feet - book sides of laximary	12,137 innear free - both sides of literiway	39,237 timent feet - beth sides of taximay		<u> </u>	County	
	S. Runway Signage		Covering 33,095 fort of Runway	Covering 9,853 feet of Runway	Covering 23,242 feet of Runway			County	
	9. Taxiway Signage	1-1	,Covering 51,374 feet of textiring	Covering 12,137 feet of textively	Covering 39,237 feet of textway		ł	County	
** ****	10 Approach Light systems for Runway 141.	\$-8, \$-6c, 8-6f, 7-6g	MA1.SR	All	1	« <i>,</i>		FAA	
	11. Approach Light systems for Recoverys 34R	8-8, 8-6c, 7-6g	:MALSR		Ali		,	FAA	•
	12 Processon Instrument Apparaich systems for Resembly 34L		ILS Category I	All				FAA	*****
	Procision festiment Approach systems for Rusresy 348	1-1	ILS Category I	, and the second	All			FAA	
	14 Airport Servelliance Redar (ASR)	6-5a	ASRII	All	·		,	FAX "	
_	(AWOS)	5-8	, No further description	All		, -		County	
	16 Rutiway Visual Range (RVR) Equipment- Rutiways 141.834R	. 14	No further description	AJI				County	
~ ****	17. Auport Rosating Beacon 18 Supplemental Wind Cones		No further description	NI -	· · · · · · · · · · · · · · · · · · ·			County	-
	19. Airport Segmented Circle	· • • • • • • • • • • • • • • • •	No firsther description	TAR	<u> </u>	······································	ļ	County	

# TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

Improvement	Map Reference (b)	Description of Proposed Project Improvements (Through Phom 2)	Phone I	Plant 2	Phase 3	Phone 4	Responsibility
20, Airfield Lighting Vault(s)	1-1	No further description	No description	No description	ļ		County
21, Very High Frequency Omni-Range/ Distance Measuring Equipment (VOR/DME)	6-5a or 7-1e	No further description	All				FAA
22 Precision Approach Path Indicator System (PAPI) -Ranways 34L&34R	1-4	For runways 34L and 34R	For narway 34E	For narway 34RL	L. 1988. 18		FAA
23. Precision Approach Path Indicator System (PAPI) -Russways 16L&16B	14	For runways 16L and 16R	For narrowy 16R	For running 16L			FAA
24 Precision Approach Path Indicesor System (PAPI) -Runway 7 R	##	For runway 7R	t	For nativery 78.			FAA
		and 19,444 square yards of POC payments.	building/structures; 21,560 against yards of AC pevernenes and 19,44 PCC aguare yards of provenents.		,		
2. Site Grading		Cut 78,200 cubic yards; Fill 4,303,269 cubic yards	Fill 1,65%,187 cubic yards	Cut 76,200 cubic yards; Fill 2,645,682 cubic yards	per a december of controller		County
3. Sike Utilisies			Included in Airfield	behided in Airfield	L		County
4. Terminal Aprim		\$79,200 square yards Portland Cement Concrete Paying and lighting	395,391 square yards PCC paving and lighting	483,809 square yards PCC pring and lighting	·		County
7. Hydraul Fuel system of Terminel Aproc	1-96	44 jet gates; 53 fleet hydrant pics; 16 isolation valve pict; 26,400 linear feet 10° distribution pipeline and 8,400 linear feet 4" connection pipeline: 2 leak detection systems; 4 hydrant servicer vehicles:	21 Jet Cates; 24 fuet hydrant pits; # isolation valve pits; 22,000 linear feet 10" distribution pipeline; 3,600 linear feet	23 let Guies; 29 fuel hydraut pits; 8 inolation valve pits; 4,400			County/Tenum
6. Remote Overright Parking in Terminal Area	1.96	16 RON parking positions		16 positions			County

,		

#### TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLANIN

Імручченин	Map Reference	Description of Proposed Project Improvements (Through Place 2)	The state of	Plant I	Passe ) Pas	ase 4 Requestibility	Ple
7. Remote Overnight Parking in Planning Area 2		5 RON parking positions	1	!5 positions	1	County	
8. Terminal Apren Area Lighting	, 1-9h	No further description	No description	No description		County	4
9. Terminal Taxilane Lighting	1-95	No fasher description	No description	No description		County	
ERMINAL BUILDING AREA							
1. Site Demolition (includes demolition of	1-96	3,675,482 subjectives of building/	2,466,426 oubse feet of	1,209,056 cubic feet of	;	County	,
terminal vehicle parking and support sees.)	1	structures; 48,989 square yards of	building/structures; 29,860	building/structures; 19,129		i	i
4		AC pavements.	square yards of AC pavements.	square yards of AC perements.			
2. Site Grading (quantities includes terminal	1-96	192 Acres, Cut 7,666,139 cubic yards; Fill-	Cut 5.091 944 4.081,044 cubic	Cut 1,714,195 cubic yards;		County	_
vehicle parking and support areas)		306,798 cubic yards	yards; Fill 255,889 cubic yards	Fill 50,909 cubic yards	;	1	1
3. Site Utilities	1.96	192 Acres	192 Acres			County	7 ~
4. Terminal Building [c]	1-96	1.44 million square feet	,520,000 square feet	919,700 square feet		Conmiy	-
3. Central Utility Plant [f]	1-96	To support 1.44 million square feet	To support 520,000 square feet	To support 919,700 square		County	-:
		of terminal building	of terminal building	feet of terminal building			i
6. Personger Loading Bridges	: 1-96	64 jet guice	21 Jet Getes	23 Jet Guies	·········· , ·······	County	
7. Pedestrian bridges to parking structures	!-96	2 bridges	1	1		County	, ,
2. Baggage Handling systems	1.96	16 baggage handling systems	7 systems	9 systems		County	;
9. Communications and Display systems		44 systems	21 systems	123 systems		Country	
10 Preponditioned Air and 490 Hz Perver			To support 520,000 square feet	To support 919,700 square fort		County	
		of terminal building	of terminal building	of terminal building			
11. Landscaping		37.9 acres	25.3 scree	12.6 scres		County	
12. Moving Sidewalks	1-96	4,000 linear feet	<del></del>	4,000 linear feet		County	
ERMINAL VEHICLE PARKING AND SUPPORT AREAS		,		-		· · · · · · · · · · · · · · · · · · ·	<u></u>
1. Size Culidies	1-94	No further description	Included in Terminal Building	Included in Terminal Building	:	Country	L
2. Short-Term Parking Structure			3,700 Spaces	3,600 Speces		County	
3. Long-Term Surface Parking in Terminal Area (Lot A)	1-96	40 acres; 4,940 apaces	3,700 Spaces	1240 spaces		County	

48 tradudes Decembricans & is and 440 Mr. Brown

	ē		

### $TABLE \, 3-9a$ Summary list of improvements at ocx and JWA recommended for Phase 2 of Airport system master plan[a]

Zeopravironiste	Map Hefacence [h]	Frequent Frequent Improvements (Through Phone 2)	Plant 1	Phote 2	Plane 3	Plane 4	<u>Perpositity</u>
4. Remote Surface Parking N. of Irvine Bivd (Public &	1-96	22 acres (approx.); 2,660 spaces (1,760	,	2,610 Spacer ( 1,760 public			County
Employee)		public spaces & 850 employee spaces)		spaces & #50 employee spaces)	i		
5. Employee Parking	1-96	22 scres; 2,750 spaces	t,700 Spaces	1,050 Spaces			County
6. Rental Car Ready Return and Service Area Facilities	1-96	34 acres	22 Acres	12 Acres			Tenant
7. Hosei Familiaca	1-96	500 mons	All				Torrect
III. Lighting	1-96	No further description	Included in Terminal Building	Included in Terminal Building			County
9. Landscaping	1-96	No further description	Included in Terminal Building	Included in Terminal Building			Country
10. Simulia Blog Facilities and Equipment	3-9b, 4-10a,	1 Pedestrian bridge & ITC station;	I Pedestrian bridge & ITC	· 2 bus shelters			County
<u> </u>	3-9x	5 bus shelters	station, 3 bus shelters	······································			
AL ACCESS AND ROADWAYS						······	
1. Site Demolition	1-96	31,396,697 cubic feet of building/structures	27,295,936 cmbic feet of	4,100,761 cubic feet of			County
}		491,337 square	building/structures; 491,537 equa	refbuilding/structures.			
1		yards of AC Pavements.	yards of AC Pavessents.	I			L
2. Sise Drainage State Rouse 133 (SR133) -	1-96	9.370 linear feet of drainage improvements		9,370 tincer feet of drainage			County
(Grading for other terminal roads has been				improvements			
included under Terminal Building Area)		i	_				
13. Sine Oulities - (SRJ33)	1-95	9,370 limes feet (SR-133 Lighting.		19,370 linear fors (SR-133		-	County
		Signing & Striping)		(Lighting, Signing & Striping)			
14. SR-133 Auxiliary Laues, Diverge laues;	**	Northbound and Scuthbound lanes		12.870 linear fact (1 tane);			County
		on SR-133 from 1-5 to arroom ratios		49,480 session feet of retaining			
Merge Lanes		On OK-135 trous 1-3 to support ramps		(angle:			

# TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCK AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

lasperretacat	Map Reference  bj	Description of Proposed Project Improvements (Through Phase 2)	Phite I	Phone 2	Photos 3	Phon 4	Bergmanitellity	n
5. SR-133 North Bound (NB) off-remps and South Bound (SB) on-samps	<del> </del>	northbound SR-133 to simpore, and sixport to nounbleowed SR-133	:	3,400 linear feet. @ grade + 159,400 square feet structures 3 issue NE cell-recep; 5,600 square feet Resain. Walt; 1,700 linear feet @ grade + 90,200 square feet structures 3 issue 58 on-resis.	The second secon		County	,
6. Trabuez Road (Sand Canyon to Hotel Site)		Widon to 4-lane primary; 2,199 linear feet	Widon to 4-lane primary; 2.190 timour fort				· County (Parsint)	_
7. Terminal Loop Routhway (surface & clevated)	1-76	) to 6 lanen at surface - 9,300 feet long; elevated at 45,200 square feet	B00 linear fort (1 Lune); 5,800 linear fort (2 Lune); 1,000 linear fort (4 Lune); 300 linear fort (5 Lune); 800 linear fort (6 Lune); 14,400 square fort (6 Lune Bridge)	20,800 square for: 3 lane bridge over RAC area.			County	to the section work in the
8. RACAT Parking Perimeter Spine Road	1-96	4 lane+median: 7.800 linear free	4 imermedian, 2,800 fines feet				Concry	<del></del>
9. Ferminel Way		4 Lanct + Median; 1,050 linear feet	4 Lames + Medium, 1,050 Bocar Sout		i		Councy	1
.10. Terminal Curbaide - Lower level (Countercial)		14,400 square fors Recirculations Ramp from Curbaide; 2,200 linear fort Terminal Curbaide; 48,000 square fort tackeralk.	14,400 square for Recirc. Ramp from Curteside; 2,300 Tecminal Curteside; 48,000 square fors sidewalk.	•	,		County	
11. Terminal Curbaids - Middle level (Activals)	•	14,400 square foot Recirculation Ramp from Curbside; 2,300 linear foot Termissal Curbside.	14,400 square fort Re-circ Ramp from Curbside; 2,300 linear fort Terminal Curbside.	·	:		Consty	;
12 Terminal Curbaide - Upper level (Departures)	1-96	14,400 square fort Recirculation Romp from Curbride; 2,300 linear foot Turnisal Curbride	114,400 square feet Re-circ. Ramp from Carbaide; 2,300 linear feet Terminal Carbaide.	· · · · · · · · · · · · · · · · · · ·	*		County	!

# $TABLE \ 3-9a$ summary list of improvements at ocx and \$\mu\$ recommended for phase 2 of airport system master plan[a]

Improviousi	Map Reference [b]	Description of Proposed Project Improvements (Through Phase 2)	Phot 1	Phine 2	Photo 3	Phone 4	Sequentity	-
13. Perimeter Road around ST parking structure	1-96	2 lanes 2,400 linear feet	2 faces 2,400 linear feet	-		<del> </del>	County	
14. Employee Parking Road (4 lane)		(2,900 linear feet (2 additional lanes + imedian)	1,450 linear feet (2 additional dance + median)	1,450 linear fees (2 additional lanes + mediae)		*	County	
13. Terminal Service Roads (2 lane)	1-96	4,200 listear feet Road (2 Lame Grade)	4,200 linear feet Road (2 Lane (Grade)	······		<b>-</b> .	County	
16, Traffic Signals	1-36	6 cach	4 cach	2 cach		<u>-</u>	County	
117. Phase I Interim Loop Road (3 lanes)	(-96	1,000 linear feet	1,000 timese feet				County	
18. Phase I Inscrim Ramp to Upper Level (2 lanes)	1-96	600 linear feet	600 linear feet				County	Agr
19. Phase I Interim Ramp to Lower Level	1-9b	600 linear feet	600 Jincar Feet			:	County	
NON-TERMINAL ROADWAYS	***************************************							
<ol> <li>Interior Marine Way Widening (Sand Carryon to Emp. Parking Rd)</li> </ol>	1	2,450 lidear foct	*2,450 linear feet	,			County	·
Air Cargo Way (Sand Carryon to Errop. Parking Rd)	1	2.784 linear feet	1	2,784 lines feet			County	-\$
3. Air Corpo Way (Emp. Parking Rd to Planning Area 3)	3 ***	11,720 linear fort		, 11,720 times feet			County	
4. Air Cargo Way (Planning Area 3)		4.160 lissear for:	-	4,160 linear Sect	<del></del>		Councy	
15. Astor Expension (Exist storm channel to "Z" Street	-	2.140 linear feet	1	2,190 linear foot		-	County	
6 'Z' Street widening	***	1.410 linear fact	···	A10 linear feet			County	
7. Planning Area 2 Access Roads	Processor and the common and the com	6,250 linear feet	6,250 linear feet	<del></del>	-		County	·-
WEST CARGO AREA							<u> </u>	
Site Demolition		24,831,458 cubic feet of building/structure 221,602 meaning yards of AC parecriteits and 225,957 square yards of PCC convenients	is: 24.333,232 cubic feet of building/structures: 148,699 segure yards of AC pavernessis and 116,022 feature yards of PCC paverness.	*546,256 cubic fort of building/structures; 72,903 1, aquate yards of AC paveracuts (and 109,835 square yards of (PCC pavements.	,		County:	
2. Site Grading and Dramage	4-114	79 Acres, Cut 144,466 cubic yards: Fill 1,462,039 cubic yards	,52 Acres; Cut 111,383 cubic yards Fill 999,089 cubic yards		<del></del>		County / Tenans	*

#### TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[s]

Lasprovenacas	Map Reference [b]	Description of Proposed Project (supercements (Through Plane 2)	Pleage 1	Plange 2	Flore )	Place 4	2-,-42	e Pie
3. Site Utilities		:79 Acres; 10,000 Timem Feet		27 Acres; 3,000 limenr feet curb	i		County / Tenn	wil
		Curb & Gutter		# gutter				
4. Cargo Tuxiweys and Aprop		21 account parking positions plus 100,000		# hardstands plus 50,000 square			County / Tent	HOL .
		square feet suziliary ramp; 170,108 square words concrete whilabilized base.	square feet qualifiery Ramp; .68.625 square yards concrete	fixes movillarry Rampy; 109,483 smaller vands concrete				1
		yards constant washanded base.	w/stabilized bear	w/minbitized base.				
5. Cargo Apron Hydrans Fucling system	411a	i I find hydram pits; Emolation valve	á faci hydrantsjá isolation valve	5 fuel hydranes;2 isolation valve			, County / Team	-
		pits: 14,500 linear feet 10" carbon	pits; 11,000 linear fort 10"	pits; 3,000 linear feet				1
		secol pipe and 2,700 limese foot	carbon sacci pipe; 1,200 linear feet				i	,
		4° carbon steel consentrice: pipe;	4" carbon steel branch connection					
		I leak detection system; 2 hydraus		steel branch connection pipe;				
		savicor vehicles.	detection system; I hydrani pervious vehicle	I hydrani servicer vehicle				1
6. Cargo Apron Lighting		89 screw (approx.); 21 of 400 Hz Power	12 of 400 Hz Power systems	9 of 400 Hz Fower systems			County / Tesa	-at
7. Cargo Warehouse / Office Building [g]	4-110	1,040,000 aquers for: bidg;	620,000 square feet hidg;	420,000 square for bldg;			County / Tees	
8 Cargo Area Entrance Roadways	4-116	61,200 square yards Asphalt Concrete	Imuranor: 34,200 square yards	, I entrance; 27,000 square yards			County / Team	
3		Pering		asphalt concrete				Í.
<ol> <li>Cargo Area Truck Docks and Staging Area</li> </ol>		163 truck docks plus 106 truck steging		85 truck docks pins 55 truck.			County / Team	est :
		spaces; 16,600 square yards oneonete. for	staging spaces; 2,950 square yards					1
		docks/truck staging		yanda concrete duckwimick staging				
10. Cargo Area Vehicle Parking	4116	950 spaces		473 spaces			County / Tour	<del></del>
11. Landscaping		55,080 square feet Landscaping		24,300 square fort			County / Tena	

ig) Would accommodate cargo carriers, U.S. Postal Service, and U.S. Custame.

### TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

Size Demolítice   3-11b   1,995.334 aquare yards of AC percencist and 997,880 aquare yards of AC percencist and 997,880 aquare yards of AC percencist and 997,880 aquare yards of AC percencist and 997,880 aquare yards of AC percencist and 997,880 aquare yards of AC percencist and 997,880 aquare yards of AC percencist.    Size Demolítico	(seprovement	May Reference [6]	Description of Proposed Project Improvements (Through Plants 2)	Pleaser 8	Ps 2		Plane 4	Responsibility
L. GENERAL AVIATION AREA  1. Size Demolition  2-12d  3.268.372 cubic feet of building/structures; 250, P40 cubic feet of building/structures; 808 square yards of PCC preventeds.  2. Size Canding and Drainage  2. Size Canding and Drainage  2. Size Canding and Drainage  3. Size Utilities  2. Size Canding and Drainage  3. Size Utilities  3. Size Utilities  3. Size Utilities  4. 12d Sources  4. 12d Sources  4. 12d Sources  4. 12d Sources  5. Size Canding and Drainage  4. 12d Sources  5. Size Canding and Drainage  6. Size Canding and Drainage  6. Size Canding and Drainage  7. Size Canding and Drainage  8. Size Canding and Drainage  9. Size Canding and Drainage  1. Size Canding and Drainage  2. Size Canding and Drainage  3. Size Canding and Drainage  3. Size Canding and Drainage  3. Size Canding and Drainage  4. Size Canding and Drainage  5. Size Canding and Drainage  6.  K. EAST CARGO AREA								
1. Size Demonition	T. Sate Demofition	3-116	and 997,\$80 square yards of PCC	prevenuents and 585,023 square	presents and 432,857 square		all controls	County
# 45,668 square yards of AC percentents & isulfong/structures; 808 square yards of AC percentents.    3   131 square yards of PC percentents.   yards of AC	L. GENERAL AVIATION AREA			<u>-i</u>				·
3. Size Utilizion	i Size Demokidon		45,068 square yards of AC personents &	building/structures; \$08 square	building/structures; 44,260 aquate yards of AC preventoria & \$1,131 aquare			County
4. Acoes Racebusy Encursors & Vehicle 7-12d 3 communes; 11,600 square yards One entrance Two entrances Courty / Tensart Parking 5. Hangar & Office (FBO) 2-12d 50,000 square feet bldg 50,000 square f		2-12d						
Parking   A Caparing   A Capa	3. Site Utilities				Management of the second of th	I		
1.     1.   1.   1.   1.   1.   1.	Parking		AC paving	One entrance	Two cotronoes		•	County / Tenant
7. General Aviation Tachways and 2-12d 54,129 square yeards Portland Cement Concrete Concrete Concrete Ramp  R. Portland Cement Concrete Paving Transient 2-12d 17,361 square feet Prefact Cement Concrete Paving 7. 129 square yeards Parking (Nov-Tennat Cook) 2. Concrete Paving 1 17,361 square feet Prefact Cement Concrete Paving 1 17,361 square yeards 2-12d Concrete Paving 1 19 General Aviation Fueting Facility 2-12d Concrete Paving 1 1,361 square yeards 1 19. Concrete Paving 1 1,361 square yeards 2-12d						<del>, , , , , , , , , , , , , , , , , , , </del>		Tenant
Ramp   Concerts Paving   Paving Familien   Paving 54, 129 aquient yards							L	
R. Portland Cement Concrete Paving Transient 2-12d 17,361 square Fort Fort fand Cement Parking (Non-Tennat Cest) Concrete Paving Parking (Non-Tennat Cest) Concrete Paving Parking (Non-Tennat Cest) Concrete Paving 19 General Aviation Feeting Facility 2-12d One All Tennat 10 Aircraft Washing Facility 2-12d One All Concry / Tennat	7. General Aviation Taxiways and	2-124		:			•	County / Tenant
Parking (Non-Tennet Code)         Concrete Paving         Paving 17.361 square yearfs           19 General Aviation Fueling Facility         2-12d         One         All         Tennet           10 Arenth Washing Earliny         2-12d         One         All         Coorey Tennet	Ramp	Y MARCHAN & Consequences on		<u>+</u>		! *		
19 General Aviation Fueling Facility 2-12d One All Tenant   10 Advantil Washing Facility 2-12d One All Concey / Tenant						,		Const
IO Aircraft Washing Facility 2-12d One Aft Country / Toners				*** ******** ********	Paring 17,361 square yards		***************	. <u> </u>
				<u>,AB</u>	-1.0	<b>*</b>	+	
	II Landscaping		No further description	No description	Ne description	r	1	County / Tenses

### $TABLE \ 3-9A \\ SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a] \\$

Songerum consent	May Reference (b)	Description of Trapand Project Improvements (Through Phase 2)	Plante I	Phone 3	Phone 3	Phase 4	Responsibility	,
IN-FLIGHT CATERING AREA								
1 Site Demoistion	2-12a	9.3 acres; 2,045 square yards of AC paveneous	4.6 Acres; 1,045 square yards of AC pevernous	4.7 Acres; 1,900 square yards of AC pavessesses			County	
2. Site Grading and Drainage	2-126	Cut 54,448 cubic yards: Fill 67 cubic yards	Cut 27,007 cubic yards; Fill 16 rubic yards	Cut 27,361 cubic yards, Fill 51 cubic yards	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		County / Tensor	
3. Site Unities	2-124		5.0 acres	3.9 acres			County / Tenast	•
4. Access Roadway Entrance & IFC Vehicle Parking	2-124	I entrance; 19,264 square yards Asphala Concrete Paving	1 entrance; 12,216 square yards Asphalt Concrete Paving	6.968 square yards Asphalt Concrete Paving			County / Tenant	
J. Service Read Entrances	2-12a	1 entrance	1 coirance	,			County / Texas	<del></del>
6. Emildings and Other Tenant Improvements	2-12a	200,000 square feet bldg	100,000 square feet	100,000 square foot			Tenant	<del>,</del>
7. Lanchscaperig	2-12a	19,204 square feet Landscaping	12,216 square fort	6.988 square feet.			Tenan	
AIR TRAFFIC CONTROL TOWER/TRACON COMMUNICA	TION EQUIPMENT							
31. Size Demotition	2-12c or 2-12f	11,848 square yards of PCC pavements.	1.617,333 colors feet of building/structures; 11,848 square yards of PCC personnes.				County	:
2. See Grading and Drainage			Cut 35,611 cubic yards			***************************************	County / FAA	
3. Size Utilities	2-12c or 2-12f		All				County / FAA	-
4. Access Roadway and Vehicle Parking			One Access Roadway; 40 perking speces.				County / FAA	
New Air Truffic Control Tower and Related     Equipment/Facilities	2-17c or 2-12f	5,000 square font base building; I control sower; I ASRI I radar	5,000 square feet base building; I congot tower, I ASRII radar				FAA	
:6. Relocation of TRACON Communication Equipment	2-12c or 2-12f	No further description	All				FAA	,

#### TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

Improvement	Map Reference (b)	Description of Proposed Project Improvements (Through Phone 2)	Places t	Phon 2		tam 4 Responsibility Ph
O. AIRPORT RESCUE AND FIREFIGHTING (ARFF) FACILITY						
ARFF - Fire Trucks & Emergency Vehicles	···	I Fleet - 4 of Fire Trucks Crash Unit and 3 Stationary Puzzos	Partial of ARFF floor	Period of ARFF flore		County ,
P. AVIATION INDUSTRIAL AND RELATED AREA			ham generally any		1	
Site Demolition	2-12%	3,725,398 cubic fort of building/structures; 71,164 square yants of AC pavements		30,774 outsic flori of building/structures; 36,354 support yards of AC perverents.	, ,	County
2. Site Oraciong and Errainage	2-126	26 seres; Cut 491,631 cubic yards		26 Acres; Cut 491,631 eubéc yards		Conety / Tesens
3 Sine Utilities		26 acres		,26 scres		County / Tenant
4. Access Readway Entrances		Onc		One	····	County / Teams
5. Buildings and Other Tenant Improvementa		No hather description		Tenan improvements		Tened
6. Lassiscaping	2-126	Na further description		Tonaci Improvements		County / Tenent
Q. AIRPORT MAINTENANCE FACILITIES (Long-Term site)						
6. Maintenance Vehicle Fuel Facility	3-121	i acre	All			
R. AIRCRAFT MAINTENANCE AREA (PA-1 & PA-3)						
11. Sac Demolisms		684,130 outric feet of building/structures; 34,634 square yards of AC prevenuents &		684, 130 cubic feet of building/structures; 36,634	,	County
2. Site Grading and Drainage	2-12b, 3-12g	:87,676 square yards of PCC pavernosts. 25 acres; Cut 979,738 cubic yards	A Service of the Serv	square yards of AC percentits 25 Acres, Cut 979,738 cubic yards	1	County / Teaset
3. Site Utilities	2-12h, 3-12a	25 acres		25 ecres		County / Terrett
Access Roadway Entrances and Vehicle Parking	2-126, 3-124	Two contrasects		Two entrasces	-	County / Teneral

# TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[6]

						,			*****
	Essperiment	Map Reference   bil	Description of Proposed Project Improvements (Through Phase 2)	Phote 1	Plane 2	Pinne 3	Phone 4	<del>Legendall</del> y	-
	5. Airline Maintenance Tintiways and Apron	Z-12b, 3-12g	85,902 square yards Fortland Comens Concrete		85,902 square yards Portland Coment Concrete Paving			County / Toness	<del> </del>
	6 Airtine Maintenance Facilities and Other Tenant improvements	2-12b, 3-12g	Hanger 152,600 square foet		Hangar 152,000 square feet			Total	;
S. AIRPORT	TRANSPORTATION CENTER (Shattle Bus Service)	·····				·····	***************************************	·····	
	Pedestrise Bridge to Irvine Transportation     Center	4-106	Pedestrius bridge to serve bus shelter at ATC	Pedestrian bridge to serve bus shelter at ATC				County	
		ų							-
T. AIRFORT	I SHIFTLE BUS MAINTENANCE FACILITY				•				
	1. Site Demolition	4-100	t i scres	_Il acres			·	County	
	2. Site Grading and Drainage		[ Lacres	II acres				Country	·
	3. Site Utilities	4-104	11 acres	·				County	
	4. Access Roadway Entrance		No finther description Qgs	<u> </u>				County	<u> </u>
	5. Bus Maintenance Facility	4-10a	No fineties description Maintenance	20 buses	10 additional buscs			County	t
			building 30,000 square fort, outdoor	•					
	1		storage and parkitus 25,500					•	
	1		sound fort, natural san fuction	:	1				
*******			facility; and buses		r				-
<del></del>	6. Lendscaping	4-10a	No further description	<del></del>				County	<u> </u>
			<u> </u>		1				<u> </u>
U. PLELDE	LIVERY, STORAGE AND DISTRIBUTION FACILITY	····						A	
	1. Sice Demolition	4-12i	6,720 cubic fees of building/structures;	7,198 square yands of AC	6.720 cubic fort of	'		County / Airlines	
			31,970 square yards of AC pavements.	pavements.	building/structures, 24,772			•	Ė
		i .			separate yields of AC pavernants.				
	2. Site Grading and Drainage	4-12i	7.7 seres; Out 74 cubic yards; Fill 39,833 Subje yards	5.6 Acres; Cut 74 cubic yearls; Fill 34,181 cubic yearls	2.1 Acres, Fill 3,652 cubic			County / Airlines	-
	3. Site Utilities	4-12i	7.7 acres	3.6 acres	(2.1 ecres	w. r. a.		County / Airlines	
** ****** *	4 Access Rosdowy Entrance	4-121	Two courses	Two entrances				County / Airlines	
	5 Pipeline Connection to Norwalk Fact Supply Line	1-56, 8-66, 4-121	One system; 14,400 linear feet of 8" Dia.	14,400 linear feet of 8" Dis.		***************************************	**********	County / Airlines	7.
	(pigetine delivery option only)		'welded Steel	welded Steel	,				

# $TABLE \ 3-9A \\ SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN(a) \\$

Suspensement	Map Reference [b]	Description of Frapased Project Improvements (Through Phone 2)	Phone (	Plans 2	Plane 3	Plane 4	Responsibility	m,
Pipeline Connection to Santa Fe Pacific Fael Supply Line (pipeline delivery option only)		One system; 250 linear feet of 10" Dia.	250 linear feet of 10" Dia. welded Steel		i		County / Airlines	T
7. Fuel Storage Facilities	•	5.25 million gallons; three 10ff diameter 30 feet high above-ground tanks	1.75 million gallons; 1 Storage Tank (100 feet die by 30 feet high)	3.5 million gallons; 2 Storage Tanks (100 feet die by 30 feet high)			County / Airlines	+
8. Fael Distribution Pipelines	4-12i	230 linear feet 10" Fill Pipeline: 1960 linear feet 14" Fill Pipeline; 760 linear feet 12" Suction Pipeline	Line: 480 linear feet 14" Fill Pipe Line: 610 linear feet 12" Suction				County / Airlines	į
9. Truck Unloading and Loading Facilities	4-12i	6 unloading stations	2 unloading stations	4 unleading stations			County / Autimes	
10. Fuel Monitoring and Control system	4-12i	3 pumps with constrols and filters	I pump with controls and filters	2 pumps with controls and filters	:		County / Airlines	t
11 Building/Portland Ceroent Concrete peving		4,250 square feet structure; 17,200 square yards 10" thick Portland Consust Concrete paying		8,500 square yards 10° thick PCC paving			Country / Airlines	
17. Lundscaping		blu further description	All					=
V. GROUND SUPPORT EQUIPMENT (GSE) AREA								
1. Site Demolities	4-12h	5,891 square yards of AC pavements.	1	2,349 square yards of AC payernesis			: County / Tenera	:
2. Site Grading and Drainage	4-12h	4.6 acres, Fill 138,092 cubic yards		4.6 Acres, Fill (38,092 cubic yards			County / Tenest	
3. Sine Unitities		4.6 scres		4.6 scres			County / Toners	
Access Roadway Entrances and Vehicle Parking	4-12h	Entrarce		1 Entrance			County / Tenant	: -
'5. Service Road Entrances	4-12h	Entrances		I Fattance			County / Tomani	

### TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[a]

Improvement	Map Saference (b)	Description of Proposed Project Improvements (Through Phase 2)	Phen I	Please 2	Place 3	Phone 4	Rays-Milky
6. CSE Facilises and Tenant Improvements	4-125	3,000 square feet fluidding; 15,617 square yards Aaphalt Concrete Paving		5,000 square feet bldg; 15,617 square yards Asphali Concrete Paving			Tennet
7. Lambcaping	4-125	15.617 square for		15,617 square for	********	-	County / Terusal
IVIRONMENTAL CONTROL FACILITIES	···········	<b>.</b>		······································	L		J
Storm Water Politation Control Facilities	1-96, \$-\$	3 Oil Water Separators @ 50,000/gal and 1 Oil Water Separator @ 20,000/gal	t Oil water separator (50,000 gal)	2 Oil water separators (50,000 gai) and I oil water separator(26,000 gai)			County
2. Noise monauring system		No further description	All	• • • • • • • • • • • • • • • • • • •			County
PPORT SYSTEMS, FACILITIES & EQUIPMENT		<u> </u>		·			<del></del>
Central Security and Constructions     system, Access Gates and Control     systems, and Security Francing	: <b>-</b>	One system - 10 gases; \$4,419 linear feet of security fence and support equipment	10 gates; 42,925 linear feet of security fence and support equipment	11,594 Insert fact of socurity fence and support equipment.		,	County
2. Alignet Vehicles		Flon	Partial of floet	Partial of faces			County
MPATIBLE REVENUE SUPPORT		······································		·	***************************************		
i. Agriculture	5-12, 3-15, 3-1c. 3-1d,7-1e	(38.9 acres	All	: I			County/Tenants
2. Galf Courses	3-3a, 3-3b, 7-3c, 7-3d, 7-3e, 7-3f	284.16 acres	**************************************				County

Supplemental Passing Analysis

<sup>[</sup>h] For environmental planning purposes, assumed to be Phone 4; the financial analysis indicates it may not be feasible to complete this project element without supplemental funding.

# TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND JWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[o]

Improvement	May Reference [b]	Proposed Project Improvements (Through Phase 2)	Phone 1	Phase 2	Pleaser 3	Phone 4	Responsibility	Ph
4. Reserced Open Space	8-62, 8-60, 8-60, 8-62, 8-60, 8-66, 7-69, 7-66	193.4 sores	AN				County	i
5. Orange County Hartral Community Conservation Plan/Habitet Conservation Fran - Habitet Reserve		29.6 screa	All	+ ^			County	!
6 Marshhum Retarding Basin		34.9 mores	All	\$ >		100 M A AA2 MARK AA	Сомпу	. 2 sengar
17. Custural/Institutional  8. Public Facilities	2-16	\$1.5 acres		All	i		Towares	1
Nebscle Maissenance Vard	5-134	:53.8 scret	<del>All</del>	·····			Tonapai	<u> </u>
b. Food Distribution Warehouse	5-136	.11.S acres	TAIL				Torum	
c IRWD Reservoir and Pumping	3-13c	5.7 mores	All	1			Towns	
d Fire Station # 1	3-138	1 2 scres	(A)I				Teneriti	-
e. California Air National Guard		24.1 scres	All	<del></del>			Tenants	. i
f. Homeless Service Providers		28.3 acres	All	1			Tenants	7
g. IRWD Facility		9.8 acres	All				Tenants	,
b. Fart Station # 2		1.0 acres	All	<u> </u>			Tenants	
9 Business Park	7-14a, 7-14b, 7-14c, 7-14d	56 & acres	All				(interpretation	1
10. Right-of-way for Alson Parkway Extension	755-75	17.67 acres		t		-	Tetests	+

County of Orange Final EIR No. 573 \$4.2 Supplemental Phasing, Analysis

### TABLE 3-9A SUMMARY LIST OF IMPROVEMENTS AT OCX AND IWA RECOMMENDED FOR PHASE 2 OF AIRPORT SYSTEM MASTER PLAN[4]

Emprovement	Map Reference  6	Description of Propused Project Ingrevenents (Through Phase 2)	Phone I.	Phase 2	Plens 3	Par 4	Responsibility	
		JWA IMPRO	/EMENTS					
				1				Τ-
Z. GENERAL AVIATION IMPROVEMENTS (I)		***************************************					:	
1 Site Preparation for Additional PBO Area	~	\$.0 Acres		:All:			County	
2. Additional Transient Parking Ramp	-	3.0 Acres	-	All			County	1
3. Additional Based Aircraft Tiedowas on East	-	3.1 Acres		All	,		County	,
. Side							ŧ	i
4. Relocation of Manua Way		750 lisear fors		All	· · · · · · · · · · · · · · · · · · ·		County	
5 New Aircraft Hangam and Reconfiguration		36 issuger units with central centrocess on		36 hangar trads with central			County / Tunant	
of West Side Tendowns		west side of authors (107,500 square fors)		restracate on west side of simon				1
1	j			(107,600 square feet)			ı	

[i] Refocution and expansion of airport quaintenance facilities, new sirport rescue and Rerfighting Sacility, and the relocation of 13 County basgar units are in the capital improvement program and are not included here.

#### 11.3 IMPACTS ANALYSIS

As discussed above, Draft EIR No. 573 analyzes the environmental impacts associated with the Phase 4 build out of the Proposed ASMP Project. To the extent the impacts projected to result during the interim phasing years (Phases 1 through 3) differ from those impacts projected at build out, Draft EIR No. 573 analyzes such interim year impacts as well.

This supplemental analysis excerpts portions of Draft EIR No. 573 relative to the interim year phasing impacts, generally, and where applicable, the impacts relative to Phase 2, specifically. Text originally appearing in Draft EIR No. 573 is presented in quotation marks with relevant page references. This section also serves to supplement the interim year phasing impacts analysis presented in Draft EIR No. 573 with additional analysis, where appropriate. When read in conjunction with the phasing year impacts analysis provided in Draft EIR No. 573, this supplemental phasing impacts analysis presents a comprehensive overview of the impacts associated with Phase 2 of the Proposed ASMP Project.

The analysis reveals that at Phase 2 of the Proposed Project, impacts will be of a lesser degree, generally, than those impacts projected to occur at Phase 4 build out. Reduced passenger and cargo aircraft operations will result in a corresponding reduction in the degree of impacts including, but not limited to, reduced noise, air quality and transportation/circulation impacts. The levels of significance for all impact categories at Phase 2, however, are substantially similar to those levels at Phase 4 build out. This is due to the fact that while the degree of project impacts at Phase 2 is less than that at Phase 4 build out, impacts at the Phase 2 level, like their counterparts at Phase 4 build out, exceed the applicable CEQA thresholds of significance. These impact categories include noise, air quality and transportation/circulation. Like Phase 4 build out, transportation/circulation impacts at Phase 2 can be mitigated below the level of significant; noise and air quality impacts remain significant and unavoidable.

All page, table and figure references contained here are to Draft EIR No. 573 (December 1999), unless otherwise noted. (Note: This section, as well as Draft EIR No. 573, intermittently refer to the ASMP as the ASMP, the proposed project or Alternative B; all such references are to the proposed project.)

#### 11.3.1 Land Use

### 11.3.1.1 Physical Community Division

"The Proposed Project at completion and during each of the four phases would not physically divide an existing community" (4.1-10), and, therefore, would not result in significant adverse impacts at the Phase 2 implementation level.

#### 11.3.1.2 Compatibility With Adjacent Land Uses

"The analysis of the proposed on-site uses [in DEIR No. 573 Sections 4.1.6.3 through 4.1.6.12] concludes that the Proposed Project does not create substantial incompatibilities between the project's land uses and adjacent off-site existing and planned land uses, in accordance with Significance Threshold ii. The order or timing of the on-site improvements will not alter this conclusion; therefore, the Proposed Project will not create substantial incompatibilities at the time of project completion or during any of the four phases." (4.1-11.) Accordingly, the Proposed ASMP Project would not result in significant adverse impacts at the Phase 2 implementation level. (This conclusion is based on the Proposed Project's 65 CNEL noise contours. See Section 5.3.4, infra, noise impacts analysis.)

### 11.3.2 General Plan Consistency

"The Proposed Project's impacts are generally analyzed according to the planned phases of construction. However, project consistency with County and City General Plans is applicable to the project actions occurring prior to Phase 1. Therefore, the General Plan Consistency analysis is not subject to analysis by phase." (4.2-11.) The Proposed ASMP Project, therefore, will not result in significant adverse impacts at the Phase 2 implementation level.

### 11.3.3 Transportation and Circulation

"Table 4.3-14 summarizes the existing plus committed intersection locations, arterial roadways, freeway mainline segments, and freeway ramps that are significantly impacted by Phase 1 (2005) of the Proposed Project. Under the existing conditions plus committed (by 2005) roadway network setting, four intersection locations would be significantly impacted, three of which would operate at LOS E and one of which would operate at LOS F, and two arterial roadway segments would be significantly impacted, both of which would operate at LOS F. Also, one freeway ramp and one continuous freeway segment would be significantly impacted and would operate at LOS F." (Draft EIR No. 573 Supplemental Analysis, supra, 3-10.)

Table 4.3-14
Phase 1 (2005) Proposed Project Impact Summary

Location	Jurisdiction	Location	Jurisdiction
IMPACTED INTERSECTION	3		
ETC East Leg NB & Irvine	County/TCA	Alicia & Paseo Valencia	Laguna Hills
Sand Canyon & Trabuco [1]	County	El Toro & Rockfield	Lake Forest
IMPACTED ARTERIAL ROA	DWAYS		
Irvine (ETC East Leg to PA-2 West Access Rd)	County	Laguna Canyon (south of El Toro) [1]	County/Caltrans Laguna Beach
IMPACTED FREEWAY SEGN	MENTS		
I-5 (SR-55 to Newport)	Caltrans		
IMPACTED FREEWAY RAM	PS		
I-5 at Red Hill (SB On-Ramp)	Caltrans/Tustin		

<sup>[1]</sup> Facilities that are also significantly impacted in existing plus Proposed Project conditions (Table 4.3-10)

"Table 4.3-16 summarizes the existing plus committed intersection locations, arterial roadways, freeway mainline segments, and freeway ramps that are significantly impacted by Phase 2 (2010) of the Proposed Project. Under the existing conditions plus committed (by 2010) roadway network setting, five intersection locations would be significantly impacted, four of which would operate at LOS E and one of which would operate at LOS F, and two arterial roadway segments would be significantly impacted, both of which would operate at LOS F. Also, one freeway ramp and one continuous freeway segment would be significantly impacted and would operate at LOS F." (Draft EIR No. 573 Supplemental Analysis, supra, 3-11.)

Table 4.3-16
Phase 2 (2010) Proposed Project Impact Summary

Location	Jurisdiction	Location	Jurisdiction
IMPACTED INTERSECTIONS			·
ETC East Leg NB & Irvine	County/TCA	Alicia & Paseo Valencia	Laguna Hills
Sand Canyon & Trabuco [1] County		El Toro & Rockfield	Lake Forest
Sand Canyon & I-5 NB Ramps [1]	Irvine/Caltrans		
IMPACTED ARTERIAL ROADWA	AYS		
Irvine (ETC East Leg to PA-2 West Access Rd)	County	Laguna Canyon (south of El Toro) [1]	County/Caltrans Laguna Beach
IMPACTED FREEWAY SEGMEN	TS		***************************************
I-5 (Sand Canyon to north of SR-55)	Caltrans		
IMPACTED FREEWAY RAMPS			
I-5 at Red Hill (SB On-Ramp)	Caltrans/Tustin		

<sup>[1]</sup> Facilities that are also significantly impacted in existing plus Proposed Project conditions (Table 4.3-10)

The Draft EIR No. 573 transportation and circulation analysis trip generation data for Phase 2 assumed the full Phase 4 build out of all nonaviation land uses (e.g., business park, regional park, golf courses, etc.). (See Draft EIR No. 573 Table 7-3 Appendix D, Part 1.) Therefore, the Draft EIR No. 573 transportation and circulation Phase 2 analysis is premised upon Phase 2 aviation land uses and Phase 4 nonaviation land uses.

#### 11.3.3.1 Mitigation Measures

Table 4.3-20A summarizes a set of potential circulation improvements that serve to mitigate the intersection, arterial roadway, freeway/tollway ramp, and freeway/tollway mainline impacts of the Proposed Project during the project's phased development from 2005 (opening phase) through Phase 2. The County's responsibility for implementing the proposed intersection, arterial roadway and freeway/tollway ramp mitigation improvements is noted on the summary table according to two types of improvements: 1) improvements wholly funded and constructed by the County, and 2) improvements whose implementation is contributed to by the County on a fair share basis (as mentioned earlier, the locations that require a fair share contribution by the project are forecasted to operate at unacceptable levels of service with or without the project).

The County's share of the cost to implement each fair share improvement location is also noted on the summary table. The timeframe identified for the implementation of each improvement is based on the results of the existing plus committed project phasing impact analysis summarized in Chapter 3.0 of this Supplemental Analysis. (See Draft EIR No. 573 Supplemental Analysis, supra, 3-14.)

#### 11.3.3.2 Level of Significance After Mitigation

"The potentially significant adverse traffic impacts of the Airport System Master Plan Proposed Project would be reduced to below a level of significance based on the implementation of [Draft EIR No. 573] Mitigation Measures T-1 through T-17 described above. In the event that Traffic Operations Strategies (TOPS) improvements are not implemented prior to project phase completion, or if such improvements fail to mitigate the identified impacts to below a level of significance, the freeway/tollway mainline impacts will remain significant and unavoidable. Regarding off-site highway improvements that are not under the sole control of the County and are subject to the implementation procedures described in Mitigation Measures T-4 and T-9, in the event such off-site improvements are not implemented by the jurisdiction in which the improvements are located before completion of the project development phase that requires the improvements, the impacts at those unimproved locations will remain significant and unavoidable." (Draft EIR No. 573 Supplemental Analysis, supra, 3-20.)

Table 4.3-20A
Proposed Project Impact Mitigation Improvements

			Date	Funding	Project
Location	Jurisdiction	Improvement	Needed	Obligation	Share
IMPACTED INTERSECT	IONS	334000000000000000000000000000000000000			
ETC East Leg NB & Irvine	County	Add third EB through lane and	2005	Fully fund	100%
		second NB right-turn lane			
Sand Canyon & Trabuco	County	Add third NB through lane	2005	Fully fund	100%
	<b>\$</b>	Add second WB left-turn lane and	[3]	Fair share	13%
	***	convert EB right-turn lane to shared			
		second through/right-turn lane [1]			
Sand Cyn & I-5 NB Ramps	Irvine	Add third NB and SB through lanes	2010	Fully fund Fully fund	100%
	Add second EB and WB through				100%
		lanes, second WB left-turn lane and		***************************************	
		NB right-turn lane [2]			
Alicia & Paseo Valencia	Laguna Hills	Add second EB right-turn lane	2005	Fair share	6%
El Toro & Rockfield	Lake Forest	Add EB right-turn lane	2005	Fair share	9%
IMPACTED ARTERIAL	ROADWAYS				
Irvine (ETC East Leg to	County	Improve to six lanes	2005	Fully fund	100%
PA-2 West Access Rd)	, i				
Laguna Canyon (south of	County/	Improve to four lanes	2005	Fair share	6%
El Toro)	Laguna Beach	7.0			
			Date	Project Sl	are of
Location	Jurisdiction	Improvement	Needed	Future T	raffic
IMPACTED FREEWAY/T	OLLWAY SEG	MENTS	<u> </u>		
I-5 at Red Hill (SB On-	Caltrans/	Convert HOV preferential lane to	2005	Fair share	3%
Ramp)	Tustin	Add a second metered mixed-flow lane			
I-5 (north of SR-55)	Caltrans	Implementation of Caltrans TOPS	2010	10% .	
I-5 (Newport to SR-55)	Caltrans	Implementation of Caltrans TOPS	2005	13%	
I-5 (Sand Canyon to	Caltrans	Implementation of Caltrans TOPS	2010	13%-15%	
Newport)		-			
Abbreviations: NB - north	bound I	EB – eastbound			
SB – south	bound '	WB – westbound			

<sup>[1]</sup> Only required with the Trabuco Road/ETC interchange option

<sup>[2]</sup> Not required with the Trabuco Road/ETC interchange option

<sup>[3]</sup> Since the Trabuco Road/ETC interchange is not needed to address impacts during the phased development of the Proposed Project, no specific date for implementing improvements associated with the interchange has been determined.

#### 11.3.4 Noise

#### 11.3.4.1 Aviation Related Noise Impacts at MCAS El Toro

#### CNEL Contours for the Proposed Project

"Developed using the assumptions described earlier, the 60, 65 and 70 dB CNEL contours for the Proposed Project for calendar year 2020 operations are shown on [Draft EIR No. 573] Figure 4.4-34. Interim years (2005, 2010 and 2015) CNEL contours for the Proposed Project are shown on Figures 4.4-35, 4.4-36 and 4.4-37, respectively." (4.4-67.) (See Draft EIR No. 573, Figures 4.4-35, 4.4-36 and 4.4-37.) As Figure 4.4-36 indicates, there are no existing residential uses inside the 65 CNEL contour for the Phase 2 project.

# 11.3.4.2 CNEL Contour Land Use Impacts for the Proposed Project

"There are no existing residential uses inside the 65 CNEL contour for any of the phasing years or the 2020 project contour. Table 4.4-29a compares the land use effects of the Proposed Project in each of the phasing years (2005, 2010, 2015 and 2020)." (4.4-67.) Like the 2020 proposed project scenario, the table indicates that while the Proposed Project will increase the total land area affected by the CNEL contours during Phase 2, there is little impact on incompatible uses. During Phase 1, there will be less total land area affected by the CNEL contours than the 1998 military contours.

### 11.3.4.3 CNEL Receptor Locations for the Proposed Project

Table 4.4-30a compares the CNEL at specific receptor locations for the Proposed Project for the years 2005, 2010, 2015 and 2020. (4.4.-70.)

Noise impacts in terms of CNEL vary among the receptor sites. While several sites experience increases, only a few achieve levels above 60 CNEL, with the number increasing during each successive phase with increased aircraft operations. During Phase 1, two locations experience levels above 60 CNEL, those designated EO and LMA. Site EO is located north of the MCAS El Toro site under the departure pattern for Runway 34. This site is currently undeveloped land owned by The Irvine Company in unincorporated Orange County. This is not considered an incompatible use. Noise levels at site EO would rise to 65.7 during Phase 2 and 67.0 during Phase 3.

Table 4.4-29a
Land Use Comparison Table, Military, CRP and Alternative B Year 2005, 2010, 2015, and 2020

(area in square miles)

	AICUZ (PIL)	UZ (PIL) 1994 1998 Year 2020 Alternative B, By Year							
	Military*	Military*	Military	CRP Alt A*	2005	2010	2015	2020	No Projec
Square Miles Within Contour:									
- 60 CNEL Contour	50.0	30.5	16.6	20.3	10.4	19.2	22	22	,
- 65 CNEL Contour	28.8	18.9	6.3	9.1	4.2	8.2	9.8	9.8	
- 70 CNEL Contour	21.6	27.0	3	5.5	1,7	3.2	3.9	3.9	
Square Miles Within Contour On Base:									
- 60+ CNEL Contour	7.3	6.7	6.3	6.2	5	5.9	6.1	6.4	į
- 65+ CNEL Contour	7.0	6.0	4.8	4.8	3.3	4.5	4.7	4.9	i
- 70+ CNEL Contour	6.4	5.2	2.9	3.2	1.7	2.9	3.1	3.2	
Square Miles of Residential:									
- 60+ CNEL Contour	7.0	1.8	0.3	0.8	0	0.3	0.6	0.7	
- 65+ CNEL Contour	0.8	0.2	0.0	0.1	0	0	0		
- 70+ CNEL Contour	0.0	0.1	0.0	0.0	0	0	0	0	
Number of Residences Inside Contour:							-		
- 60+ CNEL Contour	18368	4723	672	2034	0	787	1574	1837	
- 65+ CNEL Contour	2099	630	0	318	0	0	0	0	
- 70+ CNEL Contour	0	330	0	o	Ó	Ō	Ō		
Number of Public Schools Inside Contour:									
- 60 to 65 CNEL Contour	15	4	0	1	1 on base	1 on base	1	1	
- 65 to 70 CNEL Contour	1 onbase	1onbase	1 on base	1	0	0	1 on base	0	
- inside 70 CNEL Contour	0	0	0	0	0	0	0	0	
Number of Private Schools Inside Contour:									
- 60 to 65 CNEL Contour	7	4	4	3	1	3	3	3	
- 65 to 70 CNEL Contour	5	1	o	1	0	Ō	1	1	
- inside 70 CNEL Contour	0	0	ő	0	o	Ō	0		
Number of Colleges Inside Contour:				_	-	•	_	_	
- 60 to 65 CNEL Contour	1	0	0	0	0	0	0	0	
- 65 to 70 CNEL Contour	1 0	0	Ō	o	Ŏ	ō	0	_	
- inside 70 CNEL Contour	1 0	0	0	o	Ö	ō	Ô		
Number of Hospitals Inside Contour:			_	-	•	•		•	
- 60 to 65 CNEL Contour	1 0	l o	0	0	0	0	0	0	
- 65 to 70 CNEL Contour	0	ا	ō	ō	Ö	ō	Ö	-	
- inside 70 CNEL Contour	0	٥	ō	o	Ö	ō	ō		
Number of Churches Inside Contour:						•	•	·	
- 60 to 65 CNEL Contour	2	5	12	4	2	11	12	13	
- 65 to 70 CNEL Contour	13		ō	8	Ō	Ö	2		
- inside 70 CNEL Contour	1 1	2	Ŏ	0	Ŏ	ő	0		

<sup>\*</sup> Data from updated land use database from EIR 573.

Table 4.4-30a Comparison of CNEL, Alternative B Years 2005, 2010, 2015, and 2020

Location	2005	2010	2015	2020
AHI	49.6	55.2	56.6	56.0
ANH	53.5	57.9	58.8	59.2
AV1	56.8	59.2	60.1	60.5
AV2	55.1	60.5	61.4	61.7
AV3	56.3	58.6	59.5	59.9
AV4	56.4	59.8	60.7	61.2
CCI	50.5	53.0	53.7	53.8
CH1	30.2	34.0	35.2	35.0
DP1	50.0	53.5	54.4	55.0
DP2	47.7	51.1	52.0	52.6
EO	60.0	65.7	67.0	66.3
FE	52.4	57.2	58.1	58.5
FRI	55.1	57.5	58.1	59.4
FRE	53.8	56.3	56.9	58.2
II	53.8	56.8	57.9	57.6
12	46.2	49.2	50.3	50.2
13	41.5	44.4	45.4	45.7
13 I4	38.8	41.8	42.7	43.7
	i i		44.8	45.4
15	40.9	43.8	37.5	37.4
16	32.5	36.4		45.8
IMC	41.6	44.4	45.3	
LB1	40.8	43.3	44.3	44.6
LB2	46.6	49.6	50.6	51.0
LFI	46.8	51.3	52.1	52.4
LF2	45.6	50.2	51.2	51.1
LF3	53.6	56.1	56.8	56.0
LF4	55.5	58.0	58.6	57.9
LF5	54.0	56.5	57.1	56.7
LH1	40.4	44.2	45.1	45.5
LMA	61.5	63.3	64.2	64.5
LNI	54.4	57.8	58.8	59.2
LN2	47.2	51.6	52.5	52.9
LWI	56.8	58.0	58.8	59.1
LW2	55.9	62.2	63.1	63.4
MHE	56.6	59.1	59.8	59.8
MJD	56.7	59.9	60.8	61.2
MVI	53.3	55.8	56.4	56.1
MV2	58.3	60.7	61.4	61.4
MVC	59.6	63.7	64.6	65.0
O1	43.7	47.0	48.0	48.1
OC1	53.6	54.7	55.6	55.8
OC2	58.6	59.9	60.7	61.1
OC3	55.8	61.5	62.4	62.7
OKE	49.3	51.1	52.0	52.4
PH1	56.9	59.3	60.0	60.6
SCC	58.3	60.7	61.4	60.8
SM1	57.0	59.4	60.1	60.2
SM2	55.9	58.4	59.0	58.9
SM3	54.0	56.4	57.1	57.3
SM4	46.5	49.0	49.6	50.2
SM5	47.5	50.0	50.6	50.6
SMA	55.5	58.8	59.7	60.2
TI	41.2	44.2	45.3	45.2
WCE	52.2	56.3	57.2	57.6

Noise impacts in terms of CNEL vary among the receptor sites. While several sites experience increases, only a few achieve levels above 60 CNEL, with the number increasing during each successive phase with increased aircraft operations. During Phase 1, two locations experience levels above 60 CNEL, those designated EO and LMA. Site EO is located north of the MCAS El Toro site under the departure pattern for Runway 34. This site is currently undeveloped land owned by The Irvine Company in unincorporated Orange County. This is not considered an incompatible use. Noise levels at site EO would rise to 65.7 during Phase 2 and 67.0 during Phase 3.

Site LMA is a private school in a commercial area. Significantly, noise levels at the LMA location under 1998 conditions are 63.7 CNEL. During Phase 1, that level actually would decrease to 61.5, while rising to 63.3 during Phase 2 and 64.2 during Phase 3. The question of significant impacts for this site is resolved by the conditions of the Use Permit for the establishment, which requires that the school have no outdoor uses and be insulated to protect it from military noise higher than the forecast civilian use noise. In other words, the existing legal conditions of use impose limitations which would be expected to, and were intended to, ensure compatibility with the forecast noise levels and aviation activity. Site MVC is also a private school in a commercial area operating under a Use Permit similar to site LMA. Noise levels at MVC are 63.4 during 1998 conditions; they drop to 59.6 during Phase 1, and rise to 63.7 and 64.6 during Phases 2 and 3, respectively.

Site LW2, a residential area in Laguna Woods, would experience a decrease in CNEL of 5 dBA during Phase 1, and increases of 2.2 dBA or less during Phases 2 and 3, but no increase in CNEL above 65 dBA. Sites AV2, MV2, MVC, OC3 and SCC would experience CNEL levels above 60 dBA under Phase 2 and Phase 3. Sites AV1, AV4, OC2, PH1 and SM1 would experience noise levels above 60 dBA during Phase 3 only.

"In the approach corridor from the south, and the departure corridor to the east, most residential receptor locations will experience little differences in CNEL relative to 1998 military operations under the Proposed Project. Most changes range only between plus or minus one or two dBA CNEL during Phases 2 and 3, depending on the receptor location. These differences would be imperceptible to the average human, and none of the increases raise levels above 65 CNEL." (4.4-73.)

"In the west corridor, where the military conducted FCLP operations, CNEL under the Proposed Project during the phasing years will decrease by as much as 12 dBA." (4.4-73.)

## 11.3.4.4 SENEL Contours for the Proposed Project

"[Draft EIR No. 573] Figures 4.4-38, 4.4-39 and 4.4-40 provide 86dB SENEL noise contours for normal runway use conditions. ...¶Examination of the civilian SENEL contours indicate they are smaller than those for the military F/A-18 shown on Figure 4.4.-14. With the exception of the noisiest aircraft such as the 747-400, the MD-83 and the hushkitted 727, the 86dBA SENEL contours are contained within the noise buffer zone created by the AICUZ and the

County's land use PIL. The relative noisier aircraft are those forecasted to use OCX the least. The hushkitted 727 will use OCX under the Proposed Project only in the interim years and at relatively low activity levels. The 747-400 and MD-83 account for five percent of the departures on Runway 07 and six percent of the arrivals on Runway 34. Therefore, for the great majority of civilian aircraft using OCX under the Proposed Project, the 86 dBA SENEL contours will be confined to the existing noise buffer zone." (4.4-74.)

This conclusion applies equally during the interim phasing years since the SENEL contours are aircraft-specific and, therefore, are the same during project build out as they are during the phasing years. This is because SENEL contours are based on single events of specific types of aircraft (e.g., 7373B2s, 757PWs). The fleet mix at OCX during each of the phasing years essentially is the same as the fleet mix during build out of the Proposed Project. (See Draft EIR No. 573 Tables 4.4-26 and 4.4-27.)

### 11.3.4.5 SENEL at Receptor Locations for the Proposed Project

The discussion presented in Draft EIR No. 573 at pages 4.4-74 - 77 with respect to SENEL at receptor locations applies equally to the proposed project during the interim phasing years, as well as project build out. This is because the SENEL impacts at receptor locations are essentially the same during project build out as they are during the phasing years. As previously discussed, SENEL contours are derived based on single events of specific types of aircraft, e.g., 7373B2s, 757PWs. Because the fleet mix at OCX during each of the phasing years essentially is the same as the fleet mix during build out of the Proposed Project, the SENEL at receptor locations essentially is the same for both. (See Draft EIR No. 573 Tables 4.4-26 and 4.4-27.) See Draft EIR No. 573, pages 4.4-74 through 4.4-77 for complete discussion regarding SENEL at receptor locations for the Proposed Project.

# 11.3.4.6 Time Above Threshold Noise Levels for the Proposed Project

"[A] comparison based upon CNEL levels, as provided above, is the accepted standard and the only statistically significant methodology for predicting long-term community response to an airport noise environment. To assist in comparing the Proposed Project [during the phasing years] to the existing military operations and to further describe the noise environment associated with the Proposed Project, a Time Above ("TA") analysis was conducted." (4.4-78.)

"Tables 4.4-34a and 4.4-34b compare time above thresholds data for the Proposed Project for the years 2005, 2010, 2015 and 2020 for exposure over 24 hours and night time hours, respectively." (4.4.-78.) During a 24-hour period, at the 85 dBA TA level, the Proposed Project during Phase 1, Phase 2 and Phase 3 creates impacts only at site EO and this site is undeveloped. The military aircraft show levels above 85 dBA at 28 sites. (See Draft EIR No. 573 Table 4.4-34 for 1998 military TA.)

Table 4.4-34a

Time Above Data for Year 2020 Alternative B at Sensitive Receptor Locations
Compared with Years 2005, 2010 and 2015, 24 Hour Exposure

	Year 1998 Year 2005 Year 2010 Year 2015												ear 2020		
		24 Hours			24 Hours	- 1		4 Hours			24 Hours	H		24 Hours	
}		inutes Abov	ا م		iutes Abov	.		utes Abov	.		nutes Above	. 1		nutes Above	
Location	65 dBA	77 dBA	85 dBA		77 dBA	85 dBA		77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA
AHI	0.0	0.0	0.0	2.9	0.0	0.0	6.5	0.4	0.0	8.6	0.6	0.0	9.6	0.3	0
ANH	18.3	3.4	0.2	9.6	0.0	0.0	30.2	0.1	0.0	36.3	0.2	0.0	41.1	0.1	
AVI	18.4	5.4	0.5	19.9	0.1	0.0	35.0	0.5	0.0	41.8	0.8	0.0	47.3	0.5	
AV2	18.6	5.3	0.5	16.0	0.0	0.0	44.4	0.9	0.0	52.6	1.3	0.0	59.8	0.9	0
AV3	18.7	5.5	0.4	20.6	0.0	0.0	34.8	0.3	0.0	41.6	0.5	0.0	47.1	0.3	0
AV4	18.8	6.0	0.5	20.9	0.0	0.0	43.6	0.8	0.0	51.4	1.1	0.0	58.6	0.7	0
CCI	3.8	0.4	0.0	3.2	0.0	0.0	5.6	0.0	0.0	6.7	0.0	0.0	7.1	0	
СНІ	0.6	0.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	Ō	0	0
DP1	19.3	1.1	0.0	1.2	0.0	0.0	5.8	0.0	0.0	8.5	0.0	0.0	9.6	0	0
DP2	18.4	0.6	0.0	0.3	0.0	0.0	2.5	0.0	0.0	3.8	0.0	0.0	3.9	0	0
EO	0.5	0.0	0.0	9.6	1.9	0.3	20.5	4.4	1.0	25.7	5.8	1.4	29.4	6.1	1.1
FE	17.6	2.2	0.1	4.8	0.0	0.0	24.1	0.1	0.0	29.5	0.1	0.0	33.1	0.1	0
FR1	7.6	1.6	0.5	8.9	0.2	0.0	16.3	0.3	0.0	18.8	0.3	0.0	25.3	0.7	0
FRE	7.0	1.5	0.4	7.4	0.0	0.0	13.3	0.0	0.0	15.4	0.0	0.0	21.1	0.2	0
11	7.8	0.8	0.2	7.3	0.3	0.0	12.6	0.3	0.0	16.0	0.3	0.0	17	0.2	0
12	9.1	0.9	0.1	0.8	0.0	0.0	0.9	0.0	0.0	1.1	0.0	0.0	0.6	0	0
13	9.2	0.3	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
14	8.7	0.5	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
15	14.0	0.5	0.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
16	2.5	0.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
IMC	13.2	0.4	0.1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
LB1	1.6	0.0	0.0	0.0	0.0	0.0	0.1	0.0	0.0	0.1	0.0	0.0	0.1	0	0
LB2	17.7	0.1	0.0	0.1	0.0	0.0	1.4	0.0	0.0	2.1	0.0	0.0	1.5	0	0
LFI	11.4	0.6	0.0	0.3	0.0	0.0	1.7	0.0	0.0	2.4	0.0	0.0	1.9	0	0
LF2	13.1	0.0	0.0	0.1	0.0	0.0	0.4	0.0	0.0	0.4	0.0	0.0	0.3	0	0
LF3	7.4	1.5	0.2	6.5	0.0	0.0	11.5	0.0	0.0	13.4	0.0	0.0	11	0	
LF4	6.9	1.6	0.4	10.3	0.2	0.0	18.9	0.3	0.0	21.8	0.3	0.0	20	0.1	
LF5 LH1	6.0 2.8	1.3	0.0	6.8	0.1	0.0	12.2	0.1	0.0	14.3	0.1	0.0	13.6	0	
LMA	19.2	0.0 6.9	0.0	0.0	0.0	0.0	0.1	0.0	0.0	0.2	0.0	0.0 0.1	0.2	5.2	
LMA	19.2	5.2	0.3	31.4 14.3	2.1 0.0	0.0	54.8 31.7	3.8	0.1	64.8 38.0	5.1	0.1	74.6 43.7	- 3.2	
LN2	14.9	0.5	0.3	0.2	0.0	0.0	2.5	0.0	0.0	38.0	0.1	0.0	2.9	- 0	
LWI	18.4	3.5	0.0	22.6	0.0	0.0	26.2	0.0	0.0	31.6	0.0	0.0	35.4	0.2	
LW2	18.1	4.4	0.5	19.8	0.1	0.0	51.3	2.1	0.0	60.4	2.9	0.0	69.5	2.7	
MHE	5.8	1.8	0.0	10.1	0.1	0.0	18.4	1.0	0.0	21.7	1.1	0.0	25.1	0.8	
MJD	18.8	6.1	0.6	21.6	0.0	0.0	43.5	0.8	0.0	51.4	1.1	0.0	58.6	0.8	
MV1	4.9	1.1	0.0				9.9				0.2			0.8	
MV1	4.9	1.1	0.0	5.6	0.1	0.0	9.9	0.1	0.0	11.7	0.2	0.0	11.2	0.1	

		Year 1998			Year 2005		,	Year 2010			Year 2015			Year 2020	
		24 Hours			24 Hours		24 Hours 24 Hours					24 Hours			
1 1	Mi	nutes Abov	re	Mi	nutes Abov	e	Mi	nutes Abov	/e	Minutes Above			Mi	inutes Abov	
Location	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA
MV2	9.2	2.6	0.4	13.9	1.0	0.0	25.7	1.7	0.0	30.0	1.9	0.0	34.8	1.5	0
MVC	19.5	6.9	0.8	28.1	0.5	0.0	59.6	3.8	0.0	70.4	5.3	0.0	81.3	5.2	0
Ol	0.4	0.0	0.0	0.8	0.0	0.0	1.3	0.0	0.0	1.5	0.0	0.0	1.7	0	0
OC1	18.2	2.2	0.3	8.4	0.1	0.0	9.2	0.1	0.0	11.8	0.1	0.0	12.3	0.1	0
OC2	18.7	4.9	0.6	26.9	0.2	0.0	37.8	0.8	0.0	45.0	1.2	0.0	51.2	0.8	0
OC3	18.7	4.8	0.6	19.1	0.0	0.0	49.1	1.6	0.0	57.9	2.3	0.0	65.5	2	0
OKE	14.4	0.7	0.0	0.6	0.0	0.0	2.0	0.0	0.0	2.8	0.0	0.0	2.3	0	0
PH1	8.6	3.1	0.7	11.1	0.7	0.0	20.4	1.1	0.0	23.8	1.3	0.0	29.5	1.3	0
SCC	8.7	1.8	0.5	14.8	0.9	0.0	27.2	1.5	0.0	31.5	1.7	0.0	32.5	1.1	0
SMI	6.8	2.0	0.2	10.9	0.7	0.0	20.0	1.1	0.0	23.5	1.2	0.0	27.8	0.9	0
SM2	5.6	1.5	0.0	8.4	0.4	0.0	15.2	0.6	0.0	18.2	0.6	0.0	20.1	0.4	0
SM3	5.4	1.5	0.0	5.4	0.1	0.0	9.7	0.1	0.0	11.8	0.1	0.0	13.6	0.1	0
SM4	2.8	0.1	0.0	0.5	0.0	0.0	0.8	0.0	0.0	0.8	0.0	0.0	1	0	0
SM5	2.4	0.1	0.0	0.9	0.0	0.0	1.4	0.0	0.0	1.6	0.0	0.0	1.4	0	0
SMA	18.8	5.8	0.5	18.0	0.0	0.0	37.2	0.4	0.0	44.0	0.6	0.0	50.4	0.4	0
Tl	0.5	0.0	0.0	0.2	0.0	0.0	0.3	0.0	0.0	0.4	0.0	0.0	0.4	0	0
WCE	17.7	1.9	0.0	4.6	0.0	0.0	19.1	0.0	0.0	23.6	0.0	0.0	26.6	0	0

Table 4.4-34b
Time Above Data for Year 2020 Alternative B at Sensitive Receptor Locations
Compared With Years 2005, 2010 and 2015, Night Hours Only (10pm - 7am)

	Nig	Year 1998 ht Hours C	holy	Year 2005 Night Hours Only			H .	Year 201( it Hours (		а .	ear 2015 t Hours		•	ear 2020 t Hours	-
	M	inutes Abo	ve	Mi	nutes Abo	ve	Mi	nutes Abe	ove	Mi	nutes Ab	ove	Minutes Above		
Location	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA
AHI	0.0	0.0	0.0	0.7	0	0	1.9	0.2	0.0	2.5	0.2	0.0	2.7	0.1	0
ANH	0.5	0.0	0.0	1.7	0	0	4.4		0.0	5.3	0.0			0	0
AVI	0.5	0.0	0.0	3.2	0	0		0.0	0.0	6.0	0.0	0.0		0	0
AV2	0.5	0.0	0.0	2.6	0	0		0.1	0.0	7.6	0.1	0.0		0.1	0
AV3	0.5	0.0	0.0	3.4	0	0		0.0		6.0	0.0	0.0	H	0	
AV4	0.5	0.0	0.0	3.4	0	0		0.0	0.0	7.4	0.0	0.0	8.5	0	0
CCI	0.0	0.0	0.0	0.9	0	0	1.5	0.0	0.0	1.8	0.0	0.0	1.8	0	0
CHI	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
DPI	0.5	0.0	0.0	0.3	0	0	0.7	0.0	0.0	1.0	0.0	0.0	1.3	0	0
DP2	0.5	0.0	0.0	0	0	0	0.2	0.0	0.0	0.2	0.0	0.0		0	0
EO	0.0	0.0	0.0	1.6	0.4	0.1	4.1	1.3	0.3	5.3	1.8	0.5	6.1	1.8	0.4
FE	0.5	0.0	0.0	1	0	0	3.5	0.0	0.0	4.2	0.0	0.0	4.8	0	0
FRI	0.0	0.0	0.0	2.2	0	0	3.8	0.1	0.0	4.5	0.1	0.0	5.8	0.2	0
FRE	0.0	0.0	0.0	1.8	0	0	3.2	0.0	0.0	3.7	0.0	0.0	4.9	0	0
]]	0.0	0.0	0.0	1.4	0.1	0	3.1	0.1	0.0	4.0	0.1	0.0	4.2	0.1	0
12	0.0	0.0	0.0	0.1	0	0	0.3	0.0	0.0	0.4	0.0	0.0	0.2	0	0
13	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
14	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	
15	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
16	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
IMC	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
LBI	0.1	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
LB2	0.5	0.0	0.0	0	0	0	0.1	0.0	0.0	0.1	0.0	0.0	0.1	0	0
LFI	0.3	0.0	0.0	0	0	0	0.1	0.0	0.0	0.2	0.0	0.0	0.1	0	0
LF2	0.1	0.0	0.0	0	0	0	0.1	0.0	0.0	0.1	0.0	0.0	0	Õ	0
LF3	0.0	0.0	0.0	1.7	0	0	2.9	0.0	0.0	3.4	0.0	0.0	2.7	0	0
LF4	0.0	0.0	0.0	2.5	0	0	4.3	0.1	0.0	5.1	0.1	0.0	4.7	0	0
LF5	0.0	0.0	0.0	1.8	0	- O	3.1	0.0	0.0	3.6	0.0	0.0	3.4	0	0
LHI	0.1	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
LMA	0.6	0.0	0.0	4.9	0.5	0	7.8	0.5	0.0	9.2	0.7	0.0	10.5	0.7	0
LNI	0.5	0.0	0.0	2.4	0		4.6	0.0	0.0	5.5	0.0	0.0	6.3	0	0
LN2	0.4	0.0	0.0	0	Ō	0	0.2	0.0	0.0	0.3	0.0	0.0	0.3	0	0
LWI	0.5	0.1	0.0	3.6		Č	21		ļ	<b></b>	0.0			Ō	
LW2	0.5	0.1	0.0	3.2		C								0.4	0
MHE	0.0	0.0	0.0	2.6		0			<u> </u>	4				0.2	
MJD	0.5	0.0	0.0	3.5		0			<u> </u>					0	0
MVI	0.0	0.0	0.0	1.5		0	2.5	0.0	0.0	3.0	0.0	0.0	2.8	0	0
MV2	0.0	0.0	0.0	3.3		0	5.7	0.5	0.0	6.8	0.5	0.0	7.9	0.4	
MVC	0.6	0.0	0.0	4.5		Ō	4				<u> </u>			0.8	

	Year 1998 Night Hours Only Minutes Above		*	Nigh	ear 2005 t Hours C tutes Abo	nly	Year 2010 Night Hours Only Minutes Above			Year 2015 Night Hours Only Minutes Above			Year 2020 Night Hours Only Minutes Above		
Location	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA	65 dBA	77 dBA	85 dBA
Q1	0.0	0.0	0.0	0.1	0	0	0,1	0.0	0.0		0.0				0
OCI	0.5	0.1	0.0	1.5	0	0	1.2	0.0	0.0	1.5					0
OC2	0.5	0.1	0.0	4.3	0.1	0	5.4	0.1	0.0	6.4	0.1	<b>0</b> .0	I		
OC3	0.6	0.0	0.0	3.1	0	0	7.0	0.2	0.0	8.3	0.3	0.0		0.3	0
OKE	0.4	0.0	0.0	0.1	0	0	0.2	0.0	0.0	0.2	0.0	0.0	0.2	0	0
PHI	0.2	0.0	0.0	2.6	0.2	0	4.6	0.3	0.0	5.4	0.3	0.0	6.6	0.3	0
SCC	0.1	0.0	0.0	3.5	0.2	0	6.1	0.4	0.0	7.2	0.5	0.0	7.4	0.3	0
SM1	0.1	0.0	0.0	2.7	0.2	0	4.6	0.3	0.0	5.5	0.4	0.0	6.5	0.2	. 0
SM2	0.0	0.0	0.0	2.2	0.1	0	3.8	0.2	0.0	4.6	0.2	0.0	5.1	0.1	0
SM3	0.0	0.0	0.0	1.5	0	0	2.6	0.0	0.0	3.2	0.1	0.0	3.5	0	0
SM4	0.0	0.0	0.0	0.1	0	0	0.2	0.0	0.0	0.2	0.0	0.0	0.2	0	0
SM5	0.0	0.0	0.0	0.2	0	0	0.4	0.0	0.0	0.4	0.0	0.0	0.3	0	0
SMA	0.5	0.0	0.0	3	0	0	5.4	0.0	0.0	6.4	0.0	0.0	7.3	0	0
TI	0.0	0.0	0.0	0	0	0	0.0	0.0	0.0	0.0	0.0	0.0	0	0	0
WCE	0.5	0.0	0.0	0.9	0	0	2.8	0.0	0.0	3.4	0.0	0.0	3.9	0	1 0

During a 24-hour period, at the 77 dBA TA level the Proposed Project during the phasing years increases exposure at sites AH1 during Phase 2 and Phase 3, and at EO during all 3 Phases, and reduces exposures at all the other sites.

During a 24-hour period, at the 65 dBA level, the Proposed Project decreases exposure at 19 sites and increases exposure at 35 sites during Phase 2 relative to 1998 conditions.

During night hours, at the 85 dBA level, the Proposed Project creates no sites with this level during any of the phasing years. During night hours, at the 77 dBA level, the Proposed Project creates additional exposure at 16 sites during Phase 2 while decreasing exposure levels at two sites. During night hours at the 65 dBA level at Phase 2, the Proposed Project reduces exposure at 7 sites, makes no change at 8 sites, and increases exposure at the remaining 39 sites.

# 11.3.4.7 Summary of Aviation Noise Impacts Under the Proposed Project

"Implementation of the Proposed Project would not result in any existing residential uses inside the project case 2020 65 CNEL contour, or any interim year contours." (4.4.-91.)

"Lower cumulative and single event noise levels, but more noise events' best summarizes the differences between existing military operations at the MCAS El Toro site and the civilian aircraft operations at OCX under the Proposed Project [at the Phase 2 implementation level]. With the exception of the North Corridor (Runway 34 departures) where civilian aircraft will fly where military aircraft did not, there are general decreases in the higher levels of CNEL and SENEL in terms of both contours and at sensitive receptor sites. The noise increases in the Northern Corridor do not affect incompatible land uses. However, as discussed in [Draft EIR No. 573] Section 4.2 and in the discussion related to Mitigation Measure GPC-2, aviation noise impacts from the Proposed Project [at the Phase 2 implementation level] in the vicinity of MCAS El Toro would not be considered significant. The Proposed Project would not result in any significant aviation noise impacts in the vicinity of El Toro, and is expected generally to result in lower noise levels and effects than the existing conditions around JWA.

"Although the analysis indicates that the project would not result in any significant aviation noise impacts in the vicinity of El Toro by reference to traditional, adopted, or generally accepted significance criteria, the findings adopted for EIR No. 563 concluded that a civilian airport at MCAS El Toro would result in significantly greater numbers of total operations at the base compared to historical military levels of use, both throughout the day and during the nighttime hours. Although the Proposed Project analyzed in this study forecasts significantly fewer operations than would have occurred under the community reuse plan analyzed in EIR No. 563, the number of forecast civilian operations is still substantially greater than the "baseline" level of military operations. (EIR No. 563 compared the reuse plan project to a 1994 military operations level. This study compares the proposed project to the 1998 level of military operations, which was smaller than the 1994 level of activity.)

"As discussed earlier, the CNEL calculation factors in the number of daily operations and assigns a 'penalty weighting' to operations occurring during the nighttime hours (10 p.m. to 7 a.m.). However, the substantial increase in the absolute number of operations to and from El Toro under the Proposed Project, [at build out and at the Phase 2 implementation level] particularly during nighttime hours, is still of sufficient magnitude compared to the level of military activity that it will be considered a significant impact of the project independent of the CNEL computation and established thresholds of significance. Although there is no established threshold of significance, and although the available data indicate that any risk of project caused sleep disturbance during nighttime hours would be low, the potential for sleep disturbance during nighttime operations will be treated as a significant impact for purposes of analyzing possible mitigation measures later in this section." (4.4-91-92.)

### 11.3.4.8 Surface Ground Transportation Noise

### Long Term Ground Transportation Noise Impacts

The assessment of long-term ground transportation noise impacts associated with the Proposed Project, as presented in Draft EIR No. 573, assumes vehicle levels commensurate with Phase 4 build out of the proposed project utilizing the existing conditions roadway infrastructure. (4.4-92.) This methodology results in impacts presented under a worse case scenario. Ground transportation noise impacts associated with the Proposed Project during the interim phasing years, including Phase 2, will be less proportionately than those at Phase 4 due to the proportionate reduction in vehicle trips occurring during the interim years. However, this supplemental analysis assumes significant impacts at the Phase 2 level where also identified at Phase 4. See Draft EIR No. 573, pages 4.4-92 through 4.4-140 for complete discussion regarding long-term ground transportation noise impacts under the Proposed Project.

# 11.3.4.9 Short-Term Construction Noise Impacts from Road Improvements

"Road improvements may be required at [certain] locations and short term noise impacts on sensitive land uses may occur during construction of these improvements. County of Orange standard requirements for construction noise mitigation would apply during the construction of these improvements as described [in Draft EIR No. 573]. In addition, a mitigation measure is provided ... to mitigate potential short term noise impacts during construction of off-site road improvements under the Proposed Project." (4.4-141.) This conclusion is unaffected by the project phasing and applies equally to the Proposed Project at the Phase 2 implementation level.

### 11.3.4.10 Proposed Mitigation Measures

"The following noise mitigation measures, not already part of the Proposed Project, are proposed for adoption and implementation as part of the project:

- N-6 The County will participate in a fair share program to construct or extend existing sound walls at relevant portions of Trabuco Road between Yale and Jeffrey Road, and along Jeffrey Road just south of Irvine Boulevard, to reduce project caused traffic noise levels in any residential property to a level at or below 65 CNEL.
- N-7 For all project construction, the County will adhere to the County Noise Ordinance and, for off-site improvements in any incorporated areas, any existing applicable local jurisdiction noise ordinances regulating construction noise.
- N-8 The County will construct and implement a noise monitoring system at OCX prior to the opening for public use of the passenger terminal facilities contemplated under the Proposed Project to be completed by the end of Phase 1 of the Proposed Project (2005). This noise monitoring system will include remote microphone stations at locations substantially as shown in Figure 4.4-52 of EIR No. 573. In addition, upon completion and County acceptance of the noise monitoring system, the County will implement a noise complaint and administration program and staffing comparable to its existing program at JWA.
- N-9 Prior to the initiation of scheduled commercial operations at El Toro, the County will implement an 86 dB SENEL noise mitigation restriction for El Toro during the hours of 11:00 p.m. and 7:00 a.m. The noise level limit will be based upon quarterly energy average SENEL values for scheduled commercial operators by operator and aircraft type and will be enforced at the remote microphone stations installed and monitored pursuant to mitigation measure N-8." (4.4-183.)

## 11.3.4.11 Level of Significance After Mitigation

"Implementation of the proposed mitigation measures will result in the following levels of significance:

- Construction noise will be reduced to a level less than significant.
- Highway noise will be reduced to a level less than significant. [In the event off-site improvements proposed under Mitigation Measure N-6 are not implemented by the jurisdiction in which the improvements are located, the impacts at those unimproved locations will remain significant and unavoidable.<sup>1</sup>]

Underlined text is new text added to Draft No. 573, Section 4.4.9, unrelated to this supplemental phasing analysis. See Chapter 6.0, Other Environmental Issues, infra.

- No mitigation is proposed for the increase in aircraft operations, because mitigation to reduce aircraft operations would not be compatible with the project objectives. Therefore, these impacts remain significant and unavoidable. (See Alternatives A, H, and I in Chapter 8.0 for analysis of reduced aircraft operations.)
- Mitigation is proposed for the increase in nighttime operations and the resulting increase in the risk of sleep disturbance and impacts are reduced; however, these impacts remain significant and unavoidable." (4.4-183-184.)

## 11.3.5 Air Quality

Chapter 2.0 of the supplemental analysis to Draft EIR No. 573 evaluates in detail the potential air quality impacts associated with the Proposed Project under each development scenario (i.e., Phase 1, Phase 2, Phase 3, and Phase 4) for the MCAS El Toro site and John Wayne Airport site. The Air Quality Technical Report (Appendix N) to the supplemental analysis provides additional detail on the methodologies used to estimate emissions, analyze ambient air quality concentrations, and identify mitigation options for each of these development scenarios.

The air quality analysis focuses on four areas of concern: short-term impacts (construction), total operational emissions impacts (regional air quality), local air quality impacts due to traffic carbon monoxide (CO), and local air quality impacts due to aircraft and associated operations. The analysis measures the Proposed Project's impacts and their significance against existing physical conditions. A summary of the conclusions in this analysis with respect to the Proposed Project's impacts on Phase 2 of project development is provided below.

## 11.3.5.1 Unmitigated Short-Term Impacts (Construction Emissions)

Although construction impacts are temporary in nature, the Proposed Project would result in significant and unavoidable construction air quality impacts under each development scenario based on the preliminary estimates of construction phasing and projected construction activity levels. Table 5.3.5-1 provides a summary of the unmitigated short-term construction impacts of the Proposed Project during Phase 1 and Phase 2 of construction. As this table indicates, emissions associated with construction traffic and equipment exhaust fluctuate between different years during phase 1 and phase 2 of construction. Emissions would be greatest during the third year of each phase because the majority of construction work would occur during this year. The peak year for short-term emissions from construction under the Proposed Project is the third year of Phase 2. NO<sub>x</sub> emissions would exceed the daily threshold of 100 pounds per day for construction during all years of construction in phases 1 and 2. CO emissions would exceed the daily threshold of 550 pounds per day during years 2, 3 and 4 of phases 1 and 2. ROC emissions would exceed the daily threshold of 75 pounds per day during years 2, 3, 4, and 5 of Phases 1 and 2. SO<sub>x</sub> emissions are expected to exceed the daily threshold during year 2 of Phase 2.

Table 5.3.5-1
Unmitigated Construction Traffic and Equipment Exhaust Emissions,
Proposed Project

	Estimated Emission Rate (lb/day)									
Scenario	CO	ROC	NOx	SO <sub>x</sub>	PM <sub>10</sub>					
Phase 1 – Year 1	469.9	66.8	856.9	78.4	39.5					
Phase 1 – Year 2	871.9	114.4	1,340.9	119.4	79.1					
Phase 1 – Year 3	961.8	136.4	1,155.3	106.1	66.6					
Phase I – Year 4	908.1	1,705.9	1,009.7	85.6	51.5					
Phase 1 – Year 5	246.1	6,757.9	203.9	16.9	9.6					
Phase 2 – Year 1	320.9	38.1	599.4	50.4	31.5					
Phase 2 – Year 2	1,045.0	127.0	1,858.2	174.5	118.1					
Phase 2 – Year 3	1,195.9	154.9	1,608.4	154.4	99.5					
Phase 2 – Year 4	747.8	1,308.7	864.2	72.8	41.1					
Phase 2 – Year 5 (Half a year of Construction Activity)	421.9	8,169.4	654.1	57.0	30.3					
SCAQMD Construction Threshold	550	75	100	150	150					

Source: P&D Consultants and LSA Associates, Inc., 2001.

NOTE: [1] Numbers in bold represent emissions that exceed the SCAQMD thresholds for construction.

Although particulate emissions from vehicle and equipment exhaust are not expected to exceed the daily threshold, they need to be combined with fugitive dust emissions from other sources. Table 5.3.5-2 shows the estimated total daily fugitive dust emissions from demolition, grading, hauling/transport, excavation, dumping/reclamation, and other soil disturbance activities by phase and year for the first two phases of construction. Similar to construction traffic and equipment exhaust emissions, daily fugitive dust emissions also fluctuate between the phases and between the years within a construction phase. As can be seen in Table 5.3.5-2, the third year of Phase 2 would have the highest daily fugitive dust emission level of all construction years. Total daily fugitive dust emissions associated with construction of the Proposed Project during all of the construction years during the first two phases would also exceed the SCAQMD threshold of 150 pounds per day. These exceedances are reflected in bold type in Table 5.3.5-2.

In addition to these exceedances of the SCAQMD thresholds, the Proposed Project would also result in significant construction concentrations for  $PM_{10}$ ,  $NO_2$ , and CO at a number of receptor locations during peak years for each pollutant. These peak year concentrations would occur close to the fenceline and decrease as the plume spreads out from the property. See Draft Supplemental Analysis, Chapter 2.0, Air Quality Analysis, Figures 2-4 through 2-7 and Tables 2-18 through 2-20.

Table 5.3.5-2
Unmitigated Fugitive Dust Emissions, Proposed Project

Scenario	Fugitive Dust Emission (lb/day)
Phase 1 – Year 1	662.2
Phase 1 – Year 2	1,159.1
Phase 1 – Year 3	1,352.9
Phase 1 – Year 4	1,266.5
Phase 1 – Year 5 (Half a year of Construction Activity)	244.9
Phase 2 – Year 1	198.6
Phase 2 – Year 2	1,882.2
Phase 2 – Year 3	2,116.6
Phase 2 – Year 4	1,146.8
Phase 2 – Year 5 (Half a year of Construction Activity)	481.9
SCAQMD Construction Threshold	150

Source: P&D Consultants and LSA Associates, Inc., 2001.

NOTE: [1] Numbers in bold represent emissions that exceed the SCAQMD thresholds for construction.

## 11.3.5.2 Unmitigated Operational Impacts

Aircraft and Associated Operations Impacts – The Proposed Project would not result in exceedances of either the California Ambient Air Quality Standards (CAAQS) or the National AAQS (NAAQS) at most of the receptor site locations during Phase 2 of project development. There would, however, be exceedances of the 1-hour State standard for NO<sub>2</sub> and a continued exceedance of the 24-hour State standard for PM<sub>10</sub> projected at a number of locations surrounding both OCX and JWA. The NO<sub>2</sub> and PM<sub>10</sub> impacts cannot be mitigated to a level below significance and would, therefore, remain significant and unavoidable. See Draft Supplemental Analysis, Chapter 2.0, Air Quality Analysis, Tables 2-33 through 2-37 and Figures 2-10 through 2-12.

Traffic CO Impacts – The Proposed Project would not result in any CO hot spots at intersections in the project vicinity under Phase 2 of project development. The Proposed Project, when added to the existing conditions (1998) roadway network and traffic conditions, would continue to expose receptors near two intersections in the project vicinity to eight-hour CO concentrations exceeding the State and federal standards under Phase 2 of project development. However, this exceedance is due primarily to high existing ambient eight-hour CO concentrations. The Proposed Project would not increase the frequency or severity of this CO exceedance. Therefore, the Proposed Project at Phase 2 of project development would not result in any significant local air quality impacts due to traffic CO hot spots.

Regional Impacts — The Proposed Project at Phase 2 of project development plus existing conditions (1998) when compared to existing conditions (1998) would result in exceedances of State and federal standards for all criteria pollutants (CO, NO<sub>x</sub>, ROC, and PM<sub>10</sub>). See Table 5.3.5-3. The Phase 2 project, when compared to Phase 2 of the No Project/No Build Alternative, however, would not result in fewer-any regional air quality impacts. This is primarily due to locally generated demand being served at other regional airports under the No Project Alternative.

# 11.3.5.3 Proposed Mitigation Measures and Significance After Mitigation

A detailed description of proposed mitigation to mitigate the identified significant short-term (construction) air quality impacts is provided in Chapter 2.0, Section 2.10 of this Draft Supplemental Analysis. Tables 5.3.5-5 and 5.3.5-6 provide a summary of the mitigated construction traffic and equipment exhaust emissions for phases 1 and 2 of the Proposed Project. As can be seen from the tables, although a majority of the construction impacts can be mitigated to a level below the level of significance through implementation of proposed mitigation, NO<sub>x</sub> emissions from traffic and equipment exhaust during years 2 and 3 of phase 2 of construction remain significant and unavoidable. Fugitive dust emissions remain significant and unavoidable even after implementation of proposed mitigation for every year of construction during the first two phases of project development except for year 1 of phase 2.

A detailed description of the proposed mitigation to mitigate the operational air quality impacts is also provided in Chapter 2.0, Section 2.10 of this Draft Supplemental Analysis. As can be seen from the tables provided in Chapter 2.0, although proposed mitigation can substantially reduce the local and regional air quality impacts of the Proposed Project in Phase 2 of project development, the exceedances of the 1-hour State standard for NO<sub>2</sub> and the continued exceedence of the 24-hour State standard for PM<sub>10</sub> from aircraft and associated operations and the regional air quality impacts remain significant and unavoidable. Table 5.3.5-7 provides a summary of the significance of the project's air quality impacts after mitigation at Phase 2 of project development.

#### 11.3.5.4 Health Effects from Toxic Air Contaminants

A detailed analysis of the health effects from toxic air contaminants (TACs) that may result from the Proposed Project is provided in Chapter 2.0, Air Quality, of this Draft Supplemental Analysis. At Phase 2 of the Proposed Project, the TAC impacts will be of a lesser degree than those impacts projected to occur at Phase 4 build out. Nevertheless, the TAC impacts at Phase 2 are significant and unavoidable.

**Table 5.3.5-3** Regionwide Emissions Inventory Existing Conditions Plus Project (Pounds/Day Unless Noted) Phase 2

		Existing Co	nditions (199	(8)			Existing Conditions (1998) Plus Proposed Project (Phase 2)						
		CO	NOX	ROC	SOx	PM <sub>10</sub>		cò	NOx	ROC	SOX	PM <sub>10</sub>	
Aircraft	El Toro	1,451.49	365.51	536.16	21.47		Aircraft El Toro	6,626.97	8,242.81	1,326.72	568.65	259.78	
	JWA	7,190.15	2,579.29	515.49	199.72	39.83	JWA	12,939.34	3,846.77	795.30	307.05	62.36	
	Other Airports	41,273.20	36,139.03	2,890.80	619.17	88.43	Other Airports	98,490.25	98,941.82	11,240.86	5,404.33	771.85	
	Total Regional	49,914.84	39,083.83	3,942.45	840.36	291.71	Total Regional	118,056.56	111,031.40	13,362.88	6,280.03	1,093.74	
GSE	El Toro	1,443.08	47.02	148.03	0.62	3.48	GSE El Toro	14,041.42	1,163.01	514.64	55.95	47.52	
	JWA	5,445.93	516.29	163.02	25.08	20.46	JWA	8,500.95	938.90	267.17	36.43	39.08	
	Other Airports	57,782.48	4,631.09	819.72	67.48	38.21	Other Airports	138,040.73	12,690.39	3,194.40	589.50	333.73	
	Total Regional	64,671.49	5,194.40	1,130.77	93.18	62.15	Total Regional	160,583.11	14,792.30	3,976.21	681.88	420.33	
Energy	El Toro	29.00	166.80	1.60	17.10		Energy El Toro	99.90	574.50	5.40	58.90	19.70	
	JWA	13.20	76.10	0.80	7.40	2.50	JWA	27.90	160.70	1.60	16.10	5.40	
	Others	157.00	904.00	10.00	89.70	30.00	Others	595.00	3,426.00	34.00	346.80	116.00	
	Total Regional	199.20	1,146.90	12.40	114.20	38.20	Total Regional	722.80	4,161.20	41.00	421.80	141.10	
Fuel	El Toro			1.30	**		Fuel El Toro	**		50.24			
	JWA			8.36	**		JWA	**	***	13.12			
	Other Airports			146.52			Other Airports	**		567.09			
	Total Regional	**		156.18	**	**	Total Regional		~**	630.45		4.0=	
Airport Road		241.22	42.04	77.60	* 01	***	Airport Roadways El Toro	475.88	87.12	29.98	4.16	4.87	
1	JWA	341.23	43.24	37.50	5.81	1.13	JWA	411.45	52.20	41.54	6.11	1.69	
	Other Airports	2,476.39	413.49	83.16	4.37 10.18	5.94 7.07	Other Airports	7,287.82	1,284.79	394.64	42.86 53.13	53.10 59.66	
Airport Park	Total Regional	2,817.62	456.73	120.66	10.16	7.07	Airport Parking El Toro	8,175.15 335.87	1,424.11 30.36	466.16 9.89	2.98	2.77	
Airport Park	ing El Toro JWA	299.24	23.76	44.13	0.19	0.29	Airport Parking El Toro JWA	355.82	28.42	51.76	1.60	0.42	
	Other Airports	1,547.75	124.05	11.88	5.83	2.30		3,765.39	326.03	50.51	51.78	20.44	
1	Total Regional	1,846.99	147.81	56.01	6.02	2.59	Total Regional	4,457.08	384.81	112.16	56.36	23.63	
Roads	El Toro	2,929.00	594.00	176.00	51.00		Roads El Toro	18,279.00	5,196.00	1,597,00	347.00	1,941,00	
Roads	127 1010	2,727.00	374.00	170.00	31.00	11.00	110102	19,991.00	5,874.00	1,724.00	356.00	2,244.00	
	JWA	12,464.00	3,404.00	1,260.00	122.00	931.00	JWA2	15,708.00	4,451.00	1,540.00	175.00	1,376.00	
	3,47.	12,707.00	3,707.00	1,200.00	122.00	231.00	]	13,700.00	4,451.00	1,540.00	175.00	1,370.00	
i	Other Airports2	1,386,436.00	438,423.00	93,401.00	42,830,00	7,519.00	Other Airports	4,314,989.00	989,920.00	201,337.00	88,513.00	14.055.00	
		3,295,718.00	745,988.00	228,623.00	42,657.00	13,478.00		6,243,266.00	1,300,898.00	338,823.00	87,985.00	19,965.00	
	Total Regional2	1,401,829.00	442,421.00	94,837,00	43,003.00	8,461,00	Total Regional <sup>2</sup>	4.348,976.00	999,567.00	204,474.00	89,035.00	17,372.00	
[		3,311,111.00		230,059.00	42,830.00	14,420.00		6,278,965.00		342,087.00	88,512.00	23,585.00	
TOTAL (D	ounds/day)	1,521,279,14	488,450.67	100,255,47	44,066,94	8,862.72	TOTAL (pounds/day)	4,640,970.70	1,131,360.82	223,062.86	96,528.20	19,110,46	
	*/	3,430,561.14	796,015.67	235,477.47	43,893.94	14,821.72	1 1 1	6,570,959.70	1,443,016.82	360,675.86	96,005.20	25,323.46	
1 Revised	calculation of avera		<u></u>		· -		Change from Existing Condition		642,910.15	122,807,39	52,461.26	10,247,74	
	ance determinations				any or my		(pounds/day)	1 ' '	647,001.15	125,198.39	52,111.26	10,501.74	
				<del></del>	Later 8001 J		Change from Existing Condition	569,343.71	117,331.10	22,412.35	9,574.18	1,870.21	
= iypugi	Typographical correction (year 2005 data was inadvertently used instead of 1998 data).					(tons/year)	573,122.74	118,077.71	22,848.71	9,510.30	1,916.57		
						SCAQMD Threshold for	550	55	55	150	150		
Source: CH	Source: CH2M HILL, P&D Consultants, and LSA Associates, Inc. 2001						Operation (pounds/day)	] 330	]		130	130	

Source: CH2M HILL, P&D Consultants, and LSA Associates, Inc. 2001

Table 5.3.5-4 Regionwide Emissions Inventory Phase 2 Proposed Project/No Project (Pounds/Day Unless Noted)

		No Proj	ect (Phase 2)			*	Proposed Project (Phase 2)							
		CO	NOX	ROC	sox	PM <sub>10</sub>		CO	NOX	ROC	SOX	PM <sub>10</sub>		
Aircraft	El Toro	**		**			Aircraft El Toro	5,175.48	7,877.30	790.56	548.57	96.08		
	JWA	7,237.35	3,117.14	415.07	246.83	45.72	JWA	5,749.19	1,267.48	279.81	107.33	22.53		
	Other Airports	64,338.22	70,647.13	9,401.21	5,385.93	768.34	Other Airports	57,217.05	62,802.79	8,350.06	4,785.16	683.42		
	Total Regional	71,575.57	73,764.27	9,816.28	5,632.76	814.06	Total Regional	68,141.72	71,947.57	9,420.43	5,439.67	802.03		
GSE	El Toro		**		**		GSE El Toro	12,598.35	1,115.99	366.61	55.33	44.04		
	JWA	5,914.70	634.43	181.56	16.21	28.08	JWA	3,055.02	422.61	104.15	11.35	18.62		
	Other Airports	90,189.58	9,056.56	2,668.50	586.59	332.10			8,059.30	2,374.68	522.02	295.52		
	Total Regional	96,104.28	9,690.99	2,850.06	602.80	360.18	Total Regional	95,911.62	9,597.90	2,845.44	588.70	358.18		
Energy	El Toro	a		**	**	==	Energy El Toro	70.90	407.70	3.80	41.80	14.00		
	JWA	20.30	117.10	1.10	12.00	4.00	JWA		84.60	0.80	8.70	2.90		
	Others	492.00	2,832.00	26.00	290.00	97.00			2,522.00	24.00	257.10	86.00		
	Total Regional	512.30	2,949.10	27.10	302.00	101.00	<u>Total</u> Regional		3,014.30	28.60	307.60	102.90		
Fuel	El Toro				**	**	Fuel El Toro		**	48.94	••			
	JWA			10.23		**	JWA	1		4.76				
	Other Airports			472.61		wa	Other Airports			420.57				
	Total Regional			482.84	**	**	<u>Total</u> Regional		**	474.27				
Airport Roadway		***					,		87.12	29.98	4.16	4.87		
	JWA	147.64	18.07	8.41	0.55	1.15		1	8.96	4.04	0.30	0.56		
	Other Airports	3,864.54	803.26	271.90	37.98	51.60			871.30	311.48	38.49	47.16		
	Total Regional	4,012.18	821.33	280.31	38.53	52.75			967.38	345.50	42.95	52.59		
Airport Parking	El Toro					***	Airport Parking El Toro		30.36	9.89	2.98	2.77		
	JWA	120.73	9.92	16.28	3.01	0.28		1	4.66	7.63	1.41	0.13		
	Other Airports	2,492.05	226.97	43.40	51.64	20.38			201.98	38.63	45.95	18.14		
	Total Regional	2,612.78	236.89	59.68	54.65	20.66			237.00	56.15	50.34	21.04		
Roads	El Toro	**			w.m.		Roads El Toro.		4,602.00	1,421,00	296.00	1,930.00		
								17,062.00	5,280.00	1,548.00	305.00	2,233.00		
	JWA	6,937.00	2,238.00	600.00	112.00	952.00			1,047.00	280.00	53.00	445.00		
	Other Airports2		<b>558,499.00</b>	110,864.00	45,755.00	8,243.00	Other Airports2		<del>551,497.00</del>	107,936,00	4 <del>5,683.00</del>	6,536.00		
		2,965,980.00	559,703.00	111,572.00	45,643.00	8,228.00		2,947,548.00		110,200.00	45,328.00	6,487.00		
	Total Regional2		560,737.00	111,464.00	45,867.00	9,195.00	Total Regional2		<del>557,146.00</del>	109,637.00	46,032.00	8,911.00		
		2,972,917.00	561,941.00	112,172.00	45,755.00	9,180.00		2,967,854.00		112,028.00	45,682.00	9,165.00		
TOTAL (poun	ds/day)	3,141,756.11		124,980.27	52,497,74	10,543.65	TOTAL (pounds/day)	3,119,691,56		122,807.39	<b>5</b> 2,461.26	10,247.74		
<u> </u>		3,147,734.11		125,688.27	52,385.74	10,528.65		3,140,398.56		125,198.39	52,111.26	10,501.74		
1 Revised calc	culation of average	e trip length. Th	his revision doe	es not impact an	y of the signif	icance	Change from No Project	1 ' '	(5,289.43)	(2,172.88)	(36.48)	<del>(295.91)</del>		
	ons made in conne	ction with the p	roject.				(pounds/day)		(2,402.43)	(489.88)	(274.48)	(26.91)		
2 Typographic	cal correction.						Change from No Project		(965.32)	(396.55)	(6.66)	<del>(54.00)</del>		
·							(tons/year)	(1,338.74)	(434.44)	(89.40)	(50.09)	(4.91)		
							SCAQMD Threshold for	550	55	55	150	150		
Source: CH2M H	HLL, P&D Cons	ultants, and LS	A Associates,	Inc. 2001			Operation (pounds/day)		33	33	130	130		
								1	I					

Table 5.3.5-5
Mitigated Construction Traffic and Equipment Exhaust Emissions, Proposed Project

	Estimated Emission Rate (lb/day)								
Scenario	CO	ROC	NOx	SOx	PM10				
Phase 1 - Year 1	243.3	19.8	80.0	7.8	2.1				
Phase 1 - Year 2	368.5	28.8	99.5	11.5	4.1				
Phase 1 - Year 3	342.9	31,4	86.2	9.8	3.7				
Phase I - Year 4	363.8	37.3	69.2	7.2	3.5				
Phase 1 - Year 5°	63.6	41.1	12.5	1.3	0.6				
Phase 2 - Year 1	151.5	10.2	45.1	5.0	1.5				
Phase 2 - Year 2	515.9	36.6	146.0	17.3	5.9				
Phase 2 - Year 3	477.4	39.1	124.7	14.6	5.2				
Phase 2 - Year 4	351.9	34.6	56.4	5.7	3.0				
Phase 2 - Year 5*	227.7	60.2	46.3	5.0	2.2				
SCAQMD Construction Threshold	550	75	100	150	150				

Source: P&D Consultants, LSA Associates, Inc. 2001.

NOTE: [1] Numbers in bold represent emissions that exceed the SCAQMD thresholds for construction.

<sup>\*</sup> Half a year of construction activity.

Table 5.3.5-6
Mitigated Fugitive Dust Emissions, Proposed Project

Scenario	Fugitive Dust Emission (lb/day)
Phase 1 - Year 1	211.6
Phase 1 - Year 2	402.7
Phase 1 - Year 3	485.5
Phase 1 - Year 4	879.1
Phase 1 - Year 5*	179.7
Phase 2 - Year 1	95.3
Phase 2 - Year 2	355.8
Phase 2 - Year 3	572.9
Phase 2 - Year 4	1008.2
Phase 2 - Year 5*	409.3
SCAQMD Construction Threshold	150

Source: P&D Consultants, LSA Associates, Inc., 2001.

NOTE: [1] Numbers in bold represent emissions that exceed the SCAQMD thresholds for construction.

<sup>\*</sup> Half a year of construction activity.

Table 5.3.5-7
Significance of Air Quality Impacts After Mitigation – Phase 2

		Operations	
Criteria Pollutant	On-Airport	Off-Airport	Construction <sup>3</sup>
CO			
EMS <sup>I</sup>	S	S	LS
DISP <sup>2</sup>	LS	LS	S
ROC			
EMS	S	S	LS
DISP	LS	LS	LS
NO <sub>x</sub> or (NO <sub>2</sub> )			
EMS	S	S	S
DISP	S	LS	S
$SO_2$			
EMS	S	S	LS
DISP	LS	LS	LS
PM <sub>10</sub> or			
(fugitive dust)			
EMS	S	S	S
DISP	S	LS	S

LS = Less than Significant.

S = Significant

NA = Not applicable.

- 1 EMS = Emissions: significance determined by comparison of incremental emissions to the thresholds in Draft Supplemental Analysis, Chapter 2.0, Table 2-15.
- 2 DISP = Dispersion: significance determined by comparison of predicted ambient concentration to NAAQS and CAAQS in Draft Supplemental Analysis, Chapter 2.0, Table 2-13.
- 3 Construction impacts are based upon the peak emissions year, which are Phase 2, Year 3 for PM<sub>10</sub>, and Phase 2, Year 2 of NO<sub>c</sub>, and CO.
- 4 Significance based on NO<sub>2</sub> concentration predictions for unmitigated peak operations and estimated reductions from mitigation measures.

## 11.3.6 Landform and Topography

# 11.3.6.1 Construction Related Impacts of the Proposed Project Related to Landform and Topography

Grading operations having the most potential to impact topography are scheduled to take place during all four phases of the Proposed Project. The areas where major grading operations are planned, and the total amounts of soil to be cut or used as fill (rounded to the nearest 1,000 cubic yards), are as follows:

Phase 1 (2000-2005)

Cut: Terminal building, landside and airfield Total Cut, all areas: 6,249,000 cubic yards

Fill: Airfield, apron

Total Fill, all areas: 4,574,000 cubic yards

Planning Areas: 1,8

Phase 2 (2006-2010)

Cut: Airfield, terminal building, landside and cultural institutional area

Total Cut, all areas: 10,451,000 cubic yards

Fill: Airfield, apron

Total Fill, all areas: 10,125,000 cubic yards

Planning Areas: 1, 2, 8

Phase 3 (2011-2015)

Cut: Airport maintenance area

Total Cut, all areas: 3,578,000 cubic yards

Fill: Apron

Total Fill, all areas: 3,797,000 cubic yards

Planning Areas: 1, 2, 3

Phase 4 (2016-2020)

Cut: East cargo area

Total Cut, all areas: 2,581,000 cubic yards

Fill: Airfield, apron

Total Fill, all areas: 4,364,000 cubic yards

Planning Areas: 1, 3, 8<sup>1</sup>

In response to a comment submitted on Draft EIR No. 573, cubic yards shown are revised from those presented in Draft EIR No. 573 as the result of an adjustment unrelated to this supplemental phasing analysis. The revisions will be included as part of Final EIR No. 573 and do not constitute significant new information, nor do they result in significant impacts not previously identified.

With implementation of the Proposed Project limited through Phase 2, cut and fill operations at the end of Phase 2 will result in excess cut of approximately 2 million cubic yards. The best candidate areas to receive these fills are the areas planned for the terminal apron in Phases 3 and 4. Referring to Technical Report 19, Volume 2 of 2, Table B-4 and Figure 3 in Attachment N, the preliminary fill volumes required for Phase 3 terminal apron (Subarea 3-4 in Table B-4) and Phase 4 terminal apron (Subarea 4-5 in Table B-4) are 1.8 and 1.5 million cubic yards, respectively. This amounts to approximately 3.3 million cubic yards total volume of fill. Taking into account the 4:1 slopes from the finish grade elevations of Phases 3 and 4 apron needed to join the ground elevations constructed in Phases 1 and 2, these two areas should be sufficient to receive the 2 million cubic yards remaining after Phase 2 is completed. An alternative location to receive the excess fill is Subarea 4-2 in the airfield where Runway 7L is planned in Phase 4. This area can receive approximately 1.8 million cubic yards of fill and can be used as an alternative over the terminal apron areas. Whichever location is selected, the resulting fill will be compacted and stabilized, thus avoiding any stockpiling of fill material. The Proposed Project, therefore, will not result in significant impacts at the Phase 2 implementation level.

## 11.3.7 Soils, Geology and Seismicity

"Because the Proposed Project is a large-scale construction project, it has the potential to impact, or be impacted by, a variety of geological/geophysical factors. Impacts of the Proposed Project in relation to geology, soils, and seismicity are not anticipated to be phase-sensitive, except insofar as the timing of construction would result in the presence of buildings or other facilities that could be impacted." (4.7-13.)

## 11.3.7.1 Mitigation Measures

Final EIR No. 563 Mitigation Measure G-4, as revised by EIR No. 573, provides, in part, that "prior to obtaining a grading permit for project construction, the County of Orange will require that detailed geotechnical and hydrological reports be prepared specifically addressing any needed modifications to the existing drainages on the MCAS El Toro Site." (4.17-17). No additional mitigation measures beyond those set forth in Final EIR No 563, as revised for EIR No. 573, were identified.

## 11.3.7.2 Level of Significance After Mitigation

No significant impacts associated with issues related to soils, geology, or seismicity that cannot be mitigated to below a level of significance were identified at the Proposed Project's Phase 2 level.

# 11.3.8 Hydrology and Water Quality

"The degree of hydrology and water quality impacts is most affected by the amount of impervious surfaces within a given area: the greater the impervious surface area, the greater the runoff and corresponding impacts. In 2020, impervious surface area under the Proposed Project will be maximized for the types of land use that are proposed. Because Pre-2005 interim land uses include less total area of impervious surface cover than under the Proposed Project at build out, hydrology and water quality impacts will be less under Pre-2005 conditions than under project build out conditions. [¶] Phased development of the Project in the year 2005 (opening day), 2010 and 2015 will also have a smaller impervious surface area than the 2020 build out conditions of the Proposed Project. Therefore, there will be no greater impacts on hydrology under any of the development phases than under the Proposed Project." (4.8-17.) Because the Proposed Project at Phase 4 is not expected to result in significant hydrological impacts, impacts during the phasing years, including Phase 2, will be below the level of significant.

"[A]ny potential impacts to surface or groundwater quality that may result with implementation of the Proposed Project will be reduced to a level below significance based, in part, on improvements that will be made as part of the project and also, in part, on compliance with regulatory controls designed to maintain and improve the water quality of the MCAS El Toro receiving waters." (4.8-26.) This is also the case with respect to the interim phasing years, including Phase 2 -- any potential impacts will be below the level of significant based on project improvements and compliance with applicable regulatory controls.

# 11.3.9 Biological Resources

"Virtually all of the direct impacts at the MCAS El Toro Site and the Adjacent Federal Habitat Reserve (e.g., loss of agricultural lands, demolition, etc.) will occur in Phase 1." These impacts, however, are not significant (4.9-28-29.)

The Proposed ASMP Project will not result in significant adverse indirect impacts at the MCAS El Toro Site and the Adjacent Federal Habitat Reserve. Indirect impacts resulting from a change in the noise environment increase gradually over time with project phasing and an increase in aircraft operations. Accordingly such impacts are less during interim Phases 1 and 2 than at Phase 4 project build out. Potential short-term indirect impacts on wildlife are not considered significant and are not affected by project phasing. (See 4.9-29-30.)

The Proposed ASMP Project would not result in significant adverse impacts on native plant communities on the MCAS El Toro site or the adjacent federal Habitat Reserve. Any indirect impacts are not substantially influenced by project phasing. (See 4.9-39.)

"Biological resources associated with drainages or streambeds on the MCAS El Toro site and the adjacent federal Habitat Reserve are limited in their extent and are not well developed. A few soft bottom streambeds are subject to impacts under the Proposed Project. ... An estimated 0.64 acre of disturbed, soft bottom drainages are impacted at an insignificant level by drainage improvements

associated with the Proposed Project at Agua Chinon and at Borrego Wash. This estimated acreage includes a mulefat scrub component/open sandy wash. These washes are not locally or regionally significant, but do include limited riparian resources that do sustain riparian species. These drainages are isolated, fragmented, and do not support extensive riparian plant communities." (4.9-39-42.) The majority of these impacts occur during Phase 1, although some limited impacts occur in subsequent Phase 2.

"Due to the inclusion of the Wildlife Habitat Area, the Proposed Project is expected to result in significant, beneficial effects related to wildlife habitat on the MCAS El Toro site and not preclude opportunities for wildlife movement." (4.9-42-43.) The implementation of the Wildlife Habitat Area occurs during Phase 4. Since the Wildlife Habitat Area is not proposed as a mitigation measure for the project, its implementation during Phase 4 does not effect previous impact conclusions.

The Proposed ASMP Project will not result in significant adverse impacts on special interest species at the MCAS El Toro site or the adjacent federal Habitat Reserve. Any potential impacts that may occur are related to the loss of agricultural habitat, which occurs during the project's Phase 1. (See 4.9-43 - 45.)

#### 11.3.10 Public Services and Utilities

"The Proposed Project at MCAS El Toro will involve development between the years 2000 and 2020 in five year phasing increments. Demand for public services and facilities will increase incrementally with each phase. No phase is anticipated to result in impacts greater than build out of the project." (4.10-48.)

## 11.3.10.1 Impacts Related to Solid Waste

"The Proposed Project solid waste generation is summarized, by phase, in Table 4.10-9. The amount of construction and demolition waste produced from the Proposed Project will create a significant impact." (4.10-25.)

As Table 4.10-9 indicates, during Phase 1, the proposed project will generate approximately 3,400 tons of solid waste per year. This amount increases to 5,400 tons per year in Phase 2. This is a significant adverse impact at the Phase 2 implementation level.

## 11.3.10.2 Mitigation Measures

PS-3 The County of Orange will develop a waste reduction plan for the waste generated from demolition and construction of new facilities and implementation of the Proposed Project in order to comply with State law ABA 939 prior to the commencement of construction for each phase.

Table 4.10-9
Proposed Project Solid Waste Generation

Land Use	Generation Factor	Interim Use	Solid Waste Generated	Phase 1	Solid Waste Generated	Phase 2	Solid Waste Generated	Phase 3	Solid Waste Generated	Phase 4	Solid Waste Generated
Aviation Uses						¥					
Agriculture	ND	-	•	138.9 acres	•	-	-	•	-	-	•
Regional Park	ND	*	•	-	-	•	-	-	•	265.13	•
North Golf Course (182 golfers/day) a 3	0.5lb golfer/day 2	80 acres	0.046 tons/day	186.51 acres	0.05 tons/day	16		*	**	4	*
South Golf Course (220 golfers/day) a 3	0.5lb golfer/day 2	80 acres	0.055 tons/day	97.66 acres	0.055 tons/day	-	*	-	~	-	*
Restricted Open Space	NA NA	•	-	193.8 acres	*	<u></u>	*	•	<u> </u>	*	*
Habitat Reserve	NA	-	•	20.6 acres	*	-	•	+	•	-	-
Wildlife Habitat Area	NA	-	*	-	*	-	-		•	143.6	*
Marshburn Retarding Basin	NA NA	*	-	38.9 acres	*	•	*	-	*	*	-
Cultural/Institutional b	0.0013 tons/SF/year	-	•	*	+	545,000 SF	3.22 tons/day	-	-	-	-
Public Facilities				*		-	-	-	-	-	•
Vehicle Maintenance Yard <sup>b</sup>	0.0108 tons/SF/year		*	51.1 acres	109.27 tons/day		•	*	<b>*</b>	-	м,
Warehouse b	0.0108 tons/SF/year	670,000 <b>S</b> F	32.89 tons/day	11.21 acres	23.96 tons/day	-	*	•	*	-	-
IRWD Reservoir & Pumping	NA	-	•	5.7 acres	-	-	-	-	-	-	•
Fire Station #1d	ND <sup>4</sup>	•	•	1.14 acres	.3 cu, ft/week	-	-	h	*	-	•
Cal Air National Guard a	6 lbs/1000SF/day	-	-	22.9 acres	3.0 tons/day	•	*	*	•	-	*
Homeless Service Providers b	0.0108 tons/SF/year	-	-	26.89 acres	57.5 tons/day	-	-	_	-	-	_
IRWD Facility	NA NA	-	*	9.0 acres	-	-	*	-	•		•
Fire Station #2d	ND <sup>4</sup>		*	0.95 acres	.3 cu. ft/week	•	-	*	-	*	
Business Park <sup>a</sup>	6 lbs/1000SF/day	670,000SF	2.01 tons/day	82.46 acres	10.8 tons/day	-	-	-	-	-	-
Restaurant <sup>C</sup>	0.005 lbs/SF/day 2	30,000SF	0.075 tons/day	*	-	-	-	-	-	•	-
Child Care Center b	0.0013 tons/SF/year	20,000SF	0.11 tons/day	-		-	-		*	•	•
Subtotal Solid Waste Generated	•		35.186 tons/day	*	204.74 tons/day	-	3.22 tons/day		*	-	-
Aviation Uses							1				
Airport (OCX)	238.83 tons/MAP		-	8.8 MAP	2,101.7 tons/year	18.8 MAP	4,490 tons/year	23.4 MAP	5,588.6 tons/year	28.8 MAP	6,878.3 tons/year
Airport (JWA)	238.83 tons/MAP	*		5.5 MAP	1,313.5 tons/year	3.9 MAP	931.4 tons/year	4.6 MAP	1,098.6 tons/year	5.4 MAP	1,289.68 tons/year
Subtotal Solid Waste Generated	-	-	-	•	3,415.2 tons/year		5,421.4 tons/yr		6,687.2 tons/yr	-	8,167.98 tons/yr

#### Notes:

ND - No Data

NA - Not Applicable

- I Generation factors typically represent a 220 day work year.
- 2 Generation factors typically represent a 365 day work year.
- 3 Number of golfers were arrived at by dividing the average number of golfers anticipated at the golf courses over a three year period, by three.
- 4 Orange County Fire Authority.

Sources: Generation factors were extracted from various studies and EIRs in Los Angeles and Ventura Counties, and the City of Santa Clarita, and the California Integrated Waste Management Board.

- a Stevenson Ranch DEIR (Phase IV), Los Angeles County.
- b Guidelines for Preparation of Environmental Assessments for Solid Waste Impacts (Ventura County Solid Waste Management Department).
- c DEIR for North Hills Development (Santa Clarita).
- d Estimated Solid Waste Generation Rates for Institutions (California Integrated Waste Management Board)

### 11.3.10.3 Level Of Significance After Mitigation

"The potentially significant adverse impacts of the proposed project related to solid waste would be reduced to below a level of significance based on the implementation of Standard Condition of Approval SW1 (Solid Waste Collection Areas) and Mitigation Measure PS-3." (4.10-53.) This conclusion is unchanged at the Phase 2 implementation level.

## 11.3.11 Natural Resources and Energy

### 11.3.11.1 Project Impacts On Natural Resources

#### Water Resources

"Project-related water demand during preliminary implementation phases would be less than at build out in 2020. Consequently, neither the phased implementation nor ultimate build out of the Proposed Project will require or result in the construction of new water treatment facilities or expansion of existing facilities, or place a demand on available water supplies in excess of that capable of being served by existing entitlements and resources." (4.11-12.)

### Agricultural Resources

"Under build-out of the Proposed Project, a substantial portion (approximately 902 acres or 87%) of the existing agricultural operations on the MCAS El Toro site would be terminated, and the majority of these areas would be developed with nonaviation uses including a business park, golf courses, public facilities, and the Marshburn Retarding Basin. .... ¶Most elements of the Proposed Project that will impact/eliminate existing agricultural resources will be implemented during Phase 1." (4.11-12.) "Because this permanent loss of agricultural resource land usage on the MCAS El Toro site involves conversion of substantial acreage of farmlands, including those listed as "Prime" and/or of "Statewide Importance" to non-agricultural uses, it is considered to be a significant adverse impact of the Proposed Project." (4.11-13.) This significant adverse impact would occur during Phase 1 of the Proposed Project.

#### Mineral Resources

"Because there is sufficient supply of these materials in the region and the construction of the Proposed Project would be phased over time, the use of these materials would not constitute a significant adverse impact." (4.11-13.)

### 11.3.11.2 Project Impact on Energy Resources

### Electricity

"Project-related electricity demand during preliminary implementation phases would be less than at build out in 2020. Therefore, neither the phased implementation nor ultimate build out of the Proposed Project would result in significant adverse impacts to electricity resources." (4.11-15.) (Note: see Section 6.0, infra, for a discussion of the current State energy "crisis.")

#### Natural Gas

"[T]he existing gas mains surrounding the MCAS EL Toro site have more than adequate capacity to service the needs of the Proposed Project. ... ¶ Natural gas consumption at MCAS El Toro represents substantially less than 1% of stateside and ASA regional natural gas consumption under both existing and forecasted conditions. Therefore, impacts on energy resources created by natural gas consumption under the Proposed Project are considered to be below a level of significance. Adequate supplies are available and no development is anticipated under the Proposed Project which would be in conflict with what SCG [Southern California Gas] indicates it can support." (4.11-15 - 16.) This conclusion of significance applies to each of the phasing years as well.

#### Fossil Fuels

#### **Motor Fuels - Operation**

"At build-out, the Proposed Project would consume motor vehicle fossil fuels at volumes substantially below 1% of current (existing conditions) and forecasted conditions for the ASA region as a whole. Project-related motor vehicle fuel demand during preliminary implementation phases would be less than at build out in 2020. Consequently, neither the phased implementation nor ultimate build out of the Proposed Project will consume fossil fuel at a level that substantially exceeds existing supplies or otherwise causes regional supply and/or capacity constraints." (4.11-18.)

#### **Motor Fuels - Construction**

Construction activities at the MCAS El Toro site under the Proposed Project will be conducted across four phases. Draft EIR No. 573 Table 4.11-7 depicts construction-related energy consumption for each Phase, as well as each year within each phase. (4.11-18.) As depicted by Table 4.11-7, "diesel fuel and gasoline consumption associated with construction of the Proposed Project is substantially less than 1% of the current (existing conditions) and forecasted regional demand for these fuels. Therefore, construction of the Proposed Project will not result in significant adverse impacts to fossil fuel resources." (4.11-19.) This conclusion of significance applies during each of the phasing years as well.

### Projected Jet Fuel Consumption at JWA and OCX

Draft EIR No. 573 Table 4.11-8 depicts the estimated jet fuel consumption for JWA and OCX under the proposed project for each of the four phases.

#### Projected Jet Fuel Consumption in the Region

"[B]ecause the amount of fuel used under the Proposed Project and each of its phases [including Phase 2] exceeds the one percent threshold of significance, when compared to existing conditions, the Proposed Project causes a significant, unavoidable impact by encouraging the use of large amounts of the total jet aircraft fuel consumed in the ASA region. With the present state of jet aircraft engine technology, there are no feasible mitigation measures to address this impact." (4.11-21.)

#### Projected Aviation Gasoline Consumption at JWA and OCX

The projected level of fuel demand for the Proposed Project "is substantially less than 1% of the total current (existing conditions) and forecasted regional transportation demand for gasoline. In addition, the estimated Orange County aviation gasoline demand in 2020 under the Proposed Project represents less than a 1% increase over existing conditions from a Southern California regional aviation gasoline perspective. Therefore, the Proposed Project will not result in significant adverse impacts to this fossil fuel resource." (4.11-23.) This is also the case with respect to aviation gas consumption during the phasing years as there will be fewer aviation operations during this period than during final Phase 4.

## 11.3.11.3 Mitigation Measures

"The following mitigation measures were identified in Final EIR SA to provide partial reduction of the CRP impact to converting prime agricultural land to urban uses:

- NRE AG-1 The County, acting as the LRA for MCAS El Toro, shall use its best efforts to secure the conveyance of 40 acres of existing prime agricultural land on MCAS El Toro from the Department of Navy for the benefit of the Orange County Sheriff-Coroner. The 40-acre conveyance will be used by the Orange County Sheriff-Coroner for permanent agricultural use in conduction with the County's existing jail agricultural program at the James A. Musick Jail Facility. As a direct result of this mitigation, the conversion of prime agricultural land resulting from implementation of the Community Reuse Plan would be reduced from approximately 818 acres to 778 acres.
- NRE AG-2 As part of the interim reuse phasing strategy, the County, acting as the LRA for MCAS El Toro, will use its best efforts to continue to make available for lease the acreage currently in agricultural use on the MCAS El Toro site. This action will help compensate for the adverse economic impact resulting from base

closure, and provide a revenue stream to maintain the base property until project build out. The County will continue to make agricultural lands available for lease as long as the interim uses are consistent with the uses contemplated in the Community Reuse Plan, and as long as those uses do not compromise the County's ability to implement the Community Reuse Plan in a timely manner. However, interim agricultural use would not be allowed for a period longer than 10 years." (4.11-24.)

## 11.3.11.4 Level of Significance After Mitigation

"The impacts of the proposed project on natural resources, aggregate resources and energy resources will not be significant except as follows: the loss of agricultural land in the County of Orange, the City of Irvine and the City's sphere of influence is a significant adverse impact of the Proposed Project that cannot be mitigated to below a level of significance. The demand on jet fuel in the ASA is an impact that cannot be mitigated to below a level of significance when compared to existing conditions. The encouragement by the Proposed Project of the use of large amounts of the total jet aircraft fuel consumed in the ASA region, when compared to existing conditions, is an impact that cannot be mitigated to below a level of significance." (4.11-28.)

## 11.3.12 Aesthetics, Light and Glare

"The aesthetics, light and glare impacts analysis is provided based upon project build out in year 2020. Visual features of note constructed prior to 2020 are discussed [in Draft EIR No. 573] where applicable." (4.12-7.)

To the extent the Air Traffic Control Tower may be constructed prior to 2020 during Phase 1 of the Proposed ASMP Project, the on-site project improvements will not result in significant adverse impacts related to aesthetics during the phasing years. (See 4.12-12.)

## 11.3.12.1 Impacts Related to Off-Site Road Improvements

To the extent construction of the Eastern Transportation Corridor connector ramps will be a part of the incremental development of the Proposed ASMP Project, the project related off-site improvements will not result in significant adverse impacts to aesthetics during the phasing years. (See 4.12-12.)

### 11.3.13 Cultural Resources

#### 11.3.13.1 Historical Resources

The proposed project will not result in significant adverse impacts to historical resources at build out, or during the interim phasing years. (See 4.13-5.)

### 11.3.13.2 Archaeological Resources

The Proposed Project will not result in significant adverse impacts to archaeological resources on site at MCAS El Toro. (See 4.13-6.) This is the case at project build out, as well as during the interim phasing years.

### 11.3.13.3 Paleontological Resources

"Although no specific sensitive paleontological resources are currently known to be exposed on MCAS El Toro, the geological units present on the project site have been rated as having a moderate paleontological sensitivity potential to be exposed during ground disturbance and earthmoving activities. Therefore, County Standard Condition of Approval No. A7 will be implemented during project earthmoving activities as precautionary mitigation [monitoring of project site grading and preparation work by a County certified paleontologist]. ... Therefore, no significant impacts to paleontological resources are expected to occur under the Proposed Project." (4.13-7.) Because County Standard Condition of Approval No. A7 applies during all phases of construction, the proposed project will not result in significant adverse impacts to paleontological resources during the interim phasing years.

#### 11.3.14 Recreation

The Proposed ASMP Project will not result in significant adverse impacts to on-site recreational facilities during the phasing years. (See 4.14-12.)

The Proposed ASMP Project will not result in significant adverse impacts to regional local parks during the phasing years. (See 4.14-13.)

The Proposed ASMP Project will not cause significant adverse impacts to local and/or regional riding and hiking trails or off-road bikeways during the phasing years. (See 4.14-13-14.)

## 11.3.14.1 Class II Bikeways

"In the MCAS El Toro area, the Proposed Project's trip generation will require widening of Irvine Boulevard to its ultimate six lane master planned width, between Jeffrey Road and approximately midway between Alton Parkway and Sand Canyon Avenue (refer to Draft EIR No. 573 Section 4.3, Table 4.3-20). Irvine Boulevard is primarily four lanes currently in this segment. The widening of the section of Irvine Boulevard adjacent to the MCAS El Toro property (ETC East Leg to West Access Road) is proposed in Phase 1 (by 2005), with the remaining off-site sections being widened by the end of Phase 4. The widening of this roadway will affect the existing on-street bicycle lanes by disrupting their use during the roadway construction period. This disruption may involve a temporary closure and/or rerouting of the bicycle lanes on the north and south sides of the road. Therefore, this would be a significant short-term impact, which will be reduced to below significance with implementation of

Mitigation Measure R-1 (refer to Section 4.14.8.4)." (4.14-14.) The Proposed Project would not result in significant adverse impacts to other Class II Bikeways during the phasing years.

#### 11.3.14.2 Effects of Aircraft Noise on Use of Recreational Facilities

The Proposed Project would not cause a significant noise impact to parks and other recreational facilities that fall within the existing 1998 military 65 dB CNEL contour, including the Laguna Hills Golf Course, the Aliso Viejo Golf Course, and on-site recreational facilities. (See Draft EIR No. 573 Figure 4.14-1.) Similarly, because aircraft operations will be less during the interim phasing years than at project build out, there will be no additional noise impacts at the subject facilities during Phases 1 through 2. (See, 4.14-15.)

In the build out project condition, several existing and future planned off-site trails, Class II bikeways, parks and open space areas fall within the post-project 65dB CNEL contour. These noise impacts are considered significant. (See 4.14-15.) The 65 dB CNEL contour will be smaller proportionately during Phases 1 and 2 and, correspondingly, impacts will be less during these interim phases of the Proposed Project.

On-site recreational facilities will be affected by aircraft noise from the Proposed Project at build out. Facilities including the two proposed golf courses, the regional park, the Equestrian Center, and on-site trails will be within the project 65 dB CNEL contour. As to that portion of the 65 CNEL noise contour that covers the MCAS El Toro site, the contours for Phase 1 and Phase 2 are approximately the same as project build out. Accordingly, the potential impacts are approximately the same (no greater) for project Phases 1 and 2.

## 11.3.14.3 Consistency with Recreational Planning Policies

Project consistency with County and City General Plans is applicable to the project actions occurring prior to Phase 1. Therefore, the recreational planning policies consistency analysis is not subject to analysis by phase. (See 4.4-16-18.)

## 11.3.14.4 Mitigation Measures

## Mitigation Measure R-1

"Prior to the approval of design plans, specifications, and estimates (PS&E) for off-site roadway improvements, the County shall prepare a Construction Action Plan. The Construction Action Plan shall set forth appropriate construction practices necessary to minimize the potential disruption to properties, pedestrians, bicyclists, and motorists. The following items shall be included in the Plan:

 Public Notice: Signs providing advance notice of work to be done on a particular segment shall be posted for a period of two weeks prior to construction. Notification in a local newspaper shall be published two weeks prior to construction. Adjacent property owners and public service providers and utility companies shall also be notified two weeks prior to construction.

- Traffic Routing: Signs shall be provided to route vehicular and bicycle traffic through segments under construction. In addition, signs that suggest possible alternate routes shall be posted. Construction vehicle access to construction sites shall occur at off-peak traffic hours. Construction vehicle access routes shall be directed around residential areas. In addition, traffic control personnel shall be provided as necessary to mitigate traffic congestion and to mitigate the impact to arterial service levels during construction in accordance with local, State, and federal standards.
- Construction Staging: Storage of construction vehicles, equipment, and materials shall not occur in the immediate vicinity of residential areas and retail establishments.
- Hours of Construction: Hours of construction shall conform to established County policy unless otherwise approved by the County.
- Access to properties: Construction activities shall be arranged so that access to properties will be maintained." (4.14-20-21.)

## 11.3.14.5 Level Of Significance After Mitigation

Implementation of Mitigation Measure R-1 will reduce Phase 2 short-term impacts to bikeways on off-site roadways under construction to a level below significant. (See 4.14-21.)

## 11.3.15 Public Safety

In all cases discussed below, the likelihood of impacts is lower in Phases 1 through 3 than at project build out due to the lower number of aircraft operations forecast in the interim phasing years.

#### 11.3.15.1 Accident Likelihood

"The aircraft accident likelihood under the Proposed Project in 2020 for both JWA and OCX is shown on [Draft EIR No. 573] Table 4.15-5. This level of aircraft fatal accident risk is not extraordinary given the projected level of aircraft operations. Similar statistical risks would be expected if the operations were at another urban Southern California location. The fatal off-airport accident likelihood would be lower in the earlier years due to the lower number of aircraft operations forecast in the intermediate phasing years." (4.15-16.)

#### 11.3.15.2 Bird Strike Risk

"Although the departure corridor for Runway 34 does cross a bird movement flight pattern, the altitude separation between departing aircraft and gull flight at MCAS El Toro, coupled with the infrequent bird strikes reported at JWA, does not seem to indicate that a substantial bird strike hazard exists for the Proposed Project." (4.15-18 - 19.) To the extent Runway 34 is utilized during Phases 1 and 2, this conclusion applies equally during the phasing years.

### 11.3.15.3 Adequacy of Runway Lengths at JWA and OCX

"[T]he runway system proposed for OCX under the Proposed Project would be adequate to accommodate domestic and international flights to destinations that satisfy Orange County air travel demands under typical weather conditions existing at the site." (4.15-21.) In Phase 1, the existing 10,000-foot Runway 16R/34L will be reconstructed and readied for commercial use. (3-58.) Phase 2 airfield improvements involve construction of the parallel north-south runways to their ultimate configuration, and development of Runway 7R-25L. (3-63.)

### 11.3.15.4 Safety of Runway Configuration

"The proposed project will be in compliance with all FAA safety standards pertaining to transportation uses at the ends of runways, and therefore, the proposed uses [during all phases of the Proposed Project] will not result in a significant safety impact." (4.15-25.)

# 11.3.15.5 Proximity of JWA and OCX and Potential Airspace Conflicts

"Airspace studies [conducted in connection with the Proposed Project] demonstrate that the use of MCAS El Toro as a commercial airport and the continued commercial use of JWA can be safely and efficiently accommodated within the regional air traffic and airspace structure." (4.15-26.) This conclusion is independent of project phasing, although the relative risk increases in successive project phases as the number of aircraft operations increases.

## 11.3.15.6 Emergency Procedures for Aircraft Operations

"Although the possibility of a commercial airplane losing power in an engine during takeoff is remote, procedures are established at every airport to allow the aircraft to safely complete its takeoff and initial climb-out. At OCX these procedures would require that aircraft follow different departure paths than would be followed under normal conditions because an aircraft that lost power in one engine would have a reduced climb capability." (4.15-27.) This conclusion is independent of project phasing, although the relative risk increases in successive project phases as the number of aircraft operations increases.

#### 11.3.16 Hazardous Wastes and Hazardous Materials Use

Draft EIR No. 573 Section 4.16.6 addresses the hazardous wastes and hazardous materials use impacts associated with the Proposed Project. The impacts analysis is presented by Planning Area, eight separate subsections, each addressing one of the eight MCAS El Toro geographic planning areas. Within each Planning Area discussion, the EIR addresses separately: (i) the impacts attributable to site location, and (ii) the impacts attributable to hazardous materials use. A discussion of the impacts of the Proposed Project relative to the interim phasing years, including Phase 2, follows.

### 11.3.16.1 Impacts Attributable to Site Location

"Hazardous wastes issues currently being addressed by the DON at the MCAS El Toro site include in-service ASTs (aboveground storage tanks)/USTs (underground storage tanks), landfills, oil/water separators, active hazardous material storage areas and hazardous waste accumulation areas, PCB transformers and storage areas, burn pits, silver recovery units, potential presence of radioactive material, and pesticide storage areas... [These wastes] are found throughout the MCAS El Toro site [in locations designated "IRP" (installation restoration program) sites] and most generally tend to be concentrated in proximity to the existing airfield. The location of IRP sites in relation to the Proposed Project is shown on DEIR No. 573 Figure 4.16-2. The DON is responsible for ensuring that these sites undergo remedial action that is appropriate for the intended reuse of the MCAS El Toro site. The DON will also implement institutional controls to ensure that selected remedial actions are not compromised by proposed uses. The remediation measures and institutional controls will allow the proposed uses at the site to occur without exposing the public to any hazardous materials exceeding any established or generally accepted level of risk or exposure." (4.16-24.)

Draft EIR No. 573 presents the impacts of the proposed project attributable to site location at pages 4.16-25 through 4.16-46. In assessing potential impacts on a planning area-by-planning area basis, the analysis necessarily considered these impacts during the phasing years. That is, in assessing impacts attributable to site location within Planning Area 1, the analysis considered that under the Proposed Project, the terminal complex and parking facilities would be located in Planning Area 1. (4.16-25.) Construction of the terminal building area and terminal vehicle parking is scheduled to begin in Phase 1. Therefore, the conclusion of significance with respect to Planning Area 1 factors into consideration the Phase 1 construction of the terminal complex. Similar conclusions may be drawn with respect to each of the Planning Areas.

## 11.3.16.2 Impacts Attributable to Hazardous Materials Use

"The storage, use and handling of hazardous materials, and generation and disposal of hazardous wastes, are regulated by applicable state and federal laws and regulations, and local ordinances. A list of these applicable laws and regulations is provided in Appendix I to Draft EIR No. 573. Compliance with these regulations and [the proposed mitigation measures] will reduce potential impacts related to hazardous materials use to below a level of significance." (4.16-27.) This

significance conclusion applies equally during each of the phasing years since the requirements associated with project compliance apply during the interim phasing years, including Phase 2, as well as at project build out.

### 11.3.16.3 Level of Significance After Mitigation

"As described earlier, if the proposed mitigation is implemented, the Proposed Project is not anticipated to result in significant adverse impacts related to hazardous materials and waste, based on the existing remediation programs underway at the MCAS El Toro site, the imposition of institutional control by DON, and the lack of significant hazardous waste issues at JWA. Future uses of hazardous materials and generation of hazardous wastes by the Proposed Project may have the potential to result in significant adverse impacts to human health and the environment. Compliance with applicable federal, state, and local regulations and requirements, and proposed mitigation measures pertaining to the use of these substances will reduce potential impacts to below a level of significance. Therefore, the Proposed Project will not result in significant adverse impacts related to existing hazardous waste contamination and proposed hazardous material use." (4.16-56.)

#### 11.3.17 Socioeconomics

"Economic activity to be generated by aviation and nonaviation activities at OCX and JWA, as well as by tourism related jobs generated by visitors to Orange County using OCX and JWA, is anticipated to support a total of 146,100 jobs countywide by 2020 (Technical Report No. 16, Economic Benefits Study, November, 1999). Proportionate economic activity would be generated under the 2005, 2010, and 2015 development scenarios." (4.17-14.) See Draft EIR No. 573 Table 4.17-4, Projected On-Site Employment by phase.

# 11.3.17.1 Impacts Relating to Inducing Substantial Growth or Concentration of Employment

"The Proposed Project would result in a significant adverse impact related to inducing substantial growth or concentration of population (in this case nonresident employee population). The Proposed Project anticipates generating significant new employment opportunities in all phases, which is a significant adverse impact in all phases of the Proposed Project." (4.17-14.)

# 11.3.17.2 Impacts Relating to Inducing Substantial Growth or Concentration of Resident Population and Housing

"Since no housing units will be retained or developed in any phase of the Proposed Project, and significant new employment opportunities will be generated in all phases, this is a significant adverse impact of the Proposed Project in all phases." (4.17-15.)

# 11.3.17.3 Impacts Related to Displacement of a Large Number of People

The Proposed Project will not result in significant adverse impacts related to the displacement of a large number of people. This conclusion is unchanged during the interim phasing years. (See 4.17-17.)

# 11.3.17.4 Impacts Related to the Disruption or Division of the Physical Arrangement of an Established Community

"The Proposed Project would not result in a significant adverse impact related to the disruption or division of the physical arrangement of an established community." (4.17-17.) This conclusion is unchanged during the interim phasing years.

# 11.3.17.5 Impacts Related to Consistency with Adopted Regional Forecasts

"[Because] the projected employment and population is inconsistent with the level identified in the adopted regional growth forecasts in all phases of the Proposed Project, this is a significant adverse impact of the Proposed Project in all phases." (4.17-18.)

### 11.3.17.6 Impacts Related to the Jobs/Housing Ratio

"Because the direct impacts of the Proposed Project will slightly reduce the projected jobs/housing ratio in a jobs-rich area, the Proposed Project will not result in significant adverse impacts related to the projected jobs/housing ratio (see table 4.17-4 for project changes in ratios for 2005, 2010, and 2015)." (4.17-19.) To the extent the projected on site employment numbers are proportionately lower during the interim phasing years than at project build out, the jobs/housing ratio under the Proposed Project during the interim phasing years, including Phase 2, is proportionately lower than forecasted jobs/housing ratios. Accordingly, at Phase 2, the project will not result in significant adverse impacts related to the projected jobs/housing ratio.

# 11.3.17.7 Impacts Related to Low and Moderate Income Housing Needs

"The Proposed Project will generate significant new employment opportunities in all phases of the project without directly providing any new housing, which is a significant adverse impact of the Proposed Project in all phases." (4.17-20.)

### 11.3.17.8 Impacts Related to Housing Supply

"With respect to the removal of substantial numbers of existing housing, "in that these units will be removed in the first phase of the Proposed Project, this is a significant adverse impact in all phases of the Proposed Project." (4.17-20.)

### 11.3.17.9 Mitigation Measures

"The level of employment and population generated on site under the Proposed Project would be inconsistent with the adopted regional growth forecasts. Under the SCAG threshold of significance related to consistency with adopted regional growth forecasts, this would be a significant adverse impact of the Proposed Project. As mitigation:

SE-1 If the Proposed Project is adopted, the County of Orange will submit updated employment, population, and housing growth forecasts for the project site to SCAG for their next scheduled update of regional growth forecasts that reflect the level of activity anticipated under the Proposed Project." (4.17-21.)

## 11.3.18 Level of Significance After Mitigation

"Impacts of the Proposed Project regarding inconsistency with the adopted regional forecasts will be reduced to below a level of significance with implementation of Mitigation Measure SE-1.

Impacts of the Proposed Project related to inducing substantial growth or concentration of non-resident employee population cannot be mitigated below a level of significance.

Impacts of the Proposed Project related to increased resident population growth and the related demand for all types and prices of housing in the surrounding area, while not directly providing any new housing to meet this increased demand, cannot be mitigated below a level of significance.

Impacts of the Proposed Project related to reducing the supply of available housing in the County cannot be mitigated below a level of significance." (4.17-21.)

## 11.3.19 Risk Of Upset

#### 11.3.19.1 Accident Assessment - MCAS El Toro

## Transport Accidents - Tank Trucks

Draft EIR No. 573 Table 4.18-7 provides a breakdown by project phase of the estimated daily probabilities of accidental releases associated with tank truck delivery of jet fuel to the MCAS El Toro site.

Table 4.18-7
Proposed Project Jet Fuel Tank Truck Daily Accident
Probabilities – MCAS El Toro Site

	2005	2010	2015	2020
Truck Trips/Day	40	133	178	244
10% Cargo Loss	4.5 x 10 <sup>-4</sup>	1.5 x 10 <sup>-3</sup>	2.0 x 10 <sup>-3</sup>	2.8 x 10-3
	(High/Likely)	(High/Likely)	(High/Likely)	(High/Common)
30% Cargo Loss	1.5 x 10-4	5.0 x 10-4	6.7 x 10 <sup>-4</sup>	9.2 x 10-4
	(Medium/Reasonably Likely)	(High/Likely)	(High/Likely)	(High/Likely)
100% Cargo Loss	1.5 x 10 <sup>-4</sup>	5.0 x 10 <sup>-4</sup>	6.7 x 10 <sup>-4</sup>	9.2 x 10 <sup>-4</sup>
	(Medium/Reasonably	(High/Likely)	(High/Likely)	(High/Likely)
	Likely)			

Source: ASMP; FEMA, 1990

"For perspective, these estimated probabilities indicate that, for the level of jet fuel tank truck deliveries to OCX occurring in 2020, approximately one accident per year, resulting in a 10 percent cargo loss, would be expected. Accidents resulting in 30 percent and 100 percent cargo losses would be expected to occur approximately once every three years.

"On this basis, the representative probability for project related tank truck transport accidents to result in a release of jet fuel while enroute to the MCAS El Toro site from a fuel supplier in El Segundo is deemed to range from High/Common accidents, for a ten percent cargo loss in 2020, to Medium/Reasonably Likely accidents, for 30 percent and 100 percent cargo releases during Phase 1 (2005) of the Proposed Project. As shown in the risk matrix (Figure 4.18-1), any accident with a probability of occurrence in the High/Common or High/Likely accident range would be considered to trigger a mandatory requirement for comprehensive planning and preparedness, regardless of the severity of the accident related consequences." (4.18-20.)

"Given the relatively high probability of occurrence estimated for tank truck accidents associated with jet fuel delivery to the MCAS El Toro site during all project phases, coupled with the size of potential hazard and vulnerable zones calculated by ARCHIE for these

hypothetical accidents, the Proposed Project could result in reasonably foreseeable upset and accident conditions, involving the release of jet fuel into the environment, thus creating a hazard to the public or the environment through the routine transport and use of this fuel. As a result, highway truck transport of jet fuel to the MCAS El Toro site under the Proposed Project is deemed to result in a significant adverse impact to public health and safety under risk of upset conditions. This conclusion is independent of project phasing, although the relative risk increases in successive project phases as the number of required tank truck trips concurrently increases." (4.18-21.)

## Bulk Fuel Storage Facility Accidents

"Table 4.18-8 provides a breakdown by project phase of the total estimated daily probabilities of accidental releases associated with bulk jet fuel storage at the MCAS El Toro site." (4.18-22.) "[T]he Proposed Project would not result in reasonably foreseeable upset and accident conditions, involving the release of jet fuel into the environment, thus creating a potential hazard to the public or the environment through the routine use and storage of jet fuel. As a result, the use and storage of jet fuel at the MCAS El Toro site under the Proposed Project is deemed not to result in a significant adverse impact to public health and safety under risk of upset conditions. This conclusion is unaffected by project phasing." (4.18-23 - 24.)

## Aircraft Related Bulk Fuel Storage Facility Accidents

"Table 4.18-9 provides a breakdown by type of operation of the total estimated daily probabilities of bulk fuel storage facility accidents involving aircraft arrivals on Runways 7R and 7L at the MCAS El Toro site under the Proposed Project." (4.18-25.) "[T]he Proposed Project would be unlikely to result in reasonably foreseeable upset and accident conditions, involving the release of jet fuel into the environment, thus creating a potential hazard to the public or the environment through the routine use and storage of jet fuel. As a result, the use and storage of jet fuel at the MCAS El Toro site under the Proposed Project is deemed not to result in a significant adverse impact to public health and safety under risk of upset conditions. This conclusion is unaffected by project phasing." (4.18-27.)

## 11.3.19.2 Mitigation Measures

"On the basis of the analysis provided, the only risk of upset significant adverse impact associated with the Proposed Project requiring mitigation is the highway transport of jet fuel to the MCAS El Toro site via tank truck. To mitigate this impact, the County proposes the following measure:

- RU-1 Prior to commencement of aviation flight operations, the County shall make every reasonable effort to lease or otherwise obtain appropriate agreement and/or approval for the use of the Norwalk Pipeline and Santa Fe Pacific Pipeline Line Section 126 (LS-126) for the purpose of conveying all jet fuel to the MCAS El Toro site. The objective of this measure is to obviate the need for and eliminate the adverse impacts associated with highway truck transport of jet fuel.
- RU-2 At any such time that the County becomes the owner of the Norwalk pipeline, the County or its agents will inspect the pipeline on a regular basis to conform to the standard practice in the industry, will take corrective action to remedy any leaks, ruptures, or other hazards caused by the pipeline, and will pay any damages as required by law that are deemed by the appropriate authority to be the liability of the County.

These pipelines are existing, in-place, underground fuel conveyance facilities. Operation of the existing Norwalk petroleum transport pipeline would be subject to all pertinent state and federal regulations, including Title 49, Part 195 of the Code of Federal Regulations (14 CFR 195), which specifies operational and maintenance requirements.

Utilizing the existing Norwalk pipeline would require construction of a new, approximately 2.7 mile long pipeline from the Norwalk Pipeline at Irvine Boulevard south to the bulk fuel storage area. Accessing the Santa Fe Pipeline would require construction of a new, approximately 250 foot long supply pipeline from LS-126 to the fuel storage facility. These new pipeline connections would be constructed and completed during Phase 1." (4.18-32.)

"The potential for risk of upset conditions associated with underground petroleum fuel pipelines (i.e., fire, explosion, and airborne LOC exposures) is substantially less than for tank truck highway transport. According to the American Petroleum Institute (API), the most likely accident involving an underground pipeline would be a localized spill triggered by "outside force damage" (e.g., during construction related excavations). While such an accident could release jet fuel to the environment, the resultant spill generally would be a localized environmental contamination and cleanup issue, as opposed to a potential acute public health and safety hazard.

Implementation of the pipeline jet fuel supply option also would necessitate the construction of two (2) additional bulk fuel storage tanks at the MCAS El Toro site. Table 4.18-12 provides a breakdown by project phasing and total estimated daily probabilities of accidental releases associated with bulk jet fuel storage under the pipeline supply option.

Table 4.18-12

Bulk Jet Fuel Storage Tank Daily Accident Probabilities – OCX with Pipeline Fuel Supply Option

	2005	2010	2015	2020
Total Tanks On-Site	2	5	6	8
10% Product Loss	5.0 x 10 <sup>-7</sup>	1.2 x 10 <sup>-6</sup>	1.0 x 1.0 <sup>-6</sup>	2.0 x 10 <sup>-6</sup>
100% Product Loss	5.5 x 10 <sup>-8</sup>	1.4 x 10 <sup>-7</sup>	1.6 x 10 <sup>-7</sup>	$2.2 \times 10^{-7}$

Source: ASMP; FEMA, 1990

"As with the Proposed Project, the total probability for bulk jet fuel storage tank accidents to result in a release of jet fuel at the MCAS El Toro site under the pipeline option remains in the low/very unlikely range, regardless of project phase and independent of release size." (4.18-33.)

## 11.3.19.3 Level Of Significance After Mitigation

"Given a Low to Medium likelihood of occurrence and lack of adverse public health and safety accident consequences under risk of upset conditions, utilizing existing and new underground jet fuel conveyance pipelines will effectively mitigate the potential adverse effects associated with highway truck transport of jet fuel, while resulting in no additional or residual impacts." (4.18-33.) The implementation of Mitigation Measure RU-1 is within the control of entities other than the County of Orange. If the County is unable to implement this mitigation measure to the extent necessary to reduce the potential risks of upset associated with the Proposed Project to below a level of significance, this impact will remain significant and unavoidable.

,	

# **11.0** 12.0 PERSONS AND ORGANIZATIONS CONSULTED

In accordance with CEQA Guidelines Section 15129, the following information is provided on organizations and persons consulted in preparing the Draft EIR. Chapter 13 12 contains a list of the persons, firms, and agencies contributing to the preparation of this EIR.

The list of persons and organizations provided below is in addition to the consultation that was accomplished through the Draft EIR Notice of Preparation (NOP) process discussed in Chapter 2 of this EIR. Appendix A to this document contains a copy of the NOP and a list of individuals, agencies, and organizations to whom the NOP was sent, and Appendix B contains the comments on the NOP.

Advanced Transportation Systems, Inc

Asphalt Emulsions Manufactures Association (AEMA), James, Alan

BRAC, Ed Gilhooley, Lt. Colonel

BRAC, Tony Ray, GM13

California Air Resources Board (CARB), Bob Maxwell

CARB, Emission Inventory Group, Linda Murchinson

CARB, Archana Agrawal

CARB, Cheryl Taylor

CARB, Dennis Wade

CARB, Doug Thompson

CARB, Jacline Lourenco

CARB, Mena Shah

CARB, Nathanial Kong

Caltrans, Park-and-Ride Division

Caltrans, Air Quality Section Chief, Mike Brady

Caltrans, Sandra Freedman

City of Costa Mesa Planning Department, Kim Brandt

City of Costa Mesa Planning Department, R. Michael Robinson

City of Costa Mesa Planning Department, Willa Bowens

City of Irvine, City Managers Office, Daniel Jung

City of Irvine, Community Development Records Division, Sirpa Lee

City of Irvine, Planning Department, Diane Blaisure

City of Irvine, Planning Department, Mike Philbrick

City of Laguna Niguel, John Morgan

City of Laguna Woods, Edwin Hsu

City of Lake Forest, Development Services Department, Gail Ackerman

City of Lake Forest, Planning Department, Bob Golden

City of Mission Viejo, Planning Department, Elaine Lister

City of Mission Viejo, Clint Sherrod

City of Newport Beach, Building Department, Leslie Duarde

City of Newport Beach, Planning and Zoning Department, Robert Kahn, John Kain & Associates

City of Newport Beach, Building Department, Faisa Jurdi

City of Santa Ana, Planning Division, Ray Brantley

City of Tustin Redevelopment Agency, Dana Ogdon

City of Tustin Redevelopment Agency, Yvette Cisneros

County Librarian Office, Vince Geraghty

County of Orange, Bill Grieman

County of Orange, Bill Melton

County of Orange, Chuck Shoemaker, Chief of Site Planning,

County of Orange, Department of Real Property/Right-of-Way, John Pavlick, Chief Right-of-Way Processing

County of Orange, George Britton, Manager, Environmental and Project Planning Services Division

County of Orange, Public Facilities and Resources, Kevin Onuma

County of Orange, Public Facilities and Resources, Lance Natsuhara

County of Orange, Richard Bailey

County of Orange, Romi Archer

County of Orange, Timothy Neely, Manager, Nature Reserve of Orange County

DOE, James Sledge

DOE, Melanie Thornton

Environmental Protection Agency (EPA), OMS Division, Bryan Manning

EPA Office of Mobile Sources (OMS), Richard Wilcox

EPA OMS Division, David Brezinski

EPA OMS Division, Greg Jansen

EPA, Dennis Atkinson

EPA, Tom Braverman

FAA, Aircraft Emission Rates and Factors Dept., Julie Draper

FAA, David L. Bennett, Director of Safety Operations

FAA, John G. Bent, Manager, Aviation Data Systems Branch

**GEOSYNTEC** 

Holmes & Narver/McClier Corporation

Irvine Unified School District, Corinne Loskot

Irvine Unified School District, Shirley Cairns

John Wayne Airport (JWA)

Joyner & Furnal

JWA, Police Services, Lieutenant Orville King

JWA, Police Services, Orville King

JWA, Bonnie Streeter, Access & Noise

JWA, Jeff Roundtree, Airport Operations

K & M Consultants

LAACO, John Hathaway

LSA Associates, Deborah Pracilio

Mammoth Equities, Tucker Lewis

Newport Mesa Unified School District (NMUSD)

NTSB, Carol Floyd

Oak Ridge Private School, Patricia Burry, Proprietor

Orange County Fire Authority, Nancy Foreman, Advance Planning

Orange County Transportation Authority, Planning & Programming, Kia Mortazavi, Manager P&D Aviation

P&D Consultants, Inc.

Rainbow Disposal, Ron Shenkman, Senior Vice President

Saddleback Valley Unified School District (SVUSD)

SCAOMD, Bruce Selik

SCAQMD, Henry Hogo

SCAQMD, Joseph Cassmassi

SCAQMD, Kevin Durkee

SCAQMD, Steve Smith

SCAQMD, Thomas Chico

South Coast Air Quality Management District (SAQMD), Zorik Pirveysian

SWDIVNAVFACENGCOM, Danielle Huey

SWDIVNAVFACENGCOM, Lynn Hornecker

The Planning Center

Transportation Corridor Agency (TCA), Michelle Hart

# **12.0-13.0 LIST OF PREPARERS AND CONTRIBUTORS**

## 42.4-13.1 COUNTY OF ORANGE

Robert Aldrich

George Britton

Christopher Caliendo

Chris Crompton

Ben DeMayo

Jack Golden

Joan Golding

Michael Lapin

Ruby Maldonado

Polin Modanlou

Mark Morgan

Alan Murphy

Lance Natsuhara

Timothy Neely

Kevin Onuma

**Bob Peterson** 

Kari Rigoni

Gary Simon

Bryan Speegle

Courtney Wiercioch

Roger Yee

## **12.2** 13.2 CONSULTANTS

Advanced Transportation Systems
Andrew McKenzie

Algier & Associates Tim Algier

**Austin-Foust Associates** 

Terry Austin

Kendall Elmer

ASM Consulting

Darryl W. Stout

CABACO, Inc.

Alan Murphy

Tim Wagner

Cliff Wallace

#### CH2M Hill

Mark Bennett

Richard Burke

John Castleberry

John Dallapiazza

Lisa Diamanti

David Duran

Vince Hourigan

Doug Huxley

Ravindra Kaggal

John Knott

Doug Landwehr

Dave Lindberg

John Lowe

Natalie Young Pong

Romana Rodriguez

Suzanne Scherrer

Michael Slavick

Pamela Vanderbilt

#### **FORMA**

John Paez

John Sherwood

#### Holmes & Narver

Vic Martinez

Andy Muth

Bruce Toro

#### **HNTB** Corporation

Don Curry

Roger Deitos

Andres Garcia

George Hale

Richard Hart

Andy Herman

Keith Thompson

#### JHTM Associates, Inc.

Lawrence Serafini

## Keyser Marston Associates, Inc.

Gerald Trimble

Debbie Kern

### LSA Associates, Inc.

Michael Amling

Tony Chung

Mona McGuire DeLeon

Michele Douglas

Gary Dow

Frank Haselton

Zac Henderson

Keith Lay

Carollyn Lobell

Richard Nelson

Paul Nguyen

Jill Wilson O'Connor

Peter Pang

Avis Pope-Carter

Andrea Zullo

#### Lea & Elliot

Tim Chan

Harley Moore

Diane Woodend

#### Mestre Greve Associates

Marty Beale

Fred Greve

Vince Mestre

Ron Reeves

#### **P&D** Consultants

Ron Ahlfeldt

Steve Allison

Michael Benner

Peter Bonello

Tin Cheung

Tony DeJulio

Christine Huard-Spencer

Henry Huynh

Tim Landis

Ben Lao

Jose Mortero

Robert Olsen

Robert Price

Jeff Post

Suhair Qoborsi

Ann Reynolds

Ron Richardson Doug Sachman Ron Siecke Warren Sprague

Parsons Engineering Science Dennis Kasper

Parsons Transportation Group Tijana Hamilton Larry Wesemann

Psomas & Associates
Daniel McCroskey
Ryan McLean
Tom Wilkes

Robert Charles Lesser & Co. Robert Gardner Malee Tobias

Synergy

Mary Vigilante

The Planning Center

Al Bell Bryan Dove Brian James Donna Kelly

Craig Ramella Dick Ramella

Steve Ross

**TransSolutions** 

Belinda Hargrove Melinda Stouppe

Touchdown Engineering
David Bush
Nick Leetzee

Barbara Urban

In addition, this report has been reviewed by attorneys representing the County of Orange for legal adequacy. Communications on the Draft EIR involving attorneys representing the County are privileged under applicable attorney-client privileges established under State and federal law.

Gatzke, Dillon & Ballance
Lori D. Ballance
Michael Scott Gatzke
Michael S. Haberkorn

Mark Mispagel, Special Counsel

# **13.0** 14.0 REFERENCES

- Abrams, L., Illustrated Flora of the Pacific States, Vol. IV, Stanford University Press, Stanford, California, 1960.
- Abrams, L., Illustrated Flora of the Pacific States, Vols. I, II, and III, Stanford University Press, Stanford, California, 1923.
- Airport Land Use Commission for Orange County, Airport Environs Land Use Plan, November 16, 1995.
- Allen, Rebecca, Archaeological Survey Report, Marine Corps Air Station, El Toro. MS on file, Department of the Navy, Southwest Division. Naval Facilities Engineering Command. Planning and Real Estate Department, 1220 Pacific Highway, San Diego, California, 92132-5180, 1997.
- American Conference of Governmental Industrial Hygienists, Threshold Limit Values and Biological Exposure Indices. 1987.
- American Ornithologists' Union, Check-List of North American Birds, 6<sup>th</sup> edition, American Ornithologists' Union, Washington, D.C., 1983.
- American Ornithologists' Union, Supplement to the American Ornithologists' Union Check List of North American Birds, American Ornithologists' Union, Washington, D.C., 1995.
- American Ornithologists' Union, Thirty-seventh Supplement to the American Ornithologists' Union Check-List of North American Birds, Auk 106:532-538, 1989.
- American Petroleum Institute, Oil Spill Prevention and Response. API Website www.api.org. 1999.
- Auer, A.H. "Correlation of Land Use and Cover with Meteorological Anomalies." Journal of Applied Meteorology. 1978.
- Aviation Accident Statistics, Table 5, U.S. Carrier Operating Under, 1999.
- Awbry, F.T., and D.H. Hunsacker II, Unpublished abstract: Effects of fixed-wing military aircraft noise on California Gnatcatcher reproduction. Prepared for Acoustical Society of America 134th Meeting; December, 1997.
- Bechtel National, Inc., Draft CERCLA Groundwater Monitoring Plan, MCAS El Toro, July, 1988.
- Bond, J., Effects of Noise on Physiology and Behavior of Farm-Raised Animals. Physiological Effect of Noise. B.L. Welch and A.S. Welch, eds. Plenum Press. New York, NY, 1970.
- Brattstrom, B.H., Bibliography on the Effect of Noise on Non-Human Vertebrates. U.S. Bureau of Land Management, 1982.

- British + Horne JA, Pankhurst FL, Reyner LA, Hume K, Diamond ID, "A Field Study Of Sleep Disturbance: Effects Of Aircraft Noise And Other Factors On 5,742 Nights Of Actimetrically Monitored Sleep In A Large Subject Sample. Sleep 1994 Mar;17(2): 146-59, 1992
- Burger, J., Behavioral Responses of Herring Gulls *Larus argentatus* to Aircraft Noise, Environmental Pollution (Series A), 24:177, 1981.
- Busnel, R.G., Introduction, Effects of Noise on Wildlife, J.L. Fletcher and R.G. Busnell, eds. Academic Press. New York, NY, 1978.
- Busnel, R.G., Introduction, Effects of Noise on Wildlife, J.L. Fletcher and R.G. Bushnel, eds., p. 7-22, Academic Press, New York, New York, 1978.
- CABACO, Inc. Base Transition Plan, November, 1999.
- California Air Resource Board, 1996b. Methodology for Estimating Emissions from On-Road Motor Vehicles Volume II: EMFAC7G, CARB, Technical Support Division, Sacramento, CA.
- California Air Resource Board, Methodology for Estimating Emissions from On-Road Motor Vehicles Volume II: EMFAC7G.
- California Air Resources Board, Fact Sheet 39, 1998.
- California Air Resources Board, State Implementation Plan, 1994 and 1997.
- California Air Resources Board, Version 7G, Urban Emission Model (URBEMIS 7G).
- California Chamber of Commerce, California Environmental Compliance Manual, 1996.
- California Department of Finance, Orange MSA Annual Average Labor Force and Industry Employment, http://www.dof.ca.gov, March, 1998.
- California Department of Fish and Game, At the Crossroads: A Report on the Status of California's Endangered and Rare Fish and Wildlife. State of California Resources Agency, Sacramento, 1980.
- California Department of Housing and Community Development, California Neighborhoods, Accessed on-line http://housing.hcd.ca.gov/hpd/hrc., 1998.
- California Department of Transportation, Division of Aeronautics, Airport Land Use Planning Handbook, December, 1993.
- California Department of Transportation, Personal communication between Sandra Freedman, Public Information Officer, and Mark Harvey, P&D Consultants, July 21, 1999.
- California Department of Transportation, Transportation Project Level Carbon Monoxide Hot Spot Analysis, December, 1997.

- California Department of Transportation, UCD-ITS-RR-96-1; Transportation Project-Level Carbon Monoxide Protocol, May, 1996.
- California Employment Development Department, Orange County Employment by Industry, http://www.calmis.cahwnet.gov., 1999.
- California Motor Vehicle Stock, Travel and Fuel Forecast. Transportation System Information Program, November, 1998.
- California Native Plant Society, Sacramento. Electronic Inventory of Rare and Endangered Vascular Plants of California, Fifth Edition, 1994.
- California Office of Planning and Research, Planning, Zoning, & Development Laws, March, 1995.
- California Regional Control Board, Santa Ana Region, Order No. 94-7, NPDES No. CA 8000336, General Waste Discharge Requirements for Concentrated Animal Feeding Operations, Including Dairies, Within the Santa Ana Region, 1994.
- California Regional Water Quality Control Board, Santa Ana Region, Order No. 94-7, NPDES No. CA 8000336, General Waste Discharge Requirements for Concentrated Animal Feeding Operations, Including Dairies, Within the Santa Ana Region, 1994.
- California Regional Water Quality Control Board, Santa Ana Region, Order No. 98-42, NPDES No. CAS618006, Waste Discharge Requirements for United States Marine Corps Air Station, El Toro Storm Water Runoff, El Toro, Orange County, 1997.
- California Regional Water Quality Control Board, Santa Ana Region, Order No. 90-67, NPDES No. CA800016, Waste Discharge Requirements for John Wayne Airport Storm Water Runoff, Costa Mesa, Orange County, 1990.
- California Regional Water Quality Control Board, Santa Ana Region, Resolution revising the Amendment to the Water Quality Control Plan for the Santa Ana River Basin Incorporating a Nutrient Total Maximum Daily Loads (TMDL) for the Newport Bay/San Diego Creek Watershed (Resolution No. 98-9), 1998.
- California Regional Water Quality Control Board, Santa Ana Region, Order No. 98-42, NPDES No. CAS618006, Waste Discharge Requirements for United States Marine Corps Air Station, El Toro Storm Water Runoff, El Toro, Orange County, 1997.
- California Regional Water Quality Control Board, Santa Ana Region, Order No. 90-67, NPDES No. CA800016, Waste Discharge Requirements for John Wayne Airport Storm Water Runoff, Costa Mesa, Orange County, 1990.
- California Regional Water Quality Control Board, Santa Ana Region, Resolution No. 98-100, Resolution revising the Amendment to the Water Quality Control Plan for the Santa Ana River basin Incorporating a Nutrient TMDL for the Newport Bay/San Diego Creek Watershed (Resolution No. 98-9), 1998.

- California Regional Water Quality Control Board, Santa Ana Region, Order No. 93-16, 1993.
- California State University, Fullerton, Center for Demographic Research, Orange County Projections, 1996 Modified by Community Analysis Area.
- California State University, Fullerton, Center for Demographic Research, Orange County Progress Report, May, 1997.
- Casidy, R.B. and R.P. Lehman, Response of Farm Animals to Sonic Boom. Interim Report, Section H, National Sonic Boom Evaluation Office. Stanford Research Institute. Arlington, VA., 1967.
- Center for Demographic Research, California State University Fullerton, Orange County Projections, 1996 Modified, July, 1997.
- Chapman University, The, Economic and Business Review, Vol. 11 through Vol. 18 No. 1, December, 1999.
- Chevron Products Company, Chevron, Material Safety Data Sheet Chevron Jet Fuels A, A-1, A-50, November 18, 1997.
- City of Costa Mesa, Letter to County of Orange, September 30, 1999.
- City of Costa Mesa, Planning Department, Personal communication between Kim Brandt, Planner, and Mark Harvey of P&D Consultants, July 14, 1999.
- City of Costa Mesa, Planning Department, Personal communication between Willa Bowens and Shambhavi Rao of P&D Consultants, July 21, 1999.
- City of Costa Mesa, South Coast Metro Center Expansion project, SCH#88122116 Mitigated Negative Declaration, February 19, 1999.
- City of Irvine, About Irvine: El Toro Reuse Planning, www.ci.irvine.ca.us/about\_irvine, 1998.
- City of Irvine, About the Irvine Unified School District, July 30, 1999. www.ci.irvine.ca.us/about irvine, July 30, 1999.
- City of Irvine, Community Development Department, Development Services Division. Environmental Impact Report, City of Irvine General Plan, Amendment and Zone Change, June 9, 1983.
- City of Irvine, Community Development Department, Draft Environmental Impact Report, General Plan Amendment 37594-GA, Zone Change 37595-ZC and Annexation No. 17 for MCAS El Toro and James A. Musick Branch Jail, prepared by David Evans and Associates, Inc., March 31, 1999.
- City of Irvine, Community Development Department, Final Environmental Impact Report Planning Area 12, June 28, 1994.

- City of Irvine, Community Development Department, Records Division, Personal communication between Sirpa Lee, Records Coordinator, and Susan Roberts of P&D Consultants, July 21, 1999.
- City of Irvine, Environmental Impact Report Village 38 (14A), General Plan Amendment 87-GA-0039, EIR 87-ER-0076, October, 1990.
- City of Irvine, Environmental Impact Report, Notice of Preparation for Revised City of Irvine Planning Areas #51, #35, and #30 Annexation, General Plan Amendment, Pre-zoning, Zone Change (File Nos: 39399-GA, 39400-ZC, August 13, 1999).
- City of Irvine, General Plan Amendment, Zone Change and Annexation Draft EIR, June 14, 1999.
- City of Irvine, General Plan Parks and Recreation Element, 1995.
- City of Irvine, General Plan, April, 1995.
- City of Irvine, Planning Department, Findings and Facts in Support of Findings for Significant Environmental Effects of Planning Area 38 (Planning Area 38 Final Environmental Impact Report), Resolution No. 90-152, 1990.
- City of Irvine, Planning Department, Personal communication between Mike Philbrick, Planner, and Susan Roberts of P&D Consultants, July 4, 1999.
- City of Irvine, Planning Department, Personal communication between Diane Blaisure, and Mark Harvey of P&D Consultants, July 14, 1999.
- City of Irvine, Zoning Code Supplements No. 1 & 2, 1996.
- City of Irvine, Zoning Ordinance, 1993.
- City of Laguna Hills, General Plan, June, 1994.
- City of Laguna Niguel, General Plan, August, 1992.
- City of Laguna Niguel, Personal communication between John Morgan and Shambhavi Rao of P&D Consultants, July 28, 1999.
- City of Laguna Woods, Letter to the County of Orange, September 29, 1999.
- City of Laguna Woods, Rossmoor Business Properties Final Subsequent EIR No. 549 Environmental Impact and Mitigation Summary.
- City of Lake Forest, General Plan Recreation and Resources Element, 1994.
- City of Lake Forest, General Plan, June, 1994.
- City of Lake Forest, Letter to the County of Orange, September 27, 1999.

- City of Lake Forest, Planning Department, Personal communication between Bob Golden, Project Manager, and Shambhavi Rao of P&D Consultants, July 20, 1999.
- City of Lake Forest, Zoning Code, April, 1997.
- City of Los Angeles, LAX Master Plan, May 8, 1966.
- City of Mission Viejo Planning Commission, Resolution 91-193, Approval of Revised Vesting Tentative Tract Map 13356, Approved, adopted and signed by Susan Wilson, Chairman; Clinton Sherrod, Secretary, October 7, 1991.
- City of Mission Viejo, Approval of Vesting Tentative Tract Map 15677 by Shea Homes Ltd Partnership. Resolution 98-577. Passed, approved, adopted and signed by Norman P. Murray, Chairman; Clinton Sherrod, Secretary, July 27, 1998.
- City of Mission Viejo, General Plan, 1990, 1992, and 1998.
- City of Mission Viejo, Letter to the County of Orange, October 12, 1999.
- City of Mission Viejo, Planned Community Development Plan and Supplemental Text, May, 1991.
- City of Mission Viejo, Planning Department, Personal communication between Elaine Lister, Associate Planner, and Susan Roberts of P&D Consultants, July 27, 1999.
- City of Newport Beach, Building Department, Personal communication between Leslie Duarde, Permit Technician, and Shambhavi Rao of P&D Consultants, July 28, 1999.
- City of Newport Beach, Letter to the County of Orange, October 27, 1999.
- City of Newport Beach, Mitigated Negative Declaration; Mitigation Monitoring Program Summary, Holtze Extended Stay Hotel/Office Project, General Plan Amendment 98-10, Amendment No. 880, Use Permit 3640.
- City of Newport Beach, Planning and Zoning Department, Personal communication between Robert Kain, Planning Technician, and Mark Harvey of P&D Consultants, July 28, 1999.
- City of Orange, East Orange General Plan (December, 1989).
- City of Santa Ana, Planning Division, Personal communication between Ray Brantley, Environmental Coordinator, and Shambhavi Rao of P&D Consultants, July 16, 1999.
- City of Santa Ana, Planning Division, Personal communication between Ray Brantley, Environmental Coordinator, and Shambhavi Rao of P&D Consultants, July 20, 1999.
- City of Tustin, Draft Environmental Impact Statement (EIS)/Environmental Impact Report (EIR) for the Disposal and Reuse of Marine Corp Air Station (MCAS) Tustin, June 24, 1999.

- City of Tustin, Redevelopment Agency, Personal communication between Yvette Cisneros, Administrative Coordinator, and Susan Roberts of P&D Consultants, July 21, 1999.
- City of Tustin, Redevelopment Agency, Personal communication between Dana Ogden, Sr. Project Manager, and Susan Roberts of P&D Consultants, July 21, 1999.
- Clark, Alistair, I., McIntyre, Alun E., Perry, Roger and Lester, John H., "Environmental Pollution" (Series B) 6, 1983, pp. 245-261.
- Cottereau, P., Effects of Sonic Boom from Aircraft on Wildlife and Animal Husbandry. Effects of Noise on Wildlife. J.L. Fletcher and R.G. Busnell eds. Academic Press. New York, NY., 1972.
- County of Orange vs. City of Irvine, San Diego Superior Court No. 710120.
- County of Orange, Airport System Master Plan, Technical Report 13, Ground Access Concepts Evaluation, March 18, 1999.
- County of Orange, EIR No. 564 James A. Musick, Branch Jail Expansion Project, etc. and Recirculated Section, October, 1998.
- County of Orange, Environmental Management Agency, "County Arterial Mix Data."
- County of Orange, Environmental Management Agency, Component II, Land Use Element Advance Planning Program, June 9, 1982.
- County of Orange, Environmental Management Agency, Environmental Analysis Division, North Clear Zone Parking Lot EIR No. 512 for John Wayne Airport, 1984.
- County of Orange, Final Bee Canyon Sanitary Landfill General Development Plan/Addendum to EIR 018 (IP87-026), June, 1988.
- County of Orange, Final Environmental Impact Report 508 and Environmental Impact Statement, John Wayne Airport Master Plan and Santa Ana Heights Land Use Compatibility Program, February 26, 1985.

- County of Orange, General Plan Elements.
  - A. Growth Management Element (October, 1993)
  - B. Noise Element (February, 1993)
  - C. Public Services and Facilities Element (October, 1988)
  - D. Housing Element (September, 1993)
  - E. Safety Element (August, 1987)
  - F. Resources Element (November, 1994)
  - G. Transportation Element (September, 1995)
  - H. Land Use Element (March, 1995)
  - I. Recreation Element (March, 1994)
  - J. Long-Range Planning Framework (November, 1985)
- County of Orange, Hydrology Report, John Wayne Airport Storm Channel Facility No. F01S01 Entire Drainage System, February, 1987.
- County of Orange, Integrated Waste Management Department, Landfill Bird Monitoring and Control Project, MBA Consultants, June 6, 1996.
- County of Orange, Jail Expansion and Operation, Relocation of Interim Care Facility, Southeast Sheriff's Station, November, 1996.
- County of Orange, John Wayne Airport Hydrology Report Airport Storm Channel Facility No. F01S01 Entire Drainage System, February, 1987.
- County of Orange, John Wayne Airport, "General Aviation Noise Ordinance," Article 3, Section 2-1-30, July 1, 1985.
- County of Orange, John Wayne Airport, "Phase 2 Commercial Airline Access Plan and Regulation," October 1, 1990 December 31, 2005.
- County of Orange, John Wayne Airport, 1997-1998 Annual Report for Stormwater Discharges Associated with Industrial Activities, June, 1998.
- County of Orange, John Wayne Airport, 1997-1998 Annual Report for Stormwater Discharges Associated with Industrial Activities, June, 1998.
- County of Orange, John Wayne Airport, State General Industrial Permit, Order No. 97-03-DWQ.
- County of Orange, Marine Corps Air Station El Toro Community Reuse Plan, Final Environmental Impact Report No. 563, December, 1996.
- County of Orange, Marine Corps Air Station El Toro Community Reuse Plan Final Environmental Impact Report No. 563, as supplemented. Final Supplemental Analysis, Appendix Q, November, 1999.
- County of Orange, MCAS El Toro Community Reuse Plan, December, 1996.
- County of Orange, MCAS El Toro Community Reuse Plan, Final EIR No. 563 Final Supplemental Analysis, November, 1999.

- County of Orange, MCAS El Toro Flight Demonstration, Final Report, July 19, 1999.
- County of Orange, MCAS El Toro Local Redevelopment Authority, Preliminary Non-Aviation Concepts Report El Toro Reuse Plan, 1996.
- County of Orange, MCAS El Toro Local Redevelopment Authority, Resolution 96-2, December, 1996.
- County of Orange, MCAS El Toro, Notice of Preparation of Draft, EIR No. 573, October 16, 1998.
- County of Orange, MCAS, El Toro Local Redevelopment Authority, Preliminary Concepts Report El Toro Reuse Plan, 1996.
- County of Orange, Natural Community Conservation Plan Implementation Agreement for the Central-Coastal Subregion adopted by the Board of Supervisors April 16, 1996.
- County of Orange, Newport Bay Watershed Urban Nutrient TMDL Technical Report, September, 1998.
- County of Orange, Newport Bay Watershed Urban Nutrient TMDL Technical Report, September.
- County of Orange, Orange County Hydrology Manual, October, 1986 and Addendum No. 1, January, 1996.
- County of Orange, Orange County Hydrology Manual, October, 1986.
- County of Orange, Personal communication between Bill Grieman, Project Manager, and Shambhavi Rao of P&D Consultants, July, 1999.
- County of Orange, Personal communication between Bill Melton and Shambhavi Rao of P&D Consultants, July 26, 1999.
- County of Orange, Personal communication between Bonnie Streater, John Wayne Airport, and Steve Allison, P&D. July 27, 1999.
- County of Orange, Personal communication between Richard Bailey and Shambhavi Rao of P&D Consultants, June 26, 1999.
- County of Orange, Personal communication between Romi Archer, Project Manager, Lower Peters Canyon Project and Shambhavi Rao of P&D Consultants, July, 1999.
- County of Orange, Preliminary Environmental Assessment for the disposal and development of the MCAS El Toro Air Station Properties Alton Parkway extension project, Marlene Brajdic, 1991.
- County of Orange, Regional Flood Control Master Plan for San Diego Creek, April, 1989.
- County of Orange, Regional Flood Control Master Plan for San Diego Creek, April, 1989.

- County of Orange, Santa Ana Heights Specific Plan, 1986, Revised 1991.
- County of Orange, Technical Report 17, Airport System Master Plan for John Wayne Airport and Proposed Orange County International Airport, December, 1999.
- County of Orange, Vegetation Community Map for the Habitat Reserve, 1995.
- County of Orange, Zoning Code, January, 1999.
- Culbertson, Adams and Associates, Final Program Environmental Impact Report, Portola Hills Planned Community, Fourth Revised Glenn Ranch Planned Community, December 17, 1986.
- Daniel, Mann, Johnson and Mendenhall (DMJM), Orange County (General Services Agency). Final Environmental Impact Report 102 for Orange County Airport Alternative Futures, 1978.
- David Evans and Associates, Inc., Draft Environmental Impact Report, General Plan Amendment 37594-GA, Zone Change 37595-ZC and Annexation No. 17 for MCAS El Toro and James A Musick Branch Jail, Technical Appendices, Volume 2, prepared for the City of Irvine, Community Development Department, March 31, 1999.
- Davis, D. J. and G. T. Christianson, Firefighter's Hazardous Materials Reference Book. Van Nostrand Reinhold, 1991.
- Department of Defense, Air Installation Compatible Use Zone (AICUZ) Program Study, 1972.
- Department of Defense, Base Reuse Implementation Manual, December, 1997.
- Department of the Air Force, Headquarters Air Force Safety Center, Accident Report Summary (Loma Ridge), July 10, 1998.
- Department of Transport, "Report of a Field Study of Aircraft Noise and Sleep Disturbance,"

  Department of Safety, Environment and Engineering Civil Aviation Authority, December 1992.
- Diesel Technology Forum, "Engineering Clean Air: The Continuous Improvement of Diesel Engine Emission Performance", March 2001.
- Dieselnet.com, "Mack Trucks Demonstrates Urea-SCR NOx Control System", April 5, 2000, http://www.dieselnet.com
- Doolittle, J.H., The Airport and Its Neighbors, President's Airport Commission, May, 1952.
- Dunholter, Paul H., and R.T. Harrison, "Military Aircraft Overflight and Wilderness Perceptions," Presented at the Acoustical Society of America Conference, 1988.
- Dunnet, G.M., Observations on the Effects of Low-Flying Aircraft at Seabird Colonies on the Coast of Aberdeenshire, Scotland, Biological Conservation, 12:55-63, 1977.

- East Orange General Plan, December, 1989.
- Economic and Planning Systems Inc. (EPS), July 1996.
- El Toro Reuse Planning Authority, Millennium MCAS El Toro Reuse Plan, Appendix, March, 1998.
- Ellis, D.H., Responses of Raptorial Birds to Low Level Military Jets and Sonic Booms, U.S. Fish and Wildlife Service, Institute for Raptor Studies, 59 pp., 1971.
- Environmental Laboratory, Corps of Engineers Wetland Delineation Manual Technical Report Y-87-1. U.S. Army Engineers Waterways Experiment Station, Vicksburg, MS., 1987.
- Espmark, Y., L. Falt and B. Falt, Behavioral Responses in Cattle and Sheep Responding to Sonic Booms and Low Altitude Subsonic Flight Noise. Veterinarian Record. 94:106-113, 1974.
- Ewbank, R., The Effects of Sonic Booms on Farm Animals. The Veterinary Annual, Issue 17. C.S.G. Gnensell and F.W.S. Hill, eds. Wright, Bristol, England, 1977.
- Fay, Richard R., Hearing in Vertebrates: A Psychophysics Databook, August, 1988.
- Federal Aviation Administration, Accident/Incident Data, PT-06-084, July, 1996.
- Federal Aviation Administration, Advisory Circular, Hazardous Wildlife Attractants On or Near Airports, AC No. 150/5200-33, May 1, 1997.
- Federal Aviation Administration, Airman's Information Manual, AIM 95, ASA 1995.
- Federal Aviation Administration, Airport Pavement Design and Evaluation, Advisory Circular AC15/5320-6D, July, 1995.
- Federal Aviation Administration, Bird Strikes Accident/Incident Data, July, 1996.
- Federal Aviation Administration, Bolt Beranek and Newman, "Community Attitude Survey: John Airport," 1981.
- Federal Aviation Administration, FAA Advisory Circular 150/5300-13, Airport Design. September 29, 1989.
- Federal Aviation Administration, Federal Air Regulation Part 36.
- Federal Aviation Administration, Federal Air Regulation Part 36. Noise data taken from Advisory Circular 36-3G.
- Federal Aviation Administration, Location of Commercial Aircraft Accidents/Incidents Relative to Runways, Office of Safety Oversight, July, 1990.

- Federal Aviation Administration, Review of Joint Use Potential, El Toro Marine Corps Air Station, Orange County, California, May, 1988.
- Federal Aviation Administration, Wildlife Strikes to Civil Aircraft in the United States 1991-1997, September, 1998.
- Federal Emergency Management Agency, et al. Handbook of Chemical Hazard Analysis Procedures, 1990.
- Federal Emergency Management Agency, Flood Insurance Rate Map No. 06059C0047E, September, 1989.
- Federal Emergency Management Agency, Flood Insurance Rate Map Nos. 06059C0047E and 06059C0049E, September, 1989.
- Federal Interagency Committee on Aircraft Noise (FICAN). (The full FICAN report can be found on the internet at www.fican.org.)
- Federal Interagency Committee on Noise (FICON), Federal Interagency Review of Selected Airport Noise Analysis Issues, August 21, 1992.
- Federal Register, Environmental Protection Agency, 40 CFR Parts 69, 80 and 86, Control of Air Pollution from New Motor Vehicles; Heavy-Duty Engine and Vehicle Standards and Highway Diesel Fuel Sulfur Control Requirements; Final Rule, January 18, 2001.
- Federal Register, Environmental Protection Agency, 40 CFR Parts 9, 86, and 89, Control of Emissions of Air Pollution from Nonroad Diesel Engines; Final Rule, October 23, 1998.
- Fidell, Sanford, and D.E. Bishop, "Prediction of Acoustic Detectability," Technical Report 11949, Bolt, Beranek and Newman, 1974.
- Fidell, Sanford, and S. Teffeteller, "The Relationship Between Annoyance and Detectability of Low Level Sounds," Report 3699, Bolt, Beranek and Newman, April 1978.
- Fidell, Sanford, and S.R. Teffeteller, "Scaling the Annoyance of Intrusive Sounds," Journal of Sound and Vibration, Volume 78, pp. 291-298, 1981.
- Fidell, Sanford, K.S. Pearsons, and R.L. Bennett, "Predicting Aural Detectability of Aircraft in Noise Backgrounds," Report 2202, Bolt, Beranek and Newman, 1972.
- Fields, James M., Federal Aviation Administration and NASA Langley Research Center, "Effect of Personal and Situational Variables on Noise Annoyance: With Special Reference to Implications for En Route Noise," DOT/FAA/EE-92/03, August 1992.
- Fletcher, J.L. and T.G. Busnell, eds., Effects of Noise on Wildlife. Academic Press. New York, NY., 1978.
- Foothill/Eastern Transportation Corridor Agency, Design/Build Services for the Eastern Transportation Corridor Marshburn Channel Drainage Study, August, 1994.

- GEOFON, Inc., Summary Status Report Site Contamination Assessment and Remedial Program, Former Martin Aviation Facility and Michelson Bridge, Part Three Circulation Roadway Dewatering Area, John Wayne Airport, Orange County, March 5, 1990.
- Governor's Office of Emergency Services (OES). 1989. Guidance for the Preparation of a Risk Management and Prevention Program. State of California. November.
- Governor's Office of Emergency Services, Guidance for the Preparation of a Risk Management and Prevention Program. State of California, 1989.
- Green, D.M., and J. A. Swets, "Signal Detection Theory and Psychophysics," John Wiley and Sons, New York, 1966.
- Greenwood and Associates, Cultural Resource Investigation for the Easter Transportation Corridor, 1989.
- Greenwood and Associates, TCA EIR/EIS 2 for the Eastern Transportation Corridor, Historic Property Survey Report, 1991
- Griffin, Charles E., Alternative Airport Concept, Comments in Response to Final EIR No. 563 Draft Supplemental Analysis, April 7, 1999.
- Guidelines for Preparation of Environmental Assessments for Solid Waste Impacts (Ventura County Solid Waste Management Department)
- Harris, Cyril M., "Handbook of Noise Control," Second Edition, McGraw-Hill Book Co., 1979, including Chapter Authored by Molino.
- Henningson, Durham and Richardson (HDR), Effects of Minuteman Launch on California Least Tern Nesting Activity, Vandenberg Air Force Base, California, Technical Report, 1981.
- Hickman, J.C., Editor, The Jepson Manual Higher Plants of California, University of California Press, Berkeley, 1993.
- Holland, R.F., Preliminary Descriptions of the Terrestrial Natural Communities of California, Non-Game Heritage Program, State of California Department of Fish and Game, Sacramento, California, 1986.
- Holmes & Narver (H&N) Memorandum from Andy Muth (H&N) to Steve Allison (P&D Aviation), July 22, 1999.
- Holmes & Narver (H&N). 1999. Memorandum from Andy Muth (H&N) to Doug Sachman (P&D Aviation). H&N Transmittal No. DT-PD015. June 29.
- Institute of Transportation Engineers, Washington D.C., Trip Generation, Vols. 1-3, 6<sup>th</sup> Edition, 1997.

- INTERRA, Report to the Orange County Regional Airport Authority, Development Feasibility Analysis: El Toro Non-Aviation Reuse Alternative Millennium Plan, October, 1998.
- Irvine Ranch Water District, Procedural Guidelines and General Design Requirements, April, 1991.
- Irvine Ranch Water District, Your Water Service, Water Resources Master Plan, April, 1999, http://www.irwd.com/WaterService/MasterPlan.html, accessed September 20, 1999.
- Irvine Unified School District, Northwood High School Final Environmental Impact Report, State Clearing House #: 95051021, October 31, 1995.
- Irvine Unified School District, Notice of Determination, Negative Declaration, North Irvine Middle School State Clearinghouse Number 92031061, District Project Number 22/42014, Signed Corinne Loskot, Facilities Planning Coordinator, April 23, 1992.
- Irvine Unified School District, Personal communication between Shirley Cairns and Shambhavi Rao of P&D Consultants, July 28, 1999.
- Irvine Unified School District, Planning Area 27 (bounded roughly by Bonita Canyon Drive & the San Joaquin Hills Tollroad) Elementary School. www.isud.k12.ca.us/schools/pa27/, July 30, 1999.
- Jehl, J.R., Jr. and C.F. Cooper (eds.), Potential Impact of Space Shuttle Sonic Booms on the Biota and Geology of the California Channel Islands: Research Reports, Technical Report 80-1, San Diego State University Center for Marine Studies, 1980.
- John Wayne Airport, Accident/Incident Data; Aviation Data Systems Branch/AFS-620, March 30, 1999.
- John Wayne Airport, Groundwater Sampling Results, Southwest Parking Structure, Letter to Christopher Caliendo, July 26, 1996.
- John Wayne Airport, Information on Passengers and Operations, John Wayne Airport, Costa Mesa, March 26, 1999; National Transportation Safety Board: Brief of Accidents at Orange County/John Wayne Airport, 1964-1997; Federal Aviation Administration: Aviation Data Systems Branch/AFS-620; John Wayne Airport Accident/Incident Data; March 30, 1999.
- John Wayne Airport, Information on Passengers and Operations, March 26, 1999.
- John Wayne Airport, Operations, Bird/Other Wildlife Strike Report, April, 1999.
- Jones, J.K., D.C. Carter, H.H. Genoway, R.S. Hoffman and D.W. Rice, Revised Checklist of North American Mammals North of Mexico, Occasional Papers, Museum of Texas Technical University, No. 80, 1982.

- Joyner, W.B. & Furnal, T.E., "Predictive Mapping of Earthquake Ground Motion," U.S. Geological Survey, 1360, 1985.
- JRP Historical Consulting Services, Inventory and Evaluation of National Register of Historic Places Eligibility for Buildings and Structures at Marine Corps Air Station (MCAS) El Toro, Santa Ana, Orange County, California. JRP Historical Consulting Services, Davis, California, 1998.
- K&M Consultants, "Instrument Flight Procedures Analysis," May 1998.
- K&M Consultants, MCAS El Toro Master Development Program, Airport System Master Plan, Instrument Flight Procedures Analysis, May, 1998.
- KEA Environmental, Biological Resources Report, MCAS El Toro Habitat Preserve. National Facilities Engineering Command, Southwest Division, 1998.
- Kushlan, J.A. Effects of Helicopter Censuses on Wading Bird Colonies, Journal of Wildlife Management, 43(3):756-760, 1979.
- LAACO, Personal communication between John Hathaway, Vice President, and Susan Roberts of P&D Consultants, July 28, 1999.
- Lercher P, Stansfield S. A., Thompson S.J., Non Auditory Health Effects of Noise; Review of the 1993-1998 Period, Noise Effects-98 Conference Proceedings, p. 213. 1998.
- Lewis, R. J., Hazardous Chemicals Desk Reference. Second Edition. Van Nostrand Reinhold, 1991.
- Malcolm Lewis Associates, MCAS El Toro Utilities Technical Study, 1987.
- Mammoth Equities, Personal communication between Tucker Lewis, Project Manager, and Susan Roberts of P&D Consultants, July 28, 1999.
- Massachusetts Port Authority, Soot Deposition Study: Logan Airport and Surrounding Communities, January, 1997.
- MBA Consultants, Draft Orange County Landfill Bird Monitoring and Control Project, June, 1996.
- MCAS El Toro, Base Realignment and Closure Office, El Toro Bird Strike History, April, 1999.
- MCAS El Toro, Base Realignment and Closure Office, El Toro History of Aircraft Accidents, April, 1999.
- MCAS El Toro, Draft Final Phase II Feasibility Study Report, Site 17, September, 1997.
- MCAS El Toro, Draft Final Phase II Remedial Investigation Site 2, Magazine Road Landfill, April 30, 1997.

- MCAS El Toro, Draft Final Phase II Remedial Investigation Report, Operable Unit 2A Site 24, Volume I, March, 1997.
- MCAS El Toro, Draft Final Phase II Remedial Investigation Report, Operable Unit-3A Sites, March, 1997.
- MCAS El Toro, Draft Final Phase II Remedial Investigation Report, Site 17, April, 1997.
- MCAS El Toro, Draft Final Record of Decision, Operable Units 2A and 3A, No Action Sites, September, 1997.
- MCAS El Toro, Draft Final Soil Vapor Extraction System Design Work Plan, Site 24, May, 1998.
- MCAS El Toro, Final Work Plan Phase II Remedial Investigation/ Feasibility Study, July, 1995.
- MCAS El Toro, OU-1 Ground Water Report, 1997.
- MCAS/Department of the Navy, Draft Final Interim Record of Decision, Operable Unit 2A, Site 24 -VOC Source Area Vadose Zone, September, 1997.
- MECA, Manufactures of Emissions Control Association, "Clean Air Facts, Emission Control Retrofit of Existing Diesel Engines," http://www.meca.org
- MECA, Manufactures of Emissions Control Association, "Recent Developments in Integrated Exhaust Emission Control Technologies Including Retrofit of Off-Road Diesel Vehicles," February 3, 2000, http://www.meca.org
- Memphis State University, Effects of Noise on Wildlife and Other Animals. U.S. Environmental Protection Agency, Office of Noise Abatement and Control. Washington, D.C., 1971.
- Michael Brandman and Associates, Inc., Background Report for John Wayne Airport, Regional Water Quality Control Board Cleanup and Abatement Order No. 88-123, December 1, 1988.
- Michael Brandman Associates Inc., Foothill Circulation Phasing Plan Backbone One EIR 472. February, 1988.
- Michael Brandman Associates, Inc., Aircraft Hazard and Fuel Tank Farm Risk Assessment, August, 1989.
- Miller, Marc. Micropath Corporation, 2023 Montane Drive East, Golden, CO 80401-9123.
- Mist Systems Company, TrueFog Dust Suppression System, http://www.truefog.com/dustsuppress.html
- Morton, P. K., Environmental Geology of Orange County, CDMG OFR 79-8 LA, 1979.
- Munz, P.A., A Flora of Southern California, University of California Press, Berkeley, 1974.

- National Academy of Sciences/National Research Council, An Annotated Bibliography of Animal Response to Sonic Booms and Other Loud Noises, 1970.
- National Association of Noise Control Officials, "Noise Effects Handbook," New York, 1981.
- National Audubon Society, The Blue List for 1986 (North American birds species of concern). American Birds. Volume 40, No. 2: 227-236, 1986.
- National Business Aircraft Association, NBBA Business Aviation Fact Book, Web site: www.nbaa.org/industrydata. 1999.
- National Climatic Data Center, Climatological Data Annual Summary.
- National Transportation Safety Board, Aviation Accident Statistics, Table 10, 1982-1998, U.S. General Aviation, [Internet, WWW] ADDRESS: http://www.ntsb.gov/aviation/aviationhtml, March 30, 1999.
- National Transportation Safety Board, 14CFR121, Scheduled Service, 1982-1998 http://www.ntsb.gov/aviation/aviationhtml, March 30.
- National Transportation Safety Board, 14CFR135, Scheduled Service. 1982-1998, (Commuter Air Carriers) http://www.ntsb.gov/aviation/aviationhtml, March 30.
- National Transportation Safety Board, Aviation Accident Statistics, Table 5, 1982-1998, U.S. Carrier Operating Under 14CFR121, Scheduled and Non-Scheduled Service, [Internet, WWW] ADDRESS: http://www.ntsb.gov/aviation/aviationhtml, March 30, 1999.
- National Transportation Safety Board, Aviation Accident Statistics, Table 8, 1982-1998, U.S. Carrier Operating Under 14CFR135, Scheduled Service (Commuter Air Carriers), [Internet, WWW] ADDRESS: http://www.ntsb.gov/aviation/aviationhtml, March 30, 1999.
- National Transportation Safety Board, Brief of Accidents at Orange County/John Wayne Airport 1964-1998, 1998.
- Naval Facilities Command, Air Installations Compatable Use Zones Study (AICUZ), March, 1981.
- Naval Facilities Engineering Command, Southwest Division, Draft Historical Radiological Assessment (HRA) MCAS El Toro, May 1999.
- Naval Facilities Engineering Command, Western Division, San Bruno, CA, MCAS El Toro Master Plan Prepared By Gillespie, DeLorenzo, ASLA and Associates, Inc., 1991.
- NESCAUM, Status Report on NOx Controls for Gas Turbines, Cement Kilns, Industrial Boilers, and Internal Combustion Engines; Technology & Cost Effectiveness. http://www.nescaum.org/

- New Millennium Group, The, and Charles E. Griffin, October 5, 1999, The Alternative Airport Runway Layout Long V and Short V FAA TERPS Analysis Feasibility Study.
- Noll, Roger and Andrew Zimbalist, Sports, Jobs, and Taxes, The Economic Impact of Sports Teams and Stadiums, 1997.
- Non-road Compression Engine Statement of Principles, www.arb.ca.gov/regact/ciengine/ciengine.htm
- OCAA, Aviation Technical Report 1, Airport System Feasibility, February 27, 1998.
- OCAA, Aviation Technical Report 10, Airfield and Airspace Evaluation, November 5, 1999.
- OCAA, Aviation Technical Report 11, Traffic Analysis, October 22, 1999 (Appendix D in EIR No. 573).
- OCAA, Aviation Technical Report 12, Passenger Terminal Facility Requirements and Concepts Evaluation, November 5, 1999.
- OCAA, Aviation Technical Report 13, Ground Access Concepts Evaluation, March 18, 1999.
- OCAA, Aviation Technical Report 14, Concepts Evaluation of Other Airport Facilities, November 5, 1999.
- OCAA, Aviation Technical Report 16, Economic Benefits Study, November 5, 1999.
- OCAA, Aviation Technical Report 17, Airport System Master Plan for John Wayne Airport and Proposed Orange County International Airport, December, 1999.
- OCAA, Aviation Technical Report 19, Financial Analysis, November 5, 1999
- OCAA, Aviation Technical Report 2, Planning and Performance Parameters, February 27, 1998.
- OCAA, Aviation Technical Report 3, Existing Facilities, January 22, 1998.
- OCAA, Aviation Technical Report 4, Aviation Demand Forecasts, April 6, 1998.
- OCAA, Aviation Technical Report 5, Facility Requirements, November 5, 1999.
- OCAA, Aviation Technical Report 6, Alternatives Definition Report (2 volumes), October 15, 1999.
- OCAA, Aviation Technical Report 7, Central (Orange) County Traffic Model Version 2.8, October 22, 1999.
- OCAA, Aviation Technical Report 9, Noise Analysis, November 1999 (Appendix E in EIR No. 573).
- OCAA, Aviation Working Paper 1, Airport System Issues, January 20, 1998.

- OCAA, Aviation Working Paper 10, Traffic Mitigation Feasibility and Cost Estimates, October 22, 1999.
- OCAA, Aviation Working Paper 2, List of Preliminary Project Planning Issues, January 20, 1998.
- OCAA, Aviation Working Paper 3, JWA/OCA Simulation Assumptions, January 20, 1998.
- OCAA, Aviation Working Paper 4, Financial Issues for a Two-Airport System, November 5, 1999.
- OCAA, Aviation Working Paper 5, JWA/OCA SIMMOD Model Calibration, January 20, 1998.
- OCAA, Other Aviation Reports, Airport Plans, November 5, 1999.
- OCAA, Other Aviation Reports, Ground Access Traffic and Trip Generation, April 15, 1999.
- OCAA, Other Aviation Reports, Instrument Flight Procedures Analysis, May, 1998.
- OCAA, Other Aviation Reports, Jeppesen Analyses (Part 121), April, 1998.
- OCAA, Other Aviation Reports, MCAS El Toro Flight Demonstration, July 19, 1999.
- Office of Safety Oversight, Location of Commercial Aircraft Accidents/Incidents Relative to Runways, Report No. DOT/FAA AOV 90-1, July. 1990.
- Orange County Transportation Authority, HOV Improvements at Route 405/Route 55
  Interchange and HOV Direct Access Drop Ramps at Bear Street and Von Karman
  Avenue, August 31, 1993.
- Orange County Transportation Authority, Master Plan of Arterial Highways.
- Orange County Transportation Authority, Orange County Congestion Management Program, 1992, as amended.
- Orange County Transportation Authority, Orange County Traffic Analysis Model (OCTAM) 2.8.
- Orange County Water District, Groundwater Contour Map of Irvine Sub-Basin, November, 1997.
- P&D Consultants, Inc., Information developed for this Draft EIR based on historical data from National Transportation Safety Board (1982-1998); Federal Aviation Administration (1964-1998); John Wayne Airport (1964-1998); and projected aircraft operations, 1999.
- Phillips Brandt Reddick, Environmental Impact Report Foothill Ranch Project (EIR No. 481).

  April, 1988.
- PennzSuppress D soil stabilizer http://www.PennzSuppress.com/index.htm

- PRC Speas Associates, Air Installations Compatible (AICUZ) Use Zones Study, MCAS El Toro, prepared for Western Division, Naval Facilities Engineering Command, March, 1981.
- R.J. Meade Consulting, Inc. and William M. Boyd, Central and Coastal Subregion Natural Community Conservation Plan/Habitat Conservation Plan Joint Programmatic Environmental Impact Report No. 553 and Environmental Impact Statement, Part 3. Prepared for the County of Orange Environmental Management Agency, the United States Fish and Wildlife Service and the California Department of Fish and Game. December 5, 1995.
- R.J. Meade Consulting, Inc., Final Central and Coastal Subregion Natural Community
  Conservation Plan/Habitat Conservation Plan, Parts 1 and 2. Prepared for the County of
  Orange Environmental Management Agency, the United States Fish and Wildlife Service
  and the California Department of Fish and Game. July 17, 1996.
- Remsen, J.V., Jr., Bird Species of Special Concern in California, An annotated list of declining or vulnerable bird species. California Department of Fish and Game. Federal Aid in Wildlife Restoration, Project PR W54-R-9. Report No. 78, 1978.
- Review of Joint Use Potential, El Toro Marine Corps Air Station, Orange County, California, May, 1988.
- Robert Kahn, John Kain & Associates, Inc., Update to the April 13, 1994 North Mission Viejo Traffic Study. April 30, 1996.
- San Diego Association of Governments, San Diego Traffic Generators, July, 1998.
- Shaw, J.H. and Suppe, J., Earthquake Hazards of Active Blind-Thrust Faults Under the Central Los Angeles Basin, California. Journal of Geophysical Research, Vol.101, No. B4, pages 8623-8642, 1996.
- Silverado Constructors, Final Design Report Marshburn Retarding Basin, August, 1997.
- Silverado Constructors, Final Design Report Marshburn Retarding Basin, August, 1997.
- Skinner, M. and B. Pavlik, Inventory of Rare and Endangered Vascular Plants of California. California Native Plant Society, Sacramento. Fifth Edition, 1994.
- South Coast Air Quality Management District, Air Quality Data, 1993 to 1997, 1996 to 2000.
- South Coast Air Quality Management District, Air Quality Management Plan, 1994.
- South Coast Air Quality Management District, Air Quality Management Plan, 1997.
- South Coast Air Quality Management District, CEQA Air Quality Handbook, 1993.

- South Coast Air Quality Management District, Draft EIR for the 1997 Air Quality Management Plan, August, 1996.
- South Coast Air Quality Management District, Final Environmental Impact Report for the 1994 Air Quality Management Plan, August, 1994.
- South Coast Air Quality Management District, Guide to Agricultural PM10 Dust Control Practices, June, 1999.
- South Coast Air Quality Management District, Management Plans (1994, 1997)
- South Coast Air Quality Management District, Multiple Air Toxics Exposure Study in the South Coast Air Basin (MATES-II). March, 2000.
- South Coast Air Quality Management District, Newport Beach Fallout Study, JWA, January 15, 1993.
- South Coast Air Quality Management District, Rule 403 Agricultural Handbook, Measures to Reduce Dust from Agricultural Operations at the South Coast Air Basin, December, 1998.
- South Coast Air Quality Management District, Rules 402 and 403.
- Southern California Association of Governments, Regional Transportation Plan, 1998, 2001.
- Sports, Jobs and Taxes, The Economic Impact of Sports Teams and Stadiums, Roger G. Noll and Andrew Zimbalist, Editors, 1997, Brookings Institution Press.
- State of California Resources Agency, Sacramento, Endangered, Rare and Threatened Animals of California, 1986.
- State of California Resources Agency, Special Animals of California, Sacramento, 1998.
- State of California Resources Agency, State and Federal Endangered and Threatened Animals of California and Listing Dates, Sacramento, 1994.
- State of California, "California Airport Noise Regulations," Chapter 6, California Administrative Code, 1970.
- State of California, Code of Regulations (CCR), Part 2, Title 24, CCR, 1974.
- State of California, Department of Transportation, Traffic Volumes on the California State Highway System, Sacramento, California, June, 1997.
- State of California, Division of Mines and Geology, The Official Map of Seismic Hazard Zones for the Tustin Quadrangle, released April 15, 1998.

- State of California, Division of Traffic Operations of the Business, Transportation and Housing Agency, Department of Transportation, "1993 Annual Average Daily Truck Traffic on the California State Highway System," November 1994.
- State of California, Government Code Section 65302(f) and Section 46050.1 of the Health and Safety Code.
- State of California, Office of Emergency Services, Guidance for Conducting A Risk Management Plan, 1989.
- State of California, Section 21675, Public Utilities Code.
- Stebbins, R.C., A Field Guide to Western Reptiles and Amphibians, Houghton Mifflin Company, Boston, Massachusetts, 1985.
- Stevenson Ranch DEIR (Phase IV), Los Angeles County
- SuvaPro, Filter List, Verified Particulate Trap Systems for Diesel Engines to implement the Suva trap imperative and the Swiss Clean Air Act, LRV 1998, October 1, 2000.
- SWDIVNAVFACENGCOM, Personal Communication with Anthony DeJulio of P&D, August, 1999.
- T.W. Brown Oil Company (Brown Oil). 1999. Material Safety Data Sheet for Kerosene (Jet-A). MSDS No. U8012. January 31.
- The Keith Companies, Environmental Impact Report for One Hutton Centre, Prepared for the City of Santa Ana, SCH# 91051045, 1992.
- The Planning Center, Agricultural Lands Concept Plan, November, 1999.
- The Planning Center, Airport Site Consensus Team Final Report, 1990.
- The Planning Center, Business Park Design Guidelines, November, 1999.
- The Planning Center, Environmental Impact Report MacArthur Place, Prepared for the City of Santa Ana, SCH# 87042909, August, 1987.
- The Planning Center, Open Space Plan, A Celebration of Orange County's History, November, 1999.
- Tinsley, J. C., Youd, T. L., Perkins, D. M., and Chen, A.T.F., Evaluating Liquefaction Potential, USGS Professional Paper 1360, 1985.
- Transportation Corridor Agencies, Personal communication between Michelle Hart and Shambhavi Rao of P&D Consultants, July 28, 1999.
- Transportation Research Board, Washington D.C., Highway Capacity Manual, Special Report 209, December, 1997.

- Underwood, Jackson and Andrew York, Draft Archeological Testing and Evaluation at Site CA-ORA-1462, MCAS El Toro, Orange County, California. KEA Environmental, Inc. Ms. on file, Department of the Navy, Southwest Division. Naval Facilities Engineering Command. Planning and Real Estate Department, 1220 Pacific Highway, San Diego, California, 92132-5180, 1998.
- United States Army Corps of Engineers, Draft Training Package Wetland Delineator Certification Program for Wetland Delineator Certification Program Headquarters, U.S. Army Corps of Engineers Regulatory Branch, Washington, DC. Prepared by Wetland Research and Technology Center Environmental Laboratory, EP-W, U.S. Army Engineer Waterways Experiment Station, Vicksburg, MS, 1993.
- United States Army Corps of Engineers, Flood Plain Information Tributaries of the Upper San Diego Creek, 1974.
- United States Army Corps of Engineers, Flood Plain Information, Tributaries of the Upper San Diego Creek, 1974.
- United States Army Corps of Engineers, HEC-RAS, River Analysis System, Users Manual Version 2.0, April, 1997.
- United States Army Corps of Engineers, HEC-RAS, River Analysis System, Users Manual Version 2.0, April, 1997.
- United States Army Corps of Engineers, Wetland Delineation Manual, Environmental Laboratory, 1987.
- United States Coast Guard, CHRIS Hazardous Chemical Data. U.S. Department of Transportation, 1978.
- United States Department of Agriculture, Natural Resource Conservation Service. Field Indicators of Hydric Soils in the United States. G.W. Hurt, Whited, P.M., and Pringle, R.F. (eds.). USDA, NRCS, Forth Worth, Texas. 27 pages, 1996.
- United States Department of Agriculture, Soil Conservation Service and Forest Service, Soil Survey of Orange and Western Part of Riverside Counties, June, 1976.
- United States Department of Agriculture, Soil Conservation Service. Hydric Soils of the United States. Soil Conservation Service. In Cooperation with the National Technical Committee for Hydric Soils, Washington, DC., 1991.
- United States Department of Agriculture, Soil Conservation Service and Forest Service. Soil Survey of Orange and Western Part of Riverside Counties, June, 1976.
- United States Department of Agriculture, Soil Conservation Service. Hydric Soil Lists [Orange County]. Riverside, California Field Office. March, 1992.

- United States Department of Commerce, National Oceanic and Atmospheric Administration, National Environmental Satellite, Data, and Information Service, "Windrose Tabulations Monthly-Annual All Weather 93101 El Toro, CA, January 1984 to December 1993," National Climatic Data Center.
- United States Department of Energy, Aircraft Crash Risk Analysis Methodology, September, 1996.
- United States Department of Energy, Aircraft Crash Risk Analysis Methodology, Tables 4.5 and 4.6, September, 1996.
- United States Department of Health and Human Services, NIOSH Pocket Guide to Chemical Hazards. June, 1990.
- United States Department of Navy, Base Realignment and Closure Office. Equivalent Average Day Operations, fax received from MCAS El Toro, April, 1999.
- United States Department of the Interior, Endangered and Threatened Species, Plant and Animal Taxa; Proposed Rule. Federal Register 61(40): 7595-7613, 1996.
- United States Department of the Interior, Endangered and Threatened Wildlife and Plants. Federal Register 50 CFR 17.11 and 17.12., 1994.
- United States Department of Transportation, Federal Aviation Administration, "Integrated Noise Model (INM) Version 5.0 User's Guide," August 1995.
- United States Department of Transportation, Federal Aviation Administration, Wildlife Strikes to Civil Aircraft in the United States, 1991-1997, September, 1998.
- United States Department of Transportation, Office of Pipeline Safety (OPS), Hazardous Liquid Pipeline Accident Summaries by Commodity (1994-1998). OPS Website ops.dot.gov., 1999.
- United States Environmental Protection Agency, "Information on Levels of Environmental Noise Requisite to Protect Public Health and Welfare with an Adequate Margin of Safety," U.S. Environmental Protection Agency, Office of Noise Abatement and Control, March 1974.
- United States Environmental Protection Agency, (EPA-454/R-92-005): Guideline for Modeling Carbon Monoxide from Roadway Intersections, November, 1992.
- United States Environmental Protection Agency, 1998a. Compilation of Air Pollution Emission Factors. Volume I: Stationary Point and Area Sources AP-42, 5th Edition and Supplements, U.S. Environmental Protection Agency, Office of Air Quality Planning and Standards, Environmental Sciences Research Laboratory, Research Triangle Park, NC.
- United States Environmental Protection Agency, CAL3QHC CO Hot-Spot Analysis Model.
- United States Environmental Protection Agency, California Federal Implementation Plan (1994).

- United States Environmental Protection Agency, Clean Water Act, Section 208 Guidelines.
- United States Environmental Protection Agency, Clean Water Act, Section 303(d), 1998.
- United States Environmental Protection Agency, Clean Water Act, Section 401, Water Quality Certification.
- United States Environmental Protection Agency, Clean Water Act, Section 402, National Pollutant Discharge System.
- United States Environmental Protection Agency, Documentation for the 1996 Base Year National Toxics Inventory for Aircraft Sources, April, 1999.
- United States Environmental Protection Agency, et al., Technical Guidance for Hazards Analysis. December, 1987.
- United States Environmental Protection Agency, Non-road Engine and Vehicle Emission Study, 1991.
- United States Environmental Protection Agency, Notice of Proposed Rulemaking for Federal Implementation Plans for Southern California, Docket No. A-94-09.
- United States Environmental Protection Agency, Procedures for Emissions Inventory Preparation, Volume IV: Mobile Sources, 1992.
- United States Environmental Protection Agency, Technical Support for Development of Airport Ground Support Equipment Emissions, May, 1999.
- United States Federal Aviation Administration and the U.S. Air Force, 1997b. Emissions and Dispersion Modeling System (EDMS) Reference Manual, FAA-AEE-97-01, FAA Office of Environment and Energy (AEE-120), Washington, DC and USAF Armstrong Laboratory, Tyndall Air Force Base, FL.
- United States Federal Aviation Administration, 1985. FAA Aircraft Engine Emission Database (FAEED), FAA Office of Environment and Energy (AEF-110), Washington, DC.
- United States Federal Aviation Administration, Emissions Dispersion Modeling System (EDMS) Version 4.0. 2001.
- United States Federal Aviation Administration, Guidelines for Modeling Carbon Monoxide From Roadway Intersections, 1992.
- United States Federal Aviation Administration, Integrated Noise Model (INM).
- United States Federal Highway Administration, "FHWA Highway Traffic Noise Prediction Model," FHWA-RD-77-108, December, 1978.

- United States Fish and Wildlife Service (USFWS), Memorandum from U.S. Fish and Wildlife Service, Area Manager to Civil Aeronautics Board, Mr. Mark Atwood regarding Endangered Species Consultation John Wayne Airport, Orange County, California, SESO, No. 1-1-81-F-112, July 8, 1981.
- United States Fish and Wildlife Service, Carlsbad Field Office, a Biological Inventory of Marine Corps Air Station, El Toro, 1993.
- United States Fish and Wildlife Service, Memorandum from USFWS. Area Manager to Civil Aeronautics Board. Mark Atwood regarding Endangered Species Consultation, John Wayne Airport. Orange County, CA. SESO. No. 1-1-81-F-112, 1981.
- United States Fish and Wildlife Service, National List of Vascular Plant Species that Occur in Wetlands: 1996 Summary. Ecology Section, National Wetlands Inventory, USFWS. 209 pages, March 3, 1997.
- United States Marine Corps, Base Cleanup Plan (BCP) for Marine Corps Air Station, El Toro, CA, March, 1999.
- United States Marine Corps, Base Realignment and Closure Cleanup Plan, MCAS El Toro, March, 1999.
- United States Marine Corps, Base Realignment and Closure Office, Marine Corps Air Base, Western Area. Personal Correspondence from E.T. Gilhooley, March 25, 1999.
- United States Marine Corps, Navy Public Works Center, San Diego, 1995.
- United States Securities and Exchange Commission EDGAR System.
- Unites States Department of Transportation, Federal Aviation Administration, Wildlife Strikes to Civil Aircraft in the United States, 1991-1997, September, 1998.
- Welch, B.L. and A.S. Welch, eds., Physiological Effects of Noise. Plenum Press. New York, NY., 1970.
- Whitefield and Hagen study (Estimate of Particle Emission Indices as a Function of Particle Size for the LTO Cycles for Commercial Jet Engines, Philip D. Whitefield and Donald E. Hagan, University of Missouri-Rolla).

